

# NORTH -EAST GAS GRID PIPELINE PROJECT

# **BID DOCUMENT FOR**

# SUPPLY OF ORIFICE METER BASED METERING SKIDS ON ANNUAL RATE CONTRACT 2(TWO) YEARS

# **OPEN DOMESTIC COMPETITIVE BIDDING**

# Tender Ref. No.: 05/51/23VC/IGGL/030-FL

# **VOLUME – II OF II**



# PREPARED AND ISSUED BY

MECON LIMITED (A Govt. of India Undertaking) Delhi, India

[In case of any conflict in terms & conditions given in documents uploaded on GEM portal (Vol. I & Vol. II) and provisions/ conditions available at GeM portal (including General terms and conditions (GTC)), the terms & conditions contained in documents uploaded in GeM portal (Vol. I & Vol. II) shall prevail]

# **INDEX**

- A) MATERIAL REQUISITION
- B) VENDOR DATA REQUIREMENT
- C) TECHNICAL QUESTIONNAIRE
- D) VENDOR DRAWING/ DATA APPROVAL PROCEDURE
- E) CHECK LIST (TECHNICAL)

```
Annexure – A
Annexure – B
```

- F) JOB SPECIFICATION (REF.: Spec. No. MEC/05/21/E5/JS-030)
  - a) TECHNICAL SPECIFICATION FOR ORIFICE METER (TS No.: MEC / 05 / E5/ TS / ORFM)
  - b) TECHNICAL SPECIFICATION OF CONTROL PANEL (REF.: TS No.-MEC/05/E5/TS/CP -030)
  - c) TECHNICAL SPECIFICATION OF GAS CHROMATOGRAPHS (REF.: TS No. MEC/ 05/E5/ TS/ GC - 030)
  - d) PIPING MATERIAL SPECIFICATION

# ANNEXURE

Annexure-I -	DATASHEETS
Annexure-II -	GAS COMPOSITION PARAMETERS
Annexure-III -	P & ID
Annexure – IV -	INSTALLATION DRAWINGS
Annexure – V –	APPROVED VENDOR LIST
Annexure -VI -	QAPs
Annexure - VII -	FAT PROCEDURE/ FAT MANUAL
Annexure -VIII -	STANDARD SPECIFICATIONS

# **MATERIAL REQUISITION**

PROJECT	:	NORTH EAST GAS GRID (PHASE – 1 & 2 P/L SECTION)
ITEM	:	METERING SKID
MR NO.	:	MEC/S/05/E5/23VC/030
CLIENT	:	M/s INDRADHANUSH GAS GRID LIMITED

Item No.	Description	Unit	Qty	Remarks
Under 2	Year ARC			
A	Design, Engineering, Manufacturing, Testing, Nameplate marking, Painting, Inspection, Packaging, Forwarding, Transportation, Transit Insurances, Shipment, Unloading of skid at site, Calibration and Supply of Metering Skid which comprise of Filtration (1W+1S) & Metering (1W+1S) through Single Chamber Orifice Meter with field mounted flowcomputer (alongwith solar panel, batteries, charge controller, JB, etc.)(for item no. A.1 only), panel mounted Flow Computer including Metering Panel along with the accessories (barriers / isolators, SDC, Terminal block duly mounted, wired and fully assembled) (for A.2 to A.7), Gas Detection System (for item nos. A.2 to A.7), Scrubber (for item nos. A.1 to A.6, A.8 & A.9), Gas Chromatograph (for item nos. A.4 to A.7), Water Dew Point Analyzer (for item nos. A.4 to A.7), Consumables, Compulsory & Commissioning Spares, Special Tools Tackles including Assistance in Configuration, Interfacing, Integrated Testing & Commissioning as per Job specifications and Special Instructions enclosed.			
A.1	CHECK METERING FACILITY WITH SCRUBBER AT ONGC KSAB WHI SOURCE JORHAT (Ref P & ID No.: MEC/23VC/05/28/M/001/FEEDER-1001)	Nos.	01	
A.2	CHECK METERING FACILITY WITH SCRUBBER AT ONGC SUAB WHI SOURCE JORHAT (Ref P & ID No.: MEC/23VC/05/28/M/001/FEEDER-1002)	No.	01	
A.3	CHECK METERING FACILITY WITH SCRUBBER AT ONGC BABEJIA WHI SOURCE JORHAT (Ref P & ID No.: MEC/23VC/05/28/M/001/FEEDER-1002)	No.	01	
A.4	CHECK METERING FACILITY WITH SCRUBBER, GAS CHROMATOGRAPH, WATER DEW POINT ANALYZER, HYDROCARBON DEW POINT ANALYZER AT ONGC KORAGHAT WHI SOURCE JORHAT (Ref P & ID No.: MEC/23VC/05/28/M/001/FEEDER-1003)	No.	01	
A.5	CHECK METERING FACILITY WITH SCRUBBER, GAS CHROMATOGRAPH, WATER DEW POINT ANALYZER, HYDROCARBON DEW POINT ANALYZER AT ONGC TULAMURA (Ref P & ID No.: MEC/23VC/05/28/M/001/FEEDER-1004)	No.	01	
A.6	CHECK METERING FACILITY WITH SCRUBBER, GAS CHROMATOGRAPH, WATER DEW POINT ANALYZER, HYDROCARBON DEW POINT ANALYZER AT ONGC KHUBAL	No.	01	

Page 3 of 644

Rev.3

	FIELD (Ref P & ID No.: MEC/23VC/05/28/M/001/FEEDER- 1005)			
A.7	CHECK METERING FACILITY WITH GAS CHROMATOGRAPH, WATER DEW POINT ANALYZER, HYDROCARBON DEW POINT ANALYZER AT RT SILCHAR (Ref P & ID No.: MEC/23VC/05/28/M/001/FEEDER-1006)	No.	01	
A.8	SCRUBBER AT ONGC JANTAPATHAR SOURCE (Ref P & ID No.: MEC/23VC/05/28/M/001/FEEDER-1002)	No.	01	
A.9	SCRUBBER AT BANASKHANDI GCS (Ref P & ID No.: MEC/23VC/05/28/M/001/FEEDER-1006)	No.	01	
A.10	Assistance in Installation, Testing and Commissioning of skid at site, consisting of Scrubber, Gas Chromatograph, Water Dew Point Analyzer, Hydrocarbon Dew Point Analyzer, Filtration, Metering through Single Chamber Orifice Meter with field mounted flowcomputer / panel mounted Flow Computer including Metering Panel & Gas Detection System along with the accessories. The price shall be inclusive of Airfare Boarding, Lodging, Local Transport, Incidental, Traveling etc. & all other expenses.	Man Days	49	

# Notes:

- 1 Evaluation shall be done for complete Package.
- 2 The Metering skid supplier shall be responsible for assistance in Installation, Testing and Commissioning of quoted MR items. For this purpose, 49 man days are considered for evaluation purpose. However, payment shall be as per actual requirement. Bidders shall indicate separate price for all quoted Item nos. A.1 to A.9 against item no. A.10.
- 3 The Vendor shall be completely responsible for the design, materials, manufacture & fabrication, testing, inspection, preparation for shipment and transport of the above equipment strictly in accordance with the MR and all attachment thereto. All items shall be provided with EN 10204-3.2 certificates.
- 4 Vendor shall appoint any one of the following TPIA for inspection purpose. Vendor has to propose minimum 4 nos. of below listed agencies to be approved by IGGL/MECON.
  - a) Lloyd Register of Industrial Services
  - b) Technische Ulierwachungs Verein (TUV)
  - c) Det Norske Veritas (DNV)
  - d) AB-Vincotte
  - e) Bureau Veritas
  - f) SGS
  - g) American Bureau Services
  - h) Velosi Certification Services

Apart from inspection by TPIA, inspection shall also be performed by MECON / IGGL's delegate, as set out and specified in the codes and particular documents forming this MR.

Rev.3

# VENDOR DATA REQUIREMENT

		Prints	Certifie	d informati	ion require e Order	d after
SI. No.	Description	with Quotes	Soft Copy	Printed Matter	Date	Date Promised
a)	Piping/Instruments					
01	Dimensional outline with mounting details for each item with Model No.	5	1	6	2 W	
02	Connection by Purchaser (Piping, Electrical)	5		6	2 W	
03	Wiring Diagrams	5	1	6	2 W	
04	Parts List	5	1	6	2 W	
05	Recommended Spares with Prices	5		6		
06	Installation, operations and maintenance instructions		1	6	W/S	
07	Test certificates			6	A/C	
08	Certificate from statutory bodies	5		6	2 W	
09	Assembly Details	5	1	6	2 W	
10	Technical literature & model decoding for each item	5	1	6	2 W	
*11	Testing and inspection procedures		1	6	2 W	
*12.	Data Sheets of all the Instruments		1	6	2 W	
13.	Software Manual/ Hardware Manual etc.		1	6	2 W	
14	Functional/Loop schematics	5	1	6	2 W	
b)	Skid			6		
*1	Layout of equipment (channel mounted)	5	1	6	2 W	
*2	GA drawing with dimensional details	5	1	6	2 W	
3	Structural details	5	1	6	2 W	
*4	Piping diagrams		1	6	2 W	
5	Foundation details		1	6	2 W	
6	Single line diagram	5	1	6	2 W	

# Notes:

- 1. Categories proceeded with "\*" will be approved for fabrication by MECON LIMITED. The remaining drawings are needed for information only.
- 2. Fold all drawing to 210mm x 297mm.
- 3. Vendor to provide all printed matter and the soft copy to MECON LIMITED.
- 4. Legends:

A/C = As completed

- W/S = With Shipment
- W = Weeks

# **TECHNICAL QUESTIONNAIRE**

This questionnaire shall be duly filled in and submitted alongwith unpriced sets of <u>offers to</u> <u>avoid further queries and to ensure proper evaluation of your offer in time</u>. If this is not complied with, your offer is liable to be rejected.

ANSWER `YES', `NO' OR `NOT APPLICABLE'

SI.	Description	Bidder's Response
INO.	The second station is the falles for	
1.	Have you any deviation to the following :	
<u> </u>	General specifications (Clause Wise)	
II)	Special requirements (Clause wise)	
iii)	data sheets/ specifications.	
iv)	The scope of supply as indicated in the material requisition.	
	If `Yes', have you included the list of deviations? (If no deviations are furnished, It will be assumed that all the specifications and requirements of the subject requisition are complied with and no deviation whatsoever will be accepted after the placement of order. Further, bidder shall give a undertaking separately that in case of any technical deviation observed in the bid document, same shall be treated as null & void and tender conditions shall	
	govern)	
2.	Have you quoted for the spare parts including consumable items required for the startup and normal operation/ maintenance of the instruments?	
3.	Have you quoted for the mounting accessories, special calibration kits and equipment (complete with technical details) required, if any, for the erection, commissioning and maintenance of the instruments?	
4A)	Have you enclosed the relevant technical catalogue/ literature in ENGLISH language including model-decoding details, drawings etc. necessary for the evaluation of your offer? (Note : In case of line-mounted instruments, viz. control valves, positive displacement meters, pressure relief valves etc. for all the lines quoted, relevant dimensional details required for the installation must be furnished along with the offer)	
4B)	Have you confirmed that the documents required as per the vendor data requirements will be supplied after placement of order?	
4C)	If so, have you indicated the extra price applicable, if any?	

SI. No.	Description	Bidder's Response
5.	Have you furnished sizing, noise calculations and certified capacity curves for the instruments wherever applicable?	
6.	Have you furnished separately the shop inspection charges, if any, for inspection at your works by MECON/ Third party? (The details of the tests to be carried out on the instruments during shop inspection are to be furnished).	
7.	Have you furnished the certificates from statutory bodies viz. BASEEFA, FM, CSA etc. for the explosion proof construction/ intrinsically safe design of the instruments wherever specified?	
8.	Have you confirmed that IBR certification in form III (C) or equivalent certification from statutory bodies viz. Lloyds, Bureau Veritas, TUV etc. will be furnished wherever specified?	
9.	Have you indicated separately the Supervision charges for installation, testing and commissioning, in case the same are included in your scope of work?	
10.	Have you furnished the estimated power/utility consumption and special cable requirements, if any, for the instruments quoted?	
11.	Have you furnished the customer reference list in India/ abroad ?	

<u>Note</u>: If the answer is in negative, then furnish response thereof.

Page 2 of 2 Page 7 of 644

# VENDOR DRAWING/ DATA APPROVAL PROCEDURE

1. Vendor must take care of the following while submitting drawings and documents for review as indicated in Vendor Data Requirements enclosed.

A blank space measuring 75mm W x 40mm H shall be provided on all vendor drawings for marking review codes etc. by MECON LIMITED.

The review of vendor drawings shall be done as applicable under the following review codes:

- a) Review Code Approved : Approved
- b) Review Code Approved As Noted: Proceed with manufacture/ fabrication as per commented drawings. Revised drawing required.
- c) Review Code Not Approved: Document does not conform to basic requirements as marked. Resubmit it for review.
- 2. Review of the vendor drawings by MECON would be only to check compatibility with basic design and concepts and would in no way absolve the manufacturer/ fabricator of his responsibility to meet applicable codes, specification and statutory rules/ regulations.
- 3. For drawings/ documents indicated as FOR INFORMATION in the Vendor Data Requirement, Vendor must mark FOR INFORMATION ONLY on the submitted drawings/ documents.

	CHECK LIST (Technical)	
Bidder shall pos	itively include all the information / documents asked in this check list for p	roper evaluation.
SI. No.	Description	Bidder's response
a)	P&ID of each Quoted Skid enclosed	Yes / No
b)	Tentative Bill of Material for each Quoted items (Skid) as per the MR.	Yes / No
c)	Make, Model No. and decoding details of each supplied items.	Yes / No
d)	Letter of confirmation for "No Deviation" enclosed separately for each skid.	Yes / No
e)	Filled in data sheets of each supplied item submitted.	Yes / No
f)	Letter of confirmation that offered materials are as per data sheets enclosed separately	Yes / No
g)	GA drawing (with dimensional details)/ Catalogue of each supplied item like Cartridge Filter, PCVs, SDVs, Valves, PSV, Meters etc. submitted separately.	Yes / No
h)	Signed & Stamped copy of Tender Document including MR for Quoted section.	Yes / No
i)	Contents / Index sheets with page numbering details for attached documents / drawings / certificate enclosed	Yes / No
j)	Sizing/ Design calculation for each supplied item like Cartridge Filter, PCV & SDV, PSV, Meters etc., submitted	Yes / No
k)	Flow computer sizing details submitted.	Yes / No
I)	Documents submitted as per "Vendor Data requirement" sheets	Yes / No
m)	Filled-in Technical Questionnaire submitted	Yes / No
n)	Separate list for 2-years' recommended spares with printed price-list enclosed	Yes / No
o)	Letter of confirmation that make of the various items shall be as per "Approved Vendor List" enclosed with this tender document.	Yes / No
p)	Items and vendor list for approval of MECON / Client or the items and vendors not covered in "Approved Vendor List" of this tender document	Yes / No
q)	Estimated shipping weight & volume of each item furnished with the office	Yes / No
r)	Quoted for each supplied items as per Schedule of rates.	Yes / No

			Check List (BEC DOC	SUMENT)			ANNEXURE - /
NAME OF BIDD	ER:						
OFFER NO & D	ATE :						
QUOTED FOR I (AS PER SOR)	TEM NO. :						
				Furnished By Bidder			
Item No.	BEC CLAUSE NO.	REFERENCE DOCUMENT AGAINST THE CLAUSE	P.O. REF. NO. & DATE	INLET / OULET SIZE & RATING OF SUPPLIED SKID	SUPPLIED TO (NAME OF CUSTOMER)	Page No. Ref (Attached with Bid Document)	Remarks
~	CLAUSE NO 3.1						
м	CLAUSE NO - 3.1.1 (a)						
m	CLAUSE NO - 3.1.1 (b)						
4	CLAUSE NO 31.2						
a	CLAUSE NO 3.1.3						
ω	CLAUSE NO 3.1.4						
NOTE : BIDDER S	HALL SUBMIT THE SEPARATE CHECK LIST (BEC)	:) FOR EACH QUOTED SKID.					

						ANNEXURE - B
		(Mech	Check List anical & Instrumentatic	on Items)		
NAME OF	BIDDER:					
OFFER NC	O & DATE :					
QUOTED	FOR ITEM NO. : SOR)					
Item No.	ITEM DESCRIPITION			Furnished By Bidder		
		SIZE & Rating	Make	Offered Model No.	Page No. Ref (Attached with Bid Document)	Remarks
-	Catridge Filter					
2	Pressure safety Valve / CRV					
3	Slam Shut Valve					
4	Pressure Control Valve					
ъ	Flow Meter (USM / Turbine / RPD)					
9	Flow Control Valve					
		NB > 2", 150#				
7	Ball valve	NB > 2", 300#				
•		NB > 2", 600#				
		NB<2", 800#				
		NB > 2", 150#				
œ	Plua Valve	NB > 2", 300#				
)		NB > 2", 600#				
		NB ½" - 1½", 800#				
		NB > 2", 150#				
σ	Check Valve	NB > 2", 300#				
>		NB > 2", 600#				
		NB<2", 800#				

		300#		
01	Globe Valve	600#		
11	Flow Computer (panel mounted / field mounted)			
12	Pressure Transmitter			
13	Pressure Gauge			
14	Temperature Gauge			
15	Resistance Temperature Detector			
16	Temperature Transmitter			
17	Differential Pressure Transmitter			
18	Differential Pressure Gauge (if Applicable)			
19	Metering Control Panel			
20	Gas Chromatograph (if Applicable)			
21	LEL Dectector (if Applicable)			
22	Metering Supervisory System (if Applicable)			
23	Laptop			
24	Printer (Colour & Dot matrix)			
25	GSM Modem	-		
26	HOV actuator			

27	Solar Panel (if applicable)				
28	Battery (if applicable)				
29	Junction Box				
30	Cable Glands				
31	Battery Charger (if applicable)				
32	Cables (signal / serial / power)				
33	SS tubes				
34	SS fittings, valves & manifolds				
35	Barriers & SDC				
NOTE : BID	DER SHALL SUBMIT THE SEPARATE CHI	ECK LIST (Mechanical {	& Instrumentation Items) F	OR EACH QUOTED SKID.	





# **JOB SPECIFICATION**

# (Spec. No. - MEC/05/ 21/E5/JS-030)





#### **CONTENTS**

#### 1.0 GENERAL

#### 2.0 SCOPE OF WORK & RESPONSIBILITY

- 2.1 Scope of work
- 2.1.1 General
- 2.1.2 Design and Engineering
- 2.1.3 Procurement/ Supply, Factory testing and Acceptance
- 2.1.4 Installation, Field Calibration/ Testing and Commissioning
- 2.2 Scope of supply
- 2.3 Further Scope of Work & Supply
- 3.0 DESIGN PHILOSOPHY
  - 3.1 General
  - 3.2 Codes & Standards
  - 3.3 Instrument Design Criteria
  - 3.4 Junction Boxes & Cabling Philosophy
  - 3.5 Instrument Painting Requirements
  - 3.6 Power Supply & Power Supply Distribution

#### 4.0 PROCESS DESCRIPTION

- 4.1 Metering Skid Category
- 4.2 Process Data
- 5.0 JOB DESCRIPTION
- 6.0 TESTING AND INSPECTION
- 7.0 DOCUMENTATION
- 8.0 MISCELLANEOUS
- 9.0 SHIPPING AND PACKAGING
- 10.0 PERFORMANCE GAURANTEE
- 11.0 INSTALLATION AND COMMISSIONING
- 12.0 BIDS
- 13.0 MANDATORY SPARES
- 14.0 ATTACHMENTS





#### 1.0 GENERAL

This specification together with all annexure enclosed cover the requirement for the design, engineering, manufacturing, testing, inspection and supply of **Single chamber Orifice meter-based Metering Skid**, along with all accessories for different location as per P & IDs. Each stream of Skid shall be designed to cater flow of 100% of max. (Rated) flow capacity (as per P&ID).

The scope of work / supply includes Design, Engineering, Manufacturing, Inspection, Testing and Supervision for Installation & commissioning of metering skid consisting of (1W + 1S) Filtration, (1W + 1S) Flow measurement through Single chamber Orifice meter with field mounted flow computer (alongwith solar panel, batteries, charge controller, JB, etc.) (for item no. A.1 only), panel mounted Flow Computer with Metering Panel & its accessories (for item nos. A.2 to A.7), Gas Chromatograph (for item nos. A.4 to A.7), Water Dew Point Analyzer (for item nos. A.4 to A.7), Hydrocarbon Dew Point Analyzer (for item nos. A.4 to A.7), LEL Detection System (for item nos. A.2 to A.7), Scrubber (for item nos. A.1 to A.6, A.8 & A.9) and valves, piping, instruments & fittings as per P & ID's. The Required capacity, Pressure Rating, Quantity and Location of the Skids shall be as per P & ID.

- 1.1 The description and requirements contained in this specification are concise by necessity and cannot include all the details. However, it is the responsibility of the bidder to execute the job on a turnkey basis in accordance with the specifications and internationally recognized good engineering practices.
- 1.2 Any activity specifically not listed in this document, does not absolve the bidder of their responsibility to include such activities in their scope of work and supply, which otherwise is necessary, to complete instrumentation work for the project. All such activities shall be carried out by the bidder without any cost/ time implication.
- 1.3 In the event of any conflict between these specifications, related standards and codes, any other attachment to this package, the bidder shall follow the following documents in the order of their priority:
  - a) Job Specifications for Gas metering package
  - b) Data Sheets and Typical P&ID
  - c) Standard specifications and Technical Specification attached in the tender document.

In case of any conflict in various documents, same shall be referred to CLIENT/Mecon for clarification. Bidder shall not proceed without getting written approval in such a case.

#### 2.0 SCOPE OF WORK & RESPONSIBILITY OF BIDDER

#### 2.1 Scope of Work

- 2.1.1 General: Bidder shall be responsible for execution of the package on turnkey basis with scope of work as listed below but not limited to the following:
  - a) Design and Engineering
  - b) Procurement/ Supply, Inspection, Factory testing and Acceptance
  - c) Supervision for Installation, field calibration/ testing and commissioning.

#### 2.1.2 **Design and Engineering:**

(a) Owner shall provide the bidder with this bid package consisting of typical Piping & Instrumentation Diagrams and process data sheets. The bidder shall be responsible to carryout the design and detailed engineering based on the data provided in the bid package and in line with other technical requirements specified elsewhere in this document. Scope shall also include sizing and sizing verification for all items including where data is dependent upon detailed engineering, detailing of basic engineering designs, preparation of data sheets, coordination drawings for instruments and system-oriented items, engineering drawings etc.

#### (b) Residual Engineering

The bidder shall also be responsible for carrying out any residual basic engineering necessary for proceeding with detailed engineering like equipment/ instrument sizing, utility consumption, specifying derived data in process data sheets, type and material selection of instruments/ equipments wherever required.





#### (c) Control Room Engineering

Scope of skid supplier (bidder) for control room engineering is as explained below:

The complete skid / Metering System data (including gas composition, flow computer data and differential pressure, Pressure, Temperature and valve status) is required to be sent to remote station through CLIENT's RTU / SCADA. Gas metering panel shall be complete with Flow computers, Receiver instruments like signal selector, Power supplies, Isolating IS barriers & accessories, lamps, LEL gas monitor/ controller, converters etc. duly mounted completely wired & GC Controllers shall be mounted in same metering Panel, A separate terminal strip (designate as Telemetry Interface to RTU) shall be provided for terminating the signals to/from SCADA in this metering panel, Suitable hardware/ software is to be supplied for SCADA/RTU connectivity.

For Supervisory monitoring and control, a SCADA system is envisaged to ensure effective and reliable control, management and supervision of the consumers from centralized location. The SCADA system along with RTUs shall be provided by CLIENT. However, the bidder shall be responsible for configuring their systems and providing all the necessary details of their systems to the SCADA vendor.

#### (d) Engineering Drawings & Documents

- Vendor Data Requirements indicate the list of drawings and documents required to be supplied by the bidder, as a minimum. Bidder to note that list specifies only the major deliverables. Documents and drawings not listed but necessary for proper engineering, construction, operation and maintenance shall also be prepared by the bidder.
- ii) Bidder shall be responsible for preparation of all engineering drawings and documents including those necessary for construction like instrument index, tray layouts, location plans, cable schedules, installation standards, bill of material etc.
- iii) Bidder shall also be responsible for providing all drawings and documents for package/ sub package units
- iv) It is expected that bidder utilizes uniform data sheet formats enclosed along with this document, for preparing specifications for various instruments, including those, which are being prepared by package/ sub package vendors. Items for which no format has been attached with the document, bidder may use standard ISA formats. Use of manufacturer standard formats shall be avoided.
- v) The bidder shall supply all the documents in both hard copy and soft copy. This includes all the documentation including those for package units.
- e) The design and engineering work shall also include review of post-order vendor drawings and documents for all instruments and system-oriented items. Following methodology must be followed for drawings and documents being forwarded to Mecon.
  - i) The Bidder shall thoroughly review and approve vendor drawings for all instruments including sub-package items, before forwarding to Mecon. Only the approved drawings duly stamped and signed by a competent representative/ engineer of Bidder shall be forwarded.
  - ii) The Bidder shall be responsible for all System Engineering documents for the Gas metering systems. This shall include all related documents such Functional design specifications, sizing calculations, pressure drop calculation etc. and Engineering documents such as functional loop schematics, instrument details and cable schedule, Power supply distribution schemes etc. These documents shall be reviewed and approved by Bidder based on philosophy specified/ agreed for the engineering before forwarding to Mecon.
  - iii) All multidisciplinary fabrication and construction drawings shall be reviewed and signed by bidder's respective departmental representatives before forwarding to Mecon for review/ approval/record.
- f) Bidder shall be fully responsible for co-coordinating with all agencies concerned to ensure proper, uniform and smooth engineering. This shall include coordination with:





- All individual item suppliers for uniformity in engineering and documentation supplied by them including P&ID's, instrument specifications, installation standards etc. and obtaining all requisite drawing and documents for review, record and final documentation.
- ii) All instrument item suppliers including suppliers/ manufacturers of various system-oriented items.
- iii) Bidder's own inter-departmental coordination with departments like mechanical, piping, electrical, QC, pressure vessel, heater group etc. This shall include furnishing all necessary engineering data in the form of drawings & documents and review of drawings & data supplied by other departments.
- g) Bidder shall be responsible for preparation of all As-Built drawing / documents including
  - i) All P&IDs and GADs
  - ii) All Datasheets, specifications of instruments
  - iii) All Purchase documents.
  - iv) All System documents including hardware and software documentation.

#### 2.1.3 **Procurement/ Supply, Factory testing and Acceptance**

- a) Engineering for procurement shall include preparation of various material requisitions which shall include process data sheets, typical data sheets for instruments, instrument standard specifications, special requirements etc., evaluation of offers received from various manufacturers/vendors, preparation of Technical Bid Analysis, preparation of purchase requisition and review/ approval of vendor drawings, incorporation of Mecon comments.
- b) All Instrument items and Gas metering systems shall be procured from vendor list attached elsewhere in this package. Bidder must desist from procuring any items from vendors not approved by Mecon. Along with the bid, the bidder has to provide the list of items/ equipments with size, Make and Model Number selected form the approved vendor list only. Non-compliance to this may result into rejection of bid.
- c) Bidder shall prepare purchase requisitions for all instruments/ systems which shall consist of a consolidated purchase document including all purchase specifications including data sheets, special instructions/ requirements (if any), standard specifications/ purchase specifications, testing requirements, quality requirements etc. All purchase requisitions shall be furnished to Mecon for information/ review/ approval.
- d) Testing & calibration of all instruments, Factory Acceptance Test (FAT) and Site acceptance Test (SAT) shall be carried out by the bidder. Range/ calibration span, set points, reports etc shall be modified as per CLIENT's requirement by the bidder during FAT and SAT. CLIENT / Mecon shall witness testing of any or all items at various stages during manufacture and/or at final stage before shipment at their discretion. Testing shall be carried out as per approved procedures. No instrument shall leave manufacturer's works without factory acceptance test. All necessary changes shall be incorporated/ implemented as suggested by CLIENT / Mecon during FAT/ SAT etc. As build drawing/ documentation to be submitted by the bidder shall contain all such changes.

#### 2.1.4 Installation, Field Calibration/ Testing and Commissioning

- a) Bidder shall carryout installation of all instruments in the skid as described in this document. Installation shall include but not limited to installation of all supplied items, installation skid assemblies explained in this package, installation of junction boxes, interconnection between instruments and junction boxes, fabrication, laying and painting of cable trays, laying of all single pair and multi pair cables in the skid, JB earthing/ grounding, Field Instruments/ signal earthing/ Grounding, tagging, ferruling, cable glanding and pair/ core identification of all field cables.
- b) Distribution of power to various instruments in the skid from single point in the skid. Separate JB is required for different type of signals and also for power to field instruments (Power supply to Meter, GC and other Field Instruments shall be in separate JBs).





- c) Installation shall be carried out as per Mecon standards or as recommended in the tender. For special instruments, the installation may be carried out as per vendor/ manufacturer's recommendations however all such installation standards shall be subject to Mecon review.
- d) Bidder's scope of supply/ work shall include earthing cable/ strips etc. (along with cable tray with supports for installation) in the skid/ JB/ control room, as per the requirements of various instruments.
- e) Due to limitations in transportation of the skid, the skid has to be transported to site in different modules. Individual modules shall be so designed that there are no / minimum cabling interconnections between various modules and such interconnections if any shall be in the bidder's scope.
- f) Bidder shall quote for man day rates as per SOR for installation supervision and commissioning of the complete skids. The quoted man-day rates for installation supervision and commissioning shall be valid for 24 months from date of issuance of FOI. Bidder shall depute qualified and competent person for installation supervision / commissioning.

#### g) Testing & Calibration

Bidder scope of work includes testing of all supplied items and systems including impulse lines, pneumatic signal tubes and instrument cables and special instruments/ items if any. Bidder shall also carryout testing and calibration of all instruments as per the requirements specified elsewhere in tender document. Testing and calibration of Gas metering system shall be as described elsewhere in the document.

#### h) Commissioning

It is the responsibility of Bidder to co-ordinate and make available the services of vendors/ sub-vendors for gas metering system package, control system, etc. and other special instruments/ equipments like Gas Chromatographs, Gas flow meters, Flow computers, during installation, testing, FAT, Site acceptance, startup/ commissioning of the station. The bidder shall provide assistance during commissioning without any condition/ pre-requisite. It is the responsibility of the vendor to get the certification from site Engineer. Installation of all the loose supplied items, its interconnection etc (except control panel erection, skid erection, construction of earth pit, cabling / cable tray laying from Metering skid to control panel) shall be in the scope of supplier. In case of any dispute / conflict arising due to difference in opinion/ interpretation, the interpretation of CLIENT /Mecon shall be considered final.

#### 2.2 SCOPE OF SUPPLY

Supply of all items as indicated on the typical Piping & Instrumentation diagrams (P&ID) and other technical documents attached with this document.

#### 1) The scope of supply shall include but not limited to the following as a minimum:

- a) Skid mounted Custody transfer Gas metering system package inclusive of Gas filtration & Gas metering (through Single chamber Orifice meter) complete in all respect as per P & IDs, Process data sheets & as described in Job specification for Gas metering system package.
- b) All field instruments such as Pressure Transmitters, Differential pressure transmitters, pressure gauges, temperature gauge, Temperature Transmitters & RTDs, Pressure Safety Valve, Creep relief valves, LEL Gas detectors, etc.
- c) Supply of Gas Chromatograph is envisaged for item nos. A.4 to A.7 as per MR. The G.C Controller to be mounted in the same metering panel.
- d) Supply of fully wired metering panel for Metering skid (for item nos. A.2 to A.7) as per MR with receiver instruments & accessories such as, indicating LED lamps, required intrinsic safety barriers, isolators, TB, tube light, space heater, cable ducts & Panel Mounted Flow computers and Color Printer and its accessories, signal converters, hub, hardware required to establish various serial link / connectivity with different instruments/ items (like Printers, GSM Modem, Single chamber Orifice meter, Laptop, SCADA/ RTU, GC, FC, LEL Controller etc) mentioned elsewhere. All the control room hardware (GC controllers, LEL controller, GSM Modem, FC, printer, laptop,





communication cable required for control panel items) to be installed in the supplied metering panel. All hardware and software required for using diagnostic features of supplied instruments like Flow meters, Flow computers, GC, etc. shall be provided by the bidder.

- e) **GSM Modem** shall be provided with each flow computer for polling data from master control station (previous 35 days data) and remote data monitoring.
- f) All installation and erection materials such as impulse piping, pipe fittings and valves, copper jumpers, anchor fastener, tubes, tube fittings, SS valves & manifolds, cable tray and supports (for JB, vents, tubing etc.), cable glands, ferrules, lugs, cable ties, cable sleeves, TBs, earthing cable, foundation bolts of the skid, gaskets, companion flanges for inlet and outlet of the skid, all type of consumables and accessories for mounting of instruments, instrument supports, tray supports, canopies/ sunshields for all field mounted instruments.
- g) Supply of all types of cables such as signal, alarm, control, earthing cable, power cables. For supply of cables with each skid, Distance between metering skid (field) and Control room/ Metering panel shall be approx. 250 Meters. Supplied cables shall be intrinsically safe and shall be as per Mecon Specification. Preparation of cable trench and laying of cables and cable trays between skid and control room is not in the Scope of bidder. However, supply of all the materials like cable ferrules, cable lugs, cable ties, cable sleeves, TBs, earthing cable, cable tray supports, anchor fastener, cable trays including cable glands at both ends is in bidder's scope. Bidder shall also supply suitable power cable (for supply of UPS/ Non-UPS power to control panel, power distribution etc.) and also cable required for SCADA/ RTU connectivity. Separate runs (from field to metering panel) of cables to be considered for each Metering stream & metering to be send directly to the metering panel i.e. not through JB.
- h) The distance between Power distribution/ UPS and Control panel/metering panel shall be approx. 50 meters. The distance between Control panel/Metering panel and SCADA/ RTU panel shall be approx. 30 meters. Supply of all the materials like cable ferrules, cable lugs, cable ties, cable sleeves, TBs, earthing cable, cable tray supports, anchor fastener, cable trays including cable glands at both ends is in bidder's scope. Suitable cables of suitable length to be supplied as follow:
  - i. All type of cable from field to control room.
  - ii. Communication and signal cable from control panel to SCADA/ RTU panel, Printer & Laptop etc.
  - iii. Power supply to control panel, field instruments considering the above indicated distances.
- i) All field Instruments of Metering i.e. PT impulse line & TT Thermowell along with flange, Meter run, meters (Single chamber Orifice meter), flanges, fittings & GC impulse line shall be Cold Insulated. Extra care should be taken by the bidder to avoid damage of insulation during transportation.
- j) Perforated tray, angle tray, accessories required for cable laying and routing up to the control room through cable trench. (Only supply of these material is in bidder's scope, however cable laying, tray work and cable/ tray erection/ installation is not in bidder's scope)
- k) Junction boxes and cable glands (as per the requirement of area classification) for different types of signals such as intrinsically safe, alarm, power etc. should be supplied and mounted/ installed in metering skid. Preferably, all JBs shall be mounted in skid at height of 1.3 meter from ground/ finished floor level/ skid base frame and accessible / approachable from outside. JBs shall not to be installed inside Skid.
- Painting of pipes, equipments, instruments, control panel, enclosures, platforms, jump-over/ crossovers, as required in line with the painting specifications attached elsewhere in this document. Also flow direction to be marked on mainline piping.
- m) Galvanized iron/ copper earthing strip and earthing cables for earthing of all instrumentation items including junction boxes etc. to instrument earthing system. (Supply of earthing strip / cable is in bidder's scope. Earthing pit at a distance of 50 meters shall be considered). Making of earth pit and laying of earthing cable/ strips is not in bidder's scope.
- n) Suitable environmental enclosure for custody transfer field instruments for environmental effect protection with a provision of locking. Any other erection material necessary for installation and commissioning of special instruments, if any.





- o) Platforms & cross-over(s) / Jump-over(s) to be provided for the operation and maintenance of the equipments/ instruments/ JBs installed in the skid. Proper spacing to be maintained between the equipments for operation & maintenance. The vent & drains shall be properly supported in the skid. All the vents shall be at a height of minimum 3 meters above the working platforms. All flanges shall be connected through flexible jumpers of Minimum 3 mm thickness and 20 mm wide copper strips. Neoprene rubber sheet to be provided at all the supports for mainline pipe / valves / tubes / equipments etc. U turn flanged spool piece to be provided as spare to change the orientation of skid & the dimensions of the spool piece to be decided during detail engineering. All the extra tappings to be plugged & sealed (if not used).
- p) Any special tools/ tackles required shall be in scope of vendor.
- q) All the software used in the system and as specified elsewhere, shall be licensed in the name of CLIENT.
- r) One number of Laptop & Color Desk Jet / LaserJet Printer to be supplied with Metering skid (for item nos. A.2 to A.7) for panel mounted Flow Computers with auto switch between flow computers.
- s) Compulsory spares and commissioning spares (for each skid separately) as listed elsewhere in this document.
- t) Drawings and documents as listed elsewhere in this document.
- u) All pipes, tubing, fittings, valves, gaskets, bolts, nuts, spades, etc.
- v) A structural skid complete with necessary drip pan, walkways, staircase, platforms, gratings, handrails for access for operation and maintenance.
- w) Earth bonding system and earthing boss.
- x) Lifting lugs and spreader beam / frame, foundation Anchor bolts for the skid.
- y) Stainless steel nameplate for each skid, each tagged equipment and component.
- z) All Tie-ins (flanged connections) with nuts and bolts.
- aa) Inlet and Outlet matching flanges and Studs & nuts shall also be supplied along-with the skid.
- bb) The sun/ rain protection shed for Gas chromatograph field unit and for electronic instruments shall also be supplied by the bidder.
- cc) Operation and maintenance manual, for instruments/ equipments, as built drawing/ documentation.
- dd) The direction of Inlet & outlet is fixed. However, the orientation and detailed GAD will be approved as per the plot details and will be dealt during detailed engineering.
- ee) Please note that Bidder shall submit hard copies of all documents/ drawings to MECON. The date of receipt of these documents/ drawings at MECON shall be deemed as the date of submission. If any documents/ drawings require re-submission due to any error/ deficiency noticed during review/ approval stage, in that event the additional time required by the bidder/supplier to get the revised document/ drawing reviewed/approved by MECON shall be solely to bidder's/supplier's account and in no case the bidder/ supplier shall be entitled for any time or cost benefit. Also, P&ID shall be submitted within 2 weeks from the date of award of contract & GAD shall be submitted (after approval of P&ID & mainline equipments only) within 1 week and will be considered accordingly.

#### 2.3 Further Scope of Work and supply:

a) Vendor scope includes Design, Engineering, Manufacturing, Inspection, Testing, Transportation, Supervision for Installation & commissioning of complete supplied items (Filtration, flow measurement, Stream Flow computers (for each stream), IR type LEL Detection system, safety valves, Pressure Instruments, Temperature Instruments





Control Panel, complete integrated field instruments, piping, fittings and valves in a skid and a control room equipment with the following items described below as a minimum and as shown on the P&ID) of metering skids / metering system.

- b) Metering skids shall consist of complete metering package (consisting of Filtration and Metering. Each Metering streams of Skid should be designed for 100% of maximum flow capacity as per P& ID. Each of the Gas filtration and Metering Section shall be of required design flow capacity meeting the specified operating design conditions as per process data of respective skids. The Vendor's scope of work shall include supply of the complete Gas Filtration & Metering Skid in accordance with this specification and any other codes, standards and regulations stated herein. In each metering skid, one complete stream will be kept on a hot stand-by mode (including filtration, Metering etc.)
- c) The scope of supply as a minimum and as shown on the P& ID shall include, but not limited to the following.
- d) One gas filtration system with two gas filters (<u>1 operating + 1 standby stream</u>) (for each skid), each of 100 % of maximum flow capacity.
- e) One Gas metering system with two metering streams (for each skid) of Single chamber Orifice meter (1 operating + 1 stand by stream). Flow computer shall be provided for each meter stream with all interface accessories. Each metering stream shall be designed for 100 % of capacity.
- f) Gas Chromatograph system (for item nos. A.4 to A.7) shall consist of complete field mounted GC analyzer cabinet (closed from 3 sides with door/ access from 1 sides) and carrier and calibration gas cylinders assembled on a skid, retractable sample probe, sample tubing, sample conditioning system, and remote interface / display unit (control panel mounted) shall be supplied by the bidder along with each skid. Gas Chromatograph shall be used for Calculation of Gas Composition. The GC probe shall be installed at least 20D downstream from any flow disturbing elements such as elbows, valves, headers, tees as per ISO 10715. Separate Flanged Outlet pipe with provision for GC probe installation shall be provided for the same (It can be supplied loose). The calibration gas shall have 3 years stability with complete traceability. The on-line GC shall be certified for custody transfer application by a certifying agency equivalent to Nmi/ PTB/ NPL. The Helium Gas cylinders and Calibration Gas cylinders shall be approved by CCOE or equivalent body. The Gas Chromatograph shall be interfaced with all the Flow computers and on-line GC data to all the Flow computers shall be available all the time.
- g) The Gas Flow Meter shall be certified for custody transfer application by a certifying agency equivalent to Nmi/ PTB/ NPL. The maximum permitted velocity through meter shall be 20 meter per sec.
- h) Skid mounted field instruments like pressure gauges, Temperature gauges, Pressure & diff. Pressure transmitters, Temperature transmitters, temperature elements & thermo wells, shall be supplied along-with the skid.
- i) The metering stream (flow meter, up-stream & down stream meter tube, flow profiler, impulse tubing of Pressure transmitter, thermo- well, impulse tubing of on-line GC) shall be completely insulated to ensure an even heat transfer throughout the meter run when subjected to ambient environment.
- j) The custody transfer equipments at field like transmitters (pressure & temperature) shall be installed in an environmental enclosure to minimize the effects of ambient temperature variations and shall be lockable for prevention of unauthorized data entry.
- k) LEL gas detectors (IR Type point detectors) (quantity as per datasheet) on the skid and LEL Gas monitor in the control panel shall be provided as per data sheet and applicable standards / codes.
- Panel mounted Flow computer along with printer, GSM Modem & its accessories shall be supplied for Metering Skids (for item nos. A.2 to A.7) as per MR. The flow computers shall be control room mounted. 230 VAC, 50 Hz power supply shall be available for Control panel to be installed in control room. Flow computers and all other accessories for Metering system shall be installed in control panel. The system shall be designed in such a way that it should operate at 230VAC-50 Hz. Signal converter, signal selector, Isolating IS barriers & accessories, a separate terminal strip for all SCADA signals to/ from CLIENT's RTU duly mounted completely wired also to be provided. The entire system shall be mounted on control panels. PID controller shall be inbuilt within the flow computer, in case the supplied flow computer does not have PID control functionality, separate PID controller shall be supplied.





- m) Supply of on-line Solar Panel & Batteries alongwith field mounted Flow Computer for Metering skid (for item no. A.1 only) as per MR.
- n) On line Solar Panel & Batteries along with field mounted Flow Computer & its accessories shall be supplied for Metering skid (for item no. A.1 only) as per MR. The system shall be designed in such a way that it should operate through battery. Bidder shall submit the Battery sizing calculation, Details of battery & Solar Panel, Product Technical Literature along with offer. The Rating, make and other necessary details in support of selected model for Battery & Solar Panel shall also be submitted. The power consumption of Flow Computer, Modem and other associated utilities of online system shall be considered for Backup calculation and finalization of sizing of Battery. Charger for batteries to be kept in separate JB & Battery JBs to be provided with breathing valves.
- o) Supply of metering panel (for item nos. A.2 to A.7) to house all the required control room components including LEL detection system, GC, Flow computers, Ball valve status indication, other hard-wares like signal converter, barriers, relays, cables, connecters, power supply, modem etc. to ensure completeness of metering system is in scope of bidder. Cable gland for connecting Field signal/ control cables, power cable in control panel is also in the scope of bidder.
- p) Internal or external USB converter and cable shall also be provided with each flow computer for connecting it to USB port of Laptop for configuration.
- q) Supply of all pipes, tubing, fittings, valves, gaskets, bolts, nuts, spades, etc, are in bidder's scope.
- r) All cables (including earthing cable), cable trays, earthing strips for grounding/ earthing of skid/ panel and wiring within the skid and from skid to metering control panel (Only approved, standard armoured cable to be considered for this project). All Interconnecting cables between skid Instrument/ Junction boxes to metering control panel and inside control panel (to be located in Local equipment room in safe area). Bidder to supply required mounting accessories for Cabling, tray work etc. Supply of Cables (signal/ communication cable etc) from Metering panel to RTU/ SCADA panel is also in the scope of bidder. Distance of 30 meters between Metering panel and RTU/ SCADA panel shall be considered for supply of such cable. Suitable Power cables (UPS/ Non-UPS) for metering panel and field Instruments shall also be supplied by the bidder.
- s) Junction boxes (as per the requirement of area classification) for power, signal, alarm, instrument and control cables with suitable cable glands are in bidder's scope.
- t) A structural skid complete with necessary drip pan, walkways, staircase, platforms, crossover, gratings, handrails for access for operation and maintenance are in bidder's scope. Details of skid assembly, supporting positions, Anchor bolt layout and equipment weights to be provided prior to equipment supply.
- u) Lifting lugs and spreader beam / frame, foundation Anchor bolts, copper jumpers for flanges for the skid, Stainless steel nameplate for each tagged equipment and component; All Tie-ins with flanged connections shall be in bidder's scope. Earth bonding system and earthing boss for metering skid are in bidder's scope.
- v) Inlet and Outlet matching flanges and Studs & nuts (for skid interconnection and Inlet & Outlet piping connection), suitable Gaskets shall also be supplied along-with the skid.
- w) The sun/ rain protection shed for Gas chromatograph field unit and for electronic instruments shall also be supplied by the bidder.

#### x) <u>Softwares</u>:

- a) Flow Computer shall have type approval for flow calculation (as per AGA-3) from the internationally accredited laboratory for custody transfer application.
- b) Vendor to provide the necessary hardware/ software (licensed in favor of CLIENT) for configuration of Flow computer and Gas chromatograph. Vendor to provide all the details and required soft-wares for SCADA communication.





- c) Vendor to provide the necessary software (licensed in favor of CLIENT) for accessing all data including flow computer and Gas chromatograph.
- y) The Vendor's scope of work shall also include:
  - a) Inspection and testing of all components, sub-assemblies, and complete assemblies of items manufactured at Vendor's works, and other sub-vendor's works in accordance with approved QA/QC procedure;
  - b) Shop assembly and hydro-test.
  - c) Factory Acceptance Test (FAT) for the complete package at Vendor's works as per approved FAT procedure.
  - d) Supervision of Installation, start-up and commissioning of the complete package at site. Site acceptance test (SAT) for the complete skid as per approved SAT procedure
  - e) Preparation for shipment, packing and delivery of all packages, equipment and material to site.
  - f) Installation assistance, Start-up and commissioning assistance at site.
  - g) Preparation and submission of all documents as per requisition with the bid and after award of contract.
  - h) Preparation and submission Final Documentation / Completion files as per this specification. Two copies (hard copy and soft copy) shall be submitted along with the complete system at stores / site for each skid.
- 2.3.1 Any work not specifically mentioned but otherwise required, as per statutory rules/ codes and standards/ specifications and/or for the completion and operation of equipment to the entire satisfaction of CLIENT/Mecon have to be done by the VENDOR without any commercial implications.
- 2.3.2 The scope of work also includes the mechanical and structural detailed design of the skid, procurement of materials, preparation of fabrication drawings, detailing of internals, fabrication, inspection and testing of the piping and structural items at fabrication shop, painting, internal coating if any, preservation, transportation and undertaking Guarantee for the equipment.
- 2.3.3 The scope of supply of metering skid also includes mandatory spares mentioned elsewhere.
- 2.3.4 The VENDOR shall assume single point responsibility for all aspects of the work. This shall include timely completion, liaison with CONTRACTOR, liaison with VENDOR of specified items, co-ordination of the work, quality and guarantee for the equipment.
- 2.3.5 Where parts of the package are subcontracted and purchased by the VENDOR, these become part of the Vendor's package and it is the Vendor's responsibility to ensure that the complete package complies with the specifications, codes and standards and statutory regulations.
- 2.3.6 Scope to include all instruments shown in the P&ID/ schematic, as within the Vendor's scope.
- 2.3.7 The Vendor shall be responsible for obtaining necessary approvals, authorization and certification from local Government / Local Statutory bodies, Authorized Inspector and Third-Party Inspection Agency as applicable.
- 2.3.8 The equipment shall be suitable for the site conditions specified. All components/ consumables used shall be new and of current manufacture.
- 2.3.9 In the event of any conflict between this specifications, data sheets, related standards codes etc., vendor shall refer the matter to the PURCHASER for clarifications and only after obtaining the clarification shall proceed with the manufacture of the items in question.
- 2.3.10 Vendor shall take single point responsibility for the engineering, design, certification, procurement, inspection, testing,





supply & performance of the Gas Metering System skids along with all instruments, equipment and valves of the skids, Gas chromatographs and Metering control panels based on the data sheets and the specifications furnished and taking into consideration successful operation, safety and the established international standards for the complete skids. As a part of skid design & engineering, the following shall be undertaken/ decided/ furnished by vendor:

- a) Calculation of metering system uncertainty based on the approved design and it shall be within +/-1 %. (Vendor shall submit the Calculation for overall system uncertainty including all components of the metering system). The calibration shall be done at minimum 7 points initially. The calibration shall be done at the following nominal flow rates: 0.025 qmax, 0.05 qmax, 0.1 qmax, 0.25 qmax, 0.75 qmax, and qmax. The calibration reports should also contain verification for minimum two flow readings after adjustment of factors as per AGA 3.
- b) Based on the approved design Sizing of flow meters, Pressure relief valves.
- c) Instrument ranges to meet the Process operating and design conditions
- d) Vendor to provide detailed Noise calculation and standard used and any assumption considered.
- e) Response and transportation time calculation for Gas chromatograph
- f) All the instruments/ equipments to be procured as per the approved vendor list of CLIENT/Mecon.
- 2.3.11 Typical instrument data sheets for Single chamber Orifice meter, Gas chromatograph, pressure relief valves, field transmitters, pressure gauges, and accessories indicate materials for body, internals etc. However, this does not absolve the Vendor of the responsibility for proper selection with respect to the fluid and its operating and design conditions. Proper sizing and selection of the pipe, isolation valves, Single chamber Orifice meter, and pressure relief valves and accessories are vendor's responsibility.
- 2.3.12 All the major items like valves (Plug, Ball, Globe and Check), Filtration, Single chamber Orifice meter, Flow computers, Gas chromatograph, Pressure Safety valves, Pressure & diff. Pressure transmitters, Temperature instruments, LEL detection system etc. shall be supplied from the vendor list attached elsewhere and the offered model of equipments shall have proven track record of successful operation for at least 6 months till bid submission date.
- 2.4 Vendor shall be fully responsible for proper integration of their supplied systems with CLIENT's SCADA (RTU) systems and shall provide all the technical details to CLIENT for configuration at SCADA end. Configuration in the supplied control panel shall be bidders' responsibility.

#### 3.0 DESIGN PHILOSOPHY

#### 3.1 GENERAL

- 3.1.1 This Document together with the attachments covers the minimum requirements for the design and engineering of metering skid complete with all accessories. Bidder shall be responsible for Design, engineering, sizing, selection, manufacture and/ or procurement, of materials, components and equipment necessary for complete package.
- 3.1.2 This document provides in detail, the minimum qualitative requirements of most of the instruments. For instruments, where no such requirements are indicated in this document, the bidder shall submit the same for CLIENT's/ Mecon approval. For metering system, vendor shall submit the complete design and engineering of the metering skid for custody transfer application. The total uncertainty calculation of the metering system to be submitted as per the design data prior to the fabrication of skid.

#### 3.2 CODES AND STANDARDS

3.2.1 Design and terminology shall comply, as a minimum, with the latest edition prior to the date of bid enquiry of following codes, standard practices and publications:

AGA American Gas Association, Gas Measurement Committee





	Report No.3 – Orifice Metering of Natural Gas. Report No.7 - Measurement of Gas by Turbine Meters. Report No.9 - Measurement of Gas by Ultrasonic Meters		
ANSI / ASME	American National Standards Institute/ American Society of Mechanical Engineers. B 1.20.1 Pipe Threads.		
	<ul><li>B 16.47 Steel Pipe Flanges and Flanged Fittings.</li><li>B 16.20 Ring Joint Gaskets and Grooves for Steel Pipe Flanges.</li></ul>		
ANSI/FCI	American National Standards Institute/Fluid Controls Institute 70.2 Control valve seat leakage classification.		
API	American Petroleum Institute		
RP 520	Sizing, selection and installation of pressure relieving system in refineries. Part-I - Sizing and selection Part-II - Installation		
RP 521	Guide for pressure relieving and depressurizing systems		
P 526	flanged steel safety relief valves.		
RP 527	Seat tightness of pressure relief valves.		
MPMS	Manual of Petroleum Measurement Standards.		
RP 551	Process Measurement Instrumentation. Part 1 - Process Control and Instrumentation		
RP 552	Transmission Systems		
S 1101	Measurement of Petroleum liquid hydrocarbon by Positive Displacement meter.		
S 2000	Venting Atmospheric and low-pressure storage tank.		
S 2534	Measurement of liquid hydrocarbons by turbine meter systems.		
S 6/U ASTM	Vibration, Axial-Position and Bearing-Temperature Monitoring Systems.		
BS 1042	British Standards Measurement of fluid flow in closed conduits		
BS-1042 BS 4368	Compression coupling for tubes		
BS-4800	Colours for ready mixed paint.		
BS-5308	Part-2 Specification for PVC insulated cables.		
BS-6364	Specification for valves for cryogenic service.		
BS-7244	Flame Arrestors for general use		
DIN-43760	Temperature vs Resistance curves for RTDs.		
DIN-19234	Electrical Distance Sensors; DC interface for Distance Sensor and Signal Convertor.		
IBK	Indian Boller Regulations.		
IEC IEC 60079	Electrical Annaratus for Explosive Gas atmosphere		
IEC 60085	Thermal Evaluation and Classification of Electrical Insulation		
IEC 60332	Test on bunched wires or cables. Part 3 Cat 1		
IEC 60331	Fire resistance characteristics of electrical cables		
IEC 60529	Classification of degree of protection provided by enclosures.		
	IEC 60534-2 Industrial Process Control Valves-Flow capacity		
	IEU 00384-2 Information and compensating cables talerances and IEC 60584-3 Thermocouples extension and compensating cables talerances and		
	Identification system.		





IEC 60751 Industrial platinum resistance thermometer sensors

IS	Indian Standa	Indian Standard		
	IS-5	Colours for ready mixed paints.		
	IS-319	Specification for free cutting Brass bars, rods and sections		
	IS-1239	Mild steel tubes, tubulars and other wrought steel fittings.		
	IS-1271	Specification of Thermal Evaluation and Classification of Electrical Insulation.		
	IS-1554-	PVC insulated (heavy duty) electric cables-working		
		Part I voltage up to and including 1100 V.		
	IS-2074	Ready mixed paints, air drying, red oxide- zinc chrome.		
	IS-13947	Degree of Protection provided by enclosures for low voltage switch gear and control		
		gear.		
	IS-2148	Flame proof enclosures for electrical apparatus.		
	IS-3624	Specification for pressure and vacuum gauges		
	IS-5831	PVC insulation and sheath of electric cables.		
	IS-7358	Specifications for Thermocouples		
ISA	Instrument Society of America.			
	S-5.2	Binary logic diagrams for process operations.		
	S-7.3	Quality standard for instrument air.		
	S-75.01	Flow equations for sizing control valves.		
	ISO 5167	Measurement of fluid flow by means of orifice plates, nozzles and venturi tubes		
		inserted in circular cross-section conduits.		
NEC	National Elec	National Electric Code.		
NFPA	National Fire	National Fire Protection Association.		
	NFPA-496	Purged and pressurized enclosures for electrical equipment.		
EN	European Standard			
	EN334 European standard for pressure regulators upto 100 bar			
	EN12186 Gas Pressure Regulating Stations for Transmission and distribution			
	EN14382 Safety systems for Gas distribution			

3.2.2 In general, Bidder shall carryout engineering as per IEC/ BIS standards.

Any other standard, if necessary, can also be referred by bidder during the execution of the job, without diluting the basic requirements, however with prior information to Owner/ Owner's Representative. In any case bidder must furnish a list of codes and standards other than those specified in this document, which shall be followed by them during engineering.

#### 3.3. INSTRUMENT DESIGN CRITERIA

- 3.3.1 No instrument air shall be provided by CLIENT. Suitable high pressure gas supply trains / systems two stage pressure regulation shall be installed by vendor for proper operation of the control valves. Special care to be paid for dealing the problem arising from possible condensation due to pressure reduction. A well proven system to be used for this application. I/P converters/ Electro pneumatic positioner, Filter regulators, Actuators, positioner shall be suitable for Natural gas (sour) application.
- **3.3.2** The Equipments and instrumentation selected for the metering skids shall be rugged in design and must be well proven in the hydrocarbon industry. Prototype design or equipment of experimental nature or design undergoing testing etc. shall not be selected and supplied. Following criteria must be applied before selecting a particular instrument item: "The instruments as being offered/ supplied should have been operating satisfactorily in hydrocarbon industry like Refinery, Petrochemical and Gas Processing Plant under similar process conditions for at least 4000 hrs. from the bid due date."
- **3.3.3** The complete station shall be designed in such a way that the mean gas velocity remains within 20 Meter per second. The complete skid shall be designed for 100% of the maximum flow capacity.
- 3.3.4 Instrument Requirements for classified area:
  - a) All electronic/electrical instruments and equipments shall be suitable for area classification as per IEC codes and shall be tested by any recognized authority like BASEEFA, FM, PTB, CMRI etc. and shall be certified by CCOE.





All the configurators, gas cylinders shall be certified by CCOE.

- b) Certified Intrinsically Safe (IS) equipment as per IEC-60079-11 shall be used, in general, in hazardous area. In case intrinsically safe equipment is not available, flameproof enclosures as per IEC-60079.01 may be considered.
- c) Junction boxes and accessories required for flameproof instruments shall also be certified flameproof.
   d) All non flameproof panels and cabinets installed in classified area shall be purged as per requirements specified in NFPA-496, as a minimum.
- e) Other type of protection as specified in IEC-60079 shall not be used.

#### 3.3.5 Statutory Approvals

- a) Bidder shall be responsible for obtaining all statutory approvals, as applicable for all instruments, equipments, calibration gas cylinders and control systems.
- b) In addition, equipments/instruments/systems located in the hazardous area shall be certified by the local statutory authorities for their use in the area of their installation. In general, following certification shall be given:
  - For all intrinsically safe/ explosion proof/ flameproof equipments/ instruments/ systems or equipments with any other type of protection allowable as per this package which are manufactured abroad and certified by any statutory authority like BASEEFA, FM, UL, PTB, LCIE etc. should also have the approval of Chief Controller of Explosives (CCOE), Nagpur.
  - For all flame proof equipments manufactured locally (indigenously), the testing shall be carried out by any of the approved test house like CMRI/ERTL etc. The equipment shall in addition bear the valid approval from Chief Controller of Explosives, Nagpur and a valid BIS license.
  - For all intrinsically safe equipment manufactured locally (indigenously), the testing shall be carried out by any of the approved test house like CMRI/ERTL etc. The equipment shall in addition bear the valid approval from Chief Controller of Explosives, Nagpur.
  - Custody transfer approval from NMI/ PTB/ NPL or equivalent body from the country of origin.
  - All the supplied gas cylinders shall have CCOE approval.
  - CCOE certificate for the foreign items shall be submitted during detailed engineering, however Appropriate approval certificates from the country of origin shall be provided with the bid
- **3.3.6** In general, intrinsically safe philosophy shall be followed for all transmitters. Bidder to note that external barriers shall be selected based on entity concept.

Typically, barrier selection must be made based on the following:

- i) Analog Inputs (4-20 mA): Series 5000 of MTL/ P&F KFD series/ Eqv.
- ii) Analog Outputs (4-20 mA): Series 5000 of MTL/P&F KFD series/ Eqv.
- iii) Proximity Inputs: Series 5000 of MTL/ P&F KFD series/ Eqv.
- **3.3.7** All instruments in the skid shall be certified for IEC Zone 1 Gas Group IIA/IIB, T3.
- **3.3.8** Instruments, which are not available as per their standard design from any reputed manufacturer as intrinsic safe, can be supplied in flameproof design. All such instruments shall be certified flameproof for the area classification and requirements indicated in clause 3.3.4 above.
- **3.3.9** Flame-proof (explosion proof) junction boxes as applicable shall be certified for IEC-Zone-1, IIA/IIB for all the classified areas for flame proof instruments.
- 3.3.10 The sizing for PIPES, Safety valves, Single chamber Orifice meter is bidder's responsibility.
- 3.3.11 Any change in instrument size or revision in line sizes because of sizing shall be carried out by bidder without any financial implications to CLIENT.





- 3.3.12 All line mounted temperature elements shall be RTD type as per IEC 60751.
- **3.3.13** All the instruments shall be provided with canopies of adequate size to protect instruments from direct rain & sunlight. All such canopies shall be prefabricated type.
- 3.3.14 All the field switches shall be suitable for Flameproof and weather proof enclosure contacts. The process switch shall be silver plated with contacts rated for rating of 30 V DC, 1 Amp. This requirement shall also be applicable for any other switch contact like push button, selector switches in the intrinsically safe service. The field switches shall be normally closed type and open to alarm. Switch contacts shall be SPDT. For all field switches (except Limit Switches), differential of switch shall be less than 60% of difference between set value & operating value.
- 3.3.15 Tube Fittings used for the installation of instruments shall be tested as per BS 4368 or equivalent standards.

#### 3.4 JUNCTION BOXES AND CABLING PHILOSOPHY

#### 3.4.1 Cable Routing

3.4.1.1 Single / multiple pair cables from instrument to junction box shall be through perforated trays. Cable glands shall be provided at instrument end and junction box end. Supply of Cables from Skid to the control panel and Cable glands (required for control panel end and skid end) for these cables shall be in bidder's scope. However, cable laying shall not be in scope of vendor.

#### 3.4.2 Junction Box

- 3.4.2.1 In general separate junction boxes shall be used for the following:
  - a) 4-20 mA DC signals (IS)
  - b) LEL detector signals
  - c) Metering Signal
  - d) GC Signals
  - e) Contact signals (Field switches, Limit switches, push buttons etc.)
  - f) Interlock and shutdown signals (Solenoid valves)
  - g) Power supply to various instruments (GC, Meters)

Separate JBs and cable are required for power supply to Meters and GC. The JBs shall have side entry for Branch cable/ single pair cable and Bottom entry for multi-pair cable entry. No top entry shall be considered.

- **3.4.2.2** The multi-cable entry for 6-pair JB and 12 pair JB shall be 1" NPT (F) and 1.5" NPT (F) respectively. Each junction box shall be provided with 2 multi-cable entries from the bottom of the junction box with one plugged with weather proof plugs.
- **3.4.2.3** Junction boxes, cable glands and accessories shall be weather proof in general. Slipper type PVC sleeves shall be used over cable glands for all cable entries in junction boxes to avoid water entry in junction boxes. In case of explosion-proof components used (only for packages), the respective junction boxes, cable glands and accessories shall be certified weatherproof and explosion proof.
- 3.4.2.4 Only one multi-cable entry shall be used in the junction box. The other cable entry shall be plugged.
- 3.4.2.5 The junction boxes in the field as well as in local panel shall be provided with sufficient number of terminals to terminate all the pairs of multi-cable (including spare pairs) and shields of individual pairs as applicable.

#### 3.5 INSTRUMENT PAINTING REQUIREMENTS

- 3.5.1 All instrument impulse lines (except SS 316 Tubing) and instrument structural items shall be painted by the bidder.
- 3.5.2 The painting/ coating shall be performed in totality for all instrument items such as:
  - a) All line mounting and equipment mounted instruments.





- b) All instrument impulse piping (except SS 316 Tubing)
- c) All instrument structural items like M.S. cable trays, instrument supports and tray supports, instrument stanchion, impulse line supports etc.
- d) All surfaces of GI items wherever repair has been carried out shall also undergo painting.
- e) All cabinets/panels, base frames which have undergone repair at site shall also be painted.
- f) The final coating on external surfaces shall be applied just before handing over the plant or commissioning of the plant.
- g) Name of the manufacturer, colour and quality of all types of primers and paints shall be subject to approval of the owner/owner's representative.
- 3.5.3 Painting of other equipments shall be as per Painting specifications attached elsewhere.

#### 3.6 POWER SUPPLY & POWER SUPPLY DISTRIBUTION

- 3.6.1 230 V AC shall be available for control Panel. Vendor to suitably install Rectifiers in the control panel for powering panel and field instruments both.
- 3.6.2 Bidder to note that 230 volts power supply shall be floating neutral type. Vendor to provide isolation transformer, in case vendor requires grounded neutral.
- 3.6.3 One feeder shall be provided by Bidder in the Control panel for connecting incoming power cable. Further distribution for all the vendor supplied instruments/ equipments in control room and at field are in Bidder's scope. 24 V DC power required for the barriers, isolators and field contact interrogation and relays etc. shall be provided by the vendor through dual redundant power packs. Power packs shall be sized with the safety factor of 1.5 of the actual load requirements.
- 3.6.4 Max. 800 W solar system power shall be provided to power up the Metering panel wherever considered. Vendor to select low powered instruments & design instrumentation system to accommodate total power consumption in Max. 800 watts. Vendor to submit power consumption calculation.

#### 4.0 PROCESS DESCRIPTION

- 4.1. The bidders shall design the skid considering the worst-case process condition scenario.
  - The PSV to be designed in such a way that the set pressure can be adjusted within the operating pressure range/ set points. Separate spring(s) shall be provided, if one spring is not sufficient to meet the requirement.
  - > 10% of the maximum flow capacity shall be considered as the minimum flow.
  - ▶ Design Temperature: -29 to 65 deg C.
  - ▶ Design flow: 100% of maximum flow capacity.
  - The maximum permissible pressure drop across the complete Single chamber Orifice meter-based metering skid is 3.0 kg/cm<sup>2</sup>g.

#### 4.2. Detailed Process conditions:

The quoted skid shall be designed for flow capacity of 100% of maximum flow mentioned above and shall be suitable for Gas compositions attached with bid documents. The Heating value of Natural gas (GCV) shall be 9200 to 9880 Kcal/SCM.

4.3. The bidder shall provide certification from Original Equipment manufacturer of gas flow meter/ calibrating agency that the performance/ accuracy and repeatability of the Meters shall be within the prescribed limit as mentioned in tender during operation with above mentioned Natural gas composition. The Material (and its composition) used in Metering skids shall be suitable for Natural gas of above-mentioned composition. The bidders shall provide written confirmation





from the Original Meter Manufacturer that the meters offered by the bidder shall perform within the specified accuracy limit with the above-mentioned Gas composition.

- 4.4. All the calculations for the sizing of the valves, meters etc shall be based on the design flow capacity with worst case Process Temperature and Pressure, however for mechanical strength of the equipments, design temperature / pressure shall be considered.
- 4.5. All the Equipments shall be able to withstand maximum/ minimum design Pressure and temperature. Worst case process pressure and temperature with design flow is to be considered for noise calculation of valves.

#### **5.0** JOB SPECIFICATION

## 5.1 GENERAL

This document defines the instrument Job requirements over and above various Standards attached along with this document.

The selection of type of instruments is Bidder's responsibility. For the instruments where type is already identified in the P & IDs / data sheets, the Bidder shall follow the same. However, during the engineering procurement stage if it is found that a different type of instrument is most suited for a particular application then the same shall be referred to Mecon for review, and if suitable, the changed instrument shall be supplied by the bidder. Bidder shall also be responsible for selecting and reviewing the type of instrument where specifically indicated in P& ID.

Instrument selection and specifications shall be carried out as per specification, typical instrument data sheets and Standard Specifications in general, as appended by special requirements specified here in.

The referred Standard specification, Technical Specification together with the P&IDs, Process data sheets, data sheets, standard specifications attached with the Package, defines the requirement for the design, engineering, manufacture, fabrication & assembly, integration, calibration, factory testing, supply, packaging, shipping and documentation including deliverables, statuary and other special approval, inspection, testing overall skid performance guarantee of Gas custody transfer. If required, then installation supervision, commissioning of the same at the same terms & condition of the contract/ tender.

#### 6.0 TESTING AND INSPECTION

#### 6.1 GENERAL

- 6.1.1 All pressure boundary materials shall have certified material test reports (CMTRs) or certificate of compliance per the design code. Certifications shall be to EN 10204 Type 3.2 for pressure parts and Type 2.2 for other parts.
- 6.1.2 Vendor shall submit the QAP & FAT Manual / procedure to Mecon for review and approval. A typical Mecon QAP and FAT Manual/ procedure is attached elsewhere.
- 6.1.3 All materials and equipment shall be factory tested before shipment in the presence of Purchaser's representative. No material shall be transported to site until all required tests have been carried out and equipment is certified as ready for shipment. Acceptance of equipment or the exemption of inspection or tests thereof, shall in no way absolve vendor of the responsibility for delivering equipments meeting the requirements of the specifications.
- 6.1.4 Vendor shall furnish the following
  - Material test certificate, Hydrostatic test certificate, certificates of radiography for all line mounted items/ instruments on the skid.
  - Certificates from statutory body for hazardous area approval for all electrical items mounted on the skid.
  - Calibration certificates, certificates for custody transfer, certificates for the conformity to the standards to be submitted.
  - All other certificates mentioned in individual general specification.
- 6.1.5 Supplier shall perform the usual standard tests to maintain quality control procedures. These test certificates shall be





submitted for review before starting inspection by Purchaser. Supplier shall be responsible for testing and complete integration of the system. Detailed procedures of test and inspection shall be submitted by the supplier for review before order and mutually agreed upon.

- 6.1.6 Vendor shall include inspection by Mecon/ third party personnel at vendor's shop. For this inspection, labour, consumable, equipment and utilities as required shall be in vendor's scope. Third Party Inspectors shall be deployed by the bidder. Vendor to propose minimum 03 TPI agencies, from which one shall be selected/ approved by Client/Mecon.
- 6.1.7 Other inspection and testing requirements shall be as per respective Standard specifications of various instrument items.
- 6.1.8 Qualification of the TPI appointed by the Vendor/ supplier and deputed for witnessing at various stages of Fabrication (like Hydro-test, FAT etc) shall be provided by the supplier well in advance for our review and confirmation. The TPI involved to witness shall have relevant experience. In case of any deviation / discrepancy observed while carrying out the inspection by TPI, bidder has to take clearance for the same by MECON / CLIENT. No TPI shall be deployed for our approval and FAT shall not commence without this.

#### 6.2 **SKID**:

The following tests shall be conducted for the skid

- a. Hydro testing for the integrated skid / individual pipe spools.
- b. Pneumatic Leak test of complete skid at 7 Kg/cm2g pressure.
- c. Skid functional testing considering metering, limiting and safety characteristics.
- d. Skid piping material testing and NDT of welds as per Piping material specifications. Radiography/ X-ray shall be carried out for all welded joints and vendor shall furnish test certificate for the same. Dye-penetration test certificate shall be provided for joints wherever radiography/ X-ray is not possible.
- e. Radiography/ X-ray, Charpy impact testing for line mounted instrument items such as pressure relief valves, Single chamber Orifice meter and meter runs etc.
- f. Testing and inspection requirements for skid piping materials shall be as per specifications attached else where in the bid package.

#### 6.3 Skid Equipments

The following tests shall be conducted:

- 6.3.1 Requirements of non-destructive testing like radiography, magnetic particle test, hardness test, hydro-test, Charpy test for pressure relief valves, Single chamber Orifice meter, Profiler and meter runs shall be carried out strictly as per following specification.
  - a) 100% radiography shall be carried out on all casting. Radiography procedure and area of casting to be radiographed shall be as per ANSI B16.34 and acceptance criteria shall be as per ANSI B16.34 Annexure B. Two shots shall be taken for each area to be radiographed, as a minimum.
  - b) Radiography/ X-ray shall be carried out for all welded joints and vendor shall furnish test certificate for the same. Dye-penetration test certificate shall be provided for joints wherever radiography/ X-ray is not possible.
  - c) Each Single chamber Orifice meter along with meter runs shall be subjected to hydrostatic test with a pressure of 1.5 times the design pressure.
  - d) Charpy impact test on each heat of base material shall be conducted as per A370 for all pressure containing parts such as body, end flanges and welding ends as well as bolting material for pressure containing parts. Unless specified otherwise, the Charpy impact test shall be conducted at 0°C. The Charpy impact test specimen shall be taken in the direction of principal grain flow and notched perpendicular to the original surface of plate or forging. The minimum average absorbed energy per set of three specimens shall be 27J with an individual minimum per specimen of 22 J.
- 6.3.2 Flow Meter, Meter run, Flow computer and accessories shall be offered for pre-dispatch inspection to CLIENT and / or CLIENT's representatives. Following tests, checks shall be conducted:
  - a) Physical/ dimensional checks and workmanship. Checking of meter tube roughness. Calibration including establishing linearity and repeatability over the entire range.





- b) The calibrating agency/ Laboratory / OEM of meter shall certify that the flow meter being calibrated shall work with specified accuracy/ repeatability with the actual gas composition mentioned elsewhere in the tender documents.
- c) Functional and simulation tests including checking of hardware and software for Flow computers including Single chamber Orifice meter and gas chromatographs with all its sub-systems in fully integrated configuration.
- d) All the panels along with all instruments mounted on it including flow computers, GC controllers, LEL monitors, barriers, isolators, power supplies, Printers, Modem, LED (clustered) Lamp as applicable and accessories etc.
- e) Review of all certificates and test reports.
- f) In the event the purchaser is unable to witness a test, the test shall anyway be completed by the vendor and documents for the same shall be submitted for scrutiny before shipment.
- g) All tests as per Standard specifications attached with this bid document.
- 6.3.3 Following tests shall be carried out by vendor at their works and test certificates shall be furnished:

Calibration/ test certificates for all instruments. Calibration test reports for flow meters duly signed and certified from the recognized international laboratories / statutory weights and measures authority. Statutory body certificates for instruments. Type test report for enclosure of all electronics/ electrical equipments. Radiographic / Charpy test certificates for flow meters and meter runs. Material test certificate for all line mounted instruments. Dimensional test report Certificates for custody transfer application and other certificates mentioned elsewhere.

- 6.3.4 In addition, vendor shall also refer the inspection & testing requirements of the standard specifications attached to this requisition and follow the same.
- 6.3.5 Vendor shall include in his scope the shop inspection charges including factory acceptance test (FAT) inspection by Client's representative at vendor's works.
- 6.3.6 Inspection of Pressure Relief Valves includes:
  - a. Testing to demonstrate set point accuracy and actuation time
  - b. Calibration certificate for Pressure relief valve set pressure
  - c. Seat tightness test for pressure relief valves (shall be conducted at manufacturer's shop and certificates shall be submitted)
- 6.3.7 The following certificates are required to be submitted during inspection for review before dispatch of materials:
  - Verification of certificates as applicable for the material certificates, NDT reports like radiography/ X-ray/ diepenetration/ MP, etc., statutory certificates (from CCOE) for intrinsic safety and explosion proof, certificates of conformity etc.
  - > Visual verification for quantity, quality and workmanship.
  - > Hydro testing and pneumatic testing.
  - > Functional and performance testing including calibration, accuracy, repeatability testing.
  - Seat leakage tests & hydraulic Pressure test, actuator cycling & fail condition tests for control valves
  - Set pressure, reseat pressure & seat leakage for pressure relief valves.
  - > Calibration report of PT, TT, Flow Meters with meter run and profiler
  - Radiographic test for thermo-wells etc.
  - Bidder must detail out performance specifications of each item which shall be verified by bidder or bidder appointed agency/ Owner/ PMC during factory testing.
  - > Inspection and testing requirements as per the respective standard specification shall be referred.
  - Bidder shall submit all test records / test results for records to purchaser as bound volume along with the test procedure for each test carried out.
  - Acceptable criteria for Radiography and other NDT requirements for all the instruments / instrument castings shall be inline with those specified in 'valve/ piping specifications' for the similar service. Valve specifications/piping





specifications have been attached elsewhere in this package.

- CCOE certificate shall be supplied by the bidder for all instruments installed in hazardous area.
- Custody transfer application certificate for meters & GC
- > Approval of skid design from approving agencies mentioned elsewhere.
- 6.3.8 Wherever inspection at manufacturer's shop is waived because of any reason, the bidder shall carry out the inspection at vendor/ sub-vendor's shop and Bidder shall forward these inspection reports for verification by CLIENT before dispatch. In no case, items shall be released without proper inspection/ verification.
- 6.3.9 The inspection and testing shall be carried out as per related specifications, international codes and practices/standards, approved documents and/or any other document attached along-with specifically suggesting extent of testing to be carried out at manufacturer's works.
- 6.3.10 Items for which 'Witness Inspection' is specifically exempted, manufacturer shall forward the test certificates as desired for review.
- 6.3.11 Testing and inspection for all items shall be carried out as per approved factory testing procedures. The material shall be dispatched only after obtaining written dispatch clearance. For items where no testing is witnessed by the purchaser test certificate shall be forwarded for review before dispatch of such equipment
- 6.3.12 Contractor must detail out performance specifications of each item which shall be verified by contractor or contractor appointed agency/ Owner/ PMC during factory testing.
- 6.3.13 Inspection and testing requirements as per the respective standard specification shall be referred.
- 6.3.14 Contractor shall submit all test records / test results for records to purchaser as bound volume along with the test procedure for each test carried out.
- 6.3.15 Acceptable criteria for Radiography and other NDT requirements for all the instruments/ instrument castings shall be inline with those specified in 'valve/ piping specifications' for the similar service. Valve specifications/piping specifications have been attached elsewhere in this package.
- 6.3.16 CCOE certificate shall be supplied by the contractor for all instruments, HHC, cylinders installed/ for use in hazardous area.

#### 7.0 DOCUMENTATION

- 7.1 Detailed drawings, data and catalogues required from the Vendor are indicated by the PURCHASER in vendor data requirement sheet. The required number of reproducible and prints should be dispatched to the address mentioned, adhering to the time limits indicated.
- 7.2 Final drawings from the Vendor shall include dimensional details, weight, mounting details and any other special requirements etc for the skids. All dimensions in general shall be in millimeters.
- 7.3 Vendor shall furnish all the required software, manuals necessary to test, operate and maintain the system. All the certificates, licensed softwares etc shall be provided in name of CLIENT.

#### 8.0 MISCELLANEOUS

#### 8.1 NAMEPLATE IDENTIFICATION

In addition to the instruments and equipment nameplate, the structural skid shall be supplied with a permanent, weather resistant, stainless-steel nameplate affixed to the skid, with the following details, as a minimum: Project title and number

- Owner and Owner name
- Equipment name and tag number
- Manufacturer's name and serial number
- Skid overall dimension data.





• Skid weight data.

Each skid and all the instruments in the skid shall have a S.S nameplate attached firmly to it at a visible place furnishing the following information:

- Tag number of the skid.
- Project Name with location: "CLIENT, -----(name of site/ station)"
- Inlet size (in inch) and Outlet Size (in inch) with class rating
- Min/ Normal/ Max. Flow capacity in SM3/Hr.
- Tag number of Instruments, JB as per purchaser's data sheets
- Body sizes with class rating in inches and the Valve Cg value or meter G rating
- Set pressure range and flow capacity of pressure safety valves
- Flow range in SM3/hr for Flow meters
- Rating for all the individual instruments
- Manufacturer's name and model number

All cable should have tag no. at JB/ panel end and Instrument end. All wires terminated inside skid and control panel should have identification mark, ferrules etc. (for the termination of supplied cables, Identification Tag no, Ferrule etc. shall be provided by bidder).

#### 9.0 SHIPPING & PACKAGING

- Vendor shall indicate the shipping and packaging methodology for Metering system / Each piece of skid along with its size, weight and no. of pieces.
- > Vendor shall try to fix as many types of equipment/ instruments as possible in skid and control panel before shipment.
- If any of the items are shipped loose, then it shall be properly packed inside metallic or wooden cartoon with identification tags painted on it. Loose materials, spares etc for any two skids shall not be mixed in one cartoon/package. A System of tagging, segregation to be followed for easy identification of parts and due care to be taken to avoid intermixing of some parts of different skids.

#### 10.0 PERFORMANCE GUARANTEE

- 10.1 The VENDOR shall guarantee that all work/ job will be performed in accordance with good and sound engineering and construction practices and within the requirements of this specification. The equipment, accessories and all the materials supplied by the VENDOR shall be free from defects, shall be suitable for the use for which they are intended and shall perform in accordance with the requirements of this specification.
- 10.2 The VENDOR shall furnish a guarantee for the entire skid package comprising of all of its component/ equipments including instruments, piping, valves, fittings, internals, etc., for a period of 12 months from the date of commissioning or 24 months from the date of receipt at CLIENT store/ site, whichever is earlier.
- 10.3 The VENDOR shall take single point responsibility for the complete skid, including the sub contractor supplied components, the proprietary equipment and components included in skid package and supplied loose in accordance with this specification.

#### 11.0 INSTALLATION AND COMMISSIONING OF METERING SKID

11.1 The VENDOR shall provide **qualified and experienced** personnel for installation, field-testing and commissioning of the equipments.

## 11.2 INSTALLATION REQUIREMENTS:

- > All instruments shall be accessible from grade or a platform for operation and maintenance.
- Bidder shall consider the, "MECON installation standards "as attached or equivalent for installation of each instrument. In case, any instrument requires a special installation or any instrument not provided with installation standard the bidder shall prepare the standard and get it approved from MECON along with other documents.
- > Impulse tube/ pipe of size 1/2" shall be used as impulse lines. The Impulse pipe/ pipe fittings for instrument installation





shall be as per the piping material specifications of respective process lines. Tubing when used between manifold and the instrument shall be, 12mm OD with SS316 material of construction as a minimum. The material selected shall be suitable for the process fluid conditions.

- Instrument in gas service are to be installed only above or at least parallel to the tapping with a slope in the impulse towards the tapping to achieve self draining condition.
- > Pressure Relief valves to be installed in line only after proper flushing of the lines.
- > No unions shall be used in impulse lines instead break flanges shall be used.
- Combination of Ball & Globe or plug is acceptable for drain & vent line, Gate valves in combination with Plug/ ball / globe valve shall be used for isolation, however combination of plug valve and ball/ globe valves shall be used for equalizing service.
- Minimum  $\frac{1}{2}$ " sized valves shall be used for Instrument isolation.
- Impulse piping is to be suitably painted (except SS 316 tubing)
- > Schedule 80 seamless pipe with at least 3000 lb rating fittings should be used as a minimum for impulse piping.
- > Tube fitting shall be double ferrule type.
- Based on the installation standards for each type of instrument, bidder shall prepare Bill of materials (BOM) which indicates the requirements of different materials for installation of each instrument. However, completeness of BOM is bidder's responsibility.
- Mounting of field instruments (if remote mounted)/ JBs on the stanchion or instrument support shall be at the height of 1.3M from the grade level / finished floor level.
- The installation and erection materials like, cables (signal, control, thermocouple extension and power), cable glands, junction boxes, instrument valves and manifolds, impulse pipe and pipe fittings, pneumatic signal tubes, instrument air line fittings and valves and cable trays required for installation of complete instrumentation shall be as per standard specifications enclosed in this tender. Power supply for different instruments/ equipments having different power/ voltage rating shall be installed in different JBs.

#### 11.3 LOOP CHECKING AND COMMISSIONING

- 11.3.1 Bidder is fully responsible for all work related to loop checking (inside the skid), including cable laying, tray work, dressing, identification, ferruling, calibrations, loop testing. However, cable laying from skid JB to control panel in control room is not in bidder's scope.
- 11.3.2 Bidder shall be responsible for commissioning of each & every equipments installed in the skid and control panel. (Functionality, simulation for these equipments shall be demonstrated during FAT).

#### 12.0 BIDS: -

#### Bidders Proposal shall include the following (as minimum):

- 12.1 Bidder's proposal shall include the detailed specifications for all items of filtration and Metering skids packages and Metering packages. The proposal shall include:
  - a) Make, Model number and detailed specification along with technical details and catalogue for each applicable item of metering skid like filter, flow meter, meter tube details, flow computer, pressure relief valve, gas chromatograph, pressure/ differential pressure transmitter, Temperature transmitters, hydrocarbon (LEL) detector and pressure/ temperature gauges, limit switches, control panels, Ball valves, plug valves, Globe valves, Check valves etc.
  - b) Sizing calculations and pressure drop calculations for inlet and outlet pipeline, filtration system, Flow meters and Pressure relief valves. Calculation for Gas velocities and Pressure drop across meter runs at maximum flow and minimum pressure to be submitted. The entire system (including all the components, equipments, Instruments) shall be selected and offered in such a way that the overall pressure drop across the entire skid shall be minimum. Detailed sizing calculation for individual components and pressure drop for entire skid (including diff. Pressure across individual components of metering skid) to be provided.


#### SCOPE OF WORK FOR METERING SKID



- c) Vendor to provide Detailed Noise level calculations (and standards used and assumptions considered, if any).
- d) All design and performance characteristics.
- e) P&ID for metering skid shall be as per Bid document attached.
- f) Completely filled Datasheet of all the supplied components/ items of skid with selected single and Model of equipment/ Instrument. (Multiple make/ models shall not be mentioned/ offered).
- g) Overall dimensions of each supplied items.
- h) Overall dimensions of each skid with proposed modules of each skid considered along with estimated dimensions.
- i) Weight of each skid.
- j) Foundation details (suggestive) of Metering skid
- 12.2 All units of measurements in vendor's specification sheets shall be same as those in purchaser's data sheets.
- 12.3 All material specifications for the various parts in the vendor's specification sheets shall be to the same standard as those in purchaser's data sheets
- 12.4 Vendor shall enclose catalogues giving detailed technical specifications, selection guide for decoding Model no. of offered equipments/ instruments and other information for Single chamber Orifice meter flow meters, flow computers, pressure relief valve, gas chromatograph, LEL gas detection system, pressure/ differential pressure transmitter, temperature transmitters, Ball valves, plug valves, temperature gauges, pressure gauges, portable configurator, laptop, receiver switches, barriers and isolators, Modem, Printers etc. covered in the bid.
- 12.5 Vendor's proposal including catalogues, drawings, operating and maintenance manuals etc. shall be in ENGLISH language ONLY.
- 12.6 Vendor shall submit the sizing, rating details & specifications of all the instruments, fittings and piping items, make & model, skid details etc., subsequent to award of contract. The relevant catalogue, technical literature shall also be furnished. MECON shall review the above and vendor to note that "No post order deviation shall be granted". Vendor shall change the make and/or models of items and specifications to meet the requirement of contract without any price and delivery implications.
- 12.7 The Single chamber Orifice meter, flow computers and online Gas chromatograph shall be approved for custody transfer application. Type approval certificate for offered make & model of Flow computer, Gas Chromatograph and the quoted make & model and size of Single chamber Orifice meter to be submitted along with the bid.
- 12.8 It is compulsory that bidder utilizes uniform data sheet formats enclosed along-with this document, for preparing specifications for various instruments, including those, which are being prepared by package/ sub package vendors. Additional information shall be provided in the given data sheets, if necessary. Items for which no format has been attached with the document, bidder may use standard ISA formats. Use of manufacturer standard formats shall be avoided.
- 12.9 Vendor's shall submit details of the only offered make and model of equipment/ instrument along-with the bid. Multiple makes and / or equivalent for any item shall not be mentioned in the bid.
- 12.10 The Bidder shall submit all filled and signed formats including Data sheet, List of items etc
- 12.11 The Bidder shall provide Type approval for quoted make and model.
- 12.12 The Bidder shall furnish details mentioned in Technical questionnaire attached with tender doc.
- 12.13 VENDOR shall include in the bid, list of specific deviations, separately, if any, to this specification and all attachment



#### SCOPE OF WORK FOR METERING SKID



thereof, otherwise, the quotation will be deemed to be in compliance with the specification requirements and subsequent claims for extra arising out of non-compliance with the specification will not be considered.

#### 13.0 SPARES PHILOSOPHY:

Mandatory spares, commissioning spares and materials required for Erection and commissioning of Metering skid shall be provided with Metering skid. In case the tender consists of more than one SOR item (for supply portion), Bidder shall provide all the relevant skid-wise spares for all the SOR items.

#### 13.1 Mandatory Spares: Shall be provided separately with each skid as per the following list.

#### The list of Mandatory spares to be supplied with each quoted skid.

- a) Filter element- 02 set. (One set means all the elements required for one stream)
- b) All type of gasket and studs- 1 set (One set means a pair of gaskets of all sizes & all the studs & nuts required for each size of flanged joint)
- c) O-ring for Filter end closure -02 set (one set means one stream)
- d) All type of IO barrier/ relay/ terminal (max. of 20% or 01 no.)
- e) GC spares like Column set, Solenoid Valve, Filter, Repair Kits etc. (for item nos. A.4 to A.7).

#### 14.0 ATTACHMENTS

- a) TECHNICAL SPECIFICATION FOR ORIFICE METER (TS No.: MEC / 05 / E5/ TS / ORFM)
- b) TECHNICAL SPECIFICATION OF CONTROL PANEL (REF.: TS No.- MEC/05/E5/TS/CP -030)
- c) TECHNICAL SPECIFICATION OF GAS CHROMATOGRAPHS (REF.: TS No. MEC/ 05/E5/ TS/ GC 030)
- d) PIPING MATERIAL SPECIFICATION

#### ANNEXURE

Annexure-I -	DATASHEETS
Annexure-II-	GAS COMPOSITION PARAMETERS
Annexure-III-	P & ID

- Annexure IV INSTALLATION DRAWINGS
- Annexure -V APPROVED VENDOR LIST
- Annexure -VI QAPs
- Annexure -VII FAT PROCEDURE/ FAT MANUAL
- Annexure VIII STANDARD SPECIFICATIONS

## **INSTRUMENTATION SECTION**

## TECHNICAL SPECIFICATION FOR ORIFICE METER

TS No.: MEC / 05 / E5/ TS / ORFM



0	14.03.2013		Ritish	Vikas	Rakesh
Revision	Date	Description	Prepared by	Checked by	Approved by

Page 39 of 644

INSTRUMENTATION SECTION SPEC. No. : MEC / 05 / E5/ TS / ORFM

Page 2 of 6



### TECHNICAL SPECIFICATION FOR ORIFICE METER

Rev. 0

#### General:

- 1. Single Chamber Orifice Fitting body shall be inline mounting with flow direction clearly marked on the flow meter body to ensure correct installation for custody transfer.
- 2. Single Chamber Orifice Fitting will house and accurately position an orifice plate for differential pressure measurement.
- 3. Single Chamber Orifice Fitting design shall ensure orifice plate changing safe, quick and easy with minimum downtime.
- 4. Single Chamber Orifice Fitting body shall be of Carbon Steel A352 LCC / ASTM A 216 Gr. WCB, Carrier & Cover Plate shall be 304 SS. Body and flange rating shall be as indicated in P&ID.
- 5. Each meter shall be based on a Single chamber orifice fitting with a square edged concentric orifice plate and flange pressure taps, designed and fabricated to be fully compliant with all requirements of API MPMS 14.3 and shall include both upstream and downstream meter tubes.
- 6. Meter tube and orifice fitting material shall be carbon steel. Orifice plate material shall be 316 SS.
- 7. Dual process taps shall be provided for pressure and differential pressure connections to a second set of instrument transmitters as back up to the primary instruments.
- 8. Meter Tube Diameter and Orifice Sizing Orifice bore to meter tube diameter ratio (ß ratio) shall be kept between 0.4 and 0.6 for all combinations of orifice plates and the meter tube.
- 9. VENDOR shall submit final orifice meter sizing calculation (using actual planned wall thickness orifice bore sizes) to PURCHASER for review and approval prior to final design and fabrication.
- 10. Vendor shall submit meter uncertainty calculation based on the final design of the meter system shall consider the effects of upstream & downstream piping beyond the actual meter tubes, of the selected field instruments & the inherent inaccuracies of the API/AGA database.

1	22.02.2024			
Revision	Date	Prepared by	Checked by	Approved by
				Page 40 01 644

MECON LIMITED DELHI	TECHNICAL SPECIFICA	TION FOR ORIFICE METER	
INSTRUMENTATION	SPEC. No. : ME	C / 05 / E5/ TS / ORFM	
SECTION	Rev. 0	Page 3 of 6	08 मकान 8001:2000 Company

#### **Orifice Plates:**

- 1. Orifice plate shall be thin square edged concentric orifice plate. The material of the orifice plate shall be normally SS 316, as a minimum. These shall be fabricated as per BS 1042 Part I.
- Sizing of orifice plate shall be carried out in accordance with ISO 5167. Other sizing methods like AGA Report No. 3, BS 1042. 'Flow measurement Engineering Handbook' by R.W. Miller shall be used when specified. Supplier shall size and select appropriate orifice plate for a maximum turndown ratio of 1:3. Orifice plate(s) shall be sized for a nominal differential pressure of 2500 mm water at maximum flow and a preferred beta ratio (β) of 0.55 (allowable range 0.4 to 0.6).
- 3. Upstream and downstream straight length shall be provided based on maximum d/D ratio of 0.75 in general. Where it is difficult to meet this requirement, the actual d/D ratio can be considered for reducing the straight length as permitted by the codes. The recommended practice shall be as per API-MPMS. Recommended Practices and AGA Report No. 3 mentions the consideration of flow strengtheners where straight runs are difficult to achieve, otherwise.

#### Fabrication:

- 1. Fabrication of the orifice plate shall be in accordance with AGA Gas Measurement Committee, Report No. 3, latest edition.
- 2. The Orifice meter shall in general comprise of a concentric square edged orifice plate designed for flange tap except for particular process requirements. (Eccentric, segmental or quadrant edged orifices may be used for special services such as slurry, wet gas or highly viscous service)
- 3. Where the orifice plate flow element is coupled with a force balanced or bellows type meter, the meter shall be with a built-in adjustable dampening device.
- 4. The Orifice Plate Assembly shall be capable of withstanding differential pressure equal to full line pressure without zero or calibration change.
- 5. Orifice plate beta ratio (orifice diameter/ inside pipe diameter) shall be between 0.4 and 0.6.
- 6. Vendor may choose to increase the length of the upstream and downstream piping attached to the orifice fitting to be beyond the minimum required by API.
- 7. Meter tube shall be equipped with a reliable method to ensure correct alignment

1	22.02.2024			
Revision	Date	Prepared by	Checked by	Approved by
				Page 41 01 644

MECON LIMITED DELHI	TECHNICAL SPECIFICA	TION FOR ORIFICE METER	
INSTRUMENTATION	SPEC. No. : ME	C / 05 / E5/ TS / ORFM	
SECTION	Rev. 0	Page 4 of 6	的 위 위 위 위 위 위 위 위 위 위 위 위 위 위 위 위 위 위 위

between two sections. Dowel pins and an annular grove / ring combination are two acceptable methods of achieving meter tube alignment.

- 8. Each upstream & downstream sections of the meter tube shall be made from one continuous length of select straight and round pipe.
- 9. After confirmation of compliance for roundness and straightness for the pipes, the orifice fitting shall be bored to match the internal diameter of the meter tube pipes.
- 10. After fabrication and assembly (including grinding of weld seams) of the meter tubes and fittings, the entire meter tube shall be honed. Meter tube inner surface finish shall be better than 6.35 microns (250 micro inches) as per API MPMS 14.3. Part 2 for all cases. While API allows for a surface finish of 7.6 microns (300 micro inch) when the ß ratio is less than 0.6, for the purpose of surface roughness measurement, the ß ratio shall be assumed to be higher than 0.6. This will allow the future possibility of selecting orifice plates with a higher β ratio without any additional uncertainty in the flow measurement.
- 11. Internal surface of the finished "meter tube" shall be free from any irregularities and defects such as encrustation, deposits, pits, grooves, ridges, offsets, scoring and protrusions that may result from casting problems, welding and other manufacturing activities.
- 12. Thermowell shall be located downstream of the orifice at the recommended distance from the orifice plate. Thermowells shall be designed for full flow and the resulting increased flow velocities, which may occur during maintenance procedures.
- 13. Internal surface of the finished "meter tube" shall be free from any irregularities and defects such as encrustation, deposits, pits, grooves, ridges, offsets, scoring and protrusions that may result from casting problems, welding and other manufacturing activities.
- 14. Thermowell shall be located downstream of the orifice at the recommended distance from the orifice plate. Thermowells shall be designed for full flow and the resulting increased flow velocities, which may occur during maintenance procedures.
- 15. Flange / NPT thread pressure taps shall be located 25 mm (approx. 1 inch) from the upstream and downstream faces of the respective faces of the orifice plate. The linear dimensional tolerance shall be 1 inch  $\pm 0.03$  inch as per API 14.3.2 using the recommended design factor of  $\beta = 0.7g$ .
- 16. All the pressure taps shall be equipped with full bore valves for isolation.

17. VENDOR shall consider guidelines from ISO 5167-2 in the layout and design of the

1	22.02.2024			
Revision	Date	Prepared by	Checked by	Approved by
		,		Page

MEC	CON LIMITED DELHI	TECHNICAL SPECIFICA	<b>ATION FOR ORIFICE METER</b>	
INSTRUMENTATION SECTION		SPEC. No. : ME	SPEC. No. : MEC / 05 / E5/ TS / ORFM	
		Rev. 0	Page 5 of 6	00 4011 000 Company
	station piping minimize ben as close as necessary, th profile.	connecting to the meter run ds and elbows upstream of t possible a fully developed ey shall be kept in the sam	s and shall maximize straight run the orifice meter to allow the flow flow profile. When bends and the plane to reduce their effects	to achieve to achieve elbows are on the flow
lange	:			
1.	Differential ra water. Typical meter	nges for orifice flow meters ranges for gas, steam or va	shall not exceed 5000 mm (200 por streams shall be as follows:	) inches) of
	STATIC PRES (Kg / cm2 g (F 0.35 to 2.5 (5- 2.5 to 6 (35 - 3 Above 6 (Abo	SSURE         DIFFERE           PSIG)         (mm of V           •35)         500 - 1           85)         1250 - 1           ve 85)         As per	ENTIAL PRESSURE Vater (inches) 250 (20 - 50) 2500 (50 - 100) AGA-3 guidelines	
2.	The range 25	00 mm (100 inches) is desira	ble wherever possible.	
3.	5.21.4.3 Orific 60% and 80%	ce meter ranges shall be ch of the meter range.	osen such that the normal flow	is between
<i>l</i> leter	Runs:			
1.	All meter runs measurement measurement	s (Upstream & Downstream) unless otherwise specifi committee report No. 3, late	shall be designed for flange tap ed and fabricated as per A est edition.	) differential GA - Gas
2.	Orifice meter piping diamet orifice, in line	runs shall not be less than 1 er shall be blown up to the with the applicable standards	<ul> <li>½" nominal pipe size. Where r required size for installing the s.</li> </ul>	equired, the meter run /
3.	The straight le the standard downstream sufficient stra straightening The straighte conditions. Th within 5 to 20	ength of upstream & downstr AGA-3. Connections sha straight pipe work, other ight run is not available for vanes may be used. These ning vane material shall k ne location of Temperature D internal diameter of the pip	ream pipe works shall be in acco all be made in either the u than the orifice flange tapping the given piping geometry, meto straightening vanes shall be as be 316 SS or better to suit t sensor (on the downstream on be run.	rdance with pstream or is. In case er runs with per AGA 3. he process ily) shall be

1	22.02.2024			
Revision	Date	Prepared by	Checked by	Approved by
	•		·	Page 43 of 644

#### **TECHNICAL SPECIFICATION FOR ORIFICE METER**

SPEC. No. : MEC / 05 / E5/ TS / ORFM

INSTRUMENTATION SECTION

Rev. 0

Page 6 of 6



Orifice Flow Meters (Typical)					
UNITS: Flow> L	.iguid m*3/hr	- Gas- Sm3/hr Steam - Kg/hr Pressure> Kg	/cm*2 G Temperature >Deg C Level/Length		
General /	1.	Tag No.	**		
Process	2.	Line No.	**		
	3.	Line size &Sch	**		
	4.	Service (Fluid)	CUSTODY TRANSFER of WET NATURAL		
	8.	Application Standard	ANSI, API 14.3, ASTM, ISO 5176,		
			AGA 3 latest reports / standards.		
			ASME 16.34 and ASME 16.5		
	5.	Size	*		
	9.	End Connection	Flanged (RF).		
	10.	Max Flow Bate	**		
	11.				
	12.	TemperatureRange	-20 Deg C to +50 Deg C		
	13.	Max. Working Pressure (Kg/cm2g)	**		
Meter					
	19	Meter type	Single chamber Orifice meter (fittings		
			with orifice plate, holder, meter run and		
			flow conditioner from same OEM)		
	20.	Diff. Pressure Tapping	Flanged taps: Inbuilt (1/2" NPTF) : to		
		11 0	be provided on flanges		
	21	Thermo-well	Inhuilt: to be provided		
	22				
		Facine & Finish	RC exercite		
	23.	Facing & Finish	AF serrale		
	24.	Motor maximum	*		
	25.	Diff Proceure at Max. Flow	* (Normal is 2500 mmWC)		
	20.	Poto Potio (d/D)	* (Normal range is 0.4 to 0.6)		
	27.	Enclosure protection	WEATHER PROOF IP65		
	20.	Material – Body	CS		
	31				
	32.				
	33.	-Orifice Plate	SS 316 / suitable for the service		
Service	42.	Oper, Visc (cP)/Density (Ka/m3)	**		
0011100	43.	System Pressure Drop	**		
	44.	Gas – Mol Wt.	**		
	45.	Compressibility Factor	**		
	46.	Cp/CV	**		
	47.	Area Classification	IEC ZONE 1 GR IIA, IIB T3		
Sealing	50.	Seal Bar O ring. Orifice plate seals			
	51.				
	53.	Unit	M3		
	55.	Transmitters (for Flow measurement)	1 PT, 1 DPT and 1 TT		
Important					
	58.	Manufacturer Report/ Certificate	To be Supplied along with each meter		
	59.	Warranty Certificate	To be Supplied along with each meter		
	60.	Operation And Maintenance Manual	To be Supplied along with each meter		
	61.	Make offered	*		
	62.	Model Offered	*		

Notes

"\*" By Vendor

"\*\*" As per P&ID

1. METER SHALL BE RATED FOR THE MAXIMUM DESIGN PRESSURE AS INDICATED ABOVE.

2. THE OFFERED METER MUST BE DESIGNED AS PER APPLICABLE STANDARDS.

3. THE ORIFICE METER SHALL HAVE PIPE TAPS (1/2" NPTF) FOE DP/ FLOW COMPUTER INTERFACE.

4. DP THROUGH METER SHALL NOT EXCEED 2500 mmWC. FOR OPERATING PRESSURE BELOW 4 KG/CM2G DP THROUGH ORIFICE METERING SYSTEM SHALL NOT EXCEED 1000 MMWC.

5. BIDDER TO PROVIDE COMPLETE METERING SYSTEM (EXCEPT FOR TRANSMITTERS/ FLOW COMPUTERS) FROM ONE MAKE ONLY.

FOR EACH METER, BIDDER TO PROVIDE REPORT/ CERTIFICATE WITH FLOW CALCULATION.
 METER CASING SHALL BE TAMPER PROOF & CORROSION RESISTANT OR STEEL SUITABLE FOR OUTDOOR INSTALLATION

LOCATION TOLERANCE FOR METERS SHALL BE AS AGA 3.

9. LINE BORE I.D TOLERANCE AND ECCENTRICITY REPEATABILITY SHALL BE IN CONFORMANCE WITH AGA-3 AND ISO-5167 LATEST EDITIONS.

12. AS A MINIMUM THE REPORT/ CERTIFICATE OF METER SHALL INCLUDE ID, OD, ID BORE TOLERANCE, INSTRUMENT TAP DIAMETER AND

LOCATION, PLATE SEAL TEST, SEAL PROTRUSION, ORIFICE ECCENTRICITY, BORE INSIDE DIAMETER, BORE ROUGHNESS, BETA RATIO, MAX. DIFFERENTIAL PRESSURE (AT DESIGN FLOW RATE)

1	22.02.2024			
Revision	Date	Prepared by	Checked by	Approved by

## **INSTRUMENTATION SECTION**

# TECHNICAL SPECIFICATION FOR CONTROL PANEL & ACCESSORIES

TS. No. : MEC/05/E5/TS /CP - 030



0	20.01.10				
Revision	Date	Description	Prepared by	Checked by	Approved by

Page 45 of 644

MECON LIMITED DELHI		SPECIFICATION FOR CONTROL PANEL & ACCESSORIES				
INSRUMENTATION		SPE	SPEC. No. : MEC / 05 / E5 / TS/ CP - 030			मेकान
	SECTION	Rev.	0	Page 2 of	?11	9001:2000 Co
		CONTROI	SPECIFICAT FOR L PANEL &	TIONS		
	GENERAL	S	ite of Instal	llation		
1.	Plant and Location	. :	As p	er P&ID		
2.	Consultant	:	MEG	CON LIMITED.		
3.	Location of control	l panel :	Con	trol Room and acco	essories.	
4.	Floor	:	Cone	crete (By client)		
5.	Air Conditioning	:	Yes	(By client)		
6.	Control Panels Qua	antity :	One	for each skid (As a	applicable)	
	<u>Scope of Work</u> :					
	<ul> <li>a) Supply of 1 No (As applicable) Lamps, MCBs, Supervisory Sy specification.</li> </ul>	o. of Metering P along with all a Relays, Barrier ystem, LEL Con	Panel (free s accessories s / Isolators ntroller, GC	standing, cubicle ty like Power Supply , Selector Switch, S C Controller, FC e	ype) for each c Distribution I Signal Multipl etc. as per this	uoted iter Box (PDB iers (SDC s Technics
	<ul> <li>b) Client will pro respectively. Fu panel vendor.</li> </ul>	wide 230 VAC, urther distribution	50 Hz. at on and neces	control room for C ssary surge protect	Control panel a ion device sha	nd utilitie ll be part o
	c) Separate Termi & all the signal	inal Blocks shall ls to be terminate	l be provide ed upto RTU	d for RTU signals U TB.	with 20% spar	e termina
	d) Quantities of co and SDC Carc approval/detail	ontrol panel acc ds etc.) shall be engineering.	essories (Pu e as per th	ash Buttons, Lamp e Technical Requ	s, Relays, Zen irements durin	er Barrier 1g drawin
	e) Mounting heigh	hts :				
	i) Miniatu instrum	are and subminianents (3 rows)	ature	Bottom row Middle row	1100mm 1350mm	
	ii) Electric	push buttons			700mm	

	MECON LIMITED DELHI		SPECIFICATION FOR CONTROL PANEL & ACCESSORIES		
I	NSRUMENTATION	SPEC. No. : MEC / 05 / E5 / TS/ CP - 030			- मेकान
SECTION			Rev. 0	Page 3 of 11	0.3001:2000 Comp
		INST C	ONTROL PANEL	CONSTRUCTION	
1.	Туре	:	Self-supported, F graphic.	Free Standing, enclosed cubi	cle and Non-
2.	Lighting	:	Required for insid	le panel with door switch.	
3.	Ventilation	:	Required with lou	vers backed by wire fly scree	en & fan.
4.	Doors	:	Glass doors in arrangement.	the front of the Panel	with locking
5.	Door width	:	Each max. of 60 Panel width is ind to accommodate However the de respectively.	Omm and shall suit width dicative only. The sizes shall the required hardware spec epth and height shall be 3	of the panel. l be sufficient cified in MR. 800 & 2200
6.	Cable Entry	:	Bottom, Cable Gl external armoure plugged.	lands shall be double compre ed cables. All unused ent	ssion type for tries shall be
7.	Receptacles	:	Required for 230V	VAC (UPS).	
8.	Painting	:	<ul> <li>The finish shall</li> <li>cleaning, surface</li> <li>high grade lacque</li> <li>coats of paint in p</li> <li>high stain finish.</li> <li>following:</li> <li>a) Siemens Grey</li> <li>b) Panel internal</li> <li>IS 352</li> <li>c) Channel Base</li> </ul>	include sand blasting, grind finishing by suitable filter and er with wet sanding between panel colour shall be given fo . Panel face final colour sh r (RAL 7032) I shall have a finish color of b	ing, chemical d two coats of n coats Two or Non-glossy all be of the pale cream – lack
9.	Channel Base	:	100X50X6 MM, 1	MS material	
10.	Name Plates	:	Front of Panel laminated plastic provided on the re	Instrument name plates she with white core. Namep ear of the panel also.	all be black late shall be

	MECON LIMITED DELHI		TFICATION FOR CONT		
I	NSRUMENTATION		SPEC. No. : MEC	05 / E5 / TS/ CP - 030	मेकान 🔉
	SECTION		Rev. 0	Page 4 of 11	9001:2000 Company
11	Dimensions & Mat	erial of	f Construction		
a)	Panel dimension	:	1200(W) mm x 21 channel base) Fina	.00(H) mm x 800(D) mm (Ind alized during detail engineering	cluding ng.
b)	Control Panel	:	3 mm thick CRCA steel/5.0 mm thick HRCA steel, Welded to frame		
c)	Side & Top plates	:	2 mm thick CRCA steel, Welded to frame		
d)	Door panel	:	Glass doors in the front of the Panel		
e)	Cable gland plate	:	3 mm thick CRCA steel		
f)	Anchor Bolt Size	:	By vendor		
g)	Lifting Eye Bolt	:	Required		
h)	Anti Vibration Pad	:	Required (15mm	thick rubber pad).	

## Note :

\*Panel shall be electrically isolated from base frame.

Ţ	MECON LIMITED DELHI	SPECIFICAT	FION FOR CONTROL PANEL & ACCESSORIES				
I	NSRUMENTATION	SP	РЕС. No. : MEC / 05 / E5 / TS/ CP - 030				
	SECTION	Rev	v. 0 Page 5 of 11 00 0001:2000 CONTROL				
12	WIRING						
a)	Туре	:	General purpose, Intrinsically safe				
b)	Wiring details						
	110 VAC UPS Wirin	σ					
	External to Cabinet	5	Min. 3x2.5 mm <sup>2</sup> /copper conductor PVC insulated				
			armoured				
	Inside the cabinet	:	Min. 19 Strands, 16 AWG copper conductor PVC				
	230 VAC Wiring (No	on UPS) :	3x2.5mm <sup>2</sup> copper conductor PVC insulated				
		= ~ , •	armoured				
	Low Voltage internal	:	Min. 19 Strands, 16 AWG copper conductor PVC insulated to cabinet				
c)	Signal Wiring						
	External to Cabinet	:	1.0 mm <sup>2</sup> twin twisted individual shielded overall shielded with overall drain <b>PVC</b> insulated armoured				
	Inside the cabinet	:	Multi stranded min. 1.0 mm <sup>2</sup> copper conductor PVC insulated twin twisted and shielded.				
d)	Terminal type	:	Screw clamp type with Pressure Plate				
,	Terminal size for sig	nal :	Suitable for min. 2.5 mm <sup>2</sup> size conductor				
	Terminal size for pov	wer dist. :	Suitable for min. 4.0 mm <sup>2</sup> size conductor and higher				
	Terminal block	:	as per actual cable sizes. Clip-on type				
e)	Wiring colour code						
	Power Supply		Hot – Red. Neutral - Black. Earth – Green				
	DC Wiring	:	Positive – Red, Negative – Black				
	Alarm System	:	White				
	Control & shutdown	:	Yellow				
	Analog signals (IS)	:	Light blue				
	Identification of Cabl	e					
	Termination	:	Criss – Crossing PVC tube ferruling.				
f)	Power Indication Lam	ips :	230VAC UPS – Red color, 230VAC NUPS – Red Color				

Γ	MECON LIMITED DELHI	SPECIFICA	TION FOR CON	TROL PANEL & ACCESSORIES	
11		S	SPEC. No. : MEC / 05 / E5 / TS/ CP - 030		- क्रु मेकान
	SECTION	R	ev. 0	Page 6 of 11	5001:2000 Com
	A) PO	WER SUPPL	Y DISTRIB	UTION BOXES FOR PANE	LS
1.	Function		:	Distribute power to Individual instruments	
2.	Mounting		:	Mounted at the panel inside	
3.	Door		:	Front single door	
4.	Painting		:	As per panel painting specification	ations.
5.	Dimensions and 1	naterial const	truction		
5.1.	Box dimension		:	By vendor.	
5.2	Box plate thickness	SS	:	3mm hot rolled steel.	
6.	Cable entry		:	By vendor	
7.	Accessories		:	600V Grade DPST MCB for Distribution and SP MCB for Suitable MCB for Control par	power barriers & nel Instruments
8.	Special requireme	nts	:	MCBs for incoming feeder.	
		B) I	BULK POW	ER SUPPLY UNIT	
1.	Input	: 230 V AC,	50 Hz.		
2.	Output	: 24 V			
3.	Rating	: By Vendor			
4.	Quantity	: Dual Redu	ndant as per re	equirement	

	MECON LIMITED DELHI	SPECIFICATION FOR CO	ONTROL PANEL & ACCESSORIES	
INSRUMENTATION SECTION		SPEC. No. : M	मेकान	
		<b>Rev. 0</b>	Page 7 of 11	0.9001:2000 Company
		SPECIFICATIONS FO	OR ACCESSORIES	
1.	MCB			
	Make		: HAVELL'S/ INDO ASIAN/ MDS.	
	Qty.		: As required + 20% spare	
2.	Lamps			
	Туре		: LED Clustered Type	
	Voltage Make / Model No		: 24 VDC • SIEMENS / L & T	
	Ouantity		: As required + 20% spare	
	Quality		. The required + 20% spare	
3	Relays			
	Type Contract Tyme		: Plug in relays	
	Contact Type		$\cdot$ 3 NO + 3 NC	
	Rating		: 24V DC. 5.0 A	
	Make / Model No.		: OEN/ OMRON.	
	Quantity		: As required + 20% spare	
4.	Space Heater			
	Rating		: 60W, 230 V AC with Thermostat.	
	Make		: KEC	
5	Power Supply Unit	t:		
	Make / Model No.		: ELNOVA/ APLAB	

SPECIFICATION FOR CONTROL PANEL & ACCESSORIES

INSRUMENTATION SECTION SPEC. No. : MEC / 05 / E5 / TS/ CP - 030

Page 8 of 11

### SPECIFICATION FOR SIGNAL DISTRIBUTION CARDS

Rev. 0

SL.NO.	DESCRIPTION	DETAILS
1.	ТҮРЕ	DUAL OUTPUT CURRENT ISOLATOR
2.	INPUT	4-20 mA D.C
3.	OUTPUT	TWO NOS. 4-20 mA ISOLATED OUTPUTS AS MINIMUM
4.	ISOLATION	BETWEEN INPUT & OUTPUT, BETWEEN INPUT & OUTPUT WITH POWER SUPPLY, BETWEEN OUTPUT & OUTPUT
5.	ACCURACY	$\pm 0.1\%$
6.	POWER SUPPLY	24 V D.C
7.	TRANSMITTER POWER SUPPLY	24 V D.C
8.	OUTPUT LOAD DRIVING CAPACITY	600 Ohms MINIMUM FOR EACH OUTPUT
9.	MOUNTING	DIN RAIL (BACK OF THE PANEL)
10.	МАКЕ	MTL/P&F
11.	MODEL NO	*

NOTES:-

\* - BY VENDOR

SPECIFICATION FOR CONTROL PANEL & ACCESSORIES

**INSRUMENTATION** SECTION

SPEC. No. : MEC / 05 / E5 / TS/ CP - 030



Rev. 0

Page 9 of 11

#### SL.NO. DESCRIPTION DETAILS 1. TYPE SHUNT DIODE SAFETY BARRIER 2. APPLICATION INSTRUMENT TYPE & 4-20 mA Analog I/P from 2 wire transmitter-(i) As required + 20% Spare QUANTITY 4-20 mA Analog O/P to I/P Converter - As (ii) required +20% Spare HAZARDOUS AREA CLASSIFICATION ZONE-I, GR. I, IIA & IIB, T3 3. 4. APPROVAL AUTHORITY FM/BASEEFA/CSA/SA/CMRS 5. 24 V DC SUPPLY VOLTAGE \* 6. FUSE RATING \*

\*

\*

\*

\*

0 TO 60° C

MTL/P&F

DIN RAIL (BACK OF THE PANEL)

SPECIFICATION FOR ISOLATING BARRIERS

NOTES:- \* --By Vendor

7.

8.

9.

10.

11.

12.

14.

15.

END TO END RESISTANCE

OPEN CIRCUIT VOLTAGE

SHORT CIRCUIT CURRENT

POLARITY

MOUNTING

MAKE

MODEL NO

OPERATING TEMP

SPECIFICATION FOR CONTROL PANEL & ACCESSORIES

INSRUMENTATION SECTION SPEC. No. : MEC / 05 / E5 / TS/ CP - 030



N

Page 10 of 11

#### SPECIFICATION FOR ISOLATING BARRIERS

Rev. 0

SL.NO.	DESCRIPTION	DETAILS
1.	TYPE	SHUNT DIODE SAFETY BARRIER
2.	APPLICATION INSTRUMENT TYPE & QUANTITY	4-20 mA Analog J/P from Temperature Transmitter
3.	HAZARDOUS AREA CLASSIFICATION	ZONE-I, GR. I, IIA & IIB, T3
4.	APPROVAL AUTHORITY	FM/BASEEFA/CSA/SA/CMRS
5.	SUPPLY VOLTAGE	24VdC
6.	FUSE RATING	*
7.	END TO END RESISTANCE	*
8.	POLARITY	*
9.	MOUNTING	DIN RAIL (BACK OF THE PANEL)
10.	OPERATING TEMP	0 TO 65° C
11.	OPEN CIRCUIT VOLTAGE	*
12.	SHORT CIRCUIT CURRENT	*
13.	MAKE	MTL/P&F
14.	MODEL NO	*

NOTES:- \* --By Vendor

SPECIFICATION FOR CONTROL PANEL & ACCESSORIES

INSRUMENTATION SECTION SPEC. No. : MEC / 05 / E5 / TS/ CP - 030

Rev. 0

Page 11 of 11



	SPECIFICATION FOR ISOLATING BARRIERS					
SL.NO.	DESCRIPTION	DETAILS				
1.	TYPE	SHUNT DIODE SAFETY BARRIER				
2.	APPLICATION INSTRUMENT TYPE & QUANTITY	Limit switches (SPDT snap action Micro type) - As required + 20% Spare				
3.	HAZARDOUS AREA CLASSIFICATION	ZONE-I, GR. I, IIA & IIB, T3				
4.	APPROVAL AUTHORITY	FM/BASEEFA/CSA/SA/CMRS				
5.	SUPPLY VOLTAGE	24 V DC				
6.	FUSE RATING	*				
7.	END TO END RESISTANCE	*				
8.	POLARITY	*				
9.	MOUNTING	DIN RAIL (BACK OF THE PANEL)				
10.	OPERATING TEMP	0 TO 60° C				
11.	OPEN CIRCUIT VOLTAGE	*				
12.	SHORT CIRCUIT CURRENT	*				
14.	MAKE	MTL/P&F				
15.	MODEL NO	*				

NOTES:- \* --By Vendor

## **INSTRUMENTATION SECTION**

# TECHNICAL SPECIFICATION FOR ON LINE GAS CHROMATOGRAPH

TS. No. : MEC/05/E5/TS/GC - 030



0	27.11.2019				
Revision	Date	Description	Prepared by	Checked by	Approved by

Page 56 of 644

MECON LIMITED DELHI	Technical Specification for		
INSTRUMENTATION.	SPEC. No. : MEC /	/ 05 / E5 / TS/ GC -030	मेकान 🔿
SECTION	Rev. 0	Page 2 of 11	9001:2000 Company

## **CONTENTS**

- 1.0 GENERAL
- 2.0 CODES & STANDARD
- 3.0 DESIGN AND CONSTRUCTION
- 4.0 NAMEPLATE
- 5.0 SHIPPING
- 6.0 **REJECTION**

#### TECHNICAL SPECIFICATION FOR ON LINE GAS CHROMATOGRAPH

#### 1.0 GENERAL

**1.1** These Online gas chromatograph systems shall comprise of following components:

- a) Sample conditioning system
- b) Analyzer unit and Programmable control unit
- c) Control room mounted Interface and display unit.
- d) Portable configurator and Printer
- e) Cables, cable glands, junction box, tubing, fitting etc.
- f) Calibration gas (with 03 years stability and traceability to NPL) cylinder and carrier gas cylinders

#### Sample Conditioning System:

- i. It shall be provided to avoid all undesirable gas contamination including liquid, oil mist etc and to ensure that the gas sample is supplied at suitable pressure and temperature. It shall have fast loop wherever required. It should have a regulator with isolation valves and PSV for regulating the sample pressure & flow. The sample flow should be controlled by a needle valve and flow should be indicated through Rota meter. The enclosure for calibration gas system should have heating system with insulating jacket (uniformly to all parts of cylinders) so as to maintain minimum temperature required for stability of certified composition.
- ii. The sampling system shall be engineered, designed, fabricated and furnished completely assembled as an integral part of the Gas Chromatograph Analyser or as a separate Unit. The sampling system shall be designed on the basis of the utilities available as indicated in the job specifications.
- iii. Each sampling system component shall be capable of being removed without disassembling the entire system.
- iv. Sampling system shall be designed for easy integration with the Gas Chromatograph Analyser in the field.
- v. Sampling system design shall be such that the sample drawn for analysis is truly representative of the process stream. Extra care shall be exercised in the selection of the various components forming the sampling system.
- vi. The sampling system shall be designed to move the sample from the process to the Gas Chromatograph in the shortest possible time. Due consideration shall be given to the size of the sample tubing to avoid possible plugging. Any requirement of steam tracing of sample tubing shall be indicated in the bid.
- vii. Vendor shall clearly indicate the details of sample disposal to be provided by purchaser. When details of sample return points are indicated in purchaser's data sheet, sampling system shall be designed to meet the requirement.
- viii. Sampling system shall consist of one or more of the following components as required by the service conditions shown on the instrument data sheets : filters (coarse and fine), pressure regulators, relief valves, flow indicators, flow controllers, temperature indicators, scrubbers, heaters, coolers, dryers, sample pumps, aspirators etc. Vendor in the bid with proper reasoning shall include any additional requirement.
- ix. Where purchaser's data sheets indicate the possibility of polymer formation or presence of excess polymers, Vendor shall offer suitable design features to minimize plugging because of polymer residue.
- x. Filters shall be always dual and it shall be possible to remove and replace one of the filters without upsetting the operation of the Gas Chromatograph.

MECON LIMITED DELHI	Technical Specification for		
INSTRUMENTATION.	SPEC. No. : MEC	/ 05 / E5 / TS/ GC -030	मेकान 🔉
SECTION	Rev. 0	Page 4 of 11	9001:2000 Company

#### Analyzer Unit and Programmable control Unit

The function of this unit will be to analyze the gas sample, do the calculation of Heating Value, Relative Density (specific gravity), Compressibility Factor etc. and provide information to interface / display unit. The technical details of this unit are given as under:

- i. Type: Continuous On-line Gas Chromatograph microprocessor controlled.
- ii. No. of streams: single stream
- iii. Column: Micro pack / MEMS column with high resolution and low carrier gas consumption, Column size (length / diameter): By vendor
- iv. Detector: Thermal conductivity detector
- v. Carrier gas:
  - a) Ultra-pure Helium gas along with pressure regulators and Rota Meter flow-indication. The consumption of carrier gas should be as low as possible.
  - b) Flow rate: the minimum possible (to be defined by the bidder)
- vi Cycle time: Operator / user assignable (5 minutes maximum).
- vii. Analyser:
  - a) The design of Analyser should be based on solid-state electronics with protection from industrial EMI /RFI Interference.
  - b) Inputs / Outputs: It should communicate digitally to interface / display unit to send the analysis related information.
  - c). Analyzer & analyzer electronics should be supplied with prefabricated / installed tube connection etc., so that field tubing etc. to be done as minimum as possible at site for connection of sample line & carrier gas and calibration gas cylinders
  - d) Functions: Control of all analysis functions in a cycle from operation of sampling valves to detection of peaks and precise calculations for peak measurements, area integration and internal normalization. Average values of analysis and computed results for 8 and 24 hours shall be available automatically or on-demand. Also generation of gas analysis report (scheduled or on-demand) and average report (8 hourly average, 24 hourly averages) for gas composition, calorific values and specific gravity.
- viii. Calibration: Automatic calibration with user selectable time intervals or manually on demand as & when required. To accurately determine the relative response factors with/without single component calibrations such as high purity methane. Vendor shall indicate the accuracy and repeatability of the measurement achieved by the calibration method adopted. Also during calibration period the stream composition data of previous analysis shall be used in Flow computer. Also if the calibration results are out of range then it shall be alarmed. The composition data during alarm condition shall be rejected and shall not be used for Flow computation.
- ix. Calculations : The Analyzer should have the facility of calculating Real Net dry Heating Value, Real gross heating value, Real relative density, base compressibility factor at base pressure = 1.0332 kg/cm2 abs, base temperature=15.56 deg C as per latest standard of ISO-6976 /GPA2145-GPA2172 (user configurable) & optionally ASTM D3588 if available as standard with the GC. The tables used for these calculations should be user configurable. Initially the method table shall be configured with base pressure 1.0332 kg/cm2 abs and base

MECON LIMITED DELHI	Technical Specification for	On line Gas Chromatograph	
INSTRUMENTATION.	SPEC. No. : MEC / 05 / E5 / TS/ GC -030		मेकान क
SECTION	Rev. 0	Page 5 of 11	9001:2000 Compa

temperature 15.56 Deg.C and the heating value shall be in Kcal/scm. Provision to change to any other configuration/ standard as per site requirement

- x. The analyzer should have the facility for remote communication through CLIENT's SCADA/ RTU system.
- xi. The analyzer should be supplied with a calibration gas stand equipped with a heating system/ plate with insulating Jacket so as to ensure that calibration gas cylinder is uniformly maintained at minimum required temperature required for stable composition. Please note that GC system shall be installed in the Hazardous area. Bidder should ensure that the heating system comply the relevant standard requirement.
- xii. The analyzer should have the facility for storing at least last 480 analysis reports (last 36 hrs continuous analysis) and should have the facility for giving the average report for last 24 hours.
- xiii. Diagnostics: Online automatic diagnostics for self checking of system integrity periodically for giving alarms.
- xiv. Area classification: IEC Zone 1 Gas Group IIA / II B Temp Class T3
- xv. The offer GC should have extensive data storage/retrieval capacity to keep valuable data / results for future use. The GC archive shall store minimum most recent 1200 cycles & last 400 calibration and average result of last 64 days. Details have to be furnished in the offer.
- xvi. The Gas Chromatograph shall have retention time reproducibility character with analyzer programs stored in EEPROM and with battery backed CMOS memory. The GC shall store the data (of 3 shifts per day of 4 minute cycle average up to a month) for minimum one month. Vendor shall furnish the maximum capacity of the GC to store the data in the offer.

Type of enclosure / protection: Sample conditioning unit, analyzer, analyzer electronics shall be mounted in an enclosure certified as weather proof (NEMA-12) and explosion proof from statutory body. The field mounted equipments shall be suitable for ambient temperature 0-60 degree Celsius and under a shed of suitable size and design, so that there should not be any direct environmental effect.

A suitable selector switch JB for Isolation of Power supply to Analyser Field unit shall be provided. All Power and communication cable for Analyser/ GC shall be provided by bidder.

#### Communication Interface / Display unit:

The main function of this shall be to access all the gas analysis related parameters from field analyzer unit, send the required O/P in Analog/Digital form to Flow Computers and provide the display in control room for the composition of all the components. It should also provide the facility for calibration of analog outputs. It should have separate processor and memory boards, which should operate independent of analyzer. It can be configured to support various options selected by the user via keypad or configurator.

- i. Type: A Microprocessor based unit with alphanumeric / CRT display for input, output signals and analysis results and other parameters.
- ii. Inputs / Outputs:

DE	LIMITED LHI	Technical Specification for	On line Gas Chromatograph	
INSTRUMENTATION. SECTION		SPEC. No. : MEC / 05 / E5 / TS/ GC -030		मेकान
		Rev. 0	Page 6 of 11	9001:2000 Comito
	a) b) c)	Operator inputs via keyboard an Potential free contact output for Four nos. 4 to 20 mA (isolated) Outputs will be used for: (and sh I Gross calorific value	d / or configurator/ Laptop throu malfunction. output signals for driving 600 nall be user configurable):	ıgh serial port Ohms load resistanc
		III. Mole % $CO_2$ IV. Mole % $N_2$ Each output should be u	iser assignable.	
	d)	Dedicated 1 No. RS 485 / 232 (MODBUS ASCII) serial link f SCADA: - Complete gas composition - Calorific value (GCV and NCV - Relative density (specific gravi - Compressibility factor	C for metering supervisory sys for transmission of the followin V) ity)	tem & 1 No. RS 48 g signals to CLIEN
	e)	RS485/RS-232C communication Flow Computers of various ma FMC, OMNI, PIETRO etc. insta	n ports with modbus protocols ake like Instromet, RMG, Dar alled in the same control room.	to communicate wi iiel, Bristol Babcoc
	f) g)	It should communicate to CLIEN Complete details and document structures, frame structures, syn data addressability and interface successfully implement a seria owner).	NT SCADA through serial commutation with respect to protocol nchronizing / timing signals, ne software driver details shall be al LINK & GSM link with the	nunication. details with messag nemory locations f e furnished in order e RTU/ SCADA (b
	h)	Separate port for Laptop and Pri	nter (if provided against integra	l printer)
iii.	Interfac	e:		
	a) b)	Digital link with analyzer electro Also the communication betw display unit shall be either throu	onics in the field. veen analyzer & control roon gh OFC or copper cable.	n mounted interfac
iv.	Printing average also be	: The Color Desk Jet Printer report in a format indicating da for Alarm Printing, Alarm Repo	shall be for printing the Gas tta, time & day, type of analysis rts, Event log etc.	Analysis Report, a etc. The Printer sha
v.	Mountir	g: All mounting accessories to b	be supplied by vendor.	
vi.	All the a retrieved provisio	analysis results shall be stored in I through interface unit to po n to take print out of chromatogr	n the Field Unit / control room ortable configurator. The conf rams, results etc.	unit and it should l igurator should ha
vii.	Program	mable control unit can be eithe	r part of the analyzer in the fie	ld or it can be part

MECON LIMITED DELHI	Technical Specification for	On line Gas Chromatograph	
INSTRUMENTATION.	SPEC. No. : MEC / 05 / E5 / TS/ GC -030		मेकान 🔉
SECTION	Rev. 0	Page 7 of 11	9001:2000 Compa

#### **PORTABLE CONFIGURATOR:**

- The Portable configurator should have at least Pentium-4 processor, 10 GB hard disk, 128 MB internal RAM, serial port, CD Combo Drive, 17" LCD display. Other details are as under:
  - a) The portable configurator should provide the facility to store all the analysis results made in the past. They can be retrieved and printed out at any time. This PC will be used for following functions:
    - i) Any modification in the programs, which will be used by the analyzer.
    - ii) To Retrieve all data from the analyzer including the chromatogram and should have the facility to configure/ change user related data and should have diagnostic features to face any trouble shootings.
  - b) Power Supply: 230 V AC, 50 Hz with allowable voltage variation of +/- 10%.
  - c) All the software and operating system (Minimum Windows Vista/XP) shall be licensed in the name of CLIENT.
- ii. Cabling: Vendor to provide required interconnecting cables.

The enclosure for sampling system, Analyser & Electronics shall be complete in all respects, ready for quick installation at site.

1.2 Each gas chromatograph shall complete the analysis of sample in a maximum time of 5 minutes to a repeatability  $\pm 0.05\%$  ( $\pm 5$  Kcal at 10,000 Kcal/SCM), based on ambient temperature at site between 0°C to 50°C. The computed GHV shall not exceed an uncertainty of  $\pm 0.5\%$ . The performance and repeatability of each analyser employed in the metering system must be third party/ factory tested in an Environmental Test Chamber, from 0-50 deg. C for at least 24 hours to verify repeatability data as a part of the inspection. Manufacturer shall provide a test certificate showing, 24-hour test data with calibration gas analyzed live under flowing conditions for few GC as a sample along with the bid. Ambient temperatures will be included along with repeatability data.

- **1.3** Vendor to note that the GC skid in field consisting of analyzer field cabinet, sample conditioning system, calibration (3 years stability with traceability to NPL or equivalent) & carrier gas cylinder racks etc. shall be assembled together and mounted on a single steel structure by vendor. This skid shall be suitable for floor mounting in a rain/sun shed provided by the Vendor.
- **1.4** Gas chromatograph shall be compact, rugged and of proven design to meet the specified needs of energy measurement for natural gas.
- **1.5** Vendor to indicate the distance limitation, if any, for the communication cable. Cable shall be armoured. Vendor shall furnish complete specification of these cables and supply the same.
- **1.6** All sample tubing and the tubing between the calibration & carrier gas cylinders and analyser shall be included in the scope of supply. The tubing shall be SS316. Vendor to indicate that the distance limitation, if any, for the sample tubing.
- **1.7** Provision shall be made in the sample handling system for manual injection of grab sample in the analyzer.
- **1.8** Vendor to supply Qty of carrier gas (with 99.99% purity) suitable for 2 year continuous operation of Gas Chromatograph with regulator mechanism and automatic switch over including pressure gauges on cylinder with 3 way valve for auto switching over of standby cylinder. Gas volume per cylinder is 7 M<sup>3</sup>. The certificate for purity to be submitted from a reputed third party.
- 1.9 The vendor shall supply Qty of calibration gas (composition as mentioned in process data elsewhere) suitable for 2 year continuous operation & with regulator mechanism. A recognized laboratory must certify the calibration gas. The calibration gas specification must be approved by the purchaser prior to purchase. The calibration gas shall have a minimum certification accuracy of  $\pm 0.2\%$  for components (mole %) greater than 10 % and  $\pm 0.5$  % for components (mole%) between 1 % to 10% and  $\pm 1.0\%$  for components (mole%) below 1%. The calibration shall be traceable to NPL or

MECON LIMITED DELHI	Technical Specification for	On line Gas Chromatograph	
INSTRUMENTATION.	SPEC. No. : MEC / 05 / E5 / TS/ GC -030		मेकान
SECTION	Rev. 0	Page 8 of 11	9001:2000 Company

equivalent (the traceability certificate to be submitted) with 03 years stability. Gas volume per cylinder shall be approximately  $2.13 \text{ M}^3$ .

- **1.10** Vendor to include 1<sup>1</sup>/<sub>2</sub>" full bore ball valve with flanged ends for probe connections at the tapping point. Fully retractable probe shall be supplied along with suitable probe housing. However online extraction tool is not envisaged.
- **1.11** Vendor to include for electrical tracing of the sample tubing, in case required for the optimal performance of the system for the given operating conditions. Electrical tracing, if required, shall be certified from statutory body for the given area classification and shall be included.
- **1.12** Vendor shall note that there is no low-pressure process line to which the bypass loop can be connected and hence the same shall also be connected by vendor to the common atmosphere vent header (approximately at 3 meters height from ground) to be supplied by vendor.
- **1.13** Gas chromatograph shall be serially interfaced to all the Flow computers simultaneously for transfer of gas compositions etc. Vendor shall be totally responsible for this interface and the same shall be demonstrated by vendor during FAT.
- **1.14** Vendor shall note that the test samples to be used during inspection for testing and calibration shall be certified for the concentration of the gas components and shall have traceability. Vendor shall furnish the certificate at the time of inspection.
- **1.15** Vendor shall provide an interconnection diagram for all components from sample take off up to the analyser and also showing interconnection of analyser with programmer in control room.
- **1.16** Material of construction shall be SS316 as a minimum for sampling system, isolation valves, instrument fittings, vent lines etc. and shall be included in scope of supply by vendor. All fittings shall be double compression type Swagelok or equivalent.
- **1.17** Sample shall be returned to sample return header located in the analyser cabinet in the field. From sample header, sample shall be vented to atmosphere at a safe height. All interconnection, tubing/fittings /piping/ valves between analyser and return header shall be in vendor's scope of supply.
- **1.18** Vendor to note that separate cable entries shall be provided for power supply and signal cables in Gas Chromatograph system. All cable glands shall be weather proof and explosion proof to NEMA 4 and NEMA 7. Cable entries and cable gland sizes shall be in NPT standards. Vendor shall provide cable entry sizes and cable gland sizes as specified in the respective data sheet.
- **1.19** All the electrical items of the gas chromatograph system in the field (sampling system, analyser, electronics, isolation transformer, power switches, flame proof Lighting, junction box, glands etc) shall be certified flame proof suitable for the specified hazardous area (IEC Zone-1, Gas Gr. IIA & IIB, Temperature T3) from statutory body such as Nmi, FM, CENELEC, UL, BASEEFA, PTB etc. and shall be suitable for ambient conditions as given else where.
- **1.20** Vendor to mount the analyser, sampling system with explosion proof NEMA 7 enclosure which shall be weather proof to NEMA 4/IP 55 with suitable mounting arrangement for gas cylinders.
- 1.21 Vendor to note that for analyser, programmer and accessories shall be powered from control panel, where at one point supply of 230/110 V AC  $\pm 10\%$ , 50 Hz  $\pm 3\%$  will be given by owner. Further rectifier / transformer if required shall be provided by the vendor in their panel.
- **1.22** Vendor to ensure that Gas Chromatograph system shall not be damaged because of black outs / brown outs. Vendor to indicate steps to be taken for fail safe operation under power failure.
- **1.23** The application programs to be stored in RAM for 6 months battery backup and also for reloading without any external storage device.
- **1.24** All controller software and firmware with licenses shall be in the name of CLIENT. The system shall be supplied with an undertaking from the vendor to upgrade all software and firmware to the latest version and to incorporate all algorithm corrections and changes in line with latest industry standards. Such upgrade shall be supplied and executed free of charge by the vendor for ten years from the date of supply of the system
- **1.25** Vendor to indicate time period between successive calibrations of analyser and time period for routine preventive maintenance.
- **1.26** Vendor shall submit the certificate confirming that the natural gas analysis done by the on line GC confirms to ISO 6974-4 /IS 15130-4.

MECON LIMITED DELHI	Technical Specification for	On line Gas Chromatograph	
INSTRUMENTATION.	SPEC. No. : MEC	SPEC. No. : MEC / 05 / E5 / TS/ GC -030	
SECTION	Rev. 0	Page 9 of 11	9001:2000 Company

- 1.27 Gas Chromatograph shall transmit data to remotely mounted flow computers through serial link. Required hardware/ software for serial communication Flow computers shall be provided by the bidder. Bidder to note that the GC shall also provide the gas data to flow computers (2 nos.) supplied with SOR item. All the requisite hardware / software shall be supplied by the bidder.
- 1.28 The on line Gas chromatograph shall be certified for custody transfer application from Nmi/ PTB.

#### 2.0 CODES & STANDARD

The related standards referred to herein and mentioned below shall be of the latest editions prior to the date of the purchaser's inquiry.

IEC – 79	Electrical Apparatus for Explosive Gas Atmosphere.
IEC – 529	Degree of Protection provided by Enclosures.
ANSI/ASME	American National Standards Institute/ American Society of Mechanical
	Engr.
B. 1.20.1	Pipe Thread
B 16.5	Steel Pipe Flanges and Flanged Fitting
B 16.20	Ring joint Gaskets and Groove for Steel Pipe Flanges
API	American Petroleum Institute MPMS (Relevant Portions)
CENELEC	EN 50020 Electrical Apparatus for Potentially Explosive
	Atmospheres
DIN 50049	Document on Material Testing
IS- 2147	Degree of Protection Provided for Enclosures
IS - 2148	Flame proof Enclosures of Electrical Apparatus
IEC - 801	Electromagnetic compatibility for Industrial Process
	Measurement and Control Equipment.

#### 3.0 DESIGN AND CONSTRUCTION

- a) In general Gas Chromatograph and sampling systems shall be designed and constructed in accordance with API MPMS (Relevant portions).
- b) Process Stream sampling shall be continuous.
- c) The Gas Chromatograph performance shall be within specifications when the supply voltage varies by  $\pm 10\%$  of specified value and supply frequency varies by  $\pm 3$  HZ of specified value.
- d) Unless otherwise specified material of all components wetted by sample shall be ANSI 316 SS.
- e) Programmable control unit can be either part of the analyzer in the field or it can be part of the control room mounted interface / display unit.
- f) Unless otherwise specified, the following shall govern :
  - i. Threaded connections shall be NPT to ANSI / ASME B 1.20.
  - ii. Flanged connection shall be as per ANSI / ASME B 16.5
  - iii. Grooves of ring type joint flanges shall be octagonal as per ANSI /ASME B 16.20.
  - iv. Flange face finish shall be serrated concentric to paragraphs 6.3.4.1, 6.3.4.2, and 6.3.4.3 of ANSI / ASME B 16.5. The face finish as specified in data sheet shall have serration as follows :

Serrated	:	250 to 500 AARH
125 AARH	:	125 to 200 AARH
63 AARH	:	32 to 63 AARH

MECON LIMITED DELHI	Technical Specification for	On line Gas Chromatograph	
INSTRUMENTATION.	SPEC. No. : MEC	SPEC. No. : MEC / 05 / E5 / TS/ GC -030	
SECTION	Rev. 0	Page 10 of 11	9001:2000 Company

**3.1** Gas Chromatograph enclosures and accessories shall be suitable for the electrical area classification indicated in purchaser's data sheets. Unless otherwise specified, the enclosures shall be to the following standards:

Weather proof housing :IP 55 as per IEC -529 / IS -2147Flame proof housing :Flame proof Ex (d) as per IEC -79/IS-2148Flameproof equipment shall also be made weatherproof.

- **3.2** Intrinsically safe version of Gas Chromatograph shall be suitable for the area classification indicated in purchaser's data sheets. Unless otherwise specified, the intrinsic safety shall be to the following standards:
  - a) Intrinsically safe BS 5501/CENELEC EN 50020. All such Gas Chromatographs shall also be weatherproof to IP 55 as per IEC- 79 / IS 2147
  - b) The intrinsically safe and explosion proof equipment shall be certified by statutory bodies like BASEEFA, FM, PTB, CMRI etc.
- **3.3** In addition to the certificate from statutory body, for instruments supplied for projects in India, approval from CCOE (Chief Controller of Explosives) shall also be furnished by Vendor irrespective of their place of manufacture and the same is mandatory. In case CCOE certificate is not available now, Vendor to confirm to supply the same for each item before shipment.
- 3.4 All interconnecting wiring shall be colour coded / numbered and terminal blocks clearly identified.
- **3.5** The sampling system, Gas Chromatograph Analyser, Bottles etc. shall be supplied pre mounted on cabinet of self-standing skid, in general. The foundation of skid shall have holes to anchor it to the ground with anchor bolts. The enclosure to skid shall be weatherproof to IP 55, as a minimum.
- **3.6** The design of Gas Chromatograph system shall be in compliance with the electromagnetic compatibility requirements as per IEC 801.
- **3.7** The Gas Chromatograph shall provide isolated analog current output.
- **3.8** Process sample return point and utilities as required will be made available near the Gas Chromatograph at pressure and temperature conditions specified in the data sheet for each Gas Chromatograph as single point supply and return. In general, all the offered items are to comply with maximum pressure and temperature data specified in data sheets. In exceptional cases, standard cell or such part of instrument with a rating lower than maximum pressure and temperature can be offered provided suitable conditioning means and hardware are provided and adequate safety devices with proper hook up enabling discharge to a flare header is provided. Multi point distribution of utilities and process sample, is to be arranged by Vendor following good engineering practices.

#### 4.0 NAMEPLATE

The Gas Chromatograph and its accessories shall have a SS nameplate firmly attached to it at a visible place, furnishing the following information as applicable:

- a. Tag number as per purchaser's data sheets.
- b. Manufacturer's Serial No. and Model No.
- c. Manufacturer's name / trade mark.
- d. Range
- e. Area classification in which the equipment can be used.

MECON LIMITED DELHI	Technical Specification for	On line Gas Chromatograph	
INSTRUMENTATION.	SPEC. No. : MEC / 05 / E5 / TS/ GC -030		मेकान क
SECTION	Rev. 0	Page 11 of 11	9001:2000 Company

#### 5.0 SHIPPING

All threaded and flanged openings shall be suitably covered to prevent entry of foreign material. Each major part shall be sealed in thick plastic bags. Suitable moisture absorbent shall be provided for electronic components.

#### 6.0 **REJECTION**

Vendor shall make his offer in detail, with respect to every item of the purchaser's specifications. Any offer not confirming to this shall be summarily rejected

All documents and literatures are to be supplied in English Language. The Unit of measurement for all the parameters will be as per Data sheet.

All the Field mounted equipments, Control room mounted instruments and necessary tubing, cables, enclosures, mounting accessories are in venders scope. The system should be supplied in a ready to installed condition at site. All the operation, performance and maintenance guideline to be demonstrated during Inspection and documents to be furnished along with dispatch of the Equipments.





Doc. No. MEC/23UU/05/28/M/000/1092, R0

**MECON LIMITED** 

# PIPING MATERIAL SPECIFICATION (NORTH EAST GAS GRID PIPELINE PROJECT)

DOC. NO. MEC/23UU/05/28/M/000/1092, R0



(PROCESS & PIPELINE DESIGN SECTION) **MECON LIMITED** DELHI - 110 092



Doc. No. MEC/23UU/05/28/M/000/1092, R0



**MECON LIMITED** 

TABLE OF CONTENTS					
1.0	SCC	PE			
2.0	COE	ES AND STANDARD	S		
3.0	MAT	ERIAL SPECIFICATION	N		
4.0	CLA	CLASS DESIGNATION CODE			
5.0	PIPE	ELINE			
6.0	PIPI	NG			
7.0	FITT	INGS			
8.0	BEN	DS			
9.0	FLA	NGES			
10.0	GAS	KETS			
11.0	BOL	BOLTING			
12.0	THR	THREAD SEALANT			
13.0	VAL	VALVES			
14.0	QUI	QUICK OPENING END CLOSURE			
15.0	HYD	HYDROTESTING VENTS AND DRAINS			
16.0	PIPE	ELINE SPECIALITY IT	EMS		
TABLE-1	PIPE	WALL THICKN	ESS DETAILS F	OR	
	MAIN	NLINE/ SPURLINE			
TABLE-2	IND	EX OF PIPING MATE	RIAL SPECIFICATIONS	5	
APPENDICES-I TO	) VI	PIPING MATERIAL	SPECIFICATIONS AS	PER TABLE - 3	
PREPARED BY		CHECKED BY	APPROVED BY	DATE ISSUED	
KRISHNA PATE	L	HEMANT KUMAR	AMIT GANGAL	DECEMBER, 2021	
MANAGER		SR. MANAGER	SR. GM (0&G)		





Doc. No. MEC/23UU/05/28/M/000/1092, R0

**MECON LIMITED** 

## 1.0 <u>SCOPE</u>

This specification covers the requirements of various piping materials used in piping/ pipeline system handling Natural Gas / Re-gasified Liquid Natural Gas (RLNG) and associated utilities in the pipeline.

## 2.0 CODES AND STANDARDS

- 2.1 Pipeline and terminal facilities envisaged as a part of this project shall be designed and Engineered primarily in accordance with the provision of ASME B 31.8 Gas Transmission & Distribution Piping System Latest edition and OISD Standard 226-Natural Gas Transmission Pipeline and City Gas Distribution Networks.
- 2.2 All codes standards and specifications referred herein shall be the latest edition of such documents.
- 2.3 For sake of brevity, the initials of the society to which the codes are referred are omitted in the specification, for example, B16.5 is a code referring to ANSI/ ASME, A 105 is a code referring to ASTM.
- 2.4 In addition, MECON specifications for various piping and pipeline materials shall also be applicable.

## 3.0 MATERIAL SPECIFICATION

Piping material specifications are classified for the general purpose of selection of material for the class of services. The maximum design pressure and design temperature together with the fluid in line governs the selection of material specifications. Deviation of materials from class specifications may occur due to specific design condition. These deviations are permissible if they are equal or better than the individual class requirements.

## 4.0 CLASS DESIGNATION CODE

The piping class designation consist of three digits numbering system made up of letter, number and letter e. g. A1A, B1A, D1A, etc as follows:

First letter indicates ANSI class rating e. g. A-Class 150 B-Class 300 D-Class 600





Doc. No. MEC/23UU/05/28/M/000/1092, R0

**MECON LIMITED** 

The middle number indicates differences in the specification within the same rating and material.

The last letter indicates type of material e.g.

A-Carbon Steel

## 5.0 **<u>PIPELINE</u>**

The material for linepipe shall be as per the requirements of specification as indicated in Table-1.

## 6.0 **<u>PIPING</u>**

- 6.1 Carbon steel pipe shall be made by open hearth, electric furnace or basic oxygen process only. The steel used shall be fully killed and made with fine grain structure. The grade and wall thickness of various sizes of pipes shall be as per piping material specification for the applicable class.
- 6.2 Pipe dimension shall be in accordance with ANSI B 36.10 for carbon steel pipes and ANSI B 36.19 for stainless steel pipes.
- 6.3 All pipe threads shall conform to American Standard taper as per ANSI B 1.20.1 NPT, unless otherwise specified.
- 6.4 For butt weld end, bevel shall be in accordance to ANSI B 16.25/ API 5L as applicable.

## 7.0 **FITTINGS**

- 7.1 Fully killed carbon steel shall be used in the manufacture of fittings.
- 7.2 Threaded joints, if used shall conform to American Standard taper as per ANSI 1.20.1 NPT.
- 7.3 Dimension of socket weld/ screwed fittings shall conform to ASME B 16.11
- 7.4 Bore of socket welded fittings shall suit O. D. of pipe and its thickness.
- 7.5 Dimensions of butt welded carbon steel fittings shall be as per ASME B 16.9 / MSS-SP-75, as applicable.





Doc. No. MEC/23UU/05/28/M/000/1092, R0

**MECON LIMITED** 

- 7.6 Butt welding ends shall conform to ANSI B 16.25/ API 5L. In case of difference in thickness of matching ends, requirements of ASME B 31.8 shall apply.
- 7.7 Integrally reinforced forged branch fittings such as sockolet, threadolet, weldolet, nippolet etc. shall be as per MSS-SP-97. Fittings not covered in ASME B 16.9 and MSS-SP-97 shall conform to manufacturer's standard.
- 7.8 Fittings thickness tolerances shall match pipe thickness tolerance.

## 8.0 **BENDS**

- 8.1 Unless otherwise specified for terminal piping, the elbow of radius R=1.5 D shall only be used.
- 8.2 The radius of cold field bends shall not be less than 30 times the nominal diameter for pipes upto nominal diameter of 16" and shall not be less than 40 times the nominal diameter for pipes of nominal diameter of 18" and above. Limited use of long radius bends (R = 6D) may be permitted for reason of space constraints.

## 9.0 FLANGES

- 9.1 Flange rating shall be same as ANSI B 16.5/MSS-SP-44/ B 16.47 Series A as specified.
- 9.2 Dimensions of flanges shall be in accordance with ANSI B 16.5/ B 16.47 Series A, as applicable.
- 9.3 Neck of Weld Neck (WN) flanges to suit pipe bore and thickness.
- 9.4 Bore of Socket Welded (SW) flanges shall suit pipe O.D. and its thickness.
- 9.5 Threads for screwed flanges if used shall conform to American Standard taper as per ANSI B 1.20.1 NPT.
- 9.6 Sizes for blind flanges shall be indicated by nominal pipe sizes.
- 9.7 Carbon steel flanges faces shall have smooth finish as indicated in the material specification. Flanges faces shall have smooth finish to 125-250 micro inches AARH as per MSS-SP-6.
- 9.8 Butt welding ends of WN flanges shall conform to ANSI B 16.25.





Doc. No. MEC/23UU/05/28/M/000/1092, R0

- 9.9 Spectacle blind/ spacer & blinds shall be in accordance with ASME B 16.48 / Manufacturer's Standard. Spectacle blind shall be used for sizes up to 8" NB and spacer & blind for 10" & above shall be used.
- 9.10 Two jack screws 180° apart shall be provided for all spectacle blind assemblies. The jack screws shall be as per MECON's standard.

### 10.0 **GASKETS**

10.1 Spiral wound metallic gaskets shall conform to B 16.20 and API 601 shall be provided with graphite filler. All spiral wound gaskets shall be provided with stainless steel centering ring.

### 11.0 BOLTING

- 11.1 Nuts for stud bolts shall be American Standard Hexagonal Heavy series and double chamfered.
- 11.2 Dimension and tolerances for stud bolts and nuts shall be as per ANSI B 18.2.1 and 18.2.2 with full threading to ANSI B 1.1 Class 2A thread for bolts and Class 2B for nuts. Diameter and length of stud bolts shall be as per ANSI B 16.5/ ASME B 16.47 with full threading.
- 11.3 Threads for nuts shall be as per ANSI B 1.1, as follows:

Nuts for stud dia  $\frac{1}{4}$ " to 1" : UNC-2B Nuts for stud bolts dia  $1^{1/8}$ " to 3  $\frac{1}{4}$ " : 8UN-2B

11.4 Threads for stud bolts shall be as per ANSI B 1.1, as follows.

Studs bolts dia ¼" to 1"	:	UNC-2A
Stud bolts dia $1^{1/8"}$ to $3\frac{1}{4"}$	:	8UN-2A

11.5 Heads of jack screws shall be heavy hexagonal type. Jack screw end shall be rounded. Stud bolts shall be fully threaded with two hexagonal nuts.

## 12.0 THREAD SEALANT

12.1 Threaded joints shall be made with 1" wide PTFE Jointing tape.


#### PIPELINE LAYING AND TERMINAL WORKS FOR NORTH EAST GAS GRID PIPELINE PROJECT

Doc. No. MEC/23UU/05/28/M/000/1092, R0



### 13.0 VALVES

- **13.1** Valve ends shall be as per piping material specifications (Appendices).
- **13.2** Flange dimensions and face finish of flanged end valves shall confirm to clause 9.0 of this specification.
- **13.3** Butt welding ends of Butt Welded valves shall conform to ANSI B 16.25.
- **13.4** Face to face and end to end dimensions shall conform to applicable standards.
- **13.5** Buried valves on mainline shall be provided with stem extension, sealant, vent/drain & shall have butt welded ends.
- **13.6** Sectionalizing Valves (Block valves) installed on the main pipeline shall be Ball valves with butt welded ends and shall be full bore to allow smooth passage of cleaning pigs as well as intelligent pigs.
- **13.7** Unless specified otherwise. Valves shall confirm to the following standards:

Screwed / Socket welded / Flanged end valves (1 1/2" and below)

Ball Valves	-	BS 5351(latest)
Plug Valves	-	BS 5353(latest)
Globe Valves	-	BS 5352(latest)
Gate Valves	-	API 602(latest)

Flanged / Butt weld end Valves (2" and above)

Ball Valves	-	API 6D
Plug Valves	-	API 6D
Check Valves	-	API 6D
Globe Valves	-	BS 1873
Gate Valves	-	API 6D

13.8 Manual Valve operators shall be as indicated below, unless specified otherwise in the P&ID.



PIPELINE LAYING AND TERMINAL WORKS FOR NORTH EAST GAS GRID PIPELINE PROJECT



Doc. No. MEC/23UU/05/28/M/000/1092, R0

MECON LIMITED

### a) Gate and Globe Valves

- i) For ANSI class 150 & 300 Hand Wheel operated for size <12"NB. Gear operated for size >14" NB.
- ii) For ANSI class 600 Hand Wheel operated for size <10"NB. Gear operated for size >12" NB.

# b) Ball and Plug Valves

- For ANSI class 150, 300, 600 Wrench operated for size <u><</u>4"NB. Gear operated for size <u>></u>6"NB.
- c) Actuated Valves Actuated valves shall be as per P & ID.

### 14.0 QUICK OPENING END CLOSURE

Quick opening end closure to be installed on scrapper traps shall be equipped with safety locking devices in compliance with section VIII, division 1, UG-35.2 of ASME Boiler and Pressure Vessel code.

### 15.0 HYDRO TESTING VENTS AND DRAINS

High point vents and low point drains required for the purpose of hydro testing shall be of size 3/4" and consist of sockolet, Plug/ Ball valve for vent, Globe / Ball Valve for drain, flange & blind flange.

### 16.0 PIPELINE SPECIALITY ITEMS

Pipeline Specialty items viz., Scrapper Traps, Flow Tee, Insulating Joints, LR bends, QOEC for Venting shall be as per respective data sheets, specifications and Project Specific drawing showing Mainline & Terminal materials.



#### PIPELINE LAYING AND TERMINAL WORKS FOR NORTH EAST GAS GRID PIPELINE PROJECT

Doc. No. MEC/23UU/05/28/M/000/1092, R0



# TABLE-1 PIPE WALL THICKNESS DETAILS FOR MAINLINE/SPURLINE

SI. No.	Pipe Material Description	Size (NB)	Thickness (mm)	Length
1	API 5L Gr. X-70, PSL-2	18"	7.92	As per SOR Quantity
2	API 5L Gr. X-70, PSL-2	18"	9.53	As per SOR Quantity
3	API 5L Gr. X-70, PSL-2	18"	12.7	As per SOR Quantity
4	API 5L Gr. X-70, PSL-2	12"	7.14	As per SOR Quantity
5	API 5L Gr. X-70, PSL-2	12"	9.53	As per SOR Quantity
6	API 5L Gr. X-56, PSL-2	8"	7.04	As per SOR Quantity

# TABLE-2

# INDEX OF PIPING MATERIAL SPECIFICATIONS

Class	Service	C.A.	Basic Material	Design Code	Enclosed as
		(mm)			
D1A	Natural Gas / RLNG	1.5	ASTM A 106 Gr. B/ API 5L Gr. B/	ASME B31.8	Appendix-I
D4A	Natural Gas / RLNG	1.5	ASTM A 333 Gr. 6	ASME B31.8	Appendix-II
B1A	Natural Gas / RLNG	1.5	ASTM A 106 Gr. B/ API 5L Gr. B	ASME B31.8	Appendix-III
B4A	Natural Gas / RLNG	1.5	ASTM A 333 Gr. 6	ASME B31.8	Appendix-IV
A1A	Natural Gas / RLNG	1.5	ASTM A 106 Gr. B/ API 5L Gr. B	ASME B31.8	Appendix-V
A4A	Natural Gas / RLNG	1.5	ASTM A 333 Gr. 6	ASME B31.8	Appendix-VI

		<u>&gt;</u>			, w	r		D	т			г			ဂ		Т	
APPR	ᄝ	גַּ	SE	REV	ANSI	CORROSI	ON	TEMP °C	-29.0 3	8.0	50	100	150	200	250			
	WN S	2		S	CLASS:	600 # ALLOWAN	ICE: 1.5 M	M PRESS. KG/CM <sup>2</sup> g	104.14 10	04.14	102.2	95.04	91.97	89.35	85.07			-
			N:	DATE	SERVI	CE · NATURAL	CAS	1	BASE	MAT	FRIAL	. CARB	ON STE	ËL				
		- <sup>m</sup>		ZON			_ 0A3		DAGE			· (MATI	erial g	ROUP 1	.1)			
თ		DATE	입	- m										-				ი
			8		12. FITTINGS	SHALL BE OF SE	AMLESS CON	ISTRUCTION UP TO 16	" AND SHA	LL BE	E OF WE	LDED CO	NSTRUCI	's. FION 18"	AND AB	OVE.		
I.GAN		影	GA		3. WALL THI	CKNESS FOR LINE	PIPE USED	IN VARIOUS SECTIONS	SHALL BE	AS F	PER TAB	LE-1 OF	PMS.					ſ
IGAL		- p	ر م		4. BALL VAL	VE TO BE USED	IN MAINLINE	SHALL HAVE BUTT W	ELDED END	S EX	CEPT FC	R THE V	/ALVES U	SED FOR	НОТ ТА	PPING		Ľ
		Ŧ	i	RF IS	WHICH SH	ALL BE ONE SIDE	E BUTT WELL	DED AND OTHER SIDE	FLANGED.									ſ
					5. PROCUREN	MENT OF MATERIA	LS SHALL E	BE AS PER DETAILED I	CULTES AD		FICATION	15. <sup>2</sup> ~ % (	20° TO	165°C)				ſ
		F-	,   <sup>i</sup>	S N S	7. PRESSURE	E-TEMPERATURE	RATING INDI	CATED ARE FOR FLAN	GES ONLY	IN AC	CORDAN	UCE WITH	I ANSI B	16.5				ſ
5 PP		AA			8. FOR VALV	/ES,STEELPIPE AN	D ASSOCIAT	ED STEEL COMPONENT	S OF 2" A	ND L	ARGER 1	ОТСН Т	OUGHNES	S PROPE	RTIES S	HALL BE	AS SPECIF	
NG	5	T			IN RELEVA	ANT SPECIFICATIO	NS/CODES,	MECON'S STANDARD 1	ECHNICAL	SPECI	FICATION	NS AND	DATA SH	EETS ET	С.			ſ
MA		או	;		9. AT STATIC	ONS, BRANCH CO	NNECTIONS	SHALL BE AS PER BR.	ANCH CONN	NECTIO	ON TABL	E BELOV	V.					ſ
	1	ΡÊ		Ψ	10. ALL BUT	T WELDS SHALL E	BE 100% RA	DIOGRAPHED.										μ
		RC			11. 100% OF	SOCKET WELD	SHALL BE S	UBJECTED TO MPI/DPI	Г.									
<u>ା</u> ି ଧ	3	JI		APPR	12. PRESSUR		RATING OF	VALVE BODY SHALL E	BE AS PER	API	6D.							ſ
		G .	.	0	114 FOR PIPE	LINE SPECIALITY	ITEMS (SCR	APPER TRAP FLOW T			S FTC )		IR MATE			21	48	ן ב
)) FIC	3	TĽ			REFER D	ATA SHEET OF R	ESPECTIVE I	TEMS.	, 10, LIX 1		5 210.)						46	-   7
	į	т Н			STATIO	N DIDINC W	ΙΛΤΓΩΙΛΙ	SDECIFICATIO	NT								42	-
		L			<u>STATIO</u>	IN FIFING IN	IAIENIAI	SFECIFICATIO	<u>IN</u>					_			38	<u> </u>
		Ч И И			ITEM	SIZE		DESCRIPTION			BRANC	H CONN		>			36	] [
		<b>-</b>	'  i	<b>쮜</b>	MAINTENANCE	ALL	FLGD., BUT	TO BE KEPT MINIMUM			H H. C	S BW COUPLING				╺┩┼┼┼	32	-
ω					JOINTS					4	P PIPE	TO PIPE	<u> </u>				28	
APF		<sup>31</sup>		5		1.5" & BELOW	SOCKET WE	ELD			IS SOC	KETLETS		1	┿┩╷╷		24	<u> </u>
			$\left  \right $		PIPE JOINTS					-	T TEES	S SW/		_┏╃┼			22	100
	COM		7			2" & ABOVE	BUTT WELD	ED				DOLETS		┏┩┤┼┼			18	- 꽃   !
	7,	h-					- (			-			▁	╃┼┼┼┼	╎┢┪╎		14	Iĭ H
					DRAINS	ON LINES $\leq 1.5$ "	3/4", AS F	PER MEC/SD/05/21/15	5/03								12	
					DRAINS	ON LINES > 2"	3/4" OR A	S PER P&ID MEC/SD/	)5/21/15/0	1			┏┩┤┼┼	┼┢┿┩┼	++++	$\neg$	8.0	
N						-				4			┦╎╎┢┿╸	╈┩┼┼┼		w  - - -	5.0	
			₽∕ļ	$\square$	_	ON LINES $< 1.5$ "	3/4", AS F	PER MEC/SD/05/21/15	5/03								3.50	<u> </u> 2
	ΙÈ	5   3	9		VENTS		, , , ,		· 	-	1	╉╤┽┥	┼╉┯┽┼	++++	++++		2.50	卢뷭ㅣ!
		2   -	וא			ON LINES $\geq 2$ "	3/4" OR A	S PER P&ID, MEC/SD/0	05/21/15/0	1	- F	┛					2.00	╡╩└╵
$\Box$		-   -	ক			4 F"		AC DED MEO (CD (05 (01 /1	E /00	1							1.50	
<del>.</del>	-  <b>L</b>		븳		TEMP. CONN.	1.5	FLGD. INSIL.	AS PER MEC/SD/05/21/1	5/02	F	╿╤┤┨┤		++++	┼┼┼┼	⊣s  <del> </del>		1.00	<u> </u> #
	ÌĬ		뷗	N N			SCH. 160 N	NIPPLE WITH BALL VALV	E TO SPEC.						┽┽┼┼		0.50	]
	í <b>–</b>	3   Č	א		PRESS CONN.	3/4"		ON AS PER MEC/SD/05	/21/15/05	<u> 0.50</u> 0.75	<u>1.25</u> 2.00	2001		100000	20000	<u>728880000000000000000000000000000000000</u>	44044	
									,, .0, 00									
ο₽	7	-									F	UN PIPE	NOMINAL	DIA (INC	HES)	Page 76	6 of 644	
		>					<u> </u>	0	m			г			G		т	

	<u>}</u>				8	0		D		m		п	G		T	
APPROVED	DSGN	SECTI	REV NO													
	Р.	NAM	DATE													
_		<u> </u>	Z		PIPELINE	/PIPING DESIGN CODE	AS	ME B 🕻	31.8/ OISD 22	26		DESIGN F	ACTOR - 0.5	5		
ი		직 입		_	ITEM	NOMINAL DIAMETER (INCHES)	0.50 0.75	1.00 1.50	2.00 3.00 4.00 6.00	8.00		0.0 12.0 14.0 1	6.0 18.0 20.0 24.0	28.0 30.0		
		<u> </u>				WALL THICKNESS (MM/SCH)	S160 S160 S	S80 S80	S80 XS XS XS	14.3	:	(S S60 14.3 1	17.5 S60 S60 19.1	22.2 23.8		
A.G/	Ŧ	옷 요				MATERIAL	ASTM A10	6 GR.B	ASTM A106 GR.B	API 5L GR.B PSL2		API 5L GR.X-5	2 PSL2			
		<u> </u>			PIPE	DIMENSION STD.	B36.10		(orward 1)	API 5L	I					
-  <sup>-</sup> -		DATE	ᆔᇝ			METHOD OF MANUFACTURE, ENDS	SEAMLESS	PE	SEAMLESS BE			ŧ	BE SAW			
		···				MATERIAL AND GRADE	ASTM A 10	05	ASTM A 105 (CHA	RPY)		ASTM A 694 G	R.F-52 (CHARPY)			
			IONS NO		FLANGE	TYPE, FLANGE FACING	SW. RF 12	5AARH	WN. THICKNESS T	О МАТСН РІРЕ ТНІСИ	NESS,RF 125	ARH				
~ 믿		z	N N			DIMENSION STD.	B16.5							B 16.47A		
P		A			BLIND	MATERIAL AND GRADE	ASTM A 10	05	ASTM A 105 (CHA	RPY)						
6		l D			FLANGE		RF 125AAF	RH						<u> </u>		
MA		R				DIMENSION STD.	B16.5							B 16.47A		
- 6 린	ч	AL				MATERIAL AND GRADE	ASTM A 10	05	ASTM A 105 (CH	ARPY)				ASTM A 51	6 GR70	
AIN WIN	ਲੱ		Ĩ		BLANK	FLANGE FACING	FF 125AAR	RH						1		
	ି ପ୍ର ପ	GA	Ą			DIMENSION STD.	B16.48							MNF' STD		
(Q Å	E	S	퀑			ТҮРЕ	FIG.8 FL	ANGE				SPACER & BLI	ND			
온읚					BOLTING	STUD BOLTS (FULLY THREADED)	A 193 GR	B7, B-18	3.2							
	-	ĬĦ				NUTS (HEAVY HEXAGONAL)	A 194 GR	2H, B-18	3.2							
UT I		Ĕ			GASKET	TYPE ,MATERIAL AND Dmn. STD.	SPIRAL 60	0#, SP.V	ND SS316+GRAPH	TE FILLED, B-16.20-	ANSI B16.5,					
N		<b>E</b>			FI BOW-90		ASTM A 10	05	ASTM A 234 GR.W	PB (CHARPY)		MSS-SP-75 GR	.W PHY-52			
		H			ELBOW-45		SW,6000#	SW,3000#	BW, 1.5D							
			쮸			DIMENSION STD.	B-16.11		B-16.9			MSS-SP-75				
4					T-EQUAL		ASIM A 10	05	ASIM A 234 GR.W	PB (CHARPY)		MSS-SP-75 GR	.W PHY-52			
¥ S	.61		CES		T-RED		SW,0000#	SW,3000#	Bw 16.0							
₽ ₽						DIMENSION STD.	B-10.11	05	B-16.9			MSS-SP-75				
	2000	<b>&gt;&gt;</b> )			CAR		SCREEDOD IS	00	ASTM A 234 GR.W	S TO MATCH DIDE 1		MSS-SP-/5 GR	W PHY-52			
	COMPANY L	$\bigcirc$					B-16 11	SCRF 3000	B-16 9	55 TO MATCH FIFE I		NCC CD 75				
-   !						MATERIAI	ASTM A 10	05	ASTM A 234 GR W	PB (CHARPY)		MSS-SP-75				
						END DETAIL	SW-6000#		BW.THICKNESS	TO MATCH PIPE THI	CKNESS	MJJ-JF-/J GK				
					FITTNE	DIMENSION STD.	B-16.11		B-16.9			MSS-SP-75				
3	MEC	मव			FITTING	ТҮРЕ	COUPL FULL,H LH.,RE	LING HALF ED.	RED. CON RED. ECC	۱. ک	L					
	$ \Omega $	2				MATERIAL	ASTM A 10	05	ASTM A 105 (CH	ARPY)		ASTM A 694 G	R.F-52 (CHARPY)			
	4	4				END DETAIL	SW,6000#	SW,3000#	BW			BW				
<u>(</u> )		<u>a</u>			O'LET	DIMENSION STD.	MSS-SP97	,	MSS-SP97			MSS-SP97				
		ਜ	DRG			TYPE	SOCKOLET		WELDOLET			WELDOLET				
2 OF 2) REV	<b>IITED</b>	टेड	NO.			<u> </u>						7		Pa	age 77 of 64	44

	≥				B	0	C	,		m			וד	େ	>		Т	
APPR	<del>ק</del> 0	SE	RE					45 70	50	100								
- ROVED	R G G G G G G G G G G G G G G G G G G G	CTI	No		ANSI CLASS: 600#	CORROSION			50	100	150	200						-
	57	N N	DAT				PRESS. KG/CM	97.89 97.89	96.77	92.48	89.63	87.02						
		M ···	Z		SERVICE :	RLNG		BASE M	ATERI	AL: C	ARBON	N SIEEL	- 					
<b>о</b>		<u></u> 20	Â	_						(		IAL GIN	001 1.0)					
		≣ਾ			IOTES: -													0
A.G,	<u>5</u> 1.天	₽ G		2	. FITTINGS SH	ALL BE OF SEAM	LESS CONSTRUCTION	ON UP TO 1	S" AND	SHALL	BE OF	WELDED	CONSTRU	CTION 18"	AND ABOV	E.		
ANG/	~~			3	. WALL THICKI	NESS FOR LINEPI	PE USED IN VARIO	US SECTIONS	S SHAL	BE A	S PER	TABLE-1	OF PMS.					
		DATE	ᆔᇏ	4	. BALL VALVE	TO BE USED IN	MAINLINE SHALL H	HAVE BUTT	WELDED	ENDS	EXCEPT	FOR TH	E VALVES	USED FOR	HOT TAPF	PING		
			SCRIF		WHICH SHAL	L BE ONE SIDE E	BUTT WELDED AND	OTHER SIDE	FLAN	ED.								
			NOIL	5.	PROCUREMEN	NT OF MATERIALS	SHALL BE AS PE	R DETAILED	RELEV	ANT SP	ECIFICA	TIONS.						
▫ 등ੁੁੁ		Z	°	6.	DESIGN PRES	SSURE & TEMP. F	OR PIPELINE AND	RELATED FA	CILITIE	S ARE	92 Kg/	cm² g &	& <u>(</u> −45° TC	+65°C) F	RESPECTIV	ELY.		
N PIN		AT		/	. PRESSURE-	IEMPERATURE RA	ING INDICATED A	RE FOR FLAD	IGES O	NLY IN	ACCOR	DANCE N	MIH ANSI	B 16.5			005015	
IEN		Ē		8	IN DELEVAN	STEELPIPE AND	ASSOCIATED STEEL	COMPONEN	IS OF	2 ANL		R NOIC	H IOUGHNE	SS PROPER	RILS SHA	LL BE AS	SPECIF	IED
PA		RA					FOTIONS SHALL DE	STANDARD		CAL SP				SHEETS ETC	•			
- 문희	PF	Г	R	9.	AT STATIONS	S, BRANCH CONN	ECTIONS SHALL BE	AS PER B	ANCH	CONNE	CHON I	ABLE BE	LOW					
RIA	õ	G			). ALL BUIL V	WELD'S SHALL BE	100% RADIOGRAPH	HED. N TO MOU/DE	т									
H H N	JE	AS	PPR	12	. 100% OF SC	TEMPERATURE R	ALL BE SUBJECTEL ATING OF VALVE F	NO MPI/DP	I. RF AS	PFR A								
4 199	G				3. PIPELINE DE	ESIGN CODE - AS	SME B 31.8 & OIS	D 226.			11 00.							4
	-	II		14	4. FOR PIPELI	NE SPECIALITY IT	EMS (SCRAPPER T	RAP, FLOW <sup>-</sup>	ΈΕ, IJ,	LR BE	NDS ET	C.) AND	THEIR MAT	ERIAL DES	CRIPTIONS			
ICA		E			REFER DAT	A SHEET OF RES	PECTIVE ITEMS.											
					CTATION	DIDING MA			NT							·		
A SN		YE				PIPING MA	IERIAL SPEC					NCH CO		S			54	
			R			SIZE	DE	SCRIPTION				FFS BW		0			5 <u>2</u>	
ω			RENO			ALL	FLGD., BUT TO BE	KEPT MINIMU	JM		НН	I. COUPL	ING				28	
APISC	051		Б.									PIPE TO F						
ЩЩ Ш						1.5" & BELOW	SOCKET WELD					OCKETLE	TS					
		>)			PIPE JOINTS						TT	EES SW/	/	_	╺┛┨┼┼┼		<u>6</u> =	
	TRANK ANY					2″ & ABOVE	BUTT WELDED					ELDOLET		┍┩┤	┽╉┼┼┼		4 <u>4</u> <u>4</u>	
N N									/15 /07				NLD.	┏┛╧	┽┫┼┼┼			
					DRAINS	ON LINES $\leq 1.5^{\circ}$	J/4, AS PER MI	20/50/05/21	/15/03								5.0 N	
	$\mathbf{Z}$				DIVAINS	ON LINES > 2"	3/4" OR AS PER	P&ID. MEC/S	D/05/2	1/15/0	1		Г	╺┩╎╎┢┿┥	╺╃╂┼┼┼		5.0 Ž	N
Ĩ	E	4-1.				-				,	_		┍╼╇╡	┼┢┿┛┼┘	╡┛┼┼┼		5.50	
	2	4				ON LINES $< 1.5$ "	3/4", AS PER MI	EC/SD/05/21	/15/03					<u></u>	┙┝┫┼┼┼		<u>2.50</u> 포	
					VENTS						_		╺┫╴╴┏┿┿┥				2.00 NA	
-	~					ON LINES $\geq$ 2"	3/4" OR AS PER	P&ID, MEC/S	D/05/2	1/15/0	1						. <u></u>	H
Г Ŷ		3									_	┏┩┼┤			┽╉┼┼┼		0.75	
	N	P	DRG.		TEMP. CONN.	1.5"	FLGD. INSTL. AS PER 	R MEC/SD/05/2	21/15/02	2					╺┿┛╎╎╎		0.50	
그   으	<u> </u>		NO.								-	.250	0.200.200	000074	6000 <u>01</u> 40	0740		
	H	V4			PRESS CONN.	3/4"		NALVE IU S	1 LU.	1 1			-000004			(רא לא לא לא אין אין אין אין אין אין אין אין אין אי		-
	Ŭ						INSTALLATION AS	PER MEC/SD	/05/21,	/15/05		R	UN PIPE NO	MINAL DIA	(INCHES)	a 70 - f	644	
으문							1								Pa	ige /8 of	ю44	
			┶┶┶╋		ω	0	c			m			ח	<u>୍</u>	>	1	I	I

	>		8	0		D		m	וד	G	<b>I</b>
APPROVED	DSGN DRWN	REV NO									
	.°		<u></u>	PIPELINE/	PIPING I	DESIGN CODE	ASME B 3	31.8/ OISD 2	26 DESIGN	FACTOR - 0.5	]
7		Ĭ		ITEM	NOMINAL I	DIAMETER (INCHES)	0.50 0.75 1.00 1.50	2.00 3.00 4.00 6.	00 8.00 10.0 12.0 14.0 1	6.0 18.0 20.0 22.0 24.0	1
					WALL THIC	KNESS (MM/SCH)	S160 S160 XS XS	XS STD XS X	s XS XS 19.0 20.6 2	2.2 25.4 22.2 25.4 26.3	1
A.	프 달 6				MATERIAL			ASTM A333 GR.	6		1
ANC	AS			PIPE	DIMENSION	STD.		B36.10			1
Ě	DA				METHOD O	F MANUFACTURE, ENDS	SEAMLESS PE	SEAML	ESS BE	BE SAW	1
		REVES			MATERIAL	AND GRADE		ASTM A 350 GR.	LF2, CL-I		-
				FLANGE	TYPE, FLAI	NGE FACING	SW. RF 125AARH	WN. THICKNESS	TO MATCH PIPE THICKNE	SS, RF 125AARH	1
_		SN S			DIMENSION	STD.	B16.5	•			
P₽₽				BUND	MATERIAL	AND GRADE		ASTM A 350 GR	LF2, CL-I		
				FLANGE	FLANGE FA	CING	RF 125AARH				
EG	l di				DIMENSION	STD.	B16.5				
P≤ ∧	A	$  \downarrow$	+-1		MATERIAL	AND GRADE		ASTM A 350 GR.	LF2, CL-I		
1 문란	PH PH	<del>u</del>	₽	BLANK	FLANGE FA	CING		FF 125AARH			
Alk Alk	ନ୍ତି	ΙĻ			DIMENSION	STD.		B16.48			
R.	JI				TYPE		FIG.	8 FLANGE	SPACER &	BLIND	
6 SPE					STUD BOLT	rs (FULLY THREADED)	,	A 320 GR L7, B-1	8.2		1
	PI T			BOLTING	NUTS (HEA	VY HEXAGONAL)		A 194 GR 4, B-18	3.2		
FIC ₩	P			GASKET	TYPE, MAT	ERIAL AND Dmn. STD.	SPIRAL, SP.WND S	S316+GRAPHITE F	LLED, B-16.20-ANSI B16	5.5,	
(TA	EL				MATERIAL		ASTM A350 GR.LF2	ASTM A	420 GR.WPL6	A 420 GR.WPL6.W	
				ELBOW-90	END DETAI	L	SW,6000# SW,3000#	BW,	1.5D		
P <u>P</u>	E			ELBOW-45	DIMENSION	STD.	B-16.11	В-	16.9		
		ÊFE			MATERIAL		ASTM A350 GR.LF2	ASTM A	420 GR.WPL6	ASTM A 420 GR.WPL6.W	
·		Ř		T-EQUAL	END DETAI	L	SW,6000# SW,3000#		BW		
APF	2.0 <sup>51</sup>	S		I-RED	DIMENSION	STD.	B-16.11	в	-16.9		
ЩЩ.				CAP	MATERIAL		ASTM A350 GR.LF2	ASTM A	420 GR.WPL6		-
┋				Å.	END DETAI	L	SCRF6000 SCRF3000	BW, THK TO M	ATCH PIPE THICKNESS		-
ĺ				(UPTO 1.5")	DIMENSION	STD.	B-16.11	B-16	.9		]
<u>8</u>		$\left  \right $			MATERIAL		ASTM A350 GR.LF2	ASTM A	420 GR.WPL6	ASTM A 420 GR.WPL6.W	]
					END DETAI	L	SW-6000 SW-3000	BW, THICKNE	SS TO MATCH PIPE THIC	(NESS	
				FITTING	DIMENSION	STD.	B-16.11	B-16	.9		
	मंच MEC				TYPE		COUPLING FULL,HALF LH.,RED.	RED. CON. F	RED. ECC.		
	<u> </u>				MATERIAL		ASTM A350 GR.LF2				-
	~   ~				END DETAI	L	SW,6000# SW,3000#	BW			
	<b>H</b>   7			O'LET	DIMENSION	STD.	MSS-SP97	MSS-SP97	-1		
<del>S</del>	]   नि	DRG			TYPE		SOCKOLET	WELDOLET	-1		
2 OF 2) REV	ਟੇਤ fITED	NO.									Page 79 of 644
<u> </u>	<u> </u>		<u>+                                    </u>					-		<u>^</u>	

		<u> </u>			Φ	0		D	m			п		G		<b>工</b>		_
APPROV	DRW	DSG	SEC	REV N	ANSI		N 1 E	TEMP °C	-29.0	38.0	50	100	150	200	]			
			-13	0	CLASS: J	00 # ALLOWANC	E: 1.5	PRESS. KG/CM <sup>2</sup> g	52.1	52.1	51.10	47.52	45.98	44.60				
		.P.	N:	ATE ZON	SERVIC	E : NATURAL	GAS		BASE	e mat	ERIAL	. CARB . (MATE	ON STEI ERIAL GI	EL ROUP 1.1)				
6 A.GANGAL		H.K.	OIL & GAS	DESCR REV	1. ALL VENT 2. FITTINGS 3. WALL THIG 4. BALL VAL WHICH SH	S & DRAIN SHA SHALL BE OF SI CKNESS FOR LIN .VE TO BE USED IALL BE ONE SIE MENT OF MATER	LL BE PR EAMLESS EPIPE US IN MAINI DE BUTT 1 IAI S SHAI	OVIDED WITH PLUG VAL CONSTRUCTION UP TO ED IN VARIOUS SECTIO .INE SHALL HAVE BUTT WELDED AND OTHER SII L BE AS PER DETAILE	-VE UNLE 16" AND NS SHAL WELDED DE FLANG	SS MEN SHALL L BE AS ENDS I GED. ANT SPE	ITIONED BE OF S PER <sup>-</sup> EXCEPT	OTHERWI WELDED ABLE-1 FOR THE	SE IN P& CONSTRU OF PMS. I VALVES	dDs. ICTION 18" / USED FOR	AND ABO HOT TAP	VE. 'PING		6
PIPINO			5	IPTIONS SIONS	6. DESIGN P 7. PRESSURE	RESSURE & TEM E-TEMPERATURE	IP. FOR P RATING	IPELINE AND RELATED NDICATED ARE FOR FL	FACILITIE ANGES O	S ARE -	49 Kg	/cm² g & DANCE WI	(-29° T TH ANSI	O +65℃) I B 16.5	RESPECTI	VELY.		
5 MATERI			VATUR		8. FOR VALV IN RELEV	/ES,STEELPIPE A ANT SPECIFICATIONS BRANCH CO	ND ASSO ONS/CODI	CIATED STEEL COMPONE ES, MECON'S STANDARI	ENTS OF D TECHNI BRANCH	2"& L CAL SPI	ARGER ECIFICA	NOTCH T TIONS AN ABLE BEI	OUGHNES D DATA : OW	S PROPERTI SHEETS ETC	ES SHALI	_ BE AS SF	PECIFIED.	U
AL SF		ΡI	AL	ВY	10. ALL BUT	T WELDS SHALL	BE 100%	RADIOGRAPHED.		0011120								-
PECIFIC		ROJE	GAS	APPR	12. PRESSUR	E-TEMPERATURE	SHALL B E RATING - ASMF F	OF VALVE BODY SHAL	L BE AS	PER AF	PI 6D.							
CATIONS		CT	PIPI		14. FOR PIPE REFER D STATIO	ELINE SPECIALITY ATA SHEET OF F	( ITEMS ( RESPECTIV MATER	SCRAPPER TRAP, FLOW /E ITEMS.	' TEE, IJ,	LR BEN	NDS ET(	C.) AND 1	THEIR MA	TERIAL DESC	CRIPTIONS	>		4
300			ELI													36	7	
)# (B1/			NE	REF	MAINTENANCE	ALL	FLGD.,	BUT TO BE KEPT MINIMU	М		BRANC	H CONN	FCTIONS		r-F	34 32 30 28		
A) 3 AF	SC	- 190		ERENCES		1.5" & BELOW	SOCKE	WELD			E TEES	S BW		_	┏┪┼┼┼	20 24 22 20	ES)	3
PENDI						2" & ABOVE	BUTT V	/ELDED			P PIPE R REIN S SOC	TO PIPE FORCED KETLETS		₽₽		18 16 14	▲ (INCF	
=	N.T.S.				DRAINS	ON LINES $\leq 1.5$	" 3/4", <i>A</i>	S PER MEC/SD/05/21	/15/03		T TEES	S SW/ DOLETS		┎┛┼┼┼╴		12		_
						ON LINES $\geq 2$ "	3/4" 0	R AS PER P&ID, MEC/SI	D/05/21/	′15/01			┈┙╻ <sub>┛</sub> ┫	<mark>╶╶╴┲┿┿┩</mark>		5.0 4.0		
2		MEC	Ъ	,	VENTS	ON LINES $\leq 1.5$	" 3/4", <i>A</i>	S PER MEC/SD/05/21	/15/03			┏╼┥	┏┿┩╒╎ ┛╵╵╵╽			3.00		2
		NON	कॉन			ON LINES $\geq 2$ "	3/4" 0	R AS PER P&ID, MEC/SI	D/05/21/	′15/01			┛		┝┫╎┼┼	<u> </u>	BRANCI	
	چا	L	ন্দ্র		TEMP. CONN.	1.5"	FLGD. IN	STL. AS PER MEC/SD/05/2	1/15/02		Ч	┛ <u>┼</u> ╎┠╎ ┼╵╵┠┤				1.00 0.75 0.50		
	1. 1 OF 2	IMITE	ਸਿਟੇਤ	DRG. NO.	PRESS CONN.	3/4"	SCH. 16 INSTALL	SO NIPPLE WITH BALL VA ATION AS PER MEC/SD	ALVE TO S /05/21/15	SPEC. 5/05	0.50	<u>900-1- 2000-1- 7-00-1- 80N F</u>	20000 2000 2000 2000 2000 2000 2000 20	00000000000000000000000000000000000000	HES)	32 32 36 36		
		D													Pag	e 80 of 64	4	
	<u> </u>					0	T	D	п			п		G				Г

	····		╷		0	D	m		<u>г</u>	୍ର ଜ		I	,
APPRO1	DSG												
			5										
	P. AME		1	PIPELINE/	PIPING DESIGN CODE	ASME B	31.8/ OISD 226	DES	IGN FACTOR - 0.5				
6	DA			ITEM	NOMINAL DIAMETER (INCHES)	0.50 0.75 1.00 1.50	2.00 3.00 4.00 6.00 8.0	0 10.0 12.0 1	4.0 16.0 18.0 20.0				
		F			WALL THICKNESS (MM/SCH)	S160 S160 XS XS	XS STD S40 S40 7.9	STD S40 S	540 S40 S40 15.9				
A.G	표 물	ጽ   ብ			MATERIAL	ASTM A106 GR.B	ASTM A106 GR.B (CHARPY)	API	5L GR.B PSL2				
ANG	^ <del>0</del>	AS		PIPE	DIMENSION STD.	B36.10	, , , , , , , , , , , , , , , , , , , ,	API 5L					
_ <sup>₽</sup>	DAT		2		METHOD OF MANUFACTURE, ENDS	SEAMLESS PE	SEAMLESS B	E	BE SAW				
					MATERIAL AND GRADE	ASTM A 105	ASTM A 105 (CHARPY)						
면				FLANGE	TYPE, FLANGE FACING	SW. RF 125AARH	WN. THICKNESS TO MA	ICH PIPE THIC	CKNESS,RF 125AARH				
P		S N N			DIMENSION STD.	B16.5							
5 G					MATERIAL AND GRADE	ASTM A 105	ASTM A 105 (CHARPY)						
MA				FLANGE	FLANGE FACING	RF 125AARH							
	UF				DIMENSION STD.	B16.5							
RIA					MATERIAL AND GRADE	ASTM A 105	ASTM A 105 (CHARPY	)					
ПС	PL		श		FLANGE FACING	FF 125AARH							
II PE	ଚ <u>ଚ</u> ନ			BLANK	DIMENSION STD.	B16.48							
	JIA				TYPE	FIG.8 FLANGE			SPACER & BLIND				
		Ê	۶ 		STUD BOLTS (FULLY THREADED)	A 193 GR B7, B-1	8.2	•					
T A	ΤP			BOLTING	NUTS (HEAVY HEXAGONAL)	A 194 GR 2H, B-1	8.2						
	P			GASKET	TYPE ,MATERIAL AND Dmn. STD.	SPIRAL 300# , SP.	WND SS316+GRAPHITE FI	LED, B-16.2	0-ANSI B16.5,				
; بن	EI				MATERIAL	ASTM A 105	ASTM A 234 GR.WPB	(CHARPY)					
+ 00				ELBOW-90	END DETAIL	SW,6000# SW,3000#	BW, 1.5D						
<b>#</b> (	Ē				DIMENSION STD.	B-16.11	B-16.9						
B1/		É			MATERIAL	ASTM A 105	ASTM A 234 GR.WPB (	CHARPY)					
<u>ک</u>					END DETAIL	SW,6000# SW,3000#	BW, THICKNESS TO MAT	CH PIPE THIC	CKNESS				
		Ň		ITRED	DIMENSION STD.	B-16.11	B-16.9						
PPE F	- - - -	$\left  \right $			MATERIAL	ASTM A 105	ASTM A 234 GR.WPB (	CHARPY)					
		/		CAP	END DETAIL	SCRF6000# SCRF3000#	BW,THICKNESS TO MAT	CH PIPE THIC	KNESS				
					DIMENSION STD.	B-16.11	B-16.9						
ين  ≡	°	-+			MATERIAL	ASTM A 105	ASTM A 234 GR.WPB (	CHARPY)					
					END DETAIL	SW,6000# SW,3000#	BW, THICKNESS TO MAT	CH PIPE THIC	KNESS				
				FITTING	DIMENSION STD.	B-16.11	B-16.9						
2	MEC	≱╟			TYPE	COUPLING FULL,HALF LH.,RED.	RED. CON. RED. EC	C.					
		c			MATERIAL	ASTM A 105	ASTM A 105 (CHARPY)						
μI	"  -	4			END DETAIL	SW,6000# SW,3000#	BW						
		Þ		O'LET	DIMENSION STD.	MSS-SP97	MSS-SP97						
		기둥			TYPE	SOCKOLET	WELDOLET						
	<u>הן און און און און און און און און און או</u>	- Z				I	I	1					
	ן דר מין איז	4											
											Pag	e 81 of 644	
			+	m	0	0	m		. т	0		<b>_</b>	



		>		$\rightarrow$	8	с		D		m	-	η	G		<u> </u>
APPROVED	DRWN	DSGN	SECTIO	REV NO											
		K.P.	ž	ĂTE										_	
				Ş		PIPELINE/	PIPING I	DESIGN CODE	ASME B 3	31.8/ OISD 2	226	DESIGN FA	CTOR - 0.5		
თ		PA	입	m		ITEM	NOMINAL	DIAMETER (INCHES)	0.50 0.75 1.00 1.50	2.00 3.00 4.00 6	.00 8.00 10.0	12.0 14.0 16.0	18.0 20.0 22.0 24.0		
		m	~   ~				WALL THIC	KNESS (MM/SCH)	S160 S160 XS XS	XS STD STD S	STD STD STD	S40 S40 S30	S30 S30 S30 S30		
A.G.		된윍	ភ្				MATERIAL			ASTM A333 GR	6				
ANG,			AS				DIMENSION	STD.		B36.10					
^P		DAT		닖			METHOD O	F MANUFACTURE, ENDS	SEAMLESS PE	SEAML	ESS BE	BE,	SAW		
				SCR			MATERIAL	AND GRADE		ASTM A 350 GF	R. LF2, CL-I				
			SION	P		FLANGE	TYPE, FLA	NGE FACING	SW. RF 125AARH	WN. THICKNESS	5 ТО МАТСН РІ	PE THICKNESS,	RF 125AARH		
	_	<b>-</b>	N N	Š			DIMENSION	STD.	B16.5					_	
″ <b>≤</b> -	閜	A				BLIND	MATERIAL	AND GRADE		ASTM A 350 GF	R. LF2, CL-I			_	
		T				FLANGE	FLANGE FA	ACING	RF 125AARH					_	
	-	K					DIMENSION	STD.	B16.5					_	
		Ă		H	_		MATERIAL	AND GRADE		ASTM A 350 GR	. LF2, CL-I			4	
7 99		ЪЩ.		₽		BLANK	FLANGE FA	ACING		FF 125AARH					
	Ĕ	ି କି କି		┢			DIMENSION	STD.		B16.48					
R,		JE A		PPR			TYPE		FIG.	B FLANGE	SP/	ACER BLIND			
4 3 4	Ŕ	d í		P			STUD BOL	TS (FULLY THREADED)	/	A 320 GR L7, B-	18.2				
	⊇l	T				BOLING	NUTS (HEA	AVY HEXAGONAL)		A 194 GR 4, B-1	8.2				
<i>T</i>	5	PI				GASKET	TYPE, MAT	ERIAL AND Dmn. STD.	SPIRAL, SP.WND S	S316+GRAPHITE I	FILLED, B-16.2	0-ANSI B16.5,			
							MATERIAL		ASTM A350 GR.LF2	ASTM /	A 420 GR.WPL6	S AST	M A 420 GR.WPL6W		
	<u></u>					ELBOW-90	END DETAI	L	SW,6000# SW,3000#	BW	, 1.5D				
P_0	א	E				ELBOW-45	DIMENSION	STD.	B-16.11	B-	-16.9				
			ĔFEF				MATERIAL		ASTM A350 GR.LF2	ASTM /	420 GR.WPL6	S AST	W A 420 GR.WPL6W		
ω	_					T-EQUAL	END DETAI	L	SW,6000# SW,3000#		BW				
AP	ŝ	051	, I			I-RED	DIMENSION	STD.	B-16.11	E	3–16.9				
	= п	°				CAP	MATERIAL		ASTM A350 GR.LF2	ASTM /	420 GR.WPL6	AST	M A 420 GR.WPL6W	-	
5.	•		11				END DETAI	L	SCRF6000 SCRF3000	вw, тнк то	MATCH PIPE TH	HICKNESS		-	
╷╷	z   °					(UPTO 1.5")	DIMENSION	STD.	B-16.11	B-10	5.9			1	
<u>ة</u>   ح	<i>∗</i> ⊢						MATERIAL		ASTM A350 GR.LF2	ASTM /	420 GR.WPL6	AST	W A 420 GR.WPL6W	7	
							END DETAI	L	SW-6000 SW-3000	BW, THICKNI	ESS TO MATCH	PIPE THICKNES	S	7	
	F	_					DIMENSION	STD.	B-16.11	B-10	5.9				
2			\$₽┣		_	TTTING	TYPE		COUPLING FULL,HALF LH.,RED.	RED. CON.	RED. ECC.				
	9	<u>] </u>	c				MATERIAL		ASTM A350 GR.LF2					_	
		4   -	4				END DETAI	L	SW,6000# SW.3000#	BW					
1	<b>۱</b>	<u> </u>	₽L			OLET	DIMENSION	STD.	MSS-SP97	MSS-SP97					
	Ϊ E		击ᄝ				TYPE		SOCKOLET						
			N N N N N				<u> </u>				]			_	
0 REV		_												Page 83 of (	644
		<u> </u>		<del>- \</del>		0		0		m		η	କ ଜ		<u>т</u>

		⊳				8	0		D		m		г			G		Т	
APPR	DRW	DSG	SEC	REV															
	Ž	Ż		N		ANSI		15 M	TEMP °C	-29	38.0	50	100	150	200				F
		P.		ÂTE		CLASS: 1	30 # ALLOWANCE	<u>:</u> 1.3 IVII	<sup>VI</sup> PRESS. KG/CM <sup>2</sup> g	19.98	19.98	19.57	18.05	16.11	14.07				
ი				ZONE		SERVIC	E : NATURAL (	GAS	BA	SE MA	TERIAL:	CARBO (MATE	on stee Rial gf	EL Roup 1	.1)				
			∃  ⊼ ∞			NOTES: -													0.
A.GA		H.K.	CHKD GA			I. ALL VENTS	S & DRAIN SHALL	BE PRO	IDED WITH PLUG VA	LVE UNLE	SS MENT	TIONED O	THERWISE	E IN P&	IDs.				
		+	_ ∾			Z. FITTINGS : 3. WALL THIO	CKNESS FOR LINE	AMLESS CO PIPE USEF	) IN VARIOUS SECTION	NS SHAL	I BE AS	PFR TA	BIF-1 O	UNSTRU	CHUN TO	AND ABO	VE.		
			Ħ	REV		4. BALL VAL	VE TO BE USED	IN MAINLIN	E SHALL HAVE BUT	T WELDED	ENDS E	XCEPT F	OR THE	VALVES	USED FO	R HOT TAF	PPING		
				SIONS		WHICH SH	ALL BE ONE SIDE	E BUTT WE	LDED AND OTHER S	IDE FLAN	GED.								
<b>თ</b>	D		Z	0		5. DESIGN PF	RESSURE & TEMP	P. FOR PIP	ELINE AND RELATED	FACILITIE	S ARE 1	9 Kg/Cn	n²g & (-	-29° TO	+65°C)	RESPECTIV	ÆLY.		
	PIN		AT		·	7. PRESSURE	-TEMPERATURE	RATING IN	DICATED ARE FOR F	_ANGES (	NLY IN A	ACCORDA	NCE WITH	H ANSI	B 16.5				
	G V		UR			B. FOR VALV	ES, STEELPIPE AN	D ASSOCIA	ATED STEEL COMPON	ENTS OF	2" AND		NOTCH T	TOUGHNE DATA S	ESS PROP	ERTIES SH	ALL BE A	AS SPEC	IFIED
H	ATF	ъ	AL			AT STATIC	NS. BRANCH COM		SHALL BE AS PER	BRANCH	CONNEC <sup>-</sup>	TION TAE	BLE BELO	W		10.			-
150	RIA	RC	G			0. ALL BUTT	WELDS SHALL E	3E 100% R	ADIOGRAPHED.										
#=i		JE	Â	APPR	1	1. 100% OF	SOCKET WELD S	SHALL BE	SUBJECTED TO MPI/	DPT.									
<b>₽</b>	PΕ	CI	Ħ	H		2. PRESSUR	E-TEMPERATURE	RATING O	F VALVE BODY SHAI	L BE AS	PER API	6D.							4
1A)			Ĭ₽			4. FOR PIPE	LINE SPECIALITY	ITEMS (SC	RAPPER TRAP, FLO	V TEE, IJ,	LR BEN	DS ETC.)	AND TH	IEIR MAT	FERIAL DE	SCRIPTIONS	s F	36	
	CAT		EL			REFER D	ATA SHEET OF RE	ESPECTIVE	ITEMS.	TON								32	
H	NO		IN			STATIO	N PIPING M	IATERIA	L SPECIFICAT	<u>10N</u>	F					r	┓┥┥┥	28	F
	S		F	R		ITEM	SIZE		DESCRIPTION		r	F TEES I				┍┛╅		22	ES)
3				ERENCE		MAINTENANCE JOINTS	ALL	FLGD., BL	JT TO BE KEPT MINIM	UM		H H. CO	JPLING TO PIPE	_		_┎╃┼┼┼		18 16	(INCHE
APPE	SCALE	\$ \$		S.			1.5" & BELOW	SOCKET	WELD			R REINFO	DRCED	_				14	DIA
NDIX	••		<b>&gt;</b> )			PIPE JOINTS					[	T TEES	SW/		┍┛┼┼	┼┼┟┢┦		8.0	NAL
$ \hat{-} $	N.T.S	THE TAY					2" & ABOVE	BOLL MEI	LDED		t	D TEE W	ITH RED.		┎┨┼┼┢╸	╺╾┛		5.0	WO -
	<u>~</u>			11			ON LINES $\leq 1.5$ "	3/4", AS	PER MEC/SD/05/2	/15/03				┍┿╃	<mark>┊╴┲╼┛</mark>			4.0 3.50 3.00	L L L
		Z				DRAINS	ON LINES > 2"	3/4" OR	AS PER P&ID. MEC/S	SD/05/21	/15/01		Г	╤┛┼┼╵	╤┏╃┽┼╴			2.50	Ъ ,
2		E	H,	,										╎╏┼┼				1.50	RAN
		Co	नक			VENTS	ON LINES $\leq 1.5$ "	3/4", AS	PER MEC/SD/05/2	/15/03			┏╃╧╀╂	┿┩┼┼			▋┼┼┼┼	1.00	<b>□</b>
		$\mathbf{Z}$	1			VENTS	ON LINES $\geq 2$ "	3/4" OR	AS PER P&ID, MEC/S	SD/05/21,	/15/01							0.50	-
	S)	E	य				1 5"	FLGD, INST	L AS PER MEC/SD/05/	21/15/02		C C	0.1-1-20	20.01 0.01 0.01 0.01 0.01 0.01 0.01 0.01	-10 -10 -10 -10 -10 -10 -10 -10 -10 -10	14 16 18 20 22 22	28 32 34 37	8	
	<u>+</u>   	M	ਸ਼੍ਰੇ	RG. NO						, , -,			RUN	PIPE NO	MINAL DIA	(INCHES)			
	워		ĊN,			PRESS CONN.	3/4"	SCH. 160	NIPPLE WITH BALL V	ALVE TO	SPEC.								-
	2)	Ð					-	INSTALLA <sup>-</sup>	TION AS PER MEC/SD	/05/21/1	5/05					_			
	REV															Р	age 84 c	dt 644	
I		<u> </u>				ω	0		D		m		г			G		I	

		70		<u>₩</u>								
	DR DS	SE(	REV									
	N S		ð									
5 [		NA	DATE									
L		Ξ.	2	-								
			Ř				ASME D	31 9 / OISD 226			0.5	
		≓  ¥			PIPING DESIGN (							
	Ξ	<u>2</u> %		TIEM	NOMINAL DIAMETER (II	NCHES)	0.50 0.75 1.00 1.50	2.00 3.00 4.00 6.00 8.0		16.0 18.0 20.0 2	2.024.0	
	~	⊖ G <sub>A</sub>			WALL THICKNESS (MM)	/SCH)	S160 S160 XS XS	ASTM A106 GR.B	0   520   520   510		.9 8.7	
		۲. N		PIPE			ASIM AIUO GR.B	(CHARPY)		5L2		
$ \rightarrow$			RESCR		DIMENSION STD.					CAW DE		
					METHOD OF MANUFAC	-	SEAMLESS PE	SEAMLESS BE	^	SAW, BE		
_			NS NO	FLANGE	TYPE ELANCE EACINC	-	SW PE 125AAPH	WN THICKNESS TO M		NESS DE 1254 AR		
PIP		NA			DIMENSION STD.	,	BI6 5	WIN. THICKINESS TO MA	ATCH FIFE THICK	NESS, RF IZSAAN		
Z					MATERIAL AND GRADE	:	ASTM A 105	ASTM A 105 (CHARPY	0			
		GH		BLIND	FLANGE FACING		RF 125AARH		•			
A	<b>.</b>	Ă		FLANGE	DIMENSION STD.		B16.5					
<u></u>	PF	<b>「</b>	BA		MATERIAL AND GRADE	:	ASTM A 105	ASTM A 105 (CHARPY	)			
A	õ	<u>ନ୍</u>	₽	· · · · · ·	FLANGE FACING		FF 125AARH	· ·				
	JE	AS	PPR	BLANK	DIMENSION STD.		B16.48					
Ř	ğ		$\dashv$		TYPE		FIG.8 FLANGE			SPACER & BLIND		
	Ţ	PI			STUD BOLTS (FULLY T	THREADED)	A 193 GR B7, B-1	3.2				
0		PH		BOLTING	NUTS (HEAVY HEXAGO	DNAL)	A 194 GR 2H, B-1	3.2				
ΕA		E		GASKET	TYPE, MATERIAL AND	Dmn. STD.	SP. WND SS 316+G	RAPHITE FILLED AS PER	B 16.20/ANSI E	3 16.5		
9		$\Xi$			MATERIAL		ASTM A 105	ASTM A 234 GR.WPB (	CHARPY)	ASTM A 234 GR	.WPB-W (CHARPY)	
S		E		ELBOW-90	END DETAIL		SW,6000# SW,3000#	BW, 1.5D				
		!		ELBOW-45	DIMENSION STD.		B-16.11	B-16.9				
					MATERIAL		ASTM A 105	ASTM A 234 GR.WPB (	CHARPY)	ASTM A 234 GR	.WPB-W (CHARPY)	
A	50 <sup>251</sup>		<i>v</i> i	T-EQUAL	END DETAIL		SW,6000# SW,3000#		BW	1	. ,	
m				T-RED	DIMENSION STD.		B-16.11		B-16.9			
		▶/			MATERIAL		ASTM A 105	ASTM A 234 GR WPB (				
N.T.	~****	~		CAP	END DETAIL		SCRF6000 SCRF3000		NESS TO MATCH	PIPE THICKNESS		
اي					DIMENSION STD.		B-16.11	21, 11100	B-16.9			
					MATERIAL		ASTM A 105	ASTM A 234 GR.WPB (	CHARPY)	ASTM A 234 GR	.WPB-W (CHARPY)	
					END DETAIL		SW,6000# SW,3000#	BW, THICKN	NESS TO MATCH	PIPE THICKNESS	. ,	
	$ \mathbf{E} $				DIMENSION STD.		B-16.11		B-16.9			
	CON	मेकान		ritting	TYPE		COUPLING FULL,HALF LH.,RED.	RED. CON. RED.	ECC.			
					MATERIAL		ASTM A 105	ASTM A 105 (CHARPY)				
		3)			END DETAIL		SW.6000# SW 3000#	BW				
¥		F		O'LET			MSS-SP97	MSS-SP97				
N		$\overline{\mathbf{v}}$	s									
1 '	$ \mathbf{T} $	S I					SUCKULET	WELDULEI				



		Þ					c		D		m		П	G		т
APPROVED	DRWN	DSGN	SECTI	REV NO												
		Σ.P. NA	2													
		·	i		;		PIPELINE/	PIPING	DESIGN CODE	ASME B	31.8/ OISD	226	DESIGN F	ACTOR - 0.5		
ი		N N	e la		i	-	ITEM	NOMINAL	DIAMETER (INCHES)	0.50 0.75 1.00 1.5	0 2.00 3.00 4.00	6.00 8.00	0 10.0 12.0 14.0			
								WALL THIC	KNESS (MM/SCH)	S160 S160 XS XS	XS STD STD	STD STD	STD STD STD			
A.G		표 율	ີ່ດີ					MATERIAL			ASTM A333 G	R.6				
ANG		⊼ a	N A				PIPE	DIMENSION	STD.		B36.10					
_ <sup>₽</sup>		DA	2		,			METHOD O	F MANUFACTURE, ENDS	SEAMLESS PE	SEAM	ILESS BE				
		Ē	1	R				MATERIAL	AND GRADE		ASTM A 350 G	R. LF2, 0	CL-I			
							FLANGE	TYPE, FLA	NGE FACING	SW. RF 125AARH	WN. THICKNES	S TO MA	TCH PIPE THICKNESS	, RF 125AARH		
		,	_	IS NO				DIMENSION	STD.	B16.5	•					
זΩ₽							BLIND	MATERIAL	AND GRADE		ASTM A 350 G	R. LF2, (	CL-I			
- N		;	1				FLANGE	FLANGE FA	ACING	RF 125AARH						
EG	]	(	Π					DIMENSION	STD.	B16.5						
PA			A2					MATERIAL	AND GRADE		ASTM A 350 G	R. LF2, C	)L—I			
- 문탄		P	-	H H	2		BLANK	FLANGE F	ACING		FF 125AARH	I				
		202	G		╀			DIMENSION	STD.		B16.48					
R.		JI	À					TYPE		FIG	.8 FLANGE		SPACER & BLIND			
J 12P		E	<b>J</b>	e	i	-		STUD BOL	TS (FULLY THREADED)		A 320 GR L7, B	-18.2	•			
		Ξ	P				BOLTING	NUTS (HEA	AVY HEXAGONAL)		A 194 GR 4, B-	18.2				
FIC #			P				GASKET	TYPE, MA1	ERIAL AND Dmn. STD.	SPIRAL, SP.WND	SS316+GRAPHITE	FILLED, E	B-16.20-ANSI B16.5,			
Ă			E					MATERIAL		ASTM A350 GR.LF2	2 ASTM	A 420 G	R.WPL6			
- <u>A</u> Q			Ħ				ELBOW-90	END DETA	L	SW,6000# SW,3000	# B'	W, 1.5D				
₽5		i	E				ELBOW-45	DIMENSION	STD.	B-16.11	В	8–16.9				
								MATERIAL		ASTM A350 GR.LF	2 ASTM	A 420 G	R.WPL6			
<u>ы</u>				R			T-EQUAL	END DETA	L	SW,6000# SW,3000	#	BW				
A S		051	_	X			T-RED	DIMENSION	STD.	B-16.11		B-16.9				
R								MATERIAL		ASTM A350 GR.LF	2 ASTM	A 420 G	RWPI6			
<u>⊿</u>	000		<b>&gt;</b> )					END DETA	L	SCRF6000 SCRF300	BW. THK TO	MATCH F	PIPE THICKNESS			
X		V					PLUG   (UPTO 1.5")	DIMENSION	STD.	B-16.11	B-1	16.9				
.T S.T								MATERIAL		ASTM A350 GR.LF	2 ASTM	A 420 G	R.WPL6			
								END DETA	L	SW-6000 SW-300	O BW, THICKN	NESS TO	MATCH PIPE THICKNE	SS		
	_							DIMENSION	STD.	B-16.11	B-1	16.9				
2			ਸ				FITTING	TYPE		COUPLING FULL,HALF LH.,RED.	RED. CON.	RED. E	ECC.			
	9	21	3					MATERIAL		ASTM A350 GRUE	2 ASTM A 350 G	R.LF2				
	2	4	ц					END DETA	L	SW,6000#ISW.3000			1			
	⊨	┛	ゕ				O'LET	DIMENSION	STD.	MSS-SP97	MSS-SP97		1			
1		3	Ť	묽				TYPF			WELDOLET		1			
2 OF 2) F			रिंड	3. NO.			L				<b></b>				Pa	age 87 of 644
						<u> </u>									10	
		≻			1	<b>w</b>	0			1	т	1	די	ເ ດ		I I

# **ANNEXURE - I**

# **DATA SHEETS**

#### DATA SHEET INDEX

Sr. No.	Data Sheet No.	Description	REV.
1.	MEC/WINO/05/28/M/001/DS/BV/76 - 83	Ball Valve	0
2.	MEC/WINO/05/28/M/001/DS/PV/76 - 83	Plug valve	0
3.	MEC/WINO/05/28/M/001/DS/CV/76 - 81	Check valve	0
4.	MEC/WINO/05/28/M/001/DS/GV/76 - 83	Globe valve	0
5.	MEC/23SF/05/28/M/001/DS/001	Pressure Safety Valves & CRV	0
6.	MEC/23SF/05/28/M/001/DS-017/001	Cartridge Filter	0
7.	MEC/05/E5/DS – PG	Pressure Gauges	1
8.	MEC/05/E5/DS – TG	Temperature Gauges	1
9.	MEC/05/E5/DS – PT	Pressure Transmitter	1
10.	MEC/05/E5/DS – RTD	Resistance Temperature Detector	1
11.	MEC/05/E5/DS – TT	Temperature Transmitter	1
12.	MEC/05/E5/DS – DPT	Diff. Pressure Transmitter	1
13.	MEC/05/E5/DS/DPG-01	Diff. Pressure Gauge	0
14.	MEC/05/E5/DS – LS	Limit Switch	1
15.	MEC/05/E5/DS – PCV	Pressure Control Valve	1
16.	MEC/05/E5/DS – SDV	Slam Shut Valve	1
17.	MEC/ 05/E5/DS-UFM	ULTRASONIC FLOW METERS	1
18.	MEC/05/E5/FC-SIV	Special Instruction to Vendor for FC	1

Sheet 1 of 2

Sr. No.	Data Sheet No.	Description	REV.
19.	MEC/05/E5/DS – FC/02	Panel Mounted Flow Computer	1
20.	MEC/05/E5/DS-MSS	METERING SUPERVISORY SYSTEM	1
21.	MEC/05/E5/DS – LEL	LEL Detection System	1
22.	MEC/05/E5/GC-SIV	SPECIAL INSTRUCTIONS TO THE VENDOR (FOR GAS CHROMATOGRAPH - GC)	1
23.	MEC/ 05/E5/DS-GC	GAS CHROMATOGRAPH	1
24.	MEC/ 05/E5/Spec Laptop	LAPTOP (TYPICAL)	1
25.	MEC/SD/05/E5/I/SA-01	Proposed System Architecture for Metering Skid	0
26.	-	TYPICAL DATA SHEET FOR HYDRAULIC ACTUATOR	-
27.	MEC/ 05/E5/DS-H2S	ONLINE H2S ANALYSER	0
28.	MEC/ 05/E5/DS-MA	ONLINE MOISTURE ANALYZER SYSTEM	0

1.0	Valve I	Manufactu	rer		:					
2.0	Valve	Size (NB),	mm (inch)		:	ANSI Rati	ing : <b>150#</b>		Design Standard : API 6D	
3.0	MECO	N's Techn	ical Specific	ation No.	: MEC/	TS/05/21/002, Rev-1, E	d-1			
4.0	Conne	cting Pipel	line Design	Pressure, b	ar	: 19 kg/cm2	Design Temperature	e, °C : -29°C to +65	5°C	
5.0	Conne	cting Pip	e Specifica	tion	:					
5.1	Materia	al tor (OD) in	nm (inch)		:					
5.3	Thickn	ess, mm	nin (inch)		:					
6.0	Valve	Construct	tion Design							
6.1.	Config	uration	-		: Reduce	d Bore	Full Bore			
6.2.	End Co	onnections	5		: Flange	ed as per ASME B16.5			Butt Welded as per ASME B16.25	
6.3.	Flange	s (wherev	er applicabl	e)	:a) RF			RT		
6.4	Dell M				b) Serra	ated Smoo	th (125 to 200 microinches	s AARH)		
6.4 6.5	Valve b	ounting			: Filly \	Ng Ballupto 8" and ir Velded	Two/Three Piece Bolte	ed	Either	
0.0	rano.	souj tjpo								
7.0	Valve	Material S Part	pecificatio	n	Sn	ecified Material		Material Offe	ered (Equivalent or superior)	
7.1	Body	un	A 216 Gr. W	CB/A 234 G	r. WPB/ A 3	352 Gr. LCB/A 350 Gr. LF	2			
7.2	Bal		(A 216 Gr.W	/CB/A 234 G	r.WPB/ A 3	52 Gr.LCB/AISI 4140)+75	µENP coating/AISI410			
7.3	Body Se	eat Rings	VITON/ DEV	LON for Flo	ating type	& AISI 4140 + 75μENP cc	ating/AISI 410 for			
74	Seat Se	eat	VITON for T	ounted type MBV						
7.5	Stem		AISI 4140 +	75 micron E	NP coating	J/AISI 410 (No casting)				
7.6	Stem S	eals	VITON/PTFI							
1.1		olts/ Nuts	ASTM A 193	3 Gr. B7/ A1	94 Gr. 2H Irdness					
8.0	Corros	ion Allowa	ince		: 1 <b>.5</b> mi	m	Service :			
9.0	Locatio	on			: Above	Ground	Buried			
10.0	Stem E	xtension I	Requiremen	t	: Yes		No			
11.0	Gear C	Operator R	equirement		: Yes	$\checkmark$ for 6" and above	e No √ for 4" a	and below		
12.0	Actuate	or Require	ment		: Yes		No			
13.0	Fire Re	esistant De	esign Requii	rement	: Type tes	t as per API 607 for Flo t as per API 6EA for Tr	oating Ball Valve	/alvo		
14.0	Valve '	Testing R	equirement	t	. Type tes			valve	1	
					Test Press	sure (min.), kg/cm²(g)	Minimum Durat	tion, minutes		
14.1	Hydros	static Test		Body		32	As per A	API 6D		
14.2	Air Tes	2t		Seat		23	As per A			
14.2	/ 11 100					0.0 - 1			1	
15.0	Anti-St	atic Testin	ng Requirem	ent	: As pe	r Standard API 6D (Lat	est Ed.)			
16.0	Valve	Painting S	Specificatio	n						
16.1	Surface	e preparat	ion by Shorl	Blasting a	s per grade	SA 2 1/2, Swedish Star	ndard SIS-055 909			
16.2	For ab	ove ground	d installation	n-Three coa	ts of corros	sion resistant paint shall	be applied with minimu	m thickness of 300	micron	
	shall b	e finalized	during drav	ving approv	al stage.		our of paint shade shall i	be RAL-7030, Howe	ver any change in colour	
17.0	Lock O	pen/ Lock	Close/Norn	nally Close	Requireme	ent : As indicated in	P&ID.			
	Notes:	Vent & D	rain conne	ction for fl	pating & tr	unnion mounted valve	s shall be provided as	per details mentic	oned in MECON's TS.	
	2	Inspection	and Testin	a shall be r a shall be a	ead in conj is per attac	hed QAP, this Data She	ect. MECON's T.S., API	6D and other releva	JUZ, Rev-1,Ed-1. ant standards.	
	3.	Stops sha	all be provide	ed for posit	ve alignme	ent of ball with ports and	ensure proper installation	on of handle.		
	4.	Short patt	ern valves (	as per API	6D or othei	wise) are not permitted	Only long pattern valve	es are to be supplied	d. The When a second of the	
	5.	charpy V as ner Cl	-notch & Ha	rdness test	tor body, b stively Mir	ody adaptor, end flange	s, ball, body seat rings,	stem & studs / nuts B16.34	shall be conducted	
	6.	Compress	sed asbesto	s fibre (CA	-) shall not	be used for body sealin	g / gasket materials.			
	7.	Material for	r body shall h	ave a guara	nteed minim	um yield strength of	psi. In case the same c	annot be guaranteed,	valves shall	
	8	be provide	d with a 500 i	mm pup piec	e (integrally	welded to the valve on ear	ch side) with strength equivities and minimum and minimum.	valent to that of the co	onnecting pipe - <b>N.A.</b>	
	9.	Valves sh	all be inspe	cted and an	proved by	Purchaser before despa	atch.	ar pipe enu) shall ht	st se more than 0.5 % of pipe OD.	
REV. NO.	DATE	ZONE		DESCRIPT	ONS	BY	APPRD			
SECTIO		CESS 8	PIPING	REVISION				REFERENCES	DRG. NO.	
520110										
	NAME	DATE	CHKD	DATE						
DSGN	PM	25.04.12	AKJ	25.04.12	PROJECT	:			MECON LIMITED	
URWN				L				SCALE :		REV/
APPRO\	/ED			O.P. Jain	DATA	SHEET FOR BALL (NB> 2")	/ALVES	DATA SHEET NO.: ME	C/WINO/05/28/M/001/DS/BV/76	0

1.0	Valve Man	ufactu	rer		:				
2.0	Valve Size	(NB),	mm (inch)		: ANSI Rating :	: 150#		Design Standard :API 6D	
3.0	MECON's	Techn	ical Specific	ation No.	: MEC/TS/05/21/002, Rev-1, E	d-1			
4.0	Connecting	g Pipel	line Design	Pressure, b	ar : <b>19 kg/cm2</b>	Design <sup>-</sup>	Temperature, °C :-	-45 to 65°C	
5.0 5.1 5.2 5.3	<b>Connectir</b> Material Diameter ( Thickness,	n <b>g Pip</b> OD), r , mm	<b>e Specifica</b> mm (inch)	tion*					
6.0 6.1. 6.2.	Valve Con Bore* End Conne	struc	tion Design	1	: Reduced	Full	]		
6.3. 6.4	Flanges (w Valve Type	vherev e	er applicabl	e)	: a) RF FF b) Serrated Smoo : Floating upto 8" and Trunni		RT 00 microinches AARH <b>d above 8''</b>	i) NA NA	
7.0	Valve Mat	erial S	Specificatio	n					
7.1	Part Body		ASTM A350	Gr. LF2/ A3	Specified Material 52 Gr. LCB		N	Material Offered	
7.2	Ball		[(ASTM A35	0 Gr. LF2/ A	352 Gr. LCB/SS 316 )+75 microns E	NP]			
7.3	Body Seat F	Rings	VITON/DEV LCB/SS 316	LON for Floa 6 )+75 micro	ating type and (ASTM A350 Gr. LF2/ ns ENP coating for Trunnion Mount	A352 Gr. ed type			
7.4 7.5	Seat Seal Stem		VITON for T I(ASTM A35	MBV 0 Gr. LF2/S	S 316 )+75 microns ENPI(No casting	1)			
7.6 7.7	Stem Seals Stud Bolts/	Nuts	VITON/ PTF ASTM A320	E Gr. L7/ A19	4 Gr. 4				
8.0	Corrosion	Allowa	nas min. 35 ince	HRC hardn	ess : <b>1.5 mm</b>	Service :			
9.0	Location				: Above Ground	Buried			
10.0	Stem Exte	nsion	Requiremen	ıt	: Yes	No		_	
11.0	Gear Oper	ator R	equirement		: Yes $$ for 6" and above	No _√	upto 4"		
12.0	Gas Powe	red Ac	tuator Requ	iirement	: Yes	No			
13.0	Fire Resist	tant De	esign Requii	rement	: Type test as per API 607 for Floa	ating Ball Va	alve ed Ball Valve		
14.0	Valve Tes	ting R	equiremen	t	Test Pressure (min.), kg/cm <sup>2</sup> (g)	Minimum	Duration, minutes		
14.1	Hydrostatio	c Test		Body	32	As	per API 6D		
44.0	Ain Tast			Seat	23	As	per API 6D		
14.2	Arti Static	Tostir	a Pequirem		5.6-7	As (0.2 Ed.)	per API 6D		
16.0	Valve Pair	ntina (	Specificatio	n	. As per Standard Arrob (200	02 Eu.)			
i)	Surface pr	eparat	ion by Short	t Blasting as	s per grade SA 2 1/2, Swedish Sta	ndard SIS-0	55 909.		
ii)	For above	groun	d installatior	1-Three coa ich coat sha	ts of corrosion resistant paint shall II be within 80 to 120 micron). Cold	l be applied	with minimum thick	ness of 300 micron -7038, however any change in co	lour
	shall be fin	alized	during draw	ving approv	al stage.	our or paints			
17.0	Lock Oper	l/ Lock	Close Req	uirement	: As indicated in P&ID	المالية ومعرفه	oo nor dotaila		
	1 Thi	s Valv	e Data Shee	et shall be r	ig a trunnion mounted values shall lead in conjunction with MECON's $]$	De provided	as per details ment pecification No. ME	C/TS/05/21/002,Rev 1 ,Ed. 1	
	2. Insp 3 Sto	pection	n and Testin	ig shall be a	is per attached QAP, this Data She	eet, MECON	Vs T.S., API 6D and	d other relevant standards. andle	
	4. Sho	ort patt	tern valves (	(as per API	6D or otherwise) are not permitted	. Only long	pattern valves are t	o be supplied.	
	5. Cha	arpy V	-notch & Ha	rdness test	for body, body adaptor, end flange	es, ball, body	y seat rings, stem 8	& studs / nuts shall be conducted	
	6. Cor	npress	sed asbesto	s fibre (CAF	F) shall not be used for body sealin	ig / gasket n	naterials.		
	7. Mat	erial fo	r body shall h d with a 500	nave a guara mm pup piec	nteed minimum yield strength of	psi. In cas	se the same cannot be strength equivalent to	e guaranteed, valves shall	
	8. For	weldi	ng end, the	out of round	Iness (i. e. difference between max	kimum and r	ninimum ID at pipe	end) shall not be more than 0.5%	of pipe OD
	9. Val	Ves sh	iall be inspe I	cted and ap	proved by Purchaser before despa	atch.			
NEV. NU.			L	REVISION	IS BY	AFPRU	REFERENCES	DRG. NO.	
SECTIO		SS &			CLIENT :				
DSGN	PM 25	.04.12	AK.I	25.04 12	PROJECT :		Antin State	MECON LIMITED	
DRWN			,						
	/FD			0 P Jain		_VES		C/ WINO/05/28/M/001/DS/DV/77	REV
AFF RUV				J. F. Jaili	(ND <u>~</u> 2 )		DATA SHEET NU.: ME		U

1.0	Valve	Manufactu	ırer		:				
2.0	Valve	Size (NB),	mm (inch)		: AN	SI Rating : 150#	I	Design Standard:API 6D	
3.0	MECO	N's Techr	nical Specifi	cation No.	: MEC/TS/05/62/003, Rev	/-2			
4.0	Conne	cting Pipe	line Design	Pressure, k	kg/cm <sup>2</sup> (g) : <b>19 kg/cm2(g</b> )	Design Te	emperature, °C : •	-29°C to 65°C	
5.0	Conne	ecting Pip	e Specifica	ition					
5.1	Materi	al			:				
5.2	Diame	ter (OD), r	mm (inch)		:				
5.3	Thickn	ess, mm			:				
6.0	Valve	Construc	tion Desigr	ו					
6.1.	Patterr	1			Short	Regular		Venturi	
6.2.	End C	onnections	S		: Flanged both ends		Flange	ed as per ASME B 16.5	
					: Butt Weld both ends		Butt We	eld as per ASME B16.25	
<u> </u>	Element	. (		( <b>a</b> )	: Flanged one end, butt	weld other end			7
0.3.	Flange	s (wherev	er applicabl	le)	b) Sorrotod	Smooth (125 to 200			
						31100111 (125 to 200	microinches AARH		
7.0	Valve	Material S	Specificatio	n					-
7 1	Dedu	Part	ACTM 4046		Material		Material Offer	red (Equivalent or Superior)	_
7.1	Plug		(ASTM A210	Gr. WCB/A	234 Gr. WPB A234 Gr. WPB) + 75 microns F	NP Coating			-
7.3	Cover		ASTM A216	Gr. WCB/ A	234 Gr. WPB	in oounig			-
7.4	Stem		(AISI 4140 +	F 75 microns	ENP Coating)/ AISI 410 (No C	asting)			
7.5	Stem S	eal	PTFE/Grap	hite					
7.6	Stud Bo	olts/ Nuts	ASTM A193	Gr. B7/ A19	4 Gr. 2H				
0.0	NOTE	AISI 410	has min. 35	HRC hardne	ess . 1 E mm	6			
8.0	Corros	ION Allowa	ance		. 1.5 mm	56			
9.0	Locatio	on			: Above Ground	Buried			
10.0	Stem I	Extension	Requireme	nt	: Yes	No 📃			
11.0	Gear C	Operator R	Requirement		: Yes	No			
12.0	Gas P	owered Ac	ctuator Requ	uirement	: Yes	No			
13.0	Fire Re	esistant D	esign Requi	irement	: Type-Test as per Stand	dard API 6FA/ BS	EN: 10497		
14.0	Valve	Testing R	lequiremen	ıt	Test Pressure (min.), kg/cm2	2(g) Minimum D	uration, minutes		
14.1	Hydros	static Test		Body	32	As po	er API 6D		
14.2		2t		Seat	23 56_7	As po	er API 6D er API 6D		
17.2	Air roo				5.6 - 7	A3 pt			
15.0	Valve	Painting \$	Specificatio	on					
15.1	Surfac	e preparat	tion by Shor	t Blasting a	s per grade SA 2 1/2, Swedis	sh Standard SIS-05	55 909. With an in inclusion the all	mana of 200 minut	
15.2	For ab	ove groun	d installatio	n-Inree coa	ats of corrosion resistant pair	it shall be applied v	vith minimum thick	7038 bowever any	
	change	e in colour	shall be fin	alized durin	an be within ou to 120 micror		hade shall be tota		
16.0	Lock C	pen/ Lock	Close Req	uirement :	As indicated in P&ID	1			
	Notoo								
	INOTES:	This Valv	Data She	et shall he r	read in conjunction with MEC	ON's Technical Sn	ecification No. ME	C/TS/05/62/003 Rev2	
	2.	Inspectio	n and Testir	na shall be a	as per attached QAP. this Da	ta Sheet. MECON'	s T.S., API 6D and	d other relevant standards.	
	3.	Stops sha	all be provid	led for posit	ive alignment of plug with po	rts and ensure prop	per installation of h	nandle.	
	4.	Charpy V	- notch & H	ardness tes	t for body, plug, cover, stem	& studs/nuts shall I	pe conducted as p	er Clause No.: 3.4 & 3.5	
	F	of TS res	pectively. N	lin. Valve Bo	ody Wall Thickness as per ASM	ИЕ В16.34.	. Nat Annliach	la la	
	6	Bidder sh	all clearly w	rite valves i	material (equivalent to that o	i the connecting pip ior) offered by them	against each par	t/material of valve in the	
		space pro	ovided for V	Vherever bi	dder agrees with valves mate	erial as mentioned	above in MECON'	s data sheet, bidder	
		shall clea	rly indicate	"AGREED"					
REV. NO.	DATE	ZONE		DESCRIPT	IONS BY	APPRD			
				REVISION	NS		REFERENCES	DRG. NO.	
SECTIO	N OIL	& GAS		,	CLIENT :				
	NAME	DATE	СНКД	DATE			(		
DSGN	PM	25.04.12	AKJ	25.04.12	PROJECT :		40 0001 Comp 5h	MECON LIMITED	
DRWN									
							SCALE :		REV
APPRO	/ED			O. P. JAIN	DATA SHEET FOR PL	UG VALVES	DATA SHEET NO .: M	MEC/WINO/05/28/M/001/DS/PV/76	0
					(NB ≥ 2")				

1.0	Valve	Manufactu	Irer		:				
2.0	Valve	Size (NB)	mm (inch)		- AN	SI Rating : 15	50#	Design Standard · API 6D	
3.0	MECO	N's Techr	nical Specific	cation No	MEC/TS/05/62/003 Rev	<i>-</i> 2			
4.0	Conne	cting Pine	line Design	Prossure k	$ra/cm^{2}(a)$ : 19 ka/cm2(a)	– Doci	n Tomporaturo °C :	-15°C to 65°C	
4.0 E 0	Conne			1 1035ule, M	(g/cm/(g) . 13 kg/cm2(g)	Desig	gii reinperature, C.	-43 0 10 03 0	
5.0	Matari	oling Fip	e Specifica	lion					
5.1	wateri	al			:				
5.2	Diame	ter (OD), r	nm (inch)		:				
5.3	Thickn	ess, mm			:				
6.0	Valve Pattor	Construc	tion Design	I	· Short	Po		Venturi	
0.1.						i veç			
6.2.	End C	onnections	5		: Flanged both ends		Flang	ed as per ASME B 16.5	
					: Butt Weld both ends		Butt W	eld as per ASME B16.25	
					Elanged one and butt	wold other e			
					Flanged one end, but				-
6.3.	Flange	es (wherev	er applicabl	e)	: a) RF FF		RTJ	NA	
					b) Serrated	Smooth (125 to	o 200 microinches AARI		1
									-
7.0	Value	Matarial C							
1.0	valve	Material	specificatio	n					7
		Part			Material		Material Offe	red (Equivalent or Superior)	
7.1	Body		A 352 Gr. L	CB/A 350 Gr	LF2				
7.2	Plug		SS316/ A 35	2 Gr.LCB/A	350 Gr. LF2 with 75 µENP co	ating			
7.3	Cover		<b>ASTM A216</b>	Gr WCB/A	234 Gr WPB				
7.4	Ctore		SC 346 (No		250 C+ 1 F2				-
7.4	Stem		33 3 10 (NO	casting) / A	350 Gr. LF2				-
7.5	Stem S	ear	PTFE/Grap	nite					_
7.6	Stud Bo	olts/ Nuts	ASTM A320	Gr.L7 / AST	M A194 Gr.4				
8.0	Corros	ion Allowa	ance		: <b>1.5 mm</b>		Service :		
0.0	Locati	20			· Above Ground	Puriod			
10.0	Stom	- 	Doguiromon	.4					
10.0	Sterri t		Requirement	IL			1		
11.0	Gear		equirement		; res		]		
12.0	Gas P	owered Ac	tuator Requ	urement	: Yes	No			
13.0	Fire Re	esistant D	esign Requi	rement	: Type-Test as per Stand	lard API 6FA/	BS EN: 10497		
14.0	Valve	Testing R	equiremen	t I	Test Pressure (min.). kg/cm2	2(a) Minimu	um Duration, minutes	I	
14.1	Hydros	static Test		Body	32		As per API 6D		
				Seat	23		As per API 6D		
14.2	Air Tes	st			5.6 - 7		As per API 6D		
15.0	Valve	Painting	Specificatio	n L Diantinum		h Otan dand O		•	
10.1	Surrac	e prepara	uon by Shor	i biasting a	s per grade SA 2 1/2, SWedis	an Standard S	13-033 909.		
15.2	For ab	ove groun	d installatioi	n-Three coa	ats of corrosion resistant pair	it shall be app	lied with minimum this	ckness of 300 micron	
	(Perm	issible thi	ckness in ea	ich coat sha	all be within 80 to 120 micron	). Colour of pa	aint shade shall be RA	L-7038, however any	
	change	e in colour	shall be fin	alized durin	g drawing approval stage				
16.0	Lock C	pen/Loc	Close Rea	uirement :	As indicated in P&ID				
	Notos								
	INDICS.	Th:- \ /	Date Of	at ala = II ! · ·	and in continentian 10 MEC				
	1.	i nis vaiv	e Data Shee	et snall be r	eau in conjunction with MEC	UNS LECHNICA	a opecification No. M	EC/15/05/02/003, Rev2	
	2.	Inspection	n and Testin	ig shall be a	as per attached QAP, this Da	ta Sheet, ME0	CON's T.S., API 6D ar	nd other relevant standards.	
	3.	Stops sha	all be provid	ed for positi	ive alignment of plug with po	rts and ensure	proper installation of	handle.	
	4.	Charpv V	- notch & Ha	ardness tes	t for body, plug, cover, stem	& studs/nuts s	hall be conducted as	per Clause No.: 3.4 & 3.5	
		of TS ree	nectively M	in Valve Bo	dy Wall Thickness as ner ASN	1F B16 34			
	E	Motoriel	of the brain			0.0.0-7 .		blo	
	о. С	Dial-1-	n me boay s	man nave	equivalent to that of	ane connectif	ig pipe - not Applical		
	ь.	bluder sh	all clearly w	nie valves i	material (equivalent or super	or) offered by	mem against each pa	annaterial of valve in the	
		space pro	ovided for. V	nerever bi	dder agrees with valves mate	erial as mentio	ned above in MECON	i's data sheet, bidder	
		shall clea	rly indicate '	"AGREED"					
REV. NO.	DATE	ZONE		DESCRIPTI	ONS BY	APPRD			
			•	REVISION		1	REFERENCES	DRG NO	
SECTIO		& GAS						DRG. NO.	
		u GAO			OLILINI .				
	NAME								
	NAME	DAIE	CHKD	DAIE			मेकॉन	l	
DSGN	PM	25.04.12	AKJ	25.04.12	PROJECT :		90 5001 COMP 55	MECON LIMITED	)
DRWN	1	· · · ·	İ						
		L	ļ				CON E	<b>I</b>	
							SCALE :		
\PPRO	/ED			O. P. JAIN	DATA SHEET FOR PL	UG VALVES	DATA SHEET NO.:	MEC/WINO/05/28/M/001/DS/PV/77	0
					(NB ≥ 2")				L
					()		1	D - ··· · 0	1 - 5 1

1.0	Valve I	Manufactu	rer		:					
2.0	Valve	Size (NB)	mm (inch)			ANSLR	ating : 300#		Design Standard · API 6D	
3.0	MECO	N's Techn	ical Specific	cation No.	: MEC/1	S/05/21/002. Rev-1. Ed	-1			
4.0	Conne	cting Pipel	ine Design	Pressure h	ar	: 49 kg/cm2	Design Temperature	°C · -29°C to +65	°C	
5.0	Conne	ecting Pipe	e Snecifica	tion			Boolgir Fomporataro,		•	
5.1	Materia	al	oopeeniea		:					
5.2	Diame	ter (OD), r	nm (inch)		:					
5.3	Thickn	ess, mm			:					
6.0	Valve	Construct	tion Design	ı						
6.1.	Config	uration			: Reduced	d Bore	Full Bore			
6.2.	End Co	onnections			: Flange	ed as per ASME B16.5			Butt Welded as per ASME B16.25	
6.3.	Flange	s (wherev	er applicabl	e)	: a) RF	FF [		RT		
					b) Serra	ited Smoot	h (125 to 200 microinches	AARH)		
6.4	Ball Mo	ounting			: Floatin	g Ball upto 4" and Tru	nnion Mounted above	4"		
6.5	Valve	oody type			: Fully V	Velded	Two/Three Piece Bolte	d 📃	Either	
7.0	Valve	Material S	pecificatio	n						
	F	Part			Sp	ecified Material		Material Offe	ered (Equivalent or superior)	
7.1	Body		A 216 Gr. W	/CB/A 234 G	r. WPB/ A 3	52 Gr. LCB/A 350 Gr. LF2	2			
7.2	Ball		(A 216 Gr.W	/CB/A 234 G	r.WPB/ A 3	52 Gr.LCB/AISI 4140)+75	µENP coating/AISI410			
7.0	Body S	Seat Rings	VITON/DEV	LON for Flo	ating type a	nd AISI 4140 + 75 micron	ENP coating/AISI 410			
7.3	Soot So	oot	for Trunnio	n Mounted 1	уре					
7.5	Stem	ai	AISI 4140 +	75 micron E	NP coating	/AISI 410 (No casting)				
7.6	Stem S	eals	VITON/PTFI	E		( 0)				
7.7	Stud Bo	ots/ Nuts	ASTM A 193	3 Gr. B7/ A1	94 Gr. 2H					
	NOTE	: AISI 410	) has min. 3	35 HRC ha	irdness .		<b>a</b>			
8.0	Corros	ion Allowa	ince		: 1.5 mr	n	Service :			
9.0	Locatio	on			: Above	Ground	Buried			
10.0	Stem E	xtension I	Requiremen	it	: Yes		No			
11.0	Gear C	Operator R	equirement		: Yes	$\checkmark$ for 6" and above	No 📝 for 4" a	and below		
12.0	Actuate	or Require	ment		: Yes		No			
13.0	Fire Re	esistant De	esign Requii	rement	: Type test : Type test	as per API 607 for Flo as per API 6FA for Tru	ating Ball Valve Innion Mounted Ball V	/alve		
14.0	Valve <sup>·</sup>	Testing R	equirement	t	Test Press	ure (min.), ka/cm²(a)	Minimum Durati	ion. minutes	I	
44.4	l li colume co			Dadu		70	A	PI CD		
14.1	Hydros	static rest		Seat		57	As per A	PI6D		
14.2	Air Tes	st		ocar		5.6-7	As per A	PI 6D		
15.0	Anti St	atic Testin	a Requirem	vent	· As no	Standard API 6D (Late	et Ed )		<u>1</u>	
10.0	Anti-Ot		ig requirem	ioni	. As per		551 20.7			
16.0	Valve I	Painting S	pecificatio	n t Dlaation		CA 0 1/0 CH				
16.1	Surface	e preparat	ion by Short	t Blasting as	s per grade	SA 2 1/2, Swedish Stan	idard SIS-055 909.	m thicknoss of 300 r	microp	
10.2	(Perm	issible thir	kness in ea	ich coat sha	l be within	80 to 120 micron). Colo	ur of paint shade shall h	e RAL-7038. howe	ver anv change in colour	
	shall b	e finalized	during drav	ving approv	al stage.			,		
17.0	Lock O	pen/ Lock	Close/Norn	nally Close	Requireme	nt : As indicated in F	2&ID.			
	Notes:	Vent & D	rain conne	ction for flo	ating & tru	innion mounted valves	shall be provided as I	per details mentior	ned in MECON's TS.	
	2	Inspection	and Testin	et snall be n in shall be n	aa in conji Is ner attac	Inction with MECON's T	ecinical Specification N	NUL MEC/TS/05/21/0	nt standards	
	3.	Stops sha	ll be provide	ed for positi	ve alignme	nt of ball with ports and	ensure proper installation	on of handle.	ni standards.	
	4.	Short patt	ern valves (	as per API	6D or other	wise) are not permitted.	Only long pattern valve	s are to be supplied	l.	
	5.	Charpy V	notch & Ha	rdness test	for body, b	ody adaptor, end flanges	s, ball, body seat rings, s	stem & studs / nuts	shall be conducted	
	e	as per Cl	. 3.4 & 3.6 c	of TS respe	ctively. Min	. Valve Body Wall Thi	ckness as per ASME E	316.34 .		
	7.	Material for	r body shall h	ave a quara	nteed minim	um vield strenath of	psi. In case the same ca	annot be guaranteed	valves shall	
	••	be provide	d with a 500 i	mm pup piec	e (integrally	welded to the valve on eac	h side) with strength equiv	alent to that of the co	nnecting pipe - N.A.	
	8.	For weldin	ng end, the	out of round	lness (i. e. o	difference between maxi	mum and minimum ID a	at pipe end) shall no	t be more than 0.5% of pipe OD.	
	9.	Valves sh	all be inspe	cted and ap	proved by	Purchaser before despa	tch.	1		
REV. NO.	DAIE	ZUNE			UND IS	I BY	AFPRU	REFERENCES	DRG NO	
SECTIO	<u>N PRC</u>	CESS &	PIPING		CLIENT :					
	NAME	DATE	01.11/12	DATE						
	NAME	DATE	CHKD	DATE				and	MEAANIIMITER	
DSGN DRWN	PM	25.04.12	AKJ	25.04.12	PROJECT	:				
				1				SCALE :	·	REV
APPROV	/ED			O.P. Jain	DATA	SHEET FOR BALL V	ALVES	DATA SHEET NO : ME	C/WINO/05/28/M/001/DS/BV/78	0
1						(NB ≥ 2")		1		

F

1.0	Valve I	Manufactu	rer		:					
2.0	Valve	Size (NB),	mm (inch)		:		ANSI Rating : 300#		Design Standard : API 6D	
3.0	MECO	N's Techni	ical Specific	cation No.	: MEC	/TS/05/21/002, Rev-1, Ed	d-1			
4.0	Conne	cting Pipel	line Design	Pressure, b	ar	: 49 kg/cm2	Design Temperature,	°C:-45°C to + 65	°C	
5.0	Conne	cting Pipe	e Specifica	tion	:					
5.1	Materia	al C.	•		:					
5.2	Diame	ter (OD), n	nm (inch)		:					
5.3	Inickn	ess, mm			:					
6.0	Valve	Construct	tion Design	1						
6.1.	Config	uration	•		: Reduce	ed Bore	Full Bore			
6.2.	End Co	onnections	;		: Flang	ged as per ASME B16.5			Butt Welded as per ASME B16.25	
6.3.	Flange	s (whereve	er applicabl	e)	: a) RF			RT	NA 📃	
					b) Seri	rated Smoo	th (125 to 200 microinches	AARH)	NA NA	
6.4	Ball Mo	ounting			: Floati	ng upto 4" and Trunnio	n Mounted 6" and abov	ve		
6.5	Valve	oody type			: Fully	Welded	Two/Three Piece Bolte	d 🔄	Either	
7.0	Valve	Material S	pecificatio	n						
	F	Part			S	pecified Material		Material Offe	ered (Equivalent or superior)	
7.1	Body		A 352 Gr. L	LCB/A 350	Gr. LF2					
7.2	Bal		SS316/ A 3	52 Gr.LCB	/A 350 Gr	LF2 with 75 µENP coa	ating			
7.3	Body S	eat Rings	GR I F2 w	ith 75 Micr	on ENP c	oating for Trunnion Mo	unted type			
7.4	Seat Se	eal	VITON/DEV	VLON for T	MBV	outing for trainformer	untou typo			
7.5	Stem		[(ASTM A35	50 Gr. LF2/S	S 316 )+75	microns ENP](No casting	g)			
7.6	Stem S	eals	VITON/PTF	E	0714 4404	10-1				
1.1		· ΔISI / 10	ASTM A32	<u>0 Gr.L7 / A</u> 35 HRC ba	STM A194	1 Gr.4		ļ		
8.0	Corros	ion Allowa	ince		: 1.5 m	ım	Service :			
9.0	Locatio	on			: Abov	e Ground	Buried			
10.0	Stem E	Extension F	Requiremen	ıt	: Yes		No			
11.0	Gear C	perator R	equirement		: Yes	$\checkmark$ for 6" and above	No _√_for 4" and b	elow		
12.0	Actuate	or Require	ment		: Yes		No			
13.0	Fire Re	esistant De	esign Requir	rement	: Type tes	st as per API 607 for Flo	pating Ball Valve			
14.0	Velve	Tooting P	oquiromon		: Type tes	st as per API 6FA for Tr	unnion Mounted Ball V	alve		
14.0	varve	resting R	equirement		Test Pres	sure (min.), kg/cm <sup>2</sup> (g)	Minimum Durati	ion, minutes		
14.1	Hydros	static Test		Body		76	As per A	PI 6D		
14 2	Air Tes	st		Seal		5.6-7	As per A As per A	PI 6D		
45.0	A	- #	Denim			an Otam dami ADLOD (Lat				
15.0	Anti-St	auc resun	ig Requirem	ieni	: As p	er Standard API 6D (Lat	est Ea.)			
16.0	Valve	Painting S	Specificatio	n Dia dia mandri		- 0.4.0.4/0. Our dials Ota-				
16.1	For ab	e preparat	non by Shon d installation	i blasting as	s per grau	e SA Z 1/Z, Swedish Star	he applied with minimur	m thickness of 300 r	micron	
10.2	(Perm	issible thic	kness in ea	ich coat sha	l be withi	n 80 to 120 micron). Colo	our of paint shade shall b	e RAL-7038, howe	ver any change in colour	
	shall b	e finalized	during draw	ving approv	al stage.	,		,	, ,	
17.0	Lock C	pen/ Lock	Close/Norn	nally Close	Requirem	ent : As indicated in	P&ID			
	Notes:	Vent & D	rain conne	Ction for fig	pating & t	runnion mounted valve	s shall be provided as	per details mentio	NECIN'S IS.	
	2	Inspection	and Testin	g shall be a	is per atta	ched QAP, this Data She	et, MECON's T.S., API 6	6D and other releva	nt standards.	
	3.	Stops sha	II be provide	ed for positi	ve alignm	ent of ball with ports and	ensure proper installation	on of handle.		
	4.	Short patt	ern valves (	as per API	6D or othe	erwise) are not permitted	. Only long pattern valve	s are to be supplied		
	5.	as per rel	-notch & Hai levant mate	runess test	ior body, l lin Valve	Body Wall Thickness	s, ball, body seat rings, :	siem & stuas / nuts	Shan de conqueted	
	6.	Compress	sed asbesto	s fibre (CAI	=) shall no	t be used for body sealin	g / gasket materials.			
	7.	Material for	r body sha <b>ll</b> h	ave a guara	nteed minin	num yield strength of	psi. In case the same ca	annot be guaranteed,	valves shall	
	Q	be provided	d with a 500 i	mm pup piec	e (integrally	y welded to the valve on each difference between man	ch side) with strength equiv	valent to that of the co	nnecting pipe - N.A.	
	υ.	i or weidir	iy enu, the t		11633 (1. 8.	amerence between Max	amum anu minimum ID a	at pipe end) shall no	t be more than 0.5% of pipe OD.	
REV. NO.	DATE	ZONE		DESCRIPTI	ONS	BY	APPRD			
SECTIO		CESS &	PIPING	REVISION	CLIENT ·			REFERENCES	DRG. NO.	
5201101		2200 0			~					
	NAME	DATE	CHKD	DATE						
DSGN	PM	25.04.12	AKJ	25.04.12	PROJEC	Т:		an 02 Cana	MECON LIMITED	
DRWN				I				20115		
						SHEET FOD DALL Y		SUALE :		KEV 0
AFFRUV	Ľυ			U.F. Jain	DATA	(NB > 2")	VALVEJ	DATA SHEET NO.: ME	C/WINO/05/28/M/001/DS/BV/79	U
						(10 )		1		L

10	Valvo	Manufactu	Irer							
0.0	Value		unon (in ala)						Desire Oferedand - ADLCD	
2.0	valve	Size (NB),	mm (incn)		:		ating : 300#		Design Standard : API 6D	
3.0	MECO	in's lechn	iical Specific	cation No.	: MEC/15/05/	52/003, Rev-2				
4.0	Conne	cting Pipe	line Design	Pressure, k	(g/cm²(g) : <b>4</b> 9	kg/cm2(g)	Design Te	mperature, °C :	-45°C to 65°C	
5.0	Conne	ecting Pip	e Specifica	tion						
5.1	Materi	a			:					
5.2	Diame	ter (OD), r	nm (inch)		:					
5.3	Thickn	ess, mm			:					
6.0	Valve	Construc	tion Design	ı						
6.1.	Patterr	٦			: Short		Regular		Venturi	
6.2.	End C	onnections	5		: Flanged bot	h ends		Flange	d as per ASME B 16.5	
					· Butt Weld b	oth ends		Butt We	d as per ASME B16 25	
					: Flanged on	e end, butt weld	other end			_
6.3.	Flange	es (wherev	er applicabl	e)	: a) RF	FF		RTJ	NA	
	-	-			b) Serrated	Smoo	h (125 to 200	microinches AARH		ĩ
							.11 (120 10 200			1
7.0	Vales	Martan I.C.								
7.0	valve	Material S	specificatio	n						-
		Part			Material			Material Offere	ed (Equivalent or Superior)	
7.1	Body		A 352 Gr. L	CB/A 350 Gr	. LF2					
7.2	Plug		SS316/ A 35	52 Gr.LCB/A	350 Gr. LF2 with	75 µENP coating				
73	Cover		ASTM A216	Gr WCB/A	234 Gr WPB					1
71	Stom		SS 316 (No	casting) / A	350 Gr 1 E2					1
7.5	Stom S	ool	DTEE/Crank	bito	000 01. 21 2					-
7.5	Sterri D		PTPE/Grapi		M A404 C= 4					-
7.0	Stud Bo	oits/ inuts	ASTNI A320	Gr.L7 / AST	M A194 Gr.4					1
8.0	Corros	ion Allowa	ance		: <b>1.5 mm</b>		Se	rvice :		
9.0	Locatio	on			: Above Grour	nd	Buried			
10.0	Stom	tension	Requirement	ht.		No				
11.0	Coord			it.	· Vaa	No				
11.0	Gear					110				
12.0	Gas P	owered Ac	tuator Requ	irement	: Yes	No				
13.0	Fire Re	esistant De	esign Requi	rement	: Type-Test a	s per Standard A	PI 6FA/ BS E	N: 10497		
14.0	Valve	Testing R	equiremen	t	Test Pressure (m	in ) $ka/cm^{2}(a)$	Minimum Di	ration minutes		
						(g)	Willing and De			
14.1	Hydros	static Test		Body	7	6	As pe	r API 6D		
				Seat	5	7	As pe	r API 6D		
14.2	Air Tes	st			5.6	-7	As pe	r API 6D		
15.0	Valve	Painting \$	Specificatio	on .				•		
15.1	Surfac	e preparat	ion by Shor	t Blasting a	s per grade SA 2	1/2, Swedish Sta	ndard SIS-05	5 909.		
15.2	For ab	ove groun	d installatio	n-Three coa	ats of corrosion re	sistant paint shall	be applied w	ith minimum thicl	kness of 300 micron	
	(Perm	issible thi	kness in ea	ach coat sha	all be within 80 to	120 micron). Col	our of paint sh	ade shall be RAI	L-7038, however anv	
	change	a in colour	shall he fin	alized durin	a drawing approv	al stade	our or paint of		_ / 000, 1010101 011,	
16.0	Lock	) nen/Lock	Close Reg	uirement	As indicated in					
10.0	LOOK		. Juse ney	anomont.	no maicaleu II					
	NI 1									
	inotes:	·	Dit at							
	1.	This Valv	e Data Shee	et shall be r	ead in conjunction	h with MECON's	echnical Spe	ecification No. ME	C/TS/05/62/003, Rev2	
	2.	Inspection	n and Testin	ng shall be a	as per attached Q	AP, this Data She	et, MECON's	s T.S., API 6D and	d other relevant standards.	
	3.	Stops sha	all be provid	ed for posit	ive alignment of p	lug with ports and	l ensure prop	er installation of I	handle.	
	4.	Charpy V	- notch & Ha	ardness tes	t for body, plug, c	over, stem & stud	s/nuts shall b	e conducted as p	oer Clause No.: 3.4 & 3.5	
		of TS resi	pectively. M	in. Valve Bo	dy Wall Thickness	as per ASME B16	6.34 .			
	5.	Material c	of the body s	shall have	equivale	nt to that of the c	onnectina pip	e Not Applicab	e	
	6.	Bidder sh	all clearly w	rite valves i	material (equivale	nt or superior) off	ered by them	against each par	t/material of valve in the	
		space pro	vided for M	Vherever hi	dder agrees with	alves material as	mentioned a	bove in MECON	s data sheet bidder	
		shall doo	dy indicate '							
		andli clea	ny mulcate	AGREED	•					
					0.10		10055			
REV. NO.	DATE	ZONE		DESCRIPTI	UNS	BY	APPRD			
				REVISION	1S			REFERENCES	DRG. NO.	
SECTIO	N OIL	& GAS		7	CLIENT :					
	NAME	DATE	CHKD	DATE				ing for		
	DM	25 04 40	A 12 1	25.04.40				and		)
JOGN	PM	25.04.12	AKJ	20.04.12	FRUJEUT :					
JRWN	<u> </u>			ļ						-
								SCALE :		REV
APPRO\	/ED			O. P. JAIN	DATA SHEE	TFOR PLUG V	ALVES	DATA SHEET NO.: I	MEC/WINO/05/28/M/001/DS/PV/79	0
					(NB	≥2")			-	
					(	/			D 01	7 . 5 .

		•	-		DATA				SCALE :		REV
										ł	
ORWN				I							
DSGN	PM	25.04.12	AKJ	25.04.12	PROJECT	Г:			BO BOOL CONPERT	MECON LIMITED	
	NAME	DATE	CHKD	DATE					र मेकॉन		
		a gas			GLIEINT :						
SECTIO		8 646		REVISION					REFERENCES	DRG. NO.	
REV. NO.	DATE	ZONE		DESCRIPTI	ONS	BY	1	APPRD			
			,		0.110						
		space pro	arly indicate	"AGREED"	uuer agree	es with valves mate	eriai as	mentioned a	DOVE IN MECON	i s uata sneet, didder	
	6.	Bidder sh	nall clearly w	vrite valves r	naterial (e	equivalent or superi	ior) offe	red by them	against each pa	irt/material of valve in the	
	5.	Material of	of the body s	shall have	e	equivalent to that of	f the co	nnecting pipe	e Not Applical	ble	
	т.	of TS res	pectively. M	lin. Valve Bo	dy Wall Th	nickness as per ASN	ME B16.	.34 .	S Somuloicu do	por oladoo nor. or a oro	
	3 4	Stops sha	ail be provid /- notch & He	ied for positi ardness test	ve alignm t for body	ent of plug with point of plug cover stem	rts and & stude	ensure prope	er installation of e conducted as	nangle. per Clause No : 3.4 & 3.5	
	2.	Inspection	n and Testir	ng shall be a	as per atta	ched QAP, this Da	ata Shee	et, MECON's	T.S., API 6D ar	nd other relevant standards.	
	1	This Valv	e Data She	et shall be r	ead in cor	junction with MEC	ON's T	echnical Spe	cification No. M	EC/TS/05/62/003, Rev2	
	Notes <sup>.</sup>										
16.0	Lock C	open/ Loci	k Close Req	uirement :	As indic	cated in P&ID.					
10.0	change	e in colour	r shall be fin	alized durin	g drawing	approval stage.					
	(Perm	issible thi	ckness in ea	ach coat sha	all be withi	in 80 to 120 micron	n). Colo	ur of paint sh	ade shall be RA	L-7038, however any	
15.2	For ab	ove groun	id installation	n-Three coa	ats of corro	osion resistant pair	nt shall	be applied wi	ith minimum thic	kness of 300 micron	
15.0 15.1	Surfac	e prepara	specification by Shore	nt Blasting a	s per arad	e SA 2 1/2. Swedis	sh Stan	dard SIS-055	5 909.		
15.0	Value	Dainting	Specification								
14.2	Air Tes	st				5.6 - 7		As pe	r API 6D		
			-	Seat		57		<u>As pe</u>	r API 6D		
14.1	Hvdros	static Test		Bodv		76		As pe	r API 6D		
					Test Pres	sure (min.), kg/cm2	2(g)	Minimum Du	ration, minutes		
14.0	Valve	Testing R	Requiremen	it ,	T C		0(-)	Minis =		т	
					.,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,	por e cuin					
13.0	Fire Re	esistant D	esign Requi	irement	: Tvpe	-Test as per Stand	dard AF	PI 6FA/ BS E	N: 10497		
12.0	Gas Po	owered Ac	ctuator Requ	uirement	: Yes		No				
			-1								
11.0	Gear (	Operator R	Requirement	t	: Yes		No				
10.0	Stem E	Extension	Requiremer	nt	: Yes		No				
	_	<u>.</u>	<u>.</u> .								
9.0	Locatio	on			: Abov	e Ground		Buried			
0.0	COHOS	NOT AIIOWA	ance					Ser			
8.0	NOTE : Corros	AISI 410 I sion Allows	nas mín. 35 l ance	HRC hardne	ss : 15 m	ım		Ser	vice :		
7.6	Stud Bo	olts/ Nuts	ASTM A193	Gr. B7/ A19	4 Gr. 2H						1
7.5	Stem S	ieal	PTFE/Grap	hite							ł
7.4	Stem		(AISI 4140 +	+ 75 microns	ENP Coat	ting)/ AISI 410 (No C	asting)				ļ
7.3	Cover		ASTM A216	Gr. WCB/ A	234 Gr. WI	РВ					Į
7.2	Plug		(ASTM A21	6 Gr. WCB/ /	234 Gr. W	PB) + 75 microns E	NP Coa	ting			1
7.1	Body		ASTM A216	Gr. WCB/ A	234 Gr. Wi	PB			material Offe		t
1.0	vaive	Part	specificatio	л	N.	laterial			Material Offe	ered (Equivalent or Superior)	T
70	Value	Matorial	Specificatio	<b>N</b> D							
					b) Serr	rated	Smooth	n (125 to 200 r	nicroinches AARI	H) 🗌 NA 🗌	
6.3.	Flange	es (wherev	/er applicabl	le)	: a) RF	FF			RTJ	NA	]
					: Flan	ged one end, butt	t weld o	other end			_
					: Butt	Weld both ends			Butt V	/eld as per ASME B16.25	
6.2	End Co	onnection	s		: Flang	ged both ends			 Flang	تحصصا ged as per ASME B 16.5	
6.1	Patterr	1			: Short			Regular		Venturi	
60	Valve	Construc	tion Design	า							
5.3	Thickn	iess, mm			:						
5.2	Diame	ter (OD), r	mm (inch)		:						
5.1	Materia	al	-		:						
5.0	Conne	ecting Pip	e Specifica	ation							
4.0	Conne	cting Pipe	eline Design	Pressure, k	.g/cm⁺(g)	: 49 kg/cm2(g)	)	Design Tei	mperature, °C :	-29°C to 65°C	
				_	, , ,						
3.0	MECO	N's Techr	nical Specifi	cation No.	: MEC/	/TS/05/62/003, Rev	v-2				
2.0	Valve	Size (NB),	, mm (inch)		:		ANSI	Rating : 300	¥	Design Standard : API 6D	
1.0	vaive	พลานเสบแ									
1.0	Valvo	Manufacti	Iror								

1.0	Valve	Manufactu	irer		:				
2.0	Valve	Size (NB),	mm (inch)		: ANS	Rating : 600#		Design Standard : API 6D	
3.0	MECC	N's Techn	nical Specific	ation No.	: MEC/TS/05/21/002, Rev-1, E	d-1			
4.0	Conne	ecting Pipe	line Design	Pressure, b	oar : 92 kg/cm2	Design Temperature,	, °C:- <b>29°C to +65</b>	°C	
5.0	Conne	ectina Pip	e Specifica	tion	:				
5.1	Materi	al			:				
5.2	Diame	ter (OD), r	nm (inch)		:				
5.3	Thickn	iess, mm			:				
6.0	Valve	Construct	tion Design	1					
6.1.	Config	uration			: Reduced Bore	Full Bore			
6.2.	End C	onnections	S		: Flanged as per ASME B16.5			Butt Welded as per ASME B16.25	
6.3.	Flange	es (wherev	er applicabl	e)	: a) RF FF		RT	NA	
					b) Serrated Smoo	th (125 to 200 microinches	s AARH)		
6.4	Ball M	ountina			Trunnion Mounted		,		
6.5	Valve	body type			: Fully Welded	Two/Three Piece Bolte	d 🗌	Either	
7.0	Valve	Material S	Specificatio	n	Specified Meterial		Motorial Offe	red (Equivalent er eurorier)	
71	Body	ran	A 216 Gr W	ICB/A 234 G	Specified Material	:0	Waterial Offe	ared (Equivalent of superior)	
72	Ball		(A 216 Gr W	ICB/A 234 C	r WPB/ A 352 Gr I CB/AISI 4140)+7	2 5 uENP coating/AISI410	1		
7.3	Body S	eat Rings	AISI 4140 +	75 micron E	ENP/AISI 410				
7.4	Seat S	eal	VITON/DEV	LON					
7.5	Stem		AISI 4140 +	75 micron B	ENP coating/AISI 410 (No casting)				
7.6	Stem S	eals	VITON/PTF	E					
7.7	Stud B	olts/ Nuts	ASTM A 193	3 Gr. B7/ A1	94 Gr. 2H				
0	Corror		nas min. 3	S HRC ha	i 15 mm	Sonvice :			
0.0	Conos	SION Allowa	ince		. 1.5 mm	Service.			
9.0	Locatio	on			: Above Ground	Buried			
10.0	Stem I	Extension	Requiremer	nt	: Yes	No			
11.0	Gear (	Operator R	Requirement		: Yes $$ for 6" and above	e No √ for 4" a	and below		
12.0	Actuat	or Require	ement		: Yes	No			
13.0	Fire R	esistant De	esign Requii	rement	: Type test as per API 6FA				
14.0	Valve	Testing R	equiremen	t					
					Test Pressure (min.), kg/cm <sup>2</sup> (g)	Minimum Durat	ion, minutes		
14 1	Hydros	static Test		Body	157	As per A	PI 6D		
	i iyuro.	51010 1001		Seat	114	As per A	PI 6D		
14.2	Air Tes	st			5.6-7	As per A	PI 6D		
15.0	Anti-St	tatic Testin	a Requirem	ent	· As per Standard API 6D /I a	test Ed )			
16.0	Value	Deinting	Presificatio						
16.0	Surfac	e prenarat	tion by Shor	t Blasting a	s per grade SA 2 1/2 Swedich Sta	indard SIS-055 909			
16.2	For ab	ove aroun	id installation	n-Three coa	ats of corrosion resistant paint shall	be applied with minimu	m thickness of 300 r	micron	
	(Perm	nissible thic	ckness in ea	ch coat sha	all be within 80 to 120 micron). Col	our of paint shade shall b	be RAL-7038, howe	ver any change in colour	
	shall b	e finalized	during draw	ving approv	al stage.				
17.0	Lock C	Open/ Lock	Close/Norr	nally Close	Requirement : As indicated in	P&ID.			
	Notes:	Vent & Di	rain connec	tion for flo	ating & trunnion mounted valves	shall be provided as pe	er details mentione	d in MECON's TS.	
	1. '5'	Ins Valve	e Data Shee	et snall be r	eau in conjunction with MECON's	ecnnical Specification N	NO. MEC/15/05/21/0	JUZ, KEV 1, EQ. 1	
	2.	Stons sha	all he provid	ed for nosit	ive alignment of ball with ports and	ensure proper installatio	on of handle	ant standards.	
	4	Short pat	tern valves (	as ner API	6D or otherwise) are not permitted	Only long pattern valve	es are to be supplied	4	
	5.	Charpy V	-notch & Ha	rdness test	for body, body adaptor, end flang	es, ball, body seat rings,	stem & studs / nuts	shall be conducted	
		as per Cl	l. 3.4 & 3.6 c	of TS respe	ctively Min. Valve Body Wall Thi	kness as per ASME B	16.34 .		
	6.	Compres	sed asbesto	os fibre (CA	F) shall not be used for body seali	ng / gasket materials.			
	7.	Material fo	r body shall I	nave a guara	anteed minimum yield strength of	psi. In case the same of	cannot be guaranteed	, valves shall	
	0	be provide	d with a 500	mm pup pied	ce (integrally welded to the valve on e	ach side) with strength equ	uvalent to that of the o	connecting pipe - N.A.	
	б. 0	Valves ch	ng end, the i	out or round	uness (i. e. unerence between ma	atch	at pipe end) shall no	or be more than 0.5% of pipe OD.	
REV. NO.	DATE	ZONE		DESCRIPTI	ONS BY	APPRD			
		·		REVISION	IS	•	REFERENCES	DRG. NO.	
SECTIO	N PRO	DCESS &	PIPING		CLIENT :				
	NAME	DATE	СПКР				Aparter -		
	INAIVIE	DAIE		DATE			B BOOT CONSIS	MECONLIMITED	
DSGN	PM	25.04.12	AKJ	25.04.12	PROJECT :				
URWN	L	1	1	1			SCALE :	l	RE//
	/FD			0 P Jain	DATA SHEET FOR BALL	ALVES		C/WINO/05/28/M/001/DS/R\//80	0
					(NB≥ 2")				v

F

1.0	Valve	Manufactu	irer		:				
2.0	Valve	Size (NB),	mm (inch)		: ANSI F	Rating : 600#		Design Standard : API 6D	
3.0	MECC	N's Techn	ical Specific	cation No.	: MEC/TS/05/21/002, Rev-1, E	d-1			
4.0	Conne	ecting Pipe	line Design	Pressure, t	bar : 92 kg/cm2	Design Temperature	, °C ∶ <b>-45°C to + 6</b> 5	°C	
5.0 5.1 5.2 5.3	<b>Conne</b> Materi Diame Thickr	ecting Pip al eter (OD), r ness, mm	<b>e Specifica</b> nm (inch)	ıtion	:				
6.0 6.1. 6.2.	Valve Config End C	Construct	tion Desigr	1	: Reduced Bore	Full Bore		Butt Welded as per ASME B16.25	
6.4 6.5	Ball M Valve	ounting body type	er applicabl	e)	b) Serrated Smoo : Trunnion Mounted : Fully Welded	th (125 to 200 microinches Two/Three Piece Bolte	s AARH)	Either	
7.0	Valve	Material S	pecificatio	'n	Specified Material		Matorial Off	ared (Equivalent or superior)	1
7.1 7.2 7.3 7.4 7.5 7.6 7.7	Body Ball Body S Seat S Stem Stem S Stem S	Geat Rings eal Geals olts/ Nuts	A 352 Gr. I SS316/ A 3 SS316/ AS VITON/DE [(ASTM A35 VITON/PT ASTM A32	LCB/A 350 352 Gr.LCB TM A352 G VLON 50 Gr. LF2/S FE 20 Gr.L7 / A	Gr. LF2 3/A 350 Gr. LF2 with 75 µENP co Gr. LCB / A 350 GR. LF2 with 75 I 38 316 )+75 microns ENP](No castin ASTM A194 Gr.4	ating /licron ENP coating g)			
NQTE	:ભાષ્ટ્રિક	4i3A A <b>as</b> wa	ince35 HRC	C hardness	: 1.5 mm	Service :			
9.0	Locati	on			: Above Ground	Buried			
10.0	Stem	Extension	Requiremer	nt	: Yes	No			
11.0	Gear (	Operator R	equirement	t	: Yes for 6" and above	e No <u>√</u> fþr4" a	ind below		
12.0	Actuat	tor Require	ement		: Yes	No			
13.0	Fire R	esistant De	əsign Requi	irement	: Type test as per API 6FA				
14.0	Valve	Testing R	equiremen	ıt		1		1	
14.1	Lludro	otatia Taat		Dodu	1est Pressure (min.), kg/cm (g)		ion, minutes		
14.1				Seat	157	As per A As per A	API 6D		
14.2	Anti R	si				As per A	PI 6D		
16.0 16.1 16.2 17.0	Valve Surfac For ab ( Perm shall b Lock (	Painting S ce preparat pove groun hissible thic pe finalized Open/ Lock	Specificatic tion by Shor d installation kness in ea during drav < Close/Nori	אר t Blasting a n-Three כסג ach coat sha wing approv mally Close	as per grade SA 2 1/2, Swedish Sta ats of corrosion resistant paint shal all be within 80 to 120 micron). Col /al stage. Requirement : <b>As indicated in</b>	Indard SIS-055 909. I be applied with minimu our of paint shade shall I P&ID./ Material Requis	m thickness of 300 be RAL-7038, howe i <b>tion</b>	micron ver any change in colour	
	Notes: 1 2 3 4 5	Vent & E This Valve Inspection Stops sha Short patt Charpy V as per re	Drain conne e Data Shee h and Testin all be provid tern valves ( -notch & Ha levant mate	etion for flo et shall be r g shall be a ed for posit (as per API ardness test erial code.	oating & trunnion mounted valves read in conjunction with MECON's as per attached QAP, this Data Sh tive alignment of ball with ports and 6 D or otherwise) are not permitted t for body, body adaptor, end flang Min. Valve Body Wall Thickness	s shall be provided as p Technical Specification N eet, MECON's T.S., API ensure proper installatid 1. Only long pattern valve es, ball, body seat rings, as per ASME B16.34	per details mention No. MEC/TS/05/21/0 6D and other releva on of handle. es are to be supplied , stem & studs / nuts	ed in MECON's TS. )02,Rev 1 ,Ed. 1 ant standards. d. s shall be conducted	
	6. 7. 8. 9	Compress Material fo be provide For weldin Valves sh	sed asbesto r body shall I d with a 500 ng end, the nall be inspe	os fibre (CA have a guara mm pup pier out of round acted and a	kF) shall not be used for body sealin anteed minimum yield strength of cc (integrally welded to the valve on e dness (i. e. difference between ma poroved by Purchaser before deen	ng / gasket materials. psi. In case the same ach side) with strength equ ximum and minimum ID atch.	cannot be guaranteed uivalent to that of the at pipe end) shall n	l, valves sha <b>ll</b> connecting pipe <b>- N.A.</b> ot be more than 0.5% of pipe OD.	
REV. NO.	DATE	ZONE		DESCRIPTI	IONS BY	APPRD	REEDENCES		
SECTIO	N PRO	DCESS &	PIPING					DRG. NO.	
	NAME	DATE	СНКД	DATE					
DSGN DRWN	PM	25.04.12	AKJ	25.04.12	PROJECT :		-001 - 60**	MECON LIMITED	
APPRO\	/ED		<u> </u>	O.P. Jain	DATA SHEET FOR BALL \ (NB≥ 2")	/ALVES	SCALE : DATA SHEET NO.: ME	C/WINO/05/28/M/001/DS/BV/81	REV 0

IF.

1.0	Valve	Manufactu	urer		:							
2.0	Valve	Size (NB),	, mm (inch)		:	AM	NSI Rat	ing : 600#		Design Standard	: API 6D	
3.0	MECO	N's Techr	nical Specifi	cation No.	: MEC	/TS/05/62/003, Rev	v-2					
4.0	Conne	cting Pipe	line Design	Pressure, k	(g/cm <sup>2</sup> (g)	: 92 kg/cm2(g)	)	Design Te	mperature, °C :	-29°C to 65°C		
5.0	Conne	ectina Pip	e Specifica	ation								
5 1	Matori	al										
5.1	Inateria				•							
5.2	Diame	ter (OD), i	mm (inch)		:							
5.3	Thickn	ess, mm			:							
6.0	Valve	Construc	tion Desigr	า	Ohard			Deviler		Marturi 🗖		
6.1.	Patterr	ר			: Short			Regular		venturi		
6.2.	End C	onnection	s		: Flang	ged both ends			Flan	ged as per ASME B 16.	5	
					: Butt	Weld both ends			Butt V	veld as per ASME B16.	25	
					· Flan	and one and butt	wold	ther end				
					. 11an							1
6.3.	Flange	es (wherev	er applicab	le)	:a) R⊦				RIJ			J
					b) Ser	rated	Smooth	n (125 to 200 i	microinches AARI	H)	NA 🔄	
7.0	Valve	Material S	Specificatio	on								
		Part	ľ		Ν	/laterial			Material Offe	ered (Equivalent or	Superior)	1
7.1	Body		ASTM A216	Gr. WCB/ A	234 Gr. W	PB				\ I	. ,	
72	Plug		(ASTM A21	6 Gr WCB/	1234 Gr W	/PB) + 75 microns F	NP Coa	tina				
73	Covor		ASTM A216		224 Gr. W			ang				
7.0	Cover				END Cool		· a a time ·					
7.4	Stem		(AISI 4140 ·	+ / 5 microns	ENP COa	ung)/ AISI 410 (NO C	asung)					
7.5	Stem S	ear	PTFE/Grap									
7.6	Stud Bo	olts/ Nuts	ASTM A193	3 Gr. B7/ A19	4 Gr. 2H							l
	NOTE :	AISI 410 I	has min. 35	HRC hardne	SS							
8.0	Corros	ion Allowa	ance		: 1.5 m	าท		Sei	rvice :			
9.0	Locatio	on			: Abov	e Ground		Buried				
10.0	Stem I	İxtension	Requireme	nt	: Yes		No					
11.0	Gear C	Operator R	Requirement	t	: Yes		No					
12.0	Gas P	owered Ac	ctuator Requ	uirement	: Yes		No					
13.0	Fire Re	esistant D	esign Requ	irement	: Type	-Test as per Stand	dard Al	PI 6FA/ BS E	N: 10497			
14.0	Valve	Testing R	lequiremen	it	Test Pres	sure (min.), ka/cm2	2(a)	Minimum Du	uration, minutes	ſ		
					10011100		-(9)					
14.1	Hydros	static Test		Body		157		As pe	r API 6D			
				Seat		114		As pe	r API 6D			
14.2	Air Tes	st				5.6 - 7		As pe	r API 6D			
15.0	Value	Dointing	Creatienti									
15.0	valve	Painting	Specificatio	on t Disstance					- 000			
15.1	Surfac	e prepara	tion by Shoi	rt Blasting a	s per grad	ie SA 2 1/2, Swedis	sn Stan	dard SIS-05	5 909.			
15.2	For ab	ove groun	id installatio	n-Three coa	ats of corre	osion resistant pair	nt shall	be applied w	ith minimum thic	ckness of 300 micro	on	
	(Perm	issible thi	ckness in ea	ach coat sha	all be with	in 80 to 120 micron	ı). Colo	ur of paint sh	nade shall be RA	L-7038, however a	iny	
	change	e in colour	ˈshall be fin	alized durin	g drawing	approval stage.						
16.0	Lock C	)pen/ Locł	< Close Rec	quirement :	As indic	cated in P&ID.						
	Notes:											
	1	This Valv	e Data She	et shall be r	ead in cor	njunction with MEC	ON's T	echnical Spe	cification No. M	EC/TS/05/62/003, I	Rev2	
	2.	Inspectio	n and Testir	na shall be a	as per atta	ched QAP. this Da	ta She	et. MECON's	T.S., API 6D ar	nd other relevant st	andards.	
	3.	Stops sha	all be provid	led for posit	ive alignm	ent of plug with po	rts and	ensure prop	er installation of	handle.		
	4	Charny V	- notch & H	ardness tes	t for hody	nlug cover stem	& stude	/nuts shall h	e conducted as	ner Clause No : 34	1 & 3 5	
		of TS roc	notivoly M	in Volvo Po				21	e conducted do		10.0.0	
	F	Meterial			uy wan m	lickness as per Aoiv		04. maastina nin		- I		
	о. с	Riddor of		rito volver	motorial (-	equivalent to that of		mecung pip	e NOL APPIICAL	ne ut/matarial cfuelur	in the	
	υ.	Bidder Sh	an cleany w		nateriai (6 44.5 -	equivalent or super		neu by triem	against each pa			
		space pro	ovided for V	vnerever bi	uder agree	es with valves mate	enai as	mentioned a	bove in MECON	i s data sheet, bidd	er	
		shall clea	irly indicate	"AGREED"	•							
	1 -		1			1						
REV. NO.	DATE	ZONE		DESCRIPT	ONS	BY		APPRD	1			
				REVISION	NS				REFERENCES	r	DRG. NO.	
SECTIO	N OIL	& GAS			CLIENT :							
	NAME	DATE	CHKD	DATE					मेकॉन			
DSGN	PM	25.04.12	AKJ	25.04.12	PROJEC <sup>-</sup>	Т:			Po Scol Conpan	MECON	LIMITED	
DRWN	1											
	+		l	•					SCALE	<u> </u>		REV
	(55			о в	D				SUALE .			
APPRO	VED			O. P. JAIN	DATA	SHEETFOR PL	UG V	ALVES	DATA SHEET NO .:	MEC/WINO/05/28/M/00	01/DS/PV/80	0
						(NB ≥ 2")				-	101	

1.0	Valve	Manufactu	Irer		:					
2.0	Valve	Size (NB),	mm (inch)		:	ANSI F	Rating : 600#	E	Design Standard :API 6D	
3.0	MECC	N's Techr	ical Specific	cation No.	: MEC/TS/05	62/003, Rev-2				
4.0	Conne	ecting Pipe	line Design	Pressure, k	.g/cm²(g) : <b>9</b>	2 kg/cm2(g)	Design Te	emperature, °C :	-45°C to 65°C	
5.0	Conne	ectina Pip	e Specifica	tion						
5.1	Materi	al	e epooniou		:					
5.2	Diame	ter (OD), r	nm (inch)		:					
5.3	Thickr	ness, mm			:					
6.0	Valve	Construct	tion Design	n	_					
6.1.	Patter	n			: Short		Regular		Venturi	
6.2.	End C	onnections	5		: Flanged bo	th ends		Flange	d as per ASME B 16.5	
					: Butt Weld b	oth ends		Butt We	ld as per ASME B16.25	
					: Flanged or	ne end, butt weld	other end			
6.3.	Flange	es (wherev	er applicabl	e)	:a) RF 🗌	FF	]	RTJ	NA	]
					b) Serrated	Smoot	 h (125 to 200	microinches AARH	) NA 🗌	]
										-
7.0	Valve	Material S	Specificatio	n						-
7.4		Part			Materia			Material Offer	ed (Equivalent or Superior)	-
7.1	Body		A 352 Gr. LO	CB/A 350 Gr	250 Cr. 1 52 with					-
73	Plug		33310/ A 33	Gr WCB/A	234 Gr. WPB	75 µENP coating				-
74	Stem		SS 316 (No	casting) / A	350 Gr   F2					-
7.5	Stem S	Seal	PTFE/Graph	hite						
7.6	Stud B	olts/ Nuts	ASTM A320	Gr.L7 / AST	M A194 Gr.4					
8.0	Corros	sion Allowa	ance		: 1.5 mm		Se	rvice :		
9.0	Locati	on			· Above Grou	nd	Buried			
10.0	Stem I	Éxtension	Requiremen	ht	· Yes			1		
11.0	Gear (	Operator P	equirement	it.	· Vec	] No				
11.0	Ocar C		equirement		. 103					
12.0	Gas P	owered Ac	tuator Requ	uirement	: Yes	] No				
13.0	Fire R	esistant De	esign Requi	rement	: Type-Test a	is per Standard A	PI 6FA/ BS E	EN: 10497		
14.0	Valve	Testing R	equiremen	t	Test Pressure (r	nin.), kg/cm2(g)	Minimum D	uration, minutes		
								-		
14.1	Hydros	static Test		Body	1	57	As pe	er API 6D		
14.0	Ain Ta	-4		Seat	1	14	As pe	er API 6D		
14.2	AIFTE	st			5.0	<b>b</b> - 7	As pe	er API 6D		
15.0	Valve	Painting \$	Specificatio	on						
15.1	Surfac	e preparat	ion by Shor	t Blasting a	s per grade SA 2	1/2, Swedish Star	ndard SIS-05	5 909.		
15.2	For ab	ove groun	d installatio	n-Three coa	ats of corrosion re	esistant paint shall	be applied w	ith minimum thic	kness of 300 micron	
	(Perm	nissible thio	ckness in ea	ach coat sha	all be within 80 to	120 micron). Colo	our of paint sl	hade shall be RA	L-7038, however any	
16 0	chang	e in colour	Shall be fin	alized durin	g drawing appro	/al stage.				
10.0	LOCK		Close Req	ullement.	As mulcaleu i	I FaiD.				
	Notes:									
	1	This Valv	e Data Shee	et shall be r	ead in conjunctio	n with MECON's 1	echnical Spe	ecification No. ME	EC/TS/05/62/003, Rev2	
	2.	Inspection	n and Testin	ng shall be a	as per attached (	QAP, this Data She	et, MECON's	s T.S., API 6D an	d other relevant standards.	
	3.	Stops sha	all be provid	ed for posit	ive alignment of	olug with ports and	l ensure prop	er installation of l	handle.	
	4.	Charpy V	notch & Ha	ardness tes	t for body, plug, o	cover, stem & stud	s/nuts shall b	pe conducted as p	per relevant material code	
	F	of TS res	pectively. Mi	in. Valve Bo	dy Wall Thicknes	s as per ASME B16	.34 .	A Net America	la.	
	э. 6	Riddor ch	all clearly w	rite values :	material (equival-	ent or superior) off	ered by them	e NOT APPIICAD	t/material of valve in the	
	0.	space nr	ovided for. V	Vherever hi	dder aarees with	valves material as	mentioned a	above in MECON	s data sheet. bidder	
		shall clea	rly indicate '	"AGREED"						
		-								
REV. NO.	DATE	ZONE		DESCRIPTI	ONS	BY	APPRD			
		8 645		REVISION				REFERENCES	DRG. NO.	
		a GAS			GLIENT .					
	NAME	DATE	CHKD	DATE						
DSGN	PM	25.04.12	AKJ	25.04.12	PROJECT :			do abor Convint	MECON LIMITED	)
DRWN										
								SCALE :		REV
\PPRO\	/ED			O. P. JAIN	DATA SHEE	TFOR PLUG V	ALVES	DATA SHEET NO .:	MEC/WINO/05/28/M/001/DS/PV/81	0
					(NE	5 ≥ 2")				ـ ۲

r										
1.0	Valve M	anufacturer		:						
2.0	Value Ci		(inch)			ANCI Dating 1 90	0.4	Design Chandard	- DC 5254/DC EN ICO 47202	
2.0		ze (ND), mm	(Inch)			ANSI Kating . ou	U#	Design Standard	: B3 5351/B5 EN 150 1/292	
3.0	MECON	s recnnical	Specification No.	:						
4.0	Connect	ing Pipeline [	Design Pressure,	t:		Design Temp	erature, °C :	-29°C to +65°C		
5.0	Connec	ting Pipe Sp	ecification	:						
5.2	Material	r (OD), mm		:						
5.3	Wa <b>ll</b> Thi	ckness, mm		:						
6.0	Valve C	onstruction	Design			_				
6.1.	Configur	ation		: Reduced	Bore Welded as neu		Full Bore	$\checkmark$		
0.2.		nections		100mm	Extension Pup	s in ASTM A106	Gr.B (Sch.	160) at both ends		
6.3.1	Flanges	(wherever ap	oplicable)	: RF 🗌		FF 🗌		RTJ	NA	
6.3.2	Flange F	ace Finish		Serrated	Smo	ooth (125 to 200 mi	croinches AAR	RH)	NA	
0.4		inung		: Floating						
7.0	Valve M	aterial Spec Part	ification	ş	Specified Mater	ial		Material Offer	ed (Equivalent or superior)	
7.1	Body		ASTM A105							
7.2	Ball Bodv Sea	t	13% Cr Steel RPTFE/ DELRIN					+		
7.4	Gland		13% Cr Steel							
7.5	Stem Body Sea	1	13% Cr Steel (No Grafoil	Casting)				+		
7.7	Stem Sea	1	Grafoi							
7.8	Body Stu	ds/Nuts	ASTM A193 Gr. B	87/ A194 Gr. 2	Н			1		
8.0	Corrosio	n A <b>ll</b> owance			: 1 <b>.</b> 5 mm			Service :		
9.0	Location				: Above Gro	und	Buried			
10.0	Stem Ex	tension Requ	uirement		: Yes	No				
11.0	1.0 Gear Operator Requirement : Yes No									
12.0	Gas Ope	erated Actuat	tor Requirement		: Yes			40.407		
13.0	rile Res	istant Design	rkequirement		. Type-Test	as per API 607	63 EN 150	10497		
14.0	Valve Te	esting Requi	irement		Test Pres	sure (min.)			]	
					(kg/c	:m <sup>2</sup> (g))	Minimum D	Ouration (minutes)		
14.1	Hydrosta	atic Test		Body Seat		210  55		2		
14.2	Air Test				5.	6 - 7		15	]	
15.0	Anti-Stat	ic Testing Re	equirement		: As per BS	EN ISO 17292				
16.0	Valve Pa	ainting Spec	ification	00 000 0 ·····	SA 2 4/2 C	ich Oterate d 010	055 000			
16.1	Surface For abov	preparation to ve ground ins	by Short Blasting	as per grade bats of corros	SA 2 1/2, Swed sion resistant pai	ish Standard SIS nt shall be applie	-055 909. I with minimu	um thickness of 300	) micron	
ľ	(Permis	sible thickne	ss in each coat sl	hall be within	80 to 120 micro	n). Colour of pain	t shade sha <b>l</b>	be RAL-7038, how	ever any change in colour	
17.0	Lock Op	en/ Lock Clo	se/Normally Clos	e Requireme	nt : As indica	ted in P&ID.				
	Notes:									
	1	Charpy V-no	otch test for body,	, ball, body se	eat, gland, stem	& studs/nuts sha	be conducte	ed as per A370. The	e test shall be conducted at 0°C	
	2.	Material test	t certificates and	hydrostatic te	est reports shall	be furnished prior	to despatch.			
	3.	Detailed dim prior to man	nensional drawing	ls showing cr alves. Min \	oss-section with /alve Bodv W/a	part numbers an Thickness as	d materials s per ISO 172	nall be submitted fo	or Purchaser's approval	
	4.	All tests sha	II be as per BS E	N 12266	are body wa		55, 100, 172			
ľ	5. 6.	Valves shall Stops shall I	have ball position be provided for no	n indicator. ositive alionm	nent of ball with r	orts and ensure	proper install	ation of handle.		
	7.	Each valve	shall be provided	with a wrenc	h.					
	8. 9.	valves shall Gland packi	be inspected and ng assembly shal	a approved b II permit repa	y Purcnaser bef ir of gland packi	ore despatch. ng under full line	pressure.			
	10.	Inspection a	nd Testing shall b	be as per atta	ached QAP, this	datasheet, BS E	N 12266, othe	er relevant standard	ds&clause no. 5.0 of	
		IS NO.: ME	Gr 1 3/00/2 1/002,F	.ev. i,⊑u. l.						
REV. N	DATE	ZONE		DESCRIPTIC	INS	BY	APPRD	REFERENCES	DRG. NO.	
SECT	ION		1	1	CLIENT :					
	NAME	DATE	CHKD	DATE	ļ					
DSGN DRWN	PM	25.04.12	AKJ	25.04.12	PROJECT :			THE BUSY CLEVEN	MECON LIMITED	) _
		<u> </u>	ł		DATA OV-			SCALE :	I	REV
APPR	OVED			O.P. JAIN	DATA SHE	EIFOR BALL	VALVES	DATA SHEET NO.: M	EC/WINO/05/28/M/001/DS/BV/82	0
<u> </u>					1			L		1

1.0	Valve	Manufactu	ırer		:						
2.0	Valve	Size (NB),	mm (inch)		:	ANS	I Rating : 80	00#		Design Standard : BS EN ISO 1	17292
3.0	MECO	N's Techn	ical Specific	cation No.	:						
4.0	Conne	cting Pipe	line Design	Pressure, b	ar	: 92 kg/cm	12	Design Temp	oerature, °C : <b>-45</b>	to 65°C	
5.0	Conne	cting Pip	e Specifica	tion		N.A					
5.1 5.2	Diame	al ter (OD), r	nm (inch)								
5.3	Thickn	ess, mm									
6.0	Valve	Construc	tion Design	ı	- Dod						
6.1. 6.2.	End C	onnections	6		: Real	et welded as	per ASME E	Fuii <u>√</u> 316.11			
					100m	nm Extension	Pups in AS	TM A333 Gr.6 S	Sch. 160		
6.3.1	Flange	s (wherev	er applicabl	e)	: a) RF		FF	1	RTJ	NA	
6.3.2	Flange	Face Fini	ish	,	Serrate		Smooth (125	to 200 microinche	es AARH)		
6.4 7.0	Valve	ounting Material S	specificatio	n	: Floa	ting Ball					
71	Body	Part	ASTM A350	GR LE2	Spe	cified Material				Material Offered	
7.2	Ball		SS 304/316	with 75µEN	P Coating						1
7.3	Body S	eat	RPTFE/ DEL								]
7.4 7.5	Gland		SS304/SS31	6 + 75 uENP/	No castin	a)					+
7.6	Body S	eal	Grafoil	- TO MEIN	to ouoting	9/					1
7.7	Stem S	ea	Grafoil								]
7.8	Body S	tuds/Nuts	ASTM A320	Gr. 7/ A194	Gr.4						1
8.0	Corros	ion Allowa	ance		: 1.5 n	nm		Service :			
9.0	Locatio	on			: Abov	ve Ground		Buried			
10.0	Stem E	Extension	Requiremen	nt	: Yes		No				
11.0	Gear (	Operator R	equirement		: Yes		No				
12.0	Gas P	owered Ac	tuator Requ	uirement	: Yes		No				
13.0	Fire R	esistant De	əsign Requii	rement	: Туре	e-Test as per \$	Standard Al	PI 607			
14.0	Valve	Testing R	equirement	t	Test Pres	ssure (min.), k	g/cm2(g)	Minimum Dur	ration, minutes	Ţ	
14.1	Hydros	static Test		Body Seat		210 155			2 2		
14.2	Air Tes	st				5.6-7.0		1	15	I	
15.0	Anti-St	atic Testir	ng Requirem	ient	: Asp	er Standad B	S EN ISO 17	7292			
16.0	Valve	Painting S	Specificatio	n t Blasting o	e ner arca	10 SA 2 1/2 S	Nadich Star	dard SIS OFF OF	סו		
ii)	For ab	ove groun	d installatior	n-Three coa	its of corr	osion resistant	t paint shall	be applied with r	minimum thickne	ess of 300 micron	
	(Perm	issible thic	ckness in ea	ich coat sha	II be with	in 80 to 120 m	icron). Colo	ur of paint shade	e shall be RAL-7	038, however any change in colo	bur
17.0	Lock C	ben/ Lock	Close Reg	uirement	aistage. : Aisin	dicated in P&	ID				
	Notes:										
1.	Materi	al test cert	ificates and	hydrostatic	test repo	rts shall be fur	nished prior	to despatch.	II In a secola section of <i>C</i> .		
Ζ.	approv	al prior to	manufactur	gs showing e of the val	∠ross-se ∠es Min. \	Valve Body Wa	all Thickness	as per ISO 172	92 .	of Purchasers	
3.	All test	s shall be	as per BS:6	3755 (Part-I)		,		·			
4.	Valves	shall hav	e ball positio	on indicator.					-4-11-42 51	.U -	
5. 6	Stops	snall be pr	be provided	nsure positiv d with a wre	ve alignm	ient of ball with	i ports and e	insure proper in:	stallation of hand	die.	
7.	Valves	shall be i	nspected ar	id approved	by Purch	naser before d	espatch.				
8.	Gland	packing a	ssembly sha	all permit re	pair of gla	and packing un	der full line	pressure.			
9.	Inspec	tion and T	esting shall	be as per th	nis specifi 1	ication, BS:675	55 (Part-I) ar	nd other relevan	t standards and	clause no. 5.0 of	
10.	Charp	/ V-notch	& Hardness	test for bod	y, body a	daptor, end fla	inges, ball, b	oody seat rings,	stem & studs / n	uts shall be conducted	
REV. NO.	DATE	ZONE		DESCRIPTI	ONS	E	3Y	APPRD			
				REVISION	IS	-			REFERENCES	DRG. NC	D.
SECTION	N PRC	DATE			CLIENT	:					
	DM	25 04 40		25.04.42		т·			मेलांग		
DRWN	PW	20.04.12	Anj	23.04.12	I NOJEC				-094 - 0764		
APPROV	/ED		L	O. P. JAIN	DAT	A SHEET FO		/ALVES	DATA SHEET NO.:	MEC/WINO/05/28/M/001/DS/BV/83	REV 0
						(NB	s<2")				

1.0	Valve l	Manufactu	rer		:					
2.0	Valve	Size (NB),	mm (inch)		:	ANSI Rating : 800#			Design Standard :BS:5353	
3.0	MECO	N's Techni	cal Specific	ation No.	:					
4.0	Conne	cting Pipel	ine Design I	Pressure, k	g/cm <sup>2</sup> (g)	:	De	sign Temp	perature, °C : -29°C to 65°C	
5.0	Conne	cting Pipe	e Specifica	tion	: N.A.					
5.1 5.2	Materia	al ter (OD) ~	m (inch)		:					
5.3	Thickn	ess, mm			:					
6.0	Valve	Construct	ion Design	I						
6.1.	Patterr	1			: Short	Re	gular [		Venturi	
6.2.	End Co	onnections			: Flanged	I both ends	Flang	ed as per AS	ME B 16.5	
					: Socket	Weld both ends	Socket	Welded 3000# (	(as per ASME B16.11)	
					Socket wel	ded with Pup Piece of 100mm len	igth of A333	Gr.6 Sch. 1	60	
6.3.	Flange	s (whereve	er applicable	e)	: a) RF b) Serrate	ed Smooth (125 to	RTJ 200 microin	ches AARH	) NA .	
7.0	Valve	Material S	pecificatio	n	Sn	cified Material			Material Offered	1
7.1	Body	art		ASTM A 10	5 5				Material Offered	-
7.2	Plug (L	ubricated)		ASTM A 10	5 + 75 µENP (	Coating				
7.3	Stem (N	lo Casting)		(AISI 4140	+ 75 µENP Co	ating) /AISI 410 (No Casting)				
7.5	Gland F	Packing		Graphite/ F	TFE					
7.6	Gasket	uoning		N.A.						1
7.7	Body S	tuds/Nuts		ASTM A193	3 Gr. B7/ A194	Gr. 2H				
7.8		nt Screw	has min G	Manufactu	rer's Standard	1				l
8.0	Corros	ion Allowa	nce		: 1.5 mm					
9.0	Locatio	on			: Above G	Ground	Burie	ed 📃		
10.0	Stem E	Extension F	Requiremen	ıt	: Yes		No			
11.0	Gear C	Operator Re	equirement		: Yes		No			
12.0	Gas Po	owered Act	tuator Requ	iirement	: Yes		No			
13.0	Fire Re	esistant De	sign Requir	rement	: Type-Te	est as per Standard API 607/B	3S EN: 104	197		
14.0	Valve	Testing R	equirement	t						
14.1	Hydros	static Test		Body	Test Pressur	re (min.), kg/cm2(g) 210	N	/linimum Di	uration, minutes	
1/1 2	Air Tos	t.		Seat		155				
14.2	All Tes					5.0 - 7				
15.0 15.1	Valve Surfac	Painting S	on by Short	n t Blasting o	s ner arado C	A 2 1/2 Swedich Standard Slo	S-055 000			
15.1	For ab	ove ground	d installation	1-Three coa	ts of corrosic	on resistant paint shall be applie	ed with min	imum thick	ness of 300 micron	
	(Perm	issible thic	kness in ea	ch coat sha	all be within 8	0 to 120 micron). Colour of pair	nt shade sh	nall be RAL	-7038, however any	
16.	cnange Lock C	e in colour )pen/Lock	snall be fina Close Requ	aiized durin iirement: <b>A</b> :	y drawing app s indicated in	proval stage. n P&ID				
1	Notes:	chall have	an inhara-	t feature	aina tha lina -	recurs to oncurs that the line	Drecours -	annot com	e taner locking of the plug/	
Ι.	plug m	ovement ir	to the tape	r, i.e., valve	es shall be of	"pressure to ensure that the line "pressure-balanced design".	pressure c	annot caus	se taper locking of the plug/	
2.	Paintin	g procedu	re of valves	shall be as	per Manufac	cturer's standard				
3.	Materia	al test certi	ficates and	hydrostatic	test reports s	shall be furnished prior to despa	atch.	ial acris		
4. 5	Detaile	y v-notch to d dimensio	est for body	r, piug, sten as showing	ra stuas/nut	s will be conducted as per relev a with part numbers and materi	vant materi jals shall be	al code. e submitter	I for Purchaser's	
0.	approv	al prior to	manufacture	e of the val	ves.Min. She	ell Wall Thickness as per BS	5353.	5051111160		
6.	Valves	shall have	valve posi	tion indicate	or.					
7.	Stops :	shall be pro	ovided to er	nsure positi	ve alignment	of plug with ports and ensure p	proper insta	allation of h	andle.	
8. 9	⊏ach v Valves	aive shall shall be in	ue provided	d approved	non. I by Purchase	er before despatch				
10.	Gland	packing as	sembly sha	all permit re	pair of gland	packing under full line pressure	э.			
11.	Inspec	tion and Te	esting shall	be done as	per data she	eet, specification & BS:12266 (F	Part-I).			
12.	Minimu	im port are	a for regula	r pattern sl	nall be 55%.	and the second second second second second second second second second second second second second second second	alast i t	mandles to t	and and the same has the third set	
13	Bidder	shall clear ed for. Whe	iy write valv erever bidde	es materia er agrees w	i (equivalent o rith valves me	or superior) offered by them agaterial as mentioned above in M	∣aınst each ∕lecon's dat	part/materi tasheet. bio	ial of the valve in the space dder shall clearly	
	indicate	Agreed"								
REV. NO.	DATE	ZONE		DESCRIPTI REVISION	ONS IS	BY	REFE	RENCES	DRG. NO.	
SECTION	N OIL	& GAS			CLIENT :					
	NAME	DATE	СНКД	DATE			(4	2001-T		
DSGN	PM	25.04.12	AKJ	25.04.12	PROJECT :		s.	0001 Convint	MECON LIMITED	
DRWN								_		
	(FD		,				SCAL			REV
	-0		(	O. F. JAIN	DATA SH	(NB < 2")	DATA	SHEEL NO.:	WEG/WIND/05/28/M/001/D5/PV/82	
						····· · - /	1			

# Page 105 of 644

REV. NO. SECTIO	N OIL NAME PM	<ul> <li>"Agreed"</li> <li>ZONE</li> <li>&amp; GAS</li> <li>DATE</li> <li>25.04.12</li> </ul>	CHKD AKJ	DESCRIPTI REVISION DATE 25.04.12	ONS NS CLIENT : PROJECT :	ВҮ	REFE	RENCES	DRG. NO.			
REV. NO.	DATE N OIL	& GAS	СНКД	DESCRIPTI	INS IS CLIENT :	ВҮ	REFER	RENCES	DRG. NO.			
REV. NO. SECTIO	DATE	ZONE		DESCRIPTI REVISION	ONS NS CLIENT :	ВҮ	REFER		DRG. NO.			
REV. NO.	indicate DATE	ZONE	•	DESCRIPT		BY	REFE	RENCES	DPG NO			
	indicate	e "Agreed"				D)/						
	nrovide	ea tor. Whe	erever bidde	er agrees w	viun vaives materia	a mentioned above	in wecon's dat	asneet, bid	uer shall clearly			
13	Bidder	shall clear	ly write valv	ves materia	I (equivalent or su	perior) offered by ther	n against each i	part/materia	al of the valve in the space			
11. 12.	Inspec Minimu	tion and Te um port are	esting shall a for regula	be done as ar pattern sl	s per data sheet, s ha∥ be 55%.	pecification & BS:122	юю (Part-I).					
10.	Gland	packing as	sembly sha	all permit re	pair of gland pack	ing under full line pres	sure.					
8. 9.	⊨ach v Valves	aive shall shall be in	pe provided spected an	a with a wre	non. d by Purchaser be	fore despatch.						
7.	Stops	shall be pro	ovided to er	nsure positi	ive alignment of pl	ug with ports and ens	ure proper insta	Ilation of ha	andle.			
6.	approv Valves	shall have	e valve posi	e or the val	ves. win. Valve E or	body vvali Thickness	as per ASME I	. 10.34				
5.	Detaile	ed dimensio	onal drawing	gs showing	cross-section with	h part numbers and m	aterials shall be	submitted	for Purchaser's			
3. 4.	Charpy	y V-notch te	est for body	/, plug, ster	n & studs/nuts will	be conducted as per	relevant materia	al code.				
2.	Paintin	ig procedu	re of valves	s shall be as	s per Manufacture	r's standard.	espatch					
	plug m	ovement in	nto the tape	er, i.e., valve	es shall be of "pres	ssure-balanced desig	า".		a capar looking of the plugr			
1.	Notes: Valves	shall have	an inherer	nt feature u	sing the line press	sure to ensure that the	line pressure o	annot caus	e taper locking of the plug/			
10.	LUCK		olose iteyl	arement. A	a mulcaleu III Põ							
16	change	e in colour	shall be fina	alized durin	g drawing approva	al stage. ID						
15.2	(Perm	ove ground issible thic	a installation kness in ea	ich coat sha	als of corrosion re all be within 80 to	sistant paint shall be a 120 micron). Colour o	pplied with mini paint shade sh	nall be RAL-	-7038, however any			
15.1	Surfac	e preparati	on by Shor	t Blasting a	s per grade SA 2	1/2, Swedish Standar	d SIS-055 909.		acce of 200 misson			
15.0	Valve	Painting S	pecificatio	on								
14.2	Air Tes	st		Ceat		5.6 7						
14.1	Hydros	static Test		Body Seat		210 155						
	Vuive	- soung K	-qui enien	_	Test Pressure (m	in.), kg/cm2(g)	N	1inimum Du	iration, minutes			
14 0	Valve '	Testina Ra	equiremen	t								
13.0	0 Fire Resistant Design Requirement : Type-Test as per Standard API 607/BS EN: 10497											
12.0	Gas Po	owered Act	tuator Requ	uirement	: Yes		No					
11.0	Gear C	Operator Re	equirement		: Yes 📃		No					
10.0	Stem E	Extension F	Requiremer	nt	: Yes		No					
9.0	Locatio	on			: Above Grour	nd	Burie	d				
0.0	Corros	ion Allowal			. 1.5 mm							
8.0	Corros		nce		15 mm							
7.8	Lubrica	tuas/Nuts Int Screw		AS M A32 Manufactu	u Gr.L <i>/</i> / ASTM A19 rer's Standard	94 Gr.4						
7.6	Gasket	tudo/NL-+-		N.A.								
7.5	Gland F	Packing		Graphite/ F	PTFE							
7.4	Gland	NU Casting)		35 316 (No ASTM A10	5 casting) / A 350 G 5	DI. LF2						
7.2	Plug (Li	ubricated)		SS316/ A 3	52 Gr LCB/A 350 G	Fr. LF2 with 75 µENP o	oating					
7.1	Body			A 352 Gr. L	.CB/A 350 Gr. LF2							
7.0	Valve	Materia <mark>l</mark> S Part	pecificatio	n I	Snacific	ed Material			Material Offered	1		
6.3.	Flange	s (whereve	er applicabl	e)	: a) RF	FF F	RTJ					
					Socket welded	with Pup Piece of 100m	n length of A333	Gr.6 Sch. 16	60			
					: Socket Weld	both ends	Socket	velu as per As Welded 3000# (a	as per ASME B16.11)			
6.2.	End Co	onnections			: Flanged bot	hends [	Flange	ed as per ASN	NE B 16.5			
6.1.	Pattern	1			: Short		Regular		Venturi			
6.0	Valve	Construct	ion Design	n								
5.3	Thickn	ess, mm			:							
5.1 5.2	Materia Diamet	al ter (OD), m	nm (inch)									
5.0	Conne	ecting Pipe	e Specifica	tion	: <b>N.A.</b>							
4.0	Conne	cting Pipel	ine Design	Pressure, k	(g/cm <sup>2</sup> (g) :		Des	sign Temp	erature, °C : -45°C to 65°C			
3.0	MECO	N's Techni	cal Specific	cation No.	:							
2.0	Valve \$	Size (NB),	mm (inch)		:	ANSI Rating : 800	ŧ		Design Standard : BS:5353			
1.0	vaivel		ы "									
1.0	Valva	Manufactu	or									

				DATA SHEET FOR (	CHECK VALVE			
1.0	Valve Manufac	turer		:				
2.0	Service			:				
3.0	Valve Size (NB	), mm (inch)		:	ANSI Rating :	150#	Design Standard : API 6D	
4.0	MECON's Tech	nical Specif	ication No.	: MEC/TS/05/62/004,Rev-2				
5.0	Connecting Pip	eline Design	Pressure, b	oar : <b>19 kg/cm2</b>	Design Tem	perature, °C	: -29°C to + 65°C	
6.0	Connecting Pi	ne Snecific:	ation		5	. ,		
6.1 6.2 6.3	Material Diameter (OD), Thickness, mm	mm (inch)		· : : :				
7.0 7.1	<b>Valve Constru</b> ⊺ype	ction Desig	n	:				
7.2	End Connection	าร		<ul> <li>Flanged both ends</li> <li>Butt Weld both ends</li> <li>Flanged one end, butt weld</li> <li>Socket Weld as per ASME E</li> </ul>	other end 3 16.11			
7.3	Flanges (where	ever applicat	ole)	: a) RF FF b) Serrated Smoo		RTJ iicroinches AARH)	NA	]
8.0	Valve Material	Specificatio	on					-
81	Pa Body	rt	ASTM A 2	Specified Material		Material Of	ffered (Equivalent or Superior	-
8.2	Cover		ASTM A 2	16 Gr.WCB				-
8.3	Disc/ Plates		(ASTM A	216 Gr. WCB + 13% Cr Steel Faci	ng) /			
84	Body Seat Rings	(See Note-3)	13% Cr St	eel (Stellited) 16 Gr. WCB+13% Cr.Steel Facing	(Stellited)			-
8.5	Disc Hinge		ASTM A 2	16 Gr. WCB/ A 515 Gr. 70/ 13% C	r Steel			-
8.6	Hinge Pin		13% Cr St	eel (No Casting)				
8.7	Cover Stud Bo	ts	ASTM A 1	93 Gr. B7				-
8.9	Cover Gasket		SS 304/31	6 Spiral Wound with Grafoil				-
8.1	Spring		Inconel X	-750				
9.0	Corrosion Allov	/ance		: <b>1.5 mm</b>				
10.0	Location			: Above Ground	Buried	]		
11.0	Stem Extension	n Requireme	nt	: Yes	No	]		
12.0	Gear Operator	Requiremen	t	: Yes	No			
13.0	Gas Powered A	Actuator Req	uirement	: Yes	No			
14.0	Fire Resistant I	Design Requ	irement	:				
15.0	Valve Testing	Requiremer	nt	Toot Processo (min.) kalom2(a)	Minimum D	uration minutos	7	
45.4	L badaa atatia Taa		D. d.		Willin din D		-	
15.1	Hydrostatic Tes	st	Body Seat	23	A A	PI 6D	-	
16.0	Valve Painting	Specificati	on		•		-	
i) ii)	Surface prepar For above grou ( Permissible th	ation by Sho nd installatic ickness in e	rt Blasting a on-Three coa ach coat sha	s per grade SA 2 1/2, Swedish Sta ats of corrosion resistant paint shall all be within 80 to 120 micron).	ndard SIS-055 be applied with	909. n minimum thickne	ess of 300 micron	
17.0	Lock Open/ Loc	ck Close Red	quirement	: <b>N.A.</b>	i.			
	Notes: 1. This Val 2. Inspecti API 6D, 3. Seats sl 4. 5% of va 5. Bidder s Wherev 6. Charpy bonnet, at 0°C.T 7. Hardnes cross se measure 8. Painting	ve Data She on and Testi BS EN 1226 all be non-rri alves shall ui hall clearly v er bidder agi V' notch tesi stem, disc, t he minimum is test shall l ection shall b ection shall b procedure c	eet shall be r ng shall be a 86 and other enewable ini ndergo radic vrite all/ any rees with ME to n each he body seat, er a average ab be carried ou e taken for t e taken for t sesenting the of the valves	ead in conjunction with MECON's as per attached QAP, this Data She relevant standards. Min. Valve E tegral type. Drain Plug shall be pgraphic examination. deviation against each part/ mater CON's data sheet, bidder shall de at of base material shall be conduct of flange, welding ends as well as sorbed energy per set of three spet to n each heat of base material fo his purpose and the maximum har entire thickness.	Technical Speci set, MECON's 1 body Wall Thir provided to dr ial of valve in th araly indicate "ar ted for all press cimen shall be all pressure co dness shall not undard.	ification No. MEC/ Fechnical Specifica ckness as per A ain valve cavity e space provided greed". sure containing & d grial as per ASTM 27 J with an indiviontaining parts of ti exceed 248 HV10	TS/05/62/004,Rev-2. ation No. MEC/TS/05/62/004,Rev- SME B16.34 . for . controlling parts such as body, A370. The test shall be conducted dual minimum per specimen of 22 he valve. A full thickness based on minimum four	2, 1 J.
	9. Material	Test Certific	ates and Hy	dro Test Reports shall be furnishe	d prior to dispat	ch.		
REV. NO.	` ZONE		DESCRIPT	ONS BY	APPRD	1		
SECTIO			REVISION			REFERENCES	DRG. NO.	
SECHO	IN FRUCESS (							
	NAME DATE	CHKD	DATE			मेकांग		
DSGN	PM 25.04.12	AKJ	25.04.12	PROJECT :		3 5051 Cmt #	MECON LIMITED	
DRWN		1	1			SCALE :		RFV/
APPRO	/ED		O.P. Jain	DATA SHEET FOR CHECK (NB > 2")	VALVES	DATA SHEET NO.: N	EC/WINO/05/28/M/001/DS/CV/76	

			DATA SHEET FOR (	CHECK VALVE				
1.0	Valve Manufacturer		:					
2.0	Service		:					
3.0	Valve Size (NB), mm	ı (inch)	: ANSI Rating : 1	50#		Design Standard : API 6D		
4.0	MECON's Technical	Specification No.	: MEC/TS/05/62/004,Rev-2					
5.0	Connecting Pipeline	Design Pressure, k	ːɡ/cm²(ɡ) : <b>19</b>	Design Terr	nperature, °C	: -45°C to + 65°C		
6.0 6.1 6.2 6.3	<b>Connecting Pipe Sp</b> Material Diameter (OD), mm ( Thickness, mm	<b>Decification</b> (inch)	: : :					
7.0 7.1	Valve Construction Type	Design	:					
7.2	End Connections		<ul> <li>Flanged both ends</li> <li>Butt Weld both ends</li> <li>Flanged one end, butt weld</li> <li>Socket Weld as per ASME F</li> </ul>	other end 3 16.11				
7.3	Flanges (wherever a	pplicable)	: a) RF FF b) Serrated Smoo	 oth (125 to 200 m	RTJ	NA		
8.0	Valve Material Spec	ification	Specified Material		Material Offe	ered (Equivalent or Superior		
8.1	Body	ASTM A35	2 Gr. LCB/ A 350 GR. LF2					
8.2	Cover	ASTM A35	2 Gr. LCB/ A 350 GR. LF2				_	
0.5	Disc/ Plates	A 350 GR.	LF2 +Stellited		-			
8.4	Body Seat Rings (See N	Note-3) SS316/ AS	TM A352 Gr. LCB /					
8.5	Disc Hinge	A 350 GR. SS316/ AS	LF2 + Stellited				_	
0.0		A 350 GR.	LF2					
8.6 8.7	Hinge Pin	SS 316 (No	o casting) / A 350 Gr. LF2				_	
8.8	Nuts	ASTM A32	94 Gr.4				_	
8.9	Cover Gasket	SS 304/31	6 Spiral Wound with Grafoil					
8.1	Spring	Inconel X-	/50					
9.0	Corrosion Allowance		: 1.5 mm		1			
10.0	Location		: Above Ground	Buried	]			
11.0	Stem Extension Req	uirement	: Yes	No	]			
12.0	Gear Operator Requ	irement	: Yes	No				
13.0	Gas Powered Actuat	or Requirement	: Yes	No				
14.0	Fire Resistant Desig	n Requirement	:					
15.0	Valve Testing Requ	lirement	Test Pressure (min.), kg/cm2(g)	Minimum D	uration, minutes			
15.1	Hydrostatic Test	Body Seat	32 23	As po As po	er API 6D er API 6D			
16.0 i) ii)	Valve Painting Spec Surface preparation For above ground in: ( Permissible thickne	<b>cification</b> by Short Blasting a stallation-Three coa sss in each coat sha	s per grade SA 2 1/2, Swedish Sta ats of corrosion resistant paint sha all be within 80 to 120 micron).	andard SIS-055 II be applied wi	909. Ith minimum thickn	ess of 300 micron		
17.0	Lock Open/ Lock Clo	se Requirement	: <b>N.A.</b>					
	Notes: 1. This Valve Da	ata Sheet shall be r	ead in conjunction with MECON's	Technical Spec	cification No. MEC	/TS/05/62/004,Rev-2.		
	API 6D, BS E 3. Seats shall be 4. 5% of valves 5. Bidder shall c	N 12266 and other e non-renewable in shall undergo radio learly write all/ any	relevant standards. Min. Valve I tegral type. Drain Plug shall be graphic examination. deviation against each part/ mate	Body Wall Thi provided to d rial of valve in t	ickness as per A Irain valve cavity	ASME B16.34 . / .	/ <b>-</b> 2,	
	Wherever bid	der agrees with ME	CON's data sheet, bidder shall cl	early indicate "a	agreed".	controlling parts such as body		
	bonnet, stem,	disc, body seat, er	nd flange, welding ends as well as	the bolting mat	terial as per releva	nt material code.		
	7. Hardness tes	t shall be carried ou	ut as per relevant material code	andard				
	<ol> <li>Painting processing</li> <li>Material Test</li> </ol>	Certificates and Hy	when the sper manufacturer's St arro Test Reports shall be furnishe	ed prior to dispa	atch.			
REV. NO.	DATE ZONE	DESCRIPTI	ONS BY	APPRD				
	1.2000 - 2000 - 1	REVISION	IS	1	REFERENCES	DRG. NO.		
SECTIO	N PROCESS & PIP	PING	CLIENT :					
	NAME DATE	CHKD DATE			August of the second se			
DSGN	PM 25.04.12	AKJ 25.04.12	PROJECT :		40 8001 Carver	MECON LIMITED		
DRWN					SCALE :		REV/	
APPRO∖	/ED	O. P. JAIN	DATA SHEET FOR CHECK (NB <u>&gt;</u> 2")	VALVES	DATA SHEET NO.: MI	EC/WINO/05/28/M/001/DS/CV/77	0	
				DATA SHEET FOR C	CHECK VALVE			
-------------------	---	---	--	--	--	---	--	------------------
1.0	Valve Manufact	turer						
2.0	Service			:				
3.0	Valve Size (NB	), mm (inch)		:	ANSI Rating :	300#	Design Standard : API 6D	
4.0	MECON's Tech	nical Specifi	ication No.	: MEC/TS/05/62/004,Rev-2				
5.0	Connecting Pip	eline Design	Pressure, k	(g/cm <sup>2</sup> (g) : <b>49</b>	Design Terr	iperature, °C	: -29°C to + 65°C	
6.0	Connecting Pi	ne Snecific:	ation		0	•		
6.1 6.2 6.3	Material Diameter (OD), Thickness, mm	mm (inch)		· · ·				
7.0 7.1	<b>Valve Constru</b> e ⊺ype	ction Desig	n	:				
7.2	End Connection	าร		<ul> <li>Flanged both ends</li> <li>Butt Weld both ends</li> <li>Flanged one end, butt weld</li> <li>Socket Weld as per ASME E</li> </ul>	other end 3 16.11			
7.3	Flanges (where	ever applicab	ole)	: a) RF FF b) Serrated Smoo		RTJ icroinches AARH)	NA	
8.0	Valve Material	Specificatio	on					_
81	Par	rt		Specified Material		Material Of	fered (Equivalent or Superior	-
8.2	Cover		ASTM A 2	16 Gr.WCB				-
8.3	Disc/ Plates		(ASTM A	216 Gr. WCB + 13% Cr Steel Faci	ng) /			7
0.4	Dady Cast Diana	(0 No.1. 0)	13% Cr St	teel (Stellited)				_
8.5	Disc Hinge	(See Note-3)	ASTM A 2	216 Gr. WCB/ A 515 Gr. 70/ 13% C	r Steel			-
8.6	Hinge Pin		13% Cr St	teel (No Casting)				_
8.7	Cover Stud Bol	ts	ASTM A 1	93 Gr. B7				
8.8	Nuts		ASTM A 1	94 Gr. 2H				_
8.9	Spring		Inconel X	-750				-
9.0	Corrosion Allow	/ance		: 1.5 mm				_
10.0	Location			: Above Ground	Buried	]		
11.0	Stem Extensior	n Requireme	nt	: Yes	No	]		
12.0	Gear Operator	Requiremen	t	: Yes	No			
13.0	Gas Powered A	Actuator Req	uirement	: Yes	No			
14.0	Fire Resistant	Design Requ	irement	:				
15.0	Valve Testing	Requiremer	nt	Test Pressure (min.) ka/cm2(a)	Minimum D	uration minutes	1	
				rest ressure (min.), kg/cm2(g)	Winning D	dration, minutes		
15.1	Hydrostatic Tes	st	Body Seat	76 57	A A	API 6D API 6D	-	
10.0				•	4		-	
16.0 i) ii)	Valve Painting Surface prepara For above grou ( Permissible th	Specification ation by Sho nd installation ickness in e	<b>on</b> rt Blasting a on-Three coa ach coat sha	s per grade SA 2 1/2, Swedish Sta ats of corrosion resistant paint sha <b>ll</b> all be within 80 to 120 micron).	ndard SIS-055 be applied with	909. 1 minimum thickne	ss of 300 micron	
17.0	Lock Open/ Loc	k Close Rec	quirement	: N.A.				
	Notes:         1.       This Val         2.       Inspectiv         API 6D,       Seats st         4.       5% of var         5.       Bidder st         6.       Charpy var         bonnet, at 0°C.T       Hardness         cross se       measure         8.       Painting         9.       Material	ve Data She on and Testii BS EN 1226 hall be non-ru alves shall un hall clearly v er bidder agr V' notch test stem, disc, t he minimum is test shall h extion shall b extion shall b ements repre procedure c	et shall be r ng shall be a 66 and other enewable in ndergo radic vrite all/ any rees with MB t on each he oody seat, er a average ab oe carried or e taken for t ssenting the of the valves <u>aates and Hy</u>	ead in conjunction with MECON's as per attached QAP, this Data She relevant standards. Min. Valve B tegral type.Drain Plug shall be p graphic examination. deviation against each part/ mater ECON's data sheet, bidder shall de cat of base material shall be conduc and flange, welding ends as well as sosrbed energy per set of three spe ut on each heat of base material for this purpose and the maximum hare entire thickness. shall be as per Manufacturer's Sta dro Test Reports shall be furnishe	Technical Spec et, MECON's T ody Wall Thic rovided to dra ial of valve in th andy indicate "a ted for all press the bolting mate cimen shall be all pressure or dness shall not indicat. d prior to dispat	ification No. MEC/ rechnical Specifica kness as per AS ain valve cavity le space provided greed". sure containing & d arial as per ASTM 27 J with an indivi ontaining parts of ti exceed 248 HV10 cch.	TS/05/62/004, Rev-2. ation No. MEC/TS/05/62/004, Rev- SME B16.34 . controlling parts such as body, A370. The test shall be conducted dual minimum per specimen of 22 he valve. A full thickness based on minimum four	-2, 1 ! J.
REV. NO.	ZONE		DESCRIPT	IONS BY	APPRD			
			REVISIO	NS	•	REFERENCES	DRG.NO.	
SECTIO	N PROCESS &	PIPING	1	CLIENT :				
	NAME DATE	СНКД	DATE					
DSGN	PM 25.04.12	AKJ	25.04.12	PROJECT :		40 5001 Cart 2	MECON LIMITED	
DRWN								
APPRO\	/ED		O.P. Jain	DATA SHEET FOR CHECK (NB <u>&gt;</u> 2")	VALVES	SCALE : DATA SHEET NO.: M	1EC/WINO/05/28/M/001/DS/CV/78	REV 0

			DATA SHEET FOR	CHECK VALVE			
1.0	Valve Manufacturer	r	:				
2.0	Service		:				
3.0	Valve Size (NB), m	m (inch)	: ANSI Rating	: 300#		Design Standard : API 6D	
4.0	MECON's Technica	I Specification N	o. : MEC/TS/05/62/004,Rev-2				
5.0	Connecting Pipeline	e Design Pressu	re, bar : 49 kg/cm2	Design Terr	nperature, °C	: -45°C to + 65°C	
	O	-	-				
6.0 6.1 6.2 6.3	Material Diameter (OD), mm Thickness, mm	i (inch)					
7.0 7.1	Valve Constructio	n Design	:				
7.2	End Connections		: Flanged both ends				
			: Butt Weld both ends	d other end			
			Socket Weld as per ASME	B 16.11			
7.3	Flanges (wherever	applicable)	: a) RF FF b) Serrated Sm	 poth (125 to 200 n	RTJ nicroinches AARH)	NA	] ]
8.0	Valve Material Spe	ecification					-
81	Part Body	ASTM	A352 Gr. I CB/ A 350 GR. I F2		Material Off	ered (Equivalent or Superior	-
8.2	Cover	ASTM	A352 Gr. LCB/ A 350 GR. LF2				_
8.3	Disc/ Plates	SS316	/ ASTM A352 Gr. LCB /		-		
8.4	Body Seat Rings (See	Note-3) SS316	ASTM A352 Gr. LCB /				-
		A 350	GR. LF2 + Stellited				_
8.5	Disc Hinge	SS316	/ ASTM A352 Gr. LCB / GR 1 F2				-
8.6	Hinge Pin	SS 31	6 (No casting) / A 350 Gr. LF2				1
8.7 8.8	Cover Stud Bolts	ASTM	A320 Gr.L7				-
8.9	Cover Gasket	SS 304	/316 Spiral Wound with Grafoil				-
8.1	Spring	Incone	X-750				]
9.0	Corrosion Allowand	e	: 1 <b>.</b> 5 mm				
10.0	Location		: Above Ground	Buried	]		
11.0	Stem Extension Re	quirement	: Yes	No	]		
12.0	Gear Operator Req	uirement	: Yes	No			
13.0	Gas Powered Actua	ator Requiremen	t : Yes	No			
14.0	Fire Resistant Desi	gn Requirement	:				
15.0	Valve Testing Req	uirement	Test Pressure (min.), kg/cm2(g)	Minimum D	uration, minutes	]	
15.1	Hydrostatic Test	B	ody 76 eat 57	As p As p	er API 6D er API 6D		
16.0	Valve Painting Sp	ecification					
i) ii)	Surface preparation For above ground in (Permissible thickn	n by Short Blasti nstallation-Three ness in each coa	ng as per grade SA 2 1/2, Swedish S coats of corrosion resistant paint st shall be within 80 to 120 micron).	itandard SIS-055 nall be applied wi	5 909. ith minimum thickn	ess of 300 micron	
17.0	Lock Open/ Lock C	lose Requireme	nt : <b>N.A.</b>				
	Notes:						
	1. This Valve D	Data Sheet shall	be read in conjunction with MECON	s Technical Spe	cification No. MEC	/TS/05/62/004,Rev-2.	
	2. Inspection a	nd Testing shall	be as per approved QAP, this Data	Sheet, MECON's	s Technical Specifi	cation No. MEC/TS/05/62/004,Re	v <b>-</b> 2,
	3. Seats shall b	be non-renewabl	e integral type. Drain Plug shall b	e provided to	drain valve cavit	X	
	4. 5% of valves	s shall undergo r	adiographic examination.	- p			
	<ol> <li>Bidder snall</li> <li>Wherever bi</li> </ol>	idder agrees with	MECON's data sheet, bidder shall	clearly indicate "	the space provided agreed".	a for .	
	<ol><li>Charpy 'V' n</li></ol>	otch test on eac	heat of base material shall be cond	lucted for all pres	ssure containing &	controlling parts such as body,	
	7. Hardness te	n, disc, body sea st shall be carrie	t, end flange, welding ends as well a d out as per relevant material code	is the bolting ma	terial as per releva	ant material code.	
	8 Painting pro	cedure of the va	ves shall be as per Manufacturer's	Standard.			
	9. Material Tes	st Certificates an	d Hydro Test Reports shall be furnis	hed prior to dispa	atch.		
				405			
REV. NO.	DATE ZONE	DESCR	SIONS BY	APPRD	REFERENCES	DRG. NO.	
SECTIO	N PROCESS & PI	PING	CLIENT :		$\frown$		
	NAME DATE	CHKD DAT	:				
DSGN	PM 25.04.12	AKJ 25.04	12 PROJECT :		No BOOL CONTRACT	MECON LIMITED	
DRWN					SCALE.		חביי
APPRO\	/ED	0. P. J <i>i</i>	AIN DATA SHEET FOR CHECK (NB <u>&gt;</u> 2")	<b>K VALVES</b>	DATA SHEET NO.: M	EC/WINO/05/28/M/001/DS/CV/79	- <sup>KEV</sup>

			DATA SHEET FOR C	HECK VALVE		
1.0	Valve Manufacturer		:			
2.0	Service		:			
3.0	Valve Size (NB), mm (ind	:h)	: ANSI Rati	ng : 600#	Design Standard : API 6D	
4.0	MECON's Technical Spe	cification No.	MEC/TS/05/62/004 Rev-2			
4.0	MECON'S Technical Spe	cilication No.	. MEC/13/03/02/004,Rev-2			
5.0	Connecting Pipeline Des	ign Pressure,	bar : 92 kg/cm2	Design Temperature	e, °C : -29°C to + 65°C	
6.0 6.1 6.2 6.3	Connecting Pipe Specif Material Diameter (OD), mm (inch Thickness, mm	ication	: : :			
7.0 7.1	Valve Construction Des Type	ign	:			
7.2	End Connections		<ul> <li>Flanged both ends</li> <li>Butt Weld both ends</li> <li>Flanged one end, butt weld of Socket Weld as per ASME B</li> </ul>	Dther end 16.11		
7.3	Flanges (wherever applie	able)	: a) RF FF b) Serrated Smoot	RTJ	] NA [ ARH) NA [	
8.0	Valve Material Specifica	ation	On a stind Material		nial Offered (Environment on Overseine	
8.1	Body	ASTM A	Specified Material	Mate	erial Offered (Equivalent or Superior	
8.2	Cover	ASTM A	216 Gr.WCB			
8.3	Disc/ Plates	(ASTM A	216 Gr. WCB + 13% Cr Steel Faci	ng) /		
84	Body Seat Rings (See Note-	13% Cr S	teel (Stellited) 216 Gr. WCB+13% Cr. Steel Facing	(Stellited)		
8.5	Disc Hinge	ASTM A	216 Gr. WCB/ A 515 Gr. 70/ 13% C	Steel		
8.6	Hinge Pin	13% Cr S	teel (No Casting)			
8.7	Cover Stud Bolts	ASTM A	193 Gr. B7			
8.8 8.9	Nuts Cover Gasket	ASTM A '	194 Gr. 2H 16 Spiral Wound with Grafoil			
8.1	Spring	Inconel X	-750			
9.0	Corrosion Allowance		: 1.5 mm			
10.0	Location		: Above Ground	Buried		
11.0	Stem Extension Require	ment	: Yes	No		
12.0	Gear Operator Requirem	ent	: Yes	No		
13.0	Gas Powered Actuator R	equirement	: Yes			
14.0	Fire Resistant Design Re	quirement				
15.0	Valve Testing Requires					
10.0		ient	Test Pressure (min.), kg/cm2(g)	Minimum Duration, min	utes	
15.1	Hydrostatic ⊺est	Body	157	As per API 6D		
		Seat	114	As per API 6D		
16.0 i) ii)	Valve Painting Specific Surface preparation by S For above ground installa ( Permissible thickness in	ation hort Blasting a ation-Three co n each coat sh	is per grade SA 2 1/2, Swedish Star ats of corrosion resistant paint shall all be within 80 to 120 micron).	ndard SIS-055 909. be applied with minimum t	hickness of 300 micron	
17.0	Lock Open/ Lock Close F	Requirement	. <b>N.A.</b>			
	Notes:           1.         This Valve Data S           2.         Inspection and Te           API 6D, BS EN 12         Seats shall be no           4.         5% of valves shall clean           5.         Bidder shall clean           Wherever bidder         6.           6.         Charpy 'V' notch 1	theet shall be sting shall be 2266 and othe -renewable in I undergo radii y write all/ any agrees with M est on each he	read in conjunction with MECON's 1 as per attached QAP, this Data She r relevant standards.Min. Valve Br tegral type.Drain Plug shall be p ographic examination. deviation against each part/ materi ECON's data sheet, bidder shall cle at of base material shall be conduc	echnical Specification No. et, MECON's Technical Sp ody Wall Thickness as p rovided to drain valve or al of valve in the space pro arly indicate "agreed". ted for all pressure contair	MEC/TS/05/62/004,Rev-2. becification No. MEC/TS/05/62/004,F per ASME B16.34 . avity . byided for . bing & controlling parts such as body	Rev-2,
	<ul> <li>bonnet, stem, disc at 0°C. The minim</li> <li>Hardness test sha cross section sha</li> </ul>	c, body seat, e um average al II be carried o II be taken for	nd flange, welding ends as well as t psorbed energy per set of three spe ut on each heat of base material for this purpose and the maximum har	he bolting material as per cimen shall be 27 J with ar all pressure containing pa iness shall not exceed 248	ASTM A370. The test shall be condu h individual minimum per specimen c rts of the valve. A full thickness HV10 based on minimum four	of 22 J.
	8. Painting procedur	presenting the e of the valves ificates and H	entire thickness. s shall be as per Manufacturer's Sta vdro Test Reports shall be furnisher	ndard. I prior to dispatch		
REV. NO.	DATE ZONE	DESCRIPT	IONS BY	APPRD		
		REVISIO	NS	REFERENCE	ES DRG. I	NO.
SECTIO	N PROCESS & PIPING		CLIENT :			
	NAME DATE CHK	DATE			カー	
DSGN	PM 25.04.12 AKJ	25.04.12	PROJECT :	-10 -00 - 0 - 00	MECON LIMITE	ED
DRWN				0		
APPRO	/ED		DATA SHEET FOR CHECK	VALVES DATA SHEET	NO.: MEC/WINO/05/28/M/001/DS/CV/80	REV

			DATA SHEET FOR	CHECK VALVE			
1.0	Valve Manufacturer		:				
2.0	Service		:				
3.0	Valve Size (NB), mm (in	ch)	: ANSI Ratin	g : <b>600#</b>		Design Standard : API 6D	
4.0	MECON's Technical Spo	ecification No.	: MEC/TS/05/62/004,Rev-2				
5.0	Connecting Pipeline De	sign Pressure,	kg/cm <sup>2</sup> (g) : <b>92</b>	Design Ten	nperature, °C	: -45°C to + 65°C	
6.0 6.1 6.2 6.3	Connecting Pipe Spec Material Diameter (OD), mm (inc Thickness, mm	i <b>fication</b> h)	: : : :				
7.0 7.1	Valve Construction De Type	sign	:				
7.2	End Connections		<ul> <li>Flanged both ends</li> <li>Butt Weld both ends</li> <li>Flanged one end, butt weld</li> <li>Socket Weld as per ASME</li> </ul>	d other end B 16.11			
7.3	Flanges (wherever appli	cable)	: a) RF FF b) Serrated Smo	 oth (125 to 200 n	RTJ nicroinches AARH)	NA NA	]
8.0	Valve Material Specific Part	ation	Specified Material		Material Off	ered (Equivalent or Superior	٦
8.1	Body	ASTM A3	52 Gr. LCB/ A 350 GR. LF2		indicinal off	ered (Equivalent of Superior	_
8.2	Cover	ASTM A3	52 Gr. LCB/ A 350 GR. LF2				]
8.3	Disc/ Plates	A 350 GR	LF2 + Stellited		-		
8.4	Body Seat Rings (See Note-	3) SS316/ A	STM A352 Gr. LCB /				1
	<b>D</b>	A 350 GR	LF2 + Stellited				
8.5	Disc Hinge	SS316/ A	STM A352 Gr. LCB /				-
8.6	Hinge Pin	SS 316 (N	lo casting) / A 350 Gr. LF2				-
8.7	Cover Stud Bolts	ASTM A3	20 Gr.L7				1
8.8	Nuts Cover Casket	ASTM A1	94 Gr.4				-
8.1	Spring	Inconel X	-750				-
9.0	Corrosion Allowance		: 1.5 mm				-
10.0	Location		: Above Ground	Buried	]		
11.0	Stem Extension Require	ment	: Yes	No	]		
12.0	Gear Operator Requirer	nent	: Yes	No			
13.0	Gas Powered Actuator F	Requirement	: Yes	No 📃			
14.0	Fire Resistant Design R	equirement	:				
15.0	Valve Testing Require	ment	Test Pressure (min.), kg/cm2(g)	Minimum D	uration, minutes		
15.1	Hydrostatic Test	Body Seat	<u> </u>	As p As p	er API 6D er API 6D		
16.0 i) ii)	Valve Painting Specific Surface preparation by S For above ground instal ( Permissible thickness	<b>sation</b> Short Blasting a lation-Three co n each coat sh	as per grade SA 2 1/2, Swedish S ats of corrosion resistant paint sh all be within 80 to 120 micron).	andard SIS-055 all be applied wi	5 909. ith minimum thickn	ess of 300 micron	
17.0	Lock Open/ Lock Close	Requirement	: N.A.				
	Notes: 1. This Valve Data : 2. Inspection and T API 6D, BS EN 1 3. Seats shall be no 4. 5% of valves sha 5. Bidder shall clea Wherever bidder 6. Charpy 'V' notch bonnet, stem, dis 7. Hardness test sh 8. Painting procedu 9. Material Test Ce	Sheet shall be esting shall be 2266 and other on-renewable in lundergo radi dy write all/ any agrees with Mi test on each he c, body seat, e all be carried o re of the valves	read in conjunction with MECON's as per attached QAP, this Data Sr relevant standards. Min. Valve itegral type. Drain Plug shall be ggraphic examination. / deviation against each part/ mat ECON's data sheet, bidder shall c eat of base material shall be cond nd flange, welding ends as well a ut as per relevant material code s shall be as per Manufacturer's S with De at Reports shall be furnier	s Technical Spenet, MECON's Body Wall Th provided to d erial of valve in learly indicate ", ucted for all pre- s the bolting ma tandard. ed prior to disco	cification No. MEC Technical Specific ickness as per <i>I</i> Irain valve cavity the space provided agreed". ssure containing & terial as per releva atch.	/TS/05/62/004,Rev-2. ation No. MEC/TS/05/62/004,Rev- SME B16.34 . / . d for . controlling parts such as body, int material code.	2,
REV. NO.	DATE ZONE	DESCRIPT	IONS BY	APPRD			
		REVISIO	NS		REFERENCES	DRG. NO.	
SECTIO	N PROCESS & PIPINO	3	CLIENT :				
	NAME DATE CHK	D DATE			( And T		
DSGN	PM 25.04.12 AK	25.04.12	PROJECT :		NO BOOT CONVER	MECON LIMITED	
DRWN							
APPRO\	/ED	O. P. JAIN	DATA SHEET FOR CHECK (NB <u>&gt;</u> 2")	VALVES	SCALE : DATA SHEET NO.: M	EC/WINO/05/28/M/001/DS/CV/81	REV 0

	1.	Valve Mar	nufacturer		:					
	2.	Size			:	Rating : AN	ISI <b>150</b> #		Design Standard : BS:1873	
	3.	Purchase	r's Specifica	ation	: Refer Tech	nnical notes for G	ate & Globe Valves			
	4.	Design Pr	essure		: 19kg/cm <sup>2</sup> (	g)	Design Temperature	: -29°C to + 6	5°C	
	5.	Corrosion	Allowance		: 1.5mm	-	0	Service	:	
	6	End Conn	ections		· Elanged bot	h onde as nor ASM	IE R 16 5		7	
	0.		Inclions		Butt Weld bo Flanged one Socket weld	oth ends as per ASM e end butt weld othe both ends as per A	5 er end ASME B16.11			
	7.	Flanges (	where appli	cable)	: a) RF	FF RTJ				
					b) Serrated	Smoo	oth (125 to 200 AARH)			
	8.	Connecti	ng Pipe Spe	ecification	:					
	9.	Valve Mat	terial Specif	fication :						
	0.4	Dealer	Part		ACTN A 246 0	Materia		Material	Offered (Equivalent or Superi	or)
	9.1	Boopet (P	oltod)		ASTM A 216 G					
	9.2	Stem (Ris	ina)		13% Cr Steel	(No Casting)				
	9.5		nig) o Plug/Ball		(ASTM A 216 C	r WCB + 13% Cr	Steel Facing) /			
	5.4		e r iug/ball	турс)	13% Cr Steel (	Stellited)	oteer racing/r			
	95	Body Sea	t Rina		ASTM A 216 G	r WCB+13% Cr S	teel Facing (Stellited)			
	9.6	Stem Pac	king (Renev	wahle	Corrosion inhi	hited die formed	flexible			
	0.0	with val		stream)	graphite with k	praided anti extru	sion rings	1		
	0.7		ve open on	Sileann)	Malloable Iron	/ Cast Stool/ Eab	Stool			
	9.7	Ponnot Bo				Cast Steel/ Fab.	Steel			
	9.6	Bonnet Bo			A 193 Gr. B/					
	9.9	Bonnet N			A194 GL ZH	00.040 + 0				
	11. 12. i) ii) <b>Notes:</b> 1. 2. 3. 4. 5. 6. 7. 8	a) Body b) Seat Pnuematid Air Painting S Surface p For above ( Permissi Valve spe Valve sha Testing sh Bidder sha Wherever Charpy 'V welding et absorbed Hardness cross sect measuren Stem pacl	cTest Press pecification reparation b ground ins ble thicknes cification sh Il be design hall be as pe all clearly w bidder agre ' notch test nds as well energy per test shall be tion shall be nents repre- king shall b	sure with as: by Short Bl stallation-T ss in each neet shall the neet shall the eff or intri- er BS EN 1 rite all/ any ees with M on each the set of three e carried of the value of the value	<ul> <li>32 kg/cm<sup>2</sup>/</li> <li>23 kg/cm<sup>2</sup>/</li> <li>5.6 - 7 kg/cn</li> <li>asting as per gr hree coats of co coat shall be wi</li> <li>be read in conjuinsically fire safe (2266-1, approver y deviation again (ECON's data st eat of base material as use specimen sha but on each heat this purpose an e entire thickness le with valve oppion</li> </ul>	(g) (g) n2 (g). ade SA 2 1/2, Swe prosion resistant pathin 80 to 120 micr action with technica a. Min. Body & Born ed QAP, this specifinst each part/ mate neet, bidder shall d erial shall be condu per ASTM A370. T II be 27 J with an in t of base material for d the maximum ha s. en on stream .	dish Standard SIS-055 909 aint shall be applied with m on). al notes for Gate and Glob- net Thickness as per BS 18 cation and other relevant s rial of valve in the space p early indicate "agreed". Icted for all pressure conta- he test shall be conducted dividual minimum per spe or all pressure containing p rdness shall not exceed 24	). inimum thickne e valves. 373 . tandards. rovided for . ining parts sucl at 0°C. The min cimen of 22 J. parts of the valv l8 HV10 based	ss of 300 micron n as body, end flange, nimum average e. A full thickness on minimum four	
	8. 9.	Painting p Material T	rocedure of est Certifica	f the valve ates and H	s shall be as pe lydro Test Repo	r Manufacturer's St rts shall be furnish	andard. ed prior to dispatch.			
REV. NO.	DATE	ZONE		DESCRIP	HONS	вү А	PPRD	REFERENCES	DRG NO	
SECTION	N PRO	CESS & F	PIPING		CLIENT :				50.00.	
	NAME									
	NAME	DATE	UHKD	DATE						
DSGN DRWN	PM	25.04.12	AKJ	25.04.12	PROJECT :			-100 Europh	MECON LIMITED	
								SCALE :		REV
APPROV	ED		(	J. P. JAIN	DATA SHEE	IFUK GLOBE\ (NB≥2")	VALVES	DATA SHEET NO	.: MEC/WINO/05/28/M/001/DS/GV/76	0

	1.	Valve Ma	anufacturer		:		
	2.	Size			: Rating : ANSI 150	#	Design Standard : BS:1873
	3.	Purchase	er's Specifica	ation	: Refer Technical notes for Gate & Glo	obe Valves	
	4.	Design P	ressure		: 19 kg/cm <sup>2</sup> (g) Desi	gn Temperature	: -45°C to + 65°C
	5.	Corrosior	n Allowance		: 1.5mm	Serv	vice :
	6.	End Coni	nections		: Flanged both ends as per ASME B 16.5	i	
					Butt Weld both ends as A-16.25 Flanged one end butt weld other end Socket weld both ends as per ASME B1	16.11	with 100mm pup pieces of A106 Gr. B Sch160
	7.	Flanges (	(where appli	cable)	b) Serrated Smooth (125 t		
	8.	Connect	ing Pipe Sp	ecification	:	,	
	9	Valve Ma	terial Speci	fication :			
	~.		Part		Material	Mate	erial Offered (Equivalent or Superior)
	9.1	Body			ASTM A352 Gr. LCB/ A 350 GR. LF2		
	9.2	Bonnet (E	Bolted)		ASTM A352 Gr. LCB/ A 350 GR. LF2		
	9.3	Stem (Ris	sing) aa Diva/Dall	Turne	SS316 (No casting) / A 350 GR. LF2		
	9.4	DISC(LOO	se Flug/ball	Type)	A 350 GR   F2 with 75 Micron ENP coati	na	
	9.5	Body Sea	at Ring		SS316 + Stellite/ ASTM A352 Gr. LCB /		
			0		A 350 GR. LF2 with 75 Micron ENP coati	ng	
	9.6	Stem Page	cking (Rene	wable	Graphited Braided Asbestos with sacrifi	cal	
		with va	ve open on	stream)	inhibitor & Inconel wire reinforcement		
	9.7	Hand Wh	neel (Rising)		Malleable Iron/ Cast Steel/ Fab. Steel		
	9.8	Bonnet B	Bolts		ASTM A320 Gr.L7		
	9.9	Bonnet N	luts		ASTM A194 Gr.4		
	9.10	Bonnet G	Jaskel		Spiral Would 33 310 + Graion		
	10.	Hvdrosta	tic Test Pres	ssure			
		a) Body			: 32 kg/cm <sup>2</sup> (g)		
		b) Seat			: 23 kg/cm²(g)		
	11.	Pnuemat Air	icTest Press	sure with	: 5.6-7 kg/cm2 (g).		
	12.	Painting	Specification	ns:			
	i)	Surface p	preparation l	by Short B	asting as per grade SA 2 1/2, Swedish Star	ndard SIS-055 9	09.
	, ii) , ,	For abov ( Permiss	e ground ins sible thickne	stallation-T ss in each	hree coats of corrosion resistant paint shall coat shall be within 80 to 120 micron).	be applied with	minimum thickness of 300 micron
	Notes:						
	1.	Valve spe	ecification sl	neet shall I	be read in conjunction with technical notes f	for Gate and Glo	be valves.
	2.	Valve sha	all be desigr	ed for intri	nsically fire safe.Min. Body & Bonnet Thickne	ess as per BS 18	373 .
	ა. ⊿	Ridder of	nall deadum	rite all/ an	12200-1.	lve in the choice	provided for
	<b>+.</b>	Whereve	r bidder ann	ees with M	ECON's data sheet, bidder shall clearly indi	icate "agreed"	
	5	Charpv '\	/ notch test	on each h	eat of base material shall be conducted as	per relevant mat	terial code.
	6.	Hardness	s test shall b	e carried o	out on each heat of base material for all pres	ssure containing	parts of the valve as per relevant
		material of	code.			-	
	7.	Stem pac	cking shall b	e renewab	le with valve open on stream		
	8.	Painting	procedure of	the valve	s snall be as per Manufacturer's Standard.	a diapat-b	
REV NO	9. DATE				iyuro rest reports snaij be furnisned prior t rionis	o dispatch.	
	UNIL	LONE	1	DECONIE		REFERENC	ES DRG. NO.
SECTION	N PRO	CESS &	PIPING		CLIENT :		
	NAME	DATE	CHKD	DATE			
							-)
DSGN	РМ	25.04 12	AK.I	25.04 12	PROJECT :	10019	MECON LIMITED
DRWN						-wo1 Cave	
APPROV	′ED		C	D. P. JAIN		SCALE : DATA SHEE	ET NO.: MEC/WINO/05/28/M/001/DS/GV/77 0
					(NDCZ)	1	

	1.	Valve Ma	nufacturer		:					
	2.	Size			:	Rating :	ANSI <b>300#</b>		Design Standard : BS:1873	
	3.	Purchase	r's Specifica	ation	: Refer Tecl	hnical notes for Ga	ate & Globe Valves			
	4.	Design Pr	essure		: 49 kg/cm <sup>2</sup>	(g)	Design Temperature		: -29°C to + 65°C	
	5.	Corrosion	Allowance		: 1.5mm			Service	:	
	6.	End Conn	nections		: Flanged bot Butt Weld bo Flanged one Socket weld	h ends as per ASM oth ends as A-16.25 e end butt weld othe I both ends as per A	E B 16.5 5 r end SME B16.11			
	7.	Flanges (	where appli	cable)	: a) RF	F RTJ				
					b) Serrated	Smoo	th (125 to 200 AARH)			
	8.	Connecti	ng Pipe Spe	ecification	:					
	9.	Valve Ma	terial Specif	ication :						
			Part			Materia		Materia	Offered (Equivalent or Superi	or)
	9.1	Body			ASTM A 216 G	ir.WCB				
	9.2	Bonnet (B	solted)		ASTM A 216 G	ir.WCB				
	9.3	Stem (Ris	ing)		13% Cr. Steel	(No Casting)				
	9.4	Disc(Loos	se Plug/Ball	Type)	(ASTM A 216 C	Gr. WCB + 13% Cr	Steel Facing) /			
				•••	13% Cr Steel (	Stellited)	<b>~</b> /			
	0.5	Rody Soa	t Ding		ASTM A 216 C	wcB+12% Cr St	ool Ecoing (Stollitod)			
	9.0	bouy Sea	t King		ASTIVIAZIOG	1. WCD+13% C1 30	eer Facing (Steinted)			
	9.6	Stem Pac	king (Renev	wable	Corrosion inhi	ibited die formed f	exible			
		with val	ve open on	stream)	graphite with I	braided anti extrus	ion rings			
	0.7	Hand W/b	ool (Pising)		Malloable Iron	/ Cast Steel/ Eab	Stool			
	9.1					Cast Steel/ Fab.	steel			
	9.8	Bonnet Bo	olts		A 193 Gr. B7					
	9.9	Bonnet N	uts		A194 Gr. 2H					
	0.10	Ponnot C	ookot		Spiral Wound	SS 216 + Crofoil				
	5.10	Donnet O	asket							
	10. 11. 12. i) ii) Notes: 1. 2. 3. 4. 5. 6. 7. 8. 9.	a) Body b) Seat Pnuematii Air Painting S Surface p For above ( Permiss Valve spe Valve sha Testing sh Bidder sha Wherever Charpy 'V welding ei absorbed Hardness cross sec measurem Stem paci Painting p Material T	cTest Press specification reparation h ground ins ible thickne cification sh ll be design all clearly w all clearly w bidder agre ' notch test nds as well energy per test shall b tion shall bé nents repre king shall b roccedure o cest Certifica	sure with as: by Short Bl tallation-T ss in each heet shall I red for intri er BS EN rite all/ an es with M on each h as the bol set of thre e carried co backen for senting the e renewab f the valve ates and F	<ul> <li>76 kg/cm<sup>2</sup></li> <li>57 kg/cm<sup>2</sup></li> <li>57 kg/cm<sup>2</sup></li> <li>5.6 - 7 kg/cm</li> <li>stating as per gr</li> <li>hree coats of cc</li> <li>coat shall be with the coats of cc</li> <li>coat shall be with the coats of the coats</li></ul>	(g) (g) n2 (g). ade SA 2 1/2, Swee prosion resistant pa- tithin 80 to 120 micro nction with technica a. Min. Body & Bonr ed QAP, this specific nst each part/ mater neet, bidder shall cle erial shall be condu per ASTM A370. Th all be 27 J with an in t of base material for d the maximum har is. en on stream . r Manufacturer's Starts shall be furnishe	tish Standard SIS-055 909 int shall be applied with m on). I notes for Gate and Globe let Thickness as per BS 18 ration and other relevant s rial of valve in the space p aarly indicate "agreed". cted for all pressure contain te test shall be conducted dividual minimum per spe- r all pressure containing p dness shall not exceed 24 andard. ed prior to dispatch.	I. inimum thickne inimum thickne i73 . tandards. rovided for . ining parts such at 0°C. The min cimen of 22 J. arts of the valv 8 HV10 based	ss of 300 micron n as body, end flange, nimum average e. A full thickness on minimum four	
DEV NO	9.		est Certifica	DESCRIPTION	iyuro Test Repo	rts snall be furnishe	eu prior to dispatch.			
KEV. NO.	DAIE	ZUNE		DESCRIP	HUNS			REFERENCES	DRG. NO.	
SECTION		CESS & F		DATE	CLIENT :					
	INAME	DAIE	UNKU	DATE	1					
				1				मेकॉन /		
DSGN	PM	25.04.12	AKJ	25.04.12	PROJECT :			PO BOOT DONA	MECON LIMITED	
DRWN				<u> </u>	1					
			1	1				000	L	
APPROV	ED		(	d. p. jain	DATA SHEE	T FOR GLOBE \ (NB≥2")	ALVES	DATA SHEET NC	.: MEC/WINO/05/28/M/001/DS/GV/78	

	1.	Valve Ma	nufacturer		:				
	2.	Size			: Rating : A	NSI <b>300#</b>		Design Standard : BS:1873	
	3.	Purchase	er's Specifica	ation	: Refer Technical notes for Gat	e & Globe V	/alves		
	4.	Design P	ressure		: 49 kg/cm²(g)	Design Te	emperature	: -45°C to + 65°C	
	5.	Corrosior	n Allowance		: 1.5mm		Service	:	
	6.	End Conr	nections		: Flanged both ends as per ASME	B 16.5		7	
	7.	Flanges (	(where appli	cable)	Butt Weld both ends as A-16.25 Flanged one end butt weld other Socket weld both ends as per AS : a) RFFF RTJ	end SME B16.11		with 100mm pup pieces of A106 Gr. B Sch160	
					b) Serrated Smooth			_	
	Q	Connact	ing Dino Sn	ocification		1 (123 to 200			
	0.			fication					
	9.		Part	lication .	Material		Material	Offered (Equivalent or Superio	or)
	9.1	Body			ASTM A352 Gr. LCB/ A 350 GR 1 F	2	materia		··· /
	9.2	Bonnet (F	Solted)		ASTM A352 Gr   CB/ A 350 GR   E				
	0.2	Stom (Pic	sing)		SS216 (No casting) / A 250 GR LE	2	1		
	9.3	Sterri (Ris	sing)	<b>-</b> )	33310 (NO Casting) / A 350 GR. LF	2			
	9.4	Disc(Loos	se Plug/Ball	Type)	SS316 / ASTM A352 Gr. LCB /				
					A 350 GR. LF2 + Stellited				
	9.5	Body Sea	at Ring		SS316 / ASTM A352 Gr. LCB /				
			0		A 350 GR   F2 + Stellited		1		
	0.6	Stom Do	oking (Pono	wahla	Correction inhibited die formed fle	vible			
	9.0	StelliFat	King (Rene	wable	Corrosion infinibiled die formed fie	Sine			
	with valve open on stream				graphite with braided anti extrusion	on rings			
	9.7 Hand Wheel (Rising)				Malleable Iron/ Cast Steel/ Fab. St	eel			
	9.7 [Hand Wheel (Rising)] 9.8 Bonnet Bolts				ASTM A320 Gr I 7				
	0.0	Bonnot N	luto						
	9.9	Burnet N							
	9.10	Bonnet G	asket		Spiral Wound SS 316 + Gratoli				
	10. 11.	Hydrostat a) Body b) Seat Pnuemati Air	tic Test Pres icTest Press	ssure sure with	: 76 kg/cm²(g) : 57 kg/cm²(g) : 5.6-7 kg/cm2 (g).				
	12	Painting 9	Specification	ne.					
	12.	Surface	reportion	hy Short DI	acting as par grade SA 2 1/2 Swadi	oh Standard	SIS 055 000		
	ii)	For above ( Permiss	e ground ins	stallation-T	hree coats of corrosion resistant pair coat shall be within 80 to 120 micror	nt shall be ap	oplied with mini	mum thickness of 300 micron	
1	Notes:								
	1.	Valve spe	ecification sl	heet shall t	be read in conjunction with technical	notes for Ga	ate and Globe v	valves.	
	2.	Valve sha	all be desigr	ned for intri	nsically fire safe. Min. Body & Bonnet	t Thickness a	s per BS 1873 .		
	3.	Testing s	hall be as n	er BS FN 1	2266-1 approved QAP this specifica	ation and oth	er relevant sta	ndards.	
	4	Biddereb	all clearly w	rite all/ any	deviation against each part/ materi	al of value in	the snace prov	vided for	
		Wherevie	r biddor ocr	oog with M	ECON's data shoet hiddor shell also	arly indicate	"agrood"		
		"oh					ayreeu		
	5.	Unarpy 'V	/ notch test	on each h	eat of pase material shall be conduc	ted as per re	eevant material	coue.	
	6.	Hardness	s test shall b	e carried c	ut on each heat of base material for	all pressure	containing par	ts of the valve as per relevant	
		material of	code.						
	7.	Stem pac	king shall b	e renewab	e with valve open on stream .				
	8	Painting	procedure o	f the valve	s shall be as per Manufacturer's Sta	ndard.			
	0	Material -	Leet Cortifie	atec and U	vdro Teet Penorte shall be furnishes	I prior to diar	atch		
	J.	wateria					Jaion.		
REV. NO.	DATE	ZONE		DESCRIPT	IONS BY APP	אט	1		
							REFERENCES	DRG. NO.	
SECTION	PRO	CESS &	PIPING		CLIENT :				
				DATE			$\frown$		
	NAIVIE	DAIE		DAIE					
DOON	D14	05.04.40	ALC	05.04.40			and the second		
DOGN	PM	25.04.12	AKJ	25.04.12	FNUJEUT.		SCOT CONS		
DRWN				1					
							SCALE :		REV
APPROV	'ED		(	D. P. JAIN	DATA SHEET FOR GLOBE VA (NB≥ 2")	LVES	DATA SHEET NO	D.: MEC/WINO/05/28/M/001/DS/GV/79	0

	1.	Valve Mar	nufacturer		:					
	2.	Size			:	Rating : A	ANSI <b>600#</b>		Design Standard : BS:1873	
	3.	Purchaser	's Specifica	ation	: Refer Tech	inical notes for Ga	ate & Globe Valves			
	4.	Design Pro	essure		: 92 kg/cm <sup>2</sup> (	g)	Design Temperature		: -29°C to + 65°C	
	5.	Corrosion	Allowance		: 1.5mm			Service	:	
	6	End Conn	ections		: Flanged both	ends as per ASM	= B 16 5		7	
	0.				Butt Weld bo Flanged one Socket weld	oth ends as A-16.25 end butt weld othe both ends as per A	r end SME B16.11			
	7.	Flanges (v	where appli	cable)	:a) RF	FF RTJ				
					b) Serrated	Smoot	th (125 to 200 AARH)			
	8.	Connectir	ng Pipe Spe	ecification	:					
	9.	Valve Mat	erial Specif	ication :						
			Part			Material		Material	Offered (Equivalent or Superi	lor)
	9.1	Body			ASTM A 216 G	r.WCB				
	9.2	Bonnet (B	olted)		ASTM A 216 Gi	r.WCB				
	9.3	Stem (Risi	ing)		13% Cr. Steel (	No Casting)				
	9.4	Disc(Loos	e Plug/Ball	Type)	(ASTM A 216 G	ir. WCB + 13% Cr 9	Steel Facing) /			
					13% Cr Steel (S	Stellited)				
	9.5	Body Seat	t Ring		ASTM A 216 Gr	r. WCB+13% Cr St	eel Facing (Stellited)			
	9.6	Stem Pack	king (Renev	vable	Corrosion inhi	bited die formed fl	lexible			
		with valv	/e open on	stream)	graphite with b	raided anti extrus	ion rings			
	9.7	Hand Whe	el (Rising)	,	Malleable Iron/	Cast Steel/ Fab. S	Steel			
	9.8	Bonnet Bo	olts		A 193 Gr B7					
	0.0	Bonnet Nu	ite		A104 Gr 2H					
	0.10	Bonnet Ge	nokot		Spiral Wound 9	SS 216 + Crofoil				
	9.10	Donnet Ga	askei			55 510 + Glaion				
	11. 12. i) ii) <b>Notes:</b> 1. 2. 3. 4. 5. 6.	a) Body b) Seat Pnuematic Air Painting S Surface pr For above ( Permissi Valve spec Valve shal Testing sh Bidder sha Wherever Charpy 'V' welding er absorbed Hardness cross sect	cTest Press repecification ground ins ble thicknes cification sh Il be design hall be as pe all clearly w bidder agre notch test hds as well energy per test shall be	ure with by Short Bl tallation-T ss in each eet shall I ed for intri er BS EN rite all/ an on each h as the bol set of thre e carried c taken for	<ul> <li>157 kg/cm</li> <li>114 kg/cm</li> <li>114 kg/cm</li> <li>5.6 - 7 kg/cm</li> <li>asting as per gra hree coats of coil coat shall be with</li> <li>be read in conjurnisically fire safe 12266-1, approve y deviation again</li> <li>ECON's data sheat of base matering ting material as provide the speciment shall but on each heat this purpose and</li> </ul>	<sup>2</sup> (g) <sup>2</sup> (g) h2 (g). h2 (g). h2 (g). h3 (g). h4 (g). h5 (g	lish Standard SIS-055 909 int shall be applied with m n). I notes for Gate and Globe tet Thickness as per BS 16 eation and other relevant s ial of valve in the space pr arly indicate "agreed". cted for all pressure containe test shall be conducted dividual minimum per spe- r all pressure containing p dness shall not exceed 24	I. inimum thickne 373 . tandards. rovided for . at 0°C. The min cimen of 22 J. arts of the valv 8 HV10 based	n as body, end flange, nimum average e. A full thickness on minimum four	
	7. 8.	measurem Stem pack Painting p	tents represent ting shall be rocedure of	senting the e renewab the valve	e entire thickness le with valve ope s shall be as per	s. en on stream Manufacturer's Sta	andard.			
REV NO	9.	IVIaterial T	est Certifica	DESCRIPT	Iyaro Test Repor	ts shall be furnishe	a prior to dispatch.			
NEV. NU.	DATE	ZUNE		DESCRIP			עאד	REFERENCES	DRG. NO.	
SECTION	PRO	CESS & P	PING		CLIENT :				2	
	NAME	DATE	CHKD	DATE						
DSGN	PM_	25.04.12	AKJ	25.04.12	PROJECT :				MECON LIMITED	
DRWN										
APPROV	ED	·	C	). P. JAIN	DATA SHEET	FOR GLOBE V	′ALVES (NB≥2")	SCALE : DATA SHEET NC	.: MEC/WINO/05/28/M/001/DS/GV/80	REV 0

			DATA SHEET FOR GLOBE VALVES	
	1.	Valve Manufacturer	:	
	2.	Size	: Rating : ANSI 600#	Design Standard : BS:1873
	3.	Purchaser's Specification	: Refer Technical notes for Gate & Globe Valves	
	4.	Design Pressure	: 92 kg/cm <sup>2</sup> (g) Design Temp	erature : -45°C to + 65°C
	5.	Corrosion Allowance	: 1.5mm	Service :
	6.	End Connections	: Flanged both ends as per ASME B 16.5	
			Butt Weld both ends Flanged one end butt weld other end	
	7.	Flanges (where applicable)	: a) RF FF RTJ	
			b) Serrated Smooth (125 to 200 AARH	)
	8.	Pipe Specification :		
	9.	Valve Material Specification (or Part	Equivalent/ Superior) : Material	Material Offered (Equivalent or Superior)
	Q 1	Fait Body	ASTM A352 Gr LCB/ A 350 GR LE2	waterial Offered (Equivalent of Superior)
	9.2	Bonnet (Bolted)	ASTM A352 Gr. LCB/ A 350 GR. LF2	
	9.3	Stem (Rising)	SS316 (No casting) / A 350 GR. LF2	
	9.4	Disc(Loose Plug/Ball Type)	SS316/ ASTM A352 Gr. LCB /	
	0		A 350 GR. LF2 + Stellited	
	9.5	Body Seat Ring	SS316/ ASTM A352 Gr   CB /	
	0.0	Body Court ang	A 350 GR   F2 + Stellited	
	9.6	Stem Packing (Renewable	Corrosion inhibited die formed flexible	
	9.0	with value onen on stream)	corrosion initialited die formed nexible	
	0.7	with valve open on stream)	graphile with braided anti extrusion rings	
	9.7	Hand Wheel (Rising)	Malleable Iron/ Cast Steel/ Fab. Steel	
	9.8	Bonnet Bolts	ASTM A320 Gr.L7	
	9.9	Bonnet Nuts	ASTM A194 Gr.4	
	9.10	Bonnet Gasket	Spiral Wound SS 316 with Grafoil	
	10.	Hydrostatic Test Pressure		
		a) Body	: 157 kg/cm <sup>2</sup> (g)	
		b) Seat	: 114 kg/cm <sup>2</sup> (g)	
		,		
	11	Test Pressure with Air	$5.6 - 7.0 \text{ kg/cm}^2$ (g)	
	12	Painting Specifications:		
	i)	Surface preparation by Short B	asting as per grade SA 2 1/2 Swedish Standard SIS-05	5 909.
	ii)	For above ground installation_T	hree coats of corrosion resistant naint shall be annied w	ith minimum thickness of 300 micron
	"/ .	(Permissible thickness in each	coat shall be within 80 to 120 micron)	
			esat chai be within so to ree moreny.	
	Notes			
	1	Valve specification sheet shall	be read in conjunction with technical notes for Gate and (	Globe valves
	ı. 2	Valve shall be designed for intr	nsically fire safe. Min. Body & Ponnet Thickness of per P	S 1873
	<u>د</u> .	Testing shall be as nor BS EN	12266-1 approved OAP this specification and other relay	ant standards
	J. ⊿	Ridder shall clearly write all an	deviation against each part/material of value in the area	ant stanualus.
	4.	Whorever hidder arread with 1	y deviation against each part/ material of valve in the spa	
	· · · ·	Charpy V/ notabilitiest an and b	act of base meterial aball be conducted as non-	notorial aada
	ວ. ເ	Unarpy v notion test on each n	eat of base material snall be conducted as per relevant r	natenal code.
	ю. 7	naruness test shall be carried (	our as per relevant material code.	
	1.	Stem packing shall be renewab	ie with valve open on stream .	
	8. C	Painting procedure of the valve	s snall be as per Manufacturer's Standard.	
	9.	waterial Test Certificates and F	iyaro Test Reports shall be turnished prior to dispatch.	
REV. NO.	DATE	ZONE DESCRIP	HONS BY APPRD	
OF OTION				REFERENCES DRG. NO.
SECTION	N PRO			
				( det
	NAME	DATE ONNO DATE		मेकॉन 
DSGN	PM	25.04.12 AKJ 25.04.12	PROJECT :	MECON LIMITED
DRWN				
				SCALE : REV
APPROV	'ED	O.P. JAIN	DATA SHEET FOR GLOBE VALVES	DATA SHEET NO.: MEC/WINO/05/28/M/001/DS/GV/81 0
			(NB≥2'')	
P				·

	1.	Valve Mar	nufacturer		:			
	2.	Size			: Rating : ANSI 800#		Design Standard	: ISO 15761 /
	3.	Purchase	r's Specifica	ation	: Refer Technical notes for Gate & Globe	Valves		63.3332
	4.	Design Pr	essure		: Design T	emperature	: -29°C to + 65°C	
	5.	Corrosion	Allowance		: 1.5mm	Service	:	
	6.	End Conn	ections		<ul> <li>Flanged both ends as per ASME B 16.5 Butt Weld both ends as A-16.25 Flanged one end butt weld other end Socket weld both ends as per ASME B16.11</li> </ul>		→ with 100mm pt	ip pieces of
	7.	Flanges (v	where appli	cable)	: a) RF FF RTJ	_		1100
				10 11	b) Serrated Smooth (125 to 20	00 AARH)		
	8.	Connectii	ng Pipe Spe		: N.A.			
	9.	Valve Mat	terial Specif	ication :				
[			Part		Material	Materia	I Offered (Equivale	ent or Superior)
l [	9.1	Body			ASTM A 105			
	9.2	Bonnet (B	olted)		ASTM A 105	ſ		
	93	Stem (Ris	ina)		13% Cr. Steel (No Casting)			
∥ ⊦	0.1	Disc(Loco			SS 316 + Stellite	1		
	3.4	DISC(LOUS		Type)				
	9.5	Body Sea	t Ring		SS 316 + Stellite			
	9.6	Stem Pac	king (Renev	wable	Corrosion inhibited die formed flexible			
		with valv	ve open on	stream)	graphite with braided anti extrusion rings			
	9.7	Hand Whe	eel (Rising)		Malleable Iron/ Cast Steel/ Fab. Steel			
	9.8	Bonnet Bo	olts		A 193 Gr. B7			
	99	Bonnet Ni	uts		A194 Gr 2H			
	9.9 Bonnet Gasket				Spirel Wound SS 246 + Crofoil	-		
	9.10	Bonnet Ga	askel		Spiral wound 55 516 + Gratoli			
	11. 12. i) ii) <b>Notes:</b> 1. 2. 3. 4. 5. 6. 7.	a) Body b) Seat Pnuematic Air Painting S Surface p For above ( Permissi Valve spe Valve sha Testing sh Bidder sha Wherever Charpy 'V welding ei absorbed Hardness cross sect measuren Stem pacl	cTest Press specification reparation b ground ins ible thicknes cification sh Il be design hall be design hall clearly w bidder agre notch test notch test energy per test shall b tion shall be	sure with s: by Short BI tallation-TI ss in each ed for intri er BS EN rite all/ any ees with M on each h as the bolt set of thre e carried o taken for senting the e renewab	<ul> <li>210 kg/cm<sup>2</sup>(g)</li> <li>155 kg/cm<sup>2</sup>(g)</li> <li>5.6-7 kg/cm<sup>2</sup> (g).</li> <li>asting as per grade SA 2 1/2, Swedish Standard three coats of corrosion resistant paint shall be a coat shall be within 80 to 120 micron).</li> <li>we read in conjunction with technical notes for Gansically fire safe. Dimensions including thickness 2266-1, approved QAP, this specification and other deviation against each part/ material of valve in ECON's data sheet, bidder shall clearly indicate ast of base material shall be conducted for all pring material as per ASTM A370. The test shall be specimen shall be 27 J with an individual mining ut on each heat of base material for all pressure this purpose and the maximum hardness shall nentire thickness.</li> <li>with valve open on stream .</li> </ul>	I SIS-055 909. pplied with mini ate and Globe v as per ISO 157/ ner relevant stan the space prov "agreed". essure containi be conducted at mum per specin e containing par iot exceed 248	imum thickness of 3 valves. 61 . ndards. vided for . ng parts such as bo 0°C. The minimum nen of 22 J. ts of the valve. A ful HV10 based on min	00 micron dy, end flange, average I thickness imum four
	8.	Painting p	procedure of	the valves	s shall be as per Manufacturer's Standard.			
	9.	Material T	est Certifica	ates and H	ydro Test Reports shall be furnished prior to dis	patch.		
REV. NO.	DATE	ZONE		DESCRIPT	IONS BY APPRD	-		
	. —					REFERENCES		DRG. NO.
SECTION		CESS & G			CLIENT ·			
			1110		VEIEITT .			
DSGN DRWN	NAME PM	DATE 25.04.12	CHKD AKJ	DATE 25.04.12	PROJECT :	истрана 10001 сострана	MECON	
APPROV	ED		0	). P. JAIN	DATA SHEET FOR GLOBE VALVES (NB<2")	SCALE : DATA SHEET N	O.: MEC/WINO/05/28/M/	001/DS/GV/82 0

1.       Valve Manufacturer       ::         2.       Size       :       Rating: ANSI 800#       Design         3.       Purchaser's Specification       :       Refer Technical notes for Gate & Globe Valves         4.       Design Pressure       :       Design Temperature       :       :         5.       Corrosion Allowance       :       1.5mm       Service :       :         6.       End Connections       :       Flanged both ends as per ASME B 16.5       :		
2. Size       ::       Rating : ANSI 800#       Design         3. Purchaser's Specification       ::       Refer Technical notes for Gate & Globe Valves         4. Design Pressure       ::       Design Temporature       ::         5. Corrosion Allowance       ::       1.5mm       Service ::         6. End Connections       ::       Flanged one endo but weld other end       Design Temporature       ::         7. Flanges (where applicable)       ::       N.F       Imaged one endo but weld other end       Design Temporature       ::         8. Connecting Pipe Specification       :       N.A.       :       Smooth (125 to 200 AARH)       Design Temporature         9.1       Body Material Specification       :       N.A.       :       Material Office         9.2       Stem (Keing)       SS316 (ASTM A350 Gr. LF2       Imaged one endo textuel other end       Stem facture         9.3       Stem (Keing)       SS316 (ASTM A350 Gr. LF2       Stemite       Imaged one endo textuel other end         9.3       Stem (Keing)       SS316 (ASTM A350 Gr. LF2       Stemite       Imaged one endo textuel other         9.4       Body Stem Packung (Rerevable       Corrosion inhibited die formed factble       Imaged one endo textuel other         9.5       Body Stem Packung (Rerevable		
3.       Purchaser's Specification       ::       Refer Technical notes for Gate & Globe Valves         4.       Design Pressure       ::       Design Temperature       ::       ::         5.       Corrosion Allowance       ::       1.5mm       Service ::       ::         6.       End Connections       ::       Flanged both ends as per ASME B 16.5.       :       :       :       :       :       :       :       ::       <	Design Standard : ISO 15761	1
4.       Design Tensure       : 44         5.       Corrosion Allowance       : 15mm       Service :         6.       End Connections       : Flanged both ends as per ASME B 16.5       Design Tensure       Design Tensure         6.       End Connections       :: Flanged both ends as per ASME B 16.11       Design Tensure       Design Tensure       Design Tensure         7.       Flanges (where applicable)       : a) RF       FF       RTJ       Design Tensure       Tensure         8.       Connecting Pipe Specification :       N.A.       N.A       Endotation in the tensure       Material Offer         9.       Valve Material Specification :       N.A.       Endotation in the tensure       Material Offer         9.       Valve Material Specification :       N.A.       Endotation in the tensure       Endotation in the tensure         9.       Valve Material Specification :       N.A.       Endotation in the tensure       Endotation in the tensure       Endotation in tensure         9.       Valve Material Specification :       N.A.       Endotation in tensure       Endotation in tensure         9.       Stem (Reinwald       Corrosion inhibited dia formed flexible       Material Offer         9.       Stem (Reinwald       Corrosion inhibited dis formed flexible       Material Offer	BS:5352	
5.       Corrosion Allowance       : 1.5mm       Service :         6.       End Connections       :: Flanged both ends as per ASME B 16.5 Butt Weld both ends as per ASME B 16.11	: -45°C to + 65°C	
6.       End Connections       :       Flanged one end butt weld other end Socket weld both ends as per ASME B 16.5 Flanged one end butt weld other end Socket weld both ends as per ASME B 16.11	9 :	
0.       End both Heads as a Pri Acade         0.       End both Heads as a Pri Acade         0.       End both Heads as per ASME B16.11         0.       Socket weld both ends as per ASME B16.11         0.       Socket weld both ends as per ASME B16.11         0.       State Meld both ends as per ASME B16.11         0.       State Meld both ends as per ASME B16.11         0.       State Meld both ends as per ASME B16.11         0.       State Meld both ends as per ASME B16.11         0.       State Meld both ends as per ASME B16.11         0.       State Meld both ends as per ASME B16.11         0.       State Meld both ends as per ASME B16.11         0.       State Meld both ends as per ASME B16.11         0.       State Meld both ends as per ASME B16.11         0.       State ASME B16.11         0.       State ASME B16.11         0.       State ASME B16.11         0.       State ASME B16.11         0.       State ASME B16.11         0.       State ASME B16.11         0.       State ASME B16.11         0.       State ASME B16.11         0.       State ASME B16.11         0.       State ASME B16.11         0.       State ASME B16.11	_	
7.       Flanges (where applicable)       : a)       RF       FF       RTJ         b)       Serrated       Smooth (125 to 200 AARH)       .         8.       Connecting Pipe Specification :       N.A.         9.       Valve Material Specification :       Material       Material Offic         9.2       Bornet (Bolted)       ASTM A350 Gr. LF2       .         9.3       Stem (Rising)       SS316 (No Casting) (ASTM A350 Gr. LF2 + Stellite       .         9.4       Disc(Loose PlugBall Type)       SS316 (ASTM A350 Gr. LF2 + Stellite       .         9.5       Stem Packing (Renewable Corrosion inhibited die formed flaxible       .       .         9.5       Stem Packing (Renewable Corrosion inhibited die formed flaxible       .       .         9.4       Bonnet Bolts       ASTM A320 Gr. LF2 + Stellite       .       .         9.5       Bonnet Rolts       ASTM A320 Gr. LF2 + Stellite       .       .         9.8       Bonnet Bolts       ASTM A320 Gr. LF2 + Stellite       .       .         9.10       Bonnet Bolts       ASTM A320 Gr. LF2 + Stellite       .       .         9.10       Bonnet Bolts       ASTM A320 Gr. LF2 + Stellite       .       .         9.8       Bonnet Bolts       ASTM A320 Gr. LF2	✓ with 100mm pup pieces of A106 Gr. B Sch160	
b) Serrated       Smooth (125 to 200 AARH)         8. Connecting Pipe Specification :       N.A.         9. Valve Material Specification :       Material       Material Offe         9.2 Bonnet (Bolted)       ASTM A350 Gr. LF2		
8. Connecting Pipe Specification : N. A.     9. Valve Material Specification :     1     1     1     1     1     0 doy     1     1     8 dot     1     1     8 dot     1		
9.       Valve Material Specification :         Part       Material       Material       Material Offe         9.1       Body       ASTM A350 Gr. LF2          9.2       Bonnet (Bolted)       ASTM A350 Gr. LF2          9.4       Disc(Loose Plug/Ball Type)       SS316 (No Casting)/ASTM A350 Gr. LF2 + Stellite          9.5       Body Seat Ring       SS316/ASTM A350 Gr. LF2 + Stellite          9.5       Body Seat Ring       SS316/ASTM A350 Gr. LF2 + Stellite          9.5       Bonnet Mults       Corrosion inhibited die formed flexible          9.7       Hand Wheel (Rising)       Malleable Iron/ Cast Steel Fab. Steel          9.8       Bonnet Nuts       ASTM A194 Gr.4           9.9       Bonnet Nuts       ASTM A194 Gr.4           9.10       Bonnet Nuts       ASTM A194 Gr.4           9.10       Bonnet Cast Pressure        10       Hydrostatic Test Pressure with : 5.6-7 kg/cm2 (g).          Air        Painting Specifications:        Surface preparation by Short Blasting as per grade SA 2 1/2, Swedish Standard SIS-055 909.         <		
Part     Material     Material     Material     Material       9.1     Body     ASTM A350 Gr. LF2		
9.1       Body       ASTM A350 Gr. LF2         9.3       Stem (Rsing)       SS316 (No Casting) /ASTM A350 Gr. LF2         9.4       Disc(Loces Plug/Ball Type)       SS316 (ASTM A350 Gr. LF2 + Stellite         9.5       Body Seat Ring       SS316 /ASTM A350 Gr. LF2 + Stellite         9.5       Body Seat Ring       SS316 /ASTM A350 Gr. LF2 + Stellite         9.6       Stem Packing (Renewable       Corrosion inhibited die formed flexible         with valve open on stream)       graphite with braided anti extrusion rings         9.7       Hand Wheel (Rsing)       Malleable Iron' Cast Steel / Fab. Steel         9.8       Bonnet Bolts       ASTM A320 Gr.L 7         9.9       Bonnet Iost       ASTM A320 Gr.L 6         9.10       Hydrostatic Test Pressure       Bong         9.3       Bornet Iost       Spiral Wound SS 316 + Grafoil         10.       Hydrostatic Test Pressure       Bornet Iost         11.       PnuematicTest Pressure with       5.6-7 kg/cm2 (g).         Air       Surface preparation by Short Blasting as per grade SA 2 1/2, Swedish Standard SIS-055 909.         10.       For above ground Installation-Three coats of corrosion resistant paint shall be applied with minimum thicl (Permissible thickness in each coat shall be within 80 to 120 micron).          Nalve spacifications:       <	al Offered (Equivalent or Superi	or)
9.2       Bonnet (Boiled)       AS IM A350 Gr. LF2         9.4       Disc(Loose Plug/Ball Type)       SS316 (ASTM A350 Gr. LF2 + Stellite         9.5       Body Seat Ring       SS316 (ASTM A350 Gr. LF2 + Stellite         9.6       Stem Packing (Renewable       Corrosion inhibited die formed flexible         9.6       Stem Packing (Renewable       Corrosion inhibited die formed flexible         9.7       Hand Wheel (Rising)       Malleable Iron' Cast Steel/ Fab. Steel         9.8       Bonnet Bolts       ASTM A320 Gr.L7         9.8       Bonnet Rots       ASTM A194 Gr.4         9.10       Bonnet Casket       Spiral Wound SS 316 + Grafoil         10.       Hydrostatic Test Pressure a) Body       : 210 kg/cm²(g)         b) Seat       : 155 kg/cm²(g)         b) Seat       : 155 kg/cm²(g)         i) Surface preparation by Short Blasting as per grade SA 2 1/2, Swedish Standard SIS-055 909.         ii) For above ground installation-Three coats of corrosion resistant paint shall be applied with minimum thick (Permissible thickness in each coat shall be within 80 to 120 micron).         Notes:         1.       Valve specifications sheet shall be read in conjunction with technical notes for Gate and Globe valves.         2.       Valve shall be designed for intrinsically fire safe. Dimensions including thickness as per ISO 15761.		
9.3       Stem (Rising)       S3316 (No Casting) /ASTM A350 Gr. LF2         9.4       Disc(Loose Plug/Ball Type)       S3316 /ASTM A350 Gr. LF2 + Stellite         9.5       Body Seat Ring       S3316 /ASTM A350 Gr. LF2 + Stellite         9.6       Stem Packing (Renewable)       Corrosion inhibited dif formed flexible         with valve open on stream)       graphite with braided anti extrusion rings         9.7       Hand Wheel (Rising)       Malleable Iron' Cast Steel/ Fab. Steel         9.8       Bonnet Bolts       ASTM A194 Gr.4         9.9       Bonnet Nuts       ASTM A194 Gr.4         9.10       Bonnet Gasket       Spiral Wound SS 316 + Grafoil         10.       Hydrostatic Test Pressure       a)         a) Body       :       210 kg/cm²(g)         b) Seat       :       155 kg/cm²(g)         11.       PruematicTest Pressure with       :       5.6-7 kg/cm2 (g).         Air       :       12.       Painting Specifications:       :         i) Surface preparation by Short Blasting as per grade SA 2 1/2, Swedish Standard SIS-055 909.       :       :         ii)       For above ground installation-Three coats of corosion resistant paint shall be applied with minimum thick (Permissible thickness in each coat shall be within 80 to 120 micron).          Notes:       <		
9.4       Disc(Lose Plug/Ball Type)       \$\$316 /ASTM A350 Gr. LF2 + Stellite         9.6       Body Sear Ring       \$\$316 /ASTM A350 Gr. LF2 + Stellite         9.6       Stem Packing (Renewable       Corrosion inhibited die formed flexible         9.7       Hand Wheel (Rising)       Malleable Iron Cast Steel/Fab. Steel         9.8       Bonnet Bolts       ASTM A320 Gr.L7         9.8       Bonnet Bolts       ASTM A320 Gr.L7         9.8       Bonnet Roits       ASTM M194 Gr.4         9.10       Bonnet Gasket       Spiral Wound SS 316 + Grafoil         10.       Hydrostatic Test Pressure a) Body       : 210 kg/cm²(g)         b) Seat       : 155 kg/cm²(g)         11.       PnuematicTest Pressure with :       5.6-7 kg/cm²(g)         12.       Painting Specifications:       In         13.       Starface preparation by Short Blasting as per grade SA 2 1/2, Swedish Standard SIS-055 909.         14.       For above ground installation-Three coats of corrosion resistant paint shall be applied with minimum thick (Permissible thickness in each coat shall be within 80 to 120 micron).         Notes:         1.       Valve specification sheet shall be read in conjunction with technical notes for Gate and Globe valves.         2.       Valve shall be designed for intrinsically fire safe. Dimensions including thickness as per ISO 15761. </td <td></td> <td></td>		
9.5       Body Seat Ring       \$\$316/ASTM A350 Gr. LF2 + Stellite         9.6       Stem Packing (Renewable       Corrosion inhibited die formed flexible         9.7       Hand Wheel (Rising)       Maileable Iron/ Cast Steel/ Fab. Steel         9.8       Bonnet Bolts       ASTM A320 Gr. L7         9.9       Bonnet Nuts       ASTM A320 Gr. L7         9.10       Bonnet Gasket       Spiral Wound S\$ 316 + Grafoil         10.       Hydrostatic Test Pressure       a) Body       : 210 kg/cm²(g)         b) Seat       : 155 kg/cm²(g)         11.       PnuematicTest Pressure with       : 5.6-7 kg/cm2 (g).         Air       :       15 kg/cm²(g)         i) Surface preparation by Short Blasting as per grade SA 2 1/2, Swedish Standard SIS-055 909.       ii)         For above ground installation-Three coats of corrosion resistant paint shall be applied with minimum thick (Permissible thickness in each coat shall be within 80 to 120 micron).         Notes:       1.       Valve specification sheet shall be read in conjunction with technical notes for Gate and Globe valves,         2.       Valve shall be designed for intrinsically fire safe. Dimensions including thickness as per ISO 15761.         3.       Testing shall be as per BS EN 12266-1, approved QAP, this specification and other relevant standards.         4.       Bidder shall clearly write all and tor abg material shall be co		
9.6       Stem Packing (Renewable       Corrosion inhibited die formed flexible         9.7       Hand Wheel (Rsing)       Malleable Iron/ Cast Steel / Fab. Steel         9.8       Bonnet Bolts       ASTM A320 Gr.L7         9.9       Bonnet Bolts       ASTM A320 Gr.L7         9.10       Bonnet Gasket       Spiral Wound SS 316 + Grafoil         10.       Hydrostatic Test Pressure       a) Body       : 210 kg/cm²(g)         b) Seat       : 155 kg/cm²(g)         11.       PnuematicTest Pressure with       : 5.6-7 kg/cm2 (g).         Air       12.       Painting Specifications:         i)       Surface preparation by Short Blasting as per grade SA 2 1/2, Swedish Standard SIS-055 909.         ii)       For above ground installation-Three coats of corosion resistant paint shall be applied with minimum thicl (Permissible thickness in each coat shall be within 80 to 120 micron).         Notes:       1.       Valve specification sheet shall be read in conjunction with technical notes for Gate and Globe valves.         2.       Valve shall be designed for intrinsically fire safe. Dimensions including thickness as per ISO 15761.         3.       Testing shall be as per BS EN 12266-1.approved QAP, this specification and other relevant standards.         4.       Bidder shall clearly write all any deviation against each part material toralice and they are safe of the valves shall be conducted as per relevant materia		
with valve open on stream)       graphite with braided anti extrusion rings         9.7       Hand Wheel (Rising)       Maileable Iron/Cast Steel/Fab. Steel         9.8       Bonnet Nuts       ASTM A320 Gr.L7         9.9       Bonnet Nuts       ASTM A194 Gr.4         9.10       Bonnet Gasket       Spiral Wound SS 316 + Grafoil         10.       Hydrostatic Test Pressure       a) Body       : 210 kg/cm²(g)         b) Seat       : 155 kg/cm²(g)         11.       PnuematicTest Pressure with       : 5.6-7 kg/cm²(g)         i) Surface preparation by Short Blasting as per grade SA 2 1/2, Swedish Standard SIS-055 909.       ii)         ii) Surface preparation by Short Blasting as per grade SA 2 1/2, Swedish Standard SIS-055 909.       iii)         ii) Surface preparation by Short Blasting as per grade SA 2 1/2, Swedish Standard SIS-055 909.       iii)         iii) Surface preparation by Short Blasting as per grade SA 2 1/2, Swedish Standard SIS-055 909.       iii)         iii) Surface preparation by Short Blasting as per grade SA 2 1/2, Swedish Standard SIS-055 909.       iii)         iii       Valve specifications:       iii         10.       Valve specification sheet shall be read in conjunction with technical notes for Gate and Globe valves.         2.       Valve shall be designed for intrinsically fire safe. Dimensions including thickness as per ISO 15761. <t< td=""><td></td><td></td></t<>		
9.7       Hand Wheel (Rising)       Malleable Iron/ Cast Steel/ Fab. Steel         9.8       Bonnet Bolts       ASTM A320 Gr.L7         9.9       Bonnet Nuts       ASTM A194 Gr.4         9.10       Bonnet Gasket       Spiral Wound SS 316 + Grafoil         10.       Hydrostatic Test Pressure a) Body       : 210 kg/cm²(g)         b) Seat       : 155 kg/cm²(g)         11.       PnuematicTest Pressure with :       5.6-7 kg/cm2 (g). Air         12.       Painting Specifications:         i) Surface preparation by Short Blasting as per grade SA 2 1/2, Swedish Standard SIS-055 909.         ii) For above ground installation-Three coats of corrosion resistant paint shall be applied with minimum thick (Permissible thickness in each coat shall be within 80 to 120 micron).         Notes:       1.         Valve specification sheet shall be read in conjunction with technical notes for Gate and Globe valves.         2.       Valve shall be designed for intrinsically fire safe. Dimensions including thickness as per ISO 15761.         3.       Testing shall be as per BS EN 12266-1, approved QAP, this specification and other relevant standards.         4.       Bidder shall clearly write all any deviation against each part/ material of valve in the space provided for . Wherever bidder agrees with MECON's data sheet, bidder shall be conducted as per relevant material code.         6.       Hardness test shall be carried out on each heat of base mater		
9.8       Bonneit Bolts       ASTM A320 Gr.L7         9.9       Bonneit Nuts       ASTM A194 Gr.4         9.10       Bonnet Gasket       Spiral Wound SS 316 + Grafoil         10.       Hydrostatic Test Pressure a) Body       :       210 kg/cm²(g)         b) Seat       :       155 kg/cm²(g)         11.       PnuematicTest Pressure with       :       5.6-7 kg/cm2 (g). Air         12.       Painting Specifications:       i)       Surface preparation by Short Blasting as per grade SA 2 1/2, Swedish Standard SIS-055 909.         ii)       For above ground installation-Three coats of corrosion resistant paint shall be applied with minimum thick (Permissible thickness in each coat shall be within 80 to 120 micron).         Notes:       1.       Valve shall be designed for intrinsically fire safe. Dimensions including thickness as per ISO 15761.         3.       Testing shall be as per BS EN 12266-1, approved QAP, this specification and other relevant standards.         4.       Bidder shall clearly write all any deviation against each part/ material of valve in the space provided for Wherever bidder agrees with MECON's data sheet, bidder shall clearly indicate "agreed".         5.       Charpy 'V' notch test on each heat of base material shall be conducted as per relevant material code.         6.       Hardness test shall be carried out on each heat of base material for all pressure containing parts of the v material code.         7.		
9.9       Bonnet Nuts       ASTM A326 Gr.4         9.10       Bonnet Casket       Spiral Wound SS 316 + Grafoil         10.       Hydrostatic Test Pressure       a) Body       : 210 kg/cm²(g)         b) Seat       : 155 kg/cm²(g)         11.       PnuematicTest Pressure with : 5.6-7 kg/cm2 (g).         Air       12.       Painting Specifications:         i)       Surface preparation by Short Blasting as per grade SA 2 1/2, Swedish Standard SIS-055 909.         ii)       For above ground installation-Three coats of corrosion resistant paint shall be applied with minimum thick (Permissible thickness in each coat shall be within 80 to 120 micron).         Notes:       1.       Valve specification sheet shall be read in conjunction with technical notes for Gate and Globe valves.         2.       Valve shall be designed for intrinsically fire safe. Dimensions including thickness as per ISO 15761.         3.       Testing shall be as y deviation against each parV material of valve in the space provided for .		
9.39       Bolinet Nuis       Astim Arsy Gr.*         9.10       Bonnet Gasket       Spiral Wound SS 316 + Grafoil         10.       Hydrostatic Test Pressure a) Body       : 210 kg/cm²(g)         b) Seat       : 155 kg/cm²(g)         11.       PnuematicTest Pressure with Air       : 5.6-7 kg/cm²(g)         12.       Painting Specifications:         13.       Surface preparation by Short Blasting as per grade SA 2 1/2, Swedish Standard SIS-055 909.         ii)       For above ground installation-Three coats of corrosion resistant paint shall be applied with minimum thick (Permissible thickness in each coat shall be within 80 to 120 micron).         Notes:       1.         1.       Valve specification sheet shall be read in conjunction with technical notes for Gate and Globe valves.         2.       Valve shall be designed for intrinsically fire safe. Dimensions including thickness as per ISO 15761.         3.       Testing shall be as per BS EN 12266-1 approved QAP this specification and other relevant standards.         4.       Bidder shall clearly write all/ any deviation against each part/ material of valve in the space provided for . Wherever bidder agrees with MECON's data sheet, bidder shall clearly indicate "agreed".         5.       Charpy 'V' notch test on each heat of base material shall be conducted as per relevant material code.       6.         6.       Hardness test shall be carrired out on each heat of base material for all pressur		
9.10       Bornhet Gasket       [spiral would SS 316 + Grafoli         10.       Hydrostatic Test Pressure a) Body       :       210 kg/cm <sup>2</sup> (g)         b) Seat       :       155 kg/cm <sup>2</sup> (g)         11.       PnuematicTest Pressure with :       5.6-7 kg/cm2 (g). Air         12.       Painting Specifications:       i)         i)       Surface preparation by Short Blasting as per grade SA 2 1/2, Swedish Standard SIS-055 909.         ii)       For above ground installation-Three coats of corrosion resistant paint shall be applied with minimum thick ( Permissible thickness in each coat shall be within 80 to 120 micron).         Notes:       1.       Valve specification sheet shall be read in conjunction with technical notes for Gate and Globe valves.         2.       Valve shall be designed for intrinsically fire safe. Dimensions including thickness as per ISO 15761.         3.       Testing shall be asper BS EN 12266-1, approved QAP, this specification and other relevant standards.         4.       Bidder shall clearly write all any deviation against each pat/ material of valve in the space provided for . Wherever bidder agrees with MECON's data sheet, bidder shall clearly indicate "agreed".         5.       Charpy 'V' notch test on each heat of base material shall be conducted as per relevant material code.         6.       Hardness test shall be carried out on each heat of base material for all pressure containing parts of the v material code.         7.       Stem		
10.       Hydrostatic Test Pressure         a) Body       :       210 kg/cm²(g)         b) Seat       :       155 kg/cm²(g)         11.       PnuematicTest Pressure with       :       5.6-7 kg/cm2 (g).         12.       Painting Specifications:       :       .         13.       Surface preparation by Short Blasting as per grade SA 2 1/2, Swedish Standard SIS-055 909.       .         14.       For above ground installation-Three coats of corrosion resistant paint shall be applied with minimum thick (Permissible thickness in each coat shall be within 80 to 120 micron).         Notes:         1.       Valve specification sheet shall be read in conjunction with technical notes for Gate and Globe valves.         2.       Valve specification sheet shall be read in conjunction with technical notes for Gate and Globe valves.         2.       Valve shall be designed for intrinsically fire safe. Dimensions including thickness as per ISO 15761.         3.       Testing shall be as per BS EN 12266-1, approved QAP, this specification and other relevant standards.         4.       Bidder shall clearly write all/ any deviation against each part/ material of valve in the space provided for .         .       Valve shall be as per BS EN 12266-1 approved QAP, this specification and other relevant material code.         5.       Charpy V'n footh test on each heat of base material shall be conducted as per relevant material code.		
10.       Hydrostatic Lest Pressure         a) Body       :       210 kg/cm²(g)         b) Seat       :       155 kg/cm²(g)         11.       PnuematicTest Pressure with :       5.6-7 kg/cm²(g)         11.       PnuematicTest Pressure with :       5.6-7 kg/cm²(g)         12.       Painting Specifications:       i)         10.       For above ground installation-Three coats of corrosion resistant paint shall be applied with minimum thick (Permissible thickness in each coat shall be within 80 to 120 micron).         Notes:         1.       Valve specification sheet shall be read in conjunction with technical notes for Gate and Globe valves.         2.       Valve shall be designed for intrinsically fire safe. Dimensions including thickness as per ISO 15761 .         3.       Testing shall be as per BS EN 12266-1. Approved OAP, this specification and other relevant standards.         4.       Bidder shall clearly write all/ any deviation against each part/ material of valve in the space provided for .         Wherever bidder agrees with MECON's data sheet, bidder shall clearly indicate "agreed".         5.       Charpy 'V' notch test on each heat of base material shall be conducted as per relevant material code.         6.       Hardness test shall be carried out on each heat of base material for all pressure containing parts of the v material code.         7.       Stem packing shall be an per BERPPTONS       BY		
a) Body       :       210 kg/cm²(g)         b) Seat       :       155 kg/cm²(g)         11.       PnuematicTest Pressure with       :       5.6-7 kg/cm2 (g).         Air       12.       Painting Specifications:       i)       Surface preparation by Short Blasting as per grade SA 2 1/2, Swedish Standard SIS-055 909.         ii)       For above ground installation-Three coats of corrosion resistant paint shall be applied with minimum thick (Permissible thickness in each coat shall be within 80 to 120 micron).         Notes:         1.       Valve specification sheet shall be read in conjunction with technical notes for Gate and Globe valves.         2.       Valve shall be designed for intrinsically fire safe. Dimensions including thickness as per ISO 15761 .         3.       Testing shall be as per BS EN 12266-1, approved QAP, this specification and other relevant standards.         4.       Bidder shall clearly write all/ any deviation against each part/ material of valve in the space provided for .         Wherever bidder agrees with MECON's data sheet, bidder shall clearly indicate "agreed".         5.       Charpy V' notch test on each heat of base material shall be conducted as per relevant material code.         6.       Hardness test shall be carried out on each heat of base material for all pressure containing parts of the v material code.         7.       Stem packing shall be renewable with valve open on stream .         8.		
b) Seat       : 155 kg/cm²(g)         11. Pnuematic Test Pressure with :       5.6-7 kg/cm2 (g). Air         12. Painting Specifications:       i)         i) Surface preparation by Short Blasting as per grade SA 2 1/2, Swedish Standard SIS-055 909.         ii) For above ground installation-Three coats of corrosion resistant paint shall be applied with minimum thick (Permissible thickness in each coat shall be within 80 to 120 micron).         Notes:       1. Valve specification sheet shall be read in conjunction with technical notes for Gate and Globe valves.         2. Valve shall be designed for intrinsically fire safe. Dimensions including thickness as per ISO 15761.         3. Testing shall be as per BS EN 12266-1, approved QAP, this specification and other relevant standards.         4. Bidder shall clearly write all/ any deviation against each part/ material of valve in the space provided for . Wherever bidder agrees with MECON's data sheet, bidder shall clearly indicate "agreed".         5. Charpy V' notch test on each heat of base material shall be conducted as per relevant material code.         6. Hardness test shall be carried out on each heat of base material for all pressure containing parts of the v material code.         7. Stem packing shall be renewable with valve open on stream .         8. Painting procedure of the valves shall be as per Manufacturer's Standard.         9. Material Test Certificates and Hydro Test Reports shall be furnished prior to dispatch.         REV.NO. DATE       DATE         DSECTION PROCESS & PIPING		
11. PnuematicTest Pressure with : 5.6-7 kg/cm2 (g). Air         12. Painting Specifications:         i) Surface preparation by Short Blasting as per grade SA 2 1/2, Swedish Standard SIS-055 909.         ii) For above ground installation-Three coats of corrosion resistant paint shall be applied with minimum thick (Permissible thickness in each coat shall be within 80 to 120 micron).         Notes:         1. Valve specification sheet shall be read in conjunction with technical notes for Gate and Globe valves.         2. Valve shall be designed for intrinsically fire safe. Dimensions including thickness as per ISO 15761.         3. Testing shall be as per BS EN 12266-1, approved QAP, this specification and other relevant standards.         4. Bidder shall clearly write all/ any deviation against each part/ material of valve in the space provided for . Wherever bidder agrees with MECON's data sheet, bidder shall clearly indicate "agreed".         5. Charpy 'V' notch test on each heat of base material shall be conducted as per relevant material code.         6. Hardness test shall be carried out on each heat of base material for all pressure containing parts of the v material code.         7. Stem packing shall be renewable with valve open on stream .         8. Painting procedure of the valves shall be as per Manufacturer's Standard.         9. Material Test Certificates and Hydro Test Reports shall be furnished prior to dispatch.         REV. NO. DATE       DATE         DSCOLIPTIONS       BY         APPRD         Section		
11. PnuematicTest Pressure with : 5.6-7 kg/cm2 (g). Air         12. Painting Specifications:         i) Surface preparation by Short Blasting as per grade SA 2 1/2, Swedish Standard SIS-055 909.         ii) For above ground installation-Three coats of corrosion resistant paint shall be applied with minimum thick (Permissible thickness in each coat shall be within 80 to 120 micron).         Notes:         1. Valve specification sheet shall be read in conjunction with technical notes for Gate and Globe valves.         2. Valve shall be designed for intrinsically fire safe. Dimensions including thickness as per ISO 15761.         3. Testing shall be as per BS EN 12266-1, approved QAP, this specification and other relevant standards.         4. Bidder shall clearly write all/ any deviation against each part/ material of valve in the space provided for . Wherever bidder agrees with MECON's data sheet, bidder shall clearly indicate "agreed".         5. Charpy 'V' notch test on each heat of base material shall be conducted as per relevant material code.         6. Hardness test shall be carried out on each heat of base material for all pressure containing parts of the v material code.         7. Stem packing shall be renewable with valve open on stream .         8. Painting procedure of the valves shall be as per Manufacturer's Standard.         9. Material Test Certificates and Hydro Test Reports shall be furnished prior to dispatch.         REV. NO. DATE       ZONE         0 APRD       CLIENT :         0 BSCCTION       DATE		
Air         12. Painting Specifications:         i) Surface preparation by Short Blasting as per grade SA 2 1/2, Swedish Standard SIS-055 909.         ii) For above ground installation-Three coats of corrosion resistant paint shall be applied with minimum thick (Permissible thickness in each coat shall be within 80 to 120 micron).         Notes:         1. Valve specification sheet shall be read in conjunction with technical notes for Gate and Globe valves.         2. Valve shall be designed for intrinsically fire safe. Dimensions including thickness as per ISO 15761.         3. Testing shall be as per BS EN 12266-1 approved QAP, this specification and other relevant standards.         4. Bidder shall clearly write all/ any deviation against each part/ material of valve in the space provided for . Wherever bidder agrees with MECON's data sheet, bidder shall clearly indicate "agreed".         5. Charpy 'V' notch test on each heat of base material shall be conducted as per relevant material code.         6. Hardness test shall be carried out on each heat of base material for all pressure containing parts of the v material code.         7. Stem packing shall be renewable with valve open on stream .         8. Painting procedure of the valves shall be as per Manufacturer's Standard.         9. Material Test Certificates and Hydro Test Reports shall be furnished prior to dispatch.         REV. NO. DATE       ZONE         SECTION PROCESS & PIPING       CLIENT :         NAME       DATE         DSGN       PM </td <td></td> <td></td>		
12. Painting Specifications:         i)       Surface preparation by Short Blasting as per grade SA 2 1/2, Swedish Standard SIS-055 909.         ii)       For above ground installation-Three coats of corrosion resistant paint shall be applied with minimum thick (Permissible thickness in each coat shall be within 80 to 120 micron).         Notes:         1.       Valve specification sheet shall be read in conjunction with technical notes for Gate and Globe valves.         2.       Valve shall be designed for intrinsically fire safe. Dimensions including thickness as per ISO 15761.         3.       Testing shall be as per BS EN 12266-1, approved QAP, this specification and other relevant standards.         4.       Bidder shall clearly write all/ any deviation against each part/ material of valve in the space provided for .         Wherever bidder agrees with MECON's data sheet, bidder shall clearly indicate "agreed".         5.       Charpy 'V' notch test on each heat of base material for all pressure containing parts of the v material code.         7.       Stem packing shall be renewable with valve open on stream .         8.       Painting procedure of the valves shall be as per Manufacturer's Standard.         9.       Material Test Certificates and Hydro Test Reports shall be furnished prior to dispatch.         REV. NO.       DATE       ZONE         SECTION       PROCESS & PIPING       CLIENT :         DSGN       PM       Z5.04.12		
<ul> <li>i) Surface preparation by Short Blasting as per grade SA 2 1/2, Swedish Standard SIS-055 909.</li> <li>ii) For above ground installation-Three coats of corrosion resistant paint shall be applied with minimum thick (Permissible thickness in each coat shall be within 80 to 120 micron).</li> <li>Notes:         <ol> <li>Valve specification sheet shall be read in conjunction with technical notes for Gate and Globe valves.</li> <li>Valve shall be designed for intrinsically fire safe. Dimensions including thickness as per ISO 15761.</li> <li>Testing shall be as per BS EN 12266-1, approved QAP, this specification and other relevant standards.</li> <li>Bidder shall clearly write all/ any deviation against each part/ material of valve in the space provided for . Wherever bidder agrees with MECON's data sheet, bidder shall clearly indicate "agreed".</li> <li>Charpy 'V' notch test on each heat of base material shall be conducted as per relevant material code.</li> <li>Hardness test shall be carried out on each heat of base material for all pressure containing parts of the v material code.</li> <li>Stem packing shall be renewable with valve open on stream .</li> <li>Painting procedure of the valves shall be as per Manufacturer's Standard.</li> <li>Material Test Certificates and Hydro Test Reports shall be furnished prior to dispatch.</li> </ol> </li> <li>REV. NO. DATE ZONE DESCRIPTIONS BY APPRD         <ul> <li>REFERENCES</li> <li>SECTION PROCESS &amp; PIPING</li> <li>CLIENT :</li> <li>NAME DATE CHKD DATE</li> <li>PROJECT :</li> <li>PROJECT :</li> <li>DATA SHEET FOR GLOBE VALVES</li> </ul> </li> </ul>		
ii)       For above ground installation-Three coats of corrosion resistant paint shall be applied with minimum thick (Permissible thickness in each coat shall be within 80 to 120 micron).         Notes:       1.       Valve specification sheet shall be read in conjunction with technical notes for Gate and Globe valves.         2.       Valve shall be designed for intrinsically fire safe. Dimensions including thickness as per ISO 15761.         3.       Testing shall be as per BS EN 12266-1, approved QAP, this specification and other relevant standards.         4.       Bidder shall clearly write all/ any deviation against each part/ material of valve in the space provided for .         Wherever bidder agrees with MECON's data sheet, bidder shall clearly indicate "agreed".         5.       Charpy V' notch test on each heat of base material shall be conducted as per relevant material code.         6.       Hardness test shall be carried out on each heat of base material for all pressure containing parts of the v material code.         7.       Stem packing shall be renewable with valve open on stream .         8.       Painting procedure of the valves shall be as per Manufacturer's Standard.         9.       Material Test Certificates and Hydro Test Reports shall be furnished prior to dispatch.         REV. NO.       DATE       DESCRIPTIONS       BY         SECTION       PROCESS & PIPING       CLIENT :       SCALE :         DRWN       DATA       SCALE       PATE       <		
(Permissible thickness in each coat shall be within 80 to 120 micron).         Notes:         1.       Valve specification sheet shall be read in conjunction with technical notes for Gate and Globe valves.         2.       Valve shall be designed for intrinsically fire safe. Dimensions including thickness as per ISO 15761.         3.       Testing shall be as per BS EN 12266-1, approved QAP, this specification and other relevant standards.         4.       Bidder shall clearly write all/ any deviation against each part/ material of valve in the space provided for . Wherever bidder agrees with MECON's data sheet, bidder shall clearly indicate "agreed".         5.       Charpy 'V' notch test on each heat of base material shall be conducted as per relevant material code.         6.       Hardness test shall be carried out on each heat of base material for all pressure containing parts of the v material code.         7.       Stem packing shall be renewable with valve open on stream .         8.       Painting procedure of the valves shall be as per Manufacturer's Standard.         9.       Material Test Certificates and Hydro Test Reports shall be furnished prior to dispatch.         REV. NO.       DATE       DATE         DESCRIPTIONS       BY       APPRD         SECTION       PROCESS & PIPING       CLIENT :         DSGN       PM       25.04.12       AKJ         DRWN       O. P. JAIN       DATE FOR GLOBE VALVES <t< td=""><td>n thickness of 300 micron</td><td></td></t<>	n thickness of 300 micron	
Notes:       1.       Valve specification sheet shall be read in conjunction with technical notes for Gate and Globe valves.         2.       Valve shall be designed for intrinsically fire safe. Dimensions including thickness as per ISO 15761.         3.       Testing shall be as per BS EN 12266-1, approved QAP, this specification and other relevant standards.         4.       Bidder shall clearly write all/ any deviation against each part/ material of valve in the space provided for .         Wherever bidder agrees with MECON's data sheet, bidder shall clearly indicate "agreed".       5.         5.       Charpy 'V' notch test on each heat of base material shall be conducted as per relevant material code.         6.       Hardness test shall be carried out on each heat of base material for all pressure containing parts of the v material code.         7.       Stem packing shall be renewable with valve open on stream .         8.       Painting procedure of the valves shall be as per Manufacturer's Standard.         9.       Material Test Certificates and Hydro Test Reports shall be furnished prior to dispatch.         REV. NO.       DATE       DESCRIPTIONS       BY       APPRD         SECTION       PROCESS & PIPING       CLIENT :       REFERENCES         SECTION       PROCESS & PIPING       CLIENT :       SCALE :         DSGN       PM       25.04.12       AKJ       25.04.12       PROJECT :       DATA SHEET FOR		
Notes:       1. Valve specification sheet shall be read in conjunction with technical notes for Gate and Globe valves.         2. Valve shall be designed for intrinsically fire safe. Dimensions including thickness as per ISO 15761.         3. Testing shall be as per BS EN 12266-1, approved QAP, this specification and other relevant standards.         4. Bidder shall clearly write all/ any deviation against each part/ material of valve in the space provided for . Wherever bidder agrees with MECON's data sheet, bidder shall clearly indicate "agreed".         5. Charpy 'V' notch test on each heat of base material shall be conducted as per relevant material code.         6. Hardness test shall be carried out on each heat of base material for all pressure containing parts of the v material code.         7. Stem packing shall be renewable with valve open on stream .         8. Painting procedure of the valves shall be as per Manufacturer's Standard.         9. Material Test Certificates and Hydro Test Reports shall be furnished prior to dispatch.         REV. NO. DATE       ZONE         SECTION PROCESS & PIPING       CLIENT :         REFERENCES       REFERENCES         SECTION PROCESS & PIPING       CLIENT :         DATE       CHKD       DATE         DSGN       PM       25.04.12       AKJ       25.04.12         DRWN       O. P. JAIN       DATA SHEET FOR GLOBE VALVES       SCALE :         DATA SHEET FOR CLOBE VALVES       DATA SHEET NO.: ME		
1.       Valve specification sheet shall be read in conjunction with technical notes for Gate and Globe valves.         2.       Valve shall be designed for intrinsically fire safe. Dimensions including thickness as per ISO 15761.         3.       Testing shall be as per BS EN 12266-1, approved QAP, this specification and other relevant standards.         4.       Bidder shall clearly write all/ any deviation against each part/ material of valve in the space provided for .         Wherever bidder agrees with MECON's data sheet, bidder shall clearly indicate "agreed".       5.         5.       Charpy 'V' notch test on each heat of base material shall be conducted as per relevant material code.         6.       Hardness test shall be carried out on each heat of base material for all pressure containing parts of the v material code.         7.       Stem packing shall be renewable with valve open on stream .         8.       Painting procedure of the valves shall be as per Manufacturer's Standard.         9.       Material Test Certificates and Hydro Test Reports shall be furnished prior to dispatch.         REV. NO.       DATE       ZONE         SECTION PROCESS & PIPING       CLIENT :         NAME       DATE       CHKD         DSGN       PM       25.04.12       AKJ         DSGN       PM       25.04.12       AKJ         DRWN       O. P. JAIN       DATA SHEET FOR GLOBE VALVES       SCALE :<		
2.       Valve shall be designed for intrinsically fire safe. Dimensions including thickness as per ISO 15761.         3.       Testing shall be as per BS EN 12266-1, approved QAP, this specification and other relevant standards.         4.       Bidder shall clearly write all/ any deviation against each part/ material of valve in the space provided for .         5.       Charpy 'V' notch test on each heat of base material shall be conducted as per relevant material code.         6.       Hardness test shall be carried out on each heat of base material for all pressure containing parts of the v material code.         7.       Stem packing shall be renewable with valve open on stream .         8.       Painting procedure of the valves shall be as per Manufacturer's Standard.         9.       Material Test Certificates and Hydro Test Reports shall be furnished prior to dispatch.         REV. NO.       DATE       ZONE       DESCRIPTIONS         BY       APPRD       REFERENCES         SECTION       PROCESS & PIPING       CLIENT :         NAME       DATE       CHKD       DATE         DSGN       PM       25.04.12       AKJ       25.04.12         DRWN       Image: Constant of the constant of the constant of the constant of the constant of the constant of the constant of the constant of the constant of the constant of the constant of the constant of the constant of the constant of the constent of the constant of the constant of the constant of the consta	25	
3.       Testing shall be as per BS EN 12266-1, approved QAP, this specification and other relevant standards.         4.       Bidder shall clearly write all/ any deviation against each part/ material of valve in the space provided for . Wherever bidder agrees with MECON's data sheet, bidder shall clearly indicate "agreed".         5.       Charpy 'V' notch test on each heat of base material shall be conducted as per relevant material code.         6.       Hardness test shall be carried out on each heat of base material for all pressure containing parts of the v material code.         7.       Stem packing shall be renewable with valve open on stream .         8.       Painting procedure of the valves shall be as per Manufacturer's Standard.         9.       Material Test Certificates and Hydro Test Reports shall be furnished prior to dispatch.         REV. NO.       DATE       DESCRIPTIONS       BY       APPRD         SECTION       PROCESS & PIPING       CLIENT :       REFERENCES         SECTION       PROCESS & PIPING       CLIENT :       SCALE :         DSGN       PM       25.04.12       AKJ       25.04.12       PROJECT :         DRWN       DATE       O. P. JAIN       DATA SHEET FOR GLOBE VALVES       SCALE :         DATA SHEET FOR CLOBE VALVES       DATA SHEET NO.: ME       DATA SHEET NO.: ME		
A     Bidder shall clearly write all/ any deviation against each part/ material of valve in the space provided for .     Wherever bidder agrees with MECON's data sheet, bidder shall clearly indicate "agreed".     Charpy 'V' notch test on each heat of base material shall be conducted as per relevant material code.     Hardness test shall be carried out on each heat of base material for all pressure containing parts of the v     material code.     Hardness test shall be renewable with valve open on stream .     Set packing shall be renewable with valve open on stream .     Bidder in the space provided for .     Material Test Certificates and Hydro Test Reports shall be furnished prior to dispatch.     REV. NO. DATE ZONE DESCRIPTIONS BY APPRD     REFERENCES     SECTION PROCESS & PIPING     CLIENT :     NAME DATE CHKD DATE     DSGN PM 25.04.12 AKJ 25.04.12     PROJECT :     APPROVED     O. P. JAIN DATA SHEET FOR GLOBE VALVES	rde	
<ul> <li>4. Bidder shall clearly write any deviation against each part material of valve in the space provided for . Wherever bidder agrees with MECON's data sheet, bidder shall clearly indicate "agreed".</li> <li>5. Charpy 'V' notch test on each heat of base material shall be conducted as per relevant material code.</li> <li>6. Hardness test shall be carried out on each heat of base material for all pressure containing parts of the v material code.</li> <li>7. Stem packing shall be renewable with valve open on stream .</li> <li>8. Painting procedure of the valves shall be as per Manufacturer's Standard.</li> <li>9. Material Test Certificates and Hydro Test Reports shall be furnished prior to dispatch.</li> <li>REV. NO. DATE ZONE DESCRIPTIONS BY APPRD</li> </ul> REFERENCES           SECTION PROCESS & PIPING         CLIENT :           NAME         DATE           DSGN         PM           25.04.12         AKJ           DSGN         PM           25.04.12         AKJ           DRWN         O. P. JAIN   DATA SHEET FOR GLOBE VALVES           SCALE :           DATA SHEET FOR GLOBE VALVES	uo. Adfor	
S.       Charpy 'V' notch test on each heat of base material shall be conducted as per relevant material code.         6.       Hardness test shall be carried out on each heat of base material for all pressure containing parts of the v material code.         7.       Stem packing shall be renewable with valve open on stream .         8.       Painting procedure of the valves shall be as per Manufacturer's Standard.         9.       Material Test Certificates and Hydro Test Reports shall be furnished prior to dispatch.         REV. NO. DATE ZONE DESCRIPTIONS BY APPRD         SECTION PROCESS & PIPING         CLIENT :         NAME       DATE         DSGN       PM       25.04.12         AKJ       25.04.12       AKJ         DRWN       O. P. JAIN       DATA SHEET FOR GLOBE VALVES         APPROVED       O. P. JAIN       DATA SHEET FOR GLOBE VALVES	u 101 .	
b.       Charpy 'V' notch test on each heat of base material shall be conducted as per relevant material code.         6.       Hardness test shall be carried out on each heat of base material for all pressure containing parts of the v material code.         7.       Stem packing shall be renewable with valve open on stream .         8.       Painting procedure of the valves shall be as per Manufacturer's Standard.         9.       Material Test Certificates and Hydro Test Reports shall be furnished prior to dispatch.         REV. NO.       DATE         SECTION       PROCESS & PIPING         SECTION       PROCESS & PIPING         SECTION       PROCESS & PIPING         SECTION       PROJECT :         NAME       DATE         CHKD       DATE         PROJECT :       SCALE :         DATA SHEET FOR GLOBE VALVES       DATA SHEET NO.: ME		
b.       Hardness test shall be carried out on each heat of base material for all pressure containing parts of the v material code.         7.       Stem packing shall be renewable with valve open on stream .         8.       Painting procedure of the valves shall be as per Manufacturer's Standard.         9.       Material Test Certificates and Hydro Test Reports shall be furnished prior to dispatch.         REV. NO.       DATE         DATE       DONE         SECTION       PROJECT :         NAME       DATE         CHKD       DATE         DSGN       PM         25.04.12       AKJ         DSGN       PM         25.04.12       AKJ         25.04.12       O. P. JAIN         DATA SHEET FOR GLOBE VALVES       SCALE :         DATA SHEET FOR GLOBE VALVES	e.	
material code.       7.       Stem packing shall be renewable with valve open on stream .         8.       Painting procedure of the valves shall be as per Manufacturer's Standard.         9.       Material Test Certificates and Hydro Test Reports shall be furnished prior to dispatch.         REV. NO.       DATE       ZONE         BY       APPRD         SECTION       PROCESS & PIPING       CLIENT :         NAME       DATE       CHKD         DSGN       PM       25.04.12       AKJ         DRWN       O. P. JAIN       DATA SHEET FOR GLOBE VALVES       SCALE :         DATA SHEET FOR GLOBE VALVES       DATA SHEET NO.: ME	r the valve as per relevant	
7.       Stem packing shall be renewable with valve open on stream .         8.       Painting procedure of the valves shall be as per Manufacturer's Standard.         9.       Material Test Certificates and Hydro Test Reports shall be furnished prior to dispatch.         REV. NO.       DATE       ZONE       DESCRIPTIONS       BY       APPRD         REFERENCES         SECTION PROCESS & PIPING         SECTION       DATE       CLIENT :       REFERENCES         DSGN       PM       25.04.12       AKJ       25.04.12       PROJECT :       SCALE :         DRWN       O. P. JAIN       DATA SHEET FOR GLOBE VALVES       SCALE :       DATA SHEET NO.: ME		
8.       Painting procedure of the valves shall be as per Manufacturer's Standard.         9.       Material Test Certificates and Hydro Test Reports shall be furnished prior to dispatch.         REV. NO.       DATE       ZONE       DESCRIPTIONS       BY       APPRD         REFERENCES         SECTION PROCESS & PIPING         NAME       DATE       CLIENT :       REFERENCES         DSGN       PM       25.04.12       AKJ       25.04.12         DRWN       AKJ       25.04.12       PROJECT :       SCALE :         DATA SHEET FOR GLOBE VALVES       DATA SHEET NO.: ME       DATA SHEET NO.: ME		
9. Material Test Certificates and Hydro Test Reports shall be furnished prior to dispatch.         REV. NO.       DATE       ZONE       DESCRIPTIONS       BY       APPRD         SECTION PROCESS & PIPING         SECTION       DATE       CHKD       DATE       CLIENT :       Image: Client is the state of		
REV. NO.     DATE     ZONE     DESCRIPTIONS     BY     APPRD       SECTION     PROCESS & PIPING     CLIENT :     REFERENCES       NAME     DATE     CHKD     DATE       DSGN     PM     25.04.12     AKJ     25.04.12       DRWN     O. P. JAIN     DATA SHEET FOR GLOBE VALVES     SCALE :		
REFERENCES         SECTION       PROCESS & PIPING       CLIENT :       REFERENCES         DSGN       PM       25.04.12       AKJ       25.04.12       PROJECT :       Image: Colspan="2">SCALE :         DRWN       DATE       DATA SHEET FOR GLOBE VALVES       SCALE :       DATA SHEET NO.: ME		
SECTION       PROCESS & PIPING       CLIENT :         NAME       DATE       CHKD       DATE         DSGN       PM       25.04.12       AKJ       25.04.12         DRWN       O. P. JAIN       DATA SHEET FOR GLOBE VALVES       SCALE :         DATA SHEET NO.: ME       DATA SHEET NO.: ME       DATA SHEET NO.: ME	DRG. NO.	
NAME       DATE       CHKD       DATE         DSGN       PM       25.04.12       AKJ       25.04.12         DRWN       AKJ       25.04.12       PROJECT :       SCALE :         DATPROVED       O. P. JAIN       DATA SHEET FOR GLOBE VALVES       SCALE :		
DSGN     PM     25.04.12     AKJ     25.04.12     PROJECT :       DRWN     Image: Constraint of the second secon		
DSGN         PM         25.04.12         AKJ         25.04.12         PROJECT :           DRWN		
APPROVED O. P. JAIN DATA SHEET FOR GLOBE VALVES DATA SHEET NO.: ME	MECON LIMITED	
APPROVED O. P. JAIN DATA SHEET FOR GLOBE VALVES DATA SHEET NO.: ME	<u> </u>	
	NO.: MEC/WINO/05/28/M/001/DS/GV/83	REV 0
(NB<2")		

#### DATA SHEET FOR CARTRIDGE FILTER CLIENT : 1.0 PROJECT 2.0 JOB NO. : ITEM TAG NO. : CF-3.0 VENDOR QUANTITY : As per P&ID OPERATING CONDITIONS : 4.0 FILTER : CARTRIDGE FLUID HANDLED : NATURAL GAS VAPOUR/RLNG GAS TYPE : HORIZONTAL FLUID DENSITY(@ P&T) Kg/m3 : \*\* FLOW RATE (MMSCMD) : Refer P&ID 4.2 4.3 MOLECULAR WEIGHT : \*\* 4.4 Cn/Cv · \* FLUID VISC CENTI-POISE \* 4.5 INLET SIZE (NB) : Refer P&ID COMPRESSIBILITY FACTOR (Z) : OUTLET SIZE (NB) : Refer P&ID OPER. TEMP., °C : 0- 55 FILTRATION EFF. % : 98 4.6 OPER. PR., KG/CM<sup>2</sup>G : Refer P&ID 4.7 PARTICLE/ MESH SIZE, MICRON ≥ 3 MICRON 4.8 PR. DROP KG/CM<sup>2</sup>, CLEAN/ DIRTY : 0.2/ 0.5 Max. (SEE NOTE-2) DUST CONC : 0 - 0.2 mg/SM3 OF GAS 4.9 CORROSION ALLOWANCE, MM : 3.0 For Carbon Steel Parts CORROSIVE/ TOXIC COMPONENT : Tot sulphur incl. H2S (max.)-10 PPM(by wt) 5.0 MATERIAL OF CONSTRUCTION H2S content (max)- 5PPM (by wt.) SHELL FLANGE : SA - 105 HEAD FLANGE : SA -105 SHELL : SA-515/SA-516 Gr. 60/70 5.1 5.2 HEAD PERFORATED SHEET : 5.3 BOTTOM : SA515/ SA-516 Gr. 60/70,SA 234 Gr.WPB NOZZLE FLANGES : SA - 105 FASTENER : SA-193 Gr. B7, SA-194 Gr. 2H SUPPORT : SA516Gr.60/ SA 283 Gr. C/ IS:2062 5.4 NOZZLES : SA-106 GR. B GASKET : SS-304/316 SPIRAL WOUND WITH GRAFOIL 5.5 5.6 OTHER INTERNALS: FITTING: SA 234 Gr. WPB FILTER ELEMENT : FIBRE GLASS MEDIA TO SUIT GAS QUALITY. 5.7 6.0 DESIGN AND CONSTRUCTION 6.1 DESIGN CODE : ASME SEC-VIII DIV-1 (LATEST EDITION) besign code : Asme sec-viii Div-1 (LATEST EDITION) besign PR., Kg/CM2G : As per P&ID NO OF CARTRIDGE ELEMENT: (SEE NOTE-2) MAKE (ELEMENT) : AS PER VENDOR LIST IN MECON T.S. FIXING DETAILS : NUTS & BOLTS DESIGN TEMP., °C : (-)29 to (+) 65 O.D. (MM) X LENGTH (MM) XTHICKNESS(MM): FILTER PR. DROP, KG/CM<sup>2</sup> : . (SEE NOTE-2) DUST/DIRT HOLDING CAPACITY OF EACH CARTRIDGE ELEMENT 6.6 NAME OF VESSEL MANUFACTURER : \* TOTAL GROSS FILTERING AREA OF CARTRIDGE ELEMENTS: 6.7 FLANGE RATING : 600# FLANGE FINISH : \* 6.8 FLANGE TYPE : WNRF PSV SIZE : DPT SIZE : \* 6.9 VENT SIZE : Refer P&ID, DRAIN SIZE : Refer P&ID, UC SIZE : NR FLUSHING CONNECTION : 6.10 HEAD(TOP COVER) CONNECTION : FLANGE 6.11 QOC REQUIRED : NA 7.0 OVERALL DIMENSION & WEIGHT: OVERALL LENGTH, MM INLET FLANGE TO OUTLET FLANGE: 7.2 OVERALL HEIGHT, MM SHELL DIA. MM 7.3 EMPTY WEIGHT, KGS OPERATING WT., KGS 74 HYDROTESTWEIGHT : 8.0 ACCESSORIES 8.1 DAVIT : YES 8.2 LIFTING LUGS YES ANCHOR BOLTS& NUTS YES 8.3 8.4 BLIND FLANGE, GASKET, BOLTS&NUTS FOR VENTS&DRAINS : YES 8.5 LADDER&PLATEFORM: 9.0 PAINTING : SUITABLE TO CORROSIVE INDUSTRIAL ENVIRONMENT REF. MECON T.S. 10.0 INSPECTION & TESTING : AS PER MECON T.S. \* VENDOR TO SPECIFY/ CONFIRM \*\* GAS COMPOSITION & OTHER PROPERTIES WILL BE PROVIDED TO SUCCESSFUL BIDDER. NOTES :-1) THE TOTAL INTERNAL CROSS SECTIONAL AREA OF MOUNTED CARTRIDGE SHALL NOT BE LESS THAN INLET NOZZLE AREA 2) SUCCESSFUL BIDDER SHALL SUBMIT MECHNICAL DESIGN CALCULATION FOR FILTER & PROCESS CALCULATION (ALONG WITH DETAILED PRESSURE DROP CALCULATION) OF CARTRIDGE FILTER & CARTRIDGE ELEMENT (INCLUDING RELEVANT GRAPH, CATALOGUE ETC.) FOR MECON CLEARANCE. 3) BIDDER SHALL SUBMIT SAMPLE CALCULATIONS (FOR CATRIDGE FLEMENT SIZING & PRESSURE DROP ACROSS THE FILTER ) ALONG WITH OFFER 4) VENDOR TO SELECT/CONFIRM THE MATERIAL CONSIDERING "-29 °C TO 65 °C" TEMPERATURE . A CHARPY V-NOTCH TEST SHALL BE CONDUCTED ON 3 SAMPLE HAVING ENERGY VALUE OF 27 J AVERAGE AND MINIMUM 22 J AT 0°C. 5) FILTER ELEMENT MUST WITHSTAND A DIFFERENTIAL PRESSURE OF 1.0 KG/CM2(a) WITHOUT CRACKING AND FAILING. 6)MANDATORY SPARES SHALL BE AS PER RELEVANT CLAUSE SPECIFIED IN TENDER DOCUMENT. 7)FILTER ELEMENT SHALL BE OF SINGLE PIECE CONSTRUCTION WITH NO JOINTS IN THE ELEMENT. STACKING OF FILTER ELEMENT IS NOT ALLOWED. 8)ALL CS MATERIALS SHALL BE OF FULLY KILLED QUALITY. 9) FLANGE SHALL CONFIRM TO ASME B 16.5. 10)GASKETS SHALL CONFORM TO ASME B 16.20 11) FREQUENCY OF CLEANING SHALL BE MINIMUM 30 DAYS FOR FILTER ELEMENT DESIGN 12) PWHT REQUIRED AS PER ASME SEC-VIII (DIV-I) REV. DATE APPRD ZONE DESCRIPTIONS ΒY REVISIONS REFERENCES DRG. NO. SECTION : OIL & GAS CLIENT : MECON LIMITED NAME DATE CHKD DATE PROJECT : SGN DRWN REV SCALE CARTRIDGE FILTER APPROVED DATA SHEET NO .: 0 This document and the design it covers are the property of MECON and issued for the specific project mentioned therein. This is not to be copied or used for other projects unless expressly permitted by MECON.

				PRE	SSURE SAFET	'Y VA	LVES/CREEP F	RELIEF VA	VE	
			<u>^</u>							
UNITS : Fk	ow > Liquid	- m <sup>3</sup> /hr , G	as-Sm³/day,	Steam - kg/hr. Pres	sure -> kg/cm <sup>2</sup> g, Tempe	erature-°	C, Level/ Length-> mm	0.01		
General	01	lag No.	Di		PSV-			CRV-		
	02	Line No./ 3	bize		As per P&ID Cartridge Filter			As per P&IL		
	03	Quantity	heeled		As per P&ID			As per P&ID		
	05	Safety/ Re	lief		Safety Relief			Creep Relief		
	06	Vendor			<b>†</b>			¢		
Valve	07	Туре			Standard			Standard		
	08	Full Nozzle	Full Lift Moo	I. Nozzle	Full Nozzle Full Lift			Full Nozzle F	ull Lift	
	09	Conv / Rol	pe lown/ Bllot On	orotod	Closed			Closed		
	10	Inlet Conn	· Size & Rati	na	•			•	I	
	12	Inlet Conn	. : Facing & F	inish	RF, <del>†</del>			RF, <del>†</del>		
	13	Outlet Con	n. : Size & Ra	ating	<b>†</b>			¢		
	14	Outlet Con	n. : Facing &	Finish	RF,†			RF,♥		
	15	Cap Over	Adj. Bolt :		Required			Required		
	16	Screwed	Bolted		Bolted			Bolted		
	17	Test Gad	ar - Type		- Required			- Required		
Material	19	Body and	Bonnet		ASTM A216 Gr. WCB	3		ASTM A216	Gr. WCB	
	20	Nozzle and	d Disc		SS 316			SS 316		
	21	Spring			SS 316			SS 316		
	22	Bellows								
Ontiona	22	Decilient C	ant Cool							
Options	23	Resilient S	eal Seal							
Basis	24	Code			API 520 521 & 526			API 520 52	1 & 526 / As per M	Anufacturer's Standard
Daolo	25	Basis of S	election		Vessel Under Ext. Fire	e Case		CREEP REL	IEF (1% flow)	
Service	26	Fluid and	State		Natural Gas Vapour			Natural Gas	Vapour	
Conditions	27	Corrosive	Constituent		Tot sulphur incl. H2S	(max.)-1	0 PPM(by wt)	Tot sulphur i	ncl. H2S (max.)-10	) PPM(by wt)
	20	Corr Allow	0000		H2S content (max)- 5	PPIVI (D	/ wt.)	H2S content	(max)- 5PPIVI (by	wt.)
	20	Required F	Flow Canacity		¢			¢		
	30	Mol. Wgt.	S.G. at Rel. 1	lemp.	• •		<b><b></b></b>	• •		<b><b></b></b>
	31	Oper. Pres	ssure, kg/cm <sup>2</sup>	g	As per P&ID		l.	As per P&ID	,	4
	32	Oper. Terr	p.°C Rel. Tei	mp.°C	0-55		¢	0-55		¢
	33	Valve Disc	harges to		Atm.			Atm.		т
	34	Back Pres	s. Const. Or	Variable	Atm.		Constant	Atm.		Constant
	35	Set Pressu	ure, Kg/cm2(g	)	- (Note-7)			- (Note-7)		
	30	% Over Pr	essure % Br	w Down	₽ 20		\$	10		•
	38	Cp/Cv Co	mpressibility I	Factor	<b> † †</b>		• •	<b>++</b>		• •
	39	Viscosity a	at Rel. Temp.	(cP)	<b>ቀ</b> ቀ			<b><b> </b></b>		
	40	Vessel Wa	all Temp.,°C	Surf. Area-m <sup>2</sup>	593		¢	-		-
Orifice	41	Calculated	Area-inch <sup>2</sup>		<b>¢</b>		· .	<b></b>		Τ.
	42	Sel. Area-	inch <sup>2</sup>  Orifice	Design	<b>•</b>		Ŷ	<b>•</b>		•
	43	NO. OF VAN	ves Requ. for	capacity	Ψ •			Ψ		
	44	Actual Flor	-Inch MCanacity S	CEM	Ψ -			Ψ -		
	46	Relief Loa	d		- +			- •		
	47	Model No.	-		<b>†</b>			•		
	48	Radiograp	hy & Charpy	Test	Reqd. (100%)			Reqd. (100%	ó)	
	49	IBR Certifi	cation		Not Required			Not Required	t	
* * 1. 2. 3. 4. 5. 6. 7.	VENDOR GAS COM VENDOR VALVES S VENDOR FOR SAFE DESIGNA PSV's & C PSV's & C	TO SPECIF POSITION SHALL FUF SHALL BE 1 TO CONSIE TY VALVE TION, NO. RV'S SHAL RV'S SHAL	Y/ CONFIRM & OTHER PR NISH SIZING 100% RADIOO DER COEFFIO SIZING, FUR & RATING OF L BE SUPPLI PESSURE S	OPERTIES WILL BA CALCULATIONS T SRAPHED. JIENT OF DISCHAR NISH CERTIFIED ( PSV's & CRV's SH ED WITH INLET AN HALL BE FINALISE	E PROVIDED TO SUCE O SUPPORT HIS VALV GE AS PER ASME-SEI APACITIES AS PER A ALL BE DECIDED DUR D OUTLET COMPANIC D DURING DETAILED I	essful (e sele (c-viii (i (pi-520) Ring de On flai <u>Engine</u>	BIDDER. :CTION. Latest). ETAIL ENGINEERING. NGE. :ERING.			
REV. NO.	DATE	ZONE		DESCRIPTIONS	BY					
				REVISIONS			-	DF	(G. NO.	
SECTION :	OIL & GA	S	1		CLIENT :					
DSGN	NAME	DATE	CHKD	DATE	PROJECT :		मेकॉन	MECO		
DRWN							9001 Comp			
APPROVE	D				PSV/CRV		DATASHEET NO:			REV-0
This docun copied or	nent and the	e design it o	covers are the s unless expre	property of MECO	and issued for the spe ECON.	ecific pro	oject mentioned therein.	This is not to be		<u>.</u>

### DATA SHEET-SCRUBBER

ſ

APPRO	VED					TAG NO. SC-			DATA SHEET NO.:		0
URWN						SCRUBBER			SCALE :		REV
DSGN					PROJECT :				Soon Campan	MECON LIMIT	ED
	NAME	DATE	СНКД	DATE					मेकॉन		
SECTIC	ON O&G	(LDP)			CLIENT :						
0505		(1.0.0)		REVISIONS	0.15.15				REFERENCES	DRG. NO.	
REV. NO.	DATE	ZONE		DESCRIPTIONS		BY	APPRD				
** GAS	COMPOS	SITION & O	THER PRC	PERTIES WILL BE	PROVIDED TO	SUCCESSFUL B	BIDDER.				
¢	VENDO	R TO SPEC	IFY/ CONF	IRM							
10.0	INSPEC	TION & TE	STING : AS	PER Mecon T.S.							
9.0	PAINTIN	IG : SUITA	BLE TO CO	RROSIVE INDUSTR	RIAL ENVIRON	MENT REF. MECO	ON T.S.				
8.1	DAVIT D	ETAIL : YE	S								
8.0	ACCESS	SORIES									
7.4	EMPTY	WEIGHT, K	GS :	<del>\$</del>		OPERATING W	T., KGS :	¢			
7.3	OVERAL	L HEIGHT	, CM :	<del>\$</del>		SHELL DIA, CM	: 🕈				
7.2	OVERAL	L LENGTH	I, CM :	¢							
7.1	OVERAL	L DIMENS	ION	¢							
7.0	COVER	BOLTED	WITH DAVI	T CONNECTION							
6.9	LG CON	NECTION	Refer P&I								
6.8	VENT SI	ZE : Refer	P&ID	DRAIN SIZE : Ret	fer P&ID	UC SIZE : Re	efer P&ID	U UAIA	STILL I		
67	FLANGE	TYPF · W	NRF			PSV SIZE · REF	. ATTACHE		SHEET		
6.6	FLANGE	RATING	600#			FLANGE FINISH		I FINISH	(125 AARH)		
6.5	FIXING		S PFR SUP	PLIFR		J.D. A LENGTH	,. 1				
0.2 6.3	INTERN		SS-3041			OD XIENGTH	, ∪29 IC •⊕	00			
0.1	DESIGN	DR KGIC	DIVIE DEU.V				°C · . 20 TC	1 65			
0.0 6 1	DESIGN			אוי ווו/עבו							
0.C 6.0	DESIGN	AND CON	STRUCTIO		AFUIL	FASTENER : SA-193 GR. B7, SA-194 GR. 2H					
0.0 5.6	GACKET	1 : 15:2002 - · SS 201/2				EASTENED . 24	-304L	7 54 10			
5.4 5.5		3:3A-106	GR. B				304I 304I	0			
5.3		1:5A 516	GI. 60 CP P					A 15			
5.2		5A 516 GR.	00								
5.1	SHELL :	5A-515/SA	-516 Gr. 60	110/ SA 106 Gr. B		SHELL FLANGE	: : NA - NA				
5.0	MATERI										
1						SP. GRAVITY : *	**				
4.9	COMPR	ESSIBILITY	FACTOR	**		EXTENT OF RA	DIOGRAPH	IY - 100%	, D		
									H2S content (m	ax)- 5PPM (by wt.)	
4.8	CORRO	SION ALLC	WANCE, N	1M : 3.0		CORROSIVE/ T	OXIC COM	PONENT	: Tot sulphur incl	. H2S (max.)-10 PPM(b	oy wt)
4.7	PR. DRC	OP KG/CM <sup>2</sup>	, CLEAN/ D	IRTY : 0.05 / -		SCRUBBER PR	. DROP, KG	G/CM <sup>2</sup> : 0.	.1 MAX.		
	<u>&gt;</u> 10 MIC	RON		,							
4.6		& DUST PA	RTICLE/ M	ESH SIZE. MICRON	1	FILTERATION F	EFF. % : 98				
4.5	OPER F		G · Refer F	2810		OPER TEMP °	$C \cdot 0 = 55$				
4.4	INI FT S	IZE · Refer	P&ID			OUTLET SIZE	Refer P&ID	•			
4.2								**			
4.1				GAS / RI NG		FLOW RATE S		· Refer P	AID.		
4.0 4 1	SCRUP		STICING :					NE			
3.0	VENDOR					QUANTITY : As	per P&ID				
2.0	JOB NO	.:				ITEM NO. : SC-					
1.0	PROJEC	<b>) T</b> :				CLIENT :					

#### DATASHEET OF PRESSURE GAUGE

Project :- \*

Client :- \*

**Contractor :- \*** 

FOA	No. :- *							
1	Туре	Direct				Wetted parts material		
2	Mounting	Local				Element		
3	Dial size	150 mm				Lower body		
	Colour	White (N	on rusting	plastic		Non wetted		
		with blac	k engraving	g)		parts		
4	Case material	DIE CAS	T			Process		
		ALUMIN	NUM / SS			connection		
5	Bezel ring	Screwed	/ Bayonet			Size		
6	Window material	Shatter pr	roof glass			Rating		
7	Enclosure	Min. IP 5	5 / NEMA	4		Facing & Finish		
8	Pressure element	Bourdon	tube			Capillary		
	<b>F1</b> 1	00.016				material		
9	Element material	SS 316				Capillary		
10	Socket material	SS 316				Flushing &		
10	Socket material	55 510				Filling		
11	Accuracy	± 1% FSI	D		16	Over range	130% of range	
	_				10	protection	_	
12	Zero adjustment	Microme	ter pointer		17	Blow out	Required	
		(Internal)			17	protection		
					18	Options		
13	Connections	1⁄2" NPT	(M)			a)	Snubber	
	Conn. Location	Bottom				b)	Syphon	
14	Movement	SS 304				c)	Gauge saver	
15	Diaphragm seal					d)	Liquid filled casing	
	Туре				19	2-valve, 3-way	Required	
						manifold		
					20	Make & Model	*	
т	N. Dana	Press	ure	Desi	ign	F1: J	Leastian	Ontinus
Tag	No. Kange	Operating	Design	Ten	ıp.	Fluid	Location	Options
					Re	fer P&ID		
	-							

#### NOTES:

1) **\*\*** Information to be supplied by the Vendor / Contractor.

2) Make of the PG shall be from approved vendor list of Mecon /Client.

3) Detailed technical catalogue for offered model to be submitted by vendor/contractor during approval.

4) Offered Model shall meet the tender requirement.

0	NB	VJ	RKS	INSTRUMENTATION DATASHEETS	MECON LTD. DELHI
Rev.	Prpd. By :	Chkd. By :	Appd. By :		DS No: MEC/STD./ Pa <b>g5/1524/D\$</b> 644

#### **DATASHEET OF TEMPERATURE GAUGE (with Thermowell)**

#### Project :- \*

Client :- \*

### Contractor :- \*

FOA No	). :- *										
			GE	NERAL			FILLED	SYSTEM			
1	Туре			FILLED SYSTE	EM / BIMETALLIC	15	SAMA Class	V B			
2	Well			REQUIRED			Compensati	on CASE			
3	Mount	ting		LOCAL		16	Bulb type	ADJUST UNION	<b>TABLE</b>		
4	Dial si	ize		150 mm			Bulb mater	ial 316SS			
5	Colou	r		WHITE (Non ru black figs.)	sting plastic with	17	Bulb union threaded to	0 <sup>1</sup> / <sub>2</sub> " NPT	(M)		
6 Case material DIE CAST ALUMINIUM / SS							Extension type	RIGID			
7	Windo	ow mater	ial	SHATTER PRC	19	Bulb dia	8 mm (N	ſin)			
8	Conn.	Location	1	BOTTOM		20	Capillary material				
9	Accura	acy		±1% FSD			Armour Flexib	ole			
10     Enclosure     WEATHER PROOF TO IS2147							Armour mater				
	Enclosure class			IP 55 / NEMA 4			Capillary leng	th			
11	11   Zero adj. Screw			MICROMETER POINTER (Internal)			Overrange protection	130% O	F RANGE		
	r		BI	METAL			THERMOWELL				
12	Stem:					22	Material	SS 316			
			Туре			23	3 Construction D S'		D BAR		
		Ma	terial			24	Process connection	1 ½" FL	ANGED		
			Size			25	Gauge connection	<sup>1</sup> /2" NPT	(F)		
13	Stem of	diameter				26	Thermowell as per drg	g Drg encl	osed		
14							Options a	) LIQUID	FILLED		
							Make & Model		*		
Tag No. Range Ter			Ten	nperature ( <sup>©</sup> C)	Well Dimensions	Flange rating I			Remarks		
			Oper	aung Design	U I Dafar D&II	<u> </u>			1		
					Refer r & ll	,					

Note: 1. '\*' Information to be supplied by the Vendor / Contractor. 2. Make of the TG shall be from approved vendor list of Mecon /Client.

3. Wake frequency calculation of offered Thermowell to be submitted by the Vendor / Contractor during approval.

4. Detailed technical catalogue for offered model to be submitted by vendor/contractor during approval.

5. Offered Model shall meet the tender requirement.

0	NB	VJ	RKS	INSTRUMENTATION DATASHEETS	MECON LTD. DELHI
Rev.	Prpd. By :	Chkd. By :	Appd. By :		DS No: MEC/STD./ Pa <b>g5/425/0\$</b> 644

			DATASI	HEET OF P	RESSU	RE TRAN	SMITTER	2	
Pro	ject :- *		Diffig		<b>HE</b> SSC:			•	
Clie	nt :- *								
Con	tractor	:- *							
FO	4 No. :-	*			1				
1		GENERAL		T 1' /	1.5		MEA	SURING UNIT	
I	Function	on	Flastronia	Indicate	15	Service		Pressure	
2	Туре		Based	Smart µP	16	Element		Diaphragm	
3	Case		Mfg. Std.		17	Body Ma	iterial	Carbon Steel or be	etter
4	Mount	ing	Yoke		18	Element	Material	SS 316L	
5	Enclos	ure	Weather pro IS2147 Explosion p IS2148	oof to proof to	19	19 Process Connections		½" NPT(F)	
		Enclosure class	NEMA 4 &	z NEMA 7		Pro	ocess Conn. Locn.	Mfg. Std.	
6	Elec. A	Area Class.	Zone-I, Gr. T3, CCOE	IIA & IIB, certified	20	Over Rar Protectio	nge n	130% of Range	
7	Intrins Flame	ically safe & proof	Required						
8	Air sup	oply	N.A		21	Output N	leter	Required (W.P. &	Intr. Safe )
9	Power	supply	24 VDC		22	Mounting Accessor	g ries	Required for 2" Pip Material (SS 316)	pe Mounting –
10	Cable	entry	<sup>1</sup> /2" NPT(F)		23	Zero elev suppressi	n. & ion	Required	
11	Accura	ncy	<u>+</u> 0.025% o	of SPAN	24	2-val	ve, 3-way manifold	Required	
12	Repeat	ability	<u>+</u> 0.05%		25	Make &	z Model	*	
		TRANSMITT	ER						
13	Output		4-20  mA wire	DC, Two					
14	Trans.	Power supply	0 - 24 V D0	2					
				1	1		1		
Та	g No.	Operating Pressure	Design Press.	Design Temp.	R	ange		Fluid	Options
				R	efer P&	ID			
NO	TES:								
1. '*	'' Inforr	nation to be supplied l	by the Vend	or / Contract	or.				
2. N	lake of t	the PT shall be from a	pproved ver	ndor list of M	lecon /C	lient.			
3. C	COE Co	ertificate for offered tr	ansmitter to	be submitte	d by the	Vendor / C	Contractor d	luring approval.	
4. D	etailed	technical catalogue for	r offered mo	del to be sul	omitted b	y vendor/c	contractor d	uring approval.	
5.0	tfered N	Aodel shall meet the te	ender require	ement.			•		
6. E	nvironn	iental Enclosure with	locking arra	ngement sha	Ill be pro	vided for n	netering tra	nsmitters.	

0	NB	VJ	RKS	INSTRUMENTATION DATASHEETS	MECON LTD. DELHI
Rev.	Prpd. By :	Chkd. By :	Appd. By :		DS No: MEC/STD./ Pa <b>&amp;5/426/0\$</b> 644

#### **DATASHEET OF RTD (with Thermowell)**

Project :- \*

Client :- \*

Contractor :- \*

FOA No. :- \*

L								-			
		(	GENERA	L			13	Cable ent	ry	1/2"	NPT (F)
1	Assem	bly as per	Drg.	enclosed			14	No. of en	tries	Sin	gle
	drg.					ĺ	15	Enclosure	e type	We	ather proof to IP55, Ex-proof
							15			(CC	COE certified)
2	Туре		RTD	Class A					7	THED	MOWELL
		I	ELEMEN	Т					1	пск	AMO WELL
3	No. of	elements	Sim	olex			16	Material		SS	316
4	Calibra	ation	As p	er DIN 4376	0		17	Construction Drilled bar stock			lled bar stock
5	Eleme	nt material	Plati	num (Pt 100)	)		18	Process c	onnection	1 1/2	2" Flanged
6	Resista	ance at 0°C	100	ohms			19	Inst. conn	lection	1/2"	NPT (F)
7	7 Leads Standard					20	Thermowell as per		Drg	g. Enclosed	
8	Sheath	1					20	drg	*		
		0.I	D. 8 mr	n						TR	ANSMITTER
	Material SS 316					21	Quantity				
9	Nipple	e & Union	SS 3	16			22	Innut			
	Materi	al					LL	Input			
10	No. Ot	f wires	4 W	4 Wire			23	Output			
	•		HEAD				24	Power Su	pply		
11	Head (	Cover type	Screv	wed cap & SS	S chain		25	Mounting	5		
12	Materi	ial	Cast	Aluminium			26	Enclosure	e class		
							27	SS Tag Pla	ate	Req	uired
							20	Make &	Model	*	
							20	No.			
Temperature Well						Well		Fl	nge		
Та	Tag No Pange		Tem	perature	Dim	ensions		110	inge		Fluid
10	5 110.	Trange	Nor	Design	II	Т		Material Ratin		ace/	1 iuiu
			1101	Design	U	1		iviatel lai	Finish	l I	
						Re	efer I	P&ID			

#### NOTES:

1. '\*' Information to be supplied by the Vendor / Contractor.

2. Make of the RTD shall be from approved vendor list of Mecon /Client.

3. Detailed technical catalogue for offered model to be submitted by vendor/contractor during approval.

4. Wake frequency calculation of offered Thermowell to be submitted by the Vendor / Contractor during approval.

5. CCOE Certificate for offered RTD to be submitted by the Vendor / Contractor during approval.

6. Offered Model shall meet the tender requirement.

0	NB	VJ	RKS	INSTRUMENTATION DATASHEETS	MECON LTD. DELHI
Rev.	Prpd. By :	Chkd. By :	Appd. By :		DS No: MEC/STD./ Pa <b>ge/427/0\$</b> 644

		TEMP	ERATURE TRANS	MITTERS	;					
Units:- Flo	ow :Liquid-m3/hr,G	as-MMSCMD, Steam- kg/hr,	Pressure-Kg/cm <sup>2</sup> (G	,Temperat	ure- °C,Level/Leng	gth-mm				
1	FUNCTION	TRANSMIT	INDICATE	11	POWER SUPPLY	24 V DC				
2	ТҮРЕ	ELECTRONIC	(SMART)	12	CONDUIT CONN.	1/2" NP TF				
3	CASE	MFR STD.		13	LINEARISATION	UPSCALE DOWNSCALE				
4	MOUNTING	DIRECT ON RTD/TC	YOKE	14	ACCURACY	+ / - 0.1% FSD				
5	ENCLOSURE	Dual Chamber W. PROOF CLASS : IP - 65	EX. PROOF	15	RFI / EMI PROTECTION	REQD.				
		INTRINISICALLY SAFE		16	LOAD DRIVING CAPABILITY	600 OHMS 24 V DC				
6	AREA CLASSIFICATION	NEC, CLASS-1, DIVL1, GROUP C&D		17	MAKE	*				
7	INPUT	FROM RTD		18	MODEL NO.	*				
8	OUTPUT	4-20 mA		19 a)	OPTIONS Output Meter	WP & INTR SAFE				
9	COLD JUNCTION COMPENSATION			b)	Mounting Accessories	For 2" Pipe				
10	BURN OUT PROTECTION			c)	SS Tag Plate					
TAG NO.	Temperature Nor.	RANGE	SET	Temp. Design	SERVICE	OPTION				
TT-**	**	*	*	**	RTD (4 wire)	a, b, c				
NOTES: 1 2 3 4 5 6	OTES:         1       "** As per P & ID         2       "*' INFORMATION TO BE SUPPLIED BY VENDOR/ CONTRACTOR.         3       THE DIGITAL OUTPUT METER SHOULD INDICATE THE TEMPERATURE IN DEG. C.         4       ENVIRONMENTAL COVER TO BE PROVIDED FOR EACH TRANSMITTER (DRAWING ENCLOSED ELSEWHERE IN BID DOC.) & WITH         4       LOCKING ARRANGEMENT FOR METERING TRANSMITTERS.         5       DUAL CHAMBER ENCLOSURE SHALL BE PROVIDED         6       MAKE OF TT SHALL BE AS PER VENDOR LIST ATTACHED WITH PID DOCUMENTS.									
	1	DATA SHEET	OF TEMPERATURE	FRANSMIT	TERS	MECON LTD. DELHI				
	REV					DS No: MEC/ 05/E5/DS-TT				

DATASHEET OF DIFFERENTIAL PRESSURE GAUGE									
Project :- *									
Client :- *									
Contractor :	_ *								
FOA No. :- *									
	Туре			Direct					
	Mountin	g		Local/	Surface				
	Dial Size	e		150mm	1				
	Colour			White	with Black Nume	erals			
	Case Mate	rial		SS 316					
	Bezel Rir	ng		Screwe	d				
	Window Ma	terial		Shatter	proof Glass				
	Enclosur	e		Weathe	erproof to IP55				
	Pressure Ele	ment		Diaphg	ram				
	Element Ma	terial		SS 316					
	Socket Mate	erial		SS 316					
	Accurac	у		$\pm$ 1.5% of FSD or Better					
	Zero Adjust	ment		Micrometer Pointer					
	Connectio	on		1⁄2" NP'	ΓF				
	Connection Lo	ocation		Side/ B	lottom				
	Movemen	nt		SS 304					
	Over range pro	otection		Maxim	um static pressu	re			
	Blow out prot	ection		Requir	ed				
	5-way mani	fold		Requir	ed				
	Make & Mo	odel		*					
Tag No.RangeOperatingDesig			Design	<u>l</u>	Design	Fluid	Location		
		Pressure	sure Temperature						
			Refer	P&ID					

**NOTES:** 

- 1) **\*\*** Information to be supplied by the Vendor / Contractor.
- 2) Make of the DPG shall be from approved vendor list of Mecon /Client.
- 3) Detailed technical catalogue for offered model to be submitted by vendor/contractor during approval.
- 4) Differential Pressure Gauge shall be suitable for the maximum static pressure in both the legs.
- 5) Tapings (3/4" sw) for the DPG connections shall be provided on the inlet and the outlet nozzle of the gas filters.
- 6) Offered Model shall meet the tender requirement.

0	NB	VJ	RKS	INSTRUMENTATION DATASHEETS	MECON LTD. DELHI
Rev.	Prpd. By :	Chkd. By :	Appd. By :		DS No: MEC/STD./ Pa <b>g5/429/0</b> \$644

	DIFFERENTIAL PRESSURE TRANSMITTERS WITH LOCAL DISPLAY											
<b>TT 1</b>												
Units:-	Flow : Li	iquid-m	$\frac{1}{CENE}$	as- MMSCMI	) Steam- kg/l	1r Press	sure-	$Kg/cm^2(G)$	Temperat	ure DI	- °C Level/Length-	mm
			GENE	KAL	adianta mitte		MEASURING UNIT					
1	Function			Local Display	dicate with	24	Service		Diff. Pressure			
2	Туре			Electronic Sma	art µP Based	25	Elen	nent			Diaphragm	
3	Case			Mfg. Std.		26	Body	y Material			Carbon Steel	
4	Mounting	ç.		Yoke		27	Elen	nent Materia	1		SS 316L	
5	Enclosure	2		Weather proof Explosion proo	to IS2147 of to IS2148	28	Proc	ess Connect	ions		<sup>1</sup> /2" NPT(F)	
	E	nclosure	e class	NEMA 4 & NI	EMA 7			Proces	s Conn. Loc	en.	Bottom	
6	Elec. Are	a Class.		Zone-I, Gr.IIA	& IIB, T3	29	Diap	hragm Seal:	-		Not Requi	red
7	Intrinsical Flamepro	lly safe	&	Required					Ту	pe		
8	Air suppl	v		N.A				Wett	ed Parts Ma	ıtl.		
9	Power su	pplv		24 VDC				(	Other Mater	ial		
10	Cable ent	rv		<sup>1</sup> / <sub>2</sub> " NPT(F)			Proc	ess Conn.:-				
11	Accuracy	5		$\pm 0.075\%$ of S	PAN			Si	ze and Rati	ng		
12	12 Self Diagnostics Facility Required					Faci	ing and Fini	sh	1			
		TI	RANSM	IITTER			Capillary Material:-					
13	Output			4 – 20 mA DC	, Two wire		Armour Flexible					
14	Trans. Po	wer sup	ply	24 V DC			Armour Flexible Matl.					
		С	ONTRO	DLLER			Capillary length, mm					
15	Output						Flush / Filling Conn. with plug					
16	A/M swit	ch							MISCA	LL	ANEOUS	
	No	o. of pos	itions			30	Over	Range Prot	ection		130% of Range	
17	Set Point	Adj.				31	Opti	ons				
18	Manual R	legulato	r							a)	Intrinsically safe dig	ital Output
19	Mode									b)	5-way manifold-SS3	16 Body &
			RECOR	DFR						c)	Mounting accessorie	s for 2" Pipe
				U LIC						•)	Mounting – Material	(SS 316)
20	Chart									d)	Local display in Kg/	cm2 g
21	Chart Dri	ve								e)	SS Tag Plate	
22	Moving P	arts Ma	tl.			32	Load	I Driving Ca	pability		Not less than 600 oh	ms
23	Chart spe	ed				33	Mak	te & Mode	el		*	
Tag	No.	Diff I Kg/0 Span	Range Cm <sup>2</sup> g Set	Zero Elev. mm H <sub>2</sub> O	Zero Supp. mm H <sub>2</sub> O	Des Pre	ign ss.	Design Temp.	Control Action		Fluid	Options
DPT-**	k .	*	0 - 2	*	*	*	*	**		**	¢	a, b, c, d, e
NOTE 1) *** 2) *** 3) M	<ul> <li>NOTES:</li> <li>1) '*' - Information to be furnished by bidder.</li> <li>2) '**' As per P&amp; ID</li> <li>3) Make of DPT shall be as per vendor list attached with bid documents.</li> </ul>											

4) For installation refer MECON's installation standards.

5) Environmental cover to be provided for each transmitter.6) 5 -Way Manifold shall be provided.



	LIMIT SWITCHES					
UNIT : F	low-> Liqu	uid-M <sup>3</sup> /hr Gas-MMSCMI	D steam-kg/hr Pressure-> kg/cm <sup>2</sup> g Tempreture-> <sup>0</sup>	C Level/Length-> mm		
S.No.	D	ESCRIPTION	TECHNICAL REG	UIREMENT		
1	TYPE		Snap Action Micro			
2	Area class		IEC Zone-1, IIA, IIB, T3			
3	Limit Swit	tch & Enclosure	Weather Proof (IP65) and Flame proof (Exd)			
4	Conduit		1/2' NPTF (NOTE -3)			
5	Rating		1A @ 24 V DC			
6	Form		SPDT			
7	Quantity		One each for open & close status of SDV, valves acr	oss the USM (Quantity as per P& ID)		
8	Model No		BY VENDOR			
9 Junction Box(Ex-Proof)			One no. per SDV be provided. Limit switches duly terminated with <sup>1</sup> / <sub>2</sub> " NPT as conduit connection. All connection shall be provided with ex-proof cable gland.			
	NOTES					
	1	Vendor to furnish model I	No. and decoding detailsof limit Switch accompanied	with relevant catalogues		
		(in English) literatures.				
	2	Hazardous Certificates wit	h model No. shall be furnished along with offer			
	3	Flying leads are not accept	table. Cable shall be terminated upto JB			
1 DATA S			SHEET OF LIMIT SWITCHES	MECON LTD. DELHI		
Rev.				DS No: MEC/ 05/E5/DS- LS		

#### SPECIAL INSTRUCTIONS TO THE VENDOR

#### (FOR FLOW COMPUTERS - FC)

The purpose of this specification is to define the requirements of 'Panel Mounted Flow Computers' part of the Metering Skid.

- Vendor shall be responsible for selection of the correct model nos. of instruments to meet the purchaser's specifications. In case of model no. required to change at a later date to meet the Purchaser's Specifications, the same shall be done by the vendor without any price and delivery implications.
- 2) FC Vendor shall submit Make, Model No. with decoding details, Technical Literature and Catalogue of Flow Computer
- 3) The Flow computer along with all the accessories including Isolating IS barriers, a separate terminal strip for all SCADA signals to/from CLIENT's RTU shall be fully wired and mounted in Metering panel. (Under supplier scope).
- 4) **GSM Modem** shall be provided with each flow computer for remote data configuration, remote data uploading (previous 35 days data) and remote data monitoring.
- 5) The flow computers shall be microprocessor based, with data entry keypad and alphanumeric display with AGA firmware for natural gas flow measurement. Flow computer electronics shall be protected from industrial interferences and shall be shock and vibration proof.
- 6) Validation of Flow Computer (Flow Computation sheet comparing with standard software) confirming to latest AGA-9 (latest version) / AGA 5 / AGA-8 / GPA 2172 / GPA 2145 / AGA-7 to be submitted for all the flow Computers separately.
- 7) The flow computer shall have a facility for audit trail, web enabled and shall be compatible to calculate compressibility factor as per AGA-8 detailed method, provision for digital communication and pulse / frequency both.
- 8) The flow computer shall be certified for custody transfer applications by laboratory / institutes authorized by weights and measures authority of its country of origin such as NMI, PTB, Pigsar or other reputed International Standard laboratories such as Trans Canada Calibrations (TCC) Canada, Measurement Canada, Colorado Engineering Experiment Station Inc. (CEESI) USA., Calibration Certificates to be submitted for all the Flow Computers separately
- 9) The flow computers shall compute and display the instantaneous and totalized flow rate for each stream corrected for pressure and temperature variations. The flow computers receive data from the Gas chromatograph for calculation of compressibility factor as per AGA-8 detail method. The flow computers shall be linked to the CLIENT's RTU for providing the flow measurements of the individual stream runs and related process variables.
- 10) The flow computers shall have at least serial ports for communication purpose with RTU/SCADA, online GC and Printer, USB port / RS232 with adaptor for laptop connectivity (Port for Laptop connectivity to be provided in front side of the FC).
- 11) The power consumption of Flow Computer and other associated utilities of online system shall be considered for Backup calculation and finalization of Power requirement.
- 12) FC Vendor shall be provided all necessary assistance if required for Software Configuration, Flow Computation Checking & Serial Communication checking with RTU/SCADA, GC, & GSM modem etc.( RTU/SCADA are not in scope of vendor)
- 13) Vendor/ Supplier shall provide all necessary hardware, software, serial communication cables with connectors (for each connection 30 meter) etc. for FC connectivity to RTU/ SCADA, GC and USB port / RS232 with adaptor for Laptop. Other details required for interfacing of their Flow Computers with Purchaser's RTU/SCADA. In addition to this, the communication software shall be supplied in CD / DVD for testing the communication link.
- 14) The Flow metering parameters shall be available for future SCADA through serial communication on MODBUS. All shall be configured and available on dedicated RS 485 port.



Rev.

1

S

PECIAL INSTRUCTIONS TO THE VENDO	R
(FOR FLOW COMPUTERS - FC)	

## DATA SHEET OF PANEL MOUNTED FLOW COMPUTER

1.	а) Тур	e	:	Ele use con	ctronic microprocessor bas r Configurable Flow Compu nputations as per AGA 3).	ed, data entry key board, online ter suitable for Orifice meter (flow
	b) Mał	ĸe	:	*		
	c) Moc	lel No.	:	*		
2.	Inputs		:	a) 4	4-20 mA DC (2 Wire) su (HART Protocol) from O	perimposed with digital signal rifice Flow Meter Transmitter.
				b)	4-20 mA DC (2 Wire) so (HART Protocol) from ' representing line pressure	uperimposed with digital signal SMART' Pressure Transmitters
				c)	4-20 mA D.C. (2 wire) s (HART protocol) fro Transmitter representing	uperimposed with digital signal om 'SMART' Temperature line temperature.
				d)	RS 485/422 MODBUS spar	re.
				e)	RS-485/422/232 MODBUS	from Gas Chromatograph
				f)	Additional 4 Nos. 4-20mA digital signal (HART Proto Transmitters representing in Transmitters representing in Pressure Transmitters repre streams (In case I/Os not av shall be provided with each respective parameter indica	DC (2 Wire) superimposed with col) from 'SMART' Pressure hlet line pressure, Temperature hlet line temperature, Diff. senting DP across filters in both vailable in flowcomputer then PLC flowcomputer alongwith tion on panel).
				g)	I/Os from Water Dew point point analyzer (In case I/Os then PLC shall be provided alongwith respective param	Analyzer & Hydrocarbon Dew not available in flowcomputer with each flowcomputer eter indication on panel).
				h)	Other Standard Inputs avail	able
3.	Interfa	cing Capacity		a)	Flow computers shall be im supplied by others for feed computer through RS485/2 Detail will be provided d shall be responsible for computer with gas chroma necessary hardware, soft systems and other details re computers.	terfaced with Gas Chromatographs ing online gas composition to flow 32 MODBUS serial link. Protocol luring detail engineering. Vendor proper integration of their flow tograph. Vendor shall provide all ware etc. in vendor's supplied equired for interfacing of their flow
]	1	DATASHE	CET OF P CO	ANE	EL MOUNTED FLOW UTER	MECON LTD. DELHI

Rev.

# DS No: MEC/ 05/E5/DS-FC/02

Page 133 of 644

b) Flow computers shall be interfaced with RTU for flow metering data as well as temperature and pressure, energy flow rate and total energy, corrected volumetric flow rate and total flow, yesterday's energy total and volume total, today's running volumetric total and energy total, volume and energy integrated at 6 a.m. etc. For RTU, vendor shall finally supply as per Hardware & Protocol Details provided during detail engineering.

Keyboard capability

:

:

5.

6.

7.

Outputs

Isolation

The data entry keyboard shall have the provision to enter the following:

- a) Pressure and Temperature base factor
- b) Specific gravity and scaling factor
- c) Mole % or composition of the gas to be metered
- d) Report headings, frequency and timing of reports
- e) Selection of parameter to be displayed and on-demand printing of reports
- f) Calorific Value
- g) Flow, Pressure, temperature and density values and give compensated flow for any external conditions
- h) Gas compressibility
- i) The flow computer shall have provision to enter default values of all inputs (inclusive of gas chromatograph) low/high alarms for all inputs (inclusive of gas chromatograph) and shall be user configurable. The flow computer shall use the default values in case of any input goes beyond low/high limit.
- i) Other Standard features available
- a) 4 20 mA DC analog output for corrected volumetric flow rate, line pressure, line temperature, energy rate (4 Nos.)
- b) 4 20 mA DC analog output for corrected totalised volume and uncorrected totalised volume (in future).
- Contact Alarm outputs for unit malfunctioning, process alarm c)like low pressure etc. (3 Nos.) (In future).
- d) RS 485 serial output link for RTU / SCADA for all signals of this data sheet.
- e) USB port / RS232 with adaptor for laptop connectivity.
- f) RS-232 for Printer.
- g) RS-232 for GSM Modem.
- h) RS-485 / 232 spare.
- i) Other standard outputs available.

Power supply shall be galvanically isolated. Analog I/Os and Digital I/Os shall be opto-isolated. (Active isolators, as required, shall be included by the vendor suitable for DIN rail mounting, inside vendor supplied control panel. 24VDC for powering the isolators shall be distributed by vendor inside their cabinet. Isolating relays shall be provided for potential free contact generation from switched inputs)

<del>64</del>4

- Computations a) Volume Flow rate at standard, normal or operator specified base conditions (Sm3/ hr)
  - b) Integrated corrected volume.

1	DATASHEET OF PANEL MOUNTED FLOW COMPUTER	MECON LTD. DELHI
Rev.		DS No: MEC/ 05/E5/DS-FC/02
		Page 134 of 6

1	DATASHI	EET OF PAN COMF	EL MOUNTED FLOW PUTER	MECON LTD. DELHI
10.	Calculations standard	: a) b) c)	Volume flow rate & Total F Compressibility: AGA 8 (L Energy Rate & Integrated F	Flow: AGA Report 3(Latest) atest) Energy: AGA 5 (Latest)
		f)	Alarm log database for n cleared	ninimum 15 days alarms set an
		e)	Event log databases for last	240 parameter changes.
		d)	each contract day for up to	30 days.
		/Ł	by hour for last 30 days.	lations of selectable variables fo
		c)	Values of selected variables	s by minute for last 60 minutes an
		b)	Minimum and maximum	values of selected variables for
9.	Database	: a)	Current value of each input	and output.
			communication facility be Controller shall be provi including suitable mountin Panel)	tween Flow Computer and Flow ded with each Flow Computer ng arrangements in the Metering
		f)	Built-in Closed loop Contro (If not built-in, separ	oller (PID) functionality rate Flow Controller having
		e)	days. Archival of data for up to 1.	5 history points for 35 days.
		d)	A minimum memory to log	g 240 alarms and 240 events for 1
		c)	Super capacitor capable of without power	of storing data for three month
0.	1 Caluics	. a) b)	Parameters and program EEPROM/non-volatile men	nmed constants to stored in norv
8	Features	• • •)	Built in online diagnostics t	o detect proper functioning
		j)	All the above data shall als	so be made available by vendor in
		1)	(6 a.m.) weekly, fortnightly	w & monthly intervals etc. shall be
		:\	configurable.	urrent time) and shall be use
		h)	configurable. Today's accumulated flo	ow/energy (running total since
			current running total for integrator value at 6.00 a.n separate location (regist	the day (i.e. volume/energy) & the day (i.e. volume totalizer n. from first day) to be stored in a er) and these shall be use
		g)	Previous day's flow/energ	y (i.e. yesterday's 6.0 a.m. tota
			the basis of specific gravit % of N2 and CO2 and other Flow computer shall have AGA-Nx-19 with user	y, temperature, pressure and mole r compositions for flow computers e capability for both AGA-8 and selectable upon application
		f)	All reports shall be user cor Co-efficient of deviation fre	om ideal gas law as per AGA-8 or
		e)	Generation of Standard an	d user defined reports at printer
		(b	Linearisation of temperature	e input

Page 135 of 644

Page 136 of 644

17.	Configuration setup	:	Configuration and monitoring of operation of flow computer shall be done through configuration software running on IBM compatible PC.
18.	Ambient conditions	:	Temperature: 0 to 65 ° C. Humidity: 5 to 95 % non-condensing. Vibration: As per ISA S75-13-1989, sec 4.2 & 5.35 ECD susceptibility: Should meet IEC 801-2, level 3. Base Pressure: 1.0332 Kg/cm <sup>2</sup> Base Temperature: 15°C
19.	Enclosure	:	General purpose
20.	Mounting	:	Flush panel mounted.
21.	Tag Nos.	:	FY-**
22.	Quantity	:	**

#### GSM MODEM

a) Suitable to communicate with CLIENT's SCADA / RTU.

#### Notes:

- 1. **\*\*** Vendor to furnish.
- 2. '\*\*' As per P&ID
- 3. Separate communication port shall be provided in the flow computer for laptop & printer connectivity.
- 4. Original licensed software for retrieving the stored data, programming the Flow Computer using Laptop, software based on Windows 2000/XP shall be offered.
- 5. 2 sets of Flow Computer documentation including product literature, software/hardware manual, operating manual, maintenance instructions, Modbus addressing etc. shall be supplied.
- 6. The offered Flow Computer shall meet the requirement for Custody Transfer as mentioned in API Chapter 21.1 for audit & trial. In this regard compliance certificate from competent authority and other relevant documents shall be submitted.
- 7. **GSM Modem** shall be provided with each flow computer for remote data configuration, remote data uploading (previous 35 days data) and remote data monitoring.
- 8. The Flow Computer shall be approved for Custody Transfer by NMI / PTB / renowned Lab.
- 9. <u>Flow Computer Validation Software</u> Supplier to provide licensed Software in the name of Client for authenticating the algorithm written in the Flow Computer as per AGA-3.
- 10. Vendor shall confirm all the instruction given in "SPECIAL INSTRUCTIONS TO THE VENDOR" enclosed with these Data sheets.
- 11. The communication speed of RS485 (for SCADA) serial communication port for flow computer shall be configurable from 2400 to 19,200 bps.
- 12. The Modbus communication protocol and message structure details to be used on the RS485 serial communication port (for SCADA) for Flow Computer shall be supplied after placement of order. Vendor shall furnish all details like pin configuration and tag number

1	DATASHEET OF PANEL MOUNTED FLOW COMPUTER	MECON LTD. DELHI
Rev.		DS No: MEC/ 05/E5/DS-FC/02
		Page 137 of 64

wise MODBUS address mapping list etc. for smooth interfacing of all communication links with RTU (SCADA in future).

- 13. Vendor/ Supplier shall provide all necessary hardware, software, serial communication cables with connectors etc. in vendor's supplied systems and other details required for interfacing of their Flow Computers with Purchaser's RTU (SCADA). In addition to this, the communication software shall be supplied in CD or floppy for testing the communication link.
- 14. Vendor/ Supplier shall be fully responsible for proper integration of their supplied systems and also integration with purchaser's SCADA (RTU) systems at site and vendor shall provide all necessary assistance to purchaser's for establishing all the serial links with SCADA RTU fully functional & Operational.
- 15. The Flow Computers Terminal shall have minimum one No. of RS-485 Communication Ports for SCADA as specified in Data sheet, one no. RS-485/232 communication port for GC (Gas Chromatograph), one USB or 232 with adaptor for Laptop, One no. RS-232 for Printer. The communication protocol shall be MODBUS, the detail will be furnished during details engineering, However it shall support reading & writing as follows:
  - a) For reading function, CODE-3 or CODE-4 is required.
  - b) For writing function, GC data into the flow computer function code to be 6 (Single) and 16 (Multiple) are required.
  - c) As most of CLIENT RTU is 16 Bit registers, two registers are used for accommodating one 32-bit floating point no. Hence the flow computer (irrespective of size of the register) should be configured as 16-bit registers so that no-error is encountered in writing.
- 16. Make of FC shall be as per vendor list attached with bid documents.



### FIELD MOUNTED FLOW COMPUTER

1.	Type :	On line Solar Panel & Battery operated Microprocessor based electronic Flow Computer suitable for mounting in the field location in Hazardous area.
2.	Function :	The Flow Computer for measuring, monitoring Gas flow from single meters Run with a Gas meter. The FC measure actual gases volume, pressure and temperature and calculates compressibility factors of the gas and based on which calculates standard volume of gas. Computation of Gas Calorific values. The unit shall be complete in all respects to achieve this functionality.
3.	Inputs & Outputs :	
		a) 4-20 mA DC (2 Wire) superimposed with digital signal (HART Protocol) from Orifice Flow Meter Transmitter
		b) Temperature signal from RTD (Pt-100 – 3/4 wires) thermal element with an accuracy of $\pm 0.15$ % of measured value with thermowell. All interconnecting cable shall be screened and armoured. The cable and cabling accessories to be supplied
		by the vendor.
		<ul> <li>c) Pressure signal from built-in Pressure Sensor with an accuracy of ± 0.1 % of measured value and Range as per P&amp;ID. Pressure sensors to be individually calibrated and characteristics stored within the flow computer.</li> <li>d) Additional 4 Nos. 4-20mA DC (2 Wire) superimposed with digital sizes (UAPT, Pretacel) form Diff. Pressure Transmittees</li> </ul>
		<ul><li>e) Other Standard inputs available.</li></ul>
4.	Outputs port :	<ul><li>a) RS 232 Serial Port for PC/LAPTOP connectivity</li><li>b) RS 485 Serial Output port for SCADA facility.(Refer Note 15)</li></ul>
5.	Output Measurement:	<ul> <li>a) Corrected flow rate: Sm<sup>3</sup>/hr.</li> <li>b) Corrected Totalised volume : Sm<sup>3</sup></li> <li>c) Pressure : Kg/cm<sup>2</sup>g</li> <li>d) Temperature : °C</li> <li>e) Alarm output for unit malfunctioning</li> <li>g) Other Standared outputs available</li> </ul>
6.	Isolation :	All Inputs, Outputs and power supply shall be Individually Isolated.



#### DATA SHEET OF FIELD MOUNTED FLOW COMPUTER

Rev.

1

DS No: MEC/ 05/E5/DS/FC -01 Page 139 of 644

7.	Display :		Alphanumeric large character LCD with selectable decimal, Displaying all units, messages, alarms etc shall be
8.	Units of display :		In English. User configurable parameters like a) Corrected flow rate: Sm <sup>3</sup> /hr. b) Corrected Totalised volume : Sm <sup>3</sup> c) Pressure : Kg/cm <sup>2</sup> g d) Temperature : °C e) Energy ( Kcal) f) Yesterday's Flow ( SCMD) g) Non resettable total ( SCM) h) Days Total (SCMD) i) Co2/N2/Sp.gravity etc. etc.
9.	Power supply	:	Solar Panel (SP) Charge System with 6/12 V DC Chargeable Gelcell Battery Backup for at least 15 non sunny days and mounting Hardware for Mounting in a Hazardous area. (Refer Note 7)
10.	Configuration Setup	:	To be done in factory for all flow computer fully taking into account the process conditions, sensor & flowmeter's characteristics and calibrations for direct on site operations.
11.	Calculations standard	:	<ul> <li>a) Volume Flow calculations: AGA3 &amp; ISO 5167 (Latest).</li> <li>b) Compressibility : AGA 8 (Latest)/ GPA</li> <li>Field selectable Detailed / Gross I / Gross II Methods.</li> </ul>
12.	Features	:	<ul> <li>a) Built in diagnostics to detect proper functioning.</li> <li>b) Data security through password/key-lock facility and volume conversion and configuration to be sealed.</li> <li>c) Parameters and programmed constants shall be stored in EEPROM / non-volatile memory.</li> <li>d) Facility for entry and accessing live and stored data through external Laptop/ PC.</li> <li>e) Shall have to store at least 35 days data (on hourly basis) for flowing pressure, temperature and corrected flow with date and time stamping.</li> <li>f) Shall have to store at least last 35 days cumulative corrected flow on daily basis</li> </ul>

1

#### DATA SHEET OF FIELD MOUNTED FLOW COMPUTER

MECON LTD. DELHI

मेकान

Rev.

DS No: MEC/ 05/E5/DS/FC -01 Page 140 of 644

			<ul> <li>g) The stored data above shall be Laptops. Suitable dedicated port a flow computer for Laptops connershall be supplied.</li> <li>h) Shall have addressing facility faddress, for multi-dropping on a si ) The Time Function should be t+/- 0.01% for watchdog timer and and hour / Minute / Second forma j) Processor should be 32 bit CMF and access should be password pr k) Flow calculation shall be interpretent.</li> </ul>	retrievable by using shall be available on the ection. Software required for identification by its single telecom channel real time with accuracy d Year / Month / Day at. OS Micro -processor rotected nal selectable through	
13. Hazard	ous area	:	Certified intrinsically safe for area 1 Division 2, Groups C & D.	classification IEC Class	
14. Site co	nditions	:	Temperature –20 – 65°C (Design), saline environment. Vibration: Test 1989, Sec. 4.2 & 5.35, ECD suscep IEC- 801-2, Level 3.	Hot, humid, tropical, ted as per ISA s75-13- tibility: Should meet	
15. Enclose	ure	:	Weather Proof to IP 55 / NEMA 4 for mounting in Hazardous area as	and Compatible mentioned above.	
16. Mounti	ng	:	Outdoor		
17 Accuracy		:	0.5 % or better (Bibber shall categorically indicate the system accuracy i.e. overall accuracy considering flowmeter, PT, RTD etc.)		
18 Accessories		:	<ul> <li>a) 5 way Valve manifold –SS( ½" NP Connection.</li> <li>b) Solar Panel to Flow Computer Con (Copper cable) Minimum 10 mete</li> <li>c) 2 " pipe mounting brackets for Flow and Solar Panel.</li> </ul>	PTF) for process nnecting cable rs. w Computer (FC)	
19. Make		:	*		
20. Mode	l No.	:	*		
21. Quant	iity	:	As per P&ID.		
22. Tag n * By Vendor (	o. Vender must furnish o	: decod	FQI– xxxx (xxxx - As per P&Id) ing details & catalogue with model no.	.)	
1 DATA SHEET		FIEI	LD MOUNTED FLOW COMPUTER	MECON LTD. DELHI DS No: MEC/ 05/E5/DS/FC -01	

#### 22. **GSM MODEM**

- a) Suitable to communicate with Client's SCADA.
- b) Suitable for mounting in the Hazardous area.

#### Notes:

- 1. Original licensed software for retrieving the stored data, programming the flow computer using portable PC (Laptop), software based on Windows 2000/ XP shall be supplied in the form of CD's for each skid as per MR.
- 2. All the Hardwares and softwares to be supplied shall comfirm the Engineering Units as mentioned in these specifications.
- **3.** Flow computer documentation including product literature, software/hardware manual, operating manual, maintenance instructions, Certificates etc. shall be supplied as per vendor data requirement.
- **4.** GSM Modem shall be provided in the flow computer for remote data configuration, remote data uploading (previous 35 days data) and remote data monitoring.
- 5. Following features shall be available in Field mounted flow computer:
  - i) Energy measurement in functions and display.
  - ii) Memory allocations shall be as given below:
    - a) 512 K x 16 flash memory with non volatile copy of program code.
    - b) Programmable peripheral chip with EEPROM contains boot ladder code.
    - c) 256 K of SRAM with copy of program code and data/exact logs.
    - d) Flash memory preservation shall be achieved by 10 year data unpowered retention.
    - e) SRAM memory preservation shall be battery backed by Lithium Cell when main power is lost.
- f) Bidder / Supplier shall submit the Battery sizing calculation, Details of battery & Solar Panel, Product Technical Literature along with offer. The Rating, make and other necessary details in support of selected model for Battery & Solar Panel shall also be submitted. The power consumption of Flow Computer, Modem, PT and other associated utilities of online system shall be considered for Backup calculation and finalisation of sizing of Battery.
- 6. The installation of flow computer, its batteries, solar panel shall be in safe area (appx 15 feet) or in separate canopy outside the main skid canopy with the necessary cables and accessories shall be done. This shlold be as per the IGE/TD/13 guidelines. The FC should be visible from out side.
- 7. Field Mounted Flow Computer shall store previous 35 days data on hourly basis and cumulative corrected flow on daily basis.

1	DATA SHEET OF FIELD MOUNTED FLOW COMPUTER	MECON LTD. DELHI
Rev.		DS No: MEC/ 05/E5/DS/FC -01 Page 142 of 644

- 8. Conduit connection of <sup>1</sup>/<sub>2</sub>" NPTF with plugs shall be provided for output connection from Flow Computer.
- **9.** Bidder/Supplier shall configure and update records in Flow Computer as per technical requirement and data sheet before Final Inspection call. The process parameter and the required measuring units are already specified in Data sheets/ Tender documents and it shall be made available in Flow computer. All the specified function and features shall be demonstrated during the Final inspection.
- **10.** Bidder shall confirm all the instruction given in "SPECIAL INSTRUCTIONS TO THE VENDOR" enclosed with these Data sheets.
- **11.** The communication speed of RS232 (for SCADA) serial communication port for flow computer shall be configurable from 2400 to 19,200 bps.
- 12. The Modbus communication protocol and message structure details to be used on the RS232 serial communication port (for SCADA) for Field mounted Flow Computer shall be supplied after placement of order shall furnish all details like pin configuration and tag number wise MODBUS address mapping list etc. for smooth interfacing of all communication links with RTU (SCADA in future).
- **13.** Bidder/ Supplier shall provide all necessary hardware, software etc. in vendor's supplied systems and other details required for interfacing of their Flow Computers with Purchaser's RTU (SCADA). In addition to this, the communication software shall be supplied in CD for testing the communication link.
- 14. Bidder/ Supplier shall be fully responsible for proper integration of their supplied systems and also integration with purchaser's SCADA (RTU) systems at site and vendor shall provide all necessary assistance to purchaser's for establishing all the serial links with SCADA RTU fully functional & Operational.
- **15.** The Flow Computers Terminal shall have minimum 1 Nos. of RS-485 Communication Ports for SCADA as specified in Data sheet, one no. RS-485/232 communication port for GC (Gas Chromatograph) OR for Laptop PC. The communication protocol shall be MODBUS and shall support reading & writing as follows:
  - a) For reading function, CODE-3 or CODE-4 is required.
  - b) For writing function, GC data into the flow computer function code to be 6 (Single) and 16 (Multiple) are required.
  - c) As most of CLIENT RTU is 16 Bit registers, two registers are used for accommodating one 32-bit floating point no. Hence the flow computer (irrespective of size of the register) should be configured as 16-bit registers so that no-error is encountered in writing.



## SPECIFICATIONS FOR LEL DETECTION SYSTEM

SERVICE	:	To detect the presence of Natural Gas in the various locations in Terminal area.
NO. OF DETECTORS / COMPOSITION	:	Refer Data Sheet enclosed for quantity requirements.
ALARMS	:	To be available in the respective Control Rooms.
RELAY CONTACT FOR EACH CHANNEL	:	Required
RELAY CONTACT FOR GROUPING	:	Required


# DATA SHEET

1) SENSOR:		
TYPE	:	Poison resistant, infrared absorption type for combustible gas or vapours. (Non-selective)
CONSTRUCTION	:	Explosion proof, SS 316 body with stainless steel flame arrestors, dust cover, weather proof as per NEMA 4X or IP 65 (NOTE-1). Sensor element shall be replaceable type / plug in type. Terminal box shall be provided for further cabling. Flying leads shall not be offered. Cable entry shall be <sup>3</sup> / <sub>4</sub> " NPT (F) (NOTE-2).
ELECTRICAL AREA CLASSIFICATION	:	Zone –1, Gr. II A & II B, T3
RANGE	:	0-100 % LEL
ACCURACY	:	$\pm1$ % of gas reading @ ${\leq}50\%$ LEL, $\pm5$ % of gas reading @ > 50% to 100% LEL
MOUNTING ACCESSORIES	:	Vendor to indicate.

NOTE 1: Splash guard & weatherproof cap for protection against dust and rain shall be provided.

**NOTE 2**: The junction box shall be weatherproof to IP 55 and certified to Explosion proof.

1	DATASHEET OF LEL DETECTION SYSTEM	MECON LTD. DELHI
Rev.		DS No: MEC/ 05/E5/DS-LEL Page 145 of 644
		Page 2 of 6

# 2) GAS MONITOR / CONTROLLER:

TYPE :	Solid state electronic, panel mounted plug in modular construction			
NO. OF CHANNELS:	No of Channels shall be as per attached sheet with this specification. However, single loop integrity shall be maintained by one monitoring card per sensor. Single Channel Monitors shall be offered.			
POWER SUPPLY :	24 V DC Monitor / Controller shall supply power to sensor.			
RANGE AND : METER CALIBRATION	Adjustable between 0-100 % LEL			
INDICATORS :	Required 0 –100 % LEL.			
INDICATOR LIGHTS: (Separate for each channel)	Power on by a confidence flash for every 2 seconds. 20 % LEL 40 % LEL Malfunction (malfunction shall include short cut. line breaking, over range and earth fault.)			
OTHER FEATURES :	a) Dual adjustable set points for gas level alarm.			
	b) SPDT latching type relay contacts shall be provided for following:			
	i) 20 % LEL (each channel)			
	-alarm high (adjustable)			
	ii) 40 % LEL (each channel)			
	-alarm very high (adjustable)			
	iii) Malfunction (common)			
	iv) 20 % LEL (Common)			
	-alarm high (adjustable)			
	v) 40 % LEL (Common)			
	-alarm very high (adjustable)			
	vi) Two spare potential free outputs for			
	future use.			
	c) Relay contacts shall be suitably rated for the specified power requirements.			
	d) Reset push button (common)			
	e) Sensor calibration without alarm outputs shall be achieved by			



inhibit switch on relevant channel.

MOUNTING	:	Monitors / Controllers shall be supplied mounted in 19" racks with screwed terminals at the back. However 19" racks shall be fully wired-up in ready to use condition. Dummy plates shall be provided on spare channel slots wherever required (20 % spare slot shall be provided wherever applicable).
CALIBRATION	:	Set of fully equipped calibration gas kit consisting of at least a cylinder / bottle of calibration gas for minimum 6 months use (of known mixture of air and gas ), a pressure regulator with gauge, flexible tube / hose, adapter cap (to fit sensing head ) shall be supplied to enable online calibration of LEL detectors.

NOTE:

- 1. The Bidder shall indicate distance limitation if any between Sensor and Monitor.
- 2. RS 485 port shall be provided in monitor / controller for RTU / SCADA.
- 3. All the Digital signals from monitor / controller shall be terminated upto RTU / SCADA TB.



### SPECIAL INSTRUCTIONS TO VENDOR FOR LEL DETECTION SYSTEM

1) Vendor to note that the offered instruments shall be suitable for the following ambient conditions :

Temperature Range	:	48°C (Max.)
Humidity	:	99% (Max.)
Rainfall	:	1000mm (Max.)

- 2) Power Supply shall be from the respective control panel to each monitor rack at each terminal, Power distribution is in the Vendor's scope. If conversion to DC level is required, the system shall be based on dual redundant power packs.
- 3) Vendor shall furnish the statutory certificates of explosion proof for both LEL detector and the junction boxes along with offer.
- 4) Vendor to note that the pellister of the detector must be poison resistant to the normally known pollutants present in a hydrocarbon processing industry including sulphides, SO<sub>2</sub>, chlorides, dust etc.
- 5) Mounting details of the LEL detectors shall be furnished.
- 6) Adjustable set points shall be provided at the monitors.
- 7) Calibration gas suitable for the calibration of LEL detectors shall be included in the offer.
- 8) Mute circuits with acknowledge and reset buttons shall be provided.
- 9) Detailed configuration diagram showing connection of monitor etc. along with detailed catalogues, manuals etc. shall be furnished along with offer.
- 10) Time period required between successive calibrations of the LEL detectors shall be furnished in the offer.
- 11) Commissioning Spares for all the items shall be included in the offer. The vendor shall furnish the detailed list of the commissioning spares in the offer

1	DATASHEET OF LEL DETECTION SYSTEM	MECON LTD. DELHI
Rev.		DS No: MEC/ 05/E5/DS-LEL Page 148 of 644
<b>F</b>	·	Page 6 of 6

# LEL SYSTEM FOR EACH PANEL MOUNTED FLOW COMPUTER METERING SKID

- 1) FLAMMABLE GAS DETECTORS.
- 2) EXPLOSION PROOF JUNCTION BOXES.
- 3) WEATHERPROOF CAP.

QUANTITY (Sensors)	:	4 NOS.
MONITORS / CONTROLLERS	:	1 Set
NO OF CHANNELS	:	8 NOS. (With Common Indicating Meter)
MONITORING CARDS	:	4 NOS.
CALIBRATION KIT	:	1 Set.



Page 5 of 6

# SPECIAL INSTRUCTIONS TO THE VENDOR

### 1. Scope

The Gas Chromatograph is a part of the metering skid and shall be supplied along with respective metering skids. The GC Vendor along with skid Vendor shall execute this part.

- a) Technical specification, Data sheets attached in the bid document, covers the minimum requirements for the complete design, engineering, material selection, manufacturing, supply, nameplate marking, inspection, testing, shipping, erection, testing and commissioning of On-Line Gas Chromatographs with all accessories for Metering Skid. Controller along with Zener Barriers/Isolators shall be fully wired and mounted in the same metering Panel to be installed in the Control Room. Printer with stand (if not an integral unit with GC) along with connecting cables, shall be loose supplied. The Sampling System, Gas Analyser, Calibration, and Carrier Gas Cylinders all will be on skid mounted in field and suitable for installation in hazardous area and specified weather condition. For detail scope of work refers Bill of Material (Indicative) mentioned elsewhere in this document.
- b) In the event of any conflict between technical specification, data sheets, related standards, codes etc. the Vendor should refer the matter to the purchaser clarification and only after obtaining the same should proceed with the manufacture of the items in question.
- c) Purchaser's data sheets indicate type, minimum sampling system requirements and material of construction for Gas Chromatograph and its sampling system. However, this does not absolve the Vendor of the responsibility for proper selection with respect to the following:
  - ii. Proper design of the sampling system and gas chromatograph to measure the component of interest to the stated accuracy.
  - iii. Selection of materials for all the parts of the gas chromatograph system so as to be compatible with the process stream and surrounding atmosphere as per purchaser's data sheets.
- d) Vendor shall note that the offered instruments shall be suitable for the following ambient conditions:

Temperature Range: 65 °C (MAX), 5 - 40 °C (operating) Humidity : 100 % (max)

- e) Vendor shall be responsible for selection of the correct model no. of instruments to meet the specifications contained in the spec. In case of model no. required to be changed at later date, the same shall be done by the Vendor without any price or delivery implications.
- f) Offer shall be clear, unambiguous (with no alternatives) and complete with all data sheets, catalogues etc.
- g) Gas Chromatograph shall be connected to the Flow Computers (the scope of SKID Vendor). Hence suitable interface & integration with Flow Computer is under Vendor's scope.
- h) Separate Analyser Cabin, necessary Sample Handling System, 20D straight run shall be provided as per the ISO10715. Tubing from main pipeline to Analyser Cabin (approximate distance between Analyser Cabin and Sampling Point on the pipelines will be min. 50 mtr.), Bottles and Skid for mounting are under Vendor's scope.

Rev. Doc. No.: MEC/05/E5/GC-SIV	1	SPECIAL INSTRUCTIONS TO THE VENDOR (FOR GAS CHROMATOGRAPH - GC)	MECON LTD. DELHI	
Dage 150 of 644	Rev.		Doc. No.: MEC/05/E5/GC-SIV	

e) Vendor shall provide certificate for all the tests indicated in FAT. In addition, Vendor shall provide the 'Manufacturer's certificate of Conformity' to purchaser's specifications as per clause 2.2. of DIN 50049.

# 2. Spare Parts:-

- a) Consumable as a part of the offer.
- b) Commissioning spares as a part of the offer.
- 3. Special Tools and Tackles for Operation & Maintenance of GC. List must be furnished along with the offer. (if required)
- 4. All units of measurement for various items in Vendor's specification sheets shall be to the same standards as in purchaser's data sheets. All instruments shall be graduated in the same engineering units as indicated in purchaser's data sheets.
- 5. All the materials specifications for various parts in the Vendor's specification sheet shall be to the same standards as those in purchaser's data sheets e.g. ASTM, BS etc.
- 6. Vendor shall not offer any prototype instruments in his bid. All instruments offered should have been proven in refinery or petrochemical plants before bidding.

# 7. Drawings & Data

Vendor shall submit all the datasheets/drawings/documents as indicated below,

# a) Along with the offer

Vendor's Offer shall include a detailed specification sheet for Gas Chromatograph system which shall contain the following information:-

- i. All the details regarding type, construction, materials, accessories etc. of the On-line Gas Chromatograph along with technical catalogue.
- ii. Detailed sketch showing various components of sampling and sample conditioning system.
- iii. Sample flow rates required achieving response time, the recommended length and size of the sampling tube between process tap and sampling system.
- iv. Any special cabling requirements including shielding and grounding requirements and maximum permissible distances of separation between the field and the control room mounted units etc.
- v. A Copy of the Certificate of intrinsic safety/ flameproof from statutory body like BASEEFA, FM, PTB, CMRI etc. as applicable.
- vi. Consumption figures of electrical power and other utilities for each gas chromatograph system.
- vii. Overall dimensions of major units dimension of the skid to be mounted in field.

# b) After Placement of Order

Vendor shall submit certified drawings and specification sheets for Gas Chromatograph which shall include the following (\* marked shall be submitted for approval)

1	SPECIAL INSTRUCTIONS TO THE VENDOR (FOR GAS CHROMATOGRAPH - GC)	MECON LTD. DELHI
Rev.		Doc. No.: MEC/05/E5/GC-SIV
		rage 151 01 044

- i. \* Bill of Material of the complete G.C system.
- ii. \* G.A Drg. With overall Dimension of both Sampling unit & Panel.
- iii. Sampling interconnection details identifying each component with make and model number.
- iv. Detailed interconnection diagram for process, piping and tubing. (Hook up Diagram)
- v. Installation drawings.
- vi. Calibration curves as required for each Gas Chromatograph system.
- vii. Cabling details including shielding / grounding requirements.
- viii. Utility consumptions.
- ix. Technical Literatures, Installation & Commissioning Manuals for all supplied items.
- x. Quality Assurance Plan.
- xi. Software required for the systems.
- xii. Wiring Diagram of Panel.
- xiii. Cable Parameters (Capacitance, inductance, L/R ratio) for interconnection cable between field and the control room mounted units for intrinsically safe system.
- xiv. The details of the heat load, humidity particulate / chemical filtration etc. of the system.
- xv. Vendor shall provide certificate for the entire test indicated in FAT. In addition, vender shall provide the "manufacturer's certificate of Conformity" to purchaser's specifications as per clause 2.2 of DIN 50049.

# 8. Indicative Bill of Materials for On-Line Gas Chromatograph

# **Field Mounted Items:**

- a) Analyser for Gas Chromatograph with Necessary Certificate (CENELEC, CE, NMI etc.), 230/110 VAC / 50 Hz, One steam / one Cal., Power and signal cables and all accessories.
- b) Sample probe (SS) (inlet <sup>3</sup>/<sub>4</sub>" outlet <sup>1</sup>/<sub>4</sub>" or as recommended) with SS Relief valve and pressure gauge, Threadolet 3000 # material ASTM A-105 (as per ANSI B 16.5) thread as per ANSI B 1.20.1, necessary assembly consisting of Ball valve, manometer and Relief Valve etc.
- c) Sample line (1/16" or 1/8" SS tubing in protective Hose.) minimum 30 mtr.
- d) Sample Conditioning Cabinet with all accessories, equipments, PRV, Flow meter, PG, filters, Moisture Trap etc. and fittings.
- e) Mounting Skid for of Analyser, sampling conditioning cabinet and the bottle stand with vent system of 3 mtr. An explosion proof Junction Box with Switches for signal & power supply along with necessary power & signal cables and glands.
- f) Bottle (duly filled for Calibration Gas, Carrier Gas) & Bottle Stand with Bottle pressure regulators, Automatic switchover system for Carrier Gas, Heating plate etc.
- g) All Cables, Glands, Terminals, Tubing, structural items, Panel accessories required for installation of the system is in vendor's scope.
- h) Dedicated earthing terminal has to be provided on the skid.

# **Control Room Mounted Items:**

a) Programmable Control Unit with all inputs/outputs and serial / communication ports as mentioned in Technical Specification.

 
 1
 SPECIAL INSTRUCTIONS TO THE VENDOR (FOR GAS CHROMATOGRAPH - GC)
 MECON LTD. DELHI

 Rev.
 Doc. No.: MEC/05/E5/GC-SIV Page 152 of 644

- b) G.C controller inside metering Panel along with all accessories like Zener Barriers/Isolators, AC/DC Converter, Power Supply unit, etc, as required to be installed in metering Panel.
- c) All required Signal Cables (armored) / Fiber Optic Cables, Power Cables and RS 232 Communication Cable (min 10 -15 Mtr) for each computer and Cable Glands.
- d) Colour Desk Jet Printer with printer table including all the cables (If not the integral part of GC).
- e) The Flow computer and GC communication (on serial port) and further for SCADA if required in the scope of vendor.



# DATA SHEET OF ON-LINE GAS CHROMATOGRAPH

1.	SER Tag N	VICE COND	ITIONS :	AE - **		
2.	Quan	tity	:	**		
3.	Gas (	Composition	:	Refer Bi	d Documents	
4.	Gas I	Pressure (Oper	cating) :	**		
5.	Gas p	pressure (Desi	gn) :	**		
6.	Gas 7	ſemp ( Norma	l) :	**		
7.	Gas 7	Temp. (Design	ı) :	**		
8.	Mois	ture Content	:	Dry Gas		
9.	Line	Size / Rating /	Sch. / Flow	· · **		
Tag	No.	Line Size	Rating	Sch.	Max. Flow (MMSCMD)	Remarks
AE	_**	**	**	STD	**	

 Power Supply
 :
 230/110 V, 50 HZ

12. Instrument Air : Not Available

'\*\*' As per P& ID



# DATA SHEET FOR LAPTOP (TYPICAL)

- 1. All components / peripherals of the Laptop should be from OEM only. Authorised distributers doing local integration are disqualified and rejected.
- 2. The system offered should be modular in design with reference to both hardware / software to enable future upgradation and expandability. Only latest proven commercial realease of the hardware and software should be offered.
- 3. The Laptop shall be supplied with all necessary power, communication cables with connector of main supplier and suitable to the Indian standards.
- 4. Technical literatures / Manuals / user instructions.
- 5. The mimimum Specification of the Laptop shall be as follow:
  - 1. CPU : Intel Dual Core, 2.0 Ghz integrated cache
  - 2. Memory : RAM 1 GB DDR2 (Min.)
  - 3. HDD : HDD 160 GB (Min.)
  - 4. DVD / CD RW COMBO DRIVE
  - 5. Monitor 15"/ 17" TFT screen
  - 6. Intel Graphics Media accelerator 900 with up to 128 MB Snared Video Memory
  - 7. Integrated wireless Blue tooth
  - 8. Muiltimedia
  - 9. 56 K modem
  - 10. OS : WIN XP Professional (Licensed)
  - 11. Original Lisensed Antivirus Software & MS office
  - Port required (Min.): 1 serial, RJ 11, 3 USB port, RJ 45 -10/100/1000T Gigabit NIC, 9 Pin D connector serial ports (one male & one female) and 15 pin serial connector.
  - 13. Laptop Bag with all accessories.
  - 14. Warranty : 1 year OEM standard
  - 15. Make : Compaq / Dell / IBM / HP



Rev. :	0	
Edition	: 1	

# **TECHNICAL SPECIFICATION**

# FOR

# **GAS POWERED VALVE ACTUATORS**

# SPECIFICATION NO.: MEC/TS/05/E5/002A



# ELECTRICAL & INSTRUMENTATION (OIL & GAS SBU) MECON LIMITED DELHI 110 092

03.01.2015	Lakshi	ajain	Shulf
	Sakshi Wadhawan	Vikas Jain	Rakesh Kr. Shukla
Date	Prepared By	Checked By	Approved By

MECON LIMITED	STANDARD TECHNICAL		
REGD. OFF: RANCHI 834002	ELECTRICAL & INSTR OIL & GAS SBU		
	SPECIFICATION FOR	DOCUMENT NO.	Page 2 of 15
TITLE	GAS POWERED VALVE ACTUATORS	MEC/ TS / 05 / E5 / 002A	REVISION : 0
			EDITION : 1

# CONTENTS

- 1.0 SCOPE
- 2.0 REFERENCE DOCUMENTS
- 3.0 ACTUATOR SIZING
- 4.0 DESIGN FEATURES
- 5.0 COATING
- 6.0 INSPECTION AND TESTS
- 7.0 NAME PLATE
- 8.0 SHIPPING
- 9.0 DOCUMENTATION
- 10.0 SPARES AND ACCESSORIES

# ATTACHMENTS

- SKETCH-1
- SKETCH-2
- GAS COMPOSITION (attached elsewhere in tender document)
- PHILOSOPHY FOR AUTO-CLOSURE OF VALVES

	STANDARD TECHNICAL			
REGD. OFF: RANCHI 834002	Instruction     Electrical & INSTRUMENTATION       834002     OIL & GAS SBU, DELHI			
	SPECIFICATION FOR	DOCUMENT NO.	Page <b>3</b> of 15	
TITLE	GAS POWERED VALVE ACTUATORS	MEC/ TS / 05 / E5 / 002A	REVISION : 0	
			EDITION : 1	

## 1.0 <u>SCOPE</u>

- 1.1 This specification, together with the applicable data sheet, covers the minimum requirements for the design, manufacture, inspection, testing and shipping of valve actuators complete with accessories for quarter turn ball/ plug valves.
- 1.2 This specification indicates the minimum supply requirements and does not relieve the vendor from his responsibilities concerning the design and the safe operation of the supplied equipment.
- 1.3 The valve manufacturer shall also be responsible for proper operation of the actuator that shall develop a torque or a thrust sufficient to conveniently open and close the valve always with in the limits established by the resistance of the mechanical elements of the valve itself.
- 1.4 Valve and actuator shall be supplied as a single assembly complete in all respect and ready for installation at site.

## 2.0 <u>REFERENCE DOCUMENTS</u>

Reference is made in this specification to the latest edition of the following codes, standards and specifications:

#### **Codes and Standards**

ASME B31.8	:	Gas Transmission and Distribution Piping Systems	
ASME B16.5	:	Steel Pipe Flanges and Flanged Fittings	
ASME Sec. VIII	:	Boiler and Pressure Vessels, Code.	
ANSI B2.1	:	Pipe Threads, General Purpose	
NEC	:	National Electric Code	
IEC	:	International Electro-technical Commission	
NEMA	:	National Electrical Manufacturers Association	

In case of conflict between the requirements of this specification and above referred documents, the requirements of this specification shall govern.

	STANDARD TECHNICAL		
REGD. OFF: RANCHI 834002	ELECTRICAL & INSTR OIL & GAS SBU	मेकान 8001 Corv	
	SPECIFICATION FOR	DOCUMENT NO.	Page 4 of 15
TITLE	GAS POWERED VALVE ACTUATORS	MEC/ TS / 05 / E5 / 002A	REVISION : 0
			EDITION : 1

# 3.0 <u>ACTUATOR SIZING</u>

- 3.1 For sizing the actuator, valve manufacturer shall furnish to the actuator manufacturer the following information:
  - a) The maximum break away torque or thrust required at the valve stem with manual/ remote operation to open and close a valve at the shut off pressure in the line. The actuator shall be sized at pressure indicated in Table-2 of datasheet and meeting the safety factor of 1.25 as required by the valve operation at the shut-off pressure in the line.
  - b) The temperature correction factor.
  - c) The pressure correction factor.
  - d) As a result of points (a) to (c) the minimum required torque or thrust output of the actuator.
  - e) Maximum allowable torque or thrust output of the actuator depending on the type and size of valve.
- 3.2 The maximum time required to open and close a ball valve shall be as indicated in the data sheet.
- 3.3 Actuator manufacturer shall provide the complete model no. decoding for actuators, limit switches and solenoid valves.
- 3.4 Complete details of Gas/ Hydraulic circuit with complete sequencing of port from open to close and close to open position shall be furnished in the offer for review.
- 3.5 Manufacturer shall furnish the detailed calculation for actuator sizing after placement of order. The calculation so furnished by manufacturer shall satisfy the sizing criteria as per above clauses. Manufacturer shall agree to upgrade the actuators offered to meet the sizing criteria without any price and schedule impact.

#### 4.0 DESIGN FEATURES

4.1 The actuators shall be powered by Natural Gas from the main pipeline. The gas powered actuator shall operate at pressure indicated in Table-2 of datasheet. A typical scheme for tapping the gas from mainline with pressure reduction is shown in Sketch-1. Actuator electrical/ hydraulic circuit shall be developed as per the Sketch-2 enclosed.

	STANDARD TECHNICAL				
REGD. OFF: RANCHI 834002	ELECTRICAL & INSTR OIL & GAS SBU	Hall			
	SPECIFICATION FOR	DOCUMENT NO.	Page <b>5</b> of 15		
TITLE	GAS POWERED VALVE ACTUATORS	MEC/ TS / 05 / E5 / 002A	REVISION : 0		
			EDITION : 1		
4.2	Actuator shall be direct gas operated scotch yoke type with manual hydraulic override. The actuator shall be provided with a hydraulic pump for manual operation. If there is no gas pressure available to actuator it shall be possible to actuate Direct Pneumatic Actuator by means of Nitrogen bottles or similar pressure source. Actuator manufacturer shall furnish the capacity and set pressure of nitrogen bottle for at least two opening and two closing strokes of the actuator.				
4.3	The actuator shall be suitable for gas operating conditions and ambient temperature as specified in data sheets. The presence of methanol in the gas shall not affect the service of the Actuator.				
4.4	All materials in contact with natural gas shall be suitable for the gas composition attached with this specification.				
4.5	The actuator and its accessories shall be suitable; for outdoor installation and have weatherproof enclosure as per NEMA 4 or equivalent.				
	All compartments and housing containing electrical devices such as switches, contactors, relay, fuses, terminal box etc. shall be explosion proof suitable for NEC Class 1 Div. 2 Gr. C & D, T3 or equivalent. The cable glands shall be 1" NPT thread. The unused cable entries must be plugged off with solid metal plugs.				
	Solenoid valves shall be of explosion proof design certified for NEC Class 1 Div. 2 Gr. C & D, T3 or equivalent with moulded continuous duty coils and stainless steel valve body.				
4.6	All pressure containing parts shall b	e designed to ASME Section	VIII.		
4.7	a) The actuator shall be suitable for direct mounting to the valve without changing the standard top works of the valve and shall have the capability to be mounted or removed from the valve when the valve is in service. The actuator shall be flanged and bolted directly on the valve body or extension. The connection between actuator and the valve or between the operators, the outer casing of the extension and the valve shall be such that there is no movement between these connections when the valve is actuated by the actuator under any load.				
	b) Actuator shall be suitable for specified in the data sheet.	r installation on a vertical	stem unless otherwise		
4.8	Provision shall be made to prevent a	accidental pressure build up in	n the actuator.		
4.9	The construction of the actuator and its controls shall be such that proper manual operation and maintenance can be carried out by skilled personnel without the risk of being injured by moving parts.				

	STANDARD TECHNICAL SPECIFICATION			
REGD. OFF: RANCHI 834002	D. OFF: RANCHI     ELECTRICAL & INSTRUMENTATION       834002     OIL & GAS SBU, DELHI			
	SPECIFICATION FOR	DOCUMENT NO.	Page 6 of 15	
TITLE	GAS POWERED VALVE	MEC/ TS / 05 / E5 / 002A	REVISION : 0	
	ACTORTORS		EDITION : 1	
4.10	A position indicator on the actuator open positions.	shall show the valve in the	open, closed or partially	
4.11	Bearings shall be factory packed w for the life of the actuator.	ith grease and shall not requ	ire additional lubrication	
4.12	Unless specified otherwise in the data sheet, the actuator shall be equipped with limit valves, which immediately shut off the gas supply to actuator when the valve reaches one of its end positions.			
4.13	Actuator shall be provided with pressure gauges for pneumatic and hydraulic systems. The pressure gauges for the hydraulic system shall be in circuit with the pressure relief system.			
4.14	In selection of pressure regulator due consideration shall be given to the effect of cooling of gas at the regulator and its down stream section.			
4.15	A high pressure dehydrating filter cartridge shall be provided to remove condensate, moisture, foreign particles and any corrosive contaminants from pipeline gas.			
4.16	The sound level of the gas escaping into the atmosphere when the actuator is in operation shall not exceed 90 dBA, measured at a distance of 10 meters.			
4.17	The actuator shall be operated by either of the following two methods:			
	a) For the remote control, the actuator shall have a solenoid valve, limit switches, relays, etc. and shall be suitable for remote and local operation as per description in the attached Actuator Data Sheet(s). Electrical signal supplied is a momentary type with 1 sec duration signal, Actuator shall have a self retaining system of the above signal in its control circuit.			
	<ul> <li>b) For the local-control the actuator shall be suitable for local operation with line gas feed lines or through nitrogen bottles. The devices and accessories, which do not require the electricity for their operation, shall be provided in the actuators as per this specification. The actuators shall have a hand pump in conjunction with the oil circuit to achieve local control with hydraulic shock functionality.</li> </ul>			
4.18	The actuators shall be provided with limit switches for open and close positions. The position of switches shall be adjustable near the valve open and close positions. The limit switches shall be wired up to terminal block and shall be numbered for proper identification. The limit switches shall have 2 sets of contacts for each open and close position. The contact rating shall be as specified in the data sheet. The power to solenoid valves shall be cut-off when the actuator has travelled to extreme positions (close and open).			

	STANDARD TECHNICAL SPECIFICATION				
REGD. OFF: RANCHI 834002	ELECTRICAL & INSTR OIL & GAS SBL	Anit Buon Convert			
	SPECIFICATION FOR	DOCUMENT NO.	Page <b>7</b> of 15		
TITLE	GAS POWERED VALVE ACTUATORS	MEC/ TS / 05 / E5 / 002A	REVISION : 0		
			EDITION : 1		
4.19	The limit switches shall be wired in the actuator control circuit by the vendor so as to cut off power to the actuator once the end positions of the valve are reached. This is required to de-energize the solenoid valves in the steady state condition and failure of electrical power will not affect the valve position.				
4.20	The stroke of the Actuator shall be easily adjustable in steps of maximum $0.5^{\circ}$ for Ball/ Plug Valves.				
4.21	Speed control nozzles for adjust provided.	ing the valve speed over	a wide range shall be		
4.22	If remote control is required, a local/ remote switch shall be installed to prevent remote control during maintenance work. This switch shall be provided with a hole 12 mm in diameter for locking with pad lock in either position. This local/ remote switch shall be wired up to the junction box as per circuit diagram.				
	All control accessories, pneumatic and hydraulic, shall be mounted in an enclosure and shall be fully wired and tubed. The enclosure shall be weatherproof as per NEMA-4 or equivalent.				
4.23	All bleed and vent connections wherever required shall be piped outside the actuator cabinet so as to prevent gas pocketing inside the actuator cabinet. The actuator shall be of an automatic self purging design such that any gas pocket in the actuator will be eliminated.				
4.24	Vendor shall be responsible for integrating the potential free NO or NC contacts of Remote Telemetry Unit (RTU's) for open and close command in interlock circuit. These commands will be of momentary type with 1 sec duration.				
4.25	All mounting accessories needed for installing the actuator, tanks etc. are in manufacturer's scope of supply.				
4.26	The interconnecting cabling, interconnecting pipe work between the actuator and the valve, adapters, tubing, cable glands, junction box are in manufacturer's scope of supply.				
4.27	The actuator shall be supplied totally self-contained, wired, tubed and mounted on ball valve. In case of a separate control box, wiring and tubing between control box and actuator is in the vendor's scope. Three meters of $\frac{3}{4}$ " tubing set including all connectors between the actuator and the control box and three meters of interconnecting piping work upstream & downstream of the valve and the control box be provided. Owner shall provide 20mm ( $\frac{3}{4}$ ") SW tapping on process main line for line gas tapping.				
4.28	Threading connections shall be NPT as per ANSI B2.1 and flange connection as per ANSI B16.5. The tubing, fittings and valves shall be stainless steel with Swage Lock				

	STANDARD TECHNICAL		
REGD. OFF: RANCHI 834002	मेकान 8001 Core In		
	SPECIFICATION FOR	DOCUMENT NO.	Page 8 of 15
TITLE	GAS POWERED VALVE ACTUATORS	MEC/ TS / 05 / E5 / 002A	REVISION : 0
			EDITION : 1
	fittings or equivalent.		

# 5.0 <u>COATING</u>

The actuator, including gear boxes and piping, shall be coated as described below :

- 5.1 Removal of all rust by means of emery cloth or wire brush.
- 5.2 Short blast as per Swedish Standard No. SIS 055900 latest edition SA 2.5 one layer Primer Epoxy – Polyamide DFT 75 microns intermediate layer Epoxy – Polyamide DFT 25 microns, final layer Epoxy – Polyamide DFT 75 microns. Total DFT 175 microns and colour RAL 5012 (light blue).

## 6.0 <u>INSPECTION AND TESTS</u>

- 6.1 Test at the Actuator Manufacturer's Shop
  - a) Electrical and mechanical operating tests.
  - b) Seal test of hydraulic circuits.
  - c) Check of required functions.
  - d) Check of operating time control.
  - e) Check of limiting device operation.
  - f) Check of actuator torque or thrust.
- 6.2 Test at the Valve Manufacturer's Shop
  - a) Test and check covered by point 6.1 after assembly with ball/ plug valve.
  - b) No load test (DP=0) or load (DP max) Operations with the minimum required feeding pressure (pressure indicated in Table-2 of datasheet).
  - c) Check of the limiting device operation.
  - d) Various tests on the valve according to provisions of specific documentation.
  - e) Testing shall conform to actual field operating conditions.
- 6.3 All actuators shall be visually inspected.

6.4 Dimensional check on actuators shall be carried out as per the Purchaser approved Page 163 of 644

	STANDARD TECHNICAL			
REGD. OFF: RANCHI 834002	GOID OFF: RANCHI     ELECTRICAL & INSTRUMENTATION       834002     OIL & GAS SBU, DELHI			
	SPECIFICATION FOR	DOCUMENT NO.	Page 9 of 15	
TITLE	GAS POWERED VALVE ACTUATORS	MEC/ TS / 05 / E5 / 002A	REVISION : 0	
			EDITION : 1	

drawings.

6.5 Purchaser reserves the right to perform inspection as indicated above at Manufacturer's works prior to shipment. Manufacturer shall give reasonable access and facilities required for inspection to the Purchaser's Inspector.

In no case shall any action of Purchaser or his inspector relieve the Manufacturer of his responsibility for material, design, quality or operation of actuators.

# 7.0 <u>NAME PLATE</u>

Actuator shall have a SS name plate attached firmly to it at a visible place and reporting the following information :

- a) Instrument tag number as per Purchaser's data sheets.
- b) Manufacturer's model, trade mark, serial no etc.
- c) Max. allowable operating pressure or voltage.
- d) Area classification in which the equipment can be used.
- e) Actuator characteristics data.
- f) Inspection agency name or logo.

## 8.0 <u>SHIPPING</u>

- 8.1 All threaded and flanged opening shall be suitably protected to prevent entry of foreign material.
- 8.2 The actuator shall be supplied pre-assembled except piping/ tubing, actuator, actuator control unit, tanks and other accessories shall be packed separately.
- 8.3 Protective grease oil coating shall be applied on the surface to protect them from rusting.
- 8.4 Package shall be marked legibly with suitable marking ink the following :
  - a) Order number
  - b) Package number
  - c) Manufacturer's name
  - d) Model no. & Thrust
  - e) Tag number
  - f) Inspection agency name or logo

Page 164 of 644

MECON LIMITED	STANDARD TECHNICAL			
REGD. OFF: RANCHI 834002	GD. OFF: RANCHI     ELECTRICAL & INSTRUMENTATION       834002     OIL & GAS SBU, DELHI			
	SPECIFICATION FOR	DOCUMENT NO.	Page 10 of 15	
TITLE	GAS POWERED VALVE ACTUATORS	MEC/ TS / 05 / E5 / 002A	REVISION : 0	
			EDITION : 1	

## 9.0 DOCUMENTATION

- 9.1 At the time of vendor drawing approval the manufacturer shall submit the following documents in accordance with EN102043.1C.
  - a) A detailed specification sheet for valve actuator providing all the details regarding type, materials of construction for various parts etc.
  - b) Schematic diagrams showing the complete actuator control circuit.
  - c) A detailed dimensional drawing.
  - d) Installation drawing complete with valve assembly.
  - e) Actuator sizing calculations including relation between required torque of valve and actuator output torque.
  - f) Information asked for vide section 3.0 of this specification and actuator sizing as per Actuator Data Sheet.
  - g) Drawing showing connections by Purchaser (piping, electrical etc.)
  - h) Wiring diagram (actuator electrical circuitry) incorporating latching of momentary signals, remote/ local switch, limit switches.
  - i) Junction box terminal block nos. for I/O signals.
  - j) Parts list.
  - k) Recommended spare parts with prices.
  - 1) Assembly details (Valve & Actuator).
  - m) Clause wise list of deviations from this specification, if any.
  - Information regarding the past experience on similar actuators including (a) Size
     (b) Numbers (c) Name of Installation (d) Owner (e) Name of Contact Person (f) Date of Installation.
  - o) Technical Catalogue giving detailed Technical Specification. and other information for each type of actuator and its accessories covered in the bid.

	STANDARD TECHNICAL		
REGD. OFF: RANCHI 834002	ELECTRICAL & INSTR OIL & GAS SBU	मेकान 9001 Corv	
	SPECIFICATION FOR	DOCUMENT NO.	Page 11 of 15
TITLE	GAS POWERED VALVE ACTUATORS	MEC/ TS / 05 / E5 / 002A	REVISION : 0
			EDITION : 1

9.2 Within 30 days from the date of Purchase Order manufacturer shall submit copes of the following for Purchaser's review and approval.

- a) Documents and specifications as listed in clause 9.1 of this specification.
- b) Test certificates and certificates from statutory bodies.
- c) Manual for installation, erection instructions, maintenance and operation instructions.
- d) Complete assembly drawing of the ball valve matching with the actuator offered.

The approval of these drawings/ documents will not absolve vendor of the responsibility with respect to correct operation of the actuator. Manufacturer's quotation, catalogues, drawings, operating and maintenance manuals etc. shall be in English language.

## 10.0 SPARES AND ACCESSORIES

Vendor shall quote for two years operational spares, commissioning spares, and any special tools needed for maintenance work on the actuator and its accessories.

		STANDARD TECHNICAL	SPECI	FICATION		
REGD. OFF: 83400	REGD. OFF: RANCHI 834002 ELECTRICAL & INST OIL & GAS SB		RUMENTATION U, DELHI		Half	
	SPECIFICATION FOR		C	DOCUMENT NO.	Page <b>12</b> of 15	
TITL	.E	GAS POWERED VALVE ACTUATORS	ME	C/ TS / 05 / E5 / 002A	REVISION : 0	
					EDITION : 1	
	DATAS	HEET FOR GAS POWERE	DAC	<b>FUATORS FOR H</b>	BALL VALVES	
1.	Actuato	r Manufacturer	:	By vendor		
2.	Specific	cation for Gas Powered Actuator	:	MEC/TS/05/E5/0	002A	
3.	Actuato	or Type	:	On-Off		
4.	Tag No		:	As per the Table-1	l	
5.	Line No	Э.	:	As per the Table-1	l	
6.	Service		:	Natural Gas		
7.	Vendor	to furnish, after sizing the actuator,	the fille	ed in torque table no-2	2	
8.	Actuato	or Shut-off Pressure	:	As per the Table-1	l	
6.	Process	Conditions				
	Power (	Gas	:	Natural Gas		
	Gas Ter	mperature	:	As per the Table-1	l	
	Line Ga	as Pressure	:	As per the Table-1	l	
	Molecu	lar Weight	:	As per gas compo	sition	
	Cp/Cv		:	As per gas compo	sition	
	Compre	essibility Factor	:	As per gas compos	sition	
10.	Power ( main lin	Gas Feed Connection from ne	:	3/4" SW		
11.	Actuato (for ope	or remote operation en and close)	:	Required		
12.	Actuato	or Feed Gas	:	a) Line gas		
				b) N <sub>2</sub> Bottles		
13.	Valve F	Position Limit Switch	:	Required (SPDT SPDT contact separately)	contact for open and for close position Page 167 of 644	

MECONILI		STANDARD TECHNICA	L SPECII	FICATION				
REGD. OFF: 83400	RANCHI 02	ELECTRICAL & INST OIL & GAS SB	ELECTRICAL & INSTRUMENTATION OIL & GAS SBU, DELHI					
		SPECIFICATION FOR	D	OCUMENT NO.	Page 13 of 15			
TITL	.E	GAS POWERED VALVE ACTUATORS	MEC	C/ TS / 05 / E5 / 002A	REVISION : 0			
					EDITION : 1			
14.	Local/ and its	Remote selector switch status contact	:	Required (shall be box as per circuit c	wired up to junction liagram)			
15.	Pneuma pilot va actuato its end	atic limit valves and solenoid, lves to shut-off gas supply to r when valve reaches one of positions	:	Required				
16.	Self ret momen in the c	aining system for retaining tary open or close signals ontrol circuit	:	Required				
17.	Electric entries cable)	cal conduit connection (cable to junction box for purchaser's	:	1" NPT				
18.	Operati a) b)	ng voltage for Solenoid Valves Relays	:	24V D.C. ± 10%				
19.	Contac a) b)	t rating for Limit Switches L/R Switch (Status)	:	2A at 24V D.C.				
20.	Pad loc	k with L/R Switch	:	Required				
21.	Enclosu a)	ure for Actuator	:	Certified weather	proof as per IP-55			
	b)	Electrical items like solenoid valves, junction boxes, relays, cable glands	:	Certified weatherp Explosion proof ce Div. 2 Gr. C & D	proof toIP55 & ertified for Class 1 T3			
22.	Area C	lassification	:	NEC (Class 1,Div	. 2,Gr. C & D,T3)/equiv			
23.	Materia valves,	al of construction for all tubing, piping and fittings etc.	:	SS 316				
24.	Make &	& Model No. of Actuator	:	By vendor				
25. 26. 27.	Access Manual Time re Closing	ories Required // Hydraulic Override equired for full opening/ g of the ball valve	: : :	Required as per Sp Required as per Sp 2–3 sec. per inch.	pecification pecification Nominal valve port Dia.			

	STANDARD TECHNICAL		
REGD. OFF: RANCHI 834002	ELECTRICAL & INSTR OIL & GAS SBU	4817 83001 6000	
	SPECIFICATION FOR	DOCUMENT NO.	Page 14 of 15
TITLE	GAS POWERED VALVE ACTUATORS	MEC/ TS / 05 / E5 / 002A	REVISION : 0
			EDITION : 1

# TABLE-1

# **ITEM : GAS POWERED BALL VALVE ACTUATORS**

S.	Size	Class	Type of	Line	Gas Temp (°C)		Line Gas Pressure (kg/cm <sup>2</sup> g)		Delta P Shut Off	Remark	
INO.			Valve	INU.	Inlet	Max	Nor. Ma x		(kg/cm <sup>2</sup> g)	5	

	STANDARD TECHNICAL		
REGD. OFF: RANCHI 834002	ELECTRICAL & INSTE OIL & GAS SBU	मेकॉन 9001 Conv <sup>g</sup>	
	SPECIFICATION FOR	DOCUMENT NO.	Page 15 of 15
TITLE	GAS POWERED VALVE ACTUATORS	MEC/ TS / 05 / E5 / 002A	REVISION: 0
			EDITION : 1

# TABLE-2

# **ITEM : GAS POWERED BALL VALVE ACTUATORS**

	MR Item	Data from Ball Valve Vendor for (@ Max. Diff. Press.)			Ball Valv Figure wit Factor	e Torque th a safety of 1.25	Actuator Torque : pressure	Model Selected	
SI. No.	No., Valve Size & rating, Qty.	Break Torque (Nm)	Running Torque (Nm)	Max Allowable Valve Stem Torque (Nm)	Break Torque (Nm)	Running Torque (Nm)	Break Torque (Nm)	Running Torque (Nm)	

# NOTE :

- 1. THESE FIGURES SHALL BE USED AS BASIS FOR TESTING THE ACTUATOR PERFORMANCE DURING FACTORY TESTING. THE ACTUATOR ACCEPTANCE WOULD BE CARRIED OUT AFTER VERIFYING SUCCESSFUL TESTING COMPLETE BALL VALVE WITH ACTUATOR ASSEMBLY.
- 2. ALL TORQUE FIGURES MUST BE IN Nm.
- 3. Opening / Closing time shall be achieved at max. Design Differential pressure across the valve & actuator regulated pressure max. 24 kg/cm2(g) (approx.). However, the actuator shall open/close the valve at actuator regulated pressure of 10 kg/cm2(g) at 10 kg/cm2(g) Differential pressure across the valve and without timing restriction.



		5		4			3			2				1	
G	VEN F	DOR TO PROVIDE	DEVELO JUNCTIC	P ACTUA N BOX	TOR ELE	CTRIC TERI	CAL/HY MINAL		LIC CIRC	CUIT AS	PE MER	R SKI	ETCH LE TI	BELOW A ERMINATIO	ND N
F E	SKETCH-2	ACTUATOR ELECTRICAL/HYDRAULIC CIRCUIT TO BE DEVELOPED BY ACTUATOR/ BALL VALVE SUPPLIER WHICH	SHALL INCLUDE RELAYS. LIMIT SWITCHES, SOLENOID VALVES, LOCAL/REMOTE SWITCH, HYDRAULIC SHOCK ABSORBER WITH PUMP UNIT CONTROL AND MANUAL OPERATOR ETC.	ALL CABRING THROUGH GLANDS BY SUPPLIER					POWER אילא אילא אילא אילא אילא אילא אילא איל		COMMAND : POSITIVE PULSE(POTENTIAL FREE) FOR 1 SEC.	ED COMMAND :	PULSE (POTENTIAL FREE) FOR 1. SEC. FOR ESD VALVES	E POSITIVE PULSE FOR OTHER VALVES	F
B	CONTACT STATUS :	1. CONTACT TO CLOSE ON VALVE OPEN FOR OPEN	2. CONTACT TO CLOSE ON VALVE CLOSE FOR CLOSE LIMIT SWITCH.	3. L/R SWITCH CONTACT TO CLOSE WHEN VALVE IS ON REMOTE.				MULTICORE CABLE	ENTRY (1" NPT)		VALVE OPE	VALVE CLO	I NEGATIVE	II OTHERWIS	С  В
A	REV NO THIS DRAWIN SECTION DSGN. DRWN CHKD. VERIFIEL APPROVEI	DATE ZONE DATE ZONE CONE DATE ZONE CONE CONE CONE CONE CONE CONE CONE C	of mecon and is T. (INST.) DELHI SIG. DATE	DESCRIPTIONS REVISIONS ISSUED FOR THE PO PO 4	SPECIFIC PROJECT SPECIFICATI INSTRUM WERED VAL	BY MENTIONED	APPRD THEREIN. THI DR GAS/ AIR TUATOR: 3	s is not to	REFEREN BE COPIED OR L SCALE : N. DRG.NO. ME	ICES USED FOR OTHI .T.S CC/05/26/1 2	ER PROJE	ects unless मैट IEC(	ि EXPRESSI ठॉन DN	DRG. NO. Y PERMITTED BY MEC लिमिटेड LIMITE 72 of 1644	2D

#### 1. Application :

Water Dew Point measurement in Natural Gas at pipeline conditions, Dry as well as associate /wet Natural gas custody transfer application.

#### 2. Sampling System:

Extractive through retractable sample collection probes.

- 3. Sample Pressure: Refer P&ID.
- 4. **Sample Temperature:** Refer P&ID.

#### 5. Ambient Conditions

Temp Range : 0 °C to 55 °C, Humidity 100% (Max)

Online Analyser system shall be suitable for the ambient conditions and shall guarantee that the performance of the online Analyser shall not deteriorate up to the specific ambient conditions.

- 6. Accuracy & Repeatability 1 Deg. C or better
- 7. Range of Measurement: Refer P&ID / Gas Composition.

#### 8. Unit of Measurement / Output-:

'Water Dew Point' in Deg. C at line operating conditions.

9. **Resolution :** 0.1 °C or better

#### 10. Calibration Traceability

Calibrations traceable to National Institute of Standards and Technology (NIST)/NPL/ Any Metrological Institute

- 11. Analyser Measurement Response Time : Less than 5 Minutes
- 12. Initial warm up time : Less than 20 min

#### 13. Analyser Shed:

The analyzer shall be installed with suitable canopy, to avoid direct exposure to sunlight. Environment protection class is defined at point no 22.

#### 14. **Type of Sensor:**

The sensing mechanism of the WDP analyzer to be **`Tunable Diode LASER Absorption** Spectroscopy' type.

#### 15. Cycle Time:

In Seconds (to be specified by the bidder depending of the stability period in seconds in both way transitions from 'small to high' gas samples & from 'high to small' gas samples of their analyzer).

#### 16. Number of Streams:

One stream.

17. **Pipeline Sizes:** Refer P&ID.

#### 18. Sample Collection Probe:

0	DATASHEET OF ONLINE WATER DEW POINT ANALYZER SYSTEM	MECON LTD. DELHI
Rev.		DS No: MEC/ 05/E5/DS-WDP

Retractable sample collection probes.

#### **19.** Sample Handling System:

- The sampling handling system shall be designed to move the sample from the process to the online Analyser in the shortest possible time.
- Sampling system shall include block valves on all process sampling lines.
- The sample handling system shall be designed in such a way so as to take care of both type of gas i.e. associate/wet as well as dry natural gas.
- Sample shall be returned to the atmosphere at a safe height.
- Sample system shall have selective filtrated passage of the concerned sample to avoid ingress of oil/dust in the sample.

• NG may contain traces of H2S, Glycol & other corrosive impurities; supplied system should be designed accordingly.

#### 20. **Pressure Regulator & Sample conditioning:**

Suitable pressure regulating device & sample conditioning device should be supplied. Pressure Regulator, Filter-Separator etc., necessary tubing from pressure regulator up to the analyzer, heating arrangement, if any, all are required to be supplied by the bidder. Regulator shall have over pressure protection system.

#### 21. Instrument Air Supply:

No instrument air supply is available at site hence the vendor has to design accordingly without instrumentation air.

#### 22. Enclosure & Area Classification:

The Analyser & all accessories to be field mountable type with IP66 protection and the electrical appliances to be certified for use safely in, Class-1, Division-1, and Group C&D area. Enclosure should be designed considering, no availability of instrument air supply at the site. The external cabinets to be made of SS 316.

#### 23. Input Power Supply:

230 V 50 Hz

#### 24. Local Indication:

Programmable "through-the-glass", local indication of 'Water Dew Point' in degree C shall be available & viewable from the outside of the field cabinet without opening any cabinet etc. Water Dew point conversion methods : ISO 18453 / ASTM D1142 or equivalent using live input for pipeline pressures / Temp.

#### 25. Analog & Digital Outputs:

4-20 mA analog outputs for Water Dew point shall be available. One Ethernet and minimum two `RS 232 / 485 MODBUS RTU' serial outputs shall also be made available simultaneously for both the outputs through each port. Minimum 2 Digital Output Relays for Hi/Lo Alarm and General Fault.

#### 26. Analog & Digital Signal JB:

Analog & digital signals from the analyzer shall be terminated in separate JBs provided on the self- standing skid.

#### 27. Sample Composition (normal range):

The average composition of natural gas is provided elsewhere in the tender. Composition of Natural gas may vary, system should be designed to perform in varying natural gas compositions.

#### 28. Venting:

If required should be at least 3 meter above from the nearest working platform with proper support. All necessary piping tubing to be done by the bidder.

#### 29. Memory & Data storage :

The online Analyser system supplied shall store the data for minimum one month period with a provision of downloading the data on as and when required basis. Vendor shall furnish the maximum capacity of the online Analyser to store the data in the offer.

0	DATASHEET OF ONLINE WATER DEW POINT ANALYZER SYSTEM	MECON LTD. DELHI
Rev.		DS No: MEC/ 05/E5/DS-WDP

The system configuration & application program shall be stored on a non-volatile memory. Battery backup shall be provided in case memory is volatile. The design shall ensure that the application program and data tables remain unaffected in case of power fluctuation or failure. Clock time shall be protected with battery.

#### 30. **Communication Features**

Analog & Digital Output: 4-20 mA analog output for Water Dew point shall be available. Programmable alarm relays with Potential Free output and hermetically sealed.

Minimum 02 nos. of isolated serial port, One RS 232 /485 serial communication port for connecting CLIENT's SCADA/ RTU / flow computer (MODBUS RTU protocol) output & One RS 232 /485 isolated serial communication port for connecting the configurator / Laptop. The protocol shall be MODBUS RTU or ASCII and the same shall be user selectable. The serial ports should be user configurable for 232 and 485. One TCP/IP Ethernet port configurable for programming, Mod bus Data communication, Printing.

current process value, Day's average, max value and alarm value (set point to be configure in system) shall be communicated from analyser. The analyser should be configurable for data trending and registering the values up to 30 days. The data shall be downloaded to portable device like laptop having window software version 8 or higher.

**Note:** Complete details and documentation with respect to protocol details with message structure, frame structures, synchronizing / timing signals, memory register locations for data addressability and interface software driver details shall be furnished in order to successfully implement a serial link with the RTU. Any converter, repeater in the scope of vendor. Facility for remote monitoring of analyzer Through TCP/IP.

#### 31. **Diagnostic & validation features:**

- The online Analyser shall have auto diagnosis feature and shall display error/ warn messages about the status of every parameter and eventful malfunctioning.
- Online automatic diagnostics for periodic self-checking of system integrity & alarming.
- Vendor must specify the frequency (if required) of factory validation / calibration of the supplied system
- Provision in Analyser for Auto calibration / validation & Auto Zero checking at site by user.
- Event Logging / Audit log with Data tamper protection.

#### 32. Reporting Features

- Generation of Analysis Report (scheduled or on- demand) and Automatic Generation of 24 hrs Average maximum, minimum Reports.
- Generation of alarms, diagnostic report
- The reports & alarm log etc. shall be stored in user configurable modbus address for connectivity

#### 33. **Program Features**

The programming shall be user friendly. Field modifications of data tables and Analyser application shall be possible under password protection. Recent version of interface software compatible for windows-7 & 8 Environment with license.

0	DATASHEET OF ONLINE WATER DEW POINT ANALYZER SYSTEM	MECON LTD. DELHI
Rev.		DS No: MEC/ 05/E5/DS-WDP

#### 1. Application :

Hydrocarbon Dew Point measurement in Natural Gas at pipeline conditions, Dry as well as associate /wet Natural gas custody transfer application.

#### 2. Sampling System:

Extractive through retractable sample collection probes.

- 3. Sample Pressure: Refer P&ID.
- 4. **Sample Temperature:** Refer P&ID.

#### 5. Ambient Conditions

Temp Range : 0 °C to 55 °C, Humidity 100% (Max)

Online Analyser system shall be suitable for the ambient conditions and shall guarantee that the performance of the online Analyser shall not deteriorate up to the specific ambient conditions.

- 6. Accuracy & Repeatability 0.5 Deg. C or better
- 7. Range of Measurement: Refer P&ID / Gas Composition.

#### 8. Unit of Measurement / Output-:

'Hydrocarbon Dew Point' in Deg. C at line operating conditions.

#### 9. **Resolution :** 0.1 °C or better

#### 10. Calibration Traceability

Calibrations traceable to National Institute of Standards and Technology (NIST)/NPL/ Any Metrological Institute

#### 11. Analyser Measurement Response Time : Less than 5 Minutes

12. Initial warm up time : Less than 20 min

#### 13. Analyser Shed:

The analyzer shall be installed with suitable canopy, to avoid direct exposure to sunlight. Environment protection class is defined at point no 22.

#### 14. **Type of Sensor:**

The sensing mechanism of the analyzer to be **`Tunable Diode LASER Absorption** Spectroscopy' type.

#### 15. Cycle Time:

In Seconds (to be specified by the bidder depending of the stability period in seconds in both way transitions from 'small to high' gas samples & from 'high to small' gas samples of their analyzer).

#### 16. Number of Streams:

One stream.

17. **Pipeline Sizes:** Refer P&ID.

#### 18. Sample Collection Probe:

0	DATASHEET OF ONLINE HYDROCARBON DEW POINT	105-000 CONTROL
	ANALYZER SYSTEM	MECON LTD. DELHI
Rev.		DS No: MEC/ 05/E5/DS-HDP

Retractable sample collection probes.

#### **19.** Sample Handling System:

- The sampling handling system shall be designed to move the sample from the process to the online Analyser in the shortest possible time.
- Sampling system shall include block valves on all process sampling lines.
- The sample handling system shall be designed in such a way so as to take care of both type of gas i.e. associate/wet as well as dry natural gas.
- Sample shall be returned to the atmosphere at a safe height.
- Sample system shall have selective filtrated passage of the concerned sample to avoid ingress of oil/dust in the sample.

• NG may contain traces of H2S, Glycol & other corrosive impurities; supplied system should be designed accordingly.

#### 20. **Pressure Regulator & Sample conditioning:**

Suitable pressure regulating device & sample conditioning device should be supplied. Pressure Regulator, Filter-Separator etc., necessary tubing from pressure regulator up to the analyzer, heating arrangement, if any, all are required to be supplied by the bidder. Regulator shall have over pressure protection system.

#### 21. Instrument Air Supply:

No instrument air supply is available at site hence the vendor has to design accordingly without instrumentation air.

#### 22. Enclosure & Area Classification:

The Analyser & all accessories to be field mountable type with IP66 protection and the electrical appliances to be certified for use safely in, Class-1, Division-1, and Group C&D area. Enclosure should be designed considering, no availability of instrument air supply at the site. The external cabinets to be made of SS 316.

#### 23. Input Power Supply:

230 V 50 Hz

#### 24. Local Indication:

Programmable "through-the-glass", local indication of 'Hydrocarbon Dew Point' in degree C shall be available & viewable from the outside of the field cabinet without opening any cabinet etc. Hydrocarbon Dew point conversion methods : ISO 18453 / ASTM D1142 or equivalent using live input for pipeline pressures / Temp.

#### 25. Analog & Digital Outputs:

4 – 20 mA analog outputs for Hydrocarbon Dew point shall be available. One Ethernet and minimum two 'RS 232 / 485 MODBUS RTU' serial outputs shall also be made available simultaneously for both the outputs through each port. Minimum 2 Digital Output Relays for Hi/Lo Alarm and General Fault.

#### 26. Analog & Digital Signal JB:

Analog & digital signals from the analyzer shall be terminated in separate JBs provided on the self- standing skid.

#### 27. Sample Composition (normal range):

The average composition of natural gas is provided elsewhere in the tender. Composition of Natural gas may vary, system should be designed to perform in varying natural gas compositions.

#### 28. Venting:

If required should be at least 3 meter above from the nearest working platform with proper support. All necessary piping tubing to be done by the bidder.

#### 29. Memory & Data storage :

The online Analyser system supplied shall store the data for minimum one month period with a provision of downloading the data on as and when required basis. Vendor shall furnish the maximum capacity of the online Analyser to store the data in the offer.

0	DATASHEET OF ONLINE HYDROCARBON DEW POINT ANALYZER SYSTEM	MECON LTD. DELHI
Rev.		DS No: MEC/ 05/E5/DS-HDP

The system configuration & application program shall be stored on a non-volatile memory. Battery backup shall be provided in case memory is volatile. The design shall ensure that the application program and data tables remain unaffected in case of power fluctuation or failure. Clock time shall be protected with battery.

#### 30. **Communication Features**

Analog & Digital Output: 4-20 mA analog output for Hydrocarbon Dew point shall be available. Programmable alarm relays with Potential Free output and hermetically sealed.

Minimum 02 nos. of isolated serial port, One RS 232 /485 serial communication port for connecting CLIENT's SCADA/ RTU / flow computer (MODBUS RTU protocol) output & One RS 232 /485 isolated serial communication port for connecting the configurator / Laptop. The protocol shall be MODBUS RTU or ASCII and the same shall be user selectable. The serial ports should be user configurable for 232 and 485. One TCP/IP Ethernet port configurable for programming, Mod bus Data communication, Printing.

current process value, Day's average, max value and alarm value (set point to be configure in system) shall be communicated from analyser. The analyser should be configurable for data trending and registering the values up to 30 days. The data shall be downloaded to portable device like laptop having window software version 8 or higher.

**Note:** Complete details and documentation with respect to protocol details with message structure, frame structures, synchronizing / timing signals, memory register locations for data addressability and interface software driver details shall be furnished in order to successfully implement a serial link with the RTU. Any converter, repeater in the scope of vendor. Facility for remote monitoring of analyzer Through TCP/IP.

#### 31. **Diagnostic & validation features:**

- The online Analyser shall have auto diagnosis feature and shall display error/ warn messages about the status of every parameter and eventful malfunctioning.
- Online automatic diagnostics for periodic self-checking of system integrity & alarming.
- Vendor must specify the frequency (if required) of factory validation / calibration of the supplied system
- Provision in Analyser for Auto calibration / validation & Auto Zero checking at site by user.
- Event Logging / Audit log with Data tamper protection.

#### 32. Reporting Features

- Generation of Analysis Report (scheduled or on- demand) and Automatic Generation of 24 hrs Average maximum, minimum Reports.
- Generation of alarms, diagnostic report
- The reports & alarm log etc. shall be stored in user configurable modbus address for connectivity

#### 33. **Program Features**

The programming shall be user friendly. Field modifications of data tables and Analyser application shall be possible under password protection. Recent version of interface software compatible for windows-7 & 8 Environment with license.

0	DATASHEET OF ONLINE HYDROCARBON DEW POINT ANALYZER SYSTEM	MECON LTD. DELHI
Rev.		DS No: MEC/ 05/E5/DS-HDP

# **ANNEXURE - II**

# **GAS COMPOSITION PARAMETERS**

Page 179 of 644

C1: 80% and above

C2 heavier: it is less than or equal to 18%

CO2: 0.5 to 2% Mole

N2: 0.5 to 2% Mole

Specific Gravity: 0.57 to 0.65
## **ANNEXURE - III**

## DRAWINGS

Page 181 of 644

10		0		• 7		-		-			1	
12 1000/000		я	I	SYMBOLS	&	ABI	BRI	EVIATIONS	5		1 *	I
	SYMBOLS		SYMBOL	5			ר ר	AF	BRE	VIAT	IONS	
	VP-PLUG VALVE		PD METER	2				AE AN	ALYZEI	R SEN	NSOR	
	VB-BALL VALVE WELDED END							(64	IS CHE	COMAI	TOGRAPH)	
	VB-BALL VALVE FLANGED END		TURBINE	METER				AI AN	ALYZEI	R IND	ICATOR	
	GATE VALVE	¦∣-d┣-	INSULATIO	N JOINT (IJ)				AT AN	ALYZEI	R TRA	NSMITTER	
	GV-GLOBE VALVE	i   T	FLOW TEE	(FWT)				FY FLO	OW COI	<b>MPUT</b> I	ER	
	PRESSURE RELIEF OR SAFETY VALVE (PSV)		FIELD MO	UNTED INSTRUMENT				FQI FL(	W TO	TALIZI	ER & INDICA'	FOR
	PRESSURE REGULATING VALVE (PCV)	$  \ominus$	CONTROL F	ANEL MOUNTED INSTRUMEN	Т			LG LE	EL GA	UGE		
	(SELF ACTUATING TYPE)		ULTRASON	IC METER				ZSH/ZSL LIM	IT SW	ITCHE	ES (OPEN/CLO	DSE)
	NON RETURN VALVE		SCADA INI	DICATOR CONTROLLER OR	2			TI/TG TE	MPERA'	FURE	INDICATOR/	GAUGE
	CONTROL VALVE WITH SLAM SHUT-OFF VALVE		OTHER DE ACCESS T	VICES WITH OPERATOR 0 ADJUSTMENT.				TE TE	MPERA'	FURE	ELEMENT	
	SOLENOID VALVE							TW TH	ERMOW	ELL		
								DPI/DPG DI	F. PR	ESSUI	RE INDICATOR	R/ GAUGE
	GAS ACTUATED ISOLATION VALVE							DPT DIF	F. PR	ESSUI	RE TRANSMIT	TER
	WITH HANDWHEEL	i I						PI/PG PR	ESSUR	E IND	DICATOR/ GAU	JGE
	REDUCER							PT PR	ESSUR	E TRA	ANSMITTER	
	FLANGE							FC/FO FA	LURE	TO CI	LOSE/OPEN	
	FLAME ARRESTOR (FA)							LO LO	ск орі	EN		
RTU	REMOTE TERMINAL UNIT							FIC/PIC FLO	W/PR	ESS.	INDICATING (	CONTROLLE
	DIAPHRAM							$\langle \hat{\mathbf{I}} \rangle$ INT	ERLOC	K		
	PIG SIGNALLER							XRL LOC	AL/RE	моте	INDICATION	
-   ↑	VENT							XSL LOO	AL/RE	моте	SWITCH	
								HSH HAN	ID SWI	TCH	HIGH	
	SPECTACLE BLIND							HSL HAN	ID SWI	ТСН	LOW	
	<u></u>			1			┙╽	FT FLO	TRA	NSMIT	TTER	
	00"-FLUID-00-00-A1A/B1A/C1A/D1A/	′D4A						FE FLO	W ELEI	MENT		
1	MATERIAL SPEC.							Q.O.E.C. QUI	CK OP	ENING	G END CLOSU	RE
	LINE NO.							WDP WAT	ER DE	W PO	INT	
в	UNIT NO.							НДР НУГ	ROCAR	BON	DEW POINT	
	(P-NATURAL GAS)							111				
	LINE SIZE (NOMIN	AL)					L					
				]				1				
										-		
PEV											REFERENCES	DRG.
SEC INST.	CONCURRED BY				REV.NO	DATE	ZONE	DESCRIPTION	BY	VERIFIED	THIS DRAWING IS THE P FOR THE SPECIFIC PROJ NOT TO BE COPIED OR UNLESS EXPRESSLY PER	RUPERTY OF MECON A IECT MENTIONED THERE USED FOR OTHER PRO MITTED BY MECON.

	3			2 1	
	-				
			A	ABBREVIATIONS	
	Т	T	1	TEMP TRANSMITTER	Н
	F	R	F	FLOW RECORDER	
	P	PIR	F	PRESSURE IND.RECORDER	
	Т	'IR	J	TEMP. IND. RECORDER	
	P	SH	F	PRESSURE SWITCH HIGH	G
	P	SL	F	PRESSURE SWITCH LOW	
	L	SH	I	LEVEL SWITCH HIGH	
	L	SL	I	LEVEL SWITCH LOW	
	L	AH	I	LEVEL ALARM HIGH	F
		AL	I	LEVEL ALARM LOW	
2	D	)	C	CONDENSATE DRAIN	
	A	1A/B	LA 1	150#/300# RATING	
	c	IA/D	LA 4	400#/600# RATING	E
	L	X/PX	I	PNEUMATIC/ ELECTRIC SIGNAL TRANSMITTER	
	F	'B	F	FULL BORE	
	s	SP	S	SAMPLER	
ER	N	IG	ľ	NATURAL GAS	
	v	,	۲	VENT	D
	A	S	A	AUTO SAMPLER	
	C	P	C	CORROSION PROBE	
	C	C	C	CORROSION COUPON	
		AV	A	ACTUATED VALVE	
	S	SDV	S	SHUT DOWN VALVE	
	I	LV	L	LEVEL CONTROL VALVE	
	F	PV/PC	V F	PRESSURE CONTROL VALVE	
	F	rCV	F	FLOW CONTROL VALVE	
		GGL	IN	NDRADHANUSH GAS GRID LIMITED	
				मेकॉन लिमिटेड	
	0000			MECON LIMITED	
	SECTION	OIL & GAS DELHI	;	NORTH EAST GAS GRID (PHASE-1 & 2 P/L SECTION)	
	DRAWN	SUNIL			<b></b> ^
G.NO.	CHECKED AND VERIFIED			LEGEND / DRAWING SYMBOLS	
and issued Rein. This is Rojects	APPROVED	SIG DATE		SCALE : NTS SHEET DRG. NO THEOGRAC 108/28,007000/0601 1 0F 2	REV O
				2 1 1	

12	11 10	9	8 7	6	5 4	3	2 1	_
1000/00	DRG. NO. MEC/23VC/05/28/M/00		SYMBOLS & ABB	REVIATION	<u>S</u>			
	SYMBOLS		SYMBOLS		SYMBOLS	]	ABBREVIATIONS	1
	SHUT-OFF VALVE, GENERAL	ş	SPRING-OPERATED ACTUATOR		FILTER, GENERAL	ESD	EMERGRNCY SHUT-DOWN	
	SAFETY SHUT-OFF VALVE	γ	AUTOMATICALLY OPERATED ACTUATOR,		GAS FILTER, GENERAL	EW	DRAIN	
	CONTINIUOUS-ACTION VALVE		GENERAL	<del></del>		CCV	CAS_OPERATED CONTROL VALVE	
	RIGHT-ANGLE SHUT-OFF VALVE GENERAL	<b>Y</b>	GENERAL		CARTRIDGE FILTER	GCV	GAS-OFENATED CONTROL VALVE	
		Ŷ	VALVE CLOSED ON FAILURE OF ACTUATING		GAS SORPTION FILTER	GOV	GAS-OVER OIL OPERATED VALVE	
	THREE-WAY VALVE, GENERAL		ENERGY			HOV	HYDRAULIC-OPERATED VALVE	
	FOUR-WAY VALVE, GENERAL	Ŷ	VALVE OPENED ON FAILURE OF ACTUATING ENERGY			LC	LOCKED CLOSED	
	NEEDLE VALVE	Q	VALVE RETAINS POSITION ON FAILURE OF		"Y" TYPE STRAINER	LO	LOCKED OPEN	
		+	ACTUATING ENERGY		IN AUXILIARY AND CONTRAL FACILITIES		LOCKED POSITION	
	FUSHBUITON VALVE		TURBINE, GENERAL			LPT	LOW POINT	
	CHECK VALVE (SPRING TYPE)	0	COMPRESSOR, GENERAL		PRESSURE CONTROL VALVE, GENERAL	MCV	MOTOR CONTROL VALVE	
	NON RETURN VALVE (SWING TYPE)	œ	ROTARY PISTON COMPRESSOR		PROCESS PRIMARY LINE	MH	MANHOLE	F
	BUTTERFLY VALVE	G	PISTON COMPRESSOR	—	PROCESS SECONDARY LINE	MOV	MOTOR-OPERATED VALVE	
	CONTINUOUSLY VARIABLE BORE	0	JET BOOSTER	+++	ELECTRIC LINE	SV	SOLENOID VALVE	
	BUTTERFLY VALVE	B B	FAN, GENERAL		PNEUMATIC LINE			
	LOUVRE		COUNTERFLOW HEAT EXCHANGER		CAPILLARI LINE	PCV	PRESSURE CONTROL VALVE	
	SEPERATOR GENERAL		PLATE HEAT EXCHANGER		INSTRUMENT LINE	SAV SSV	SAFETY SHUT-OFF VALVE	<sub>E</sub>
	2 STAGE SEPERATOR				HEAT TRACING LINE	SBV	PRESSURE RELIFE VALVE	
	CYCLONE SEDEDATOD		FINNED HEAT EXCHANGER WITH VENTILATOR		JACKETED LINE		SETDAINT	
		Ŷ	BLIND PLATE		JACKETED & HEATED LINE		THERMOCOUPLE	
	CONDENSATE TRAP	<b>\$</b> "	FIGURE 8" PLATE, BLIND PLATE IN		VENT STACK	 	THERMOWELL	
7	FLOWMETER GENERAL	, , , , , , , , , , , , , , , , , , ,	FUNCTION FIGURE 8" PLATE COCCLE PLATE IN		SCRAPPER TRAP			
	METER GENERAL	Ŷ	FUNCTION		LIQUID PUMP GENERAL	NRV	NON-RETURN VALVE	
⊈- •	TURBINE METER	þ	RESTRICTED ORIFICE		CENTRIFUGAL PUMP	ECV	CHECK VALVE	
₽.	VORTEX-SHEDDING FLOWMETER	ÎÌ	INSULATING FLANGE		LIQUID KING FUMF	FCV	FLOW CONTROL VALVE	
	DIAPHRAGM METER		INSULATING COUPLING	U U U	DIAPHRAGM PUMP	PGS	POWER GAS SUPPLY	
	MANUALLY OPEDATED ACTUATOR CENERAL		STRAIGHTENING VANES			IAS	INTRUMENT AIR SUPPLY	
	DIAPHRAGM ACTUATOR		RUPTURE DISK	CP	CORROSION PROBE	NC NO	NORMAL CLOSED	c
	PISTON ACTUATOR		SIGHT GLASS. GENRAL EXPANSION JOINT		CORROSION COUPON			
	ELECTROMAGNETIC ACTUATOR		SILENCER					
	HYDRAULIC OR PNEUMATIC ACTUATOR,	6	PULSATION DAMPENER		SCRUBBER			-
	GENERAL		MIXING NOZZLE					
		D	CAP				INDRADUANUSU CAS	-
В			FLANGE CONNECTION			IGGL	GRID LIMITED	E
							मेकॉन लिमिटेइ	
_								4
 						0 0 1 COMPE	MECON LIMITED	
						SECTION OIL & GAS	NORTH EAST GAS GRID (PHASE-1 & 2 P/L SECTION)	
								4
					REFERENCES         DRG.NO.           © Copyright MECON LDMITED         All rights reserved.		LEGEND / DRAWING SYMBOLS	
SEC INST.	CONCURRED BY		REV.NO DATE Z	ONE DESCRIPTI REVISION	ON BY VERIFIED THIS DRAWING IS THE PROPERTY OF MECON AND ISSU TO THE SPECIFIC PROVECT MENTIONED THEREIN. THIS NOT TO BE COPIED OR USED FOR OTHER PROJECTS UNLESS EXPRESSLY PERMITED BY MECON.	APPROVED SIG	DRG. NO. MELODOVC 108/28/NJ 060/0001 2 0F 2 0	<u>,</u>



REVISION

CONCURRED BY

SEC

	3			2	1		
A	<u>NOTES:</u> 1. ALL V	ENTS SHA	LL BE LOCAT	ted Min. Height of	3 MTR. ABOVE THE HIGHES	т	
	2. METE MOUN EDITIC CHEC	ring levi Ring Shal Ited Flow Dn. Howev K Meterin	L BE DONE COMPUTER. ER THERE S IG.	By Orifice Meter ( Installation Shall Hall be two Strea	SINGLE CHAMBER) WITH FIE BE AS PER AGA3 LATEST MS OF METERING (1W+1S)	ld For	н
	3. ENVIR	ONMENTAL ATION OF	ORIFICE MET	E REQUIRED FOR MET TER, METER RUN & I	ERING INSTRUMENTS. MPLUSE LINES OF METERING	G	
	INSTR 4. PROV	RUMENT TO	BE PROVID	ed. To hook up meter	ING PR. & TEMP., FLOW		
	5. INLET 6. ALL 7. SIZE, ENGI 8. SKID	METERS WI & OUTLE BALL VALV RATING & NEERING. SHALL BE	t of skid s es shall be set press symmetric	SHALL BE PROVIDED E FULL BORE ONLY. SURE OF PSV's TO B ACROSS CENTRE LIN	with companion flanges. E decided during detailei IE.	D	G
	9. FUTU HOWE	RE PROVIS	SION FOR CA	PACITY AUGMENTATION CONFIRM.	N HAS BEEN PROVIDED.		
	VALV	E/GLOBE	VALVE.	COMBINATION OF A	BALL VALVE AND A PLUG		
	12. NOTE 13. NOTE 14. NOTE	DELETED DELETED DELETED					F
							E
							D
						ŀ	
	<u>V</u>	ALVE LE	EGEND				с
			BALL VA BALL VA	LVE FLANGE END LVE BW END/UPTO	1 <sup>1</sup> /2" SW END.		
			PLUG VA PLUG VA GLOBE V	ALVE BWY END/OPTO ALVE FLANGE END /ALVE.	172 SW END.	-	
	į		IN	DRADHAI GRID	NUSH GAS		в
				 मेकॉन	लिमिटेड		
		<b>1</b>		MECON	LIMITED		
	SECTION	01 CO™		NORTH	EAST GAS GRID		
		UMAR		(PHASE-1	& 2 P/L SECTION)		
).		A.K.BHARTI		P&IDFORCHEC	K METERING FACILITY (TYPE	;-I)	
ISSUED THIS IS CTS	APPROVED	SIG	(S.GUPTA)	scale : nts drg. no. <b>melgas</b> vc/o	8/28/9/06/4+4+DER-1001	REV	
		DATE	U6.12.24	2	1	لنسا	



&	NOTES:	2 1
	1. ALL VENTS SHALL BE LOCATED OPERATING LEVEL.	D MIN. HEIGHT OF 3 MTR. ABOVE THE HIGHEST
	2. METERING SHALL BE DONE BY PANEL MOUNTED FLOW COMPL EDITION. HOWEVER THERE SHA CHECK METERING.	Y ORIFICE METER (SINGLE CHAMBER) WITH CONTROL H JTER. INSTALLATION SHALL BE AS PER AGA3 LATEST ALL BE TWO STREAMS OF METERING (1W+1S) FOR
İ	3. ENVIRONMENTAL ENCLOSURE F INSULATION OF ORIFICE METER	REQUIRED FOR METERING INSTRUMENTS. R, METER RUN & IMPLUSE LINES OF METERING
   	4. PROVISION SHALL BE KEPT TO PARAMETERS WITH SCADA.	D HOOK UP METERING PR. & TEMP., FLOW
	<ol> <li>INLET &amp; OUTLET OF SKID SHA</li> <li>ALL BALL VALVES SHALL BE F</li> <li>SIZE, RATING &amp; SET PRESSUF ENGINEERING.</li> </ol>	ALL BE PROVIDED WITH COMPANION FLANGES. FULL BORE ONLY. RE OF PSV'S TO BE DECIDED DURING DETAILED
	8. SKID SHALL BE SYMMETRIC AN 9. FUTURE PROVISION FOR CAPA HOWEVER M/S IGGL HAS TO (	CROSS CENTRE LINE. ICITY AUGMENTATION HAS BEEN PROVIDED. CONFIRM.
	10. EACH VENT LINE SHALL BE C VALVE/GLOBE VALVE. 11. NOTE DELETED.	LOMBINATION OF A BALL VALVE AND A PLOG
דו       	<ol> <li>NOTE DELETED.</li> <li>FLOW OF 52000 SCMD CON THE COMBINED FLOW OF 0</li> <li>SOLAR PANEL SYSTEM TO CONTROL ROOM ALONG WIT INSTALLED AT INJECTION F</li> <li>NOTE DELETED.</li> </ol>	NSIDERED FOR BABEJIA INJECTION POINT IS ONGC BABEJIA & ONGC JANTAPATHAR. BE INSTALLED AT SOURCE POINT & FH PANEL MOUNTED FLOW COMPUTER TO BE POINT.
-    		_
		ε
		D
	VALVE LEGEND	c = 1 1105 510
		E PDAGE END E BW END/UPTO 1/2" SW END. /E BW END/UPTO 1/2" SW END.
GC	니었다. Plug valv 니었다. Globe val	VE FLANGE END
		ORADHANUSH GAS GRID LIMITED
		मेकॉन लिमिटेड
	-	MECON LIMITED
	SECTION OIL & GAS LOCATION DELHI DESIGNED UMAR	NORTH EAST GAS GRID (PHASE-1 & 2 P/L SECTION)
10.	DRAWN UMAR CHECKED AND A.K. BHARTI	P & ID FOR CHECK METERING FACILITY (TYPE-II)
eserved. ID ISSUED N. THIS IS JECTS	VERIFIED SIG (S.GUPTA) DF	cale: NTS REV RG. NO. MELOSOVC/08/28/04/06/44400R-1002 3
3	DATE 10.10.24	2 1



<b>`</b>									∖	10.10.24		AS PER CLIENT	'S REQUIREMENT	UMAR	AKB			
										19.02.24		AS PER CLIEN	t's comments	UMAR	AKB	REFERENCES	0	RG.NO
									$\mathbb{A}$	04.01.24		GENERAL	REVISION	UMAR	AKB	C Copyright MECON LIMITE	D – All righ	ts res
REV	INST.								REV.NO	DATE	ZONE	DES	CRIPTION	BY	VERIFIED	THIS DRAWING IS THE PROPE FOR THE SPECIFIC PROJECT	RTY OF MECO	N AND
SEC		C	CONCURRED	BY								RE	ISION			NOT TO BE COPIED OR USE UNLESS EXPRESSLY PERMITT	FOR OTHER	PROJE
-	12			11	10	9	8	7				6		5		4		

	1			•		1		-
۵	<u> </u>	I		2	I			
<u> </u>	NOTES:							
1	1. ALL N	ENTS SHAL	l be locat	'ED MIN. HI	EIGHT OF 3 MT	R. ABOVE THE HIGHES	r	
	OPER	ATING LEVEL						
	2. METE PANE	ring shall L mounted	FLOW COM	by orifice Puter. Ins	METER (SINGL TALLATION SHA	LL BE AS PER AGA3 L	ATEST	Н
i	EDITIO	ON. HOWEVE	R THERE S	HALL BE TV	VO STREAMS O	F METERING (1W+1S)	FOR	
i	CHEC	K METERING	<i>.</i>					
	3. ENVIE	RONMENTAL	ENCLOSURE	REQUIRED	FOR METERING	g instruments. Ise lines of metering		
1	INSTR	RUMENT TO	BE PROVIDE	ED.			Í	
	4. PROV	ISION SHALL	L BE KEPT	то ноок и	JP METERING F	PR. & TEMP., FLOW		
	5. INLET	METERS WIT & OUTLET	h Scada. Of Skid S	HALL BE P	ROVIDED WITH	COMPANION FLANGES.		
Í	6. ALL	BALL VALVE	s shall be	FULL BOR	E ONLY.			
	7. SIZE, ENGI	, RATING & NEERING.	SET PRESS	URE OF PS	SV's TO BE DE	CIDED DURING DETAILED	)	
i	8. SKID	SHALL BE	SYMMETRIC	ACROSS C	ENTRE LINE.			
i	9. FUTU HOWI	IRE PROVISI EVER M/s I	on for cai Ggl has to	PACITY AUG CONFIRM.	MENTATION HAS	S BEEN PROVIDED.		
ļ	10. EACH	VENT LINE	SHALL BE	COMBINATI	ON OF A BALL	VALVE AND A PLUG		
	11. 20D	UPSTREAM	SHALL BI	E MAINTAI	NED FOR GC	SAMPLE POINT. APAR	r	
3"v12"	FRO DEW	M GC, NEC POINT) AI	ESSARY AN LSO TO BE	NALYZER ( INSTALLE	WATER DEW P D.	OINT & HYDROCARBO	N	
μ	12. NOT	E DELETED.	00 COMD (	ONGINEDE	D BOD KODIC	THE INTERMENT DOTS		
	THE	COMBINE	) FLOW OF	ONGC KO	RAGHAT & ON	IGC NAMBAR.	1 15	F
₽₹	14. SOL CON	AR PANEL TROL ROOM	SYSTEM TO	) BE INST TTH PANRI	ALLED AT SOU	URCE POINT & LOW COMPUTER TO F	ar I	
~~~~~	INST	ALLED AT	INJECTION	POINT.			-	
1								
į								F
i								
								D
							Ì	
								c
		ALVE LEO	<u>GEND</u>					
		чр₫н	BALL VAI	LVE FLANG	E END			
			BALL VAI	LVE BW EN	וD/UPT0 1/2' 11/2 חוקע	'SWEND. "SWEND		
		- - - - - - - - - - - - - -	PLUG VA	LVE FLANG	E END	SW LND.		
		ч₩	GLOBE V	ALVE.				
	6		IN	DRAI	HAND	ISH GAS		
		IGGL		GR	ID LIN	AITED		B
	/			÷	कॉन जि	गिन्देव		
					াদ্যাল লে	1465		
	-00	मकान		MEC	ON L	MITED		_
	SECTION	01 COM.						
	LOCATION	DELHI		(PI	NORTH EA IASE-1 &	ST GAS GRID 2 P/L SECTION)		
	DESIGNED	UMAR				· ·		•
NO.		A.K.RHAPTI		P&ID	OR CHECK ME	TERING FACILITY (TYPE-	-ш)	
ND ISSUED	VERIFIED			SCALE : NT	5		REV	
in. This is Jects	APPROVED	DATE	(S.GUPTA) 10.10.24	drg. no.Pm	29.9.vc/08/28	MOGAFEEDER-1003	3	



		12			11	10	9		8	7		1		6		5		4		
SE	c		C	ONCURRED	ЗY			-						REV	ISION			NOT TO BE COPIED OR USE UNLESS EXPRESSLY PERMIT	D FOR OTHER PR ED BY MECON.	ROJE
REV		IST.									REV.NO	DATE	ZONE	DESC	RIPTION	BY	VERIFIED	THIS DRAWING IS THE PROP FOR THE SPECIFIC PROJECT	ERTY OF MECON MENTIONED THEF	AND
											$\mathbb{A}$	04.01.24		GENERAL	REVISION	UMAR	AKB	C Copyright MECON LIMIT	D - All rights	re
											A	19.02.24		AS PER CLIEN	t's comments	UMAR	AKB	REFERENCES	DRO	G.NO
<b>`</b>						]					A	10.10.24		AS PER CLIENT	'S REQUIREMENT	UMAR	AKB			
											_									

	3			2		1			
A									
<u> </u>	NOIES:								
!	1. ALL V	ENTS SHA	LL BE LOCAT	'ED MIN. H	HEIGHT OF	3 MTR. ABOVE THE H	IIGHEST	·	
;	OPER/	ATING LEV	EL.						
Í									I
	Z. MEIE	RING SHAL	L BE DUNE		E MEIER (	SINGLE CHAMBER) WIII		TECT	Н
		N HOWEN	FR THERE S	HALL RE 1	WALLAHON	MS OF METERING (1W.	⊥1S) F		
i	CHEC	K METERIN	IC INCLUS				113/1		
	3. ENVIR	RONMENTAL	. ENCLOSURE	REQUIRE	D FOR MET	ERING INSTRUMENTS.			
i	INSUL	ATION OF	ORIFICE MET	ER, METER	RRUN &	MPLUSE LINES OF ME	TERING		-
!	INSTR	UMENI IC	BE PROVIDI	<u>-</u> D.					
	4. PROV	ISION SHA	ll be kept	to hook	UP METER	ING PR. & TEMP., FLO	WC		
	PARA	METERS W	ith scada.						
!	5. INLET	& OUTLE	t of skid s	HALL BE	PROVIDED	WITH COMPANION FLAN	IGES.		
	6. ALL I	BALL VALV	es shall be	FULL BO	RE ONLY.				G
	7. SIZE,	RATING &	SET PRESS	URE OF F	PSV's TO B	e decided during de	TAILED		
	ENGI	NEERING.							
i i	8. SKIU	SHALL BE	STMMETRIC	ACRUSS (	CENTRE LIP				
1	9. FUIU HOWF	VFR M/s	IGGI HAS TO	CONFIRM	GMENTATIO	n has been providel	).		
	10 FACH		IF SHALL BE	COMBINAT	ION OF A	BALL VALVE AND A PL	LUG		
	VALV	E/GLOBE	VALVE.	001101111			200		
i	11. 20D	UPSTREA	M SHALL B	E MAINTA	INED FOR	GC SAMPLE POINT.	APAR'	r	
1	FRO	GC, NE	CESSARY AI	VALYZER	(WATER D	EW POINT & HYDRO	CARBO	N	
5"x12"	DEW	POINT)	ALSO TO BE	INSTALL	ÈD.				
머케	12. NOTE	DELETED					-	_	
	13. CON INST	TROL ROU	DM ALONG T SOURCE P	OINT	SL MOUNT	ED FLOW COMPUTER	TOB	96	F
	1101		bookon i						
t 🖞									
≥⊈'∞!									
<u>⊣</u> ¦∣									
IOTE-9									
I									
i									
i									Е
Í									
									ľ
									-
	V/	ALVE LE	EGEND						C
		н⊳бн	BALL VA	LVE FLAN	GE END				
		-12-	BALL VA	LVE BW E	ND/UPTO	1/2" SW END.			
		┙╳┙	PLUG VA	LVE BW B	end/upto	1 <sup>1</sup> /2" SW END.			
		ы∕дн	PLUG VA	LVE FLAN	GE END				
		ч¤а⊦	GLOBE V	ALVE.					
				<b>n r</b> · ·	<b>n</b> · ·		~		
			IN	DRA	DHA]	NUSH GA	S		в
		ICCI I		GF	SID 1	LIMITED			
		of the last		<u> </u>					
				-	$\rightarrow$	निमिनेट			
		<b>20</b> \			नकान	191465			
		(كيم							
	6	नणन / २		мг	NOT	LIMITED			
	800	0 1 COMP							
	SECTION		;						
	LOCATION	DELHI		/ <i>.</i> _	NORTH	EAST GAS GRID	)		
	DESIGNED	UMAR		(P	HASE-1	& 2 P/L SECTI	ON)		
		UMAR							A
	CHECKED			рьт		K METERING PACIFIC	(TYDB)	<sub>πν</sub> ,	
υ.	AND	A.K.BHARTI		r et ID	FOR CHISC	a Mining FACILITY (	(11 <b>PB</b> -	-**	
served.	VERIFIED			0011-	TC				
D ISSUED	APPROVED	SIG	(S.GUPTA)	SCALE : N		87. of 644		REV	
LCTS		DATE	10.10.24	URG. NO.	MECY2SVC/	Je/28/11/0617FEEDER-1	1004	3	



					_														
L																			
1										▲	10.10.24		AS PER CLIENT	's requirement	UMAR	AKB			
										∕	19.02.24		AS PER CLIEN	I'S COMMENTS	UMAR	AKB	REFERENCES	D	RG.N
											04.01.24		GENERAL	REVISION	UMAR	AKB	C Copyright MECON LIMITE	D - All right	its re
	REV	INST.								REV.NO	DATE	ZONE	DESC	RIPTION	BY	VERIFIED	THIS DRAWING IS THE PROPI	ERTY OF MECON MENTIONED TH	N AN
	SEC		С	ONCURRED	BY								REV	ISION			NOT TO BE COPIED OR USE UNLESS EXPRESSLY PERMITT	D FOR OTHER ED BY MECON.	PROJ
L		12			11	10	9	8	7				6		5		4		

	3			2		1		1
<u> </u>	NOTES:							
	1. ALL V	ENTS SHA	LL BE LOCAT	ed Min. Hei	GHT OF	3 MTR. ABOVE THE HIG	HEST	
	2. METEI PANEI EDITIC	RING SHAL L MOUNTEI )N. HOWEV	L BE DONE D FLOW COM ER THERE SI	By Orifice Puter. Inst. Hall be two	Meter (S Allation D stream	Single Chamber) with Shall be as per ag, Is of metering (1W+	CONTROL A3 LATEST 1S) FOR	н
	3. ENVIR	ONMENTAL	ENCLOSURE	REQUIRED I ER, METER I	FOR MET RUN & II	ering instruments. Mpluse lines of mete	RING	
i	INSTR 4. PROV	:UMENT TO ISION SHAI	be provide	ed. To hook ui	> Meteri	NG PR. & TEMP., FLOW	v	
	Parai 5. Inlet	WETERS WI & OUTLE	th Scada. T of skid s	HALL BE PR	ovided N	WITH COMPANION FLANG	ES.	
İ	<ol> <li>ALL I</li> <li>SIZE, ENGIN</li> </ol>	BALL VALVI RATING & NEERING.	LS SHALL BE \$2 SET PRESS	URE OF PSV	. ONLY. ''s TO BI	e decided during det	AILED	G
	8. SKID 9. FUTU	Shall be Re provis	SYMMETRIC	ACROSS CEI PACITY AUGM	NTRE LIN	e. I has been provided.		
	HOWE 10. EACH VALV	VER M/s VENT LIN F/GLOBF V	iggl has to E shall be /aivf	CONFIRM.	N OF A	BALL VALVE AND A PLU	IG	
.4" <u>x</u> 12"	11. 20D FROI DEW	UPSTREA I GC, NE POINT) A	M SHALL BI CESSARY AN ALSO TO BE	8 MAINTAIN ALYZER (W INSTALLED	ED FOR Ater di	GC SAMPLE POINT. A W POINT & HYDROCA	APART ARBON	
	12. NOTE 13. SOLA CON	R PANEL R PANEL ROL ROO	SYSTEM TO M ALONG W	) BE INSTA ITH PANEL	LLED AT MOUNTI	SOURCE POINT & ED FLOW COMPUTER 1	IO BE	F
	INST	ALLED AT	INJECTION	POINT.				
NOTE-9								
i I								E
								D
	<u></u>	ALVE LE	<u>EGEND</u>					c
		+D¢l+ +D¢l+	Ball Vai Ball Vai	LVE FLANGE LVE BW END	end D/upto	1/2" SW END.		
			PLUG VA PLUG VA	LVE BW EN LVE FLANGE	d/upto : End	11/2" SW END.		
		HMH	GLUBE V					
	6		IN	DRAD	HAN	NUSH GAS	5	в
		GGL		GRI	D I	LIMITED		
]				मे	कॉन	लिमिटेड		
	50000	HITT COMPA		MEC	ON	LIMITED		
	SECTION	OIL & GAS DELHI		(PH	NORTH	EAST GAS GRID & 2 P/L SECTIO	N)	1
	DESIGNED	UMAR			-	/	, ,	•
NO.	AND VERIFIED	A.K.BHARTI		SCALE : NTC	UR CHEC	A MISILAGING FACILITY (]	.1FB-V)	
IN. THIS IS JECTS	APPROVED	SIG DATE	(S.GUPTA) 10.10.24	DRG. NO.PME	gevc/d	8/28/01/06/44Eder-10	05 3	1



REV

INST.

CONCURRED BY

3 NOTES: A NOTES: 1. ALL VENTS SHALL OPERATING LEVEL 2. METERING SHALL PANEL MOUNTED	2 1 . BE LOCATED MIN. HEIGHT OF 3 MTR. ABOVE THE HIGHEST BE DONE BY ORIFICE METER (SINGLE CHAMBER) WITH CON' FLOW COMPUTER. INSTALLATION SHALL BE AS PER AGA3 LA R THERE SHALL BE TWO STREAMS OF METERING (1W+1S) F ENCLOSURE REQUIRED FOR METERING INSTRUMENTS. WRIFICE METER, METER RUN & IMPLUSE LINES OF METERING	TROL H TEST OR
A NOTES: 1. ALL VENTS SHALL OPERATING LEVEL 2. METERING SHALL PANEL MOUNTED	. BE LOCATED MIN. HEIGHT OF 3 MTR. ABOVE THE HIGHEST BE DONE BY ORIFICE METER (SINGLE CHAMBER) WITH CON FLOW COMPUTER. INSTALLATION SHALL BE AS PER AGA3 LA R THERE SHALL BE TWO STREAMS OF METERING (1W+1S) F ENCLOSURE REQUIRED FOR METERING INSTRUMENTS. WRIFICE METER, METER RUN & IMPLUSE LINES OF METERING	TROL H TEST OR
OPERATING LEVEL 2. METERING SHALL PANEL MOUNTED	BE DONE BY ORIFICE METER (SINGLE CHAMBER) WITH CON FLOW COMPUTER. INSTALLATION SHALL BE AS PER AGA3 LA R THERE SHALL BE TWO STREAMS OF METERING (1W+1S) F ENCLOSURE REQUIRED FOR METERING INSTRUMENTS. MRIFICE METER, METER RUN & IMPLUSE LINES OF METERING	TROL H (TEST OR
	r there shall be two streams of metering (1W+1s) f Enclosure required for metering instruments. Rifice meter, meter run & impluse lines of metering	OR
EDITION. HOWEVE	RIFICE METER, METER RUN & IMPLUSE LINES OF METERING	
INSULATION OF C	BE PROVIDED.	H
A. PROVISION SHALL PARAMETERS WITH 5. INLET & OUTLET	. BE KEPT TO HOOK UP METERING PR. & TEMP., FLOW 4 SCADA. OF SKID SHALL BE PROVIDED WITH COMPANION FLANGES. 5 SHALL BE FULL BORF ONLY.	
7. SIZE, RATING & ENGINEERING. 8. SKID SHALL BE	SET PRESSURE OF PSV'S TO BE DECIDED DURING DETAILED	G
9. FUTURE PROVISIO HOWEVER M/S IO 10. EACH VENT LINE	IN FOR CAPACITY AUGMENTATION HAS BEEN PROVIDED. GL HAS TO CONFIRM. SHALL BE COMBINATION OF A BALL VALVE AND A PLUG	
VALVE/GLOBE VA 11. 20D UPSTREAM FROM GC, NEC DEW POINT) AI	LVE. SHALL BE MAINTAINED FOR GC SAMPLE POINT. APAR ESSARY ANALYZER (WATER DEW POINT & HYDROCARBO SO TO BE INSTALLED.	r N
12. NOTE DELETED.	SYSTEM TO BE INSTALLED AT SOURCE POINT &	F
	INJECTION POINT.	5
		H
		Ē
		D
	SEND	с
	BALL VALVE FLANGE END BALL VALVE BW END/UPTO 1 <sup>1</sup> /2" SW END.	
	PLUG VALVE BW END/UPTO 1/2" SW END. PLUG VALVE FLANGE END GLOBE VALVE.	H
IGGL	INDRADHANUSH GAS GRID LIMITED	в
	मेकॉन लिमिटेड	
SECTION OIL & CAS	MECON LIMITED	]
LOCATION DELHI DESIGNED UMAR	NORTH EAST GAS GRID (PHASE-1 & 2 P/L SECTION)	
IO. CHECKED AND VERIFIED CHECKED A.K.BHARTI	P & ID FOR CHECK METERING FACILITY (TYPE-	VI)
ID ISSUED N. THIS IS JECTS APPROVED SIG DATE	S.GUPTA) SCALE : NTS 08.01.25 DRG. NO. MELGEN/C/08/28/M/061/FEEDER-1006	REV 4

UMAR

UMAR AKB

BY VERIFIED

GENERAL REVISION

DESCRIPTION

REVISION

REV.NO DATE ZONE

DRG.

REFERENCES

C Copyright MECON LIMITED - All rights THIS DRAWING IS THE PROPERTY OF MECON AN FOR THE SPECIFIC PROJECT MENTIONED THERE NOT TO BE COPIED OR USED FOR OTHER PRO UNLESS EXPRESSIVE PREMITTED BY MECON

# ANNEXURE - IV INSTALLATION DRAWING

## LIST OF STANDARD DRAWINGS

SI. No.	DRAWING No.	DESCRIPTION	REV.	
1	MEC/05/E5/SD/PT/001	INSTALLATION DIAGRAM FOR PRESSURE TRANSMITTER.	0	
2	MEC/05/E5/SD/RTD/002	MOUNTING DETAILS FOR RTD ON THERMOWELLS.	0	
3	MEC/05/E5/SD/DPG/003	INSTALLATION DIAGRAM FOR DIFERENTIAL PRESSURE GAUGE.	0	
4	MEC/05/E5/SD/PG/004	INSTALLATION DIAGRAM FOR PRESSURE GUAGE.	0	
5	MEC/05/E5/SD/TG-RTD/005	INSTALLATION DIAGRAM FOR TEMPERATURE GUAGE & RTD.	0	
6	MEC/05/E5/SD/RTD-TW/006	MOUNTING DETAILS FOR THERMOWELLS (RTD).	0	
7	MEC/05/E5/SD/TG-TW/007	MOUNTING DETAILS FOR THERMOWELLS (T G).	0	
8	MEC/05/E5/SD/JB/009	SUPPORT DETAILS FOR JUNCTION BOXES.	0	
9	MEC/05/E5/SD/CY/011	FABRICATED CANOPY FOR INSTRUMENTS.	0	









Г	5 4 3 2 1	
6		G
	[ <sup>-</sup> ]	
F		F
	I-I/Z FLANGED NOZZLE	
	THERMOWELL	
E		E
	INSTRUMENTATION SCOPE	
D	- PIPING SCOPE	D
	NOTES:	
	1) PIPING SCOPE INCLUDES SUPPLY & INSTALLATION OF 1-1/2" SOCK-O-LET	
C	1-1/2" NOZZLE, $1-1/2$ " FLANGE ALONG WITH GASKET, STUD & NUTS.	С
	2) INSTRUMENTATION SCOPE INCLUDES SUPPLY OF FLANGED THERMOWELL ALONG WITH THE INSTRUMENT.	
	_,	
		П
B		В
	REV NO DATE ZONE DESCRIPTIONS BY APPRD	Ħ
	REFERENCES DRG. NO. THIS DRAWING IS PROPERTY OF MECON AND IS ISSUED FOR THE SPECIFIC PROJECT MENTIONED THEREIN. THIS IS NOT TO BE COPIED OR USED FOR OTHER PROJECTS UNLESS EXPRESSLY PERMITTED BY MECON.	
	SECTION : ELECTRICAL/INSTN. मैकान लिमिटेड	
A	NAME     Date     CHKD     Date       DSGN     R.K.S     P.S     INSTALLATION     DIAGRAM	A
	DRWN IRFAN TEMPERATURE GUAGE & RTD	
	APPROVED D.G.M D.G.M SCALE : REV DRG.NO. MEC/05/E5/SD/TG-RTD/005 0	
	5   4   3   2   Page 196 of 644	1























## ANNEXURE – V

## LIST OF VENDORS FOR BOUGHT-OUT ITEMS

Page 208 of 644

### **INSTRUMENTATION**

#### I) PRESSURE REGULATOR AND SLAM SHUT VALVE

- 1) M/s Pietro Fiorentini S.P.A. (Italy)
- 2) M/s Emerson Process Management (Singapore)
- 3) M/s RMG-Regel Messtechnik (Germany)
- 4) M/s Nirmal Industrial Controls (India)
- 5) M/s Gorter Controls (Netherlands)
- 6) M/s Dresser

#### II) **PID CONTROLLER**

- 1) M/s ABB
- 2) M/s EUROTHERN
- 3) M/s TATA HONEYWELL
- 4) M/s MASIBUS

## III) ORIFICE FITTINGS (METER RUN, FLOW CONDITIONER & ORIFICE PLATE, ORIFICE ASSLY)

- 1) M/s Emerson Process Management (Daniel), USA
- 2) M/s Canalta Controls, Canada
- 3) M/s FMC, USA
- 4) M/s Pietro Fiorentini, Italy

#### IVA) PANEL MOUNTED FLOW COMPUTERS

- 1) M/s Barton Instruments System Ltd.
- 2) M/s Daniel Measurement and Controls
- 3) M/s Instromet International, Belgium
- 4) M/s RMG Messtechnik GmbH
- 5) M/s Omni Flow Computers Inc.

### IVB) FIELD MOUNTED FLOW COMPUTER

- 1) M/s Barton Instruments System LLC
- 2) M/s Daniel Measurement and Controls
- 3) M/s Bristol Babcock

#### V) GAS CHROMATOGRAPH

- 1) ABB Ltd , India
- 2) Daniel Measurement & Control Asia Pacific, India
- 3) Instromet International, NV
- 4) RMG Regal+Messtechnik GmbH

### VI) L.E. L DETECTION SYSTEM

Page **1** of **12** Page 209 of 644

- 1) Crowcon Detection Instruments Ltd
- 2) Detection Instruments (I) Pvt Ltd
- 3) Detector Electronics Corporation
- 4) MSA Mines safety appliances.
- 5) Oldham France S.A.
- 6) Honeywell

### VII) CONTROL AND SIGNAL CABLES

- 1) M/s ASSOCIATED CABLES
- 2) M/s ASSOCIATED FLEXIBLES
- 3) M/s DELTON Cables Ltd, India
- 4) M/s BROOK
- 5) M/s KEI Industries Ltd INDIA
- 6) M/s Suyog Electricals Ltd, India
- 7) M/s Thermo Cables Ltd
- 8) M/s Udey Pyrocables Pvt Ltd, India
- 9) M/s T C Communication Ltd.

#### VIII) ZENER BARRIERS/ISOLATORS

- 1) M/s MTL
- 2) M/s P & F

### IX) <u>RTDs</u>

- 1) M/s General Instruments Ltd., Mumbai
- 2) M/s Nagman Sensors (Pvt.) Ltd.
- 3) M/s Pyro Electric, Goa

### X) PRESSURE, TEMPERATURE & DIFF. PRESSURE TRANSMITTER

- 1) M/s Fisher Rosemount (Emerson)
- 2) M/s Yokogawa
- 3) M/s Fuji
- 4) M/s Honeywell

#### XI) <u>PRESSURE GAUGES, TEMPERATURE GAUGES & DIFFERENTIAL</u> <u>PRESSURE GAUGE</u>

- 1) M/s AN Instruments Pvt. Ltd., New Delhi
- 2) M/s General Instruments Ltd., Mumbai
- 3) M/s WIKA

### XII)A <u>SS TUBE</u>

1) Sandvik, Sweden

Page **2** of **12** Page 210 of 644

- 2) Choksy Tube Co Ltd.
- 3) Heavy Metals & Tubes Ltd.
- 4) Nuclear Fuel Complex, India
- 5) Scorodite
- 6) Ratnamani Metal & Tubes Ltd
- 7) Jindal Saw

#### XII)B SS TUBE FITTINGS

- 1) M/s Swagelok (USA)
- 2) M/s Parker (USA)
- 3) M/s Excellsior
- 4) M/s Reliance
- 5) M/s Multimetal
- 6) M/s Comfit
- 7) M/s Aura Inc
- 8) M/s Arya Crafts
- 9) M/s Swastic

#### XII)C SS VALVES & MANIFOLDS

- 1) M/s Swagelok (USA)
- 2) M/s Parker (USA)
- 3) M/s Excellsior
- 4) M/s Comfit
- 5) M/s Aura Inc
- 6) M/s Arya Crafts
- 7) M/s Swastic

#### XIII) JUNCTION BOXES AND CABLES GLANDS

- 1) M/s EX-PROTECTA
- 2) M/s FLAMEPROOF CONTROL GEARS
- 3) M/s BALIGA
- 4) M/s FLEXPRO ELECTRICALS

#### XIV) PUSH BUTTONS/LAMPS:

- 1) L&T
- 2) SIEMENS

#### XV) <u>MCB'S:</u>

- 1) HAVELL'S
- 2) INDO ASIAN
- 3) MDS

## XVI) <u>RELAYS:</u>

- 1) OEN
- 2) JYOTI

Page **3** of **12** Page 211 of 644

#### XVII) POWER SUPPLY UNIT:

- 1) ELNOVA
- 2) APLAB

#### XVIII) CONTROL ROOM EQUIPMENT CONTROL PANEL & ACCESSORIES

- 1) M/s Keltron Controls Ltd., Kerala
- 2) M/s RITTAL
- 3) M/s Pyrotech
- 4) M/s Positronics Pvt. Ltd.
- 5) M/s ABB Instruments Ltd., New Delhi
- 6) M/s Emerson Process Management (I) Pvt. Ltd.
- 7) M/s Rockwell Automation (I) Ltd., Ghaziabad
- 8) M/s Siemens Ltd.
- 9) M/s Tata Honeywell Ltd.

#### XIX) <u>PRINTERS</u>

- 1. Compaq
- 2. Dell
- 3. IBM
- 4. HP
- 5. EPSON

#### XX) ACTUATOR

- 1. Rotork
- 2. Schuck
- 3. Biffi
- 4. Ledeen

#### **MECHANICAL**

#### i) PIPE CARBON STEEL TO INDIAN STANDARDS

- 1. A.S.T. PIPES PVT. LTD. (AST GROUP)
- 2. ADVANCE STEEL TUBE LTD.
- 3. APL APOLLO TUBES LTD. (ER. BIHAR TUBES LTD.
- 4. ASIAN MILLS PVT. LTD.
- 5. ASRANI TUBES LIMITED
- 6. DADU PIPES (P) LTD.
- 7. ESSAR STEEL LIMITED(ER HAZIRA PIPES MILL)
- 8. GAURANG PRODUCTS PVT LTD. (AST GROUP)
- 9. GOODLUCK STEEL TUBES LTD.
- 10. HI-TECH PIPES LIMITED
- 11. INDUS TUBE LIMITED
- 12. JINDAL INDUSTRIES LTD
- 13. JINDAL PIPES LTD.
- 14. JINDAL SAW LTD (KOSI WORKS)
- 15. JOTINDRA STEEL & TUBE LTD
- 16. LALIT PIPES AND PIPES LTD.

#### Page **4** of **12** Page 212 of 644

- 17. MAHARASHTRA SEAMLESS LTD.
- 18. MAN INDUSTRIES (INDIA) LTD. PITHAMPUR
- 19. MAN INDUSTRIES (INDIA) LTD. ANJAR
- 20. MUKAT TANKS & VESSELS LTD.
- 21. NEZONE TUBES LIMITED
- 22. NORTH EASTERN TUBES LIMITED
- 23. PRATIBHA INDUSTRIES LIMITED
- 24. PRATIBHA PIPES & STRUCTURAL LTD.
- 25. PSL LTD (CHENNAI)
- 26. PSL LTD (V1, V2 & NC)
- 27. RAMA STEEL TUBES LTD.
- 28. RATNAMANI METALS AND TUBES LTD.
- 29. RAVINDRA TUBES LIMITED
- 30. SAMSHI PIPE INDUSTRIES LIMITED
- 31. SURYA ROSHNI LTD.
- 32. SWASTIK PIPES LTD.
- 33. UTKARSH TUBES & PIPES LTD. (FORMLY BMW)
- 34. WELSPUN CORP. LIMITED (DAHEJ)
- 35. ZENITH BIRLA (INDIA) LIMITED

#### ii) PIPE & TUBULARS TO A.P.I. STANDARDS

- 1. ARCELORMITTAL TUBULAR PRODUCTS ROMAN SA, ROMANIA
- 2. BHEL (TRICHY),INDIA
- 3. DALMINE SPA (ENQUIRY TO TENARIS), UAE
- 4. EEWKOREA CO. LTD (GERMANY), KOREA
- 5. EEW KOREA CO. LTD. (KOREA), KOREA
- 6. EISENBAU KRAMER GMBH, GERMANY
- 7. HYUNDAI RB CO. LTD. SOUTH KOREA
- 8. ILVA LAMIERE E TUBI SRL (ENQ TO ILVA SPA, ITALY
- 9. INOX TECH. SPA, ITALY
- 10. ISMT LTD. AHMEDNGR, INDIA
- 11. ISMT LTD. BARAMATI, INDIA
- 12. JINDAL PIPES LTD., INDIA
- 13. JINDAL SAW LTD. (KOSI WORKS), INDIA
- 14. JINDAL SAW LTD. (NASHIK WORKS), INDIA
- 15. LALIT PIPES AND PIPES LTD. INDIÁ
- 16. MAHARASHTRA SEAMLESS LTD., INDIA
- 17. MAN INDUSTRIES (I) LTD. (PITHAMPUR), INDIA
- 18. MUKAT TANKS & VESSELS LTD., INDIA
- 19. PRATIBHA INDUSTRIES LIMITED, INDIA
- 20. RATNAMANI METALS AND TUBES LTD., INDIA
- 21. SIDERCA S.A.I.C (ENQUIRY TOTENARIS), UAE
- 22. SUMITOMO METÀL IND. LTD., INDIA
- 23. SURYA ROSHNI LTD., INDIA
- 24. SWASTIK PIPES LTD, INDIA
- 25. TATA STEEL UK LIMITED (FORMERLY C702)
- 26. TUBOS DE ACERO DE MEXICO SA (ENQ. TENARIS), UAE
- 27. TUBOS REUNIDOS SA SPAIN
- 28. UMRAN STEEL PIPE INC (TURKEY), TURKEY
- 29. VALCOVNY TRUB CHOMUTOV, CZECH REPUBLIC
- 30. VALLOUREC AND MANNESMANN TUBES, FRANCE
- 31. WELSPUN CORP LIMITED (DAHEJ), INDIA

#### iii) <u>PIPE/TUBE CS (SEAMLESS) TO ASTM STDS</u>

- 1. ARCELORMITTAL TUBULAR PRODUCTS ROMAN SA, ROMANIA
- 2. BHEL (TRICHY), INDIA
- 3. CHANGSHU SEAMLESS STEEL TUBE CO. LTD., CHINA
- 4. DALMINE SPA (ENQUIRY TO TENARIS, UAE
- 5. HEAVY METALS & TUBES LIMITED (MEHSANA), INDIA

Page **5** of **12** Page 213 of 644

- 6. ISMT LTD. AHMEDNGR, INDIA
- 7. ISMT LTD. BARAMATI INDIA
- 8. JFE STEEL CORPORATION, UAE
- 9. JINDAL SDAW LTD (NASHIK WORKS) INDIA
- 10. KLT AUTOMOTIVE AND TUBULAR PRODUCTS LTD., INDIA
- 11. MAHALAXMI SEAMLESS LIMITED, INDIA
- 12. MAHARASHTRA SEAMLESS LTD, INDIA
- 13. PRODUCTS TUBULARES S.A.U, SPAIN
- 14. RATNADEEP METAL TUBES LTD., INDIA
- 15. STAINEEST TUBES PVT LTD., INDIA
- 16. SUMITOMO METAL IND. LTD., INDIA
- 17. TUBOS REUNIDOS SA SPAIN
- 18. VALCOVNY TRUB CHOMUTOV, CZECH REPUBLIC
- 19. VALLOUREC ANDMANNESMANN TUBES FRANCE
- 20. YANGZHOU CHENGDE STEEL PIPE CO. LTD DUBAI (UAE)

#### iv) PIPE CARBON STEEL (WELDED) TO ASTM STDS

- 1. EEW KOREA CO. LTD. (GERMANY), KOREA
- 2. EEW KOREA CO. LTD. (KOREA), KOREA
- 3. EISENBAU KRAMER GMBH, GERMANY
- 4. HYUNDAI RB CO. LTD., SOUTH KOREA
- 5. INOX TECH. SPA, ITALY
- 6. JINDAL SAW LTD (KOSI WORKS), INDIA
- 7. LALIT PIPES AND PIPES LTD., INDIA
- 8. MAN INDUSTERIES (I) LTD. (PITHAMPUR), INDIA
- 9. MAN INDUSTRIES (INDIA) LTD. ANJAR, INDIA
- 10. MUKAT TANKS & VESSELS LTD., INDIA
- 11. RATNAMANI METALS AND TUBES LTD., INDIA
- 12. SUMITOMO METAL INDIA LTD., INDIA
- 13. TATA STEEL UK LIMITED
- v) <u>Valve</u>

#### a) Globe Valves

- 1) M/s Weir BDK Valves (Aunit of Weir India Pvt. Ltd.)
- 2) M/s Datre Corpn (Calcutta)
- 3) M/s KSB Pumps Ltd., Coimbatore, India
- 4) M/s L&T Audco
- 5) M/s Neco Schuber & Salzer Ltd. (New Delhi)
- 6) M/s Niton Valve India Pvt. Ltd., India
- 7) M/s Ornate Valves (Mumbai)
- 8) M/s Panchavati Valves & Flages (P) Ltd., India
- 9) AV Valves Ltd., India
- 10) BHEL (Trichy), India
- 11) Econo Valves Pvt Ltd, India
- 12) Fouress Engg (I) Ltd (Aurangabad), India
- 13) Leader Valves Ltd, India
- 14) Oswal Industries Ltd, India
- 15) Petrochemical Engineering Enterprises, India (Fouress Group)
- 16) Sakhi Engineers Pvt Ltd., India
- 17) Shalimar Valves Pvt Ltd., India
- 18) Steel Strong Valves India Pvt Ltd, India
- 19) Petro Valves Pvt. Limited, Ahmedabad
- 20) Fluid Line Valves Co. (P) Ltd., India
- 21) MICON Engineers (Hubli) (P) Ltd., India
- b) Check Valves

Page **6** of **12** Page 214 of 644

- 1. M/s Advance Valves Pvt. Ltd., Noida
- 2. M/s Aksons & Mechanical Enterprises, Mumbai
- 3. M/s Larsen & Toubro Limited (M/s Audco India Limited, Chennai)
- 4. M/s AV Valves Ltd., India
- 5. M/s Weir BDK Valves (A unit of Weir India Pvt. Ltd.)
- 6. M/s BHEL, Trichy
- 7. M/s Datre Coroportion Limited, Calcutta
- 8. M/s Leader Valves Ltd., Jalandhar
- 9. M/s Neco schubert &Salzer Ltd., New Delhi
- 10. M/s Niton Valves Industries (P) Ltd., Mumbai
- 11. M/s Precision Engg.Co., Mumbai
- 12. Econo Valves Pvt Ltd, India
- 13. Fouress Engg (I) Ltd (Aurangabad)
- 14. KSB Pumps Ltd (Coimbattore), India
- 15. NSSL Ltd. (Neco Schubert & SalzerLtd)
- 16. Oswal Industries Ltd, India
- 17. Panchvati Valves & Flanges Pvt Ltd, India
- 18. Petrochemical Engineering Enterprises, India (Fouress Group)
- 19. Sakhi Engineers Pvt Ltd
- 20. Shalimar Valves Pvt Ltd
- 21. Steel Strong Valves India Pvt Ltd, India
- 22) Fluid Line Valves Co. (P) Ltd., India
- 22. MICON Engineers (Hubli) (P) Ltd., India
- c) Plug Valves
- 1) M/s Breda Energia Sesto Industria Spa, Italy
- 2) M/s Fisher Sanmar Ltd., Chennai
- 3) M/s Larsen & Toubro Ltd., (Audco) New Delhi
- 4) M/s Nordstrom Valves, USA
- 5) M/s Serck Audco Valves, UK
- 6) M/s Sumitomo Corporation India Pvt. Ltd., New Delhi
- 7) M/s Z Corporation, Korea
- 8) M/s Hawa Valves (India) Pvt. Ltd., Mumbai
- 9) M/s Steel Strong Valves India Pvt. Ltd., Navi Mumbai
- 10) M/s Econo Valves Pvt. Ltd., India (WSSL Ltd. Group Co.)
- 11) M/s Flow-Serve PTE (Mfr. SERCK), India
- 12) M/s Galli Cassina SPA, Italy
- d) Ball Valves
- 1. M/s Hawa Valves (India) Pvt. Ltd, Navi Mumbai
- 2. M/s Larsen & Toubro (Audco), India
- 3. M/s Oswal Industries Ltd., India
- 4. M/s Virgo Engineers Ltd., Delhi
- 5. M/s Boteli Valve Group Co. Ltd., China
- 6. M/s Cameron Italy s.r.l., Italy
- 7. M/s Dafram S.P.A., Italy
- 8. M/s Fangyuan Valve Group Co. Ltd., China
- 9. M/s Franz Schuck GmbH, Germany
- 10. Kita Mura Valve Manufacturing Co.Ltd., India
- 11. Petrol Valve S.R. Italy
- 12. Piplviesse S.P.A. Italy
- 13. Tormene Gas Technology S.P.A. Valvetalia Group, Italy
- 14. Valbeot S.R.L. Italy
- 15. KMC Corporation, South Korea
- 16. MSA a.s. Czeek Republic
- 17. OMS Aleri, Italy

Page **7** of **12** Page 215 of 644

- 18. PCC Valves s.r.l. Italy
- 19. Perar s.p.a. (Engineering. To TRP srl), Italy
- 20. Italy s.r.l., Italy
- 21. MIR Valves, Malaysia
- vi) <u>Split Tee</u>

1) M/s lpsco, Canda

2) M/s TD Willamsons, USA

#### vii) <u>Flanges</u>

- 1. M/s Aditya Forge Ltd., Vadodara
- 2. M/s Amforge Industries Ltd., Mumbai
- 3. M/s CD Engineering Co., Ghaziabad
- 4. M/s Echjay Forgings Pvt. Ltd. (Bombay), Mumbai
- 5. M/s Echjay Industries Ltd., Rajkot
- 6. M/s Forge & Forge Pvt. Ltd., Rajkot
- 7. M/s Golden Iron & Steel Works, New Delhi
- 8. M/s JK Forgings, New Delhi
- 9. M/s Metal Forgings Pvt. Ltd., Mumbai
- 10. M/s Perfect Marketings Pvt. Ltd., New Delhi
- 11. M/s Sky Forge, Faridabad
- 12. M/s S&G, Faridabad
- 13. Chaudhry Hammer Works Ltd, India
- 14. JAV Forgings (P) Ltd, India
- 15. Kunj Forgings Pvt Ltd, India
- 16. MS Fittings Mgf. Co. Pvt. Ltd.
- 17. R.N. Gupta & Co. Ltd, India
- 18. R.P. Engineering Pvt Ltd, India
- 19. Sanghvi Forgings & Engineering Ltd
- 20. Shri Ganesh Forgings Ltd., India
- 21. Uma Shankar Khandelwal & Co., India
- 22. Sawan Engineers, Baroda
- 23. Stewarts & Lloyds of India Ltd., Kolkata
- 24. Engineering Services Enterprises
- 25. Abasi Engineersing Works, India
- 26. Anandmayee Forgings Pvt Ltd, India
- 27. CD Industries., India
- 28. Fivebros Forgings Vot Ltd., India
- 29. Good Luck Engineering Co., India
- 30. Korea Flange, South Korea
- 31. Lal Metal Forge Ltd, India
- 32. Melesi Officine
- 33. Amlrojie Melesi & C. srl. Italy
- 34. Nicola Galperti & Figlio S.P.AIndia
- 35. Paramount Forge, India
- 36. Pradeep Metal Limited, India
- 37. Punjab Steel Works (the), India
- 38. R.D.Forge, India
- 39. Shah Industrial & Comml. Corporation, India
- 40. Ulma Forja S. Coop.
- 41. Vivial Forge Pvt. Ltd., Vadodara
- viii) Fittings
  - 1. M/s Commercial Supplying Agency, Mumbai
  - 2. M/s Dee Development Engineers Ltd.
  - 3. M/s Eby Industries, Mumbai
  - 4. M/s Flash Forge Pvt. Ltd., Vishakhapatnam

Page **8** of **12** Page 216 of 644
- 5. M/s Gujarat Infra Pipes Pvt. Ltd., Vadodara
- 6. M/s M.S. Fittings Mfg. Co. Pvt. Ltd., Kolkata
- 7. M/s Stewarts & Lloyds of India Ltd., Kolkata
- 8. M/s Teekay Tubes Pvt. Ltd., Mumbai
- 9. M/s Pipe Fit,Baroda
- 10. M/s Sky Forge, Faridabad
- 11. M/s S&G, Faridabad
- 12. M/s Sawan Engineers, Baroda
- 13. Eby Fasteners, India
- 14. R.N. Gupta & Co. Ltd, India
- 15. Exten Engg Pvt Ltd
- 16. Sivananda Pipe & Fittings Ltd
- 17. Chero Piping SPA, Italy
- 18. CSA Fittings, India
- 19. EBY Fasteners, India
- 20. Fittnox SRL, Italy
- 21. Keonsae High Pressure Co. Ltd., South Korea
- 22. Munro & Miller Fittings Ltd., U.K.
- 23. TK Corporation, South Korea
- 24. Tube Turn (India) Pvt Ltd., India
- 25. Topaz Piping Industries, India
- 26. Technoforge SPA, Italy
- 27. P.K. Tubes & Fittings Pvt. Ltd., India
- 28. Vivial Forge Pvt. Ltd., Vadodara
- ix) <u>Gaskets</u>
  - 1. IGP Engineers (P) Ltd., Madras
  - 2. Madras Industrial Products, Madras
  - 3. Dikson & Company, Bombay
  - 4. Banco Products (P) Ltd., Vadodara
  - 5. Goodrich Gaskets Pvt Ltd
  - 6. Starflex Sealing India Pvt Ltd, India
  - 7. Teekay Meta Flex Pvt Ltd
  - 8. UNIKLINGER Ltd
  - 9. HEM Engg. Corp.
  - 10. Unique Industrial Packing Pvt. Ltd.

# x) <u>Fasteners</u>

- 1. Nireka Engg. Co. (P) Ltd., Calcutta
- 2. Precision Taps & Dies, Bombay
- 3. AEP Company, Vithal Udyoug Nagar
- 4. Fix Fit Fasteners, Calcutta
- 5. Precision Engg. Industries, Baroda
- 6. Echjay Forgings Pvt. Ltd., Bombay
- 7. Capital Industries, Bombay
- 8. Boltmaster India Pvt Ltd, India
- 9. Deepak Fasteners Limited, India
- 10. Fasteners & Allied Products Pvt Ltd, India
- 11. Hardwin Fasteners Pvt Ltd, India
- 12. J.J. Industries, India
- 13. Multi Fasteners Pvt Ltd, India
- 14. Nexo Industries, India
- 15. Pacific Forging & Fasteners Pvt Ltd, India
- 16. Pioneer Nuts & Bolts Pvt Ltd, India
- 17. Precision Auto Engineers, India
- 18. President Engineering Works, India
- 19. Sandeep Engineering Works, India

Page **9** of **12** Page 217 of 644

- 20. Syndicate Engineering Industries, India
- 21. BEA SRL, Italy
- 22. Korea Parts & Fasteners (KPF), South Korea
- 23. Kundan Industries Ltd., India
- 24. Mega Engineering Pvt. Ltd., India
- 25. OME Metallurgica ERBESE S.R.L, Italy
- 26. Pankaj International, India
- 27. Udehra Fasters Ltd., India

# xi) <u>Welding Electrodes</u>

- For Mainline Lincon make
   For Terminal –For root pass
  - For Terminal –For root pass Lincon Make
    - For other passes Lincon, D&H or equivalent make

# PROCESS

# I) CARTRIDGE FILTER

- 1) M/s Grand Prix Fab (Pvt.) Ltd. (New Delhi)
- 2) M/s Multitex Filteration Engineers Ltd. (New Delhi)
- 3) M/s Perry Equipment Corp. (USA)
- 4) M/s Siirtec NIGI SPA (Italy)
- 5) M/s Axsia Howmar Ltd. (UK)
- 6) M/s Faudi Filters Systems GmbH (Germany)
- 7) M/s Filtan Filter Anlagenbau GmbH (Germany)
- 8) M/s Plenty Filters (UK)
- 9) M/s Forain S.r.l. (Italy)
- 10) M/s Ravi Techno Systems (Mumbai)
- 11) M/s Gujarat Otofilt (Ahmedabad)
- 12) M/s Nirmal Industrial Controls (India)
- 13) M/s Flash Point
- 14) M/s Fil Sep Equipments Pvt. Ltd.

# II) PRESSURE SAFETY VALVES

- 1) M/s Keystone Valves (India) Pvt. Ltd. (Baroda)
- 2) M/s Sebim Sarasin Valves India (P) Ltd
- 3) M/s Tyco Sanmar Ltd. (New Delhi)
- 4) M/s Parcol SPA, Italy
- 5) M/s Tai Milano SPA, Italy
- 6) M/s Emerson Process, Singapore
- 7) M/s Instrumentation Ltd., Palghat

# III) <u>KO DRUM</u>

- 1) MULTITEX
- 2) GRAND PRIX

# IV) FILTER SEPARATOR

Page **10** of **12** Page 218 of 644

- 1) MULTITEX
- 2) GRAND PRIX
- 3) FIL SEP
- 4) ULTRA FILTER

# V) <u>CONDENSATE TANK</u>

- 1) MULTITEX
- 2) GRAND PRIX
- 3) NIRMAL INDUSTRIAL CONTROLS
- 4) MEENAKSHI ASSOCIATES
- 5) AERO ENGINEERS

# VI) SCRUBBER

- 1) MULTITEX
- 2) GRAND PRIX

# VII) CARTRIDGE ELEMENT MAKE

- 1) PECO
- 2) PALL
- 3) FILTERITE
- 4) VELCON
- 5) FACET
- 6) BURGESS MANNING

For porous metal cartridge

- 1) FUJI
- 2) PALL
- 3) PORAL
- 4) MOTT

# VIII) QUICK OPENING CLOSURE MAKE

- 1) Peerless
- 2) Grinell
- 3) Peco,
- 4) Siirtec
- 5) Huber Yale
- 6) G.D. Engineering

### Note-1

For procuring bought out items from vendors other than those listed above / for items not having vendor list, the same may be acceptable subject to the following :-

a) The vendor/ supplier of bought out item(s) is a regular and reputed manufacturer/ supplier of said item(s) for intended services and the sizes being offered is in their regular manufacturing/ supply range.

b) The vendor/ supplier should not be in the Holiday list of CLIENT/MECON/OTHER PSU.

c) Should have supplied at least 50% of required quantity or minimum 1 number whichever is higher of maximum size and rating of item(s) as required for intended services.

Page **11** of **12** Page 219 of 644 Successful bidder shall submit documentary evidences i.e. PO copies, Inspection Certificate etc. for the above at the time of drawing approval. Bidder may fulfill the above requirement using multiple PO in last seven years reckoned from bid due date. In case of ARC contracts, 50% qty. shall be determined based on quantity derived from FOA skid quantities specified in ARC FOA. Since Qtys are very large therefore Vendor can opt for Multiple makes in order to meet delivery schedule.

For MECHANICAL Items - category (v) VALVES, for makes of following Items , i.e., (b) CHECK VALVES , (c) PLUG VALVES and (d) BALLVALVES, valve manufacturer's valid API 6D certificate shall be submitted, even if his name appears in the above vendor list. Manufacturers of BALL / PLUG / CHECK valves, whose API 6D license is not found in active status during drawing approvals will not be accepted.

# Note-2

The details of vendors indicated in this list are based on the information available with MECON; Contractor shall verify capabilities of each vendor for producing the required quantity with. PMC does not guarantee any responsibility on the performance of the vendor. It is the contractor's responsibility to verify the correct status of vendor and quality control of each party and also to expedite the material in time.

For MECHANICAL Items - category (v) VALVES, for makes of following Items , i.e., (b) CHECK VALVES , (c) PLUG VALVES and (d) BALLVALVES, valve manufacturer's valid API 6D certificate shall be submitted, even if his name appears in the above vendor list. Manufacturers of BALL / PLUG / CHECK valves, whose API 6D license is not found in active status during drawing approvals will not be accepted.

# **ANNEXURE - VI**

# QAPs

Page 221 of 644

																		F	ORM NO. 11.20(4.4)F-09 REV-0
	$\bigcirc$	CONTRACTOR				QUA	LITY A	S S	SURA	NCE	PLA	Ν	PROJECT	:					
6		ORDER NO. & DATE						F	OR				PACKAGE	NO.:					
2	मेकॉन	SUB-CONTRACTOR				STR	UCTURAI	LΑ	AND M	ECHA	ANICA	L	PACKAGE	NAME :	Ball Va	lve, Plug	Valve, Check Valve, Glo	obe Valve	
*0	Soot Campon	ORDER NO. & DATE					ΕQ	DUI	IPMEN	Т									
INCT						CODES FOR EXTENT	OF INSPECT	101	N, TESTS	, TEST (	CERTIFI	CATES	& DOCUN	IENTS :					
INST	RUCTIONS FOR	R FILLING UP :				Code Description			Code esc	rintion				Code		Descrinti	מר	Code DOCUMEN	TS.
1.	. QAP shall be s	submitted for each of the equi	ipment separately	with break	up 1.	Visual			18. An	nplitude	Test			34.		Internal I	nspection Report	D1. Approved G	A drawings
	of assembly/s	ub-assembly & part/compone	nt or for group of	equipment	2.	Dimensional			19. Sp	onge Te	est					by Contra	ctor	D2. Information	and other
	having same s	specification.				3. Fitment & Ali	gnment		20. Du	ust/ Wate	er Ingres	s Test		35.		Hardness	Test	reference d	rg/ stamped
2.	. Use numerical	I codes as indicated for extent	of inspection & t	ests and		4. Physical Test	(Sample)		21. Fri	iction Fac	ctor lest			36.		Spark Tes	st for Lining	drgs release	ed for mfg.
	for extent of i	rest certificates & documents	ed as applicable f	or the plan	1011 5. t 6		st (Sample)		22. AU 23. Pe	inesion i rforman	rest rest/r	haracte	ristic	37.		Safety De	n wice Test	D3. Relevant ca	/Item no /
	and equipmen	napeetion & tests may be add	cu us applicable l			7. Magnetic Par	.sc ticle Test (MP	T)	25.10	irve		Indiacte	i i buc	39.		Fase of M	laintenance	Identificatio	n
3.	. Separate iden	tification number with quantit	y for equipment s	hall be		8. Radiography	Test	-,	24. No	b Load/ F	- ree Runi	ning Tes	t	40.		Fire Test	(Type Test)	D5. Matchmark	s details
	indicated whe	rever equipment having same	specifications be	longing 9.		Dye Penetrat	ion Test		25. Lo	ad/ Over	rload Tes	at -		41.		Charpy V	Notch Test	D6. Line/ Layou	t diagram
	to different fa	cilities are grouped together.				<ol><li>Metallograph</li></ol>	ic Exam.		26. Me	easureme	ent of Sp	eeds		42.		Operation	al Torque Test	D7. Approved e	rection
4.	. Weight in kilo	grams must be indicated unde	er Column-5 for ea	ach item.		11. Welder's Qua	alification &		27. Ac	coustical	l Test			43.		ENP (Elec	troless Nickel Plating)	procedures	
	Estimated wei	ights may be indicated where	ver actual weights	are not		Weld Procedu	ure Test		28. Ge	eometrica	al Accura	Cy	~	44		Execution	1	D8. Unpriced su	ib P.O. with
	avaliable.					Procedure	lest and kepa	1	29. Re An	curacy	ity and P	USILIOIIIII	iy	45		Anti-Stati	r Test	ments if ar	
						13. Heat Treatme	ent		30. Pro	oving Te	st			46.		Hydrostat	ic Double Block &	D9. Calibration	Certificate of
	ABBREVIAT	IONS USED :	KEY TO SYMBO	OLS :		14. Pressure Test	t		31. Su	Irface Pre	eparatior	n				Bleed Tes	t	all measurir	ng instruments
	SV	: SUB VENDOR	* : TO BE FILLED E	BY VENDOR		<ol><li>15. Leakage Test</li></ol>	t		32. Ma	anufactu	rer's Tes	t Certific	ates	47.		Functiona	l Test	and gauges	
	MFR	: MANUFACTURER	** : TEST TO BE PER	RFORMED, IF A	PPLICABLE	16. Balancing for				bought	-out item	IS		48.		Pneumati	c Double Block &	D10. X-Ray Repo	orts
	TPI	: DESIGNATED THIRD PAR	TY INSPECTION A	GENCY		17. Vibration Tes	t		33. IB	R/ Other	r Statutor	y agenc	ies	40		Bleed Tes	t		
	D D								0	inpliance		ite		49.			ы ст		I
	Ŵ	: WITNESS												50		JINIF IL	51		
			EQUIPMENT D	ETAILS							INSF	PECTION	AND TEST	-S			Test Certificates &	Acceptance Criteria	REMARKS/
SI.	Descri	ption (with equipment	Identification	Quantity	Unit	Manufacturer's	Expected		Raw Ma	terial an	d In-Pro	cess	Fina	l Inspec	tion/ Te	st by	Documents to be	Standards/ IS/ BS/	SAMPLING PLAN
No.	heading	g, place of use and brief	No.	No./M	Weight	Name and Address	Schedule of	М	St	tage Insp	Dection	MECON	MED /CV/	TDIA	CONTR	MECON	submitted to MECON	ASME/ MECON TS /	
		specifications)	(MR/SOR Item No.)		(Kg)		гіпаі шэрп.	IVIE	IFR/5V 11			MECON	MLK/20	IPIA	CONTR	MECON		DOCUMENTS:	
1		2	3	4	5	6	7		8	9	10	11	12	13	14	15	14	15	16
1.0	Ball Valve, Plu	ig Valve, Check Valve,		Refer	*	*	*												
	Globe Valve																		
				(1110, 5010)					As per at	ttached s	sheet 2 t	o 10							
																	MEC/W/INO/0E/28/M/001		
																QAP NO.	MEC/WINO/05/28/M/001	L/QAP-002	REV O
																SHEET ?	OF 10		C C
													7						
	For CONTRAC	TOR					For SUB-COM	NTR	RACTOR										
							(Valve Manu	fact	turer)										
								-	,										

												QAP NO.	: MEC/WINO/05/28	5/M/UU1/QAP-UU2				FU	KM NO. 11.20(4.4)F-09 KEV-0
	EQUIPMENT	DETAILS					]	INSPECTIO	N AND TES	TS			Test Certificates &	Acceptance Criteria		Inspect	ion Codes		REMARKS
SI.	Description (with equipment	Identification	Quantity	Unit	Ra	w Material	and In-Pro	cess	F	inal Inspe	ction/ Test b	у	Documents to be	Standards/ IS/ BS/		& Samp	oling Plan		
NO.	specifications)	NO.	NO./M	(Kg)	MFR/SV	TPIA	CONTR	MECON	MFR/SV	TPIA	CONTR	MECON	submitted to MECON	ASME/ MECON TS / API /Normsand DOCUMENTS:	MFR/SV	TPIA	CONTR	MECON	-
1	2	3	4	5	8	9	10	11	12	13	14	15	14	15	16A	16B	16C	16D	
1.01	Body	<u>Material</u> As per MR/SOR/ Alternate Material accepted			1,2	1,2	-	-	-	-	-	-	1. D1 2. Report	<ol> <li>D1</li> <li>MECON'S TS &amp; DS</li> <li>Relevant Material Standard</li> <li>Manufacturer's Specification</li> </ol>	P	R	R	R	
		by MECON			4	4	-	-	-	-	-	-	Material Test Certificates	<ol> <li>Relevant Material Standard</li> <li>MECON's TS &amp; D.S.</li> </ol>	Р	W	R	R	
					5	5	-	-	-	-	-	-	Material Test Certificates	<ol> <li>Relevant Material Standard</li> <li>MECON's T.S.</li> <li>MECON's D.S.</li> </ol>	Ρ	W	R	R	
					6 **	6 **	-	-	-	-	-	-	Test Report	1. ASME B16.34, Appendix-IV 2. MECON's T.S. & DS	Р	W	R	R	Forgings, welds, wrought weld ends
					7 **	7 **	-	-	-	-	-	-	Test Report	1. ASME B16.34, Appendix-II 2. MECON's T.S. & DS	Ρ	W	R	R	Wet MPI for 100% of internal surfaces of all castings & forgings & bevel surfaces (MPI/ DP)
					8 **	8 **	-	-	-	-	-	-	Test Report	1. ASME B16.34, Appendix-I 2. MECON's T.S.& DS	Ρ	W	R	R	All castings as per clause 5.1.4 b) of T.S., all welds, weld ends of all cast
					9**	9**	-	-	-	-	-	-	Test Report	1. ASME B16.34, Appendix-III 2. MECON's T.S.& DS	Р	W	R	R	Bevel Surfaces (by MPI/ DP)
					13	13	-	-	-	-	-	-	Report/ Material Test Certificates	<ol> <li>Relevant Material Standard</li> <li>MECON's T.S.&amp; DS</li> </ol>	Р	R	R	R	
					35	35	-	-	-	-	-	-	Material Test Certificates	<ol> <li>Relevant Material Standard</li> <li>MECON's T.S.</li> <li>MECON's D.S.</li> </ol>	Ρ	W	R	R	
					41	41	-	-	-	-	-	-	Material Test Certificates	<ol> <li>Relevant Material Standard</li> <li>MECON's T.S.</li> <li>MECON's D.S.</li> </ol>	Р	W	R	R	

# OAP No. : MEC/WINO/05/28/M/001/0AP-002

												QAP No.	: MEC/WINO/05/28	3/M/001/QAP-002				FU	RM NO. 11.20(4.4)F-09 REV-0
	EQUIPMENT	DETAILS					I	NSPECTION	N AND TEST	S			Test Certificates &	Acceptance Criteria		Inspecti	on Codes		REMARKS
SI. No.	Description (with equipment heading, place of use and brief	Identification No.	Quantity No./M	/ Unit Weight	Rav	w Material stage in	and In-Pro	cess	F	inal Inspe	tion/ Test b	ру	Documents to be submitted to MECON	Standards/ IS/ BS/ ASME/ MECON TS /		& Samp	ling Plan		
	specifications)			(Kg)	MFR/SV	TPIA	CONTR	MECON	MFR/SV	TPIA	CONTR	MECON		API /Normsand DOCUMENTS:	MFR/SV	TPIA	CONTR	MECON	-
1	2	3	4	5	8	9	10	11	12	13	14	15	14	15	16A	16B	160	16D	
1.02	2 Closure/ Body Adapter/ Tail	Material			1.2	-	-	-	-	-	-	-	1. D1	1. D1	P	R	R	R	
1102	Piece / Bonnet	Manufacturer to indicate (to be approved											2. Report	<ol> <li>2. MECON's TS &amp; DS</li> <li>3. Relevant Material Standard</li> <li>4. Manufacturer's Specification</li> </ol>				K	
		by MECON)			4	4	-	-	-	-	-	-	Material Test Certificates	<ol> <li>Relevant Material Standard</li> <li>MECON's D.S.&amp; TS</li> </ol>	Р	W	R	R	
					5	5	-	-	-	-	-	-	Material Test Certificates	<ol> <li>Relevant Material Standard</li> <li>MECON's T.S.</li> <li>MECON's D.S.</li> </ol>	Р	W	R	R	
					6**	6**	-	-	-	-	-	-	Test Report	1. ASME B16.34, Appendix-IV 2. MECON's T.S. & DS	Р	W	R	R	Forgings, welds, wrought weld ends
					7**	7**	-	-	-	-	-	-	Test Report	1. ASME B16.34, Appendix-II 2. MECON's T.S. & DS	Р	W	R	R	Wet MPI for 100% of internal surfaces of all castings & forgings & bevel surfaces (MPI/ DP)
					8**	8**	-	-	-	-	-	-	Test Report	1. ASME B16.34, Appendix-I 2. MECON's T.S.& DS	Р	W	R	R	All castings as per clause 5.1.4 b) of T.S., all welds, weld ends of all cast
					9**	9**	-	-	-	-	-	-	Test Report	1. ASME B16.34, Appendix-III 2. MECON's T.S. & D.S.	Р	W	R	R	Bevel Surfaces (by MPI/ DP)
					13	13	-	-	-	-	-	-	Report/ Material Test Certificates	<ol> <li>Relevant Material Standard</li> <li>MECON's T.S.</li> <li>MECON's D.S.</li> </ol>	Р	R	R	R	
					35	35	-	-	-	-	-	-	Material Test Certificates	<ol> <li>Relevant Material Standard</li> <li>MECON's T.S.</li> <li>MECON's D.S.</li> </ol>	Р	W	R	R	
					41	41	-	-	-	-	-	-	Material Test Certificates	<ol> <li>Relevant Material Standard</li> <li>MECON's T.S.</li> <li>MECON's D.S.</li> </ol>	Р	W	R	R	

# OAD No + MEC/WINO/05/28/M/001/OAD-002

												QAP No.	: MEC/WINO/05/28	3/M/001/QAP-002				FC	RM NO. 11.20(4.4)F-09 REV-0
	EQUIPMENT	DETAILS					I	NSPECTION	N AND TEST	S			Test Certificates &	Acceptance Criteria		Inspecti	ion Codes		REMARKS
SI. No.	Description (with equipment heading, place of use and brief	Identification No.	Quantity No./M	/ Unit Weight	Rav	w Material stage ir	and In-Pro	cess	F	inal Inspe	ction/ Test l	ру	Documents to be submitted to MECON	Standards/ IS/ BS/ ASME/ MECON TS /		& Samp	oling Plan		
	specifications)			(Kg)	MFR/SV	TPIA	CONTR	MECON	MFR/SV	TPIA	CONTR	MECON		API /Normsand DOCUMENTS:	MFR/SV	TPIA	CONTR	MECON	
1	2	3	4	5	8	9	10	11	12	13	14	15	14	15	16A	16B	16C	16D	
1.03	3 Top Cover/Disc /	Material			1.2	-	-	-	-	-	-	-	1. D1	1. D1	P	R	R	R	
	Hinge Pin (For Check Valves)	Manufacturer to indicate (to be approved											2. Report	<ol> <li>2. MECON's TS &amp; DS</li> <li>3. Relevant Material Standard</li> <li>4. Manufacturer's Specification</li> </ol>					
		by MECON)			4	4	-	-	-	-	-	-	Material Test Certificates	<ol> <li>Relevant Material Standard</li> <li>MECON's D.S. &amp; T.S.</li> </ol>	P	W	R	R	
					5	5	-	-	-	-	-	-	Material Test Certificates	<ol> <li>Relevant Material Standard</li> <li>MECON's T.S.</li> <li>MECON's D.S.</li> </ol>	Р	W	R	R	
					6 **	6 **	-	-	-	-	-	-	Test Report	1. ASME B16.34, Annex-E 2. MECON's T.S. & D.S.	Р	W	R	R	Forgings, welds, wrought weld ends
					7 **	7 **	-	-	-	-	-	-	Test Report	1. ASME B16.34, Annex-C 2. MECON's T.S. & D.S.	Р	W	R	R	Wet MPI for 100% of internal surfaces of all castings & forgings & bevel surfaces (MPI/ DP)
					8 **	8 **	-	-	-	-	-	-	Test Report	1. ASME B16.34 Annex-B 2. MECON's T.S. 3. MECON's D.S.	Р	W	R	R	All castings as per clause 5.1.4 b) of T.S., all welds, weld ends of all cast
					9**	9**	-	-	-	-	-	-	Test Report	1. ASME B16.34, Appendix-III 2. MECON's T.S. & D.S.	Р	W	R	R	Bevel Surfaces (by MPI/ DP)
					13	13	-	-	-	-	-	-	Report/ Material Test Certificates	<ol> <li>Relevant Material Standard</li> <li>MECON's T.S.</li> <li>MECON's D.S.</li> </ol>	Р	R	R	R	
					35	35	-	-	-	-	-	-	Material Test Certificates	<ol> <li>Relevant Material Standard</li> <li>MECON's T.S.</li> <li>MECON's D.S.</li> </ol>	Р	W	R	R	
					41	41	-	-	-	-	-	-	Material Test Certificates	<ol> <li>Relevant Material Standard</li> <li>MECON's T.S.</li> <li>MECON's D.S.</li> </ol>	Р	W	R	R	

# OAD No + MEC/WINO/05/28/M/001/OAD-002

												QAP No.	: MEC/WINO/05/28	/M/001/QAP-002				FUR	RM NO. 11.20(4.4)F-09 REV-0
	EQUIPMENT	DETAILS					I	NSPECTION	AND TEST	ſS			Test Certificates &	Acceptance Criteria		Inspecti	on Codes		REMARKS
SI.	Description (with equipment	Identification	Quantity	Unit	Rav	w Material	and In-Pro	cess	F	inal Inspe	ction/ Test b	)y	Documents to be	Standards/ IS/ BS/		& Samp	ling Plan		
No	heading place of use and brief	f No	No /M	Weight		stage in	spection				,		submitted to MECON	ASME/ MECON TS /			5		
110.	specifications)			(Ka)	MER/SV/			MECON	MED/SV/	ΤΡΙΛ	CONTR	MECON	Submitted to MECON	API /Normsand	MED/SV/	ΤΡΙΛ	CONTR	MECON	-
	specifications			((((	11117,51			MECON	11117,51			MECON			1111730	11174	CONTR	MECON	
1	2	2	4			0	10	11	12	12	14	15	14		104	100	100	100	
1	2	3	4	5	8	9	10		12	13	14	15	14	15	16A	168	160	16D	
1.04	Trunnion (for Trunnion	Material			1,2	1,2	-	-	-	-	-	-	1. D1	1. D1	Р	R	R	R	
	Mounted	Manufacturer											2. Report	2. MECON's TS & DS					
	Valves)	to indicate												3. Relevant Material					
		(to be												Standard					
		approved												4. Manufacturer's					
														Specification					
					1	1							Matorial Tost	1 Bolovant Material	D	۱۸/	D	D	
		by MLCON)			- T	7	_	_	_	-	-	_	Cortificatos	1. Kelevant Material	Г	vv		ĸ	
													Certificates						
														2. MECON'S D.S. & T.S.					
					5	5	-	-	-	-	-	-	Material Test	1. Relevant Material	Р	W	R	R	
													Certificates	Standard					
														2. MECON's T.S.					
														3. MECON's D.S.					
					13	13	-	-	-	-	-	-	Report/ Material Test	1. Relevant Material	Р	R	R	R	
					15	10							Certificates	Standard	•				
													Certificates						
														2. MECON'S T.S.					
														3. MECON S D.S.					
					35	35	-	-	-	-	-	-	Material Test	1. Relevant Material	Р	W	R	R	
													Certificates	Standard					
														2. MECON's T.S.					
														3. MECON's D.S.					
					41	41	-	-	-	-	-	-	Material Test	1. Relevant Material	Р	W	R	R	
													Certificates	Standard					
														2. MECON's T.S.					
														3. MECON's D.S.					
	1				43 **	43 **	-	-	-	-	-	-	1. Test Report	1. MECON's T.S.	Р	н	R	R	
					15	10							2 Material Test	2 MECON'S D S				, in	
													Cortificatos for	3 ASTM B733 Std					
													certificates for	A Manufacturaria					
													composition,						
													naraness,	Specification					
													thickness &						
													integrity						
1.05	Ball /Disc / Plug / Obturator	<u>Material</u>			1,2	1,2	-	-	-	-	-	-	1. D1	1. D1	Р	R	R	R	
		As per MR/											2. Report	2. MECON's TS & DS					
		Alternate												3. Relevant Material					
		Material												Standard					
		accepted												4 Manufacturer's					
1														Specification					
					4	Α							Matarial Test	1 Dolovent Material		147			
					4	4	-	-	-	-	-	-		1. Kelevant Material	Р	VV	К	к	
													Certificates						
														Z. MECON'S D.S. & T.S.					
					5	5	-	-	-	-	-	-	Material Test	1. Relevant Material	Р	W	R	R	
													Certificates	Standard					
														2. MECON's T.S.					
1														3. MECON's D.S.					
1					6**	6**	-	-	-	-	-	-	Test Report	1. ASME B16.34.	Р	W	R	R	Forgings, welds.
						Ŭ								Appendix-IV	•				wrought weld ends
1																			
1																			
					/**	/**	-	-	-	-	-	-	Test Report	1. ASME B16.34,	Р	W	R	R	wet MPI for 100%
1														Appendix-II					of internal surfaces
1														2. MECON's T.S.					of all castings &
														3. MECON's D.S.					forgings & bevel
L																			

												QAP No. :	: MEC/WINO/05/28	/M/001/QAP-002				FOI	RM NO. 11.20(4.4)F-09 REV-0
	EQUIPMENT	DETAILS		-			I	NSPECTION	AND TEST	ſS			Test Certificates &	Acceptance Criteria		Inspecti	on Codes		REMARKS
SI. No.	Description (with equipment heading, place of use and brief	Identification No.	Quantity No./M	Unit Weight	Rav	v Material stage ir	and In-Pro	cess	F	inal Inspec	tion/ Test b	ру	Documents to be submitted to MECON	Standards/ IS/ BS/ ASME/ MECON TS /		& Samp	ling Plan		
	specifications)			(Kg)	MFR/SV	TPIA	CONTR	MECON	MFR/SV	TPIA	CONTR	MECON		API /Normsand DOCUMENTS:	MFR/SV	TPIA	CONTR	MECON	
1	2	3	4	5	8	9	10	11	12	13	14	15	14	15	16A	16B	16C	16D	
					8**	8**	-	-	-	-	-	-	Test Report	<ol> <li>ASME B16.34, Appendix-I</li> <li>MECON's T.S.</li> <li>MECON's D.S.</li> </ol>	Ρ	W	R	R	All castings as per clause 5.1.4 b) of T.S., all welds, weld ends of all cast valves
					9**	9**	-	-	-	-	-	-	Test Report	<ol> <li>ASME B16.34, Appendix-III</li> <li>MECON's T.S.</li> <li>MECON's D.S.</li> </ol>	Ρ	W	R	R	Bevel Surfaces (by MPI/ DP)
					13	13	-	-	-	-	-	-	Report/ Material Test Certificates	<ol> <li>Relevant Material Standard</li> <li>MECON's T.S.</li> <li>MECON's D.S.</li> </ol>	Р	R	R	R	
					35	35	-	-	-	-	-	-	Material Test Certificates	<ol> <li>Relevant Material Standard</li> <li>MECON's T.S.</li> <li>MECON's D.S.</li> </ol>	Ρ	W	R	R	
					41	41	-	-	-	-	-	-	Material Test Certificates	<ol> <li>Relevant Material Standard</li> <li>MECON's T.S.</li> <li>MECON's D.S.</li> </ol>	Ρ	W	R	R	
					43	43	-	-	-	-	-	-	<ol> <li>Test Report</li> <li>Material Test Certificates for composition, hardness, thickness &amp; integrity</li> </ol>	<ol> <li>MECON'S T.S.</li> <li>MECON'S D.S.</li> <li>ASTM B733 Std.</li> <li>Manufacturer's Specification</li> </ol>	Ρ	W	R	R	
1.06	Stem	<u>Material</u> As per MR/ Alternate Material accepted			1,2	1,2	-	-	-	-	-	-	1. D1 2. Report	<ol> <li>D1</li> <li>MECON's TS &amp; DS</li> <li>Relevant Material Standard</li> <li>Manufacturer's Specification</li> </ol>	Ρ	R	R	R	
		by MECON			4	4	-	-	-	-	-	-	Material Test Certificates	<ol> <li>Relevant Material Standard</li> <li>MECON's D.S. &amp; T.S.</li> </ol>	Р	W	R	R	
					5	5	-	-	-	-	-	-	Material Test Certificates	<ol> <li>Relevant Material Standard</li> <li>MECON's T.S.</li> <li>MECON's D.S.</li> </ol>	Ρ	W	R	R	
					6**	6**	-	-	-	-	-	-	Test Report	<ol> <li>ASME B16.34, Appendix-IV</li> <li>MECON's T.S.</li> <li>MECON's D.S.</li> </ol>	Р	W	R	R	Forgings, welds, wrought weld ends

												QAP No.	: MEC/WINO/05/28	/M/001/QAP-002				FO	RM NO. 11.20(4.4)F-09 REV-0
	EQUIPMENT	DETAILS					11	NSPECTION	AND TES	ГS			Test Certificates &	Acceptance Criteria		Inspecti	on Codes		REMARKS
SI.	Description (with equipment	Identification	Quantity	/ Unit	Rav	N Material	and In-Proc	cess	F	inal Inspec	tion/ Test t	у	Documents to be	Standards/ IS/ BS/		& Samp	ling Plan		
No.	heading, place of use and brief	i No.	No./M	Weight		stage ir	ispection						submitted to MECON	ASME/ MECON TS /					
	specifications)			(Kg)	MFR/SV	TPIA	CONTR	MECON	MFR/SV	TPIA	CONTR	MECON		API /Normsand DOCUMENTS:	MFR/SV	TPIA	CONTR	MECON	
1	2	3	4	5	8	9	10	11	12	13	14	15	14	15	16A	16B	16C	16D	
					7**	7**	-	-	-	-	-	-	Test Report	1. ASME B16.34, Appendix-II 2. MECON's T.S. 3. MECON's D.S.	Р	W	R	R	Wet MPI for 100% of internal surfaces of all castings & forgings & bevel
					8**	8**	-	-	-	-	-	-	Test Report	<ol> <li>ASME B16.34, Appendix-I</li> <li>MECON's T.S.</li> <li>MECON's D.S.</li> </ol>	Р	W	R	R	All castings as per clause 5.1.4 b) of T.S., all welds, weld ends of all cast
					9**	9**	-	-	-	-	-	-	Test Report	<ol> <li>ASME B16.34, Appendix-III</li> <li>MECON'S T.S.</li> <li>MECON'S D.S.</li> </ol>	Р	W	R	R	Bevel Surfaces (by MPI/ DP)
					13	13	-	-	-	-	-	-	Report/ Material Test Certificates	<ol> <li>Relevant Material Standard</li> <li>MECON's T.S.&amp; DS</li> </ol>	Р	R	R	R	
					35	35	-	-	-	-	-	-	Material Test Certificates	<ol> <li>Relevant Material Standard</li> <li>MECON's T.S.</li> <li>MECON's D.S.</li> </ol>	Р	W	R	R	
					41	41	-	-	-	-	-	-	Material Test Certificates	<ol> <li>Relevant Material Standard</li> <li>MECON's T.S.</li> <li>MECON's D.S.</li> </ol>	Р	W	R	R	
					43	43	-	-	-	-	-	-	<ol> <li>Test Report</li> <li>Material Test Certificates for composition, hardness, thickness &amp; integrity</li> </ol>	<ol> <li>MECON'S T.S.</li> <li>MECON'S D.S.</li> <li>ASTM B733 Std.</li> <li>Manufacturer's Specification</li> </ol>	Ρ	W	R	R	
1.07	Metal Seats / Retainer Ring (If provided)	<u>Material</u> As per MR/ Alternate Material accepted			1,2	1,2	-	-	-	-	-	-	1. D1 2. Report	<ol> <li>D1</li> <li>MECON's TS &amp; DS</li> <li>Relevant Material Standard</li> <li>Manufacturer's Specification</li> </ol>	Р	R	R	R	
		by MECON			4	4	-	-	-	-	-	-	Material Test Certificates	<ol> <li>Relevant Material Standard</li> <li>MECON's D.S. &amp; T.S.</li> </ol>	Р	W	R	R	
					5	5	-	-	-	-	-	-	Material Test Certificates	<ol> <li>Relevant Material Standard</li> <li>MECON's T.S.</li> <li>MECON's D.S.</li> </ol>	Ρ	W	R	R	

|--|

												QAP No.	: MEC/WINO/05/28	3/M/001/QAP-002				FC	0RM NO. 11.20(4.4)F-09 REV-0
	EQUIPMENT	DETAILS					Ι	NSPECTION	N AND TEST	ſS			Test Certificates &	Acceptance Criteria		Inspecti	ion Codes		REMARKS
SI. No.	Description (with equipment heading, place of use and brief	Identification No.	Quantity No./M	/ Unit Weight	Rav	w Material stage ir	and In-Pro	cess	F	inal Inspe	ction/ Test b	ру	Documents to be submitted to MECON	Standards/ IS/ BS/ ASME/ MECON TS /		& Samp	oling Plan		
	specifications)			(Kg)	MFR/SV	TPIA	CONTR	MECON	MFR/SV	TPIA	CONTR	MECON		API /Normsand DOCUMENTS:	MFR/SV	TPIA	CONTR	MECON	
1	2	3	4	5	8	9	10	11	12	13	14	15	14	15	16A	16B	16C	16D	
					6**	6**	-	-	-	-	-	-	Test Report	<ol> <li>ASME B16.34, Appendix-IV</li> <li>MECON's T.S.</li> <li>MECON's D.S.</li> </ol>	Р	W	R	R	Forgings, welds, wrought weld ends
					7**	7**	-	-	-	-	-	-	Test Report	1. ASME B16.34, Appendix-II 2. MECON's T.S. 3. MECON's D.S.	Р	W	R	R	Wet MPI for 100% of internal surfaces of all castings & forgings & bevel surfaces (MPI/ DP)
					8**	8**	-	-	-	-	-	-	Test Report	1. ASME B16.34, Appendix-I 2. MECON's T.S. 3. MECON's D.S.	Р	W	R	R	All castings as per clause 5.1.4 b) of T.S., all welds, weld ends of all cast
					9**	9**	-	-	-	-	-	-	Test Report	<ol> <li>ASME B16.34, Appendix-III</li> <li>MECON's T.S.</li> <li>MECON's D.S.</li> </ol>	Р	W	R	R	Bevel Surfaces (by MPI/ DP)
					13	13	-	-	-	-	-	-	Report/ Material Test Certificates	<ol> <li>Relevant Material Standard</li> <li>MECON's T.S. &amp; D.S.</li> </ol>	Р	R	R	R	
					35	35	-	-	-	-	-	-	Material Test Certificates	<ol> <li>Relevant Material Standard</li> <li>MECON's T.S.</li> <li>MECON's D.S.</li> </ol>	Р	W	R	R	
					41	41	-	-	-	-	-	-	Material Test Certificates	<ol> <li>Relevant Material Standard</li> <li>MECON's T.S.</li> <li>MECON's D.S.</li> </ol>	Р	W	R	R	
					43	43	-	-	-	-	-	-	<ol> <li>Test Report</li> <li>Material Test Certificates for composition, hardness, thickness &amp; integrity</li> </ol>	<ol> <li>MECON's T.S.</li> <li>MECON's D.S.</li> <li>ASTM B733 Std.</li> <li>Manufacturer's Specification</li> </ol>	P	W	R	R	
1.08	Bolting Material (Studs & Nuts)	<u>Material</u> As per MR/ Alternate Material accepted			1,2	1,2	-	-	-	-	-	-	1. D1 2. Report	<ol> <li>D1</li> <li>MECON's T.S. &amp; D.S.</li> <li>Relevant Material Standard</li> <li>Manufacturer's Specification</li> </ol>	Р	R	R	R	Alongwith thickness measurement for ENP Coating.
		by MECON			4	4	-	-	-	-	-	-	Material Test Certificates	<ol> <li>Relevant Material Standard</li> <li>MECON's D.S. &amp; T.S.</li> </ol>	Р	W	R	R	

|--|

												QAP NO.	: MEC/WINO/05/28	/M/001/QAP-002					ORM NO. 11.20(4.4)F-09 REV-0
	EQUIPMENT	DETAILS					I	NSPECTION	I AND TEST	S			Test Certificates &	Acceptance Criteria		Inspecti	on Codes		REMARKS
SI.	Description (with equipment	Identification	Quantity	/ Unit	Rav	w Material	and In-Proc	cess	F	inal Inspec	tion/ Test b	)V	Documents to be	Standards/ IS/ BS/		& Samp	oling Plan		
No.	heading, place of use and brief	No.	No./M	Weight		stage in	nspection			•		,	submitted to MECON	ASME/ MECON TS /			5		
	specifications)			(Ka)	MFR/SV	TPIA		MECON	MFR/SV	ΤΡΙΑ	CONTR	MECON		API /Normsand	MFR/SV	ΤΡΙΑ	CONTR	MECO	
	op contraction to y			((19)								1.20011		DOCUMENTS:					
1	2	3	4	5	8	9	10	11	12	13	14	15	14	15	164	16B	160	16D	
<u> </u>	LL				5	5		-	-	-	-	-	Material Test	1 Relevant Material	P	H	R	R	
					5	5							Cortificatos	Standard					
													Certificates						
														2. MECON'S T.S.					
														3. MECON's D.S.					
					6**	6**	-	-	-	-	-	-	Test Report	1. ASME B16.34,	Р	W	R	R	Forgings, welds,
														Appendix-IV					wrought weld ends
														2. MECON's T.S.					_
														3. MECON's D.S.					
					7**	7**							Test Report	1 ASME B16 34	D	\W/	P	D	Wet MPI for 100%
					,	,							rest Report	Appondix II	· ·	••			of internal surfaces
														2. MECON'S T.S.					of all castings &
														3. MECON'S D.S.					forgings & bevel
																			surfaces (MPI/ DP)
			1		8**	8**	-	-	-	-	-	-	Test Report	1. ASME B16.34,	Р	W	R	R	All castings as per
														Appendix-I					clause 5.1.4 b) of
														2. MECON's T.S.					T.S., all welds, weld
														3. MECON'S D.S.					ends of all cast
																			valves
					Q**	0**							Tect Deport	1 ASME B16 34	D	۱۸/	D	D	Boyol Surfaces
					9	9	-	-	-	-	-	-	Test Report	Appendix III	F	vv	r r	r.	bever surfaces
														Appendix-III					(DY MP1/ DP)
														2. MECON'S T.S.					
					13	13	-	-	-	-	-	-	Report/ Material Test	1. Relevant Material	Р	R	R	R	
													Certificates	Standard					
														2. MECON's T.S. & D.S.					
					35	35	-	-	_	-	-	_	Material Test	1 Relevant Material	Р	W	R	R	
					55	55	_	_	_		_	_	Cortificator	Standard	Г	vv	ĸ	ĸ	
													Certificates						
														2. MECON'S T.S.					
			1											3. MECON'S D.S.					
1				1	41	41	-	-	-	-	-	-	Material Test	1. Relevant Material	Р	W	R	R	
1				1									Certificates	Standard					
			1											2. MECON's T.S.					
1				1										3. MECON's D.S.					
1.09	Assembled Valves			1	-	_	-	-	1.2	1.2	-	_	Report	1. D1	Р	W	W	W	
									-/-	-,-				2. MECON's T.S.	· ·				
1					-	-	-	-	3	3	-	-	Report		Р	W	W	W	
			1		-	-	-	-	14	14	-	-	1. Report	1. D1	Р	W	W	W	
													2. Test Certificates	2. MECON's T.S.					
1				1										3. MECON'S D.S.					
			1											H. Applicable Standard					
1																			
			1		-	-	-	-	15	15	-	-	1. Report	1. D1	Р	W	W	W	
			1										2. Test Certificates	2. MECON's T.S.					
1				1										3. MECON's D.S.					
			1											4. Applicable Standard					
			1																
			1						40	40			1 Doport	1 D1		P	п		
1				1			-	-	40	40	-	-	1. Repuil	2. MECON's T.S.		к	ĸ	ĸ	
			1										2. Test Certificates	3. MECON's D.S.					
1				1										4. Applicable Standard					
			1																

												QAP No.	MEC/WINO/05/28	/M/001/QAP-002				FO	RM NO. 11.20(4.4)F-09 REV-0
	EQUIPMENT	DETAILS					I	NSPECTION	AND TEST	ſS			Test Certificates &	Acceptance Criteria		Inspectio	on Codes		REMARKS
SI. No.	Description (with equipment heading, place of use and brief	Identification No.	Quantity No./M	Unit Weight	Rav	w Material stage in	and In-Prod rspection	cess	F	inal Inspec	ction/ Test b	ру	Documents to be submitted to MECON	Standards/ IS/ BS/ ASME/ MECON TS /		& Sampl	ling Plan		
	specifications)			(Kg)	MFR/SV	TPIA	CONTR	MECON	MFR/SV	TPIA	CONTR	MECON		API /Normsand DOCUMENTS:	MFR/SV	TPIA	CONTR	MECON	
1	2	3	4	5	8	9	10	11	12	13	14	15	14	15	16A	16B	16C	16D	
							-	-	42 **	42 **	-	-	<ol> <li>Report</li> <li>Test Certificates</li> </ol>	<ol> <li>D1</li> <li>MECON's T.S.</li> <li>MECON's D.S.</li> <li>Applicable Standard</li> </ol>	P	W	W	W	If Applicable
					-	-	-	-	37	37	-	-	Certificates		-	R	R	R	
					-	-	-	-	44	44	-	-	<ol> <li>Report</li> <li>Test Certificates</li> </ol>	<ol> <li>D1</li> <li>MECON's T.S.</li> <li>MECON's D.S.</li> <li>Applicable Standard</li> </ol>	Р	W	R	R	
					-	-	-	-	45 **	45 **	-	-	<ol> <li>Report</li> <li>Test Certificates</li> </ol>	<ol> <li>D1</li> <li>MECON's T.S.</li> <li>MECON's D.S.</li> <li>Applicable Standard</li> </ol>	Р	W	W	W	If Applicable
					-	-	-	-	46 **	46 **	-	-	<ol> <li>Report</li> <li>Test Certificates</li> </ol>	<ol> <li>D1</li> <li>MECON's T.S.</li> <li>MECON's D.S.</li> <li>Applicable Standard</li> </ol>	Р	W	W	W	If Applicable
					-	-	-	-	47	47	-	-	<ol> <li>Report</li> <li>Test Certificates</li> </ol>	<ol> <li>D1</li> <li>MECON's T.S.</li> <li>MECON's D.S.</li> <li>Applicable Standard</li> </ol>	P	W	W	W	Refer Note 3 of Table 2 of TS no. MEC/ TS / 05 / E5 / 002A
					-	-	-	-	48 **	48 **	-	-	<ol> <li>Report</li> <li>Test Certificates</li> </ol>	<ol> <li>D1</li> <li>MECON's T.S.</li> <li>MECON's D.S.</li> <li>Applicable Standard</li> </ol>	Р	W	W	W	If Applicable
					-	-	-	-	49 **	49 **	-	-	<ol> <li>Report</li> <li>Test Certificates</li> </ol>	<ol> <li>D1</li> <li>MECON'S T.S.</li> <li>MECON'S D.S.</li> <li>Applicable Standard</li> </ol>	Р	W	W	W	If Applicable
					-	-	-	-	50	50	-	-	<ol> <li>Final Report</li> <li>Final Certificates</li> </ol>	<ol> <li>D1</li> <li>MECON's T.S.</li> <li>MECON's D.S.</li> <li>Applicable Standard</li> </ol>	P	W	W	W	One valve per size per rating
1.10	) Complete documentation check and compilation				-	-	-	-	-	-	-	-	<ol> <li>Final Report</li> <li>Final Certificates</li> </ol>	<ol> <li>D1</li> <li>MECON'S T.S.</li> <li>MECON'S D.S.</li> <li>Applicable Standard</li> </ol>	Р	P / H / R	Н	Н	
1.11	Issue I.C. / I.R.N and give Dispatch clearance				-	-	-	-	-	-	-	-	<ol> <li>Final Report</li> <li>Final Certificates</li> </ol>	<ol> <li>D1</li> <li>MECON's T.S.</li> <li>MECON's D.S.</li> <li>Applicable Standard</li> </ol>	R	Р	Н	н	
1.12	2 Actuator Tests					As per A	Ctuator C	Quality As	surance	Plan (to	be submi	tted by v	endor for approva	)					
1	(If Applicable)																		
	<ol> <li>MAKE : As per App</li> <li>VENDOR shall establish app</li> <li>Vendor shall do RT for Body</li> <li>Vendor shall do UT/RT for E</li> </ol>	proved Ma roved WPS-PQ adapter to PU odt to Body ad	I R-WPQ fo P piece we lapter wel	I : <b>in Te</b> r the wel elding an ding witr	I nder Do dings duly w nd RT report nessed by T	cument witnessed I t shall be ro PIA (Applic	t or App by TPIA (Ap eviewed by cable for ful	roved P pplicable for MECON & ly welded b	TR as portain the second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second secon	er Tend ed body va icable for v only)	ler condi alves OR val valves with v	itions . ves with w welded pup	elded pup pieces and si pieces and size 2" and	ze 2" and above only). above only)	<u>I</u>	<u> </u>	<u> </u>	<u> </u>	

For CONTRACTOR

For SUB-CONTRACTOR (Valve Manufacturer)

QAP NO.: MEC/WINO/0

05/28/M/001/QAP-002	REV
	0



# QUALITY ASSURANCE PLAN FOR FLANGES

MECON Limited (A Govt. of India Enterprise)	QUALITY ASSURANCE PLAN FOR FLANGES	PROJECT: ITEM: FLANGES QAP NO.: MEC/WINO/05/21/M/001/S026/QAP/A
		Page 2 of 4

	APPLICABL	E CODES AND SPEC	FICATIONS: N	ICATIONS: MEC/TS/05/21/025, R-0 and MEC/TS/05/21/026, R-0 WITH AMENDMENT						IS SCOPE OF INSPE		
S. NO.	STAGE	CHARACTERISTICS	CATEGORY	METHOD OF CHECK	QUANTUM OF CHECK	REFERENCE DOCUMENTS	ACCEPTANCE NORMS	RECORD	Vendor/ Sub Vendor	TPI	MECON	
1.	Review of PO / Drg.	Review of PO Doc. / Drg.	Critical	Scrutiny / Verification	Each doc. of P.O.	Appl. Spec. / STD	-	-	Р	R	R	
2.	Raw Material	1. Manufacturing process of steel     2. Visual     3. Dimensional     4. Chemical	Critical	Verification with M.T.C.	Each Heat	Appl. Material Specification / STD	As per tender document / Material Specification / STD	Material Test Certificate and MI Register	Р	R	R	
		Composition		Analysis	Lacinneat							
3.	Forgings	1. Reduction Ratio     2. Temperature	Critical	Measurement	Minimum 1 per size	Standard Manufacturing Procedure	Std. Procedure	Forging process record / internal Register	Р	R	R	
		3. Forging Dimensions		Measurement			AMSE B16.5					
4.	Heat Treatment (as applicable)	Heat Treatment Cycle	Major	Verification of Heat Treatment Cycle	Each Heat / HT Lot	As per Tender Doc. / Material Specification / STD	As per Tender Doc. / Material Specification / MECON Std	T.P.M. Sheet, Heat Treatment Graph	Р	R	R	
5.	Mechanical Testing (as applicable)	1. Tensile Test (TS, YS, RA%, EL%)	Major	Tensile Testing	One / HT / Lot / Group	As per Tender Doc. / Material Specification / STD	Std. Procedure	Mechanical Test Report & T.C.	Р	W	R	
		2. Hardness		IMPACT Testing			As per Tender Doc. / Material Specification / MECON Std					
		3. Charpy V-Notch Test										



	APPLICABL	E CODES AND SPEC	FICATIONS: N	ICATIONS: MEC/TS/05/21/025, R-0 and MEC/TS/05/21/026, R-0 WITH AMENDMENTS						SCOPE OF INSPECTION		
S. NO.	STAGE	CHARACTERISTICS	CATEGORY	METHOD OF CHECK	QUANTUM OF CHECK	REFERENCE DOCUMENTS	ACCEPTANCE NORMS	RECORD	Vendor/ Sub Vendor	TPI	MECON	
6	NDT	1. DP	Major	DP Testing	100%	As per Tender Doc. / Material Specification / STD	As per Tender Doc. / Material Specification / MECON Std	Mechanical Test Report & T.C.	Р	W	R	
		2. UT		Ultrasonic Flaw Detector								
		3. MPI		MPI Testing								
7.	Final Inspection	Visual & Dimensions	Major	Visual / Measurement	100%	As per Tender Doc. / Material Specification / STD	As per Tender Doc. / Material Specification / MECON Std	Dimension Report Format	Р	W	W/R	
8.	Making, Colour Coding, Rust Prevention & Packing	Making, Colour Coding, Rust prevention & Packing	Major	Visual	100%	As per Tender Doc. / Material Specification / STD	As per Tender Doc. / Material Specification / MECON Std	Packing List	Р	W	W/R	
10.	Certification & Release Notes	Inspection Release Note as per EN 10204 Type 3.2	Major	Verification of PO Spec. & QAP	100%	As per Tender Doc. / Material Specification / STD	As per Tender Doc. / Material Specification / MECON Std	Release Note	R	IR	IR	
11.	Shipping	Verification of surface coating / type of packing	Major	-	-	As per Tender Doc. / Material Specification / STD	As per Tender Doc. / Material Specification / MECON Std	Shipping documents	Р	R	R	

MECON Limited	QUALITY ASSURANCE PLAN FOR FLANGES	PROJECT: ITEM: FLANGES QAP NO.: MEC/WINO/05/21/M/001/S026/QAP/A
(A Govt. of India Enterprise)		Page 4 of 4

APPLICABLE CODES AND SPECIFICATIONS: MEC/TS/05/21/025, R-0 and MEC/TS/05/21/026, R-0 WITH AMENDMENTS									SCOPE OF INSPECTION		
S.	STAGE	CHARACTERISTICS	CATEGORY	METHOD OF	QUANTUM	REFERENCE	ACCEPTANCE	RECORD	Vendor/	TPI	MECON
NO.				CHECK	OF	DOCUMENTS	NORMS		Sub		
					CHECK				Vendor		

Note:

- 1. MAKE: As per Approved Make List in Tender Document or Approved PTR as per Tender conditions.
- 2. All items shall be provided with EN 10204 Type 3.2 certificate.
- 3. All inspection related documents shall be reviewed (Sign & Stamp) by the contractor.

Legends: H – Hold (Offer for Witness & obtain clearance), W – Witness, R – Review, A – Approval, I – Information, IR – Issue Release Note, C – Certify, X – Submit, PO – Purchase Order, PR – Purchase Requisition, SR – Stress Relieving, MPI – Magnetic Particle Inspection, DI-Dye Penetrant Test, UT – Ultrasonic examination, TS – Technical Specification, WPS – Welding Procedure Specification, PQR – Procedure Qualification Record, WQT – Welder Qualification Test.

All the NDT / Leak Testing / Heat Treatment / Special manufacturing procedures have to be specially approved or only previously approved procedures have to be used. In case of conflict between purchase specification, contract documents and ITP more stringent conditions shall be applicable. The document describes generally the requirements pertaining to all types of Flanges. Requirements specific to the item are only applicable.



# QUALITY ASSURANCE PLAN FOR FITTINGS



# **QUALITY ASSURANCE PLAN** FOR

FITTINGS

QAP NO.: MEC/WINO/05/21/M/001/S026/QAP/B

PROJECT:

**ITEM: PIPE FITTINGS** 

Page 2 of 4

AP		ODES AND SPEC	CIFICATIONS: ME	C/TS/05/21/0	25, R-0 and M	EC/TS/05/21/026	6, R-0 WITH AMEN	DMENTS	SCOPE OF INSPECTION			
S. NO.	STAGE	COMPONENT	CHARACTERI- STICS	METHOD OF CHECK	QUANTUM OF CHECK	REFERENCE DOCUMENTS	ACCEPTANCE NORMS	RECORD	Vendor/ Sub Vendor	TPI	MECON	
1.	Material	FITTINGS	Fully killed steel	MTC	100%	PO & Std. Spec.	PO QAP & Std. Spec.	MTC	Р	R	R	
2.	Inspection	FITTINGS										
i)	Visual Inspection	FITTINGS	Visual Imp.	Visual Internal & External Surface	100%	PO & Std. Spec.	PO QAP & Std. Spec.	Inspection Report	Р	W	R	
ii)	NDT	FITTINGS	Soundness of Tee & Butt Welds	UT, RT & MPI	100%	PO & Std. Spec.	PO QAP & Std. Spec.	Inspection Report	Р	W & Evaluation of RT Films	R	
iii)	NDT	FITTINGS	Forgings	WET MPI	100%	PO & Std. Spec.	To comply with MSS-SP-53	Inspection Report	Р	W	R	
iv)	NDT	FITTINGS	End Laminations	UT for Distance of 25 mm on ends.	100%	PO & Std. Spec.	Any lamination than 6.35 mm not accepted	Inspection Report	Р	W	R	
V)	Testing Destructive	FITTINGS	Properties of Mech. / Chemical & Impact Test	Chemical by Spectro and other test as per ASTM A - 370	As per Heat / Lot	PO & Spec.	MSS-SP-75	IMP Lab Report	Ρ	W	R	



# QUALITY ASSURANCE PLAN FOR

# FITTINGS

QAP NO.: MEC/WINO/05/21/M/001/S026/QAP/B

PROJECT:

**ITEM: PIPE FITTINGS** 

Page 3 of 4

AP	PLICABLE CO	ODES AND SPE	CIFICATIONS: ME	C/TS/05/21/0	25, R-0 and M	IEC/TS/05/21/026	6, R-0 WITH AMEN	NDMENTS	SCOP	E OF INSPE	CTION
S. NO.	STAGE	COMPONENT	CHARACTERI- STICS	METHOD OF CHECK	QUANTUM OF CHECK	REFERENCE DOCUMENTS	ACCEPTANCE NORMS	RECORD	Vendor/ Sub Vendor	TPI	MECON
3.	Final Inspection	FITTINGS	Visual & Dimensional	-	As per lot	PO & Spec.	PO & Spec.	Inspection Report	Ρ	W	W/R
4.	Marking	FITTINGS	Identification manufacturer's Name, nominal diameter end thickness malts & Tag No.	By painting	100%	PO & Spec.	-	-	Ρ	W	W/R
5.	Certification & Release Notes	FITTINGS	Inspection Release Note as per EN 10204 Type 3.2	Verification of PO Spec. & QAP	100%	As per Tender Doc. / Material Specification / STD	As per Tender Doc. / Material Specification / MECON Std	Release Note	R	IR	IR
6.	Shipping	FITTINGS	Verification of surface coating / type of packing	-	-	As per Tender Doc. / Material Specification / STD	As per Tender Doc. / Material Specification / MECON Std	Shipping documents	Ρ	R	R



# **QUALITY ASSURANCE PLAN PROJECT:** FOR **ITEM: PIPE FITTINGS** FITTINGS QAP NO.: MEC/WINO/05/21/M/001/S026/QAP/B

Page 4 of 4

APPLICABLE CODES AND SPECIFICATIONS: MEC/TS/05/21/025, R-0 and MEC/TS/05/21/026, R-0 WITH AMENDMENTS									SCOPE OF INSPECTION		
S. NO.	S. NO.       STAGE       COMPONENT       CHARACTERI- STICS       METHOD OF CHECK       QUANTUM 										MECON
Note											

- 1. MAKE: As per Approved Make List in Tender Document or Approved PTR as per Tender conditions.
- All items shall be provided with EN 10204 Type 3.2 certificate. 2.
- 3. All inspection related documents shall be reviewed (Sign & Stamp) by the Contractor.

Legends: H – Hold (Offer for Witness & obtain clearance), W – Witness, R – Review, A – Approval, I – Information, IR – Issue Release Note, C – Certify, X – Submit, PO - Purchase Order, PR - Purchase Requisition, N-Normalizing, N&T - Normalizing & Tempering, SA - Solution annealing, N & SR - Normalizing & Stress relieving.

All the NDT / Leak Testing / Heat Treatment / Special manufacturing procedures have to be specially approved or only previously approved procedures have to be used. In case of conflict between purchase specification, contract documents and ITP more stringent conditions shall be applicable. The document describes generally the requirements pertaining to all types of Fittings. Requirements specific to the item are only applicable.



# INSPECTION AND TEST PLAN FOR

ASSORTED PIPES

STANDARD SPECIFICATION NO. MEC/TS/05/62/59A Page 1 of 3

# **INSPECTION AND TEST PLAN**

FOR

ASSORTED PIPES (FOR METERING SKID)

1	31.05.2024	ISSUED FOR IMPLEMENTATION	Sachin Kumar	Sachin Singhal	Anil Kumar
Rev. No.	Date	Purpose	Prepared	Checked by	Approved by

ITP No. : 05/21/012E/001

# Bo age 1:2000 Control

# INSPECTION AND TEST PLAN FOR

# **ASSORTED PIPES**

# STANDARD SPECIFICATION NO. MEC/TS/05/62/59A

	STACE	COMPONENT	T CHARACTERISTICS	METHOD OF	OUANTUM	M REFERENCE	CE ACCEPTANCE		SCOPE OF INSPECTION		
SL.NO.	STAGE		CHARACTERISTICS	METHOD OF CHECK	QUANTUM OF CHECK	REFERENCE DOCUMENTS	ACCEPTANCE NORMS	RECORD	Manufacturer	TPIA	Client/Mecon
1.0	Raw material	Plate/Sheet/	(i)Chemical Composition: CE	Testing	One sample per heat	Relevant Code & Technical Specification	As per code & Technical Specification	Mill test Certificate/ report	Н	R	R
1.0 inspe	ispection	Strip	(ii)Mechanical properties YS,UTS,EL	Testing	One sample per heat	Relevant Code & Technical Specification	As per code & Technical Specification	Inspection report	Н	R	R
2.0	Manufacturing, welding (HFW & SAW)	Pipe	Welding Speed, Current, Voltage, Frequency, Offset, Height of Flash & Trim	Record of Parameter	100% by Vendor	Relevant Code & Technical Specification	Approved WPS	Internal report	Р	R	R
3.0	Product Testing										
3.1	Destructive Testing	Pipe	Mechanical Properties	Tensile, Flattening, Reverse Bend, Hardness, Impact & Macro	One sample per heat	Relevant Code & Technical	As per code & Technical	Lab Report	w	W	W
			Chemical Properties	Chemical Composition	_	Specification	specification				
3.2	Non Destructive Testing	Pipe	Surface & Internal Inspection	UT/RT or as per code & specification	100% by Vendor	Relevant Code & Technical Specification	As per code & Technical Specification	Inspection Report/ Internal Record	W	R	R

ITP No. : 05/21/012E/001

# INSPECTION AND TEST PLAN FOR



# **ASSORTED PIPES**

STANDARD SPECIFICATION NO. MEC/TS/05/62/59A Page 3 of 3

		COMPONENT		METHOD OF	QUANTUM	REFERENCE	ACCEPTANCE	DECODD	SCOPE	OF INSP	ECTION
SL.NO.	SIAGE	COMPONENT	CHARACTERISTICS	СНЕСК	<b>OF</b> CHECK	DOCUMENTS	NORMS	RECORD	Manufacturer	TPIA	Client/Mecon
3.3	Heat Treatment (As Applicable)	Pipe	Stress Relieving, Normalizing, Tempering, Solution Annealing etc.	Temperature Recording	Each Pipe	Relevant Code & Technical Specification	Relevant Code & Technical Specification	HT Chart	Р	R	R
4.0	Hydrostatic test	Pipe	No leak	Visual	Each Pipe	Relevant Code & Technical Specification	As per code & Technical Specification	Inspection Report, Hydraulic Graph	W	W	RW
5.0	Final inspection	Pipe	Straightness, Outer Diameter (Pipe Body, Pipe Ends), Out of Roundness, Thickness, Length, Surface Lamination, Root Face, Bevel Angle	Visual, dimensional	100% by vendor	Relevant Code & Technical Specification	As per code & Technical Specification	Inspection Report	W	W	RW
6.0	Galvanizing (If Applicable)	Pipe	Integrity of Galvanized Coating		Each Pipe	Relevant Code & Technical Specification	As per code & Technical Specification	Inspection Report	Р	R	R
	Legends/Notes:		·	<u></u>	<u>.</u>	•					
	W-witness, R-Review, RW- Minimum 10% witness and 100% Review,										
NOTE: Hydrostatic test for the pipes can be allowed to be conducted along with the Skid assembly instead of hydrostatic testing at the pie manufacturer/supplier works. In that case, hydrostatic test of pipes shall be witnessed at the time of hydrotesting of metering skid assembly only.											

																		FORM NO.	11.20(4.4)F-09 REV-0		
		CONTRACTOR					QUAL	ITY A	S S U	IRANC	E PL	A N			PROJECT :						
		ORDER NO. & D	ATE						FΟ	R					BID DOCUMENT N	0. :					
2	मेकॉन 🚶	SUB-CONTRACT	OR				STRU	CTURAL	AN	D MECH	IANIC	CAL			ITEM NAME : CAR	TRIDGE FILTER					
~~ <u>`</u>	Pot Cunpan	ORDER NO. & D	ATE					EO	UIPI	MENT				SPEC. NO.: MEC/TS/05/62/017, REV-01 & MEC/TS/05/62/019, REV-0							
								L	-												
INSTR	UCTIONS FOR	R FILLING UP :				CODES FOR EXTENT	OF INSPE	TION, TES	TS, TE	EST CERTI	ICATES	5 & DOCU	MENTS :								
								,	,												
1. (	QAP shall be s	ubmitted for each o	of the equipme	nt separately with break	up	Code Description		Code		De	escription	1			Code	Des	cription	Code DOCUMENTS:			
(	of assembly/su	ub-assembly & part	t/component o	r for group of equipment		1. Visual 2. Dimonsional		18.		Amplitude I	est +				34.	Inte	rnal Inspection Report	D1. Approved GA dra	wings		
21	lise numerical	codes as indicated	for extent of i	nspection & tests and		3 Fitment & Al	ianment	20		Dust/ Wate	er Inarese	s Test			35	Har	iness Test	reference dra/ st	amped		
	submission of t	test certificates & c	locuments. Ad	ditional codes & descriptic	n	<ol> <li>Physical Tes</li> </ol>	t (Sample)	21.		Friction Fac	tor Test	0 1 000			36.	Spa	k Test for Lining	dras released for	mfa.		
f	for extent of ir	nspection & tests m	nay be added a	as applicable for the plant		5. Chemical Te	st (Sample)	22.		Adhesion T	est				37.	Cal	ration	D3. Relevant catalogu	les		
ā	and equipment	t				6. Ultrasonic Te	est	23.		Performanc	e Test/C	Characteris	tic		38.	Safe	ty Device Test	D4. Bill of matl./Item	no./		
3. 9	Separate ident	ification number wit	h quantity for	equipment shall be		7. Magnetic Pa	rticle Test (M	PI)		Curve					39.	Eas	e of Maintenance	Identification			
i	indicated where	ever equipment ha	ving same spe	cifications belonging		8. Radiography	Test	24.		No Load/ F	ree Runn	ning Test			40.	Fire	Test (Type Test)	D5. Matchmarks deta	ils		
1	to different fac	clincies are grouped t	togetner. atod undor Col	ump E for each itom		9. Dye Penetra	tion lest	25.		Load/ Over Moacuromo	10ad Tes	it woode			41.	Cha	py v-Notch Test	D5. Line/ Layout diag	ram		
	Estimated weight	this may be indicat	ed wherever a	ctual weights are not		11. Welder's Ou	alification &	20.		Accoustical	Test	eeus			43	ENE	(Electroless Nickel Plating)	procedures			
	available.					Weld Proced	ure Test	28.		Geometrica	I Accurac	v				Exe	cution	D8. Unpriced sub P.O	.with		
						12. Approval of	Test and Rep	29.		Repeatabilit	y and Po	sitioning			44.	Pair	ting	specification and	amend-		
						Procedure				Accuracy					45.	Ant	Static Test	atic Test ments, if any			
1	ABBREVIATI	ONS USED :	KEY TO SY	MBOLS :		13. Heat Treatm	nent	30.		Proving Tes	st				46.	Hyc	rostatic Double Block &	D9. Calibration Certific	ate of		
	CONTR	: CONTRACTOR	* : MFR/ CO	ONTRACTOR - AS APPLICABL	E	14. Pressure Te	st	31.		Surface Pre	paration				47	Blee	d Test	all measuring inst	ruments		
	мгк н		EI** : IESI IG	D BE PERFORMED, IF APPLIC	ABLE	15. Leakage Tes	ST.	32.		for bought-	ers rest	c Certificat	es		47.	Fun	matic Double Block &	D10 Y-Ray Reports			
, i	R	: REVIEW				17. Vibration Te	st	33.		IBR/ Other	Statutor	v agencies	5		-10.	Blee	d Test	DIG. A Ray Reports			
1	Ŵ	: WITNESS				171 1010001110		55.		compliance	certificat	e				5.00					
F	Р	: PERFORM																			
			E	QUIPMENT DETAILS	T	1						INSPEC	TION AND 1	TESTS			Test Certificates &	Acceptance Criteria	REMARKS/		
SI.	Description	(with equipment	Identification	Quantity	Unit	Manufacturer's	Expected	R	aw Ma	terial and In	-Process			Fi	inal Inspection/ Test	: by	Documents to be	Standards/ IS/ BS/	SAMPLING PLAN		
No.	heading, plac	e of use and brief	No.	NO./M	Weight	Name and Address	Schedule of		St	age Inspect	on						submitted to MECON	ASME/ Norms and			
	spec	LIICALIOIIS)	-	(AS per MK)	(Kg)		rinai mspii.	MFR		TPI		MECON	MFR		TPI	MECON		Documents			
1		2	3	4	5	6	7	8		9		10	11	-	12	13	14	15	Butt welding to be		
								1,2,3,4	P	1,2,3,4	W	-	1,2,3	P	1,2,3 W	1,2,3 F	1,2,3,4,5,6,8,9,11,12	D1,D2 &	radiographed 100%		
	CARTRIDGE F.	ILIEK/SCRUBBER	-	-	-			5,0,35,41	P	5,0,35,41	VV P	-	9,14	P	9,14,44 W	9,14,44	13,14,31,32,34,35 41 44 DQ	ASME SEC-VIII, DIV-1			
								11,12	г	11,12	ĸ		31 32 44	P	31.32 R	31.32	50,77,17	ASHL SLC-IA			
			ļ				I			I											
																QAI	NO.		REV		
	FOR MECON (S	tamp & Signature)			For CONT	KACIOR/ SUB-CONTRA	CIOR									CUT	ET 1 OF 1		U		
						(Stamp & Signature)										SHE					

\* To be field by party as per index above & approved by MECON

																	FORM NO. 1	1.20(4.4)F-09 REV-0		
	CONTRACTOR					QUALIT	Y ASSUF	RANC	E PLA	Ν			PROJECT :							
	ORDER NO. & D	ATE					FOR						BID DOCUMENT	NO. :						
7	मेकॉन SUB-CONTRACT	OR				STRUCT	URAL AND	MECH	HANICA	L			ITEM NAME : PR	ESSURE S	AFETY VALVE/CREEP RELIEF VALVE					
-0	GRDER NO. & D	ATE					EQUIPM	ENT					SPEC. NO.: MEC/	TS/05/62/	056, RE\	/-01				
INST	RUCTIONS FOR FILLING UP :				CODES FOR EXTENT	OF INSPECT	ION, TESTS, TE	ST CERT	TIFICATES	& DOC	UMENT	S :								
1.	OAP shall be submitted for each	of the equipm	ent separately with break	up	Code Description		Code	Ĺ	Description				Code		Descri	otion	Code DOCUMENTS:			
	of assembly/sub-assembly & par	t/component o	r for group of equipment		1. Visual		18.	Amplitud	le Test				34.		Interna	al Inspection Report	D1. Approved GA drav	vings		
-	having same specification.				2. Dimensional		19.	Sponge 7	Test	<b>-</b> .			25		by Cor	tractor	D2. Information and o	ther		
2.	Use numerical codes as indicates	for extent of	Inspection & tests and Iditional codes & descripti	on	3. Fitment & A 4. Physical Tes	ignment t (Sample)	20.	Dust/ Wa	ater Ingress Factor Test	lest			35. 36		Hardne Spark	ess lest Test for Lining	reference drg/ sta	mped		
	for extent of inspection & tests (	nav be added a	as applicable for the plant		5. Chemical Te	st (Sample)	22.	Adhesion	n Test				37.		Calibra	tion	D3. Relevant catalogu	es		
	and equipment	,	···· · · · · · · · · · · · · · · · · ·		6. Ultrasonic Te	est	23.	Performa	ance Test/C	naracteri	istic		38.		Safety	Device Test	D4. Bill of matl./Item r	no./		
3.	Separate identification number v	vith quantity fo	r equipment shall be		<ol><li>7. Magnetic Pa</li></ol>	rticle Test (MPI	)	Curve					39.		Ease o	f Maintenance	Identification			
	indicated wherever equipment h	aving same spe	ecifications belonging		8. Radiography	Test	24.	No Load/	/ Free Runn	ing Test			40.		Fire Te	st (Type Test)	D5. Matchmarks detail	s		
4	to different facilities are grouped	l together.	olumn E for oach itom		9. Dye Penetra	tion Test	25.	Load/ Ov	erload lest	ode			41.		Charpy	V-Notch Test	D6. Line/ Layout diagr	am		
4.	Estimated weights may be indice	ated wherever :	actual weights are not		11. Welder's Ou	alification &	20.		ral Test	eus			43		FNP (F	lectroless Nickel Plating)	procedures			
	available.				Weld Proced	lure Test	28.	Geometri	ical Accurac	y			101		Execut	ion	D8. Unpriced sub P.O.	with		
					12. Approval of	Test and Repai	r 29.	Repeatat	bility and Po	sitioning	9		44.		Paintin	g	specification and a	amend-		
					Procedure			Accuracy					45.		Anti-St	atic Test	ments, if any			
	ABBREVIATIONS USED :	KEY TO SY	MBOLS :		13. Heat Treatm	ent	30.	Proving	Test				46.		Hydros	tatic Double Block &	D9. Calibration Certific	ate of		
	MER · MANUEACTUR	* : MFR/ CO	RE DEDEORMED TE ADDI TCARL	F	14. Pressure Tes	5C +	31. 32	Manuface H	reparation	Certifica	tec		47		Function	lest mal Test	all measuring instr	uments		
	H : HOLD	. 1251 10	DE FERIORIED, II AFFEICADE		16. Balancing		52.	for boual	ht-out item	s certinea	1003		48.		Pneum	atic Double Block &	D10. X-Ray Reports			
	R : REVIEW				17. Vibration Te	st	33.	IBR/ Oth	er Statutor	, agencie	es				Bleed	Test	,,,			
	W : WITNESS							complian	ce certificat	e										
	P : PERFORM						1			ICDECT		TECTO						DEMARKS/		
SI	Description (with equipment	Identification		Unit	Manufacturer's	Expected	Raw Mat	erial and	In-Process	ISPECTI	ON AND	Fin:	al Inspection/ Te	st hv		Documents to be	Acceptance Criteria Standards/ IS/ BS/	SAMPLING PLAN		
No.	heading, place of use and brief	No.	No./M	Weight	Name and Address	Schedule of	Sta	age Inspe	ection							submitted to MECON	ASME/ Norms and			
	specifications)			(Kg)		Final Inspn.	MFR		М	ECON	MF	R		ME	CON		Documents			
	2	2		-				TF	Ы	10			TPI	· · ·			45	16		
1	2	3	4	5	6	/	в 1.2.3 Р	9	w	-	1.2.3	Р	1.2.3 W	1.2.3	.3 R	14	15 D1.D3.D8.D10	16 47		
	PRESSURE SAFETY VALVE (PSV)	- 1	-	-			4,5 P	5,41	Ŵ	-	14,15	P	14,15 W	14,47	R	31,32,34,41,44,47	ASME SEC-VIII, DIV-1	100%		
	/CREEP RELIEF VALVE (CRV)						8,41 P	8	R		31,32	Р	44,47 W				MECON TS			
											44,47	Р	31,32 R	-			APPROVED DS			
							<u>├</u> ── <del> </del> ──						<u>├</u>	+	+					
															QAP N	0.		REV		
	For MECON (Stamp & Signature			For CONTI	RACTOR/ SUB-CONTRAC	TOR									L			0		
					(Stamp & Signature)										SHEET	1 OF 1				
		·													1					

\* To be field by party as per index above & approved by MECON

		CON	ITRACTOR													C	LIENT								
		ORE	DER NO. & DATE					QU	<b>ALITY</b>	Λ A	SSUR	RAN	ICE I	PL	AN	PA	ACKAGE	NO	). (	030					
50.90	मेकान 01:2000 Company	SUB	-CONTRACTOR			_		INST	RUME	NT	FOR ATIO	N E	QUII	PM	ENT	PA	ACKAGE	NA NA	ME	мет	FERING SKID				
INS	TRUCTIONS	S FOR F	FILLING UP :			_	CODES F	FOR EXTEN	T OF INS	SPEC	TION,	TEST	rs, tes	ST C	CERTIFIC	CATI	ES & DO	CUM	IENTS:						
1.	QAP shall b assembly/su	be subm ub-asser	nitted for each of the equipme nbly & part/component or for	nt separately or group of e	with quip	n break up of oment having	Code Do	escription		(	Code I	Descr	iption			Co	ode Deso	cripti	ion		DOCUMENTS:				
2	same specif	fication.	des as indicated for extent	of inspecti	0.0	& tests and	1. Ma Uni	rking, Taggii ique Srl. No	1g , Finish		12. Ro	outine	e test as	s per	relevant	IS	23. Elect	rical	/Conduit E	ntry	D1. Approved GA	drawings			
2.	submission	of test	certificates & documents. A	dditional coc	les &	& description	2. Din	nensions	1/111511		12(a).	Acce	ssories	che	ck		24. Oper	ation	al & functi	onal	D2. Approved sin	ngle line/			
	for extent o	of inspe	ction & tests may be added a	s applicable	for	the plant and	3. Fitr	nent & Aligr	iment		13. T	ype to	est as p	er re	levant IS/	/	check	• Te	et		schematic dia	gram ta sheet / D	20		
3.	Separate ide	entificat	tion number with quantity for	equipment s	shall	be indicated	5. Che	emical Test			14. In	npuls	e Test	u			26. Flam	e Pro	oof Test		D3. Approved da	la sileet / Do	<i>.</i>		
	wherever ec	quipmer	nt having same specifications l	belonging to	liffe	rent facilities	6. Ult	rasonic Test	a Test ()		15. H	yster	esis				27. Three	ading	g		D4. Approved bil	l of materia	ls		
4.	Weight in	togethe	er. (T) must be indicated und	er column 5	5 fo	r each item.	7. Ma 8. Rac	liography Te	st	(PT)	10. A	ccura nclos	cy ure Pro	otecti	on Test		28. Proce Check	ss C	onnection		D5. Unpriced P.C D6. Calibration C	Certificate			
	Estimated w	veights i	may be indicated wherever act	ual weights a	re n	ot available.	9. Sur	face Defect (	DPT)		18. 0	Calibr	ation				29. Indica	ator (	Operation		of all measur	ing			
ABI	BREVIATIO	NS USE	ED :				10. Me a) ]	asurement of Before HV T	IR Value	e	19. N 20. I	voise P/Haz	& Vibi zardous	ratio s are	n a/COC/		Check 30. Comn	unio	cation with		instruments				
COI	NTR :	CONT	RACTOR				b) .	After HV Te	st		Т	est/C	alibr./1	Гуре	Test Cert	t.	Field	Com	municator						
MF	R ·	MANI	IFACTURER				11. Hig	sh Voltage te	st/Dielect	ric	21. C	)ver ra Repea	ange pr tability	rotec	tion		31. IR at a 32. N2 Tes	mb.	& at 500V	DC					
1011 1	K .	1012 11 11 1	STRETORER		11(a). Pa	int Test			22(a) l	Impac	et Test			3	3. Respon	se T	ime test								
TPI	: 7	THIRD	PARTY INSPECTION AGE	ED BY	34. Heat	eat Treatment     35. Hardness Test     41. Charpy V-Notch Test       NP (Electroless Nickel Plating) 44. Pressure Test     45. Fire Test (Type Test)       ti-Static Test     45. Fire Test (Type Test)																			
		עעום	LK)		46. Anti-S	Static Test	INICKCI I	iaun	g) 44. I	10550	ie rest			4.	5. Phe 1es	n (1	ype rest)	akage test							
							11(b) Pne	umatic Test f	or Intern	al se	al & ext	ernal	tightne	ess a	nd functio	onal	test (as pe	r EN	334 for PC	CV 8	& 14382 for SDV)				
							11(c) Pres	ssure retainin ctional Test	g parts		11(a) 11(g)	Plug/ Leak	seat lea Test	акад	e test	1	11(e) Pack $11(h)$ Ru	ing i ining	leakage test g Test						
							11(i) Actu	ator stroking	g Time		(8)								6						
			EOUIPMEN	T DETAIL	s		11(j) Acco	eptance Tests	s as per re	eleva	nt IS/ ot	ther st	tandard	1 TIO	DN AND	) Т	ESTS				Test Certificates &	Acceptan	REMAR		
					5																documents to be	ce Criteria Standards	KS/ SAMPLI		
SI. No	Desciptio Of item	i I	Indentificatum No.	Quantity		Name and Addre	s Expec	al Inspection	Raw Mat	erial a	nd inproc	ess stag	ge inspect	10n		F	inal Inspectio	n/Tes	st by		MECON	/IS/ BS/ASM	NG		
				No/M	Т				MFR		TP	I	CLIEN	JT/	MFR		TPI		CLIENT	,		E/ Norms	I LANY		
													MECC	ON					MECON			Document			
1	2		3	4	5	6		7	8		9	•	10		11		12		13		14	15	16		
01	Pressure gaug Temperature	ges / gauges	Refer P&ID, Tech. Spec. and Data	*											1,2,16,1	в	1,2,16,1 8,21,22	w	1,2,16,18,	в	1,2,16,18,20,21,22,D	D3 & Tech.	10%		
01	/ Diff. Pressure gauges	e	Shts.	Nos.	-			-	-	-	-	-	-	-	20		20	R	21,22,20	ĸ	5,D6	Spec., EN 837-1	Witness		
	Pressure Transmitter /	Diff.													1.12(a).		1,12(a), 23.28.1	***	1.12(a).23			D2			
02	Pressure Transmitter /		Refer P&ID, Tech. Spec. and Data Shts.	* Nos.	-			-	-	-	-	-	-	-	23,28,16	Р	6,18,22, 29,30	w	,28,16,18, 22,29,30,2	R	1,12(a),23,28,16,18,2 2,29,30,20,D5,D6	& Tech.	10% Witness		
	Temperature														9,30,20		20	R	0			Spec.			
	Desistance Ter			1						t				$\square$	1,2,5,25,		1,2, 25 16 3		1 2 5 25 1			D2 &			
03	Detector with	mp.	Refer P&ID, Tech. Spec. and Data Shts.	* Nos.	-	-		-	-	-	-	-	-	-	16,32,18 ,33,20,2	Р	25,10,5 2,18,33,	w	6,32,18,33	R	1,2,5,25,16,32,18,33, 20,27,23,D5,D6	Tech.	10% Witness		
	Thermowell														7,23		5,20	R	,20,27,23			Spec.			
			Refer P&ID. Tech Snee and Data												1.2.24 1		1,2,24,1 8	W	1.2.24 18			D3 & Tech.	100%		
04	Flow Comp	uters	Shts.	* No.	-	-		-	-	-	-	-	-	-	8,20	Р	20	R	20	R	1,2,24,20,18,D5,D6	Spec., AGA-	Witness		
									1					1								3,5,7,8,9			

05	Orifice flow Meter	Refer P&ID, Tech. Spec. and Data Shts.	* No.	-	-	-	4,5,6,7, 8,9,34, 35,41	Р	4,5,6, 7,8,9, 34,35 ,41	R	4,5,6 ,7,8, 9,34, 35,4 1	R	1,2,11(f) ,18,11(a ),25,11( g),20,23	Р	1,2,11(f) ,18,11(a ),25,11( g),23 20	W R	1,2,11(f),1 8,11(a),25 ,11(g),20, 23	R	1,2,4,5,6,7,8,9,34,35,41 ,11(f),18,11(a),25,11(g) ,20,23,D5,D6	D3 & Tech. Spec., AGA-3	100 % Witness
06	LEL Detection System	Refer P&ID, Tech. Spec. and Data Shts.	* Nos.	-	-	-	-	-	-	-	-	-	1,2,18,2 0,24,23	Р	1,2,18,2 4,23 20	W R	1,2,18,20, 24,23	R	1,2,18,20,24,23,D5,D 6	D3 & Tech. Spec.	10% Witness
	Gas														1,2,24,2	w				D1 D2 D3	
07	Water Dew Point Analyzer, Hydrocarbon Dew Point Analyzer	Refer P&ID, Tech. Spec. and Data Shts.	* Nos.	-	-	-	-	-	-	-	-	-	1,2,12,1 3,20,24, 23	Р	12,13,20	R	1,2,12,13, 20,24,23	R	1,2,12,13,20,24,23,D 5,D6	,D4 & Tech. Spec.	100% Witness
08	Control Panel	Refer P&ID, Tech. Spec. and Data Shts.	* Nos.	-	-	-	-	-	-	-	-	-	1,2,3,10, 11,11(a) ,12,13,2 4,20	Р	1,2,3,10, 11,11(a) ,24 12,13,20	W R	1,2,3,10,1 1,11(a),12 ,13,24,20	R	1,2,3,10,11,11(a),12, 13,24,20,D5,D6	D1,D2,D3 ,D4 & Tech. Spec.	100 % Witness
09	Cables	Refer P&ID, Tech. Spec. and Data Shts.	* Nos.	-	-	-	-	-	-	-	-	-	1,2,4,5,1 0,11,12, 13,11(j)	Р	1,2,4,5,1 0,11,11( j) 12,13	W R	1,2,4,5,10, 11,12,13,1 1(j)	R	1,2,4,5,10,11,12,13,1 1(j),D5,D6	D3 & Tech. Spec., rel. IS / other std.	10% Witness

Note :-

1. This is Typical QAP. Bidder to prepare the QAP considering the tests / stds. (as applicable) & submit for review/approval. Review/Witness category & stage category will be finalized during approval.

2. Test Certificates of Barrier, SDC, SS tube, fittings, Manifolds, GSM Modem, Laptop, Printer, Junction Boxes, Cable Glands, solar panel, batteries, charge controller, canopies, cable tray, earthing wire, etc. shall be reviewed by TPI.

\*Qty as per MR /PO and P& Id requirement.

P: Perform, W: Witness, R: Review

For MECON (Stamp & Signature)

for CONTRACTOR /SUB-CONTRACTOR (Stamp & Signature) QAP No.: MEC/05/E5/QAP - 030

R0

-						011.0.1		0110			A NI				01507					FORM NO.	11.20(4.4)F-09 REV-0		
100		_				QUALI	ITAS	501		: PL/	AN												
मेकॉन	ORDER NO. & DAT	E					NOWDI			т				PAC									
The second	SUB-CONTRACTO	{				1	NSIRU	MEN		N				PAC	CKAGE N	VAME :				1			
	ORDER NO. & DA	E					EQU	ЛРМ	ENI					ITE	M NAME	: GAS	POWERED	ACTUAT	OR				
INSTRUCTIO	NS FOR FILLING UP :				CODES FOR EXTENT	OF INSPECTION	ON, TESTS	, TEST		CATES 8	& DOCUM	IENTS :											
1. QAP sh	all be submitted for each o	the equipment	separately w	ith break up	Code Description		Code		L	Descriptio	on				Code			Descrip	tion	Code DOCUMENTS:			
of asse	mbly/sub-assembly & part/	component or fo	or group of eq	uipment .	1. Visual		18.		Amplitude	e Test					34.			Interna	Inspection Report	D1. Approved GA draw	ings		
having	same specification.	or outont of inc	nantion 9. toot	to and	2. Dimensional	nnont	19.		Sponge T	est	an Toot				25			by Cont	ractor	D2. Information and of	her		
2. Use nui submis	sion of test certificates & do	cuments. Addit	ional codes &	description	4. Physical Test	(Sample)	20.		Friction Fri	actor Tes	ss rest				35. 36.			Spark 1	est for Lining	dras released for r	npeu nfa.		
for exte	nt of inspection & tests ma	y be added as a	pplicable for	the plant	5. Chemical Tes	t (Sample)	22.		Adhesion	Test					37.			Calibra	ion	D3. Relevant catalogue	s		
and eq	uipment identification number wit	h guantitu far a	auinmont cho	ll bo	<ol> <li>Ultrasonic Tes</li> <li>Magnetic Part</li> </ol>	iele Teet (MDI)	23.		Performa	nce Test,	/Character	ristic			38.			Safety	Device Test	D4. Bill of matl./Item n	0./		
indicate	d wherever equipment has	ing same specif	ications belor	naina	8. Radiography	Test	24.		No Load/	Free Rur	nnina Test				39. 40.			Fire Te	st (Type Test)	D5. Matchmarks detail	5		
to diffe	ent facilities are grouped t	ogether.		.99	9. Dye Penetrati	on Test	25.		Load/ Ove	erload Te	est				41.			Charpy	V-Notch Test	D6. Line/ Layout diagra	am		
4. Weight	in kilograms must be indic	ted under Colu	mn-5 for each	n item.	10. Metallographi	c Exam.	26.		Measuren	nent of S	Speeds				42.			Operat	onal Torque Test	D7. Approved erection			
Estimat	ed weights may be indicate	d wherever act	ual weights ar	re not	11. Welder's Qua Weld Procedu	lification &	27.		Accoustic	al Test cal Accur	acv				43.			ENP (El	ectroless Nickel Plating)	procedures D8 Unpriced sub P O	with		
availab					12. Approval of T	est and Repair	20.		Repeatab	ility and	Positioning	g			44.			Painting	)	specification and a	mend-		
					Procedure				Accuracy						45.			Anti-Sta	itic Test	ments, if any			
	VIATIONS USED :	KEY TO SY	MBOLS :		13. Heat Treatme	nt	30.		Proving T	est renaratio	n				46.			Hydros Blood T	atic Double Block &	D9. Calibration Certific	ate of		
CONTR	. CONTRACTOR	· · · · MPRy CO	NTRACTOR - AS A	FFLICABLE	15. Leakage Test		51.		Surface r	reparado	511				47.			Functio	nal Test :	an measuring maa	umento		
					a) Piston Sea													a) Elect	rical and pneumatic functional test.				
					b) Pneumatic	& Hydraulic												b) Insu	ation Test of Electrical components.				
					Connection													d) Chec	k of limiting device operation.				
																		e) No lo	ad test (DP=0) & load (DP max) Operations with				
																		the min	imum required				
																		f) Manu	pressure. al Override functional test				
																		g) Verif	ication of actuator regulated Pressure.				
																		h) Actu	ated Valve Open/cliose at 10 kg/cm2(g) actuator				
																		regulat	ed pressure & 10 kg/cm2(g) differential pressure				
MFR	: MANUFACTURER	** : TEST TO	BE PERFORMED, I	F APPLICABLE	16 Palancing		32.		Manufact	urer's Te	st Certifica	ates			10			Droum	the Valve (at Valve manufacturer 3 works)	and gauges			
	. HOLD				10. balancing				TOT DOUGH	t-out iter	1115				40.			Fileum	are bouble block &	DIO. A-Kay Reports			
R	: REVIEW				17. Vibration Test	:	33.		IBR/ Othe	r Statuto	ory agenci	es						Bleed T	est				
W	: WITNESS								compliance	ce certific	cate												
P	: PERFORM	EOUIF	MENT DETAI	LS							INSPECT	TION ANI	D TESTS	S					Test Certificates &	Acceptance Criteria	REMARKS/		
SI. De	scription (with equipment	Identificati	on Quantity	Unit	Manufacturer's	Expected	R	Raw Ma	aterial and I	n-Proces	55	1		Final I	Inspectio	on/ Tes	st by		Documents to be	Standards/ IS/ BS/	SAMPLING PLAN		
No. hea	ding, place of use and brief	No.	No./M	Weight	Name and Address	Schedule of		St	tage Inspec	tion							1		submitted to MECON	ASME/ Norms and			
	specifications)	(AS per Mi	K)	(Kg)		Final Inspir.	MFF	۲	TP	Ί	MECON	Ν	1FR		TPI		MEC	N		Documents			
1	2	3	4	5	6	7	8	1	9		10		11	_	12		13		14	15	16		
							5,6/7/8/9*	•	6/7/8/9*,			1,2,3,1	5								100%		
G	AS POWERED ACTUATOR		As per PC	- 10			,14,15(a), 42.23	Р	14,15(a), 42.23	w#	. I	(D),31, 4 47	4 p	1,2 b)	44 47	w	1,2,3,15( b) 44 47	R	1 2 3 5 6 7 8 9 14 15(a) 15(b) 23	D2 D3 D4 D6 MECON TS	(For prototype test -		
			· ·				72,23	<u> </u>	72,25			(ד,ד	+ '	- 0,,	(ד,דד,	**	0),++,+/	ĸ	1,2,3,3,0,7,0,3,17,13(0),13(0),23	D2,D3,D4,D0, PIECON 13	10%)		
									5	R		32,33	R	31,	,32,33	R	31,32,33	R	31,32,33,42,44,47				
*Tests	as applicable shall be carrie	ed out on hydrau aulic & pnoump	ulic & pneuma tic cylinder	atic cylinder.																			
NOTE :	-	autic & prieuria	ue cynnuer.																				
1. 3.2 I	nspection Report shall be p	rovided.																					
2. Test	No. 47 shall also be carried	out on valve+a	ctuator asser	nbly at valve	manufacturer's works.																		
5. All tr	e roj internartest reports		incates shall t	Je i eviewed l	UY 1F1.																		
F 145	CON (Champ & Cirration )																	QAP NO	. MEC/05/E5/STD./QAP/AV		REV		
For ME	LON (Stamp & Signature)			FOR CONTR	(Stamp & Signature)													SHEFT	1 OF 1		1		
					(samp a signature)													SALLI					

# **ANNEXURE - VII**

# FAT PROCEDURE/ FAT MANUAL

# FAT PROCEDURE / FAT MANUAL For METERING SKIDS

(This is Typical FAT Procedure. Bidder to prepare & submit the Factory Acceptance Test Procedure for review / approval)

Page 249 of 644

MECON LIMITED DELHI	FAT PROCEDU FOR MET	RE / FAT MANUAL ERING SKIDS	
INSTRUMENTATION	Doc. No: MEC	C/05/E5/FAT- 030	मेकान 🗛
SECTION	Rev. 1	Page 2 of 21	0 3001:2000 Company

# TABLE OF CONTENTS

# 1. PREFACE

Introduction Scope Reference Documents Orientation of Witnesses Test Certificates Action List

# 2. VISUAL INSPECTION

Skid Review Test Certificates

# **3.** FACTORY ACCEPTANCE TEST

- Hydrostatic Test Pneumatic Leak Test Functional Test of flow computer Transmitter Test Procedure Orifice Flow Meter Control Panel Gas Chromatograph LEL system Solar Panel
- 4. SPARES
- 5. PAINTING
- 6. DOCUMENTS REVIEW
- 7. ANNEXURES

MECON LIMITED DELHI	FAT PROCEDU FOR MET	RE / FAT MANUAL ERING SKIDS					
INSTRUMENTATION	Doc. No: MEC	C/05/E5/FAT- 030	मेकान अ				
SECTION	Rev. 1	Rev. 1 Page 3 of 21					

## 1. PREFACE

# Introduction:

This document defines procedure for Factory acceptance Test (FAT) to be carried out for Metering Skid(s). The purpose of carrying out FAT is to check functionality of the entire natural Gas Skid with respect to agreed technical documents.

NOTE: All the tests, inspection, checking, data-configuration etc shall be conducted as per FAT procedure by the vendor and the same shall be submitted to Mecon for review. The records / test reports/ calibration reports/ certificates, configuration sheet for flow computers and other relevant document should be prepared / obtained by the vendor and sent to Mecon for review.

#### Scope.

The FAT will confirm the compliance of ORIFICE based Metering Skid with the project specifications. The purpose of FAT is to check the performance of each component as well as entire skid. Upon completion of the test described in the following procedures, the system will be considered to be ready to be dispatched to the site. All the physical & functional tests are described in the remaining sections. When each functional test has been completed, the TPIA will indicate acceptance of the tests by signing the test certificates.

All the equipments / instruments/ items shall be installed (as far as possible) for demonstration during FAT. In case some of the items can not be erected/ installed during FAT, list of such item along with the reason for not installing the same shall be provided before start of FAT. FAT shall not be conducted before our approval of such list. In case difference in calculated values (calculated by different Instruments) is observed for any parameter, the list of such parameters along with values and suitable reason for such deviation to be submitted to us in advance for our review. In case deviation is observed in the parameter-values displayed by various components, the same shall be recorded and may result in to rejection of FAT.

All connectivity/ its simulation mentioned elsewhere in tender (like Laptop, Printers, SCADA/RTU, ORIFICE, FC, GSM Modem, GC, etc) shall be established and demonstrated during FAT. Non-compliance to any of these shall result in rejection of Metering skid

#### **Reference Documents:**

#### **Design Documents of Skid**

#### **Document No.**

- 1 P & ID for Skid Drawing No. -----
- 2 GA Drawing for Skid Drawing No. -----
- 3 Base frame and Foundation Drawing No. -----
- 4 Base frame Calculation Doc. No. -----
- 5 Quality assurance plan Mechanical Items Doc. No. -----

MECON LIMITED DELHI	ECON LIMITED FAT PROCEDURE / FAT MANUAL DELHI FOR METERING SKIDS								
INSTRUMENTATION	Doc. No: MEC	%/05/E5/FAT- 030	मेकान 🗛						
SECTION	Rev. 1	Page 4 of 21	0.9001:2000 Company						

\_\_\_\_\_

6 Quality assurance plan – Instrumentation Equipment Doc. No. -----

7 Material Test certificate, Radiography test report, Hydro-test certificate

# **Datasheet / Drawing of Equipments:**

- 1. Upstream/ Downstream Flow straighteners Drawing No.\_\_\_\_\_
- 2. Pressure Transmitter Data Sheet No.\_\_
- 3. Differential Pressure Transmitter- Data Sheet No.\_\_\_\_\_
- 4. RTD with Temperature Transmitter– Data Sheet No.\_\_\_\_\_
- 5. Pressure Gauge Data sheet No.\_\_\_\_\_
- 6. Temperature Gauge Date sheet No.\_\_\_\_\_
- 7. Flow Computer Data Sheet No.\_\_\_\_\_
- 8. Pressure safety Valve Data Sheet No.\_\_\_\_\_
- 9. Control Panel Data Sheet No.\_\_\_\_\_
- 10. Ball Valves ------ make Drawing No.\_\_\_\_\_
- 11. Plug Valves Drawing No.\_\_\_\_\_
- 12. Check Valves- Drawing No.\_\_\_\_\_
- 13. Cartridge Filter Design data Doc. No.\_\_\_\_\_
- 14. Orifice Flow meter Data sheet No.\_\_\_\_\_
- 15. Gas Chromatograph –Data sheet No.\_\_\_\_\_

# **Orientation of Witnesses:**

All representatives shall be briefed on details/ description/ operating principles of the ORIFICE Metering Skid for this project before commencing the FAT.

# **Test Certificates:**

Upon completion of the tests mentioned in this document, Test Certificates should be filled with the results and signed / stamped by all the parties.

# Action List:

Any discrepancies noted during the functional tests shall be defined, recorded and summarized in the Action List Form. Once proper action has been taken on those points, this Action List shall be signed / stamped by Customer/ TPIA duly filled with the results.

# 2. VISUAL INSPECTION OF SKID.

# 2.1 Skid Review:

The Skid will be inspected for installation of all the components as per approved P & ID and approved G.A. Drawing. Dimensional Checking shall be done as per approved G.A. Drawing. Skid will also be inspected for correctness of installed Equipments/ instruments and approachability for maintenance. The Skid will be inspected for proper support with neoprene sheet / rubber pads / clamps for the major equipments/ Instruments and pipes / tubes.
MECON LIMITED DELHI	FAT PROCEDURE / FAT MANUAL FOR METERING SKIDS		
INSTRUMENTATION	Doc. No: MEC	C/05/E5/FAT- 030	मेकान
SECTION	<b>Rev. 1</b>	Page 5 of 21	9001:2000 Company

### 2.2 Test Certificate for Visual Inspection:

Upon completion of the visual inspection described in this section, the test certificate for Visual Inspection of skid should be dully filled and signed by both parties.

### **3.** FACTORY ACCEPTANCE TEST:

### **3.1** Hydrostatic Test (if not performed earlier)

- 1. Complete skid shall be offered for hydrostatic test.
- 2. Non Corrosive water shall be used for hydro testing at ambient temperature.
- 3. During the hydro test, Flow meter & filter cartridges / elements shall not be installed with pipeline as internals will get damaged in contact with water.
- 4. All valves shall be kept at open position.
- 5. Either inlet or outlet pipe shall be closed with blind flange and the other side will be connected with water line (hose pipe) to fill up water in side the skid.
- 6. After filling water, the skid shall be pressurized to 1.5 times design pressure.
- 7. Maintain this pressure for six hours.
- 8. Pressure should not exceed hoop stress of the line pipe at any given moment of time.
- 9. No leakage is allowed through any joints and shell.
- 10. Record pressure and temperature reading at 30 minutes interval.

### 3.2 Pneumatic Test

- 1. After completion of hydro-test, drain out water from the skids.
- 2. Pass the air through the entire skid for some time to dry out the pipeline.
- 3. Assemble the Flow meter and impulse tubes in pipe line.
- 4. Keep all valves open and close either inlet or outlet side pipe with blind flange.
- 5. Pressurize the entire skid by nitrogen up to max. operating pressure and hold for one hour.
- 6. No leakage/ drop in pressure is allowed.
- 7. Check the leakage using soap water.

#### **3.3** Functionality Test of Flow Computer

- 1. Configuration check, parameter settings/ checking etc shall be carried out as per approved Data Sheet/ FDS.
- 2. Volumetric flow rate, Mass flow rate by simulation of Flow meter signal using Function Generator/ any other equipment. Flow, Energy and Mass Totalizers, shall be also be checked.
- 3. Line Temperature and pressure, Diff. Pressure etc. shall be checked by inputting respective analog signals.
- 4. Interface data of flow computer with RTU Measurement by Laptop.
- 5. Visual checking of instrumentation cable connection including cable gland, cable tag nos., JB mounting installation, cable dressing etc. as per approved drawing, wiring diagram.
- 6. Facility of entry of Atmospheric pressure and temperature by operator to be verified.
- 7. Various Units of measurement (Pressure, flow rate, heating value, energy etc.) shall be checked as per approved documents.
- 8. Simulation of Flow Computer output to SCADA by GSM Modem.
- 9. GC data input to Flow computer from SCADA via GSM modem.

MECON LIMITED DELHI	FAT PROCEDURE / FAT MANUAL FOR METERING SKIDS		
INSTRUMENTATION	Doc. No: MEC	C/05/E5/FAT- 030	मेकान
SECTION	<b>Rev. 1</b>	Page 6 of 21	9001:2000 Company

- 10. Proper entry of the Orifice flow meter error curve, K factors etc.
- 11. Checking of Audit Trail facility in the flow computer and its printing.
- 12. Modbus address details to be verified.
- 13. Availability and functioning of USB ports with connecting cables for communication of various devices/Instruments/ Equipments (like GC, ORIFICE, RTU/ SCADA, Printers, Laptop etc) with flow computers.(Functioning of standalone Software for flow calculation verification to be checked.)

### **3.4 Transmitter Test Procedure** (Pressure, Differential Pressure, Temperature):

Provide power supply to the transmitter and check the local indication, in FC, RTU/SCADA TB & output during the pneumatic test. Calibration report (from approved Lab) and Hart functionality to be checked

### 3.5 Orifice Flow Meter :

The Orifice Flow meter has been manufactured by ------ and is calibrated at ----------- using Natural Gas as per approved data sheet. The calibration Certificates shall be submitted for review during FAT. Insulation of metering system and Environmental enclosure for custody metering equipments shall be checked visually during FAT.

#### **3.6 Control Panel :**

- 1. Check the complete functional test as per loop diagram / Data sheet.
- 2. Hot stand by functionality of Power supply in control panel to be checked.
- 3. Visual Check: Wiring diagram and marking
- 4. Operation of Fan, Light and other items.
- 5. Check the painting quality, cable identification mark, tags, cable dressing,
- 6. Door alignment, locks and overall dimensions to be checked.
- 7. IS ground and panel ground shall be independent and to be checked.
- 8. Check the availability of spare items & space.

# **3.7 Gas Chromatograph:**

- 1. GC controller/ display unit and interface unit Installation in control panel shall be checked as per approved data Sheet/ FDS.
- 2. Parameter settings/ software loading checking, default values etc shall be carried out for GC controller/ remote configurator as per approved data Sheet/ FDS.
- 3. Probes, Tools for insertion/ removal in pipe shall be also be checked.
- 4. Field Analyser unit with stand shall be also be checked as per Data sheet.
- 5. Calibration Gas cylinders, carrier gas cylinders, self acting pressure regulator for Cylinders (calibration gas/ carrier gas) shall be checked.
- 6. Report for composition of calibration gas shall comply to the specifications.
- 7. Shed for Analyser, Cable for remote configurator/ controller etc shall also be checked.
- 8. Communication between GC and Laptop shall also be checked and GC data simulation to be checked.
- 9. Simulation of GC data input to Flow computer, Metering supervisory system and RTU/ SCADA via GSM modem.

MECON LIMITED DELHI	FAT PROCEDURE / FAT MANUAL FOR METERING SKIDS		
INSTRUMENTATION	Doc. No: MEC	C/05/E5/FAT- 030	मेकान अ
SECTION	<b>Rev. 1</b>	Page 7 of 21	0.3001:2000 Company

- 10. Availability and functionality checking of communication with flow computers, Laptop, Metering supervisory system, RTU/ SCADA, Printer
- 11. GC Calibration shall be performed and chromatogram for 3 consecutive runs shall be recorded as base for future. Also heating value as displayed by GC and as calculated by FC shall be checked for exact matching and shall comply to GPA 2145 / 2172.
- 12. Modbus address details to be verified.

# **3.8** LEL Detection system:

- 1. Installation and Simulation of LEL monitor/ controller to be checked.
- 2. Installation & functioning of LEL Detectors to be checked.
- 3. LEL calibration gas cylinder, Regulator, portable LEL detectors etc. to be verified.
- 4. Verification of calibration report for Calibration mixture, CCOE certification for cylinders etc to be done.
- 5. Signals in RTU to be verified.
- 6. Modbus address details to be verified.

### 3.9 Solar Panel:

- 1. For each stream, check the functional test as per approved Document.
- 2. No. of Batteries as per approved Battery sizing Calculation to be checked
- 3. Output Voltage of each streams shall be recorded.
- 4. Visual Check: Support for Solar Panel, Battery enclosure/ mounting, Cable, identification

#### 4.0 Spares:

List to be prepared as per P.O. terms/ tender documents for all the mandatory spares, commissioning spares, cables, and materials required for erection. All the spares shall be verified by TPIA during FAT test. All spares shall be identified with proper tags. Foundation bolt and mating flanges shall be checked during FAT.

#### 5.0 Painting:

As per Std. Specification (Specification For Shop & Field Painting - MEC/S/05/21/07, Rev-0)

#### 6.0 Documents Review:

After completion of all tests, Documents like material test certificates, Radiography test, Hydrotest, calibration/ test reports etc. for Ball valves/ Plug valves/ CRV/ NRV/ PSV/Filters, pipes, fittings, Tee/ elbow/ weld joints, shall be reviewed to check the compliance with the tender specifications and approved QAP for the equipments / instruments. All the documents shall be signed & stamped by TPI & bidder.

# 7.0 Annexure to FAT Manual:

Apart from FAT procedure, FAT Manual shall have Annexure which shall include the following:

MECON LIMITED DELHI	FAT PROCEDURE / FAT MANUAL FOR METERING SKIDS		
INSTRUMENTATION	Doc. No: ME	C/05/E5/FAT- 030	मेकान
SECTION	Rev. 1	Page 8 of 21	0 3001:2000 Company

- a) List of all the loose items with details to be provided by vendor (like Mandatory Spares, commissioning spares, Consumables, Parts of metering skids, Parts of GC analyzer and shed, Parts of Control panel, mating flanges, Gaskets, materials required for installation of skid, Hand held communicators, portable LEL detector, Calibration gas, Printers with its accessories, laptop with its accessories, Connecting and power cable for laptop/ Printers/ peripherals, Earthing strips, Cable Trays, cable glands, plugs, blind flanges, lugs, ferrules, earthing Cables, Mounting Accessories, any other hardware required for establishing various connectivity mentioned elsewhere, Software etc.) to be enclosed as Annexure in the FAT Manual for our review/ approval. The items mentioned in the list (Annexure) shall be demonstrated during FAT.
- b) List of all the reports (like calibration reports for various Instruments, test reports for all the equipments, CCOE certificates etc.), with details like Report no., date, description etc shall also be prepared as Annexure of FAT manual. The same shall also be reviewed during FAT

Any discrepancy observed during FAT shall be recorded/ noted in the Action list. Issue of Inspection Release note/ Dispatch clearance shall be subject to compliance of all the FAT Check-list points/ Action list & submission of TPI release note for the complete skid.

MECON LIMITED DELHI		FAT PROCEDURE / FAT MANUA FOR METERING SKIDS	
INSTRUMENTATION SECTION		Doc. No: MEC/05/E5/FAT- 030	Hone I
		Rev. 1 Page	9 of 21
		TEST CERTIFICATE	
		<b>VISUAL INSEPCTION</b>	
Project	:		
Client	:		
P.O. No	:	Dtd	
Sr. No	POINT INSPECTED RESU		RESULTS FOUND OK
1	Skid checl	ked as per P & ID and GA Drawing	YES / NO
2.	Process pa	arameter in Name Plate	YES / NO
3	Lifting ho	ok	YES / NO
4	Copper ju	mper for all flanges	YES / NO
5	Earthing C	Connection at base frame	YES / NO
6	Insulation	for Metering system	YES / NO
7	Instrumen	tation cable connection, cable dressing, cable th	ay covers and
	JB mounti	ing installation etc. as per wiring diagram.	YES / NO
8	Identificat	ion Tags for cable and all Instruments	YES / NO
9	Painting colour of pipe line (golden Yellow)		YES / NO
10	Spares as per P.O. requirement		YES / NO
11	Foundatio	n bolt and Matting flanges	YES / NO
12.	Support fo	or all the equipments/ Instruments, Pipe, tubes,	Trays YES / NO
13.	Adequacy	of O&M Platform and approachability	YES / NO
14.	Canopy to	all transmitters	YES / NO
16	Operating	handles on all valves	VES / NO

TPIA

Client /Mecon Representative

MECON LIMITED DELHI INSTRUMENTATION SECTION			FAT PROCEDURE / FAT MANUAL FOR METERING SKIDS Doc. No: MEC/05/E5/FAT- 030		AL	
						के मेकान
			<b>Rev. 1</b>	Page	10 of 21	3001:2000 Com
			TEST CERT	IFICATE		
			HYDROTEST	<u>COFSKID</u>		
Project	:					
Client	:					
P.O. No	:		Dtd			
Test Pressu	re :					
Test Media Duration	:	Non Corros 6 Hours	ive Water			
Sr. No.	TIM	1E	Pres	sure		
1						
2						
4						
5						
6						
7						
8						
	<b>POIN</b>	[ INSPECT]	ED		<u>RESULT FO</u>	<u>DUND OK</u>
	) Look to	at at flange i	into and other conr	actions	VEC /	NO
a,	) Leak te	st at mange j		lections	165/	
Company I	Representat	tive	TPIA		Client /Mecon	Representative
Date:						

MITED I	FAT PROCEDUR FOR METE	E / FAT MANUAL RING SKIDS	
TATION	Doc. No: MEC	/05/E5/FAT- 030	मेकान
DN	Rev. 1	Page 11 of 21	9001:2000 Company
	<u>TEST CERTII</u>	FICATE	
	PNEUMATIC TES	ST OF SKID	
:			
:			
:	Dtd		
•	Max. Operating Pressure		
:	Nitrogen Gas		
:	60 minutes		
' INSPF4	TFD	RESULT FOUND O	)K
		MESOLI FOUND (	
nge joints	and other connections	YES / NO	
	MITED I TATION DN : - : - : - : - : - : - : - : - : - : -	MITED       FAT PROCEDUR FOR METE         TATION       Doc. No: MEC.         DN       Rev. 1         TEST CERTIN         PNEUMATIC TES         :       Dtd         :       Max. Operating Pressure         :       Otd         :       Otd         :       Otd         :       Otd         :       Otd         :       Otd         :       Otd	MITED       FAT PROCEDURE / FAT MANUAL FOR METERING SKIDS         TATION       Doc. No: MEC/05/E5/FAT- 030         NN       Rev. 1       Page 11 of 21         TEST CERTIFICATE         PNEUMATIC TEST OF SKID         :          :          :          :          :          :          :       Max. Operating Pressure         :       Nitrogen Gas         :       60 minutes         RESULT FOUND C         age joints and other connections       YES / NO

TPIA

Client /Mecon Representative

MECON LIM DELHI	MECON LIMITED DELHI     FAT PROCEDURE / FAT MANUAL FOR METERING SKIDS       INSTRUMENTATION     Doc. No: MEC/05/E5/FAT- 030			
INSTRUMENTATION SECTION		Rev. 1 Page 12 of 21		00 मेकान 9001:2000 Company
		<u>TEST CERTI</u>	FICATE NSPECTION	I
Project	:			
Client	:			
<b>P.O.</b> No	:	Dtd		
Sr. No	. No Parameters Requirement			<b>Result Found</b>
1.	Base Fram	ne Dimension As per approved		
]	Foundatio	n drawing		YES / NO
2.	Height of	inlet pipe from base in mm		YES / NO
3.	Height of	Outlet pipe from base in mm		YES / NO
4.	Upstream	flow straightener Length in mr	n	YES / NO
5. 6	Down stre	am now straightener Length in		YES / NO
0. 7	Cable and	trave from Metering system to	control room meters	VES / NO
8	Painting th	lickness of pipe line as per app	roved	
	Specificati	ions	10,00	YES / NO
9.	Dimension	n of Control Panel as per Appro	oved Drawing	YES / NO

TPIA

**Client /Mecon Representative** 

MECON DE	LIMITED CLHI	FAT PROCEDURE / FAT MANUAL FOR METERING SKIDS		
INSTRUM	ENTATION	Doc. No: MEC	Doc. No: MEC/05/E5/FAT- 030	
SECTION		Rev. 1	Page 13 of 21	0.9001:2000 Company
		FIELD TARNS	MITTERS	
Project	:			
Client	:			
P.O. No	:	Dtd		

SR. No.	ITEM DESCRIPTION	LOCAL, PAI INDICATION	REMARKS	
1.	Inlet Pressure Transmitter	O Accepted	O Rejected	
2.	Inlet Temp. Transmitter	O Accepted	O Rejected	
3.	Diff Pressure Transmitter- Filter 1	O Accepted	O Rejected	
3.	Diff Pressure Transmitter- Filter 2	O Accepted	O Rejected	
4.	Metering Pressure Transmitter Stream1	O Accepted	O Rejected	
5.	Metering Pressure Transmitter Stream2	O Accepted	O Rejected	
6.	Metering Temperature Transmitter – Stream1	O Accepted	O Rejected	
7.	Metering Temperature Transmitter – Stream 2	O Accepted	O Rejected	
8.	Functioning of HART communication of transmitters	O Accepted	O Rejected	

TPIA

Client /Mecon Representative

MECON LIMITED DELHI		FAT PROCEDU FOR MET	FAT PROCEDURE / FAT MANUAL FOR METERING SKIDS		
INSTRUMENTATION SECTION		Doc. No: ME	Doc. No: MEC/05/E5/FAT- 030		
		<b>Rev. 1</b>	Page 14 of 21	90 9001:2000 Company	
<u>FAT CHECK LIST</u> <u>FLOW COMPUTER</u>					
Project	:				
Client	:				
P.O. No	:	Dtd			

Tag No.: Stream-1

Inspection / Simulation done for	CHECKED/ OBSERVED	REMARKS
Volumetric Flow rate (SCMH)	O Accepted O Rejected	
Totaliser (SCM)	O Accepted O Rejected	
Mass Flow rate (kg/ hr)	O Accepted O Rejected	
Energy Flow rate (Kcal/ hr)	O Accepted O Rejected	
Pipe Line Temperature °C	O Accepted O Rejected	
Pipe Line Pressure Kg/Cm2g / Bar(g)	O Accepted O Rejected	
Atmospheric pressure entry at site condition	O Accepted O Rejected	
Base Pressure/ Base Temperature entry	O Accepted O Rejected	
Configuration check, parameter settings	O Accepted O Rejected	
PID controller Analog O/P check	O Accepted O Rejected	
Functioning of PID loop for FCV by simulation	O Accepted O Rejected	
Interface data of flow computer with RTU – Measurement by Laptop	O Accepted O Rejected	
Various Units of measurement (Flow rate, heating value, energy)	O Accepted O Rejected	
Flow computer o/p to SCADA	O Accepted O Rejected	
GC data input to Flow computer from SCADA	O Accepted O Rejected	
K factors for meter error curves	O Accepted O Rejected	
Audit trail printout (to be attached)	O Accepted O Rejected	
Software for flow calculation verification	O Accepted O Rejected	

**Company Representative** 

TPIA

Client /Mecon Representative

Date

MECON LIMITED DELHI			FAT PROCEDURE / FAT MANUAL FOR METERING SKIDS					
INSTRUM	INSTRUMENTATION		Doc. No: MEC/05/E5/FAT- 030		मेकान			
SECTION		Re	ev. 1	Page 15 of 21	9001:2000 Company			
FAT CHECK LIST         FLOW COMPUTER         Project       :								
Client	:							
P.O. No	:		Dtd					

Tag No.: Stream-2

Inspection / Simulation done for	CHECKED/ OBSERVED	REMARKS
Volumetric Flow rate (SCMH)	O Accepted O Rejected	
Totaliser (SCM)	O Accepted O Rejected	
Mass Flow rate (kg/ hr)	O Accepted O Rejected	
Energy Flow rate (Kcal/ hr)	O Accepted O Rejected	
Pipe Line Temperature °C	O Accepted O Rejected	
Pipe Line Pressure Kg/Cm2g / Bar(g)	O Accepted O Rejected	
Atmospheric pressure entry at site condition	O Accepted O Rejected	
Base Pressure/ Base Temperature entry	O Accepted O Rejected	
Configuration check, parameter settings	O Accepted O Rejected	
PID controller Analog O/P check	O Accepted O Rejected	
Functioning of PID loop for FCV by simulation	O Accepted O Rejected	
Interface data of flow computer with RTU – Measurement by Laptop	O Accepted O Rejected	
Various Units of measurement (Flow rate, heating value, energy)	O Accepted O Rejected	
Flow computer o/p to SCADA by GSM modem.	O Accepted O Rejected	
GC data input to Flow computer from SCADA by GSM modem.	O Accepted O Rejected	
K factors for meter error curves	O Accepted O Rejected	
Audit trail printout (to be attached)	O Accepted O Rejected	
Software for flow calculation verification	O Accepted O Rejected	

**Company Representative** 

TPIA

Client /Mecon Representative

MECON LIMITE DELHI	D FAT PROCEDU FOR MET	JRE / FAT MANUAL TERING SKIDS		
INSTRUMENTATI	ON Doc. No: ME	C/05/E5/FAT- 030		
SECTION	<b>Rev. 1</b>	Page 16 of 21		
	TEST CERT	IFICATE		
	<u>CONTROL</u>	PANEL		
Project :				
Client :				
P.O. No :	Dtd			
Type of Skid :				
POINT INS	PECTED	OBSERVATION/ RESULT FOUND		
1) Wiring conn	ection as per approved wiring diagr	ram Yes / NO		
2) Functional C	heck as per data sheet	Yes/ NO		
3) Hot stand by	functionality of Power supply, Mo	dem Yes/ NO		
4) Door alignm	ent	Yes/ NO		
5) Dimensions	/ Drawings	Yes/ NO		
6) Painting		Yes/ NO		
7) Identification	n mark/ Tagging (Cable, Instrument	ts) Yes/ NO		
8) Cable Dress	ng, Plugging of spare holes	Yes/ NO		
9) Fan, Tube L	ight	Yes/ NO		
10) Adequacy o	f Installed spares for future use	Yes/ NO		
Remarks (if any):				

TPIA

Date:

**Company Representative** 

Client /Mecon Representative

MECON DE	MECON LIMITED FAT PROCEDURE / FAT MANUAL DELHI FOR METERING SKIDS							
INSTRUMENTATION SECTION		Doc. No: MEC/	Doc. No: MEC/05/E5/FAT- 030					
		Rev. 1	Rev. 1 Page 17 of 21					
<u>FAT CHECK LIST</u> ORIFICE FLOW METER								

P.O. No : ----- Dtd. -----

Tag No.: Stream-1

Inspection / Simulation done for	CHECKED/	OBSERVED	REMARKS
Make/ Model/ for Flow meter	O Accepted	O Rejected	
Installation of Orifice Meter in skid	O Accepted	O Rejected	
Calibration report	O Accepted	O Rejected	

Tag No.: Stream- 2

Inspection / Simulation done for	CHECKED/	OBSERVED	REMARKS
Make/ Model/ for Flow meter	O Accepted	O Rejected	
Installation of Orifice Meter in skid	O Accepted	O Rejected	
Calibration report	O Accepted	O Rejected	

**Company Representative** 

TPIA

**Client /Mecon Representative** 

MECON LIMITED DELHI		FAT PROCEDU FOR MET	FAT PROCEDURE / FAT MANUAL FOR METERING SKIDS				
INSTRUMENTATION SECTION		Doc. No: ME	Doc. No: MEC/05/E5/FAT- 030				
		<b>Rev. 1</b>	Page 18 of 21	9001:2000 Company			
		GAS CHROMA	TOGRAPH				
Project :							
Client :							
P.O. No	:	Dtd					

Inspection / Simulation done for	CHECKED/ OBSERVED	REMARKS
MAKE/ Model of GC	O Accepted O Rejected	
GC controller/ remote configurator Installation in control	O Accepted O Rejected	
panel		
Parameter settings/ configuration/ application software	O Accepted O Rejected	
loading, default values etc for GC controller/ remote		
configurator		
Probes, Tools for insertion/ removal	O Accepted O Rejected	
Calibration Gas cylinders, carrier gas cylinders, self	O Accepted O Rejected	
acting pressure regulator for Cylinders (calibration gas/		
carrier gas) with stand, mounting arrangement		
Field Analyser with Stand	O Accepted O Rejected	
Composition of calibration gases	O Accepted O Rejected	
Shed for Analyser, Cable for remote configurator/	O Accepted O Rejected	
controller, communication with various devices,		
instruments, equipments etc		
Communication between GC and Laptop and GC data	O Accepted O Rejected	
simulation		
Simulation of GC data input to Flow computer, Metering	O Accepted O Rejected	
supervisory system and RTU/ SCADA and via GSM		
Functionality/ simulation of communication ports for	O Accepted O Rejected	
communication with flow computers, Laptop, Metering		
supervisory system, RTU/ SCADA, Printers,		
Interface of GC with RTU/SCADA, Flow computers,	O Accepted O Rejected	
Calibration methods. Various Units of measurement	O Asserted O Dejected	
(heating value/ anergy)	O Accepted O Rejected	
Communication with DTU/SCADA	O Assented O Dejected	
CC data input to Flow computer through	O Accepted O Rejected	
	O Accepted O Rejected	
Calibration run for CC	O Accortad O Paicated	
Calibration full for Q consecutive runs	O Accepted O Rejected	
Chromatogram for 5 consecutive runs	O Accepted O Rejected	

Client /Mecon Representative

MECON LIMITED DELHI	FAT PROCEDURE / FA FOR METERING	AT MANUAL SKIDS	
INSTRUMENTATION	Doc. No: MEC/05/E5	/FAT- 030	के मेकान
SECTION	Rev. 1	Page 19 of 21	9001:2000 Comps
	TEST CERTIFICA	TE	
	SOLAR PANE	<u>:L</u>	
Project :			
Client :			
P.O. No :	Dtd		
POINT INSPECTED	OBS	SERVATION/ RESULT F	OUND
1) Functional check of Sc	lar panel in stream-1	Yes / NO	
2) Output Voltage	VDC in stream-1	Yes/ NO	
3) No. of(Ah)	Batteries per stream	Yes/ NO	
4) Functional check of Sc	lar panel in stream-2	Yes / NO	
5) Output Voltage	VDC in stream-2	Yes/ NO	
6) Identification mark/ tag	ging	Yes/ NO	
7) Adequacy of the Solar arrangement, fittings,	r panel system mounting requires cable and other ha	ardware Yes/ NO	
Remarks (if any):			
Company Representative		Client /Mecon Ro	enresentative

MECON LIMITED DELHI		FAT PROCEDU FOR MET	FAT PROCEDURE / FAT MANUAL FOR METERING SKIDS Doc. No: MEC/05/E5/FAT- 030			
INSTRUMENTATION		Doc. No: ME				
		Rev. 1	Page 20 of 21	9001:2000 Company		
Project	:	<u>ACTION</u>	<u>LIST</u>			
Client :						
P.O. No	P.O. No : Dtd					

# Type of Skid :

Sr. No.	Description	Action By
- 1		
<u>l.</u>		
2.		
3.		
4.		
5.		
6.		
7.		
8.		
9.		
10.		
11.		
12.		

**Company Representative** 

TPIA

# Client /Mecon Representative

MECON LIMITED DELHI		FAT PROCEDURE / FA FOR METERING	FAT PROCEDURE / FAT MANUAL FOR METERING SKIDS				
INSTRUMENTATION SECTION		Doc. No: MEC/05/E5/FAT- 030		मेकान			
		Rev. 1 Page 21 of 21		0 9001:2000 Company			
FAT CHECK LIST REVIEW OF DOCUMENTS Project :							
Client :							
P.O. No : Dtd							

SR.	DOCUMENTS REVIEWED	CHECKED	REMARKS
<b>No.</b> 1.	Radiography Test for all items of skid (as per List prepared by vendor and attached as Annexure)	O Accepted O Rejected	
2.	Dye Penetration Test for all items of skid (as per List prepared by vendor and attached as Annexure)	O Accepted O Rejected	
3.	Test Certificate (for Filtration system, PSV, CRV, NRV, Flow profiler, Flow meter, Isolation Ball Valves, Plug valves, Flow computer, PT, DPT, PG, DPG, Thermo-well, RTD, LEL detection system, GC, Control panel, JBs, Pipe spools etc.)	O Accepted O Rejected	
3.	Hydro test for all items of skid (as per List prepared by vendor and attached as Annexure)	O Accepted O Rejected	
4.	Material test Report for all items of skid (as per List prepared by vendor and attached as Annexure)	O Accepted O Rejected	
5.	Certification as per 3.2 for all Pressure Equipments (as per List prepared by vendor and attached as Annexure)	O Accepted O Rejected	
6.,	Calibration Report for PT, DPT, TT, PSV, CRV, Flow Computer, GC, ORIFICE, LEL sensors, LEL detection system, Calibration Gas (for FC & LEL)	O Accepted O Rejected	
7.	Welding Procedure Specification, Welder qualification.	O Accepted O Rejected	
8.	Configuration Sheet for Flow computers, GC	O Accepted O Rejected	
9.	Compliance Certification for Painting of skid including all items (Filtration system, Valves, PSVs, CRVs etc) and control panel	O Accepted O Rejected	
10.	Material Correlation Chart & Welding Joints Correlation Chart	O Accepted O Rejected	

TPIA

Client /Mecon Representative

# **ANNEXURE - VIII**

# **STANDARD SPECIFICATIONS**

# Standard Specifications

1.	Technical specification of Cartridge Filter	MEC/TS/05/62/017, Rev-1
2.	Specification of Assorted Pipe	MEC/TS/05/62/59A, Rev-0
3.	Technical Specification of Pressure Safety Valve	MEC/TS/05/62/056, Rev-1
4.	Specification for Seamless Fittings & Flanges up to 400 mm (16") NB	MEC/TS/05/21/025, Rev-0
5.	Specification for Gaskets, Nuts and Bolts	MEC/S/05/21/19, Rev-0
6.	Specification for Piping, Fabrication and Erection	MEC/S/05/21/06, Rev-0
7.	Specification for Vents, Drains and Wells, Pressure tapping	MEC/S/05/21/15, Rev-0
8.	Specification for Flushing and Testing	MEC/S/05/21/11, Rev-0
9.	Technical Specification for Ball Valve	MEC/TS/05/21/002, Rev-1
10.	Technical Specification for Plug Valve	MEC/TS/05/62/003, Rev-2
11.	Technical Specification for Check Valve	MEC/TS/05/62/004, Rev-2
12.	Specification For Shop & Field Painting	MEC/S/05/21/07, Rev-0
13.	Specification for Installation of Instruments	MEC/S/05/26/01, Rev-0
14.	Specification for Instrument Tubing	MEC/S/05/26/02, Rev-0
15.	Specification for Inlet, Outlet Sections and Flow Straightner	MEC/S/05/26/03, Rev-0
16.	Specification for Instrument Tube Fittings	MEC/S/05/26/04, Rev-0
17.	Specification for Instrument Valves and Manifolds	MEC/S/05/26/05, Rev-0
18.	Specification for Junction Boxes and Cable Glands	MEC/S/05/26/06, Rev-0
19.	Specification for Signal Cable	MEC/S/05/26/07, Rev-0
20.	General Technical Specification for Instrumentation	MEC/S/05/26/08, Rev-0
21.	Specification for Cabling	MEC/S/05/E5/21, Rev-0
22.	Specification for Earthing	MEC/S/05/26/23 A, Rev-0

# TECHNICAL SPECIFICATION FOR CARTRIDGE FILTER & ACCESSORIES (Dry Gas Filters)

# SPECIFICATION NO. : MEC/TS/05/62/017, Rev-2



Page 272 of 644

<u>CONTENTS</u>					
<u>Sl.</u> <u>No.</u>	<b>Description</b>	Page No.			
1.0	GENERAL	2			
2.0	SCOPE OF SUPPLY		2		
3.0	SCOPE OF SERVICES		2		
4.0	DESIGN		3		
5.0	TECHNICAL REQUIR	EMENTS	4		
6.0	INSPECTION AND TE	STING	5		
7.0	PROTECTION AND PA	AINTING	6		
8.0	PACKAGING AND ID	ENTIFICATION	7		
9.0	SPARE PARTS		8		
10.0	INFORMATION/ DOC TO BE SUBMITTED V	8			
11.0	INFORMATION/ DOCUMENTS/ DRAWINGS 9 TO BE SUBMITTED BY SUCCESSFUL VENDOR				
12.0	ORDER OF PRECEDE	NCE	10		
REVISION	PREPARED BY	CHECKED BY	APPROVED BY		

# 1.0 **GENERAL**

1.1 This specification outlines the minimum requirements under which the manufacturer shall design, manufacture, test and supply Cartridge Filter for separating solid contaminants from the gas stream.

# 2.0 SCOPE OF SUPPLY

2.1 The vendor's scope of supply shall be Cartridge Filter alongwith accessories complete in all respects as per data sheet enclosed. The detailed scope of work is as follows :

The filters shall comprise of the following :

- Filter vessel alongwith necessary filtering cartridge.
- Necessary nozzles, companion flanges, blind flanges (with nuts, bolts, gaskets etc.) on the filter vessel as specified in the data sheets.
- Cartridge supporting arrangement inside the vessel.
- Quick opening closure for maintenance and filter element replacement.
- Instruments, viz. PSV as per Instrumentation inputs.
- Necessary supports for filter vessel.
- Cleats for platforms and ladders.
- Foundation bolts & base plate for embedding/ grouting into civil foundation.
- Documents clause 10, 11 & elsewhere in spec.
- Spare parts for two years normal operation.

# 3.0 SCOPE OF SERVICES

• Engineering, design and manufacturing.

- Procurement of raw materials etc. from sub-vendors.
- Preparation and submission of documentation for design approval by purchaser/ consultant.
- Inspection and testing as per T.S.
- Surface preparation, protective coating and painting as per T.S.
- Packaging for transportation to site and supply.

# 4.0 **DESIGN**

4.1 Following codes & standards (latest edition) shall be followed for design, manufacture, testing etc. of the equipment.

ASME Sec-VIII Div-1	:	Boiler and Pressure Vessel Code
ASME Sec-IX	:	Welding and Brazing Qualifications
ASME Sec-II & ASTM	:	Material Specifications
ANSI B16.5	:	Pipe Flanges & Pipe Fittings
ANSI B16.1	:	Forged Steel Fittings Socket Welded & Threaded
ASME B16.47	:	Large Diameter Steel Flanges
ANSI B36.10	:	Welding & Seamless Wrought Steel Pipe.

4.2 For purpose of material selection national code of the country of origin shall also be acceptable provided the vendor specifically establishes, to the satisfaction of the purchaser, the equivalence or superiority of the proposed material with respect to those specified.

# 5.0 **TECHNICAL REQUIREMENTS**

5.1	The cartridge shall be from approved make.
5.2	Filter elements must withstand a pressure of 1.0 kg/cm <sup>2</sup> (g) without breaking or failure.
5.3	At least 300mm space from bottom tangent line to be provided.
5.4	Filtering efficiency shall be as per enclosed data sheet.
5.5	Particle size shall be as indicated in the data sheets.
5.6	The end closure to be high pressure quick release type interlock to open only when vessel is completely depressurised.
5.7	Filter element should be suitable for specified mole% of CO <sub>2</sub> .
5.8	Core of filter element shall be of SS material.
5.9	A davit/ hinged arrangement shall be provided for the closure for convenient handling. The closure shall have perfect sealing arrangement to prevent leakage.
5.10	The equipment shall be of the type as mentioned in the data sheets and shall meet the duty requirements and performance parameters as mentioned therein.
5.11	Vendor shall submit calculations for sizing of the equipment together with all supporting documents/ catalogues/ nomographs etc. with the bid. The type, model and number of cartridge shall be selected based on allowable pressure drop and supplier's recommendation. The total internal cross sectional area of mounted cartridge shall not be less than inlet nozzle area for inlet size upto 150 NB. The calculation for the selected number of cartridge shall be furnished, alongwith the bid.

5.12	Suitable baffle plates shall be provided in the vessels for proper fluid flow distribution. Vessel diameter shall be minimum twice the diameter of inlet nozzle. All internal nuts and bolts shall be of stainless steel irrespective of material of construction of vessel.
5.13	All nozzles/ pipes on the vessel shall be of seamless construction. All nozzle less than or equal to 50 NB size shall be provided with 2 Nos., 6mm thick stiffeners at 90 degress to each other. All nozzles above 80 NB size, shall be provided with reinforcement pads.
5.14	All flanges shall be WNRF except DPT connection which is to be socket welded.
5.15	Dimensions of flanges including shell flanges, blind head cover flanges, nozzle flanges and blind flanges shall be as per ANSI B16.5. Larger flanges shall be as per ANSI B16.47.
5.16	Pressure parts joined by butt welds shall be with full penetration welds. Where both sides welding is not accessible, root run by tungsten inert gas process or backing strip, shall be used to ensure full penetration. Backing strip if used, shall be removed after welding.
5.17	Vessels shall be post weld heat treated, whenever it is required due to service requirement or due to code requirements. Vessels shall be post weld heat treated as a complete unit and no welding shall be permitted after the post weld heat treatment is completed.
5.18	For vessels in stainless steel construction, lower allowable stress values shall be considered as per ASME Code for their design.
5.19	Filter vessel shall be provided with lifting and earthing lugs. Fire proofing and insulation supports shall be provided if indicated in data sheet.
6.0	INSPECTION AND TESTING
6.1	Equipment shall be subjected to stage wise expediting, inspection and testing at vendor's/ sub-vendor's works by purchaser/ its authorised inspection agency. Vendor shall submit Quality Assurance (QA) procedures before commencement of fabrication. Approved QA procedures shall form the basis for equipment inspection.

6.2		Testing at vendor's works shall include but not limited to the following:			
		Non destructiv	ve tests such as	radiogra	aphy, dye penetration tests.
		Hydrostatic te	st at 150% of de	esign pr	essure for the vessel.
		• Any other test	s as per data sh	eets/ sta	ndards/ codes.
6.3		Any or all the tests, at purchaser's option, shall be witnessed by purchaser/ its authorised inspection agency. However, such inspection shall be regarded as check-up and in no way			
		absolve the venuor of	uns responsion	ity.	
6.4		Extent of radiography	shall be 100%.		
7.0		PROTECTION A	AND PAINT	ING	
7.1		All exposed carbon st outside to remove sca blasting as applicable be St. 3 and incase 0055900.	teel parts to be ale, rust, dirt an . Minimum acc of blast clean	painted d other eptable ing sha	shall be thoroughly cleaned from inside and foreign materials by wire brushing and sand standard in case of power tool cleaning shall Il be Sa 2 <sup>1</sup> / <sub>2</sub> as per Swedish Standard SIS
7.2		Non-ferrous materials, austenitic stainless steels, plastic or plastic coated materials, insulated surfaces of equipment and pre-painted items shall not be painted.			
7.3		Stainless steel surfaces both inside and outside shall be pickled and passivated.			
7.4		Machined and bearing surfaces shall be protected with varnish or thick coat of grease.			
7.5		Depending on the environment, following primer and finish coats shall be applied.			
		Environment	Description		
	i)	Normal Industrial	Surface Preparation	:	Sa 21/2
			Primer	:	2 coats of Redoxide zinc chromate each 25 microns (min.) thick.

			Finish Coat	:	2 coats of synthetic enamel, each 25 microns (min.) thick.
	ii)	Corrosive Industrial	Surface Preparation	:	Sa 21/2
			Primer	:	2 coats of Epoxy zinc chromate each 35 microns (min.) thick.
			Finish Coat	:	2 coats of Epoxy high build paint each 100 microns (min.) thick.
	iii)	Coastal and Marine	Surface Preparation	:	Sa 21/2
			Primer	:	2 coats of high build Chlorinated Rubber zinc phosphate, each 50 microns (min.) thick.
			Finish	:	2 coats of chlorinated rubber paint, each 35 microns (min.) thick.
	iv)	All Environment (temp. 80-400°C)	Surface Preparation	:	Sa 21/2
			Finish	:	2 coats of heat resistant aluminium paint suitable for specified temp. each 20 $\mu$ thick. (All values refer to dry film thickness).
7.6		The colour of finish c	coat shall be inti	imated t	o vendor after placement of order.

# 8.0 PACKAGING AND IDENTIFICATION

8.1 All packaging shall be done in such a manner as to reduce the volume. The equipment shall be dismantled into major components suitable for shipment and shall be properly packed to provide adequate protection during shipment. All assemblies shall be properly match marked for site erection.

8.2	Attachments, spares parts of the equipment and small items shall be packed separately in wooden-cases. Each item shall be appropriately tagged with identification of main equipment, its denomination and reference number of the respective assembly drawing.
8.3	Detailed packing list in water-proof envelope shall be inserted in the package together with equipment.
8.4	Each equipment shall have an identification plate giving salient equipment data, make, year of manufacture, equipment number, name of manufacturer etc.
9.0	SPARE PARTS
9.1	Vendor shall submit his recommended list of spare parts for two years with recommended quantities and operation of the equipment. Proper coding and referencing of spare parts shall be done so that later identification with appropriate equipment will be facilitated.
9.2	Recommended spares and their quantities should take into account related factors of equipment reliability, effect of equipment downtime upon production or safety, parts and availability of vendor's service facilities around proposed location of equipment.
9.3	Vendor shall also submit a list of recommended commissioning spares with quantities.
10.0	INFORMATIONS/ DOCUMENTS/ DRAWINGS TO BE SUBMITTED WITH THE OFFER
	Contractor shall submit with the offer four copies each of the following:
10.1	Manufacturer's complete descriptive and illustrative catalogue/ literature.
10.2	The completion schedule activity wise.
10.3	In case of failure to submit the documents listed above, the offer may be rejected.

# Information/ documents/ drawings to be submitted By successful vendor

Successful vendor shall submit six copies unless noted otherwise, each of the following :

- 11.1 Inspection & test reports for all mandatory tests as per the applicable code. Test reports for any supplementary tests, in nicely bound volumes.
- 11.2 Material test certificates (physical) property, chemical composition, make, heat treatment report etc.) as applicable for items in nicely bound volumes.
- 11.3 Statutory test certificates, as applicable.
- 11.4 Filled in QAP for Owner's/ Consultants approval. These QAP's shall be submitted in four copies.
- 11.5 WPS & PQR, as required.
- 11.6 Within two(2) weeks of placement of order, the detailed fabrication drawings alongwith mechanical design calculations for Owner's/ Consultants approval. These drawings shall be submitted in four copies.
- 11.7 Detailed completion schedule activity wise, within one week of placement of order.
- 11.8 Weekly & fortnightly progress reports for all activities including procurement.
- 11.9 Purchase orders of bought out items soon after placement of order.
- 11.10 Manufacturer's drawings for bought out items, in 4 copies, for Owner's/ Consultant approval within 4 weeks.
- 11.11 Manufacturer related information for design of civil foundation & other matching items within 6 weeks of LOI.

11.12 All approved drawings/ documents as well as inspection and test reports for Owner's/ Consultants reference/ record in nicely category wise bound volumes separately.

Note: All drawings, instructions, catalogues, etc. shall be in English language and all dimensions shall be metric units.

# 12.0 **ORDER OF PRECEDENCE**

The following order of precedence shall govern in interpretation of various requirements and data :

- Data Sheets
- This Specification
- Codes & Standards
- Vendors Standards

# TECHNICAL SPECIFICATION FOR SCRUBBER WITH ACCESSORIES

# SPECIFICATION NO .: MEC/TS/05/62/019



MECON LIMITED Delhi	PROCESS & PIPING DESIGN SECTION DELHI	TECH	NICAL SPECIFICATION SCRUBBER WITH ACCESSORIES	Reference of the second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second
TECHNICAL SPECIFICATION NO. : MEC/TS/05/62/019			REV-0	Page 1 of 13

<u>C O N T E N T S</u>					
<u>SI.No.</u>	<b>Description</b>				
1.0	GENERAL		2		
2.0	SCOPE OF SU	PPLY	2		
3.0	SCOPE OF SE	RVICES	3		
4.0	DESIGN		4		
5.0	MATERIALS		6		
6.0	TECHNICAL R	EQUIREMENTS	7		
7.0	INSPECTION A	AND TESTING	9		
8.0	PROTECTION	AND PAINTING	12		
9.0	PACKAGING A	ND IDENTIFICATION	13		
10.0	SPARE PARTS		14		
11.0	INFORMATION/ DOCUMENTS/ DRAWINGS TO BE SUBMITTED WITH THE OFFER.		14		
12.0	INFORMATION/ DOCUMENTS/ DRAWINGS TO BE SUBMITTED BY SUCCESSFUL VENDOR		14		
PREPARED BY		CHECKED BY	APPROVED BY		

<b>MECON LIMITED</b> Delhi	PROCESS & PIPING DESIGN SECTION DELHI	TECH	NICAL SPECIFICATION SCRUBBER WITH ACCESSORIES	Авіт 4001 Сигорії
TECHNICAL SPECIFICATION NO. : MEC/TS/05/62/019			REV-0	Page 2 of 13

# 1.0 **GENERAL**

1.1 This specification outlines the minimum requirements under which the manufacturer shall design, manufacture, test and supply Scrubbers alongwith accessories including pressure safety valve, level gauges for separating entrained liquid & solid contaminants from the gas stream.

# 1.2 Order of Precedence

The following order of precedence shall govern in interpretation of various requirement and data.

- Data sheets.
- This specification.
- Codes & standards.
- Vendors standards.

# 2.0 SCOPE OF SUPPLY

- 2.1 The vendor's scope of supply shall be as per Data Sheet & Drawing enclosed. The detailed scope of supply is as follows :-
  - Scrubbers complete in all respect.
  - Companion flanges with bolting and gaskets for all nozzles.
  - Cover flanges for manholes, handholes and inspection openings etc. with bolting and gaskets.
  - Test blind flanges in addition to companion flanges.
  - All internals complete with supports and fasteners.
  - Tower davits.
  - Manhole davits.
  - All external welded attachments like platforms and ladder cleats, insulation supports if any and pipe support cleats etc.

<b>MECON LIMITED</b> Delhi		PROCESS & PIPING DESIGN SECTION DELHI	TECHNICAL SPECIFICATION SCRUBBER WITH ACCESSORIES		10 5001 Carp 5"		
TECHNICAL SPECIFICATION NO. : MEC/TS/05/62/019			REV-0	Page 3 of 13			
<ul> <li>Lifting lugs/ erection lugs.</li> <li>Template for skirt supports of equipment; template shall be supplied within 2 months from the date of Fax of Intent.</li> <li>Sight glasses and protector glasses.</li> <li>Cleats for earthing connections.</li> <li>Name plate with bracket.</li> <li>GA drawings for approval shall be submitted with in one month from FOI.</li> <li>Foundation/ Anchor/ Holding down bolts with nuts and washers. Shall be supplied within one month from FOI.</li> </ul>							
•	Lapping tool with transportation ring etc.						
•	Welding/ wire rods for weld seal gaskets.						
•	Special tools/ tackles.						
•	Attachment for painter's trolley.						

- Documents as called in "Vendor Data Requirement Sheet".
- Spare parts for two years normal operation.
- Pressure Safety Valve and Level Gauge complete in all respect.

# 3.0 SCOPE OF SERVICES

- Engineering, design and manufacturing.
- Procurement of raw materials etc. from sub-vendors.
- Preparation and submission of documentation & GA drawings for design approval by purchaser/ consultant.
  - QAP
- Inspection and testing as per T.S.

MECON LII Delhi	MITED PI	ROCESS & PIPING DESIGN SECTION DELHI	TECHNICAL SPECIFICATION SCRUBBER WITH ACCESSORIES	मित्र मेल्सन मित्र 2001 Carp In						
TECHNICAL SPECIFICATION NO. : MEC/TS/05/62/019			REV-0	Page 4 of 13						
	Surface prep     Packing for	paration, protective transportation to sit	coating and painting as per -	T.S.						
4.0	DESIGN	DESIGN								
4.1	Complete design of responsibility of the specification, this specification, the specification of the specification of the specification of the specification of the specification of the specification of the specification of the specification of the specification of the specification of the specification of the specification of the specification of the specification of the specification of the specification of the specification of the specification of the specification of the specification of the specification of the specification of the specification of the specification of the specification of the specification of the specification of the specification of the specification of the specification of the specification of the specification of the specification of the specification of the specification of the specification of the specification of the specification of the specification of the specification of the specification of the specification of the specification of the specification of the specification of the specification of the specification of the specification of the specification of the specification of the specification of the specification of the specification of the specification of the specification of the specification of the specification of the specification of the specification of the specification of the specification of the specification of the specification of the specification of the specification of the specification of the specification of the specification of the specification of the specification of the specification of the specification of the specification of the specification of the specification of the specification of the specification of the specification of the specification of the specification of the specification of the specification of the specification of the specification of the specification of the specification of the specification of the specification of the specification of the specification of the specification of the specification of the specification of the specification	Complete design of equipment as per latest code of construction shall be the responsibility of the supplier. Strict compliance with the requirements of code/equipment specification, this specification and any other referred document shall be fully ensured.								
4.2	Following codes & standards (latest edition) shall be followed for design, manufacture, testing etc. of the equipment.									
	ASME Sec-VIII Div-7	I : Boiler ar	nd Pressure Vessel Code							
	ASME Sec-IX	: Welding	and Brazing Qualifications							
	ASME Sec-II & ASTI	M : Material	Specifications							
	ANSI B16.5	: Pipe Flai	nges & Pipe Fittings							
	ANSI B16.11	: Forged S	Steel Fittings Socket Welded a	& Threaded						
	ASME B16.47	: Large Di	ameter Steel Flanges							
	ANSI B36.10	: Welding	& Seamless Wrought Steel P	Pipe.						
4.3	Detailed design calc and submitted to F approval from Purc final. Any change/ c Code/enquiry docu incorporated. For hi lower allowable stre	ulations for adequa Purchaser/ Consultant haser/ Consultant comment, made on ments, without a gh alloy steel parts esses shall be adop	acy of strength of all individua ant for comments and appro- the design of equipment sh design calculations to satisfy ny commercial implications /components designed as pe ted, unless otherwise specifie	al parts shall be made oval. After receipt of hall be considered as y the requirements of s, shall have to be r ASME Sec VIII Div.I ed. After the approval						

4.4 Any modification in design parameters as may be required by Purchaser/ Consultant after placement of order/ at any time shall have to be suitably incorporated. Commercial implications, if any, due to such changes shall be mutually discussed and settled.

of various documents subsequent deviations shall be discouraged.

4.5 Wind and seismic analysis shall be in accordance with IS: 875 and IS: 1893 unless otherwise stated in specifications.

MECON LIMITED Delhi	PROCESS & PIPING DESIGN SECTION DELHI	TECH	NICAL SPECIFICATION SCRUBBER WITH ACCESSORIES	Asir Julian Carl
TECHNICAL SPECIFICATION NO. : MEC/TS/05/62/019			REV-0	Page 5 of 13

Detailed vibration analysis for wind forces and seismic forces shall be submitted alongwith the design calculations. For the design consideration, wind forces shall be increased by 20% to cater for the effect of piping system, platforms and ladders etc.

- 4.6 Design of supports and anchor/foundation bolts shall be the responsibility of the supplier. The type, number, location and any other information if furnished in the specification sheets shall be strictly adhered to. However, adequacy of the same must be checked by the supplier. In no case diameter of anchor bolts for skirt support shall be less than M 24 and for other M 16.
- 4.7 Forces & moments coming on the nozzles shall be furnished after placement of order by Purchaser/ Consultant which shall be considered in the design by the supplier, without any commercial implication. Alternatively supplier shall indicate maximum permissible forces & moments for review and confirmation by Purchaser/ Consultant.
- 4.8 Unless otherwise specified, removable internal parts shall be designed in units as large as can be installed through the nearest/upper manhole or opening, Demisters, may however, be installed through the nearest lower manhole or opening.

Trays, distributors, baffles and support beams shall be designed in such a way that deformation of the shell due to operating pressure and thermal expansion does not occur.

Bolts and nuts for fixing internals shall be 18/8 SS and minimum size of bolt shall be M10.

- 4.9 Support for packing/internals shall be designed for the worst condition. In case of packing liquid hold-up of minimum 20% of packing volume shall be considered unless otherwise specified.
- 4.10 Unless otherwise specified minimum corrosion allowance for CS and low alloy steel shall be 1.5 mm or as specified in data sheet.
- 4.11 The lifting lugs shall be designed with a shock factor of 2.

# 5.0 **MATERIALS**

5.1 For purpose of material selection national code of the country of origin shall also be acceptable provided the vendor specifically establishes, to the satisfaction of the purchaser, the equivalence or superiority of the proposed material with respect to those specified.
MECON LIMITED Delhi	PROCESS & PIPING DESIGN SECTION DELHI	TECH	NICAL SPECIFICATION SCRUBBER WITH ACCESSORIES	
TECHNICAL SPECIFICATION NO. : MEC/TS/05/62/019		REV-0	Page 6 of 13	

5.2 All materials, whatsoever, required to complete the supplies shall be procured by the supplier and all such materials shall be covered with due identifiable material test certificates. 5.3 Materials indicated in specification sheets are recommended for the required services conditions. Supplier, however, may use better or equivalent material with prior approval of Purchaser/ Consultant. Details of such materials must be indicated in the offer with proper reference. 5.4 Unless otherwise specified all C.S. material used for pressure parts shall be fully killed. 5.5 Unless otherwise specified copper or copper alloys shall not be used. Steel conditions with copper content upto 0.35% maximum are acceptable. 5.6 Materials for low temperature service shall be impact tested (Charpy) at the lowest design temperature in accordance with requirements of code/ specification. 5.7 For coarse grained and high tensile materials in carbon steel (UTS > 45 kg/mm<sup>2</sup>) and low alloy steel, guarantee impact strength shall be ensured at a temperature 15°C below envisaged hydraulic test temperature as precaution against brittle fracture during hydraulic test. 5.8 C.S. and low alloy steel exceeding thickness of 50mm shall be vacuum degassed except for plate ring flanges. 5.9 For AISI 316 and 316L material, ferrite content for plates, pipes and tubes shall be maximum 0.6% and for forging maximum 1%. 5.10 Intergranular corrosion test shall be carried out on all S.S. materials as per ASTM A 262 practice C if specified. The corrosion rate shall not exceed 0.6 mm/year. 5.11 All plate material over 50mm thickness shall be ultrasonically tested both on surfaces and edges as per ASTM A 435. 5.12 All forgings shall be ultrasonically tested as per ASTM A 388 for thickness greater than 100 mm with the criteria shown in ASME VIII Div. 2 para AM-203-2. In case any defect is found no repair by welding shall be allowed. ANSI flanges, for equipment designed as per ASME SCE VIII Div. I need not be ultrasonically tested. 5.13 For IS 2062 materials without mill test certificates check test, if called for by the inspector, shall be performed without any commercial implication.

TECHNICAL SPECIFICATION NO. : MEC/TS/05/62/019

REV-0

Page 7 of 13

### 6.0 **TECHNICAL REQUIREMENTS**

- 6.1 The equipment shall be procured from a source possessing proven track record for trouble free and reliable operation.
- 6.2 At least 300mm space from bottom tangent line to be provided unless otherwise stated.
- 6.3 Scrubber efficiency shall be 98% minimum
- 6.4 Particle size shall be as indicated in the data sheets.
- 6.5 Internal should be suitable for specified mol% of  $CO_2$  and gas composition, as given in data sheets.
- 6.6 Internal shall be of SS material.
- 6.7 The equipment shall be of the type as mentioned in the data sheets and shall meet the duty requirements and performance parameters as mentioned therein.
- 6.8 Vendor shall submit calculations for sizing of the equipment together with all supporting documents/ catalogues/ nomographs etc. with the bid. The type, model and number of cartridge/ cyclone shall be selected based on allowable pressure drop and supplier's recommendation. The total internal cross sectional area of cyclone shall not be less than inlet nozzle area for inlet size upto 150 NB. The calculation for the selected number of cyclone shall be furnished, alongwith the bid.
- 6.9 Suitable baffle plates shall be provided in the vessels for proper fluid flow distribution. Vessel diameter shall be minimum twice the diameter of inlet nozzle. All internal nuts and bolts shall be of stainless steel irrespective of material of construction of vessel.
- 6.10 All nozzles/ pipes on the vessel shall be of seamless construction. All nozzle less than or equal to 50 NB size shall be provided with 2 Nos., 6mm thick stiffeners at 90 degress to each other. All nozzles above 80 NB size, shall be provided with reinforcement pads.
- 6.11 All flanges shall be WNRF.
- 6.12 Dimensions of flanges including shell flanges, blind head cover flanges, nozzle flanges and blind flanges shall be as per ANSI B16.5. Larger flanges shall be as per ANSI B16.47.
- 6.13 Pressure parts joined by butt welds shall be with full penetration welds. Where both sides welding is not accessible, root run by tungsten inert gas process or backing strip, shall be used to ensure full penetration. Backing strip if used, shall be removed after welding.

MECON LIMITED Delhi	PROCESS & PIPING DESIGN SECTION DELHI	TECH	NICAL SPECIFICATION SCRUBBER WITH ACCESSORIES	र संबर्धन 2001 COLP11
TECHNICAL SPECIFICATION NO. : MEC/TS/05/62/019		REV-0	Page 8 of 13	

- 6.14 Vessels shall be post weld heat treated, whenever it is required due to service requirement or due to code requirements. Vessels shall be post weld heat treated as a complete unit and no welding shall be permitted after the post weld heat treatment is completed.
- 6.15 For vessels in stainless steel construction, lower allowable stress values shall be considered as per ASME Code for their design.
- 6.16 Vessel shall be provided with lifting and earthing lugs. Fire proofing and insulation supports shall be provided if indicated in data sheet.

# 7.0 **INSPECTION AND TESTING**

- 7.1 Equipment shall be subjected to stagewise expediting, inspection and testing at vendor's sub-vendor's works by purchaser/ its authorised inspection agency. Vendor shall submit Quality Assurance (QA) procedures before commencement of fabrication. Approved QA procedures shall form the basis for equipment inspection.
- 7.2 Testing at vendor's works shall include but not limited to the following:
  - Non destructive tests such as radiography, dye penetration tests.
  - Hydrostatic test at 150% of design pressure for the vessel.
  - Any other tests as per data sheets/ standards/ codes.
- 7.3 All raw materials shall be inspected at source and test certificates to enable proper identification shall be submitted.
- 7.4 All equipment shall be inspected during various stages of manufacture starting from identification of raw materials to completion. The equipment shall be considered acceptable for despatch only after final certification for acceptance is issued by the inspector.
- 7.5 Bought out items or items sub-contracted to other sub-suppliers shall also be inspected at the sub-supplier's works.
- 7.6 Inspection by third party, if specified, shall be arranged by the supplier at free of cost. It shall be responsibility of the supplier to make available to the inspector all the new/revised drawings, calculations and other enquiry documents.
- 7.7 Inspection order on third party shall also include specific instructions for marking copies of all correspondence from inspecting authorities to purchaser/ consultant and reporting monthly progress of the order to purchaser/consultant Complete responsibility of getting approval of drawings/ calculations and documents from inspecting authority shall be that of the supplier.

MECON LIMITED Delhi	PROCESS & PIPING DESIGN SECTION DELHI	TECH	NICAL SPECIFICATION SCRUBBER WITH ACCESSORIES	
TECHNICAL SPECIFICATION NO. : MEC/TS/05/62/019		REV-0	Page 9 of 13	

- 7.8 In case of site fabricated/ assembled equipment same inspection agency shall be responsible for inspection, testing at site.
  7.9 In case of castable refractory, 3 Nos. slab of each size shall be cast at site before application and tested to meet the refractory specification in the presence of the
- 7.10 Unless otherwise stated gaskets used during testing shall be same as specified for operating conditions. After testing, gaskets used during testing shall be replaced by new gaskets.
- 7.11 The following NDT requirements are mandatory in addition to the requirements of code/ specifications.

### a) Ultrasonic Examinations

inspector.

- i) Butt weld in thickness  $\geq$  50mm as supplement to radiography.
- ii) Full penetration welds of nozzle attachments on equipments shell/head of thickness  $\geq$  50mm as substitute to radiography.

# b) Magnetic particle/ liquid penetrant examination

- i) All edges of plates and openings in shell of C.S. having thickness over 50mm and low alloy steel/ S.S. having thickness over 25mm.
- ii) Root-run and final layer of all butt welds.
- iii) Fillet welds of 3½% nickel and S.S.
- iv) Each layer of weld deposit in case of S.S. overlay.
- v) Knuckle surface of dished ends/ toriconical Sections and pipe bends.
- vi) Skirt to head joint.
- vii) In case of heat treated equipment final examination as stated above for all weld surfaces shall be carried out after heat treatment.

# c) Radiography

i) Radiography when called for shall be applicable to all pressure welds i.e. longitudinal and circumferentials.

MECON LIMITED Delhi	PROCESS & PIPING DESIGN SECTION DELHI	TECH	NICAL SPECIFICATION SCRUBBER WITH ACCESSORIES	Astr Burger	
TECHNICAL SPECIFICATION NO. :	MEC/TS/05/62/019		REV-0	Page 10 of 13	

- ii) When formed heads are made of welded plates/ petal construction all the weld seams prior to forming and after forming shall be fully radiographed.
  - iii) All the weld T joints shall be radiographed.
- iv) Radiography examination of welds in Cr-Mo and Cr-Mo steel shall preferably be carried out after heat treatment. If radopgrahy is carried out prior to heat treatment the welding and adjacent areas of base metal shall be examined by MP/DP examination after heat treatment.
- 7.14 All completed equipment shall be tested hydrostatically as per the requirements of specification/ codes in presence of the inspecting authority. Pneumatic test of completed equipment shall be carried out only when specially mentioned in the specification sheets. Water used for testing of S.S. equipment shall not have a chloride content exceeding 30 ppm, C.S. 30 ppm.
- 7.15 When required as per specifications/ code, strain gauge measurements shall be carried out on outside circumstance during hydraulic testing. The results shall be plotted both during pressurising and depressurising and procedure of such strain measurements shall have prior approval of consultant.
- 7.16 Any or all the tests, at purchaser's option, shall be witnessed by purchaser/ its authorised inspection agency. However, such inspection shall be regarded as check-up and in no way absolve the vendor of this responsibility.

### 8.0 **PROTECTION AND PAINTING**

- 8.1 All exposed carbon steel parts to be painted shall be thoroughly cleaned from inside and outside to remove scale, rust, dirt and other foreign materials by wire brushing and sand blasting as applicable. Minimum acceptable standard in case of power tool cleaning shall be St. 3 and incase of blast cleaning shall be Sa 2½ as per Swedish Standard SIS 0055900.
- 8.2 Non-ferrous materials, austenitic stainless steels, plastic or plastic coated materials, insulated surfaces of equipment and pre-painted items shall not be painted.
- 8.3 Stainless steel surfaces both inside and outside shall be pickled and passivated.
- 8.4 Machined and bearing surfaces shall be protected with varnish or thick coat of grease.

MECON LIMITED Delhi	D	PROCESS & PIPING DESIGN SECTION DELHI		TEC	HNICAL SPECIFICATION SCRUBBER WITH ACCESSORIES	ABT COFF
TECHNICAL SPECI	IFICATION NO. : I	MEC/TS/C	05/62/019		REV-0	Page 11 of 13
8.5	Depending on	the env	vironment, follov	wing prir	mer and finish coats sha	all be applied.
	Environment		<b>Description</b>			
i)	Normal Indust	trial	Surface Preparation	:	Sa 21⁄2	
			Primer	:	2 coats of Redoxide 25 microns (min.) thic	zinc chromate each k.
			Finish Coat	:	2 coats of synthetic microns (min.) thick.	enamel, each 25
ii)	Corrosive Indu	ustrial	Surface Preparation	:	Sa 21⁄2	
			Primer	:	2 coats of Epoxy zind microns (min.) thick.	c chromate each 35
			Finish Coat	:	2 coats of Epoxy high microns (min.) thick.	build paint each 100
iii)	Coastal and M	larine	Surface Preparation	:	Sa 21⁄2	
			Primer	:	2 coats of high build zinc phosphate, each thick.	Chlorinated Rubber 50 microns (min.)
			Finish	:	2 coats of chlorinated 35 microns (min.) thic	l rubber paint, each k.
iv)	All Environme (temp. 80-400	nt )°C)	Surface Preparation	:	Sa 21⁄2	
			Finish	:	2 coats of heat resist suitable for specified thick. (All values thickness).	ant aluminium paint temp. each 20 μ refer to dry film

8.6 The colour of finish coat shall be intimated to vendor after placement of order.

# 9.0 PACKAGING AND IDENTIFICATION

- 9.1 All packaging shall be done in such a manner as to reduce the volume. The equipment shall be dismantled into major components suitable for shipment and shall be properly packed to provide adequate protection during shipment. All assemblies shall be properly match marked for site erection.
- 9.2 Attachments, spares parts of the equipment and small items shall be packed separately in wooden-cases. Each item shall be appropriately tagged with identification of main equipment, its denomination and reference number of the respective assembly drawing.
- 9.3 Detailed packing list in water-proof envelope shall be inserted in the package together with equipment.
- 9.4 Each equipment shall have an identification plate giving salient equipment data, make, year of manufacture, equipment number, name of manufacturer etc.

### 10.0 **SPARE PARTS**

- 10.1 Vendor shall submit his recommended list of spare parts with recommended quantities and <u>itemised prices</u> for first two years of operation of the equipment. Proper coding and referencing of spare parts shall be done so that later identification with appropriate equipment will be facilitated.
- 10.2 Recommended spares and their quantities should take into account related factors of equipment reliability, effect of equipment downtime upon production or safety, cost of parts and availability of vendor's service facilities around proposed location of equipment.
- 10.3 Vendor shall also submit a list of recommended commissioning spares with quantities and the itemised prices.

# 11.0 INFORMATIONS/ DOCUMENTS/ DRAWINGS TO BE SUBMITTED WITH THE OFFER

Contractor shall submit with the offer four copies each of the following:

- 11.1 Enclosed data sheet duly filled in.
- 11.2 Manufacturer's complete descriptive and illustrative catalogue/ literature.
- 11.3 The completion schedule activity wise.

MECON LIMIT Delhi	ED	PROCESS & PIPING DESIGN SECTION DELHI	TECH	INICAL SPECIFICATION SCRUBBER WITH ACCESSORIES	A STATE OF THE STATE
TECHNICAL SPE	CIFICATION NO. :	MEC/TS/05/62/019		REV-0	Page 13 of 13
11.4	Basic Design (	Calculation			
11.5	In case of fail	ure to submit the docume	ents list	ed above, the offer ma	ay be rejected.
12.0	INFORMATI SUCCESSFU	<u>on/ Documents/</u> L vendor	DRA	WINGS TO BE	SUBMITTED BY
	Successful Co following :	ontractor shall submit s	ix copi	es unless noted othe	erwise, each of the
12.1	Inspection & for any supple	test reports for all manda ementary tests, in nicely b	tory tes ound v	sts as per the applicab olumes.	le code. Test reports
12.2	Material test or report etc.) as	ertificates (physical) prop applicable for items in n	erty, cl icely bo	nemical composition, n bund volumes.	nake, heat treatment
12.3	Statutory test certificates, as applicable.				
12.4	Filled in QAP for Owner's/ Consultants approval. These QAP's shall be submitted in four copies with in 15 days from LOI.				
12.5	WPS & PQR, a	as required.			
12.6	Within two(2) weeks of placement of order, the detailed fabrication drawings alongwith design calculations for Owner's/ Consultants approval. These drawings shall be submitted in four copies.				
12.7	Detailed comp order.	pletion schedule activity v	vise, (B	ar Chart) within one v	veek of placement of
12.8	Weekly & fort	nightly progress reports f	or all a	ctivities including procu	urement.
12.9	Purchase orde	ers of bought out items so	on afte	er placement of order.	
12.10	Manufacturer's drawings for bought out items, in 4 copies, for Owner's/ Consultant approval within 4 weeks.				
12.11	Manufacturer related information for design of civil foundation & other matching items within 6 weeks of LOI.				
12.12	All approved drawings/ documents as well as inspection and test reports for Owner's/ Consultants reference/ record in nicely category wise bound volumes separately.				
	<u>Note</u> : All dra dimen	awings, instructions, cata sions shall be metric units	logues, s.	etc. shall be in Engl	ish language and all

# PROCESS & PIPING DESIGN SECTION MECON LIMITED DELHI 110 092



# TECHNICAL SPECIFICATION FOR ASSORTED PIPES

# SPECIFICATION NO. : MEC/TS/05/62/59A, R-0

\Akjha\work contrac\standard TS for Pipe Line WC (Vol-II) MASTER FOLDER\59A assorted pipe spec\Specification for Assorted Pipes 59A.doc

Page 297 of 644

<b>MECON LIMITED</b> Delhi	PROCESS & PIPING DESIGN SECTION	TECH	INICAL SPECIFICATION FOR ASSORTED PIPES	<b>ये</b> कॉन
TECHNICAL SPECIFICATION NO. :	MEC/TS/05/62/59A		REV-0	PAGE 1 OF 9

# CONTENTS

SI.No.	Description	Page No.
1.0	GENERAL	2
2.0	IBR PIPES	4
3.0	HYDROSTATIC TEST	5
4.0	MARKING & DESPATCH	5
ANNEXURE-I	: HYDROSTATIC TEST	7

Prepared By :	Checked By :	Approved By :

\\Akjha\work contrac\standard TS for Pipe Line WC (Vol-II) MASTER FOLDER\59A assorted pipe spec\Specification for Assorted Pipes 59A.doc

<b>MECON LIMITED</b> Delhi	PROCESS & PIPING DESIGN SECTION	TECH	NICAL SPECIFICATION FOR ASSORTED PIPES	<b>र्ट्र्ड्</b> मेकॉन
TECHNICAL SPECIFICATION NO. :	MEC/TS/05/62/59A		REV-0	PAGE 2 OF 9

# 1.0 **GENERAL**

1.1 All pipes and their dimensions, tolerances, chemical composition, physical properties, heat treatment, hydrotest and other testing and marking requirements shall conform to the latest codes and standards specified in the Material Requisition (MR). Deviation(s), if any, shall be clearly highlighted in the offer.

### 1.2 **Testing**

- 1.2.1 Test reports shall be supplied for all mandatory tests as per the applicable material specifications. Test reports shall also be furnished for any supplementary tests as specified in the MR & Clauses 1.10 & 1.11.
- 1.2.2 Material test certificates (physical property, chemical composition & treatment report) shall also be furnished for the pipes supplied.

# 1.3 Manufacturing Processes

- 1.3.1 Steel made by Acid Bessemer Process shall not be acceptable.
- 1.3.2 All longitudinally welded pipes other than IS:3589 should employ automatic welding.
- 1.4 Pipe shall be supplied in single or double random length of 4 to 7 and 7 to 14 meters, respectively.
- a) Seamless and E.R.W. pipes shall not have any circumferential seam joint in a random length. However, in case of E.FS.W. pipe, in one random length one welded circumferential seam of same quality as longitudinal weld is permitted. This weld shall be at least 2.5 m from either end. The longitudinal seams of the two portions shall be staggered by 90°. Single random length in such cases shall be 5 to 7m.
  - b) Unless otherwise mentioned in the respective material code, E.FS.W. pipes < 36'' shall not have more than one longitudinal seam joint and E.FS.W. pipes  $\geq 36''$  shall not have more than two longitudinal seam joints.

<b>MECON LIMITED</b> Delhi	PROCESS & PIPING DESIGN SECTION	TECH	INICAL SPECIFICATION FOR ASSORTED PIPES	<b>र्ट्र्ड्</b> मेकॉन
TECHNICAL SPECIFICATION NO. :	ATION NO. : MEC/TS/05/62/59A		REV-0	PAGE 3 OF 9

- 1.6 Pipe with screwed ends shall have NPT external taper pipe threads conforming to ASME/ ANSI B1.20.1 upto 1.5" NB & IS:554 for 2" to 6" NB.
- 1.7 Pipe with bevelled ends shall be in accordance with ASME B16.25. Weld contours shall be as follows:

Material	Wall Thickness	Weld Contour
Carbon Steel (Except Low	Upto 22mm	Figure 2 Type A
Temp. Carbon Steel)	> 22mm	Figure 3 Type A
Alloy Steel Stainless Steel &	Upto 10 mm	Figure 4
Low Temp. Carbon Steel	>10 mm & Upto 25 mm	Figure 5 Type A
	> 25 mm	Figure 6 Type A

- 1.8 Gavanished pipes shall be coated with zinc by hot dip process conforming to IS:4736/ ASTM A 153.
- 1.9 All austenitic stainless steel pipes shall be supplied in solution annealed condition.

### 1.10 **I.G.C. Test for Stainless Steels**

1.10.1 For all austenitic stainless steel pipes, intergranular corrosion test shall have to be conducted as per following:

ASTM A262 practice " B " with acceptance criteria of "60 mils/ year (max.)"

#### OR

ASTM 262 practice "E" with acceptance criteria of "No cracks as observed from 20X magnification" & "Microscopic structure to be observed from 250X magnification".

- 1.10.2 When specifically asked for in MR for high temperature application of some grades of austenitic stainless steel (eg.SS 309, 310, 316, 316H etc.), ASTM A262 practice "C" with acceptance criteria of "15 mils/ year (max.)" shall have to be conducted.
- 1.10.3 For the IGC test as described in 1.10.1 & 1.10.2, two sets of samples shall be drawn from each solution annealing lot; one set corresponding to highest carbon content and the other corresponding to the highest pipe

<b>MECON LIMITED</b> Delhi	PROCESS & PIPING DESIGN SECTION	TECH	INICAL SPECIFICATION FOR ASSORTED PIPES	<b>र्ट्र्ड्</b> मेकॉन
TECHNICAL SPECIFICATION NO. :	MEC/TS/05/62/59A	REV-0	PAGE 4 OF 9	

thickness. When testing in is conducted as per Practice "E", photograph of microscopic structure shall be submitted for record.

- 1.11 All welded pipes indicated as 'CRYO' & 'LT' in MR shall be impact tested per requirement and acceptance criteria of ASME B31.3. The impact test temperature shall be -196° C & -45° C for stainless steel and carbon steel, respectively, unless specifically mentioned otherwise in MR.
- 1.12 Pipes under 'NACE' category shall meet the requirements given in MR-01-75.
- 1.13 Specified heat treatment for carbon steel & alloy steel and solution annealing for stainless steel pipes shall be carried out after weld repairs. Number of weld repairs at the same spot shall be restricted to maximum two by approved repair procedure.
- 1.14 For black or galvanised pipes to IS:1239, the minimum percentage of elongation shall be 20%.

# 2.0 **IBR PIPES**

### 2.1 **IBR Documentation**

- 2.1.1 Pipes under purview of IBR shall be accompanied with IBR certificate original in Form IIIA, duly approved and countersigned by IBR authority/ local authority empowered by the Central Boiler Board of India. Photocopy of the original certificate duly attested by the local boiler inspector where the supplier is located is the minimum requirement for acceptance.
- 2.1.2 For materials 1 <sup>1</sup>/<sub>4</sub> Cr- <sup>1</sup>/<sub>2</sub> Mo (ASTM A335 Gr. P11/ A691 Gr. 1 <sup>1</sup>/<sub>4</sub> Cr) & 2 <sup>1</sup>/<sub>4</sub> Cr-1Mo (ASTM A335 Gr.P22/ A691 Gr. 2 <sup>1</sup>/<sub>4</sub> Cr.), from III-A approved by IBR shall include the tabulation of  $E_t$ ,  $S_c$  &  $S_r$  values for the entire temperature range given below.  $E_t$ ,  $S_c$  &  $S_r$  values shall be such that throughout the temperature range

		E <sub>t</sub> / 1.5 S <sub>r</sub> / 1.5	≥ ≥	Sa
		Sc	$\geq$	-
whe	re,			
S₄	:	Allowable	stress a	at the working metal temperature.
E <sub>t</sub>	:	Yield point	t (0.2%	proof stress at the working metal
		temperatu	ire).	

<b>MECON LIMITED</b> Delhi	PROCESS & PIPING DESIGN SECTION	TECH	INICAL SPECIFICATION FOR ASSORTED PIPES	<b>र्ट्र्ड्</b> मेकॉन
TECHNICAL SPECIFICATION NO. : MEC/TS/05/62/59A			REV-0	PAGE 5 OF 9

- $S_c$  : The average stress to produce elongation of 1% (creep) in 1,00,000 hrs at the working metal temperature.
- S<sub>r</sub> : The average stress to produce rupture in 1,00,000 hrs. at the working metal temperature and in no case more than 1.33 times the lowest stress to produce rupture at this temperature.

S <sub>A</sub> (psi)	Temperature (°F)											
Material	500	600	650	700	750	800	850	900	950	1000	1050	1100
A335 Gr. P11	17200	16700	16200	15600	15200	15000	14500	12800	9300	6300	4200	2800
A 691 Gr. 11/2 Cr	18900	18300	18000	17600	17300	16800	16300	15000	9900	6300	4200	2800
A335 Gr. P2/	17900	17900	17900	17900	17900	17800	14500	12800	10800	7800	5100	3200
A691 Gr. 2 ¼ Cr												

- Note:  $S_A$  values given above are as per ASME B31.3-1999. Values shall be as per the latest edition prevailing.
- 2.2 For carbon steel pipes under IBR, the chemical composition shall conform to the following;

Carbon (max.)	:	0.25%
Others (S, P, Mn)	:	As prescribed in IBR regulation.

The chemical composition as indicated in this clause is not applicable for pipes other than IBR services.

# 3.0 HYDROSTATIC TEST

Refer Annexure – I.

# 4.0 MARKING AND DESPATCH

- 4.1 All pipes shall be marked in accordance with the applicable codes, standards and specifications. In addition, the purchase order number, the item code & special conditions like "IBR", " CRYO", "NACE", etc., shall also be marked.
- 4.2 Pipes under "IBR", "CRYO", & "NACE" shall be painted in red stripes, light purple brown stripes & canary yellow stripes, respectively, longitudinally throughout the length for easy identification.
- 4.3 Paint or ink for marking shall not contain any harmful metal or metallic salts such as zinc, lead or copper which cause corrosive attack on heating.

<b>MECON LIMITED</b> Delhi	PROCESS & PIPING DESIGN SECTION	TECH	INICAL SPECIFICATION FOR ASSORTED PIPES	<b>र्यस्थि</b> मेकॉन
TECHNICAL SPECIFICATION NO. : MEC/TS/05/62/59A			REV-0	PAGE 6 OF 9

- 4.4 Pipes shall be dry, clen and free from from moisture, dirt and loose foreign materials of any kind.
- 4.5 Pipes shall be protected from rust, corrosion and mechanical damage during transportation, shipment and storage.
- 4.6 Rust preventive used on machined surfaces to be welded shall be easily removable with a petroleum solvent and the same shall not be harmful to welding.
- 4.7 Both ends of the pipe shall be protected with the following material:

Plain end	:	Plastic cap
Bevel end	:	Wood, Metal or Plastic cover
Threaded end	:	Metal or Plastic threaded cap

- 4.8 End protectors to be used on bevelled ends shall be securely and tightly attached with belt or wire.
- 4.9 Steel end protectors to be used on galvanised pipes shall be galvanised.

<b>MECON LIMITED</b> Delhi	PROCESS & PIPING DESIGN SECTION	TECH	INICAL SPECIFICATION FOR ASSORTED PIPES	<b>र्ट्र्ड्</b> मेकॉन
TECHNICAL SPECIFICATION NO. : MEC/TS/05/62/59A			REV-0	PAGE 7 OF 9

# ANNEXURE-I

# 3.0 HYDROSTATIC TEST

- 3.1 All pipes shall be hydrostatically tested.
- 3.2 The mill test pressure shall be as follows:

# 3.2.1 Seamless, E.R.W. & Spiral Welded

# a) Carbon Steel

Material Standard	Test Pressure Standard
ASTM A 106 Gr. B	ASTM A 530
API 5L Gr. B, Seamless	API 5L
API 5L, E.R.W.	API 5L
API 5L, Spiral	API 5L
ASTM A333 Gr.3 & 6, Seamless	ASTM A 530
ASTM A 333 Gr. 3 & 6, E.R.W.	ASTM A 530

# b) Seamless Alloy Steel

Material Standard	Test Pressure Standard
ASTM A335 GR.P1, P12, P11, P22, P5, P9	ASTM A 530
ASTM A268 TP 405, TP410	ASTM A530

# c) Seamless Stainless Steel

Material Standard	Test Pressure Standard
ASTM A312 Gr.TP304, 304L, 304H, 316,	ASTM A 530
316L, 316H, 321, 347	

# d) Seamless Nickel Alloy

Material Standard	Test Pressure Standard
ASTM B161 UNS No.2200	ASTM B161
ASTM B165 UNS No.4400	ASTM B165
ASTM B167 UNS No.6600	ASTM B167
ASTM B407 UNS No.8800	ASTM B407

<b>MECON LIMITED</b> Delhi	PROCESS & PIPING DESIGN SECTION	TECHNICAL SPECIFICATION FOR ASSORTED PIPES		<b>र्ट्र्ड्</b> मेकॉन
TECHNICAL SPECIFICATION NO. :	MEC/TS/05/62/59A		REV-0	PAGE 8 OF 9

# e) Welded Nickel Alloy

Material Standard	Test Pressure Standard		
ASTM B725 UNS No.2200, 4400	ASTM B725		
ASTM B517 UNS No.6600	ASTM B517		
ASTM B514 UNS No.8800	ASTM B514		

# 3.2.2 Electric Fusion Welded

# a) Carbon Steel & Alloy Steel E.FS.W. (16" & above)

Material Standard	Test Pressure Standard
API 5L Gr.B	P=2ST/ D
ASTM A 671 Gr.CC65, 70 (Cl.32)	S=90% of SMYS (except for API 5L
ASTM A 672 Gr.C60, 65, 70 (Cl.12,22)	Gr.B)
ASTM A 671 Gr.CF60, 65, 66, 70 (Cl.32)	S=85% of SMYS for API 5L Gr.B
ASTM A 691 Gr. 1/2 Cr, 1Cr, 1 1/4Cr, 2	T=Nominal Wall Thickness
¼Cr, 5Cr, 9Cr (Cl.42)	D=O.D. of Pipe

# b) Stainless Steel E.FS.W. (2" to 6")

The hydrostatic test pressure in kg/  $cm^2$  for the following materials shall be as given below:

Material Gr.1: ASTM A312 TP304/ 304H/ 316/ 316H/ 321/ 347 welded

Material Gr.2: ASTM A312 TP 304L/ 316L welded

	Pipe Sche	dule: S10	Pipe Sche	dule : S40	Pipe Schedule : S80		
Size	Material Gr.1	Material Gr.2	Material Gr.1	Material Gr.2	Material Gr.1	Material Gr.2	
2″	100	80	155	130	230	190	
3″	80	60	155	130	230	190	
4″	80	50	155	130	230	190	
6″	65	35	90	75	155	130	

\Akjha\work contrac\standard TS for Pipe Line WC (Vol-II) MASTER FOLDER\59A assorted pipe spec\Specification for Assorted Pipes 59A.doc

<b>MECON LIMITED</b> Delhi	PROCESS & PIPING DESIGN SECTION	TECHNICAL SPECIFICATION FOR ASSORTED PIPES		<b>र्ट्र्ड्</b> मेकॉन
TECHNICAL SPECIFICATION NO. :	IEC/TS/05/62/59A		REV-0	PAGE 9 OF 9

# c) Stainless Steel E.FS.W. (8" and above).

Material Standard	Test Pressure Standard
ASTM A358 TP 304L, 304, 304H,	P = 2ST/D
316L, 316, 316H, 321, 347	S = 85% of SMYS
(Classes 1, 3 & 4)	T = Nominal Wall Thickness
	D = O.D. of Pipe
ASTM A358 TP 304L, 304, 304H,	P = 2ST/D
316L, 316, 316H, 321, 347	S = 72% of SMYS
(Classes 2 & 5)	T = Nominal Wall Thickness
	D = O.D. of Pipe

# 3.2.3 Carbon Steel Pipes to IS Standards

Material Standard	Test Pressure Standard
IS :1239	IS :1239
IS :3589	IS :3589

# PROCESS & PIPING DESIGN SECTION MECON LIMITED DELHI – 110 092



# TECHNICAL SPECIFICATION FOR PRESSURE SAFETY VALVES

# SPECIFICATION NO. : MEC/TS/05/62/056, Rev-1

D:\TS\PSV - TS - 056 (R-1).doc

<b>MECON LIMITED</b> Delhi	PROCESS & PIPING DESIGN SECTION	TECHNICAL SPECIFICATION FOR PRESSURE SAFETY VALVES		Афіт 3001 Сол И
TECHNICAL SPECIFICATION NO. :	MEC/TS/05/62/056		REV-1	Page 1 of 9

			<u> </u>	<u>S</u>		
<u>SI.No.</u>	<b>Description</b>				<u>P</u>	age No.
1.0	GENERAL					2
2.0	VALVE SIZING	3				5
3.0	VALVE CONST	FRUCTION	I			5
4.0	NAMEPLATE					7
5.0	INSPECTION	& TESTIN	G			7
6.0	SHIPPING					9
7.0	GUARANTEE					9
8.0	REJECTION					9
Revision No.	Date	9	Revised by	Che	ecked by	Approved by
1			K.P. Singh	A.	K. Johri	Niraj Gupta
PREPARED BY	:	CHECKE	D BY :		APPROVED	) BY :
K.P. SINGH		A.K. JOH	IRI		NIRAJ GUI	ΡΤΑ

n

<b>MECON LIMITED</b> Delhi	PROCESS & PIPING DESIGN SECTION	TECHNICAL SPECIFICATION FOR PRESSURE SAFETY VALVES		
TECHNICAL SPECIFICATION NO. :	MEC/TS/05/62/056	REV-1		Page 2 of 9

# 1.0 **GENERAL**

#### 1.1 **Scope**

- 1.1.1 This specification together with the attached data sheets covers the requirements for the design, materials, nameplate marking, testing and shipping of pressure safety valves.
- 1.1.2 The related standards referred to herein and mentioned below shall be of the latest editions prior to the date of the Purchaser's enquiry :

ASME B 1.20.1	:	Pipe threads
ASME B 16.5	:	Pipe flanges and flanged fittings
ASME B 16.20	:	Ring joint gaskets and grooves for steel pipe flanges
ASME Sec.VIII	:	Boiler & pressure vessels codes for unfired pressure vessel
API RP 520 (Part-I & II)	:	Sizing, selection and installation of pressure relieving devices in refineries
API RP 521	:	Guide for pressure relieving and depressurising systems
API 526	:	Flanged steel safety-relief valves
API 527	:	Commercial seat tightness of refineries relief valve with metal to metal seats
DIN 50049	:	Document on material testing
IBR	:	Indian boiler regulations

- 1.1.3 In the event of any conflict between this specification, data sheets, related standards, codes etc, the Vendor should refer the matter to the Purchaser for clarifications and only after obtaining the same, should proceed with the manufacture of the items in question.
- 1.1.4 Purchaser's data sheets indicate the selected valve's relieving area, materials for the body, bonnet, disc, nozzle, spring, indicative inlet/outlet connection sizes, bellows etc. However, this does not relieve the Vendor of the responsibility for proper selection with respect to the following :

<b>MECON LIMITED</b> Delhi	PROCESS & PIPING DESIGN SECTION	TECHNICAL SPECIFICATION FOR PRESSURE SAFETY VALVES		
TECHNICAL SPECIFICATION NO. :	MEC/TS/05/62/056 RE		REV-1	Page 3 of 9

- a) Sizing calculations and selection of valve with proper relieving area to meet the operating conditions indicated.
- b) Selection of materials for all parts of the valve suitable for the fluid and its conditions indicated.
- 1.1.5 All process-wetted parts, metallic and non-metallic, shall be suitable for the fluids and service specified by the Purchaser. The service gas composition shall be as given in Annexure-I.

#### 1.2 Bids

- 1.2.1 Vendor's quotation shall include a detailed specification sheet for each pressure safety valve which shall provide all the details regarding type, construction materials, relieving area, relieving capacity, orifice letter designation, overpressure, blowdown, operating pressure, etc., and any other valve accessories.
- 1.2.2 All the units of measurement for various items in the Vendor's specification sheets shall be to the same standards as those in Purchaser's data sheets.
- 1.2.3 All the material specifications for various parts in the Vendor's specification sheets shall be to the same standards as those in Purchaser's data sheets.
- 1.2.4 Deleted.
- 1.2.5 Vendor shall enclose catalogues giving detailed technical specifications and other information for each type of pressure safety valve covered in the bid.
- 1.2.6 Vendor's quotation, catalogues, drawings, operating and maintenance manual, etc., shall be in English.
- 1.2.7 Vendor's quotation shall include detailed sizing calculation for each pressure safety valve. Published data for certified discharge coefficient and certified flow capacities and actual discharge area shall be furnished. Data used by Vendor without the above mentioned supported documentation shall, on prima-facie basis, be rejected.
- 1.2.8 All valves shall have been type tested for capacity as per ASME. A copy of the certificate shall be provided.
- 1.2.9 Vendor shall also quote separately for the following :
  - a) Two years recommended operational spares for each pressure relief valve and its accessories. List of such spares without price shall be indicated alongwith technical bid and separately with price.
  - b) Any specific tools needed for maintenance work.

<b>MECON LIMITED</b> Delhi	PROCESS & PIPING DESIGN SECTION	TECH PRES	INICAL SPECIFICATION FOR SSURE SAFETY VALVES	- 
TECHNICAL SPECIFICATION NO. : MEC/TS/05/62/056			REV-1	Page 4 of 9

1.2.10 Vendor's quotation shall include general arrangement and sectional drawings showing all features and major parts with reference numbers and material specification.

#### **IMPORTANT**

The drawings to be submitted alongwith the bid shall be in total compliance with the requirement of technical specification and data sheets of the valves with no exception & deviation.

1.2.11 Vendor's quotation shall include Quality Assurance Plan (QAP) enclosed with this tender duly signed, stamped & accepted.

#### 1.3 Drawings and Data

- 1.3.1 Detailed drawings, data, catalogues required from the Vendor are indicated by the Purchaser in this specification. The required number or reproducibles and prints should be dispatched to the address mentioned, adhering to the time limits indicated.
- 1.3.2 Within two weeks of placement of order, Vendor shall submit six copies of certified drawings and specification sheets for each pressure safety valve for Purchaser's final approval. These documents shall specially include the following :
  - a) Flange face to face dimension.
  - b) Height of the complete valve assembly.
  - c) Weight of the complete valve assembly.
  - d) Cold bench set pressure for the valve to be tested at atmospheric temperature and back pressure.
  - e) The cold test medium to be used for bench test in case it is different from air.
  - f) Horizontal reaction force at center line of valve outlet.
  - g) Relieving capacity of the valve under the same operating conditions.
  - h) Over pressure and blowdown/ reclosing pressure for each valve.
- 1.3.3 Vendor shall provide test certificates for all the tests indicated in clause 5.0 of this specification. In addition Vendor shall provide the Manufacturer's certificate of conformity to Purchaser's specifications as per clause 2.2 of Din 50049.
- 1.3.4 Within 30 days from the approval date, Manufacturer shall submit to Purchaser one reproducible and six copies of the approved drawings, documents and specifications as listed in clause 1.3.2 above.
- 1.3.5 Prior to shipment, Manufacturer shall submit one reproducible and six copies of the following:

<b>MECON LIMITED</b> Delhi	PROCESS & PIPING DESIGN SECTION	TECH PRES	INICAL SPECIFICATION FOR SSURE SAFETY VALVES	
TECHNICAL SPECIFICATION NO. : MEC/TS/05/62/056			REV-1	Page 5 of 9
-				

- a) Test certificates for all the tests indicated in clause 5.0 of this specification.
- b) Manual for installation, erection, maintenance and operation instructions, including a list of recommended spares for the valves.

#### 2.0 **VALVE SIZING**

2.1 Sizing shall be carried out using the formulae mentioned in the following standards, whenever the sizing code mentioned in the Purchaser's data sheets refers to them:

Sizing	Code	Standard	

- API API RP 520 Part-I
- ASME ASME boiler and pressure vessel code section VIII titled - Unfired pressure vessels
- IBR Indian Boiler Regulations Paragraph 293
- 2.2 Discharge co-efficient of Vendor's pressure safety valves shall be minimum 0.975 as per API 520. However, for valves covered under IBR, regulations of IBR shall govern.
- 2.3 For flanged pressure safety valves, the orifice letter designation and the corresponding relieving area indicated in the Purchaser's data sheet shall be as per API 526. For a valve of given inlet and outlet sizes and letter designation, relieving area of the valves offered by Vendor shall meet those in API-526, as a minimum.
- 2.4 The discharge capacity of selected pressure safety valves shall be calculated based on certified ASME capacity curves or by using ASME certified discharge coefficient and actual orifice area. Higher valve size shall be selected in case pressure relief valve discharge capacity is less than the required flow rate.
- 2.5 The definitions of various terminologies used in Purchaser's data sheets are as per paragraph 3.1 of API RP 520 Part-I.

### 3.0 VALVE CONSTRUCTION

#### 3.1 **Body**

- 3.1.1 Unless otherwise mentioned end connection details shall be as below :
  - a) Threaded end connections shall be to NPT as per ASME B 1.20.1.
  - b) Flanged end connections shall be as per ASME B 16.5.

<b>MECON LIMITED</b> Delhi	PROCESS & PIPING DESIGN SECTION	TECH PRES	INICAL SPECIFICATION FOR SSURE SAFETY VALVES	AGIT CAN DA
TECHNICAL SPECIFICATION NO. : MEC/TS/05/62/056			REV-1	Page 6 of 9

c)	Flanged face finish shall be serrated concentric to paragraphs 6.3.4.1, 6.3.4.2
	and 6.3.4.3 of ASME B 16.5. The face finish as specified in data sheets, shall
	have serrations as follows.

Serrated	:	250 to 500 microinches AARH
125 AARH	:	125 to 200 microinches AARH
63 AARH	:	32 to 63 microinches AARH

- 3.1.2 For flanged valves, inlet and outlet sizes & ratings and center to flange face dimensions shall be in accordance with API-526. Dimensional tolerances shall be as mentioned therein.
- 3.1.3 Body drain with a plug shall be provided as a standard feature on every pressure safety valve.

#### 3.2 **Trim**

- 3.2.1 The term `trim' covers all the parts of the valves exposed to and in contact with the process fluid except for the body and bonnet assembly.
- 3.2.2 Valves shall in general be of the full nozzle full lift type, unless otherwise specified.
- 3.2.3 Wherever stelliting of disc and nozzle has been specified, it stands for stelliting of the seat joint and the entire disc contour, unless otherwise mentioned.
- 3.2.4 Resilient seat/ seal or `O' rings wherever used shall be suitable for pressure and temperature conditions specified.

#### 3.3 Bonnet and Spring

- 3.3.1 All valves shall be provided with a cap over the adjusting bolt.
- 3.3.2 Lifting lever shall be provided whenever the fluid to be relieved is steam or air.
- 3.3.3 Valve spring design shall permit an adjustment  $\pm$  5% of the set pressure as a minimum.
- 3.3.4 Carbon Steel spring shall be cadmium/ nickel plated.
- 3.3.5 The allowable tolerances in set pressures are as below :

 $\pm$  0.14 kg/cm<sup>2</sup>(g) for set pressures upto and including 5 kg/cm<sup>2</sup>(g);  $\pm$ 3% for set pressure above 5 kg/cm<sup>2</sup>(g).

3.3.6 Bonnet shall be of the enclosed type in general. Open type of bonnet may be used only for non-toxic fluids.

<b>MECON LIMITED</b> Delhi	PROCESS & PIPING DESIGN SECTION	TECH PRES	INICAL SPECIFICATION FOR SSURE SAFETY VALVES	Hatr BEDT CONV
TECHNICAL SPECIFICATION NO. : MEC/TS/05/62/056			REV-1	Page 7 of 9

#### 3.4 **Pilot**

- 3.4.1 Wherever pilot operated valves are specified, pilot shall be non-flowing type and shall be designed fail safe.
- 3.4.2 All accessories like back flow preventer, pilot filter etc. required for proper operation of pilot operated valves as per indicated service conditions shall be included.
- 3.4.3 Wherever the body is part of flow path, body material shall be same as trim material, as a minimum.

#### 4.0 **NAMEPLATE**

- 4.1 Each pressure safety valve shall have a S.S. nameplate attached firmly to it at a visible place, furnishing the following information:
  - a) Tag number as per Purchaser's data sheets.
  - b) Manufacturer's serial no. or model no.
  - c) Manufacturer's name/ trade mark.
  - d) Nominal flanged size in inches and rating in lbs. for both inlet and outlet.
  - e) Orifice letter designation.
  - f) Valve set pressure.
  - g) Cold bench test set pressure.

Unit of the above pressures shall be marked in the same units as those followed in Purchaser's data sheets.

#### 5.0 **INSPECTION & TESTING**

- 5.1 Unless otherwise specified, Purchaser reserves the right to test and inspect all the items at the Vendor's works.
- 5.1.1 Purchaser's Inspector shall perform inspection and witness test on all valves as indicated in the Quality Assurance Plan (QAP) attached with this specification.
- 5.2 Vendor shall submit the following test certificates and test reports for Purchaser's review:
  - a) Material test certificate from the foundry (MIL certificate) for each valve body and bonnet castings, nozzle, disc etc.
  - b) Certificate of radiography / x-ray for valve castings. 100% radiography shall be carried out for all valve castings with body rating of 600# and above. A minimum of two shots shall be taken for all curved portion of the body and bonnet.

<b>MECON LIMITED</b> Delhi	PROCESS & PIPING DESIGN SECTION	TECH PRES	INICAL SPECIFICATION FOR SSURE SAFETY VALVES	и 3017 Ссков 1
TECHNICAL SPECIFICATION NO. : MEC/TS/05/62/056		REV-1	Page 8 of 9	

c) Hydrostatic test reports for all valve bodies and functional test reports for all valves as per clause 5.3 and 5.4 of this specification.

d) IBR certificate in Form III item 11 and shall be furnished for all safety valves in steam service in addition to Form III C. Form III C shall also be furnished for pressure relief valves in distribution network.

#### 5.3 Hydrostatic Test

5.3.1 Each pressure safety valve body and nozzle shall undergo hydrostatic test as per outlet flange and inlet flange ANSI rating, respectively. However all the safety valves castings covered under IBR shall be tested as per IBR regulations. There shall not be any visible leakage during this test.

### 5.4 Functional Tests

- 5.4.1 Assembled valves shall be subjected to functional tests as below :
  - a) Cold bench set pressure test

Pressure relief valve shall be tested for opening at specified set pressure and also for seat tightness.

b) Seat Leakage test as per API

Whenever the specified set pressure is less than or equal to 70 kg/cm<sup>2</sup>g, the valve shall meet the seat tightness requirements specified in API RP-527. The maximum permissible leakage rates for conventional and balanced bellow valves against various sizes shall be as specified therein. Whenever the specified set pressure exceeds 70 kg/cm<sup>2</sup>g, the Vendor shall submit the leakage rates of valves for approval by the Purchaser.

Where bubble tightness has been specified, there shall be no leakage or bubbles of air at the specified percentage of set pressure.

c) Valve lift test

### 5.5 Witness Inspection

All pressure safety valves shall be offered for pre-despatch inspection for following as a minimum :

- a) Physical dimensional checks and workmanship
- b) Hydrostatic test as per clause 5.3 of this specification.
- c) Functional test on representative samples.
- Review of all certificate and test reports as indicated in clause 5.2 of this specification.

<b>MECON LIMITED</b> Delhi	PROCESS & PIPING DESIGN SECTION	TECH PRES	INICAL SPECIFICATION FOR SSURE SAFETY VALVES	
TECHNICAL SPECIFICATION NO. : MEC/TS/05/62/056		REV-1	Page 9 of 9	

In the event of tests being not witnessed by Purchaser, the tests shall anyway be completed by the Vendor and documents for same submitted for scrutiny.

#### 6.0 **SHIPPING**

- 6.1 Valves shall be supplied as a whole, complete with all the accessories like cap, lifting lever, test gag, etc.
- 6.2 All threaded and flanged opening shall be suitably protected to prevent entry of foreign material.

#### 7.0 **<u>GUARANTEE</u>**

- 7.1 Manufacturer shall guarantee that the materials and machining of valves and fittings comply with the requirements in this specification and in the Purchase Order.
- 7.2 Manufacturer is bound to replace or repair all valve parts which should result defective due to inadequate engineering or to the quality of materials and machining.
- 7.3 If valve defect or malfunctioning cannot be eliminated, Manufacturer shall replace the valve without delay,
- 7.4 Any defect occurring during the period of Guarantee shall be attended to by making all necessary modifications and repair of defective parts free of charge to the Purchaser as per the relevant clause of the bid document.
- 7.5 All expenses shall be to Manufacturer's account.

#### 8.0 **REJECTION**

8.1 Vendor shall make his offer in detail with respect to every item of the Purchaser's specifications. Any offer not conforming to this shall be summarily rejected.

Edition:1

# SPECIFICATION FOR SEAMLESS FITTINGS & FLANGES [SIZE UPTO DN 400 mm (16") NB]

# SPECIFICATION NO.: MEC/TS/05/21/025



(OIL & GAS SBU) MECON LIMITED DELHI 110 092

Page 317 of 644

MECON LIMITED REGD. OFF: RANCHI	STANDARD TECHNICAL		
834002	OIL & GAS SBU,	मुकान 9001:2000 Contract	
	DOCUMENT NO		Page 1 of 1
TITLE	SEAMLESS FITTINGS & FLANGES [SLZE LIPTO DN 400	MEC/TS/05/21/025	REVISION : 0
	mm (16") NB]		EDITION : 1

# CONTENTS

# SL.NO. DESCRIPTION

- 1.0 SCOPE
- 2.0 REFERENCE DOCUMENTS
- 3.0 MANUFACTURER'S QUALIFICATION
- 4.0 MATERIALS
- 5.0 DESIGN AND MANUFACTURE
- 6.0 INSPECTION AND TESTS
- 7.0 TEST CERTIFICATES
- 8.0 PAINTING, MARKING AND SHIPMENT
- 9.0 DOCUMENTATION

PREPARED BY:	CHECKED BY:	APPROVED BY:	ISSUE DATE :
(Shalini Singh)	(Sunil Kumar)	(A.K. Johri)	Dec. 2008

MECON LIMITED REGD. OFF: RANCHI	STANDARD TECHNICAL		
834002	OIL & GAS SBU,	DELHI	18) 라하다 8001:2000 Contract
	DOCUMENT NO. SEAMLESS FITTINGS & MEC/TS/05/21/025		Page 1 of 5
TITLE			REVISION : 0
	mm (16") NB]		EDITION : 1

# 1.0 SCOPE

This specification covers the minimum requirements for the design, manufacture and supply of following carbon steel flanges and fittings of size upto DN 400 mm (16") to be installed in onshore pipeline systems handling non-sour hydrocarbons in liquid or gaseous phase including Liquefied Petroleum Gas (LPG) :

- Flanges such as welding neck flanges, blind flanges, spectacle blinds, spacers and blinds etc.
- Seamless fittings such as tees, elbows, reducers, caps, outlets etc.

# 2.0 **REFERENCE DOCUMENTS**

2.1 Reference has been made in this specification to the latest edition (edition enforce at the time of issue of enquiry) of the following Codes, Standards and Specifications :

ASME B31.4	-	Pipeline Hydrocarboi	Transportation ns and Other Liqu	Systems ids	for	Liquid
ASME B31.8	-	Gas Transmission and Distribution Piping Systems				
ASME B16.5	-	Pipe Flanges and Flanged Fittings				
ASME B16.9	-	Factory Made Wrought Steel Butt Welding Fittings				
ASME B 16.11	-	Forged Stee	el Fittings, Socket	Welding and	d Threa	Ided
ASME B 16.48	-	Steel Line B	Blanks			
ASME Sec VIII	-	Boiler and Constructior	Pressure Vess n of Pressure Vess	sel Code sels	- Rule	es for
ASME Sec IX	-	Boiler and Brazing Qua	Pressure Vesse alifications	el Code -	Weldin	g and
ASTM A 370	-	Standard Te Testing of S	est Methods and later list of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the	Definitions fo	or Mec	hanical
MSS-SP-25	-	Standard Manual Manual Manual Manual Manual Manual Manual Manual Manual Manual Manual Manual Manual Manual Manu And Unions	arking System for	<sup>·</sup> Valves, Fitt	ings, F	langes

MECON LIMITED REGD. OFF: RANCHI	STANDARD TECHNICAL	Bo ADDIT 2000 CONTRACT	
834002	OIL & GAS SBU, DELHI		
		DOCUMENT NO.	Page 2 of 5
TITLE	SEAMLESS FITTINGS & FLANGES [SLZE LIPTO DN 400	MEC/TS/05/21/025	REVISION : 0
	mm (16") NB]		EDITION : 1

MSS-SP-97 - Forged Carbon Steel Branch Outlet Fittings - Socket Welding, Threaded and Butt welding Ends.

2.2 In case of conflict between the requirements of this specification and the requirements of above referred Codes and Standards, the requirements of this specification shall govern.

# 3.0 MANUFACTURER'S QUALIFICATION

Manufacturer who intends bidding for fittings must possess the records of a successful proof test, in accordance with the provisions of ASME B16.9 / MSS-SP-75 as applicable.

# 4.0 MATERIAL

- 4.1 The Carbon Steel used in the manufacture of flanges and fittings shall be fully killed. Material for flanges and fittings shall comply with the material standard indicated in the Purchase Requisition. In addition, the material shall also meet the requirements specified hereinafter.
- 4.2 Each heat of steel used for the manufacture of flanges and fittings shall have Carbon Equivalent (CE) not greater than 0.45 calculated from check analysis in accordance with the following formula:

 $CE = C + \frac{Mn}{6} + \frac{Cr + Mo + V}{5} + \frac{Ni + Cu}{15}$ 

Carbon contents on check analysis shall not exceed 0.22%.

4.3 For flanges and fittings specified to be used for Gas service or LPG service, Charpy V-notch test shall be conducted on each heat of steel. Unless specified otherwise, the Charpy V-notch test shall be conducted at 0°C in accordance with the impact test provisions of ASTM A 370 for flanges and fittings.

The average absorbed impact energy values of three full-sized specimens shall be 27 joules. The minimum impact energy value of any one specimen of the three specimens analysed as above, shall not be less than 22 Joules.

When Low Temperature Carbon Steel (LTCS) materials are specified for flanges and fittings in Purchase Requisition, the Charpy V-notch test requirements of applicable material standard shall be complied with.

MECON LIMITED REGD. OFF: RANCHI	STANDARD TECHNICAL	B B B B B B B B B B B B B B B B B B B	
834002	OIL & GAS SBU,		
		DOCUMENT NO.	Page 3 of 5
TITLE	SEAMLESS FITTINGS & FLANGES [SLZE LIPTO DN 400	MEC/TS/05/21/025	REVISION : 0
	mm (16") NB]		EDITION : 1

- 4.4 For flanges and fittings specified to be used for Gas service or LPG service, Hardness test shall be carried out in accordance with ASTM A 370. Hardness testing shall cover at least 10% per item, per size, per heat, per manufacturing method. A full thickness cross section shall be taken for this purpose and the maximum hardness shall not exceed 248 HV<sub>10</sub>.
- 4.5 In case of RTJ (Ring Type Joint) flanges, the groove hardness shall be minimum 140 BHN. Ring Joint flanges shall have octagonal section of Ring Joint.

# 5.0 DESIGN AND MANUFACTURE

- 5.1 Flanges such as weld neck flanges and blind flanges shall conform to the requirements of ASME B16.5.
- 5.2 Spectacle blind and spacer & blind shall conform to the requirements of ASME B 16.48.
- 5.3 Fittings such as tees, elbows, reducers, etc. shall be seamless type and shall conform to ASME B16.9 for sizes DN 50 mm (2") to DN 400 mm (16") (both sizes included) and ASME B 16.11 for sizes below ON 50 mm (2").
- 5.4 Fittings such as weldolets, sockolets, nippolets, etc. shall be manufactured in accordance with MSS-SP-97.
- 5.5 Type, face and face finish of flanges shall be as specified in Purchase Requisition.
- 5.6 Flanges and fittings manufactured from bar stock are not acceptable.
- 5.7 All butt weld ends shall be bevelled as per ASME B 16.5 / ASME B 16.9 / MSS-SP-97 as applicable.
- 5.8 Repair by welding on flanges and fittings is not permitted.
- 5.9 Stub-in or pipe to pipe connection shall not be used in the manufacture of tees. Tees shall be manufactured by forging or extrusion methods. The longitudinal weld seam shall be kept at 90° from the extrusion. Fittings shall not have any circumferential joint.

MECON LIMITED	STANDARD TECHNICAL	80 9001 COUNT	
834002	OIL & GAS SBU,		
		DOCUMENT NO.	Page 4 of 5
TITLE	SEAMLESS FITTINGS & FLANGES [SLZE LIPTO DN 400	MEC/TS/05/21/025	REVISION : 0
	mm (16") NB]		EDITION : 1

# 6.0 INSPECTION AND TESTS

- 6.1 The Manufacturer shall perform all inspections and tests as per the requirement of this specification and the relevant codes, prior to shipment at his works. Such inspections and tests shall be, not but limited to the following :
  - a) All flanges and fittings shall be visually inspected. The internal and external surfaces of the flanges and fittings shall be free from any strikes, gauges and other detrimental defects.
  - b) Dimensional checks shall be carried out on finished products as per ASME B16.5 for flanges, ASME B16.48 for spacers and blinds and ASME B16.9 / MSS-SP-97 as applicable for fittings and as per this specification.
  - c) Chemical composition and mechanical properties shall be checked as per relevant material standards and this specification, for each heat of steel used.
  - d) All finished wrought weld ends subject to welding in field, shall be 100% tested for lamination type defects by ultrasonic test. Any lamination larger then 6.35 mm shall not be acceptable.
- 6.2 Purchaser's Inspector reserves the right to perform stage wise inspection and witness tests, as indicated in clause 6.1 of this specification at Manufacturer's Works prior to shipment. Manufacturer shall give reasonable notice' of time and shall provide, without charge, reasonable access and facilities required for inspection, to the Purchaser's Inspector.

Inspection and tests performed / witnessed by Purchaser's Inspector shall in no way relieve the Manufacturer's obligation to perform the required inspection and tests.

# 7.0 TEST CERTIFICATES

Manufacturer shall furnish the following certificates:

- a) Test certificates relevant to the chemical analysis and mechanical properties of the materials used for manufacture of flanges and fittings as per relevant standards and this specification.
- b) Test Reports on non destructive testing.
- c) Certificates for each fitting stating that it is capable of withstanding without leakage a test pressure, which results in a hoop stress equivalent to 100 % of

MECON LIMITED REGD. OFF: RANCHI	LIMITED STANDARD TECHNICAL SPECIFICATION		
834002	OIL & GAS SBU	18 3001:2000 Convert	
		DOCUMENT NO.	Page 5 of 5
TITLE	SEAMLESS FITTINGS & FLANGES [SLZE LIPTO DN 400	MEC/TS/05/21/025	REVISION : 0
	mm (16") NB]		EDITION : 1

the specified minimum yield strength for the pipe with which the fitting is to be attached without impairment of serviceability.

## 8.0 PAINTING, MARKING AND SHIPMENT

- 8.1 After all inspection and tests required have been carried out; all external surfaces shall be thoroughly cleaned to remove grease, dust and rust and shall be applied with standard mill coating for protection against corrosion during transit and storage. The coating shall be easily removable in the field.
- 8.2 Ends of all fittings and weld neck flanges shall be suitably protected to avoid any damage during transit. Metallic or high impact plastic bevel protectors shall be provided for fittings and flanges. Flange face shall be suitably protected to avoid any damage during transit.
- 8.3 All flanges and fittings shall be marked as per applicable dimension / manufacturing standard.

# 9.0 DOCUMENTATION

Documentation to be submitted by Manufacturer to Company is summarized below. Number of Copies (Hard copies / soft copies etc.) shall be as indicated in CONTRACT document / Material Requisition.

- 9.1 At the time of bidding, Manufacturer shall submit the following documents:
  - a) Reference list of previous supplies of similar fittings of similar specification.
  - b) Clausewise list of deviations from this specification, if any.
  - c) Brief description of the manufacturing and quality control facilities at Manufacturer's works.
  - d) Manufacturer's qualification requirement as per clause 3.0 of this specification.
  - e) Quality Assurance Plan (QAP) enclosed with this tender duly signed, stamped and accepted.
- 9.2 Prior to shipment, the Manufacturer shall submit test certificates as listed in clause 7.0 of this specification.
- 9.3 All documents shall be in English Language only.

Edition:1

# SPECIFICATION FOR FLANGES AND WELDED FITTINGS [SIZE DN 450 mm (18") AND ABOVE]

# SPECIFICATION NO.: MEC/TS/05/21/026



(OIL & GAS SBU) MECON LIMITED DELHI 110 092

Page 324 of 644
MECON LIMITED REGD. OFF: RANCHI	STANDARD TECHNICAL SPECIFICATION OIL & GAS SBU, DELHI		
834002			मिकान मिकान 12 9001:2000 Commen
		DOCUMENT NO.	Page 1 of 1
TITLE	FLANGES AND WELDED	MEC/TS/05/21/026	REVISION : 0
	(18") AND ABOVE]		EDITION : 1

# CONTENTS

# SL.NO. DESCRIPTION

- 1.0 SCOPE
- 2.0 REFERENCE DOCUMENTS
- 3.0 MANUFACTURER'S QUALIFICATION
- 4.0 MATERIALS
- 5.0 DESIGN AND MANUFACTURE
- 6.0 INSPECTION AND TESTS
- 7.0 TEST CERTIFICATES
- 8.0 PAINTING, MARKING AND SHIPMENT
- 9.0 DOCUMENTATION

PREPARED BY:	CHECKED BY:	APPROVED BY:	ISSUE DATE :
(Shalini Singh)	(Sunil Kumar)	(A.K. Johri)	Dec. 2008

MECON LIMITED	STANDARD TECHNICAL SPECIFICATION		
01L & GAS SBU, DELHI		, DELHI	10 मेकान 9001:2000 Control
		DOCUMENT NO.	Page 1 of 7
TITLE	FLANGES AND WELDED	MEC/TS/05/21/026	REVISION : 0
	(18") AND ABOVE]		EDITION : 1

# 1.0 **SCOPE**

This specification covers the minimum requirements for the design, manufacture and supply of following items to be installed in pipeline system handling hydrocarbons in liquid or gaseous phase including Liquefied Petroleum Gas (LPG) :

- Carbon Steel Welded Fittings 450 mm (18") NB and above, such as tees, elbows, reducers, caps, outlets etc.
- Carbon Steel Flanges 450mm (18") NB and above, such as welding neck flanges, blind flanges, spectacle bind, spacers & blinds etc.

This specification does not cover the above mentioned items which are to be installed in pipeline system handling sour hydrocarbons (liquid/ gas) service as defined in NACE Standard MR-01-75.

# 2.0 **REFERENCE DOCUMENTS**

2.1 Reference has been made in this specification to the latest edition (edition enforce at the time of issue of enquiry) of the following Codes, Standards and Specifications :

ASME B31.4	-	Pipeline Transportation Systems for Liquid Hydrocarbons and Other Liquids
ASME B31.8	-	Gas Transmission and Distribution Piping Systems
ASME B16.5	-	Pipe Flanges and Flanged Fittings
ASME B16.9	-	Factory Made Wrought Steel Butt Welding Fittings
ASME B 16.11	-	Forged Steel Fittings, Socket Welding and Threaded
ASME B 16.48	-	Steel Line Blanks
ASME Sec VIII	-	Boiler and Pressure Vessel Code - Rules for Construction of Pressure Vessels
ASME Sec IX	-	Boiler and Pressure Vessel Code - Welding and Brazing Qualifications
ASTM A 370	-	Standard Test Methods and Definitions for Mechanical Testing of Steel Products.

Page 326 of 644

	STANDARD TECHNICAL SPECIFICATION			
834002	OIL & (	OIL & GAS SBU, DELHI		
			DOCUMENT NO.	Page 2 of 7
TITLE	FLANGES AND WELE	DED 0 mm	MEC/TS/05/21/026	REVISION : 0
	(18") AND ABOVE]			EDITION : 1
	MSS-SP-44 -	Stee	el Pipeline Flanges 22" & 2	6" and above.
	MSS-SP-25 -	Stan and	idard Marking System for Unions	Valves, Fittings, Flange
	MSS-SP-97 -	Forg Weld	jed Carbon Steel Branch ding, Threaded and Butt w	Outlet Fittings - Socke elding Ends.

2.2 In case of conflict between the requirements of this specification and the requirements of above referred Codes and Standards, the requirements of this specification shall govern.

# 3.0 MANUFACTURER'S QUALIFICATION

The design of fittings shall be established by mathematical analysis contained in ASME Sec. VIII/ ASME B31.3. The design of fittings for which mathematical analysis is not available shall be established by proof testing. These records shall be submitted at the time of bidding, qualifying the complete range of fittings offered. Manufacturer who intends bidding for fittings must posses the records of a successful proof test in accordance with the provisions of ASME B16.9 and/ or MSS-SP-75. These records shall be submitted at the time of bidding, qualifying the complete range of fittings offered. Failure to submit such records at the time of bidding may become a cause of rejection of the offer.

# 4.0 **MATERIALS**

- 4.1 The steel used in the manufacture of fittings and flanges shall be fully killed carbon steel with a grain size of ASTM 7 or finer as defined in ASTM E112. This requirement shall not apply to quenched and tempered fittings. The basic material for fittings and flanges shall be as indicated in the Material Requisition. Additionally, the material shall also meet the requirements specified hereinafter.
- 4.2 Each heat of steel used for the manufacture of fittings and flanges shall have carbon equivalent (CE) not greater than 0.45 calculated from check analysis in accordance with the following formula:

	Mn	Cr + Mo + V	Ni + Cu
CE =	C + +	+	
	6	5	15

4.3 Carbon contents on check analysis shall not exceed 0.22%.

MECON LIMITED REGD. OFF: RANCHI	STANDARD TECHNICAL		
834002	002 OIL & GAS SBU, DELHI		10 मेकान 10 9001:2000 Confine
		DOCUMENT NO.	Page 3 of 7
TITLE	FLANGES AND WELDED	MEC/TS/05/21/026	REVISION : 0
	(18") AND ABOVE]		EDITION : 1

4.4 For flanges and fittings specified to be used for Gas service or LPG service, Charpy V-notch test shall be conducted on each heat of steel. Unless specified otherwise, the Charpy V-notch test shall be conducted at 0°C in accordance with the impact test provisions of ASTM A 370 for flanges and MSS-SP-75 for all fittings.

The average absorbed impact energy values of three full-sized specimens shall be 27 joules. The minimum impact energy value of any one specimen of the three specimens analysed as above, shall not be less than 22 Joules.

When Low Temperature Carbon Steel (LTCS) materials are specified in Material Requisition for flanges and fittings, the Charpy V-notch test requirements of applicable material standard shall be complied with.

- 4.5 Hardness test shall be carried out as per ASTM A370 for each heat of steel used. A full thickness cross-section shall be taken for this purpose and the maximum hardness of base metal, weld metal and heat affected zone shall not exceed 248 HV<sub>10</sub>. Hardness testing shall cover at least 10% per item, per size, per heat, per manufacturing method.
- 4.6 One transverse guided weld bend test shall be performed for each lot of welded fittings produced from the same heat in accordance with provisions of MSS-SP-75. The dimension "A" in guided bend test shall not exceed 4.0 times the nominal wall thickness and dimension "B" shall be equal to A+2t+3.2mm, where "t" is nominal wall thickness.
- 4.7 One transverse weld tensile test shall be conducted on each heat/ lot of welded fittings in accordance with the requirements of MSS-SP-75.
- 4.8 In case of RTJ (Ring Type Joint) flanges, the groove hardness shall be minimum 140 BHN. Ring Joint flanges shall have octagonal section of Ring Joint.

# 5.0 DESIGN AND MANUFACTURE

- 5.1 Flanges such as weld neck flanges and blind flanges shall conform to the requirements of ASME B16.5 upto size DN 600mm (24") excluding DN 550mm (22"), MSS-SP-44 for sizes DN 550mm (22") and ASME B16.47 for sizes DN 650mm (26") and above.
- 5.2 Spectacle blind and Spacer & blind shall conform to the requirements of API 590 upto sizes DN 600mm (24"). For sizes above DN 650mm (26") and above, Spectacle blind and Spacer & blind shall conform to Manufacturer's standard.

	STANDARD TECHNICAL SPECIFICATION		मेकान मुख्या:2000 Com/M	
834002	OIL & GAS SBU,			
	FLANGES AND WELDED     DOCUMENT NO.       FLANGES AND WELDED     MEC/TS/05/21/026	DOCUMENT NO.	Page 4 of 7	
TITLE		MEC/TS/05/21/026	REVISION : 0	
	(18") AND ABOVE]		EDITION : 1	
5.3	Type, face and face finish of flang	es shall be as specified in	Material Requisition.	
5.4	Fittings such as tees, elbows and reducers shall be either welded or seamless type. All fittings shall comply with the requirements of MSS-SP-75. Fittings such as weldolets etc. shall be manufactured in accordance with MSS-SP-97.			
5.5	Tees shall be manufactured by forging or extrusion method. Stub-in or pipe to pipe connection shall not be used in the manufacture of tees. The longitudinal weld seam shall be kept at 90° from the extrusion. Fittings shall not have any circumferential weld joint.			
5.6	All butt weld ends shall be bevelle as applicable for flanges and MSS	ed as per ASME B16.5/ MS S-SP-75 / MSS-SP-97 as a	SS-SP-44/ ASME B16.47 pplicable for fittings.	
5.7	Inside weld projection for welded fitting shall not exceed 1.6 mm. The reinforcement of inside weld seam shall be removed for a distance of 100mm from each end of welded fittings.			
5.8	All welds shall be made by welders and welding procedures qualified in accordance with provisions of ASME Sec. IX. The procedure qualification shall include impact test for weld/ heat affected zone, hardness test and guided bend test and shall meet the requirements of Clauses 4.4, 4.5 and 4.6 of this specification, respectively.			
5.9	Repair by welding on flanges and parent metal of fittings is not allowed. Repair of weld seam by welding shall be carried out by welders and welding procedures duly qualified as per ASME Section IX and API 1104 and records for each repair shall be maintained. Repair welding procedure qualification shall include all tests which are applicable for regular production welding procedure qualification.			
6.0	INSPECTION AND TESTS			
6.1	The Manufacturer shall perform all inspections and tests as per the requirement of this specification and the relevant codes, prior to shipment at his works. Such inspections and tests shall be, but not limited to, the following :			
6.1.1	All fittings and flanges shall be visually inspected. The internal and external surfaces of the fittings shall be free from any strikes, gouges, burrs and other detrimental defects.			
6.1.2	Dimensional checks shall be carr MSS-SP-44/ ASME B16.47 as app	ied out on finished produ plicable for flanges and AS	cts as per ASME B16.5/ SME B16.9/ MSS-SP-75 /	

MECON LIMITED	STANDARD TECHNICAL		
834002	2 OIL & GAS SBU, DELHI		मित्र मेकान मेकान 1000:2000 Commit
		DOCUMENT NO.	Page 5 of 7
TITLE	FLANGES AND WELDED	MEC/TS/05/21/026	REVISION : 0
	(18") AND ABOVE]		EDITION : 1

MSS-SP-97 as applicable for fittings and as per this specification. Fittings not covered in MSS-SP-75 shall be checked as per Manufacturer's standard.

- 6.1.3 Chemical composition and mechanical properties shall be checked as per relevant material standards and this specification, for each heat of steel used.
- 6.1.4 The non-destructive inspection shall be carried out as given below :
  - a) All butt and repair welds for welded fittings shall be examined 100% by radiography. Acceptance limits shall be as per API 1104.
  - b) When elbows of size ≥ 18" NB are manufactured, the first elbow of each radius, diameter and wall thickness shall be ultrasonically checked for sufficient wall thickness in areas where a minimum wall thickness is to be expected. This shall be followed by random inspection of one out of every three elbows of the same radius, diameter and wall thickness.
  - c) All finished wrought weld ends shall be 100% tested for lamination type defects by ultrasonic test. Any lamination larger than 6.35mm shall not be acceptable.
  - d) Magnetic particle or liquid penetrant examination shall be performed on cold formed butt welding tees with extruded outlets, that are subjected to an extreme fiber elongation of greater than 5% shall be carried out as per the Supplementary Requirement SR3 of MSS-SP-75.
  - e) Welds which cannot be inspected by radiographic methods shall be checked by ultrasonic or magnetic particle methods. Acceptance criteria shall be as per ASME Section VIII Appendix 12 and Appendix 6, respectively.
- 6.2 Purchaser's Inspector reserves the right to perform stagewise inspection and witness tests, as indicated in Clause 6.1 of this specification at Manufacturer's Works prior to shipment. Manufacturer shall give reasonable notice of time and shall provide, without charge, reasonable access and facilities required for inspection, to the Purchaser's Inspector. Inspection and tests performed/ witnessed by Purchaser's Inspector shall in no way relieve the Manufacturer's obligation to perform the required inspection and tests.

MECON LIMITED	ECON LIMITED       STANDARD TECHNICAL SPECIFICATION         GD. OFF: RANCHI       OIL & GAS SBU, DELHI		
834002			मित्र मेकान मेठान 9001:2000 Contrain
		DOCUMENT NO.	Page 6 of 7
TITLE	FLANGES AND WELDED	MEC/TS/05/21/026	REVISION : 0
	(18") AND ABOVE]		EDITION : 1

# 7.0 **TEST CERTIFICATES**

Manufacturer shall furnish the following certificates :

- a) Test certificates relevant to the chemical and mechanical properties of the materials used for manufacture of flanges and fittings as per relevant standards and this specification.
- b) Test Reports on radiography, ultrasonic inspection and magnetic particle examination.
- c) Test reports of heat treatment carried out as per the specification.
- d) Welding procedures and welders' qualification reports.
- e) Test certificates for each fitting stating that it is capable of withstanding without leakage a test pressure which results in a hoop stress equivalent to 100% of the specified minimum yield strength for the pipe with which the fitting is to be attached without impairment of serviceability.

# 8.0 PAINTING, MARKING AND SHIPMENT

- 8.1 After all required inspection and tests have been carried out, all external surfaces shall be thoroughly cleaned to remove grease, dust & rust and shall be applied with standard mill coating for protection against corrosion during transit and storage. The coating shall be easily removable in the field. Manufacturer shall furnish the details of paint used at the time of bidding.
- 8.2 Ends of all fittings and weld neck flanges shall be suitably protected to avoid any damage during transit. Metallic or high impact plastic bevel protectors shall be provided for fittings and flanges. Flange face shall be suitably protected to avoid any damage during transit.
- 8.3 All fittings and flanges shall be marked as per applicable dimension / manufacturing standard.
- 8.4 Package shall be marked legibly with suitable marking ink to indicate the following :
  - a) Manufacturer's Name
  - b) Type of flange(s) and fittings(s)
  - c) Nominal diameter, thickness and material grade
  - d) Purchase order number and item serial number

MECON LIMITED	STANDARD TECHNICAL		
834002	OIL & GAS SBU, DELHI		10 मेकान 10 9001:2000 Colificati
		DOCUMENT NO.	Page 7 of 7
TITLE	FLANGES AND WELDED	MEC/TS/05/21/026	REVISION : 0
	(18") AND ABOVE]		EDITION : 1

# 9.0 DOCUMENTATION

- 9.1 Manufacturer shall furnish at the time of bidding, the following documents:
  - a) Reference list of similar supplies including all relevant details, viz. Project, Year, Client, Location, Size and Service.
  - b) Record of successful qualification test of fittings in compliance with the requirement of this specification.
  - c) Brief description of the manufacturing, heat treatment and quality control facilities of the Manufacturer's Works.
  - d) Clause-wise list of deviations from this specification, if any.
- 9.2 Within three weeks of placement of order, Manufacturer shall submit four copies of method of manufacture, testing and quality control procedure for raw material and finished product for Purchaser's approval.

Once the approval has been given by Purchaser, any changes in design, material and method of manufacture shall be notified to the Purchaser, whose approval in writing of all changes shall be obtained before the flanges and fittings are manufactured.

- 9.3 Within four weeks from the approval date, Manufacturer shall submit six copies of all documents as listed in Clause 9.2 of this specification.
- 9.4 Prior to shipment, the Manufacturer shall submit six copies of the test certificates as listed in Clause 7.0 of this specification.
- 9.5 All documents shall be in English Language only.

# SPECIFICATION FOR GASKETS, BOLTS & NUTS

# **SPECIFICATION NO.: MEC/S/05/21/19**



(OIL & GAS SBU) MECON LIMITED DELHI 110 092

Page 333 of 644

MECON LIMITED	STANDARD TECHNICAL		
834002	OIL & GAS SBU,	मित्र मेकान 9001:2000 Contract	
		DOCUMENT NO.	Page 1 of 1
TITLE	<b>GASKETS, BOLTS &amp; NUTS</b>	MEC/S/05/21/19	REVISION : 0
			EDITION: 1

# <u>CONTENTS</u>

# SL.NO. DESCRIPTION

- 1.0 GASKETS
- 2.0 NUTS AND BOLTS

PREPARED BY:	CHECKED BY:	APPROVED BY:	ISSUE DATE :
(Shalini Singh)	(Sunil Kumar)	(A.K. Johri)	Jan. 2009

MECON LIMITED REGD. OFF: RANCHI	STANDARD TECHNICAL		
834002	OIL & GAS SBU	, DELHI	日 初日 12000 Control
		DOCUMENT NO.	Page 1 of 1
TITLE	<b>GASKETS, BOLTS &amp; NUTS</b>	MEC/S/05/21/19	REVISION : 1
			EDITION: 1

# **AMENDMENT STATUS**

SI. No.	Clause / Paragraph / Annexure / Exhibit / Drawing Amended	Page No.	Revision	Date	By (Name)	Verified (Name)

MECON LIM		STANDARD TECHNI	ICAL S	PECIFICATION			
OIL & GAS SBU, DELHI						मुकान मुकान 9001:2000 Comm	
			DOCUMENT NO.				
TITLE	TITLE GASKETS, BOLTS & NUTS MEC/S/05/21/19				<b>REVISION: 0</b>		
						EDITION: 1	
		TECHNICAL		ES FOR GASKETS			
1.0	All gasl Vendor	kets shall conform to the cod shall strictly comply with MR /	des/sta / PR st	ndards and specifica pulations and no dev	tions giv iations s	ven in the requisitions hall be permitted.	
2.0	Process accorda	of manufacture, dimensions ance with the requirements of	and the ma	tolerances not speci inufacturer's standard	fied in ds.	requisition shall be	
3.0	Test re in the r	ports shall be supplied for all n equisition.	mandat	ory tests for gaskets	as per t	he standards specif	
4.0	Chemic reports	al composition and hardness on samples.	of RTJ	gaskets shall also be	e furnish	ned in the form of t	
5.0 For Spiral wound material following shall be furnished:							
	a.	Manufacturer's test certificate material specifications.	e for fil	er material and spira	al mater	ial as per the releva	
	b.	Manufacturer's test certificate ability & recovery as per the re	e for r elevan	aw materials and te t material specificatio	sts for ns.	compressibility / se	
6.0	Full fac	e gaskets shall have bolt holes	s punch	ned out.			
7.0	Filler m	aterial for spiral wound gasket	ts shall	not have any colour	or dye.		
8.0	All spira CS unle	All spiral wound gaskets shall be supplied with Outer ring. Material of the outer ring shall CS unless other wise specified in the MR.					
9.0	For spir materia shall be	ral wound gaskets, material o I. In addition to the requirem provided for the following:	of Inne ients a	er Compression ring s per code and as sp	shall be becified	e same as Spiral Si in the MR, inner rir	
	a.	Sizes 26" and above.					
	b.	Class 900 and above.					
10.0	Hardne specifie	ss of metallic RTJ gaskets sha d in MR :	all not e	exceed the values spo	ecified b	elow unless otherw	
	Ring G	asket Material	Maxii	num Hardness (BH	IN)		
	Soft Iro Carbon 5 Cr. ½ Type 30 Type 30	n steel 2 Mo 04, 316, 321, 347 04L, 316L	90 120 130 140 120				

MECON LIMITED REGD. OFF: RANCHI	STANDARD TECHNICAL	SPECIFICATION	
834002	OIL & GAS SBU,	, DELHI	मित्रान मुख्या:2000 Contract
		DOCUMENT NO.	Page 2 of 4
TITLE	<b>GASKETS, BOLTS &amp; NUTS</b>	MEC/S/05/21/19	<b>REVISION: 0</b>
			EDITION: 1

- 11.0 Face finish of metallic RTJ gaskets shall be 32 to 63 AARH.
- 12.0 Gaskets of different types and sizes shall be placed in separate shipping containers and each container clearly marked with the size, rating, material specification and item code.
- 13.0 All items shall be inspected and approved by MECON Inspector or any other agency authorized by MECON.
- 14.0 Any additional requirements specified in the requisition, shall be fully complied with.
- 15.0 Non-metallic ring gaskets as per ASME B16.21 shall match flanges to ASME B16.5 upto 24" (except 22" size) and to ASME B16.47B above 24" unless specified otherwise. For 22" size, the matching flange standard shall be MSS-SP44 unless specified otherwise.
- 16.0 Spiral wound gasket as per ASME B16.20 shall match flanges to ASME B16.5 upto 24" (except 22" size) and to ASME B16.47B above 24" unless specifically mentioned otherwise. For 22" size, the matching flange standard shall be MSS-SP44 unless specified otherwise.
- 17.0 The following abbreviations have been used in the Material Requisition for Spiral Wound Gaskets :

(I)	:	Inner Ring
(0)	:	Outer Ring
ĊĂF	:	Compressed Asbestos Fibre
GRAFIL	:	Grafoil Filler

		TED STANDARD TECHNICAL SPECIFICATION			
834002	KANCHI	OIL & GAS SBU, DELHI			
			DOCUMENT NO.	Page 3 of 4	
TITL	Ξ	GASKETS, BOLTS & NUTS	MEC/S/05/21/19	<b>REVISION: 0</b>	
				EDITION: 1	
1.0	The pr marking / stand accorda	<b>TECHNICAL NOTES</b> ocess of manufacture, heat treat g for all stud bolts, m/c bolts, jack s ards and specification given in the ince with the material specification	<b>5 FOR BOLTS &amp; NUTS</b> tment, chemical & mech crews & nuts shall be in a requisition. The applicab shall be stamped on each	anical requirements an accordance with the cod le identification symbol bolt and nut. Vendor sh	
2.0	Test respecific	eports shall be supplied for all ations.	mandatory tests as pe	er the relevant mater	
3.0	Materia Mechar	l test certificate shall also be ical Requirement)	furnished. (Heat Analysis	s, Product Analysis a	
4.0	Stress I materia	Rupture Test as detailed in ASTM A I irrespective of the temperature.	A453 shall be carried out	for all ASTM A453 bolti	
5.0	All bolti for nuts	ng shall be as per ANSI B 18.2.1 fo	or studs. M/c bolts and jac	kscrews and ANSI BI8.2	
6.0	Threads fit for s	s shall be unified (UNC for 1" dia ar tuds, M/c bolts and jackscrews and	nd BUN for> 1" dia) as per class 2B fit for nuts.	ANSI B.1.1 with class	
7.0	Stud bo be in ac	olts shall be threaded full length wit accordance with the requirement as p	th two heavy hexagonal nu per ANSI B 16.5.	uts. Length tolerance sh	
8.0	The nui by the l	ts shall be double chamfered, semi not forged process and stamped as	i-finished, heavy hexagona per respective material sp	al type and shall be ma ecification.	
9.0	Heads rounde	of jackscrews and m/c bolts shall d.	be heavy hexagonal type	e. Jackscrew end shall	
10.0	Each s contain case 'Cl	ze of studs & m/c bolts with ne ers marked with size and material s RYO' is specified in the requisition.	uts and jackscrews shall specifications. 'CRYO' shall	be supplied in separa be marked additionally	
11.0	All iten agency	ns shall be inspected and approve authorized by MECON.	ed (stagewise) by MECOI	N inspector or any oth	
12.0	The hea	at treatment for stud bolts & nuts sl	hall be as per code unless	mentioned otherwise.	
13.0	All aust unless s	enitic stainless steel bolts, nuts, scr specified otherwise in the material s	rews shall be supplied in so pecification.	olution annealed condition	

MECON LIMITED	STANDARD TECHNICAL	SPECIFICATION	
834002	OIL & GAS SBU,	, DELHI	मिठ मिठान प्रती
		DOCUMENT NO.	Page 4 of 4
TITLE	<b>GASKETS, BOLTS &amp; NUTS</b>	MEC/S/05/21/19	<b>REVISION: 0</b>
			EDITION: 1

- 15.0 Stud bolts, nuts & jackscrews shall be impact tested wherever specified in the material specification and also where the material specification is indicated as "CRYO". For" S.S. nuts and bolts minimum impact energy absorption shall be 27 Joules and test temperature shall be -196°C unless mentioned otherwise. For other materials impact energy and test shall be as per respective code.
- 16.0 Bolts / nuts of material of construction B7M / 2HM shall be 100% Hardness tested as per supplementary requirement S3 of ASTM A 193.
- 17.0 When specified as galvanized, the studs, m/c bolts and nuts shall be 'hot dip zinc coated' in accordance with requirements of 'class C' of 'ASTM A 153'. As an alternative, electro-galvanizing as per IS 1573, 'Service Grade Number 2' is also acceptable.
- 18.0 All Stud Bolts of Bolt diameter size 1" and above shall be provided with three nuts irrespective of whatever has been specified elsewhere in the MR.

# SPECIFICATION FOR PIPING FABRICATION AND ERECTION

SPECIFICATION NO.: MEC/S/05/21/06



(OIL & GAS SBU) MECON LIMITED DELHI 110 092

Page 340 of 644

OIL & GAS SBU	DOCUMENT NO. MEC/S/05/21/06	Page 1 of 1 REVISION : 0 EDITION : 1
IPING FABRICATION AND ERECTION CRIPTION ERAL PE S FOR WORK RICATION CTION DING CTION ECTION ECTION FECTIVE COATING SHING ROSTATIC TESTING EXURE-1 - WELDING SPECIFIC EXHIBIT-A - ELECTRODE EXHIBIT-A - ELECTRODE EXHIBIT-A - ELECTRODE	DOCUMENT NO. MEC/S/05/21/06	Page 1 of 1 REVISION : 0 EDITION : 1
IPING FABRICATION AND ERECTION CRIPTION ERAL PE S FOR WORK RICATION CTION DING CTION ECTION ECTION FECTIVE COATING FING ROSTATIC TESTING EXURE-1 - WELDING SPECIFIC EXHIBIT-A - ELECTRODE EXHIBIT-A - ELECTRODE EXHIBIT-A - STRESS REL	MEC/S/05/21/06 TENTS CATION QUALIFICATIONS TEST REC IEF HEAT TREATMENT PRO	REVISION : 0 EDITION : 1
ERECTION CRIPTION ERAL PE S FOR WORK RICATION CTION CTION DING CTION ECTION ECTION FECTIVE COATING SHING ROSTATIC TESTING EXURE-1 - WELDING SPECIFIC EXHIBIT-A - ELECTRODE EXHIBIT-B - STRESS REL EXHIBIT-B - STRESS REL	TENTS CATION QUALIFICATIONS TEST REC IEF HEAT TREATMENT PRO	EDITION : 1
CRIPTION ERAL PE S FOR WORK RICATION CTION DING CTION ECTION ECTION FECTIVE COATING FROSTATIC TESTING EXURE-1 - WELDING SPECIFIC EXHIBIT-A - ELECTRODE EXHIBIT-A - ELECTRODE EXHIBIT-A - STRESS REL EXHIBIT-B - STRESS REL	TENTS CATION QUALIFICATIONS TEST REC IEF HEAT TREATMENT PRO	CORD CEDURE SPECIFICATIO
ERAL PE S FOR WORK RICATION CTION DING CTION ECTION ECTION FECTIVE COATING SHING ROSTATIC TESTING EXURE-1 - WELDING SPECIFIC EXHIBIT-A - ELECTRODE EXHIBIT-B - STRESS REL EXHIBIT-B - STRESS REL	CATION QUALIFICATIONS TEST REC IEF HEAT TREATMENT PRO	CORD CEDURE SPECIFICATIO
ERAL PE S FOR WORK RICATION CTION DING CTION ECTION FECTIVE COATING SHING ROSTATIC TESTING EXURE-1 - WELDING SPECIFIC EXHIBIT-A - ELECTRODE EXHIBIT-B - STRESS REL EXHIBIT-B - STRESS REL	CATION QUALIFICATIONS TEST REC IEF HEAT TREATMENT PRO	CORD CEDURE SPECIFICATIO
PE S FOR WORK RICATION CTION DING CTION ECTION FECTIVE COATING SHING ROSTATIC TESTING EXURE-1 - WELDING SPECIFIC EXHIBIT-A - ELECTRODE EXHIBIT-B - STRESS REL EXHIBIT-B - STRESS REL	CATION QUALIFICATIONS TEST REC IEF HEAT TREATMENT PRO	CORD CEDURE SPECIFICATIO
S FOR WORK RICATION CTION DING CTION ECTION FECTIVE COATING CHING ROSTATIC TESTING EXURE-1 - WELDING SPECIFIC EXHIBIT-A - ELECTRODE EXHIBIT-B - STRESS REL EXHIBIT-B - STRESS REL	CATION QUALIFICATIONS TEST REC IEF HEAT TREATMENT PROD	CORD CEDURE SPECIFICATIO
RICATION CTION DING CTION ECTION FECTIVE COATING CHING ROSTATIC TESTING EXURE-1 - WELDING SPECIFIC EXHIBIT-A - ELECTRODE EXHIBIT-B - STRESS REL EXHIBIT-B - STRESS REL	CATION QUALIFICATIONS TEST REC IEF HEAT TREATMENT PRO	CORD CEDURE SPECIFICATIO
TION DING CTION ECTION FECTIVE COATING SHING ROSTATIC TESTING EXURE-1 - WELDING SPECIFIC EXHIBIT-A - ELECTRODE EXHIBIT-B - STRESS REL EXHIBIT-B - STRESS REL	CATION QUALIFICATIONS TEST REC IEF HEAT TREATMENT PRO	CORD CEDURE SPECIFICATIO
DING CTION ECTION FECTIVE COATING SHING ROSTATIC TESTING EXURE-1 - WELDING SPECIFIC EXHIBIT-A - ELECTRODE EXHIBIT-B - STRESS REL	CATION QUALIFICATIONS TEST REC IEF HEAT TREATMENT PRO	CORD CEDURE SPECIFICATIO
ECTION ECTION FECTIVE COATING SHING ROSTATIC TESTING EXURE-1 - WELDING SPECIFIC EXHIBIT-A - ELECTRODE EXHIBIT-B - STRESS REL EXHIBIT-B - STRESS REL	CATION QUALIFICATIONS TEST REC IEF HEAT TREATMENT PRO	CORD CEDURE SPECIFICATIO
ECTION FECTIVE COATING HING ROSTATIC TESTING EXURE-1 - WELDING SPECIFIC EXHIBIT-A - ELECTRODE EXHIBIT-B - STRESS REL	CATION QUALIFICATIONS TEST REC IEF HEAT TREATMENT PRO	CORD CEDURE SPECIFICATIO
ECTIVE COATING HING ROSTATIC TESTING EXURE-1 - WELDING SPECIFIC EXHIBIT-A - ELECTRODE EXHIBIT-B - STRESS REL	CATION QUALIFICATIONS TEST REC IEF HEAT TREATMENT PRO	CORD CEDURE SPECIFICATIO
EXURE-1 - WELDING SPECIFIC EXURE-1 - WELDING SPECIFIC EXHIBIT-A - ELECTRODE EXHIBIT-B - STRESS REL	CATION QUALIFICATIONS TEST REC IEF HEAT TREATMENT PRO	CORD CEDURE SPECIFICATIO
EXURE-1 - WELDING SPECIFIC EXHIBIT-A - ELECTRODE EXHIBIT-B - STRESS REL	CATION QUALIFICATIONS TEST REG IEF HEAT TREATMENT PRO	CORD CEDURE SPECIFICATIO
EXURE-1 - WELDING SPECIFIC EXHIBIT-A - ELECTRODE EXHIBIT-B - STRESS REL	CATION QUALIFICATIONS TEST REG IEF HEAT TREATMENT PROP DESCEDUES CREATED AT O	CORD CEDURE SPECIFICATIO
EXHIBIT-A - ELECTRODE EXHIBIT-B - STRESS REL	QUALIFICATIONS TEST REC IEF HEAT TREATMENT PRO	Cord Cedure Specificatio
EXHIBIT-C - STANDARD EXHIBIT-D - COUPON TE EXHIBIT-E - WELDER'S II EXHIBIT-F - RADIOGRAP EXHIBIT-G - WELDING SI	PROCEDURE SPECIFICATION ST RECORD DENTIFICATION CARD HIC PROCEDURE FOR PIPE PECIFICATION CHART	N NO.
EXURE-2 - DESTRUCTIVE TEST	ING OF WELDED JOINT - BL	JTT WELDS
Exure-3 - Ultrasonic Inspe	CTION	
EXURE-4 - RADIOGRAPHY		
CHECKED BY:	APPROVED BY	Y: ISSUE DA
		I
	EXURE-4 - RADIOGRAPHY CHECKED BY:	EXURE-4 - RADIOGRAPHY CHECKED BY: APPROVED B

MECON LIMITED REGD. OFF: RANCHI	STANDARD TECHNICAL	STANDARD TECHNICAL SPECIFICATION		
834002	OIL & GAS SBU	, DELHI	मुत्र मेकान 9001:2000 Contract	
		DOCUMENT NO.	Page 1 of 1	
TITLE	PIPING FABRICATION AND	MEC/S/05/21/06	REVISION : 0	
	ERECTION		EDITION: 1	

# **AMENDMENT STATUS**

SI. No.	Clause / Paragraph / Annexure / Exhibit / Drawing Amended	Page No.	Revision	Date	By (Name)	Verified (Name)

MECON LIMITED	STANDARD TECHNICAL	SPECIFICATION	
834002	OIL & GAS SBU, DELHI		येत्र मेकान मेकान 9007:2000 Californi
		DOCUMENT NO.	Page 1 of 61
TITLE	PIPING FABRICATION AND	MEC/S/05/21/06	REVISION: 0
	ERECTION		EDITION: 1

#### 1.0 GENERAL

This specification covers general requirements of fabrication and erection of aboveground and trench piping systems at site. The specification covers the scope of work of contractor, basis of work to be carried out by contractor and standards, specifications and normal practice to be followed during fabrication and erection by the contractor.

# 2.0 SCOPE

Generally the scope of work of contractor shall include the following :

- 2.1 Transportation of required piping materials, pipe support and all other necessary piping materials from Owner's storage point or contractor's storage point (in case of contractor's scope of supply) to work site / shop including raising store requisitions for issue of materials in the prescribed format & maintaining an account of the materials received from Owner's stores.
- 2.1.1 Piping materials include the following but not limited to the same.
  - a. Pipes (All sizes and schedule)
  - b. Flanges (All sizes, types & Pressure ratings).
  - c. Fittings (All sizes, types and schedule)
  - d. Valves (All sizes, types and Ratings)
  - e. Gaskets (All sizes, types & Ratings)
  - f. Bolts, Nuts or M/C Bolts (All types)
  - g. Expansion Joint / Bellows (All types)
  - h. Specialty items like online filters, ejectors, sample coolers, steam traps, strainers, air traps etc.
  - i. Online instruments like control valve, orifice flange, rotameter, safety valves etc.
- 2.2 Shop & field fabrication and erection of piping in accordance with documents listed under Cl. 3.0 i.e. 'BASIS OF WORK' including erection of all piping materials enumerated above.
- 2.3 Fabrication and erection of pipe supports like shoe, saddle, guide, stops, anchors, clips, cradles, hangers, turn buckles, supporting fixtures, bracket cantilevers, struts, teeposts including erection of spring supports and sway braces.
- 2.4 Fabrication
- 2.4.1 Fabrication of piping specials like special radius bends, reducers, mitres etc.

- 2.4.2 Fabrication of plain and threaded nipples from pipes as required during erection.
- 2.4.3 Fabrication of swage nipples as and when required.
- 2.4.4 Fabrication of odd angle elbow like 60°, 30° or any other angle from 90/45° elbows as and when required.
- 2.4.5 Fabrication of flange, reducing flange, blind flange, spectacle blinds as and when required.
- 2.4.6 Fabrication of stub-in connection with or without reinforcement.
- 2.4.7 Grinding of edges of pipes, fittings, flanges etc. to match mating edges of uneven / different thickness wherever required.
- 2.5 Modifications like providing additional cleats, extension of stem of valve, locking arrangement of valves etc. as and when required.
- 2.6 Preparation of Isometrics, bill of materials, supporting details of all NON-IBR lines upto 2-1/2" within the unit battery limit and get subsequent approval from Engineer-in-Charge as and when called for.
- 2.7 Obtaining approval for drawings prepared by contractor from statutory authority, if required.
- 2.8 Spun concrete lining of the inside of pipes 3" NB & above including fittings and flanges as required in accordance with specification.
- 2.9 Rubber lining inside pipes, fittings, flanges as and when required, in accordance with specification.
- 2.10 Radiography, stress relieving, dye penetration, magnetic particle test etc. as required in specification.
- 2.11 Performing PMI using alloy analysers as per 'Standard Specification for Positive Material Identification at Construction Sites, 6-82-0002'.
- 2.12 Casting of concrete pedestals and fabrication & erection of small structures for pipe supports including supply of necessary materials.
- 2.13 Providing insert plates from concrete structures and repair of platform gratings around pipe openings.
- 2.14 Making material reconciliation statement and return of Owner's supply left over materials to Owner's storage.
- 2.15 Flushing and testing of all piping systems as per standard specification for inspection, flushing and testing of piping systems (Specification No. MEC/S/05/21/11).

MECON LIMITED REGD. OFF: RANCHI	STANDARD TECHNICAL SPECIFICATION		
834002	OIL & GAS SBU, DELHI		येत्र मिकान मेकान मेकार:2000 Contract
		DOCUMENT NO.	Page 3 of 61
TITLE	PIPING FABRICATION AND	MEC/S/05/21/06	REVISION: 0
	ERECTION		EDITION: 1

# 3.0 BASIS FOR WORK

- 3.1 The complete piping work shall be carried out in accordance with the following
- 3.1.1 "Approved for Construction" drawings and sketches issued by MECON to the Contractor Plans and/or Isometrics.
- 3.1.2 "Approved for Construction" drawings and sketches issued by Turn-key bidders to the Contractor Plans and/or Isometrics.
- 3.1.3 Approved Process licensor's standards and specifications.
- 3.1.4 Drawings, sketches and documents prepared by contractor duly approved by Engineer-in-Charge' (such as isometrics and offsite piping etc.)
- 3.1.5 Approved construction job procedures prepared by Contractor as stipulated in 2.16
- 3.1.6 MECON specifications/documents as below :
  - a. Process and Instrument Diagram.
  - b. Piping Materials Specification
  - c. Piping support standards.
  - d. Line list / Number
  - e. Piping support index.
  - f. Standard specification of NDT Requirement of Piping
  - g. Welding specification charts for piping classes.
  - h. Standard specification for Pressure Testing of Erected Piping System.
  - i. Welding specification for fabrication of piping
  - j. Any other MECON or OTHER specifications attached with Piping Material Specification or special condition of contract.
  - k. Procedure for storage, preservation and positive identification of materials Contractors works / stores.
- 3.1.7 Following codes, standards and regulations
  - ASME B 31.3 : Process Piping
  - ASME Sec. VIII : Code for unfired pressure vessel.
  - c. IBR Regulations

a.

b.

Page 345 of 644

MECON LIMITED	STANDARD TECHNICAL SPECIFICATION OIL & GAS SBU, DELHI		
834002			गठ मेकान 10 मेकान 10 मेकान
		DOCUMENT NO.	Page 4 of 61
TITLE	PIPING FABRICATION AND	MEC/S/05/21/06	REVISION: 0
	ERECTION		EDITION: 1
d.	IS:823 :	Code for procedure for	Manual Metal Are Welding
e.	NACE Std.	Code for Sour Service	es material requirements

NACE Std. : Code for Sour Services material requirements MR.

Note : All codes referred shall be latest edition.

#### 3.2 Deviations

Where a deviation from the "Basis of Work" and approved job procedure described above is required or where the basis of work does not cover a particular situation, the matter shall be brought to the notice of Engineer - in - Charge and the work carried out only after obtaining written approval from him in each case.

# 4.0 FABRICATION

#### 4.1 Piping Material

Pipe, pipe fittings, flanges, valves, gaskets, studs bolts etc. used in a given piping system shall be strictly as per the "Piping Material Specification" for the "Pipe Class" specified for that system. To ensure the above requirement, all piping material supplied by the Owner / Contractor shall have proper identification marks as per relevant standards / MECON's specifications / Licensors specification. Contractor shall provide identification marks on left over pipe lengths wherever marked up pipe lengths have been fabricated / erected. Material traceability is to be maintained for AS., S.S., NACE, LTCS, material for Hydrogen service and other exotic materials by way of transferring heat number, etc. (hard punching) as per approved procedure. This shall be in addition to colour coding for all piping materials to avoid mix-up.

#### 4.2 **Fabrication**

- 4.2.1 All fabrication shall be carried out in accordance with piping general arrangement drawings, (prepared by CONTRACTOR and approved by COMPANY) including this specification and codes as specified in section 2.0.
- 4.2.2 CONTRACTOR shall be responsible for working to the exact dimensions as per the approved drawings. Dimensional tolerances to be adopted during implementation of fabrication work shall be as per attached sketch "TOLERANCES FOR FABRICATION".
- 4.2.3 Flange bolt holes shall generally straddle the established centre lines unless other orientation is required and as called out in approved drawings.
- 4.2.4 Threading shall be NPT to ANSI B 1.20.1. Threading shall preferably be done after bending, forging or heat treatment operation. However if it is not possible, precaution shall be taken to protect threading against deformation. Thread shall be clean cut with no burrs or stripping. Dies shall be new, sharp and properly designed for piping material. Ends shall be reamed to remove burrs.

MECON LIMITED         STANDARD TECHNICAL SPECIFICA           REGD, OFF: RANCHI		SPECIFICATION	
834002	OIL & GAS SBU, DELHI		मुकान 10,0001:2000 Control
		DOCUMENT NO.	Page 5 of 61
TITLE	PIPING FABRICATION AND	MEC/S/05/21/06	REVISION: 0
	ERECTION		EDITION: 1

- 4.2.5 All threaded joints shall be aligned properly. The pipe entering unions shall be true to centrelines so as to avoid forcing of union coupling during make up. Damaged threads shall be cut from the end of run and the pipe shall be rethreaded.
- 4.2.6 Immediately before testing the piping, all threads of pipe and fittings shall be thoroughly cleared of cuttings, fuel oil or other foreign matter. The male threads shall be sealed with thread sealant and the piping made up sufficiently for the thread to seize. Sealant shall be teflon tape.
- 4.2.7 Seal welding of threaded connections when specified shall include the first block valve, cover all threads. The joint shall be cleaned of all cutting oil and other foreign material and made up dry to full thread engagement. Instrument threaded connections which are frequently subjected to testing and maintenance shall not be seal welded.
- 4.2.8 All threaded connections shall be protected from rusting by applying greases or oil when in operating condition.
- 4.2.9 When socket weld fittings or valves are used, pipe shall be spaced approximately 1/16" to avoid bottoming which could result in excessive weld stress.
- 4.2.10 Where the ends of the piping components being welded have an internal surface misalignment exceeding 1.6mm, the wall of the component extending internally shall be trimmed by machining so that the adjoining internal surface will approximately flush.

For the purpose of common understanding the construction job procedure, to be submitted by the contractor, shall include proposal for

- Maximizing prefabrication, inspection and testing at fabrication shop with minimum field joints.
- Positive material identification, handling, storage & preservation.

#### 4.3 Dimensional Tolerances

Dimensional tolerances for piping fabrication shall be as per MECON Standard Specification. The Contractor shall be responsible for working to the dimensions shown on the drawings. However, the Contractor shall bear in mind that there may be variations between the dimensions shown in the drawing and those actually existing at site due to minor variations in the location of equipments, inserts, structures etc. To take care of these variations "Field Welds" shall be provided during piping fabrication. An extra pipe length of 100 mm over and above the dimensions indicated in the drawing may be left on one side of the pipe at each of the field welds. During erection, the pipe end with extra length at each field weld, shall be cut to obtain the actual dimension occurring at site. Isometrics, if supplied may have the field welds marked on them. However, it is the responsibility of the Contractor to provide adequate number of field welds. In any case no extra claims will be entertained from the Contractor on this account. Wherever errors / omissions occur in drawings and Bills of Materials it shall be the Contractor's responsibility to notify the Engineer-in-Charge prior to fabrication or erection.

MECON LIMITED	MECON LIMITED       STANDARD TECHNICAL SPECIFICATION         REGD. OFF: RANCHI       OIL & GAS SBU, DELHI		
834002			मुकान 10,0001:2000 Cantra
		DOCUMENT NO.	Page 6 of 61
TITLE	PIPING FABRICATION AND	MEC/S/05/21/06	REVISION: 0
	ERECTION		EDITION: 1

# 4.4 IBR Piping

- 4.4.1 Contractor shall be supplied generally with all drawings for steam piping falling under the purview of Indian Boiler Regulations duly approved by Boiler Inspectorate. The Contractor shall carry out the fabrications, erection and testing of this piping as per requirements of Indian Boiler Regulations and to the entire satisfaction of the local Boiler Inspector. The Contractor shall also get the approval of IBR inspector for all fabrication and testing done by him at his own cost. All certificates of approval shall be in proper IBR forms.
- 4.4.2 Approval of boiler inspector on the drawings prepared by the contractor shall be obtained by the contractor at his own cost.

#### 4.5 Pipe Joints

The piping class of each line specifies the type of pipe joints to be adopted. In general, joining of lines 2" and above in process and utility piping shall be accomplished by butt welds. Joining of lines 1-1/2" and below shall be by socket welding / butt welding / threaded joints as specified in "Piping Material Specifications". However, in piping 1-1/2" and below where socket welding/ threaded joints are specified butt - welds may be used with the approval of Engineer-in-Charge for pipe to pipe joining in long runs of piping. This is only applicable for non-galvanized piping without lining.

Flange joints shall be used at connections to Vessels, Equipment's, Valves and where required for ease of erection and maintenance as indicated in drawings.

#### 4.6 Butt Welded and Socket Welded Piping

End preparation, alignment and fit-up of pipe pieces to be welded, welding, pre-heat, postheating and heat treatment shall be as described in the welding specification and NDT specification.

#### 4.7 Screwed Piping

In general, Galvanized piping shall have threads as per IS:554 or ANSI B 2.1 NPT as required to match threads on fittings, valves etc. All other piping shall have threads as per ANSI B 2.1, tapered unless specified otherwise.

Threads shall be clean cut, without any burrs or stripping and the ends shall be reamed. Threading of pipes shall be done preferably after bending, forging or heat treating operations. If this is not possible, threads shall be gauge checked and chased after welding heat treatment etc.

During assembly of threaded joints, all threads of pipes and fittings shall be thoroughly cleaned of cuttings, dirt, oil or any other foreign matter. The male threads shall be coated with thread sealant and the joint tightened sufficiently for the threads to seize and give a leakproof joint.

MECON LIMITED REGD. OFF: RANCHI	MECON LIMITED         STANDARD TECHNICAL SPECIFICATION           REGD. OFF: RANCHI		
834002	OIL & GAS SBU, DELHI		येत्र मिकान मेकान 9007:2000 Contrain
		DOCUMENT NO.	Page 7 of 61
TITLE	PIPING FABRICATION AND	MEC/S/05/21/06	REVISION: 0
	ERECTION		EDITION: 1

Threaded joints to be seal-welded shall be cleaned of all foreign matter, including sealant and made up to full thread engagement before seal welding.

#### 4.8 Flange Connections

All flange facings shall be true and perpendicular to the axis of pipe to which they are attached. Flanged bolt holes shall straddle the normal centerlines unless different orientation is shown in the drawing.

Wherever a spectacle blind is to be provided, drilling and tapping for the jack screws in the flange, shall be done before welding it to the pipe.

#### 4.9 Branch Connections

Branch connections shall be as indicated in the piping material specifications. For end preparation, alignment, spacing, fit-up and welding of branch connections refer welding specifications. Templates shall be used wherever required to ensure accurate cutting and proper fit-up.

For all branch connections accomplished either by pipe to pipe connections or by using forged tees the rates quoted for piping shall be inclusive of this work.

Reinforcement pads shall be provided wherever indicated in drawings/ specifications etc.

#### 4.10 Bending

Bending shall be as per ASME B31.3 except that corrugated or creased bends shall not be used.

Cold bends for lines 1-1/2" and below, with a bend radius of 5 times the nominal diameter shall be used as required in place of elbows wherever allowed by piping specifications. Bending of pipes 2" and above may be required in some cases like that for headers around heaters, reactors etc.

The completed bend shall have a smooth surface, free from cracks, buckles, wrinkles, bulges, flat spots and other serious defects. They shall be true to dimensions. The flattening of a bend, as measured by the difference between the maximum and minimum diameters at any cross-section, shall not exceed 8% and 3% of the nominal outside diameter, for internal and external pressure respectively.

# 4.11 Forging and forming

Forging and forming of small bore fittings, like reducing nipples for piping 1-1/2" and below, shall be as per ASME B 31.3.

MECON LIMITED         STANDARD TECHNICAL SPECIFIC           REGD. OFF: RANCHI		SPECIFICATION	
834002	OIL & GAS SBU, DELHI		मुकान 10,0001:2000 Cantra
		DOCUMENT NO.	Page 8 of 61
TITLE	PIPING FABRICATION AND	MEC/S/05/21/06	REVISION: 0
	ERECTION		EDITION: 1

#### 4.12 Mitre Bends and Fabricated Reducers

The specific application of welded mitre bends and fabrication reducers shall be governed by the Piping Material Specifications. Generally all 90 deg. mitres shall be 4-piece 3-weld type and 45 deg. mitres shall be 3-piece 2-weld type as per MECON Standard unless otherwise specified. Reducers shall be fabricated as per directions of Engineer-in-Charge. The radiographic requirements shall be as per Material Specifications for process and utility systems and NDT Specification for steam piping under IBR, radiographic requirements of IBR shall be complied with.

# 4.13 Cutting and Trimming of Standard Fittings & Pipes

Components like pipes, elbows, couplings, half-couplings etc. shall be cut / trimmed / edge prepared wherever required to meet fabrication and erection requirements, as per drawings and instructions of Engineer-in-Charge. Nipples as required shall be prepared from straight length piping.

# 4.14 Galvanised Piping

Galvanised carbon steel piping shall be completely cold worked, so as not to damage galvanised surfaces. This piping involves only threaded joints and additional external threading on pipes may be required to be done as per requirement.

# 4.15 Jacketed Piping

The Jacketing shall be done in accordance with MECON Specification or Licensors specification as suggested in material specification or special condition of contract.

Pre-assembly of jacketed elements to the maximum extent possible shall be accomplished at shop by Contractor. Position of jumpover and nozzles on the jacket pipes, fittings etc. shall be marked according to pipe disposition and those shall be prefabricated to avoid damaging of inner pipe and obstruction of jacket space. However, valves, flow glasses, in line instruments or even fittings shall be supplied as jacketed.

# 4.16 Shop Fabrication / Prefabrication

The purpose of shop fabrication or pre-fabrication is to minimise work during erection to the extent possible. Piping spool, after fabrication, shall be stacked with proper identification marks, so as facilitate their withdrawal at any time during erection. During this period all flange (gasket contact faces) and threads shall be adequately fabricated by coating with a removable rust preventive. Care shall also be taken to avoid any physical damage to flange faces and threads.

#### 4.17 Miscellaneous

4.17.1 Contractor shall fabricate miscellaneous elements like flash pot, seal pot, sample cooler, supporting elements like turn buckles, extension of spindles and interlocking arrangement of valves, operating platforms as required by Engineer-in-Charge.

MECON LIMITED REGD. OFF: RANCHI	STANDARD TECHNICAL SPECIFICATION OIL & GAS SBU, DELHI		मिकान 10 अवता:2000 Contrib
834002			
		DOCUMENT NO.	Page 9 of 61
TITLE	PIPING FABRICATION AND	MEC/S/05/21/06	<b>REVISION: 0</b>
	ERECTION		EDITION: 1

#### 4.17.2 Spun Concrete Lining

The work of inside spun concrete lining of pipes and specials of diameter 3" and above shall be done as per material specifications and special condition contract.

#### 4.17.3 Fabrication of pipes from plate

Pipes shall be fabricated at site as and when required as per the specifications attached and the actual Piping Material Specification.

# 5.0 ERECTION

# 5.1 Cleaning of Piping before Erection

Before erection all pre-fabricated spool pieces, pipes, fittings etc. shall be cleaned inside and outside by suitable means. The cleaning process shall include removal of all foreign matter such as scale, sand, weld spatter chips etc. by wire brushes, cleaning tools etc. and blowing with compressed air/or flushing out with water. Special cleaning requirements for some services, if any shall be as specified in the piping material specification or isometric or line list. S.S jacketed piping requiring pickling shall be pickled to remove oxidation and discolouring due to welding.

# 5.2 Piping Routing

No deviations from the piping route indicated in drawings shall be permitted without the consent of Engineer- in-Charge.

Pipe to pipe, pipe to structure / equipments distances / clearances as shown in the drawings shall be strictly followed as these clearances may be required for the free expansion of piping / equipment. No deviations from these clearances shall be permissible without the approval of Engineer-in-Charge.

In case of fouling of a line with other piping, structure, equipment etc. the matter shall be brought to the notice of Engineer-in-Charge and corrective action shall be taken as per his instructions.

# 5.3 Cold Pull

Wherever cold pull is specified, the Contractor shall maintain the necessary gap, as indicated in the drawing. Confirmation in writing shall be obtained by the Contractor from the Engineerin-Charge, certifying that the gap between the pipes is as indicated in the drawing, before drawing the cold pull. Stress relieving shall be performed before removing the gadgets for cold pulling.

MECON LIMITED STANDARD TECHNICAL SPECIFICATION REGD. OFF: RANCHI			
834002	OIL & GAS SBU, DELHI		ये मेकान 10 4007:2000 Contract
		DOCUMENT NO.	Page 10 of 61
TITLE	PIPING FABRICATION AND	MEC/S/05/21/06	REVISION: 0
	ERECTION		EDITION: 1

#### 5.4 Slopes

Slopes specified for various lines in the drawings / P&ID shall be maintained by the Contractor. Corrective action shall be taken by the Contractor in consultation with Engineer-in-Charge wherever the Contractor is not able to maintain the specified slope.

# 5.5 Expansion Joints / Bellows

Installation of Expansion Joints/Bellows shall be as follows:

- All Expansion joints / Bellows shall be installed in accordance with the specification and installation drawings, supplied to the Contractor.
- ➢ Upon receipt, the Contractor shall remove the Expansion Joints/ Bellows from the case(s) and check for any damage occurred during transit.
- > The Contractor shall bring to the notice of the Engineer-in-Charge any damage done to the bellows / corrugations, hinges, tie-rods, flanges / weld ends etc.
- ➢ Each Expansion Joint / Bellow shall be blown free of dust / foreign matter with compressed air or cleaned with a piece of cloth.
- For handling and installation of Expansion Joints, great care shall be taken while aligning. An Expansion Joints shall never be slinged from bellows corrugations / external shrouds, tie / rods, angles.
- An Expansion Joints / Bellow shall preferably be slinged from the end pipes / flanges or on the middle pipe.
- All Expansion Joints shall be delivered to the Contractor at "Installation length", maintained by means of shipping rods, angles welded to the flanges or weld ends or by wooden or metallic stops.
- Expansion Joints stop blocks shall be carefully removed after hydrostatic testing. Angles welded to the flanges or weld ends shall be trimmed by saw as per manufacturer's instructions and the flanges or weld ends shall be ground smooth.
- > The pipe ends in which the Expansion Joint is to be installed shall be perfectly aligned or shall have specified lateral deflection as noted on the relevant drawings.
- > The pipe ends / flanges shall be spaced at a distance specified in the drawings.
- The Expansion Joint shall be placed between the mating pipe ends / flanges and shall be tack welded/bolted. The mating pipes shall again be checked for correct alignment.
- Butt-welding shall be carried out at each end of the expansion joint. For flanged Expansion Joint, the mating flanges shall be bolted.

MECON LIMITED REGD. OFF: RANCHI 834002		STANDARD TECHNICAL	SPECIFICATION		
		OIL & GAS SBU, DELHI		मिकान 93 900 1:2000 Control	
			DOCUMENT NO.	Page 11 of 61	
TITLE	E	PIPING FABRICATION AND	MEC/S/05/21/06	REVISION: 0	
		ERECTION		EDITION: 1	
		After the Expansion Joint is installed and Expansion Joints are in correct and guided.	ed the Contractor shall ens ct alignment and that the	sure that the mating pip pipes are well support	
	$\triangleright$	The Expansion Joint shall not have parallelism of restraining rings or b	any lateral deflection. The ellows convolutions.	e Contractor shall mainta	
	$\triangleright$	Precautions			
		<ul> <li>For carrying out welding, earthing lead shall not be attached with the Expansio Joint.</li> </ul>			
		• The Expansion bellow shall be protected from arc weld spot and welding spatter.			
	<ul> <li>Hydrostatic Testing of the system having Expansion Joint shall be shipping lugs in position. These lugs shall be removed after testing is over.</li> </ul>				
5.6	Flange	e Connections			
	While f the flar any str connec connec the fin approve	itting up mating flanges, care shall langes for trueness, so that faces of tresses in the pipes and the equipations to pumps, turbines, comprisions to these equipments shall be al alignment of the equipment is al of Engineer-in-Charge.	be exercised to properly ali the flanges can be pulled t ment nozzles. Extra care ressors, cold boxes, air checked for misalignment over. The joint shall be	gn the pipes and to che ogether, without induci shall be taken for flan coolers etc. The flan c, excessive gap etc. af made up after obtaini	
	Tempo compre any for	rary protective covers shall be reta essors and other similar equipments eign material from entering these e	ained on all flange connec s, until the piping is finally quipments.	tions of pumps, turbing connected, so as to ave	
	The as flange sequen	sembly of a flange joint shall be c faces is uniformly compressed. To ce. All bolts shall extend completely	lone in such a way that t achieve this the bolts shal hough their nuts but no	he gasket between the I be tightened in a prop t more than 1/4".	
	Steel to after br	o C.I. flange joints shall be made u ringing flange flush with gaskets wit	p with extreme care, tigh h accurate pattern and late	tening the bolts uniforr eral alignment.	
5.7	Vents	and Drains			

High point vents and low point drains shall be provided as per the instructions of Engineer-in-Charge, even if these are not shown in the drawings. The details of vents and drains shall be as per piping material specifications / job standards.

MECON LIMITED	CON LIMITED       STANDARD TECHNICAL SPECIFICATION         GD. OFF: RANCHI       OIL & GAS SBU, DELHI		
834002			मिकान 9001:2000 Control
		DOCUMENT NO.	Page 12 of 61
TITLE	PIPING FABRICATION AND	MEC/S/05/21/06	<b>REVISION: 0</b>
	ERECTION		EDITION: 1

#### 5.8 Valves

Valves shall be installed with spindle / actuator orientation / position as shown in the layout drawings. In case of any difficulty in doing this or if the spindle orientation / position is not shown in the drawings, the Engineer-in-Charge shall be consulted and work done as per his instructions. Care shall be exercised to ensure that globe valves, check valves, and other unidirectional valves are installed with the "Flow direction arrow "on the valve body pointing in the correct direction. If the direction of the arrow is not marked on such valves, this shall be done in the presence of Engineer-in-Charge before installation.

Fabrication of stem extensions, locking arrangements and interlocking arrangements of valves (if called for), shall be carried out as per drawings / instructions of Engineer-in-Charge.

# 5.9 Instruments

Installation of in-line instruments such as restriction orifices, control valves, safety valves, relief valves, rotameters, orifice flange assembly, venturimeters, flowmeters etc. shall form a part of piping erection work.

Fabrication and erection of piping upto first block valve / nozzle / flange for installation of offline Instruments for measurement of level, pressure, temperature, flow etc. shall also form part of piping construction work. The limits of piping and instrumentation work will be shown in drawings / standards / specifications. Orientations / locations of take-offs for temperature, pressure, flow, level connections etc. shown in drawings shall be maintained.

Flushing and testing of piping systems which include instruments mentioned above and the precautions to be taken are covered in flushing, testing and inspection of piping. Care shall be exercised and adequate precautions taken to avoid damage and entry foreign matter into instruments during transportation, installation, testing etc.

#### 5.10 Line Mounted Eqipments / Items

Installation of line mounted items like filters, strainers, steam traps, air traps, desuperheaters, ejectors, samples coolers, mixers, flame arrestors, sight glasses etc. including their supporting arrangements shall form part of piping erection work.

#### 5.11 Bolts and Nuts

The Contractor shall apply molycoat grease mixed with graphite powder (unless otherwise specified in piping classes) all bolts and nuts during storage, after erection and wherever flange connections are broken and made-up for any purpose whatsoever. The grease and graphite powder shall be supplied by the Contractor within the rates for piping work.

#### 5.12 Pipe Supports

Pipe supports are designed and located to effectively sustain the weight and thermal effects of the piping system and to prevent its vibrations. Location and design pipe supports will be shown in drawing for lines 2" NB. However, any extra supports desired by Engineer-in-Charge

MECON LIMITED REGD. OFF: RANCHI 834002		STANDARD TECHNICAL	SPECIFICATION	
		OIL & GAS SBU, DELHI		मिकान मिकान मिकार
			DOCUMENT NO.	Page 13 of 61
TITLE	Ξ	PIPING FABRICATION AND	MEC/S/05/21/06	REVISION: 0
		ERECTION		EDITION: 1
	shall al	so be installed.		
	No pipe	e shoe / cradle shall be offset unless	specifically shown in the o	drawings.
	Hanger move c	r rods shall be installed inclined in a during expansion.	direction opposite to the o	direction in which the pi
	Preset is over obtain setting Engine	pins of all spring supports shall be r . Springs shall be checked for the the correct positioning in cold cond in operating condition. The followin er-in-Charge and necessary confirma	removed only after hydrost range of movement and dition. These shall be sub ng points shall be checked ation in writing obtained ce	atic testing and insulation adjusted if necessary sequently adjusted to h after installation, with the ertifying that :
	-	All restraints have been installed co	rrectly.	
	-	Clearances have been maintained a	s per support drawings.	
	-	Insulation does not restrict thermal	expansion.	
	-	All temporary tack welds provided of	during erection have been	fully removed.
	-	All welded supports have been fully	v welded.	
6.0		WELDING		
		Welding of pipelines shall be done a	as per applicable codes an	d Annexure-1
7.0		ERECTION		
7.1		Pre-fabrication and Field Assen	nbly	
		Extent of pre-fabrication shall be p view the following :-	ourely at the discretion of	CONTRACTOR keeping
7.1.1		Field joint shall be decided by CON fabricated pieces to site.	TRACTOR keeping in view	the transportation of pr
7.1.2		There can be some variations arrangement drawings and those a the location of equipments, struct provided, permitting assembly and	in the dimensions and actually occurring at site o tures, cut out etc. Adeq erection of pipe work with	level appearing in the due to minor variations uate field joints shall out major modification.
7.2		Supporting		
		Location and design of pipe sup	pports shown in approve	d drawings and suppo

MECON LIMITED	STANDARD TECHNICAL SPECIFICATION OIL & GAS SBU, DELHI		
834002			मिकान मिकान मिकान
		DOCUMENT NO.	Page 14 of 61
TITLE	PIPING FABRICATION AND	MEC/S/05/21/06	REVISION: 0
	ERECTION		EDITION: 1
7.2.1	Supports shall be installed in such of a line.	a way that they do not co	ontribute to over stressi
7.2.2	Fabrication and erection of addition in COMPANY's view are required for out by CONTRACTOR at no extra co	nal supporting elements an or proper supporting of the ost.	d structural fixtures whi e system, shall be carri
7.2.3	All temporary supports, elements be removed after completion of wo	required for alignment, en rk.	ection and assembly sh
7.3	Equipment hook-up		
7.3.1	Prior to hook-up, the alignment ensure that no undue stresses shal	and trueness of flange fa I be induced in the system	aces shall be checked while hooking up.
8.0	<b>INSPECTION</b>		
8.1			
8.2	CONTRACTOR shall provide all fac of their inspection without any extr	ilities/ assistance to COMP a charge.	ANY for proper execut
8.3	All piping work shall be subjecte fabrication. CONTRACTOR shall sufficiently in advance, in order to e	d to inspection by COMF furnish to COMPANY de enable COMPANY to arrang	PANY at any time duri etailed work programi ge for inspection.
9.0	PROTECTIVE COATING		
9.1	All above ground piping system sh with specification for shop & field p	all be applied with protect ainting.	ive coating in accordar
9.2	All under ground portion of piping a CONTRACTOR shall prepared pro approval of COMPANY. Procedure a coating to be adopted. Coating of coating procedure. Total dry film t microns. Compatible primer and fir shall only be applied. Coating integ length of coated pipe work. Coating for design temperature.	system shall be coated wit cedure for epoxy paintin- shall include surface prepa- of pipes shall not comme chickness to be achieved s hish coat as recommended grity shall be checked by "In g to be supplied by CONT	h three layer P.E. coating g of burried pipeline aration, brand and type ence without approval shall not be less than 3 I by coating manufactu Holiday detector" over RACTOR shall be suita
9.3	Once the coating has been accepted In order to protect coated pipe fro for stone, rock and any other substances shall be removed befor ask for a 100mm padding of clear s hard soil area. No additional par admissible.	ed by COMPANY, backfiling m damage, the excavated hard substance detrimen re lowering the pipe in the sand under and above pipe syment on account of pa	operation can be started trench shall be examinated to coating. All sume trench. COMPANY me line in rocky or otherw adding shall however

MECON LIMITED	STANDARD TECHNICAL SPECIFICATION OIL & GAS SBU, DELHI		10 1000 Contract
834002			
		DOCUMENT NO.	Page 15 of 61
TITLE	PIPING FABRICATION AND	MEC/S/05/21/06	REVISION : 0
	ERECTION		EDITION: 1

# 10.0 **FLUSHING**

Completed piping systems shall be flushed by CONTRACTOR with fresh water, to clean the pipe of all dirt, debris, and foreign material. CONTRACTOR shall prepare a procedure for flushing of the system for approval of COMPANY. Flushing shall not be commenced without approval of flushing procedure.

- 10.1 CONTRACTOR shall perform all activities like dismantling and reinstalling of all strainers, in line instruments etc. before and after completion of flushing.
- 10.2 Flushing shall be considered as complete only after inspection and approval by COMPANY.
- 10.3 Disposal of muck and flushing media shall be arranged by CONTRACTOR as directed by COMPANY, in such a manner that it does not spoil the adjacent installation. CONTRACTOR shall obtain COMPANY approval regarding the place and method to be adopted for disposal of debris.
- 10.4 Record of flushing giving following details shall be submitted by CONTRACTOR to COMPANY for its approval and records :
  - a) Date of flushing
  - b) Identification of line : flushed-line number

# 11.0 HYDROSTATIC TESTING

- 11.1 Completed piping system as approved by COMPANY shall be hydrostatically tested in the presence of COMPANY. The general requirements of hydrostatic testing shall be in accordance with codes specified in section 2.0.
- 11.2 CONTRACTOR shall prepare hydrostatic test procedure based on specified codes. The hydrostatic test shall commence only after approval of procedure by COMPANY.
- 11.3 Piping system shall be hydrostatically tested to a pressure corresponding to 1.4 times the design pressure.
- 11.4 Fresh water shall be used as test media. CONTRACTOR shall locate the source of water supply and arrange for transportation of water to test site. CONTRACTOR shall arrange at his own cost the water analysis and confirm that water is suitable for testing. In case any corrosion inhibitor is to added, the same shall be done after approval of COMPANY.
- 11.5 Lines repaired subsequent to hydrostatic test shall be retested using the same procedure as originally adopted. However COMPANY may waive such retest in case of minor repairs by taking precautionary measures to ensure sound construction.

MECON LIMITED	STANDARD TECHNICAL SPECIFICATION			
834002	OIL & GAS SBU, DELHI		10 1-2000 Commit	
		DOCUMENT NO.	Page 16 of 61	
TITLE	PIPING FABRICATION AND ERECTION	MEC/S/05/21/06	REVISION: 0	
			EDITION: 1	
11.6	All equipment and instruments used for hydrostatic test shall be approved by COMPANY before start of tests.			
11.7	Pressure gauges shall be installed on line to measure test pressures. In case of longer lines two or more pressure gauges shall be installed as directed by COMPANY. One gauge shall be installed at the discharge of the pressurising pump. Pressure gauge used for hydrostatic testing shall be calibrated with dead weight tester in the presence of Engineer-in-charge. Range of pressure gauge shall generally be 1.5 times the test pressure.			
11.8	Orifice plates and restriction orifices shall not be installed until hydrostatic testing is completed. Temporary gaskets shall be used during testing.			
11.9	First block valve of pressure instruments shall be half open & plugged at the time of hydrostatic testing. Temperature connections shall be blanked off during testing.			
11.10	All equipments, in line instruments, relief valves shall be disconnected from piping system by means of blinds during testing. Control valves shall be replaced by spool pieces during testing.			
11.11	High point vents and low point drain required for testing in addition to those marked in the drawings shall be provided by CONTRACTOR at his own cost.			
11.12	All welded and screwed joints shall be kept clean for detecting leaks during testing.			
11.13	Test pressure shall be maintained long enough to facilitate complete inspection of the system. Minimum duration of test shall be 6 hours unless other wise specified. Pressurising equipment shall be isolated immediately after test pressure is attained.			
11.14	After successful completion of hydrostatic testing, the piping system shall be dewatered. All lines shall be completely dried using compressed air. CONTRACTOR shall make his own arrangement for supply of compressed air. Drying of lines shall be considered complete on approval by COMPANY.			
11.15	Test Records			
	The records in duplicate shall be prepared and submitted by CONTRACTOR as below			
	<ul> <li>a) Date of test</li> <li>b) Identification of pipe tested</li> <li>c) Test pressure</li> <li>d) Test results</li> <li>e) Signature of CONTRACTOR</li> <li>f) Approval signature by COM</li> </ul>	l - line number PANY.		

MECON LIMITED	STANDARD TECHNICAL SPECIFICATION OIL & GAS SBU, DELHI		Фортория 12000 Солония		
834002					
		DOCUMENT NO.	Page 17 of 61		
TITLE	PIPING FABRICATION AND	MEC/S/05/21/06	REVISION : 0		
	ERECTION		EDITION: 1		
			ANNEXURE		
	WELDING S	SPECIFICATION			
1.0	GENERAL				
	This specification shall be followed for the fabrication of all types of welded joints carbon steel above ground natural gas service piping systems.				
	The welded pipe joints shall include the following :				
	a) All line pipe joints of the longitudinal and circumferential butt welded.				
	b) Attachments of castings, forgings, flanges.				
	c) Welded manifold headers and other sub-assemblies				
	d) Welded branch connections with or without reinforcing pads.				
	e) Joints in welded/ fabrication piping components.				
	f) The attachments of smaller connections for vents drain drips and oth instrument tappings.				
	Any approval granted by the Engither the contractor of his responsibilities	ineer-in-charge or owner's i es & guarantees.	nspectors shall not relie		
1.1	Applicable Codes & Standards				
	All welding work, equipments for welding, heat treatment, other auxiliary functions ar the welding personnel shall be as per the requirements of the latest editions of the following approved standards and procedures :-				
	<ul><li>i) Code for gas transmission and distribution piping systems. ANSI B31.8.</li><li>ii) Code for petroleum refinery piping, ANSI B31.3.</li></ul>				
	In addition, the following codes/ specifications referred to in the relevant code of fabrication shall be followed for the welding/ brazing qualifications, consumable qualifications and non destructive test procedures.				
	<ul> <li>i) Standard for welding of pipelines and related facilities API-1104.</li> <li>ii) Material Specifications - Welding rods, electrodes and filler materials - ASME Sec IIC.</li> </ul>				
	<ul> <li>iii) Code for non destructive examination ASME Sec-V.</li> <li>iv) Qualification standard for welding and brazing procedure and welders, brazers, welding and brazing operators - ASME Sec-I</li> </ul>				

MECON LIMITED		STANDARD TECHNICAL SPECIFICATION			
834002	OIL & GAS SBU, DELHI		गुरु गुरुगन 10 3001:2000 Convint		
			DOCUMENT NO.	Page 18 of 61	
TITLE	PI	PING FABRICATION AND	MEC/S/05/21/06	REVISION : 0	
		ERECTION		EDITION: 1	
	In the specif bindin	e event of any difference du ication, over and above thos ig.	e to the additional requir e obligation as per codes,	ements mentioned in th this specification shall b	
1.2	Base Material				
	a) In general carbon steel is used in this plant. The details of material				
	h)	<ul> <li>specifications are given in the welding Specification Chart.</li> <li>The contractor shall provide the Manufacturer's test certificates for every hear of the materials supplied by him.</li> </ul>			
	5)				
1.3A	Filler Materials				
	a) The Contractor shall provide all the necessary welding electrodes, fille materials, etc. required for the execution of the work.				
	b)	b) The welding electrodes/ filler wires supplied by the Contractor shall conform t the class specified in the welding specification chart. The materials shall be o the make approved by the Engineer-in-charge.			
	c)	The electrode shall be suitable for the welding process recommended and bas metal used. Unless otherwise specified physical properties of the weld produced by a electrode recommended for the welding of a particular bas metal shall not be lower than the minimum valves specified for the base meta and shall correspond to the physical properties of the class of electrod adopted. The choice of electrode shall be made after conducting the require tests on the electrodes as per relevant standards, and shall be the sol prerogative of the Engineer-in-charge.			
	d)	d) Tungsten electrodes used shall conform to ASME Sec. II C SFA 5. specification. Thoriated Tungsten electrodes shall not be permitted due possible radiation hazard. Instead, ceriated Tungsten Electrodes (EWCe-2 equivalent) shall be used for GTA Welding.			
	e)	Electrode qualification test records should be submitted as per the <b>Exhibit</b> (attached) in respect of the electrodes tested by the contractor, for obtaini the approval of the Engineer-in-charge.			
	f)	The Contractor shall su Manufacturers giving detail for each batch of electrodes	bmit batch test certific s of physical and chemical s to be used.	ate from the electroc tests carried out by then	
	g)	All electrodes shall be purp prevent deterioration. The kept in holding ovens a Manufacture. Out-of-the o shall not exceed the limits electrodes shall be handled	chased in sealed containe electrodes removed from at temperature recommendent ven time of electrodes be recommended by the elect with care to avoid any dar	rs and stored properly to the containers shall be ended by the electroc efore they are consume trodes manufacturer. The mage to flux covering.	
MECON LIMITED		STANDARD TECHNICAL SPECIFICATION OIL & GAS SBU, DELHI			
---------------	-----------	----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------	----------------------------------------------------------------------------------------------------------------------------------------------------	----------------------------------------------------------------------------------------------------------------------------	--
834002				गुठा निकान 10 मकान 1000 Contract	
			DOCUMENT NO.	Page 19 of 61	
TITLE	PI	PING FABRICATION AND	MEC/S/05/21/06	REVISION: 0	
		ERECTION		EDITION: 1	
	h)	All low hydrogen type of minimum and stored in ove of the electrode Manufactur	electrodes shall be rebal ens kept at 80-100°C befo er shall be followed if avai	ked at 350°C for 1 hour re use. Recommendations lable.	
	i)	The electrodes, filler wires earth and other foreign mat	and flux used shall be fre ter which can affect the qu	ee from rust, oil, greases, uality of welding.	
1.3B	SHIE	LDING & PURGING GAS			
	a)	Argon gas used in GTA welding for shielding purposes shall be 99.995% pure. The purity of the gas shall be certified by the manufacturer. The rate of flow for shielding purposes shall be established through procedure qualification tests. Normally this rate may be 12-20 CFH.			
	b)	Argon gas with a purity level of 99.995% shall be used for purging.			
	c)	When GTAW process alone or a combination of GTA Wand SMAW processes is recommended for the production of a particular joint, the purging shall be maintained during the root pass and for the first filling pass to minimize oxidation on the inner side of the pipe, unless otherwise specified in Welding Specification Chart.			
	d)	Initial purging shall be maintained for sufficient period of time so that at least 4-5 times the volume between the dams is displaced, in order to completely remove the entrapped air. In no case should the initial purging period be less than 10 minutes. High gas pressure should be avoided.			
	e)	After initial purging, the flo where only a slight positive volume (up to 1/2 cubic for adequate. Systems of large should be established during	w of the backing gas sho pressure prevails. For sys ot) to be purged, a gas flo or volume may require hig g procedure qualification to	uld be reduced to a point stems, which have a small w rate of 6-CFH is usually gher flow rates and these ests.	
	f)	Gas backing (purging) is no	t required for socket type	of welded joints.	
	g)	Dams, used for conserving completion of the welding, dams is not possible after w	inert gas during purging and shall be accounted f velding, use of water-solub	g, shall be removed after or. Wherever, removal of le dams should be made.	
1.4	Weld	ling Consumables			
	The neces	Contractor shall provide at sary for the execution of lene, etc. and these should be	this own expense all t the job such as electro ear the approval of the COI	he welding consumables des filler wires, oxygen, MPANY.	

MECON LIMITED	STANDARD TECHNICAL		
834002	OIL & GAS SBU, DELHI		मुकान 10,4007:2000 Californi
		DOCUMENT NO.	Page 20 of 61
TITLE	PIPING FABRICATION AND	MEC/S/05/21/06	REVISION: 0
	ERECTION		EDITION: 1

## 1.5 Equipment & Accessories

- 1.5.1 The Contractor should have the arrangement of sufficient number of welding and cutting equipments, auxiliaries and accessories of sufficient capacities so as to meet the target schedule.
- 1.5.2 All the equipment for performing the heat treatment, including transformers, thermocouples, flow meters, automatic temperature recorders with suitable calibration arrangement etc. shall be provided by the Contractor, at his own expense and these should bear the approval of the COMPANY.
- 1.5.3 Contractor shall make necessary arrangements at his own expense for providing the radiographic equipment, radiographic films, and all the equipment/ materials required for carrying out the dye penetrant/ magnetic particle test for satisfactory and timely completion of the job.
- 1.5.4 Redoing of any work necessitated by faulty equipments or operation used by the Contractor, will be done at his own expense.

### 1.6 Welding Processes

- 1.6.1 Welding of various materials under this specification shall be carried out using Shielded Metal Arc Welding (SMAW) Process with the approval of the Engineer-in-charge.
- 1.6.2 The welding processes to be employed are given in the welding specification chart. Any deviation desired by the Contractor shall be obtained through the express consent of the Engineer-in-charge.
- 1.6.3 Automatic and semi-automatic welding processes shall be employed only with the express approval of the Engineer-in-charge. The welding procedure adopted and consumables used shall be specifically approved.
- 1.6.4 A combination of different welding processes or a could be employed for a particular joint only after duly qualifying the welding procedure as per the requirements of code of fabrication to be adopted and obtaining the approval of the Engineer-in-charge.

### 1.7 End Preparation

## 1.7.1 End Preparation

The edges to be welded shall be prepared to meet the joint design requirements by gas cutting, machining or grinding method. After gas cutting, oxides shall be removed by chipping or grinding.

## 1.7.2 Cleaning

a) The ends to be welded shall be properly cleaned to remove paint, oil, greases, rust, oxides, sand, earth and other foreign matter. The ends shall be

MECON LIMITED	STANDARD TECHNICAL		
834002	OIL & GAS SBU	初日 初日 初日 初日 初日 12000 Call 日 日	
		DOCUMENT NO.	Page 21 of 61
TITLE	PIPING FABRICATION AND	MEC/S/05/21/06	REVISION : 0
	ERECTION		EDITION: 1

completely dry before the welding commences.

b) On completion of each run, craters, welding irregularities, slag etc., shall be removed by grinding and chiseling. Wire brushes used for cleaning stainless steel joints shall have stainless steel wires and the grinding wheels used for grinding stainless steel shall be of a suitable type. Separate grinding wheels and wire brushes should be used for carbon steels and stainless steels.

1.8

### Alignment and Spacing

- a) Prior to alignment, the contractor shall inspect the pipe ends inside and outside for damage, dents, laminations etc. Pipe for welding shall be set up correctly spaced. Temporary attachment of any kind shall not be welded. Every effort shall be made to reduce misalignment by the use of clamp and rotation of pipes to the best fit. For pipes of same nominal wall thickness, the off set should not exceed 1.6mm. Any branch connections sleeve shall be at least 150mm from any other weld. The welds for fitting shall be so located that top of the weld shall not come within 50mm of any other weld. The use of internal line up clamps is mandatory for diameters 10" and above. However, in case where it is impossible to use internal line up clamp, external line up clamp may be used.
- b) Tack welds, for maintaining the alignment, of pipe joints shall be made only by qualified welders using approved WPS. Since the tack welds become part of the final weldment they shall be executed carefully and shall be free from defects. Defective tack welds must be removed prior to the actual welding of the joints.

### 1.9 Weather Conditions

- 1.9.1 The parts being welded and the welding personnel should be protected from rain and strong winds. In the absence of such a protection no welding shall be carried out.
- 1.9.2 During field welding using GTAW process, particular care shall be exercised to prevent any air current affecting the welding process.

## 1.10 Welding

## 1.10.1 **Root Pass**

- a) Root pass shall be made with electrodes/ filler wires recommended in the welding specification chart. The preferable size of the electrode is 2.5mm diameter (12 SWG) but in no case greater than 3.25mm (10 SWG).
- b) Uphill welding shall be adopted for welding pipes weld fixed with its axis horizontal. Downward technique of welding shall not be used for welding of pipes in horizontal position, unless specifically permitted by Engineer-in-charge for a particular case.

MECON LIMITED		STANDARD TECHNICAL SPECIFICATION OIL & GAS SBU, DELHI		
834002				नकान मकान 10 मकान
			DOCUMENT NO.	Page 22 of 61
TITLE	P	PIPING FABRICATION AND	MEC/S/05/21/06	REVISION: 0
		ERECTION		EDITION: 1
	c)	The root pass of but joints penetration with complete pipe shall not exceed .4mm	s should be executed prop fusion of the root edges. wherever not specified by	perly so as to achieve full Weld projection inside the the applicable code.
	d)	Any deviations desired f electrodes indicated in the after obtaining express app	rom the recommended welding specification char roval of the Engineer-in-ch	welding technique and t should be adopted only arge.
	e)	Welding shall be continuous	s & uninterrupted during a	pass.
	f)	On completion of each run removed by grinding and ch	n, craters, welding irregul niselling.	arities, slag etc., shall be
	g)	While the welding is in progress care should be taken to avoid any kind o movement of components, to prevent occurrence of weld cracks.		
	h) Fillet welds shall be made by shielded metal arc/ GTAW w irrespective of the thickness and class of piping.			GTAW welding process
	i)	i) Peening shall not be used unless specified in the welding specificat		
1.10.2	Join	it Completion		
	a)	Joint shall be completer recommended in the weldir exceed 4 mm in diameter temperature applications.	d using the class of ng specification chart. Size for stainless steels and	filler wires/ electrodes, of the electrode shall not alloy steels used for low
	b)	Two weld beads shall not b	e started at the same poin	t in different layers.
	b)	Butt joints shall be complet at the joint edge and a grad	ted with a cover layer tha dual notch free surface.	t would effect good cover
	d)	Each weld joints should hav	ve a workman like finish.	
	e)	Weld identification mark sh the weld. Metal stamping si shall be used on thin wall p	all be stamped clearly at e hall not be used on the thi ipes for identification.	ach joint, just adjacent to n wall pipe. Suitable paint
	f)	No painting shall be done u	ntil the weld joint has beer	hydrostatically tested.
1.10.3	Diss	similar Welds		
	Whe	ere welds are to be produced b	petween carbon steels and	l alloy steels, preheat and

Where welds are to be produced between carbon steels and alloy steels, preheat and post weld heat treatment requirements shall be those specified for corresponding alloy steels and filler wire / electrodes shall correspond to ER 70 S-G or AWS E-7016/7018 type. For welds between two dissimilar Cr-Mo low alloy steels, preheat and post weld

MECON LIMITED	STANDARD TECHNICAL		
834002	OIL & GAS SBU	येत् मेकान 10,4001:2000 Cantra	
		DOCUMENT NO.	Page 23 of 61
TITLE	PIPING FABRICATION AND	MEC/S/05/21/06	<b>REVISION: 0</b>
	ERECTION		EDITION: 1

heat treatments shall be those specified for higher alloy steel and electrodes used shall correspond to those specified for steel of lower alloy content. For carbon steel or alloy steel to stainless welds, use of filler wire / electrodes E/ER-309/E-310/E NiCr Fe-3 shall be made. The welding procedure, electrodes / filler wires to be used shall be approved by the Engineer-in-Charge.

## 1.11 Heat Treatment

# 1.11.1 Preheating

- a) Preheating requirements for the various materials shall be as per the welding specification chart attached. No welding shall be carried out without preheating the joint to 10°C (50°F) when the ambient temperature is below 10 degree.
- b) Preheating shall be performed using resistance or induction heating methods. Preheating by gas burners, utilising any acetylene or oxy-propane gas mixtures, with neutral flame may also be carried out when permitted by the Engineer-incharge.
- c) Preheating shall extend uniformly to atleast three times the thickness of the joint, but not less than 50mm, on both sides of the weld.
- d) Preheating temperature shall be maintained over the whole length of the joint during welding. Temperature indicating crayons or other temperature indicating devices shall be provided by the contractor to check the temperature.
- e) Preheating temperature shall be maintained over the whole length of the joint during welding. Temperature recorders shall be provided by the Contractor to record the temperature.

## 1.11.2 **Post Heating**

In case of alloy steel materials such as Cr-Mo steels, if the post weld heat treatment is not performed immediately after welding, the weld joint and adjacent portion of pipe, at least 50 mm on either side of weld, shall be uniformly heated to 300°c. This temperature shall be maintained for half an hour minimum, and then wrapped with mineral wool before allowing it to cool to room temperature. If the Post Heating temperature specified in the Welding Specification Charts exceeds 300°C, the same shall be followed. Similarly, if the welding specification chart specifies post-heat time, the same shall be applicable. Post weld heat treatment as specified in the Welding Specification Chart shall be carried out later on.

## 1.11.3 **Postweld Heat Treatment**

a) Post weld heat treatment, wherever required for joints between pipes, pipes an fittings, pipe body and supports shall be carried out as per the relevant specifications, applicable standards and the instructions of the Engineer-incharge. In this regard procedure qualification to be done before carrying out

MECON LIMITED	STANDARD TECHNICAL SPECIFICATION			
834002		OIL & GAS SBU,	DELHI	中の1-2000 Cantil
			DOCUMENT NO.	Page 24 of 61
TITLE	PI	PING FABRICATION AND	MEC/S/05/21/06	REVISION: 0
		ERECTION		EDITION: 1
		PWHT in production welds.		
	b)	The heat treatment of requirements laid down in A	welded joints shall be ANSI B31.8 and welding sp	carried out as per the ecification chart.
	c) The contractor shall submit for the approval of the Engineer-in-charge, the details of the post weld heat treatment procedure, as per <b>Exhibit</b> ` attached, that the propose to adopt for each of the materials/ assembly/ painvolved, well before carrying out actual heat treatment.			
	d)	Post weld heat treatment induction heating equipmer	shall be done by using at as decided by the Engine	an electric resistance or eer-in-charge.
	e) While carrying out local post weld heat treatment, technique of application heat must ensure uniform temperature attainment at all points of the po being heat treated. Care shall be taken to ensure that width of treated I over which specified post weld heat treatment is carried out, the tempera attained is atleast as that specified in the relevant applicable standards/ coo			
	f) Throughout the cycle of heat treatment, the portion outside the heated shall be suitably wrapped under insulation so as to avoid any ha temperature gradient at the exposed surface of pipe. For this pu temperature at the exposed surface of the pipes should not be allow exceed 400°C.			outside the heated band s to avoid any harmful pipe. For this purpose should not be allowed to
	g)	The temperature attained recorded by means of thermocouples should be a location along the periph thermocouples attached per upto 12" and above. However minimum number of thermo	by the portion under thermocouple pyrometers attached to the pipe direct ery of the pipe joint. T er joint shall be 1 upto 6" ver the Engineer-in-charge pocouples to be attached if t	heat treatment shall be s. Adequate number of tly at the equally spaced he minimum number of dia, 2 upto 10" dia and 3 can increase the required found necessary.
	h)	Automatic temperature rec employed. The calibration Engineer-in-charge prior to approval should be obtained	corders which have been chart of each recorder sh o starting the heat tread d.	duly calibrated should be ould be submitted to the tment operation and his
	i)	Immediately on completion treatment charts/ records a (whenever required as per to Engineer-in-charge for hi	on of the heat treatme longwith the hardness test the welding specification s approval.	nt, the post weld heat results on the weld joints chart), shall be submitted
	j)	Each joint shall bear an ide piping sketch to be prepare should appear on the cor same identification numb corresponding radiographi	entification number which s ed by the contractor. The j responding post weld he ers shall also be follow c films. The chart cont	shall be maintained in the oint identification number at treatment charts. The ed for identification for caining the identification

STANDARD TECHNICAL SPECIFICATION OIL & GAS SBU, DELHI			
		18 मेकान 18 मेकान 19 मेकार 2000 Contribut	
	DOCUMENT NO.	Page 25 of 61	
RICATION AND	MEC/S/05/21/06	REVISION: 0	
		EDITION: 1	
and piping sketc folders.	h shall be submitted to	the Engineer-in-charge i	
Iness of the heat a nt, shall be measu he maximum hardunt shall be subject the specified limit,	ffected zone as well as of ured using suitable hardr ness specified in the weldin ted to re-heat treatment at the contractor's own ex	the weld metal, after hea ness tester and shall no ng specification chart. Th when hardness measure penses.	
tractor shall arrang of all joints teste inspector.	ge for the hardness testir d. These records shall b	ng and shall maintain th be checked by the plar	
e Weld Joints			
shall be free from a ieved by brushing.	adherent weld spatter, sla	g, dirt or foreign matte	
Inspection & Testing			
General			
ver's inspector shall vork is being perfo r all means and fac	I have free access to all co ormed. The contractor sha ilities necessary for carryin	ncerned areas, where th all also offer the Owner g out inspection.	
ner is entitled to d ication and erection ollowing objectives	epute his own inspector to n of pipelines are being d :-	o the shop or field when lone with (but not limite	
To check the conf various welding equ To supervise the we To supervise the we	formance to relevant sta lipments and the welding p elding procedure qualification elder performance qualifica	ndards and suitability operformance. on. tions.	
or shall intimate tion tests welding inspector to be pre	sufficiently in advance works and acceptance to sent to supervise them.	the commencement ests, to enable the pla	
edure Qualification	ons		
Procedure Qualific requirements of Al b requirements by the welding proce d) immediately afte	cation shall be carried ou PI 1104/ ASME Sec-IX or o the contractor at his expe edure specifications in fo er the receipt of the order.	it in accordance with the other applicable codes ar ense. The contractor sha format as per <b>Exhibit-</b>	
tŀ d`	ne welding proce ) immediately afte	ne welding procedure specifications in fo ) immediately after the receipt of the order.	

MECON LIMITED	STANDARD TECHNICAL		
834002	OIL & GAS SBU	मुकान 10,0001:2000 Control	
		DOCUMENT NO.	Page 26 of 61
TITLE	PIPING FABRICATION AND	MEC/S/05/21/06	REVISION: 0
	ERECTION		EDITION: 1

b) COMPANY's inspector will review, check and approve the welding procedure submitted and shall release the procedure for procedure qualification tests. The procedure qualification test shall be carried out by the Contractor under field conditions at this own expense. A complete set of test results in format as per **Exhibit-D** (attached) shall be submitted to the COMPANY's inspector for approval immediately after completing the procedure qualification test and atleast 2 weeks before the commencement of actual work. Standard tests as specified in the code shall be carried out in all cases. In addition to these tests, other tests like radiography, macro/ micro examination, hardness testers, dye penetrant examination, Charpy V-notch etc. shall be carried out on specimens. It shall be the responsibility of the contractor to carry out all the tests required to the satisfaction of the COMPANY's Inspector. The destructive testing of welded joints shall be as per **Annexure-2** and ASME Sec-IX.

### 1.13.3 Welder's Qualification

- a) Welders shall be qualified in accordance with the API 1104/ ASME IX and other applicable codes by the contractor at his expense. The butt weld test pieces of the qualification test shall meet the radiographic tests requirements as mentioned in this specification. The COMPANY's inspector shall witness the test and certify the qualification of each welder separately. Only those welders who have been approved by the COMPANY's inspector shall be employed for welding. Contractor shall submit the welder qualification test reports in the standard format and obtain express approval, before commencement of the work. No welder shall be permitted to work without the possession of the identify card. It shall be the responsibility of contractor to carry out Qualification tests of welders.
- b) The welders shall always have in their possession the identification card as shown in **Exhibit-E** and shall produce it on demand by the COMPANY's Inspector. It shall be the responsibility, of the Contractor to issue the identify cards after it has been duly certified by the COMPANY. If a welder is found to perform a type of welding for which he is not qualified, he shall be debarred from doing any further work. All welds performed by an unqualified welder shall be cut and redone by a qualified welder at the expense of the Contractor.

## 1.13.4 Visual Inspection

Inspection of all welds shall be carried out by COMPANY as per the latest editions of the applicable codes and specifications. All finished welds shall be visually inspected for parallel and axial alignment of the work, excessive reinforcement, concavity of welds, shrinkage, cracks, under cuts, dimensions of the weld, surface porosity and other surface defects. Under-cutting adjacent to the completed weld shall not exceed the limits specified in the applicable standard/ code.

MECON LIMITED	STANDARD TECHNICAL		
834002	OIL & GAS SBU	मेकान मेकान 8001:2000 Combin	
		DOCUMENT NO.	Page 27 of 61
TITLE	PIPING FABRICATION AND	MEC/S/05/21/06	<b>REVISION: 0</b>
	ERECTION		EDITION: 1
1.13.5	Non-destructive Examination		

The non destructive examination shall mainly consist of examination using x-ray radiography as detailed in **Annexure-4**.

Radiographic examination of one hundred percent (100%) girth welds will be required by the COMPANY. Welds shall meet the standards of acceptability as set forth in API 1104 and as per the requirements laid in subsequent paragraphs.

The CONTRACTOR shall make all the arrangements for the radiographic examination of work covered by this specification at his expense.

The COMPANY will review all the radiographs of welds and inform the CONTRACTOR regarding unacceptable welds. The decision of the COMPANY shall be final and binding in this regard.

All requirements mentioned in the specification shall be arranged and executed by the CONTRACTOR through his own resources. In addition, for pipes with wall thickness 9.5mm and above, ultrasonic inspection is required in the following cases as per **Annexure-3** of this specification.

- a) On the first 100 welded joints corresponding to each automatic (GTAW/ GMAW) welding procedure used.
- b) When 20mm or more are cut from the pipe end as supplied, the ends shall be ultrasonically inspected for an additional length of 20mm to ensure no lamination exist.
- c) When welds are repaired.
- d) When in the opinion of COMPANY, ultrasonic inspection is required to confirm or clarify defects indicated by radiography.
- e) When automatic procedure is used at least 10cm on each weld shall be ultrasonically inspected at COMPANY's discretion.

In addition, ultrasonic inspection may be required for certain critical weldings of the pipeline (i.e. tie-ins, welding of valves, flanges) randomly selected at COMPANY's discretion. All fillet and groove welds other than that radiographed shall be subjected to dye pentrant/ MP inspection. The non destructive test system used for inspecting welds must be approved by the COMPANY.

Weld quality is judged on the basis of the acceptability criteria mentioned below :

Any weld which as a result of radiographic and / or ultrasonic examination in the opinion of COMPANY **exhibits** imperfections greater than the limits stated in API-1104 latest edition or as superseded in this article shall be considered defective and shall so be marked with an identification paint marker.

MECON LIMITED	STANDARD TECHNICAL SPECIFICATION		
834002	OIL & GAS SBU	OIL & GAS SBU, DELHI	
		DOCUMENT NO.	Page 28 of 61
TITLE	PIPING FABRICATION AND	MEC/S/05/21/06	REVISION: 0
	ERECTION		EDITION: 1
	In addition to the API-1104 require cracks regardless of size of location i. Any amount of inadequate is unacceptable.	ements, the welds contain are unacceptable. penetration of the root bea	ing cracks including crate ad as defined by API-110
	ii. Any amount of incomplete API-1104 is unacceptable.	e fusion between the root	and bevel as defined b
	iii. Unrepaired burn through a	reas are unacceptable.	
	Contractor shall appoint agency for list of agency (ies) enclosed in the	r carrying out the radiogra bid document.	phy works at site from the
	• The Radiographic Examinat the contractor as per <b>Exh</b> Inspector prior to employr Level-III in Radiographic te Procedure shall be establist be consistently achieved u to film distance, geometr technique and procedure at in Article 2 as well as Arti shall be equal to or bette ASME Sec.V. Source sid radiographic procedure / te relevant codes of Fabricati where in the technical spe responsible for carrying of radiography of welds repain	tion procedures to be adop <b>ibit-F</b> and shall be got ap nent. A person qualified to esting shall prepare the pro- hed to demonstrate that the nder the most unfavorable ic unsharpness, thickness dopted shall conform of the cle 22 of ASME Sec.V. The r than the requirements dependerameter shall the echnique. The acceptance of ion and over riding require ecifications of the contract out Radiography; rectifica- red/rectified at his cost.	bed shall be submitted by proved from the Owner to ASNT Level-II or ASN becedure. The Radiograph he required sensitivity ca e parameters (e.g. source s etc.). The radiograph e requirements mentioned the IQI sensitivity obtained mentioned in Article 2 of the used in establishin criteria shall be as per the ements if mentioned els the Contractor shall be ation of defects and references
	• The extent of Radiography Contractor. For welds betw Examination shall be the materials being welded. A particular piping class, the included.	shall be as per specificative en dissimilar materials, t more stringent of the tw Wherever random Radiog dissimilar materials weld	ions to be supplied to th he extent of Radiographi vo recommended for th raphy is called for, in joints shall essentially b
	• Type of Radiation source a carrying out radiographic e where in the contract) for then Radiography shall be a	and film to be used shall examination. However if sp some critical material required done using X-Rays only.	be as per <b>Exhibit-H</b> fo pecifications (as given els uire usage of X-Radiation
	• The Contractor shall fulfill while handling X-ray and G	all the statutory and ow amma-ray equipments.	ner's safety requirement

MECON LIMITED	STANDARD TECHNICAL SPECIFICATION			
834002	OIL & GAS SBU	OIL & GAS SBU, DELHI		
		DOCUMENT NO.	Page 29 of 61	
TITLE	PIPING FABRICATION AND	MEC/S/05/21/06	REVISION: 0	
	ERECTION		EDITION: 1	
	• In case of random radiogra the Owner's Inspector and if he instructs the contract radiographs, to the Owner evaluation by a person qua with Article 2 of ASME qualification of the NDT per his approval prior to start of	aphy, the joints for Radiog the Radiography shall be p ctor to do so. The contra 's Inspector immediately a alified to ASNT Level-II in R Sec.V. The certificate of ersonnel shall be submitted of job.	raphy shall be selected by performed in his presence, actor shall furnish all the fter processing along with Radiographic testing, inline ASNT / ISNT Level II d to owner's inspector for	
	• The Contractor shall provide the Owner's Inspector, all the necessary facilitie at site such as a dark room with controlled temperature, illuminator (viewe suitable for varying densities, a duly calibrated electronic densitometer wi batteries, magnifying glass, tracing papers, ruler, marking pencils etc. to enab him to review the radiographs.			
	• Where random radiography is specified, the first weld of each welder s completely radiographed. In the case of pipe of size 6" and below, the fi welds shall be completely radiographed.			
	• For each weld performed by a welder found unacceptable, two addition checks shall be carried out on welds performed by the same welder. To operation is iterative and the of two additional welds for each weld deen unsatisfactory shall be continued till such time that two consecutive welds satisfactory quality are found for every defective weld.			
	• The Contractor shall carry expense. To avoid the po- welder remaining undetect promptly arrange for R accumulation of defective j	out these additional radio ossibility of too many def ted for a long period to t Radiographic Examination oints.	graphic testing at his own ective welds by a single ime, the Contractor shall so that there is no	
	• Contractor shall quote rates to be radiographed by X-ra	s for X-ray as well as Gamr y in Table of <b>Exhibit-H.</b>	na Ray for joints indicated	
1.13.6	Check shots			
	(a) Owner / Engineer- in- char joints radiographed on a d shots as directed.	rge or his representative sl lay for check shots. Contra	nall select 5% of the total actor shall carry out check	
	(b) Weld profiles of check sho the earlier Radiographs. In and earlier Radiographs, radiographed by particular re-shot films shall be comp	ots shall be compared with n the event of anyone var contractor shall re-shoot Radiography agency on t ared with the originally sub	a weld profile observed in riation in the check shots the entire lot of joints he particular date. All the mitted films.	

MECON LIMITED REGD. OFF: RANCHI	STANDARD TECHNICAL		
834002	OIL & GAS SBU	मेकान मेकान मेकार मेकार	
		DOCUMENT NO.	Page 30 of 61
TITLE	PIPING FABRICATION AND	MEC/S/05/21/06	<b>REVISION: 0</b>
	ERECTION		EDITION: 1

## 1.13.7 Magnetic Particle & Liquid Penetrant Examination

- a) Whenever such tests are specified, the tests shall be carried out on joints chosen by the Owner's inspector, as per ASME Section V article 6 and 7 respectively. The tests are to be performed by a person possessing a valid ASNT / ISNT Level-II qualification in the method being used.
- (b) For austenitic stainless steels and other nonmagnetic materials, liquid (dye) penetrant test shall be carried out. For carrying out this test, the materials shall be brought within a temperature limit of 15° to 50°C.

## 1.13.8 Hardness Test

Hardness requirements for welds shall be as per the Welding Specification Chart / Non Destructive Examination Specification attached elsewhere in the contract. Hardness testing shall be carried out by Vickers Hardness Tester during welding procedure qualification and shall be cross sectional. For production welds, hardness testing shall be carried out by portable digital hardness testers. Poldi hardness tester shall not be permitted. Contractor shall produce documentary evidence/calibration certificate to the Owner's Inspector and obtain approval of the hardness testing equipment.

## 1.13.9 **Proof Tests**

Hydrostatic and pneumatic tests shall be performed as per the requirements laid down in the respective flushing & testing specification/ applicable codes to demonstrate the soundness of the welds. The tests shall be conducted only after fulfilling the requirements of visual examination, radiography etc. and after the entire work has been certified by the Owner's inspector, as fit for subjecting to such test.

## 1.14 **Repairs of Welds**

- a) Defects ascertained, through the inspection methods, which are beyond acceptable limits shall be removed from the joint completely by the process of chipping and grinding.
- b) When an entire joint is judged unacceptable, the welding shall be completely cut and the edges be suitably prepared as per required alignment tolerances. The welded joint shall again be examined following standard practices.
- c) No repair shall be carried out without prior permission of the Owner's inspector.
- d) Where random radiography is specified, the test welds of each welder shall be completely radiographed. In the case of pipes of sizes 6" and below, the first two welds shall be completely radiographed.
- e) For each weld found unacceptable due to a welder's fault, two additional checks should be carried out on welds performed by the same welder. This operation is interactive and the procedure of radiographing two additional

MECON LIMITED	STANDARD TECHNICAL		
834002	OIL & GAS SBU	मुकान 10,0001:2000 Cantra	
		DOCUMENT NO.	Page 31 of 61
TITLE	PIPING FABRICATION AND	MEC/S/05/21/06	<b>REVISION: 0</b>
	ERECTION		EDITION: 1

welds for each weld deemed unsatisfactory shall be continued till such time that the two consecutive welds of satisfactory quality are found for every defective weld.

The contractor shall carry out these additional radiographic testing.

To avoid the possibility of too many defective welds by a single welder remaining undetected for a long period of time, the Contractor shall promptly arrange for radiographic examination so that there is no accumulation of defective joints.

### 1.15 Limitations on Repairs

Only one attempt at repair of any region is permitted. Repairs are limited to a maximum 30% of the weld length. For internal or external repairs which open the weld root, only 20% of the weld length may be repaired. Repairs opening the root must only be carried out in the presence of COMPANY. The minimum length of a repaired area shall be 100mm as measured over the recapped length. Welds containing cracks shall be cut out and rebevelled to make a joint, COMPANY shall authorise all repairs.

### 1.16 Weld Rejected by Accumulation of Defects

Where a weld is rejected by the accumulation of defect clause, as defined by API-1104 and this specification, repairs within these limitations are permitted. Defects in the filling and capping passes shall be repaired preferentially.

## 1.17 DOCUMENTS TO BE SUBMITTED BY CONTRACTOR (4 COPIES EACH)

- a) Electrode and Welding Consumable Qualification Records as per **Exhibit-A**, for the Welding Consumables tested and approved for the work.
- b) Batch Test Certificates, for the Electrodes used, obtained from the Electrode Manufacturers.
- c) Proposed Heat Treatment Procedure as per **Exhibit-B.**
- d) Heat Treatment Charts.
- e) Weld joint hardness test results.
- f) Welding Procedure Specifications as per **Exhibit-C** immediately after receipt of the order.
- g) Welding Procedure Qualification records as per **Exhibit-D**.
- h) Welder Performance Qualification records as per **Exhibit-E** immediately after conducting Welder Qualification Tests.
- i) Radiography Procedure as per **Exhibit-F** and other NDT procedures.
- j) Radiographic test Report along with Radiographs and other NDT reports.
- Piping Sketch (Isometric) giving all the details regarding the pipe specifications, welded joints, joints radiographed magnetic particle, tested, ultrasonic tested, penetrant tested, joints heat treated, WPS used, welders identification number, etc.

MECON LIMITED REGD. OFF: RANCHI 834002		STANDARD TECHNICAL SPECIFICATION				
		OIL & GAS SB	OIL & GAS SBU, DELHI			येत्र मेकान संस्थार संस्थार संस्थार संस्थार संस्थार संस्थार संस्थार संस्थार संस्थार संस्थार संस्थार संस्थार संस्थार संस्थार संस्थार संस्थार संस्थार संस्थार संस्थार संस्थार संस्थार संस्थार संस्थार संस्थार संस्थार संस्थार संस्थार संस्थार संस्थार संस्थार संस्थार संस्थार संस्थार संस्थार संस्थार संस्थार संस्थार संस्थार संस्थार संस्थार संस्थार संस्थार संस्थार संस्थार संस्थार संस्थार संस्थार संस्थार संस्थार संस्थार संस्थार संस्थार संस्थार संस्थार संस्थार संस्थार संस्थार संस्थार संस्थार संस्थार संस्थार संस्थान संस्थार संस्थार संस्थार संस्थार संस्थार संस्थार संस्थार संस्थार संस्थार संस्थार संस्थार संस्थार संस्थार संस्थार संस्थार संस्थार संस्थार संस्थार संस्थार संस्थार संस्थार संस्थार संस्थार संस्थार संस्थार संस्थार संस्थार संस्थार संस्थार संस्थार संस्थार संस्थार संस्थार संस्थार संस्थार संस्थार संस्थार संस्थार संस्थार संस्थार संस्थार संस्थार संस्थार संस्थार संस्थार संस्थार संस्थार संस्थार संस्थार संस्थार संस्थार संस्थार संस्थार संस्थार संस्थार संस्थार संस्थार संस्थार संस्थार संस्थार संस्थार संस्थान संस्थार संस्थार स्थार स्थार स्थार स्थार स्थार स्थार स्थार स्थार स्थार स्थार स्थार स्थार स्थार स्थार स्थार स्थार स्थार स्थार स्थार स्थार स्थार स्थार स्थार स्थार स्थार स्थार स्थार स्थार स्थार स्थार स्थार स्थार स्थार स्थार स्थार स्थार स्थार स्थार स्थार स्थार स्थार स्थार स्थार स्थार स्थार स्थार स्थार स्थार स्थार स्थार स्थार स्थार स्थार स्थार स्थार स्थार स्थार स्थार स्थार स्थार स्थार स्य स्थार स्थार स्य स्य स्य स्य स्थार स्थार स्थार स्थार स्थार स्थार स्थार स्थार स्य स्य स्य स्य स्य स्य स्य स्य स्य स्य
			D	OCUMENT N	0.	Page 32 of 61
TITL	E	PIPING FABRICATION AND	ME	C/S/05/21	/06	REVISION : 0
		ERECTION				EDITION: 1
						<b>EXHIBIT</b>
						Sheet 1 of
		ELECTRODE QUALIF	ICATIO	<u>NS TEST R</u>	ECORE	<u>)</u>
A :	Tested	at	Date	e :		
	(Site Na	ame)			٦	Test Period :
	Manufa	cturer's Name	:			
	Brand N	lame		:		
	Batch N	lumber & size Tested	:			
	Classific	cation & Code	:			
	Intende	ed for Welding in positions		:		
	In com	pination with (if any)		:		
	Code of	Reference (used for testing)	:			
	Special	requirements (if any)	:			
B:	All - W	eld Tensile Test				
	Base M	aterial used		:		
	Pre-hea	t temperature	:			
	Postwe	d Heat Treatment Details		:		
	Visual E	Examination		:		
	Radiog	aphic Examination Results		:		

		STANDARD TECHNICAL SPECIFICATION								
834002	KANCHI		OI	L & GAS	S SBU, C	DELHI			मिकान 8001:2000 Combin	
						DO	CUMENT	NO.	Page 33 of 61	
TITL	E	PIPING	FABRICA		D	MEC/	/S/05/2	21/06	REVISION: 0	
			ERECTIO	N					EDITION: 1	
	Tensile	Test Result	ts			:			Sheet	: 2 of 4
SI. No.	Identifi Numbe	cation U.T r	S. Yield F	Point		Elonga	ition		Remarks	
 C :	Impac	t Test Res	ults							
	Test Te	mperature				:	Notch	in :		
	Type of	Specimens	s (Impd, Cl	narpy	:	Size of	Specim	iens :		
Speci	imen No.	Imp	oact Value		Averag	ge				
1. 2. 3. 4. 5.										
D :	Chemi	cal Analys	is Result							
	Electro	le size usec	: t							
	Batch N	lo.								
	 %С	%S	%P	%Si	%Mn	%	Cr	%Ni	%Mo	
									Page 375 of 6	644

MECON LIMITED STA		STANDARD 1	ECHNICA			
834002	OIL & GAS SBU, DELHI			मेकान		
					DOCUMENT NO.	Page 34 of 61
TIT	E	PIPING FABRICATI	ON AND	М	EC/S/05/21/06	REVISION: 0
		ERECTION				EDITION: 1
E :	Fillet \	<b>Weld Test Results</b>				Sheet 3
	Weldin	g Positions		:		
	Base M	laterials	:			
	Size of	electrode used	:			
	Visual 1	Inspection Results		:	1)	
					2) 3)	
	Micro T	est Results				
	Fractur	e Test Results	:			
	Remark	<s< td=""><td></td><td>:</td><td></td><td></td></s<>		:		
F:	Other	Test Results				
	i)	Transverse Tensile Tes	t	:		
		In Combination with		:		
		Base Material used		:		
		Position of Welding		:		
		Preheat Temperature	:			
		Postweld Heat Treatme	ent :			
		Radiography	:			
	 Identifi	cation No.	U.T.S.	Frac	ture in	Remarks

MECON LIMITED REGD. OFF: RANCHI	STANDARD TECHNICAL		
834002	OIL & GAS SBU,	येत् मेकान 10, 4007:2000 Carthan	
		DOCUMENT NO.	Page 35 of 61
TITLE	PIPING FABRICATION AND	MEC/S/05/21/06	REVISION: 0
	ERECTION		EDITION: 1

## 2. Guide Bend Test

Position	ID No.	Root, Face or Side Bend	Remarks
	1 2 3 4 5		

3. Any other tests

Conclusions

Approved By :

MECON LIN		STANDARD TECHNICAL		
834002	OIL & GAS SBU, DELHI		मेकान मेठा:2000 Confil	
			DOCUMENT NO.	Page 36 of 61
TITL	E	PIPING FABRICATION AND	MEC/S/05/21/06	REVISION: 0
		ERECTION		EDITION: 1
				EXHIBIT-B
		STRESS RELIEF HEAT TREATM	ENT PROCEDURE SPECI	Sheet 1 of 2 FICATION
	Name o Name o	of the Heat - Treatment : of the Project :	Specification Reference	
1.	Genera	al Details		
	Name o	of the Equipment :		
	Name o	of the Assembly/ Part :		
	Assemb	oly/ Part Drawing No. :		
	Materia	l :		
2.	Furnac	e Details		
	Type of	f Heating : Gas/ Oil/ Elec. Res./ Indu	uction (Tick Mark)	
	Capacit Maximu	y (size) : ım Temp. (°C)		
	Method	of temp. measurement :		
	Atmosp	here Control :		
3.	Heat T	reatment Cycle Details		
	Chargin	ıg Temp. °C :		
	Rate of	Heating, °C/Hr :		
	Soaking	J Temp., °C :		
	Soaking	J Time, Hrs. :		
	Rate of	Cooling, °C/Hr :		
	Mode o	f Cooling :		

MECON LIMITED	STANDARD TECHNICAL		
834002	OIL & GAS SBU, DELHI		येत् मेकान 10,4001:2000 Cantra
		DOCUMENT NO.	Page 37 of 61
TITLE	PIPING FABRICATION AND	MEC/S/05/21/06	<b>REVISION: 0</b>
	ERECTION		EDITION: 1

Sheet 2 of 2

4. Other Details, if any : .....

- 5. The following documents are to be furnished : along with these specification :
  - i) Material Test Certificates
  - ii) Assembly/ Part Details

MECON LIMITED REGD. OFF: RANCHI 834002STANDARD TECHNICA OIL & GAS SB		ICAL	SPECIFICATION		
		OIL & GAS	S SBU,	, DELHI	00 मेकान 18 मेकान 19 मेकान
				DOCUMENT NO.	Page 38 of 61
TITLE		PIPING FABRICATION AN	ND	MEC/S/05/21/06	REVISION: 0
		ERECTION			EDITION: 1
		STANDARD PRO	CEDU	IRE SPECIFICATION NO	EXHIBIT-C Sheet 1 of 2
foi	r	Welding of	: 	Pipe and F	Fittings
• Pro ma	ocess achine	& type es).			(Details of special
• Ma the	ateria e proc	edure applied, grade of steel	l, type	of pipe, Reference Specifi	(Pipes to which cation).
• Dia an	amete d thic	er and wall thickness kness to which procedure is a	applica	able)	(Series of dia
• Joi	int De	sign			
• Fil	ler Me	etal and Number of Beads			
• Ele	ectrica	I or Flame Characteristics			
• Po	sition				
• Dii Do	rectio wnhi	n of Weldings I, Mixed)			(Uphill,
• Ni	ımber	of Welders			
• Tir	ne La	pse between passes			
• Ty	pe of	Line-up Clamp			
• Re	emova ercent	l of Line-up Clamp age of welding carried out be	fore r	emoval of clamps)	(Minimum
• Cle	eaning	]			
• Pro	eheat	, Stress Relief			
• Sh	ieldin	g Flux			
• Sp	eed c	f Travel			
• Sk	etche	s and Tabulations (to be atta	iched)		
• Wi	ire Sp	eed (rate of wire speed and v	variati	on range)	
• Mi	nimuı	n No. of passes which must b	be cor	npleted before discontinuir	ng weld.

MECON LIMITED	STANDARD TECHNICAL		
834002	OIL & GAS SBU	18 मेकान 9001:2000 Canth	
		DOCUMENT NO.	Page 39 of 61
TITLE	PIPING FABRICATION AND	MEC/S/05/21/06	REVISION : 0
	ERECTION		EDITION: 1
• Minimu	m No. of welders required for the fi	irst pass and second pass :	Sheet 2 of 2

Tested :		Welder	
Approved	:		Welding Supt
Accepted	:		Chief Engineer

	STAND					
34002		OIL & GAS SBU, DELHI				
			DOCUMENT NO		Page 40 of 61	
TITLE	PIPING FABR	ICATION AND	MEC/S/05/21/0	06	REVISION: 0	
	EREC	.110N			EDITION: 1	
		COUPON 1	EST RECORD		<b>EXHIBIT-D</b> Sheet 1 of 2	
Location Date weld	M. Tempe Win Amperag	ate welder Welding Time . rature g break used ge Size	Roll Weld F. Weathe Typ Fille Pipe kin Wall th	Mark Mark Conditio Voltage oe of weldi r Metal . Size of re d and Grad ickness	Fixed position ne of day n e ng machine einforcement de	
Bead No	1 2	3 4	5 	6 of electrod	7 Size of electrode e	
Coupon Stencil Dimension of P (inch <sup>2</sup> )	1 2 ed	3 4	5 	6 Origin m Load nsile S/ in. ocation	7 Original nal area of plate plate area	
Procedure Welder Max. tensile str	ength	Qualif Lin min. tensile str	ying Test e Test ength		Qualified Disqualified	
Avg. tensile str 1 2	ength	Remarks or	tensile strength		·····	
3 4						

RECON LIMITED	STANDARD TECHNICAL	SPECIFICATION				
834002	OIL & GAS SBU	, DELHI	現 利 和 可 可 可 一 れ 一 和 一 和 一 和 一 和 一 和 一 和 一 和 一 和 一 和 一 和 一 和 一 和 一 和 一 和 一 和 一 和 一 和 一 和 一 和 一 和 一 和 一 和 一 和 一 和 一 和 一 和 一 和 一 和 一 和 一 和 一 和 一 和 一 和 一 和 一 和 一 和 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 本 一 本 本 一 本 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 一 本 一 一 一 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 本 一 本 一 本 一 本 一 本 一 本 本 一 本 本 一 本 本 一 本 本 一 本 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 一 本 本 本			
		DOCUMENT NO.	Page 41 of 61			
TITLE	PIPING FABRICATION AND	MEC/S/05/21/06	REVISION: 0			
	ERECTION		EDITION: 1			
			Sheet 2 of 2			
Remarks on	Bend Tests					
1						
2						
3						
4	4					
Remarks on	Nick Tests					
1						
2						
3						
4						
Other Tests						
		( Use ba	ck for additional remarks )			

MECON LIMITED	STANDARD TECHNICAL		
834002	OIL & GAS SBU,	, DELHI	मेकान मेकान 8001:2000 Contribut
		DOCUMENT NO.	Page 42 of 61
TITLE	PIPING FABRICATION AND	MEC/S/05/21/06	REVISION : 0
	ERECTION		EDITION: 1
			EXHIBIT-
	WELDER's IDEN	TIFICATION CARD	
Name	:		
Identification	:		
Date of Testing	:		
Valid Unit	:		
Approval of We	lding :		
Welding Position	n :		
Material	:		
Diameter	:		
Wall Thickness	:		

Type of Welding Consumable :

Approved By :

Employer's Signature with Seal

MECON LIMITED	STANDARD TECHNICAL			
834002	OIL & GAS SBU	मुकान अल्ला:2000 Control		
		DOCUMENT NO.	Page 43 of 61	
TITLE	PIPING FABRICATION AND	MEC/S/05/21/06	REVISION: 0	
	ERECTION		EDITION: 1	
EXHIBIT-F Sheet 1 of 1 RADIOGRAPHIC PROCEDURE FOR PIPE WEIDING				

- 1. Location
- 2. Date of Testing
- 3. Name of Supervised Contractor
- 4. Material
- 5. Dia. & Thickness
- 6. Type of Weld Joint
- 7. Radiation Source (X-ray, gamma ray)
- 8. Type of equipment (external/ internal)
- 9. Intensifying screens and material
- 10. Filter type and placement mask, diaphragm lead screen etc. adjacent to radiation sources or specimen.
- 11. Geometric relationship (source local spot size, max and min source strength, object to film distance, radiation angle with respect to weld and film).
- 12. Limit of film coverage
- 13. Film type and make
- 14. Exposure Time
- 15. Processing (time temperature for development stop bath or rinse, fixation, washing, drying etc.)
- 16. Density
- 17. Sensitivity
- 18. Type of penetrameter

Approval of the COMPANY

Signature of CONTRACTOR with seal

MECON LIMITED	STANDARD TECHNICAL					
834002	OIL & GAS SBU,	初日 初日 初日 10 10 10 10 10 10 10 10 10 10				
TITLE	PIPING FABRICATION AND ERECTION	DOCUMENT NO. MEC/S/05/21/06	Page 44 of 61 REVISION : 0 FDITION : 1			
L	WELDING SPECIFICATION CHART     EXHIBIT-G       Sheet 1 of 2					
Class :						
Material Spec	cification :					
Pipes	: API 5L Gr. X 60, API 5L Gr.	B API 5L Gr. X 42				
Fittings	: A 105, A234 Gr. WPB. MSS-	-SP-75, Gr. WPHY42, MSS-	S			
Flanges	: A 105, MSS-SP-44 Gr. F42,	MSS-SP as Gr. WPH 60				
Other	: 44 Gr. F6C					
Base Metal of I	NCL :					
Welding Proces	sses : Groove Joints : Butt					
Root Pass <u>SMA</u> <u>SMAW</u>	<u>AW</u> Filler Pass <u>SMAW</u> Root Pass <u>SN</u>	<u>MAW</u> Filler Pass <u>SMAW</u> Fill	er Joints/ Socket Joints :			
Welding Materi	ials : Groove Joints : Butt					
Root Pass Root Pass	E6010G/ E7010G E7010/ E7018G/ E8018G Filler	Filler Pass F7010G/ Pass F7016/ E7018G/	E8010G/ E8818G ' E80118G			
Filler Joints/ Sc	ocket Joints : E7016/ E7018/ E7018	G/ E8018G				
Backing Page _	Cons	sumable :				
Gases : Purgine	g Sheil	ding				
Gas Compositio	on : Purging	Sheilding				
Preheating: 10	Preheating : 10 min for all welds, 100°C Post heating					
Post weld heat	Post weld heat treatment :					
Holding temp. : 595-650 CHolding Time : 1 Hr. per inch thkRate of heating : 200 C/hr max.Min holding time : 1 hr.Method of cooling : ControlledRate of cooling : 200°C/hr max.						
Mechanical pro	Mechanical property requirements :					

MECON LIMITED	STANDARD TECHNICAL		
834002	OIL & GAS SBU	यह मेकान मेकान मेकार 12000 Calify	
		DOCUMENT NO.	Page 45 of 61
TITLE	PIPING FABRICATION AND	MEC/S/05/21/06	REVISION: 0
	ERECTION		EDITION : 1

Sheet 2 of 2

Charpy `V' notch impact test valve :

Normal	:	22 J	
Average		:	27 J
At temperature		:	0 °C
Hardness		:	300 HV10 (for weld & HAZ)

Code of fabrication : ANSI B31.8; API 1104 and welding specifications.

			ST/	ANDARD 1	TECHN		<b>SPECIFICATION</b>		
834002		OIL & GAS SBU, D			DELHI		99 मेकान 199 मेकान 199 मेकान		
							DOCUMENT N	10.	Page 46 of 61
TITLI	Ξ	PII	PING F	ABRICAT		ID	MEC/S/05/21	L <b>/06</b>	REVISION : 0
			E	RECTION					EDITION: 1
					TEC	CHNIC	CAL NOTES		
1.	Welding the rea specific be appl	g, heat quirem ation. icable.	treatn ent of In case	nent and ANSI Bi of conflic	non de 31.8/ / ct betw	estruc API-1: een co	tive testing shall l 104 and addition 10de and specificati	be carrie Ial requ ion more	ed out in accordance with irement specified in the e stringent conditions shall
2.	No weld temper	ding sh ature is	all be o s below	carried ou v 10°C (50	it witho ) °F).	out pre	heating the joint	to 10°C	(50 °F) when the ambient
3.	Preheat	shall	be appl	lied while	welding	g the	following material	as detai	led below :
	API 5L	Gr. B			:	Thick and i	ness upto nclusive of 12mm	1	.00 °F min.
	A 105 MSS-SF A 234 ( MSS-SF	9-44 Gr Gr. WP 9-75-W	. F60 B PHY60			Thick 12 m	ness beyond m	2	200 °F
4.	For fille filler pa	t weld sses.	s comp	olete weld	ling ma	ay be	carried out using	the ele	ctrodes recommended for
5.	5. All weldments & HAZ shall meet the hardness requirements of 300 HV10 during procedure qualification. If the hardness exceeds 300 HV10 the joints shall be heat treated at temp. 1100-1250 °F for one hour. The heating and cooling rates shall be decided during procedure qualification subject to a maximum of 200 °C/Hr. Hardness testing shall be carried out by Vickers hardness tester during welding procedure qualification test only. No hardness test is required for production welds.				0 HV10 during procedure eat treated at temp. 1100- decided during procedure g shall be carried out by only. No hardness test is				
6.	The ele	ctrode	s used	shall mee	t the fo	ollowir	g additional requi	rement	
	<u>Specific</u>	ation	<u>UTS (</u>	Min.) (As	welded	<u>1)</u>	Impact (As weld	<u>ded)</u>	
	E7018-	G		52.7 kg	/mm²		20 ft. lb. at 0°C		
	E7018-	[		52.7 kg	/mm²		-		
	E6010		-			-			
	E6018		-			20 ft.	lb. at 0°C		
7.	All the	weldm	ents &	HAZ shall	meet t	he im	pact test requirem	nent of 2	0 ft. lb at 0°C.

MECON LIMITED REGD. OFF: RANCHI	STANDARD TECHNICAL		
834002	OIL & GAS SBU,	यत्र मेकान 9001:2000 Commit	
		DOCUMENT NO.	Page 47 of 61
TITLE	PIPING FABRICATION AND	MEC/S/05/21/06	<b>REVISION: 0</b>
	ERECTION		EDITION: 1

## ANNEXURE-2

### 1.0 **DESTRUCTIVE TESTING OF WELDED JOINT - BUTT WELDS**

### 1.1 **Preparation**

Having passed the visual and the non destructive inspection, the test weld shall be subjected to mechanical test.

After satisfactory completion of all visual and non destructive testing the procedure test weld shall be set aside for a period not less than 24 hours. No further work on the test weld and no cutting of test specimens from the weld be performed till a period of at least 24 hours has expired.

Weld specimens, for pipe diameter greater than or equal to 12.3/4" shall be taken from the positions indicated in Fig. 1 of this specification from areas as free from defects as possible. For this reason it is necessary to take the previous non destructive tests into account. The minimum no. of tests to be carried out is given in Table-I of this specification.

The test shall be carried out at laboratories approved by COMPANY. The specimens shall be prepared in accordance with the figure given in the paragraphs which refer to the individual test.

### <u>Table-I</u>

## TYPE AND NUMBER OF TEST SPECIMENS FOR PROCEDURE QUALIFICATION TEST

Pipe Size,				Νι	umber of	Specime	ns			
Out-side	Tensil	Tensil	Nick	Root	Face	Side	Macro	Hard-	Impa	Total
diameter	e API	e ISO	Break	Bend	Bend	Bend		ness	ct	
Inches										
		Wall <sup>-</sup>	Thickness	s > ½ in	ch (12.7r	nm) anc	l under			
Under 2 3/8	0	0	2	2	0	0	0	0	0	4
2 3/8 to 41/2 incl.	0	0	2	2	0	0	0	0	0	4
Over 4½ less than 12 3/4	2	0	2	2	2	0	2	2	12	24
12 3/4 and over	2	2	4	4	4	0	2	2	24	44
			Wall Thi	ckness >	1/2 inch	(12.7mm	ı)			
4½ and smaller	0	2	0	0	0	2	0	0	0	4
Over 4½ less than 12 3/4	2	0	2	2	2	0	2	2	12	24
12 3/4 and over	2	2	4	0	0	8	2	2	24	44

MECON LIMITED	STANDARD TECHNICAL		
834002	OIL & GAS SBU	मुकान 10,4007:2000 Californi	
		DOCUMENT NO.	Page 48 of 61
TITLE	PIPING FABRICATION AND	MEC/S/05/21/06	REVISION: 0
	ERECTION		EDITION: 1

## 1.2 Tensile Strength

Specimens for pipe diameter over 12 3/4" shall be taken from the position indicated in Fig. 1 of this specification. Two API type specimen shall be taken for pipe diameter greater than or equal to 12 3/4".

### 1.3 Nick-Break Test

### 1.3.1 **Preparation**

Specimens for nick-break test with notches thus worked can break in the base metal, instead of in the fusion zone; therefore an alternative test piece may be used after authorisation by the COMPANY with a notch cut in the reinforcement of outside weld bead to a maximum depth of 1.5mm measured from the surface of the weld bead.

### 1.4 Macroscopic Inspection

### 1.4.1 **Preparation**

Specimens shall be taken from the positions indicated in Fig. 1 of this specification and shall be prepared in accordance with ASTM E2 and E3.

The width of the macrosection has to be at least three times the width of the weld. The section is to be prepared by grinding or polishing and etching to clearly reveal the weld metal and heat effected zone.

### 1.4.2 **Method**

Specimens shall be carefully examined under the microscope with a magnification of at least 25 times. The COMPANY may ask for a macrograph with 5 times magnification for documentation purposes.

### 1.4.3 **Requirements**

Under macroscopic examination, the welded joints shall show good penetration and fusion, without any defect exceeding the limits stated in the evaluation criteria of the nick break test.

### 1.5 Hardness Test

### 1.5.1 **Preparation**

The prepared macrosection is to be used for hardness testing using the Vickers method with 100 N (10 kg) load. Indentations are to be made along traverses each approximately 1mm below the surface at both side of the weld.

MECON LIMITED	STANDARD TECHNICAL		
834002	OIL & GAS SBU	येत्र मिकान मेकान मेकार 2000 Californi	
		DOCUMENT NO.	Page 49 of 61
TITLE	PIPING FABRICATION AND	MEC/S/05/21/06	REVISION: 0
	ERECTION		EDITION: 1

In the weld metal a minimum of 6 indentations equally spaced along the traverses are to be made. The HAZ indentations are to be made along the traverses for approximately 0.5mm each into unaffected material, and starting as close to the fusion line as possible.

One indentation at each side of the weld along each traverse has to be made on parent metal. Reference is made to fig. 3 of this specification. The indentation are to be made in the adjacent region as well as on the opposite side of the macrosection along the specified traverses.

### 1.5.2 **Method**

The test shall be carried out in accordance with Recommendation ISO R81, Vickers hardness, using laboratory type machine controlled as per-recommendation of ISO R 146 and using a diamond pyramid penetrator set at 2.37 rad. (136) with a load of 100 N (10 kg).

### 1.5.3 **Requirements**

Hardness value shall not exceed 300 H VI0 . In case of a single reading slightly (+10 HV) higher than the specified limit, further indentations should be made to check if the high value was an isolated case.

All the hardness values obtained from the heat effected zone shall not exceed 100 HV with respect to the average hardness of the values obtained for the base metal.

If these additional tests mentioned above give a hardness within the specification limit, the slightly higher value may be accepted.

## 1.6 Charpy-V-notch Impact Test

1.6.1 Specimens shall be taken from the position indicated in Fig. 1 of this specification. The test specimens will be prepared in accordance with ISO R 148. Charpy V-notch specimens shall have dimensions as given in Fig. 3 of the specification.

Three test specimens shall be taken from each sample and they shall be cut and worked so that their length is transversal and perpendicular to the weld bead with the notch position as shown in Fig. 4 of this specification. The notch shall be perpendicular to the roller surface. The test specimens width shall depend upon the pipe wall nominal thickness as following :

Nominal wall thickness in mm	Test Specimens width in mm
> 12	12
> 9.5 and $\leq$ 12	7.5

MECON LIMITED	STANDARD TECHNICAL SPECIFICATION OIL & GAS SBU, DELHI			
834002			मुख्य मेकान मुख्य मेकान मुख्य 1:2000 Control	
		DOCUMENT NO.	Page 50 of 61	
TITLE	PIPING FABRICATION AND	MEC/S/05/21/06	REVISION: 0	
	ERECTION		EDITION: 1	
	 ≥ 7 and ≤ 9.5	5		
	< 7	2.5		
1.6.2	Test Method			
	The test shall be carried out as indicated in ISO R 148 "Beam Impact Test V-notch.			
	Test pieces shall be immersed in a thermostatic bath and maintained at the test temperature for at least 15 minutes. They shall than be placed in the testing machine and broken within 5 seconds of their removal from the batch.			
1.6.3	Requirements			
	The impact energy shall be greater or equal to :-			
	Test Specimens in Aver mm Thre Joul	age of e specimens e (min) (Note-2)	Minimum Single Value Joules (Note 1)	
	10	27	22	
	7.5	21.5 18 5	17.5	
	2.5	10.0	8.0	
	Note:1)Only one value is permitted	l to be lower than average	ge upto the value specified.	
1.7	Bend Test Requirements			
	The bend test specimens shall be made and tested as per the requirements of API-1104 sixteenth edition - May, 1983 except that the dimensions of Jig for guided bend test fig. 5 para 2.6 API-1104 shall be modified as follows :			
	Padius of the plupger • ^!	– 2 <del>+</del>		

Radius of the plunger	°A'	= 2 t
Radius of the die	° <b>B'</b>	= 3 t + 1.6mm
Width of the die	°C'	= 50.8

The acceptance criteria shall however be as per para 2.643 and 2.653 of API-1104 sixteenth edition - May, 1983.

Note t = Thickness of Specimen (nominal)

MECON LIMITED REGD. OFF: RANCHI	STANDARD TECHNICAL SPECIFICATION OIL & GAS SBU, DELHI		
834002			00 मेकान 9001:2000 Confination
		DOCUMENT NO.	Page 51 of 61
TITLE	PIPING FABRICATION AND	MEC/S/05/21/06	<b>REVISION: 0</b>
	ERECTION		EDITION: 1

## ANNEXURE-3

### 1.0 ULTRASONIC INSPECTION

In addition to the radiographic inspection ultrasonic inspection is required as per conditions mentioned in **Annexure-1** of this specification.

This section concerns manual ultrasonic inspection. However ultrasonic inspection by automatic equipment may be used if approved by the COMPANY.

#### 1.1 Equipment and Operators

The CONTRACTOR who carries out the ultrasonic inspection shall have sufficient qualified personnel, equipment and instrument at his disposal to be able to effect the tests without hindering or delaying the pipeline assembly operations.

- Calibrate the equipment;
- Perform an operational test under production conditions;
- Interpret the screen picture;
- Evaluate the size and location of reflectors

Specification for Ultrasonic Testing Procedure

• Interpret the type of defects detected.

The COMPANY has the option of checking the ability of personnel employed for ultrasonic testing by means of qualification tests.

The CONTRACTOR appointed to carry out UT inspection shall supply the instruments necessary for their execution on site.

#### 1.2

### Qualification

Before work begins the CONTRACTOR shall present a specification describing the proposed U.T. procedure qualification.

This specification shall be state, as an indication only but not limited to the following information :

- Type of U.T. equipment used
- Type and dimensions of transducers
- Frequency range
- Details for calibration
- Coupling medium
- Inspection technique
- Record details
- Reference to the welding procedure where it is intended to adopt the specification.
- Temperature range of the joints to be inspected.

MECON LIMITED REGD. OFF: RANCHI	STANDARD TECHNICAL SPECIFICATION OIL & GAS SBU, DELHI		
834002			येत्र मेकान 8001:2000 Confin
		DOCUMENT NO.	Page 52 of 61
TITLE	PIPING FABRICATION AND	MEC/S/05/21/06	REVISION: 0
	ERECTION		EDITION: 1

## 1.3 Qualification of Ultrasonic Inspection Procedure

The ultrasonic inspection procedure shall be approved by the COMPANY. Before inspection begins, the COMPANY may require the qualification test of the ultrasonic inspection procedure. This specification test consists in testing (under normal operating conditions) some CONTRACTOR welds made according to the same production procedure, when there are typical defects the test intends to detect.

### 1.4 **Test Procedure**

Circumferential weld shall be inspected from both sides using angled. Probes.

The surface with which the probe comes into contact shall be free of metal spatter, dirt, iron oxide and scales of any type: therefore it shall be necessary to clean a strip at least 50mm wide on both sides of the weld with steel - wire brushes and anyhow the cleaned strip must be atleast wide enough to allow full skip examination.

If during the test, echoes of doubtful origin appear, it shall be necessary to inspect a convenient area on the pipe surface, close to the weld, with a straight beam transducer in order to check whether any manufacturing defects are present which could have interfered with the ultrasonic beam.

By way of an example, the equipment shall include but not be limited to the following :

- Ultrasonic equipment and coupling medium
- Sample sections for calibration of instruments.
- Equipment for cleaning of surface to be examined.
- Rulers calibrated in centimeters for exact location of the position of defects.

The characteristics of the above-listed instruments and equipment shall guarantee:

- a) that the required standards of the inspection procedure, as previously established and approved by the COMPANY, are satisfied.
- b) continuous operation

All the instruments and equipment shall be approved by the COMPANY before being used. The COMPANY has the authority to reject any item which is considered unsuitable. The decision of the COMPANY is final. The CONTRACTOR appointed to carry out ultrasonic inspection shall also ensure the operational efficiency and maintenance of the instruments and equipment, and shall immediately substitute any item rejected by the COMPANY.

All the instrument and equipment necessary for carrying out ultrasonic inspection on site shall satisfy the requirements laid down by the public board of institutions which regulate "safety at work".

MECON LIMITED REGD. OFF: RANCHI	STANDARD TECHNICAL SPECIFICATION OIL & GAS SBU, DELHI		
834002			वि मेकान मेकान 9001:2000 Confide
		DOCUMENT NO.	Page 53 of 61
TITLE	PIPING FABRICATION AND	MEC/S/05/21/06	<b>REVISION: 0</b>
	ERECTION		EDITION: 1

### 1.5 Ultrasonic Instruments

The ultrasonic instruments :

- Shall be each pulse type, able to generate, receive and display, on the screen a cathode ray tube (CRT) pulse, at frequencies between 1 and 6 mhz. The useful part of the CRT screen shall be at least 70m wide and at least 50mm high.
- Shall have various amplification, with steps of 1 or 2 dB over a range of a least 60 dB.
- The regulation control shall be accurate to within 1 dB and this accuracy shall be certified by the instrument manufacturer.
- May be powered by a battery or an electric generator. In the first case, the autonomy of operation (endurance) of the instrument shall be sufficient to carry on working without frequent interruptions, and the instrument shall be equipped with an automatic switch which switches it off when the battery runs down, in the second case, there must be a voltage stabilising device with a tolerance of 2 volts.

### 1.6 **Probes**

The probes used shall have dimensions, frequencies, and a refraction angle suited to the type of steel, the diameter the thickness of the pipeline and to the joint design.

## 1.7 **Reference Sample Pieces**

The efficiency of the equipment used, the effective refraction angle of the probe, and the beam output points, shall be checked using a  $V_1$  and  $V_2$  sample block, IIW type or the calibration block ASTM E-428.

For the calibration of runs and the regulation of detection sensitivity during the test, a calibration piece be used. This piece shall be taken from the production material, and will be at least 150mm long (measured in the direction of the axis), and at least 50mm wide (measured in the direction of the circumference), (see Fig. 4 of this specification).

In the middle of the inside and the outside surface of the calibration piece a groove shall be made. The groove will have a rectangular cross-section, a flat bottom and the following dimensions :-

- Depth : 1 +/- 0.1mm
- Breadth (measured parallel to the 150mm side) : 1 +/- 0.1mm
- Length (measured parallel to the 50mm side) not less than 30mm.

In addition, the calibration piece shall have a hole, 2mm in diameter, passing through its thickness and positioned so that during calibration the echoes returning from the two grooves do not interfere with those returning from the hole.

MECON LIMITED REGD. OFF: RANCHI	STANDARD TECHNICAL SPECIFICATION OIL & GAS SBU, DELHI		
834002			10 मेकान 10 मेकान 10 0007:2000 Contribut
		DOCUMENT NO.	Page 54 of 61
TITLE	PIPING FABRICATION AND	MEC/S/05/21/06	REVISION: 0
	ERECTION		EDITION: 1

## 1.6 Calibration

For a precise check of the sound paths necessary for a full inspection of the weld joint, the probe shall be moved (half skip and full skip distance) until internal and external notches on the test piece are detected (see Fig. 5 of this specification).

The relevant defect limits the path lengths on the time base. The calibration of reference sensitivity is obtained by utilising the through drilled test hole in the thickness of the reference block to draw the distance amplitude correction curve relevant to the test probe.

Calibration shall be carried out according to the following procedure : place its internal vertex until the maximum height of echo is displayed on the screen; this echo is adjusted to 80% of full screen height by means of the sensitivity adjuster set in dB. Without varying the amplification, the probe placed at full skip distance from the hole is moved to detect the external vertex the hole until the maximum height of echo is obtained. The straight line connecting the peaks of the two echoes obtained by the above procedure, represents the 100% reference level, while the one connecting the two points at half height of the same echoes represents "50% reference level".

The two straight lines shall be marked on the screen with a pen. Calibration shallbe repeated each time tests are re-started at intervals not longer than 30 minutes during normal operations; each time the conditions fixed in advance are altered. This calibration is applicable provided that the crystal of the probe is  $8 \times 9$ mm size. Should this size of the crystal be different, the value of the sensitivity obtained from the calibration by a crystal of a different size shall be brought to the value of sensitivity obtained from the calibration by a  $8 \times 9$ mm crystal. The sensitivities of the two different size probes shall be compared through the echoes obtained on the notch of the test piece with the probe position at half skip of the distance.

### 1.9 **Regulation of Amplification during Production Testing**

The amplification during production testing shall be obtained by adding 2-6 dB (according to the surface condition of the pipe and its cleanness) to the reference amplification.

### 1.10 **Qualification of Ultrasonic Testing Operators**

Before the inspection begins or during the same inspection, the COMPANY may require a qualification test for the ultrasonic equipment operators.

### 1.11 Evaluation of Indications given by Ultrasonic Tests

Each time that echoes from the weld bead appear during production testing, the instrument amplification shall be altered to coincide with the reference amplifications and the probe shall be moved until maximum respond is obtained paying attention all the time to the probetube coupling.
MECON LIMITED	STANDARD TECHNICAL		
834002	OIL & GAS SBU	मुकान 10,0001:2000 Cantra	
		DOCUMENT NO.	Page 55 of 61
TITLE	PIPING FABRICATION AND	MEC/S/05/21/06	REVISION: 0
	ERECTION		EDITION: 1

If, under these conditions, the height of the defer echo is equal to or greater than that of the reference echo, the defect shall be evaluated. If the defect has also been detected by the radiographic and or visual examination, the dimensions shall be judged according to the type of examination which detects the greater defects. Returns which are less than 50% of the reference echo, will not be considered. It returns are above 50% but lower than 100% of the reference echo, and if the operator has good reasons to suspect that the returns are caused by unfavorably oriented cracks, he shall inform the COMPANY. Moreover, when there is a defect to be repaired such defect shall be removed for a length corresponding to the one where no more return echo is given.

#### 1.12 **Other Equipment**

The use of rules calibrated in centimeters, attached if possible to the probe, for the precise location of the position of welding defects, it recommended. Defect location is effected by measuring the projection distance between the probe output and the reflecting surface.

The operators carrying out the tests shall have besides the probing instrument, tools for cleaning the pipe surface (files, brushes, etc.) as well as the coupling liquid or paste appropriate for the temperature of the section to be examined.

MECON LIMITED		STANDARD TECHNICAL					
834002	OIL & GAS SBU, DELHI		DELHI	मिकान 13 3001:2000 Confu			
			DOCUMENT NO.	Page 56 of 61			
TITLE	P	IPING FABRICATION AND	MEC/S/05/21/06	REVISION: 0			
		ERECTION		EDITION: 1			
				ANNEXURE-4			
		RADIC	<b>OGRAPHY</b>				
1.0	<u>SCO</u>	PE					
	This main	annexure covers the radiogra pipeline. The welded joints sh	phic inspection of all type all include the following :	es of welded joints of the			
	i)	Full girth welds on the mair adopted.	nline construction including	double jointing of pipe, if			
	ii)	Welds for installation of blo and tie-ins.	ck valves, insulating joints	and other appurtenances			
	iii)	Welds at scraper launching	and receiving barrels				
	iv)	Terminal Piping.					
2.0	<u>APPI</u>	LICABLE STANDARDS					
	This specification shall apply in conjunction with the following (all latest edition) :						
	i)	) API 1104, Standard for welding pipelines and related facilities.					
	ii)	ANSI B31.8, code for Gas T	ransmission and Distribution	on Piping Systems.			
	iii)	ANSI B31.4, Code for Liquid Petroleum Transportation Piping System.					
	iv)	ASTM E94, Recommended	practice for Radiographic T	esting.			
	v)	ASTM, E 142, Standard Met	hod for Controlling Quality	of Radiographic Testing.			
	vi)	The American Society for r SNT-TC-1A Supplement-A.	non-destructive Testing. Re	ecommended Practice No.			
3.0	PROCEDURE						
3.1	The CON <sup>-</sup>	The radiographic examination procedure to be adopted shall be submitted by the CONTRACTOR as per <b>Exhibit-F.</b>					
3.2	The   of C requi	The procedure of radiographic examination shall be qualified to the entire satisfaction of COMPANY prior to use. It shall include but not be limited to the following requirements.					
	i)	Lead foil intensifying scre exposures.	ens, at the rear of the	film shall be used in all			

	LIMITED       STANDARD TECHNICAL SPECIFICATION         DFF: RANCHI       OIL & GAS SBU, DELHI				
834002			मुकान मुकान कार्ग		
		DOCUMENT NO.	Page 57 of 61		
TITLE	PIPING FABRICATION AND	MEC/S/05/21/06	REVISION: 0		
	ERECTION		EDITION: 1		
	ii) Type 2 and 3 films as per A	STM E-94 shall be used.			
	iii) A densitometer shall be us density shall be 2.0 and 3.5 of the film shall not exceed	sed to determine film den 5 through out the weld. Th 0.30.	sity. The transmitted film e unexposed base density		
	iv) Radiographic identification interpretation reports and t	n system and documen heir recording system.	ntation for radiographic		
3.3	The CONTRACTOR shall qualify ea to use.	ch procedure in the preser	nce of the COMPANY prior		
3.4	The procedure of radiographic endensity, clarity and contrast so that weld, and the outline and holes of	xamination shall produce at defects in the weld or ir the pentrameter are clearly	radiographs of sufficient the pipe adjacent to the discernible.		
3.5	All the girth welds of mainline shall be subjected to 100% radiographic examination. The CONTRACTOR shall furnish all the radiographs to the COMPANY, immediately after processing them, together with the corresponding interpretation reports on approved format. The details of the radiographs alongwith the joint identification number shall be duly entered in a register and signed by the CONTRACTOR and submitted to the COMPANY for approval.				
3.6	When the radiation source and the film are both on the outside of the weld and located diametrically opposite each other, the maximum acceptable length of film for each exposure shall not exceed the values given in Table 4 of API 1104. The minimum film overlap, in such cases, shall be 40mm. The ellipse exposure technique may be used on nominal pipe sizes of 2 inch and smaller provided that the source to film distance used is a minimum of 12 inches.				
3.7	Three copies of each acceptable radiographic procedure (as outlined in Specification no. MEC/S/O5/62/02) and three copies of radiographic qualification records, shall be supplied to COMPANY. One set of the qualifying radiographs on the job shall be kept by the CONTRACTOR's authorised representative to be used as a standard for the quality of production radiographs during the job. The other two sets shall be retained by COMPANY for its permanent record.				
3.8	Three copies of the exposure charts relating to material thickness, kilo voltage, source to film distance and exposure time shall also be made available to aCOMPANY by the CONTRACTOR.				
3.9	The CONTRACTOR shall, on a dail number (2) approximate chainage the specified acceptance standard unacceptable defects observed. It and welder on piping drawing and	y basis, record for each ra of weld location, (3) whet ls and (4) the nature and must be possible to relate pipeline alignment drawing	adiograph (1) radiography her or not the welds meet approximate location of to a particular butt weld J.		
	and welder on piping drawing and		j.		

MECON LIMITED	STANDARD TECHNICAL SPECIFICATION					
834002	OIL & GAS SBU, DELHI		मुत्र मेकान 9001:2000 Control			
		DOCUMENT NO.	Page 58 of 61			
TITLE	PIPING FABRICATION AND	MEC/S/05/21/06	REVISION: 0			
	ERECTION		EDITION: 1			
3.10	Each day's production of processed radiographs shall be properly packaged separately, identified by at least the (1) date, (2) radiographic unit, (3) job locations, (4) starting and ending progress survey stations and (5) shall include original and three copies of the daily radiographic record. The package shall be submitted to the COMPANY daily when possible, but in no event later than noon of the following day.					
3.11	The CONTRACTOR shall provide al with controlled temperature, film v radiographs.	I the necessary facilities at viewer etc. to enable the (	site, such as a dark room COMPANY to examine the			
3.12	The CONTRACTOR, if found nece examination suiting the local cond the approval of the COMPANY.	essary, may modify the p litions prevailing. This sha	procedure of radiographic II, however, be subject to			
3.13	COMPANY shall have free access to	all the CONTRACTOR's w	ork facilities in the field.			
3.14	Any approval granted by the COMPANY shall not relieve the CONTRACTOR of his responsibilities and guarantees.					
4.0	RADIATION SOURCE					
4.1	Radiographic examination shall be carried out using x-radiations. Radiographic examination by Gamma rays may be allowed, at the discretion of the COMPANY, in case of inaccessible joints.					
4.2	Whenever possible, pipeline welds inside the pipe, on the pipeline axis	s will be inspected by pla s, with a radiation of 6.28 r	cing the radiation source ad. (360°).			
	If it is impossible to place the radiation source inside the pipe, the weld will be inspected with the source on the outside. An overlap of at least 40mm at the ends of each film shall be required to ensure that the first and last location increment numbers are common to successive films and to establish that no part of a weld has been omitted.					
5.0	LEVEL OF QUALITY					
	The quality level of radiographic sensitivity required for radiographic inspection shall be at least equivalent to the values in Figure-6.					
6.0	<b>PENETRAMETERS</b>					
6.1	The image quality indicator (abbre the welding procedure and during be measured with the wire image shall be selected according to DIN source on the outside, a penetrame smaller wire of the penetrameter complete weld is radiographed in	eviation : IQI) shall be use normal line production. Ra quality indicator (Penetrar 54109 or ISO 1027. For ra eter shall be placed on eac turned towards the end of a single exposure using a	ed for the qualification of diographic sensitivity shall meter). The penetrameter adiographs made with the sh side of the film with the of the film itself. When a source inside the piping,			

MECON LIMITED REGD. OFF: RANCHI	IECON LIMITED STANDARD TECHNICAL SPECIFICATION EGD. OFF: RANCHI		
834002	OIL & GAS SBU,	मुकान 101:2000 Contract	
		DOCUMENT NO.	Page 59 of 61
TITLE	PIPING FABRICATION AND ERECTION	MEC/S/05/21/06	REVISION: 0
			EDITION: 1

four penetrameters approximately equally spaced around the circumference shall be used. During the procedure qualification, IQI shall be placed both on the source side and on the film side. The sensitivity obtained with IQI on the source side shall not be less than the values shown in Fig. 6 of this specification.

The sensitivity limit may be considered to have been reached when the outline of the IQI, its identification number and the wire of the required diameter show up clearly on the radiographs.

The COMPANY may authorise use of types of IQI other than those planned, provided that they conform with recognised standards and only if the CONTRACTOR is able to demonstrate that the minimum sensitivity level required is obtained. For this demonstration, a test shall be carried out comparing the IQI specified and the CONTRACTOR's to show up the identification number and other details of the proposed IQI, which must be visible in the test radiograph.

#### 7.0 FILM IDENTIFICATION MARKERS

All films shall be clearly identified by lead numbers, letters, and/ or markers. The image of the markers shall appear on the films, without interfering with the interpretation. These markers positions shall also be marked on the part to be radiographed and shall be maintained during radiography.

#### 8.0 **PROTECTION AND CARE OF FILM**

- 8.1 All unexposed films shall protected and stored properly as per the requirements of API 1104 standard and ASTM E 94.
- 8.2 The exposed and unexposed film shall be protected from heat, light, dust and moisture. Sufficient shielding shall be supplied to prevent exposure of film to damaging radiation prior to and following the use of the film for radiographic exposure.

## 9.0 **RE-RADIOGRAPHY**

- 9.1 The weld joints shall be re-radiographed in case of unsatisfactory quality of the radiographs, at the expense of the CONTRACTOR.
- 9.2 All the repaired weld joints shall be re-radiographed at no extra cost to the COMPANY in the same manner as that followed for the original welds. In addition, the repaired weld area shall be identified with the original identification number plus the letter `R' to indicate the repair.
- 9.3 When evaluating repair film, radiographers shall compare each section (exposure) of the weld with the original film to assure repair was correctly marked and original defect removed.
- 9.4 The COMPANY will review prior to any repair of welds, all the radiographs of welds which contain, according to the CONTRACTOR's interpretation, unacceptable defects.

MECON LIMITED	<b>MECON LIMITED</b> STANDARD TECHNICAL SPECIFICATION REGD. OFF: RANCHI				
834002	OIL & GAS SBU,	10 4007.2000 Convert			
		DOCUMENT NO.	Page 60 of 61		
TITLE	PIPING FABRICATION AND	MEC/S/05/21/06	REVISION: 0		
	ERECTION		EDITION: 1		
	The final disposition of all unaccept	able welds shall be decide	d by the COMPANY.		
10.0	QUALIFICATION OF RADIOGR	APHERS			
10.1	Pipeline radiographers shall be qua and to the full satisfaction of COMP	lified in according with the PANY.	e requirement of API 1104		
10.2	Certification of all the radiographer the CONTRACTOR to the COMPAN production radiography. The certific	rs, qualified as per 10.1 ab Y before a radiographer wi cate record shall include :	ove, shall be furnished by Il be permitted to perform		
	<ul> <li>i) Background and experience record</li> <li>ii) Training course record</li> <li>iii) Technical examination record</li> <li>iv) Doctor's report on radiographer's Oaecuer 0-1 acquity eye test.</li> <li>v) Date of qualification.</li> </ul>				
10.3	The radiographers shall be require use, prior to performing the work a	ed to qualify with each rac assigned to him in accordar	diographic procedure they nee with the specification.		
11.0	PRESERVATION OF RADIOGRA	<u>IPHS</u>			
11.1	The radiographs shall be processed for at least three years. All the rad preservation alongwith necessary d	d to allow storage of films liographers shall be preser locumentation.	without any discoloration nted in suitable folders for		
11.2	All radiographs shall become prope	rty of the COMPANY.			
12.0	EQUIPMENT AND ACCESSORIE	S			
12.1	CONTRACTOR shall make necessary arrangement at his own expense, for providing the radiographic equipment, radiographic films and the accessories for carrying out the radiographic examination for satisfactory and timely completion of the job.				
12.2	For carrying out the mainline radiographic examination the CONTRACTOR shall be equipped with suitable mobile/ stationary type with rooms. These shall have all the required facilities for film processing. Film viewer used shall be equipped with the film illuminator that has a light source of sufficient intensity and can be suitably controlled to allow viewing film densities upto 4.0 without damaging the film.				
13.0	RADIATION PROTECTION				
13.1	CONTRACTOR shall be responsible man with or near radiation sources	for the protection and per	rsonal monitoring of every		
13.2	The protection and monitoring shal	ll comply with local regulat	ions.		

MECON LIMITED	I LIMITED STANDARD TECHNICAL SPECIFICATION		
834002	OIL & GAS SBU	मेकान मेकान 9091:2000 Complet	
			Page 61 of 61
TITLE	PIPING FABRICATION AND ERECTION	MEC/S/05/21/06	REVISION: 0
			EDITION: 1

13.3 In view of visual hazards in the handling of radioactive source of material, CONTRACTOR shall be solely responsible for complying with all rules and regulations set forth by Atomic Energy Commission or any other Government agencies of India in this regard and COMPANY shall not be responsible and shall be kept indemnified by the CONTRACTOR for default(s) of whatever nature by the Contractor. Safety equipment as considered adequate by the COMPANY for all necessary personnel shall be made available for use and maintained for immediate and proper use by the CONTRACTOR.

#### 14.0 DISPLAY OF SAFETY INSTRUCTIONS

14.1 The safety provisions shall be brought to the notice of all concerned by display on a notice board at prominent place at the work spot. The person responsible for the "safety" shall be named by the CONTRACTOR.

#### 15.0 ENFORCEMENT FOR SAFETY REGULATIONS

15.1 To ensure effective enforcement of the rules and regulations relating to safety precautions, the arrangement made by the CONTRACTOR shall be open to inspection by COMPANY or its representatives.

### 16.0 **FIRST AID INDUSTRIAL INJURIES**

- 16.1 CONTRACTOR shall maintain first aid facilities for its employees and those of its subcontractors.
- 16.2 CONTRACTOR shall make outside arrangements for ambulance service and for treatment of industrial injuries. Names of those providing these services shall be furnished to COMPANY prior to start of work and their telephone no. shall be posted prominently in CONTRACTOR's field office.
- 16.3 All critical industrial injuries shall be reported promptly to the COMPANY and a copy of CONTRACTOR's report covering each personal injury requiring the attention of physician shall be furnished to the COMPANY.

#### 17.0 **NO EXEMPTION**

17.1 Not withstanding the above there is nothing in these clauses to exempt the CONTRACTOR from the operation of any other act or rules in force.

# SPECIFICATION FOR VENTS, DRAINS AND WELLS

SPECIFICATION NO.: MEC/S/05/21/15



(OIL & GAS SBU) MECON LIMITED DELHI 110 092

Page 404 of 644

MECON LIMITED REGD. OFF: RANCHI	STANDARD TECHNICAL	B and the second	
834002	OIL & GAS SBU		
		DOCUMENT NO.	Page 1 of 1
TITLE	VENTS, DRAINS AND WELLS MEC/S/05/21/15		REVISION: 0
			EDITION: 1

# CONTENTS

SL. NO.	DESCRIPTION OF DRAWING	DRAWING NO.
1.	Vent & Drain for Line 2" & above	MEC/SD/05/21/15/01
2.	Wells Installation 1½ Dia Taps	MEC/SD/05/21/15/02 (Sheet 1 of 2)
3.	Wells Installation 1½ Dia Taps	MEC/SD/05/21/15/02 (Sheet 2 of 2)
4.	Vent & Drain for lines 1 <sup>1</sup> / <sub>2</sub> " & below	MEC/SD/05/21/15/03
5.	Pressure Tapping	MEC/SD/05/21/15/05

PREPARED BY:	CHECKED BY:	APPROVED BY:	ISSUE DATE :
(Binita Brahma)	(Sunil Kumar)	(A.K. Johri)	Feb. 2009

Page 405 of 644

MECON LIMITED	STANDARD TECHNICAL	B Storie Contract	
834002	OIL & GAS SBU		
		DOCUMENT NO.	Page 1 of 1
TITLE	VENTS, DRAINS AND WELLS	VENTS, DRAINS AND WELLS MEC/S/05/21/15	
			EDITION: 1

# **AMENDMENT STATUS**

SI. No.	Clause / Paragraph / Annexure / Exhibit / Drawing Amended	Page No.	Revision	Date	By (Name)	Verified (Name)
<u> </u>						











# SPECIFICATION FOR FLUSHING AND TESTING OF PIPING SYSTEMS

SPECIFICATION NO.: MEC/S/05/21/11



(OIL & GAS SBU) MECON LIMITED DELHI 110 092

Page 412 of 644

MECON LIMITED REGD. OFF: RANCHI	STANDARD TECHNICAL		
834002	OIL & GAS SBU, DELHI		मित्र मेकान 9001:2000 Continue
		DOCUMENT NO.	Page 1 of 1
TITLE	FLUSHING AND TESTING OF	MEC/S/05/21/11	REVISION : 0
	PIPING SYSTEMS		EDITION: 1

# CONTENTS

# SL. NO. DESCRIPTION

- 1.0 SCOPE
- 2.0 REFERENCES
- 3.0 INSPECTION
- 4.0 FLUSHING
- 5.0 TESTING
- 5.1 EXTENT OF TESTING
- 5.2 GENERAL REQUIREMENT/TEST PREPARATION FOR TESTING
- 5.3 TESTING MEDIA, TEST PRESSURE AND TEST PRESSURE GAUGES.
- 5.4 TESTING PROCEDURE
- 5.5 COMPLETION OF TESTING
- 5.6 TEST RECORDS

PREPARED BY:	CHECKED BY:	APPROVED BY:	ISSUE DATE :
(Shalini Singh)	(Sunil Kumar)	(A.K. Johri)	Dec. 2008

MECON LIMITED REGD. OFF: RANCHI	STANDARD TECHNICAL		
834002	OIL & GAS SBU, DELHI		मिकान 9001:2000 Control
		DOCUMENT NO.	Page 1 of 1
TITLE	FLUSHING AND TESTING OF	MEC/S/05/21/11	REVISION: 0
	PIPING SYSTEMS		EDITION: 1

# AMENDMENT STATUS

SI. No.	Clause / Paragraph / Annexure / Exhibit / Drawing Amended	Page No.	Revision	Date	By (Name)	Verified (Name)

MECON LIMITED REGD. OFF: RANCHI	STANDARD TECHNICAL		
834002	OIL & GAS SBU	OIL & GAS SBU, DELHI	
		DOCUMENT NO.	Page 1 of 8
TITLE	FLUSHING AND TESTING OF	MEC/S/05/21/11	REVISION: 0
	PIPING SYSTEMS		EDITION: 1

# 1.0 <u>SCOPE</u>

This specification covers the general requirements for Inspection, flushing and testing of piping systems. However testing of steam lines falling under IBR shall also be governed by Indian Boiler Regulations.

Flushing and testing of all piping system shall be witnessed by the Consultant Representative / Engineer-in- Charge.

# 2.0 <u>REFERENCE</u>

ASME B31.3-2004	:	Process Piping
IBR	:	Indian Boiler Regulations

# 3.0 INSPECTION

During various stage and after completion fabrication and erection, the piping system shall be inspected by the Consultant Representative / Engineer- in - Charge to ensure that :

- Proper piping material has been used.
- Piping has been erected as per drawings and the instruction of the engineer- in charge.
- All supports have been installed correctly.
- Test preparations mentioned in this specification have been carried out.

## 4.0 FLUSHING

Flushing of all lines shall be done before pressure testing.

Flushing shall be done by 'fresh potable water' or 'dry compressed air, wherever water flushing is not desirable' to clean the pipe of all dirt, debris or loose foreign materials.

Required pressure of water, flushing shall meet the fire hydrant pressure or utility water pressure. For air flushing the line, system will be pressurised by compressed air at the required pressure which shall be 50 psi maximum. The pressure shall then be released by quick opening of a valve, already in the line for this purpose. This procedure shall be repeated as many times as required till the inside of the pipe is fully cleaned.

In line instruments like control valves, orifice plates, rotameters, safety valves and other instruments like thermowells which may interfere with flushing shall not be included i m the flushing circuit.

		STANDARD TECHNICAL SPECIFICATION			
	834002	OIL & GAS SBU, DELHI		विकान मुकान मुकान	
			DOCUMENT NO.	Page 2 of 8	
	TITLE	FLUSHING AND TESTING OF	MEC/S/05/21/11	REVISION: 0	
		PIPING SYSTEMS		EDITION: 1	
From all permanent strainers the screens/meshes shall be removed before flu Screens/meshes shall be re- installed after flushing but before testing. During flushing temporary strainers shall be retained. These shall be removed, or and reinstalled after flushing, but, before testing. In case an equipment such as column, vessel, exchanger etc. forms part of a circuit during flushing, this shall be done with the approval of Engineer- in - O However equipment thus included in the circuit, shall be completely cleaned an with comprss4ed air, after flushing is completed.				removed before flushing. re testing. shall be removed, cleaned etc. forms part of a piping of Engineer- in - Charge. upletely cleaned and dried	
		<ul><li>Engineer- in - Charge. If necessary, proper temporary drainage shall be provided by the contractor.</li><li>Care shall be taken during flushing so as not to damage/spoil work of other agencies. Precautions shall also be taken to prevent entry of water/foreign matter into equipment, electric motors, instruments, electrical installations etc. in the vicinity of lines being flushed.</li></ul>			
		The contractor shall carry out all the activities required before, during and after th flushing operation, arising because of flushing requirements, such as but not limited to the following. Dropping of valves, specials, distance pieces, online instruments and any other pipin part before flushing. The flanges to disengaged for this purpose shall be envisaged be the contractor and approved by the Engineer-in-Charge. These flanges shall be provided with temporary gaskets at the time of flushing.			
		After flushing is completed and approved, the valve, distance pieces, piping specetic shall be re-installed by the contractor with permanent gaskets. However, flange equipment nozzles and other places where isolation is required during testing, temporary gaskets shall be provided. Records in triplicate shall be prepared and submitted by the Contractor for each pip system for the flushing done in the proforma provided / approved by EIC.			
	5.0	TESTING			
		Pressure testing, in general shal otherwise specified, herein. Lines have a sensitive leak test. For IBR	I be as per clause 345 carrying highly hazardous lines, 'IBR Regulations' sha	of ASME B31.3, unless s / poisonous fluids must all also be followed.	

MECON LIMITED REGD. OFF: RANCHI	CHI STANDARD TECHNICAL SPECIFICATION OIL & GAS SBU, DELHI		सित्त कान सित्त मेकान सित्त गर्द्र कार्य
834002			
		DOCUMENT NO.	Page 3 of 8
TITLE	FLUSHING AND TESTING OF	MEC/S/05/21/11	<b>REVISION: 0</b>
	PIPING SYSTEMS		EDITION: 1

#### 5.1 <u>Extent of testing</u>

With the exclusion of instrumentation, piping system fabricated or assembled in the field shall be tested irrespective of whether or not they have been pressure tested prior to site welding of fabrication.

To facilitate the testing of piping systems, vessels and other equipments may be included in the system with the prior approval of Engineer-in-charge, if the test pressure specified is equal to or less than that for the vessels and other equipments.

Pumps, compressors and other votary equipments shall not be subjected to field test pressures.

Lines which are directly open to atmosphere such as vents, drains, safety valves, discharge need not be tested, but all joints shall be visually inspected wherever necessary such lines shall be tested by continuous flow of fluid to eliminate the possibility of blockage. However, such lines if provided with block valve shall be pressure tested upto the first block valve.

Seats of all vales shall not be subjected to a pressure in excess of the maximum cold welding pressure of the valve. Test pressure applied to vales shall not be grater than the manufacturer is recommendation nor less than that required by the applicable code. Where desirable set pressure is less than test pressure, test shall be made through an open valve.

Instruments in the system to be tested, shall be excluded from the test by isolation or removal, unless approved otherwise by the Engineer-in-charge. Restrictions which interfere with filling, venting and drawing such as orifice plates etc. shall not be installed unless testing is complete.

Control valves shall not be included in the test system. Where by-passes are provided test shall be performed through the by-pass end/or necessary spool shall be used in place of the control valve.

Pressure gauges which are part of the finished system, but cannot withstand test pressure shall not be installed until the system has been tested. Where piping systems to be tested are directly connected at the battery limits to piping for which the responsibility tests with other agencies, the piping to be tested shall be isolated from such piping by physical disconnection such as valves or blinds.

#### 5.2 <u>General Requirement/Test preparation for Testing</u>

Test shall be carried out with permanent gaskets installed unless specified otherwise or instructed by the Engineer-in- charge.

MECON LIMITED	STANDARD TECHNICAL SPECIFICATION						
834002	OIL & GAS SBU	OIL & GAS SBU, DELHI					
		DOCUMENT NO.	Page 4 of 8				
TITLE	FLUSHING AND TESTING OF	MEC/S/05/21/11	REVISION: 0				
	PIPING SYSTEMS		EDITION: 1				
	No pressure test shall be carried out against close valve unless approved by the Engineer-in-charge. The Engineer-in-charge shall be notified in advance by the contractor, of the testing sequence and programme, to enable him to be present for witnessing the test. The contractor shall be fully responsible for making arrangements with the local boiler inspector to witness the tests for steam lines falling under IBR. IBR certificates for these tests shall be obtained in the relevant IBR forms and furnished to the						
	All piping systems to be hydrostat the systems purged of air before th	ically tested shall be vent te test pressure is applied.	ed at the high points and				
	Wherever in the line any void is evalve, safety valve, check valves et	existing due to any reasor c. it shall be filled with tem	ns, for absence of control porary spools.				
	All joints welded, screwed or flang test. Before pressuring the lines, e it from rest and any other foreign m	ed shall be left exposed f each weld joint shall be clea natter.	or examination during the aned by wire brush to free				
	Where a system is to be isolated inserted between the companion designed in accordance with applic	l of a pair of companion flanges. Minimum thickne able design code.	flanges, a blank shall be ess of the blank shall be				
	Open ends of piping system w compressors, turbines or whereve disconnected prior to hydrostatic te flanges of same rating as the piping	where blanks cannot be r equipment or pipe spool esting, shall be blinded – c g system being tested.	used, such as pumps, have been receivered or ff by using standard blind				
	Pressure gauges used in testing shall be installed as close as possible to the lowest point in the piping system to be tested, to avoid overstressing of any of the lower portion of the system. For longer lines and vertical lines, two or more pressure gauges shall be installed at locations selected by the Engineer-in-charge. For lines containing check valves any of the following alternatives shall be adopted for pressure testing. Wherever possible pressurise up-stream side of valve.						
	Replace the valve by a temporary spool and re-install the valve after testing.						
	Provide blind on valve flanges and test the upstream and downstream of the line separately and remove the blind after testing. All these flanges, temporary gaskets shall be provided during testing and shall be replaced by permanent gaskets subsequently. For check valves in lines 1-1/2" and below, flapper or seat shall be						

MECON LIMITED	STANDARD TECHNICAL	STANDARD TECHNICAL SPECIFICATION OIL & GAS SBU, DELHI		
834002	OIL & GAS SBU			
		DOCUMENT NO.	Page 5 of 8	
TITLE	FLUSHING AND TESTING OF	MEC/S/05/21/11	REVISION: 0	
	PIPING SYSTEMS		EDITION: 1	
	removed during testing (if possible be refitted.	e). After completion of test	ting the flopper/ seat shall	
	Gas lines when hydrostatically to supports during testing as directed	ested shall be provided v by Engineer-in-charge.	with additional temporary	
	Piping which is spring or counter where the weight of the fluid wou supports shall be removed only at drained.	<ul> <li>weight supported shall link</li> <li>weight support overload the support.</li> <li>ter testing is completed a</li> </ul>	be temporarily supported, Retaining pins for spring nd test fluid is completely	
	When testing any piping system, air or steam of approximately 2 kg/cm <sup>2</sup> (g) may be used as preliminary test to detect missing gaskets etc. as this avoids the necessity of purging the gas to make repairs. However, this method may not be used for this purpose, if the steam temperature is more than the design temp. of the line.			
	For jacketed pipes testing of core pipe is continuously packed, befor separately as a system for piping jacket shall be tested as separate s	pipes shall be done on in re it is jacketed. The ou with discontinuous jacketin system.	dividual pieces where the ter jacket shall be tested ng, the core pipe and the	
5.3	Testing Modes, Test pressure and	Test Pressure Gauges		
5.3.1	Testing Modes			
	In general all pressure test shall be hydrostatic using iron free water, which is clean and free of silt. Maximum clorine content in water for hydrostatic testing for MS piping shall be 15-20 ppm. Air shall be used for testing only if water would cause corrosion of the system or overloading of supports etc. in special cases as directed by Engineer-in-charge.			
	If operating fluid in the line is much lighter than testing fluid, the additional weight of testing fluid may render piping supports (as designed) inadequate. This will call for additional temporary supports. The typical examples are flare and vapor lines. It is preferable that hydrostatic testing is avoided in such systems and instead pneumatic testing may be specified.			
	Where air/water tests are undesir shall be used as the testing media These test fluids shall be specified	able substitute fluid such a um, with due consideration in the line list given to the t	as gas, oil, methanol etc. n to the hazards involved. contractor.	

MECON LIMITED	STANDARD TECHNICAL		
834002	OIL & GAS SBU, DELHI		मिकान मिकान 8001:2000 Contra
		DOCUMENT NO.	Page 6 of 8
TITLE	FLUSHING AND TESTING OF	MEC/S/05/21/11	REVISION: 0
	PIPING SYSTEMS		EDITION: 1

#### 5.3.2 <u>Test Pressure</u>

The hydrostatic/pneumatic test pressure shall be as indicated in the line list or as per the instruction of Engineer-in-charge.

The selection of the piping system for one individual test shall be based on the following :-

Test pressure required as per line list.

Maximum allowable pressure for the material of construction of piping depending upon the above requirements and based on construction progress, maximum length of piping shall be included in each test.

## 5.3.3 <u>Test Pressure Gauge</u>

All gauge used for field testing shall have suitable range so that the test pressure of the various system falls in 35% to 65% of gauge scale range. Pressure gage shall be minimum of 150 mm. Size of Bourdon shall not be less than 75% of nominal diameter of dial range. Gauge shall be of a good quality and in first class working condition.

Prior to the start of any test or periodically during the field test programmes, all test gauges shall be calibrated using a standard dead weight gauge tester or other suitable approved testing apparatus. Any gauge having an incorrect zero reading or error of more than  $\pm 2\%$  of full scale range shall be discarded. The Engineer-in-charge shall check the accuracy of master pressure gauge used for calibration.

## 5.4 <u>Testing Pressure</u>

## 5.4.1 <u>Hydrostatic Test</u>

All vents and other connections used as vents shall be kept open while filling the line with test fluid for complete removal of air. For pressurising and depressurising the system, temporary isolating valves shall be provided if valves, vents, drains do not exist in the system.

Pressure shall be applied only after the system/line is ready and approved by the Engineer-in-charge.

Pressure shall be applied by means of a suitable test pump or other pressure source which shall be isolated from the system as the desired test pressure is reached and stabilised in the system.

	STANDARD TECHNICAL	STANDARD TECHNICAL SPECIFICATION OIL & GAS SBU, DELHI		
834002	OIL & GAS SBU,			
		DOCUMENT NO.	Page 7 of 8	
TITLE	FLUSHING AND TESTING OF	MEC/S/05/21/11	REVISION: 0	
	PIPING SYSTEMS		EDITION: 1	
	A pressure gauge shall be provide the required pressure.	ed at the pump discharge t	for guiding the system to	
	The pump shall be attended const pump shall be isolated from the sys	antly during the test by ar stem wherever the pump is	n authorised person. The to be left unattended.	
	Test pressure shall be maintained inspection of all joints for leakage pressure test, shall be re-tested to be maintained for a minimum of fou	d for a sufficient length o or signs of failure. Any jo the specified pressure afte ur hours.	of time to permit through int found leaking during a er repair. Test period shall	
	The pump and the piping system to be tested are to be provided with separate pressure indicating test gauges. There gauges are to be checked by the standard test gauge before each pressure test.			
	Care shall be taken to avoid increduring the test.	ease in the pressure due	to atmospheric variation	
5.4.2	<u>Air Test</u>			
	When testing with air, pressure s compressor shall be portable type v	hall be supplied by mean with a receiver after cooler	s of a compressor. The & oil separator.	
	Piping to be tested by air shall have that the joints can be examined for	ve joints covered with a so leaks.	pap and water solution so	
	All other activities shall be same as	per hydrotesting procedur	re (specified above).	
5.5	Completion of Testing			
	After the hydrostatic test has been completed, pressure shall be released in a manner and at a rate so as not to endanger personnel or damage equipments.			
	All vents and drains shall be opened before the system is to be drained and shall remain open till all draining is complete, so as to prevent formation of vacuum in the system. After draining lines/systems shall be dried by air.			
	After testing is completed the test blinds shall be removed and equipment/piping isolated during testing shall be connected using the specified gaskets, bolts and nuts. These connections shall be checked for tightness in subsequent pneumatic tests to be carried out by the contractor for complete loop/circuit including equipments (except rotary equipments).			

MECON LIMITED REGD. OFF: RANCHI	D STANDARD TECHNICAL SPECIFICATION		
834002	OIL & GAS SBU,	यहार मेकान 9001:2000 Confu	
		DOCUMENT NO.	Page 8 of 8
TITLE	FLUSHING AND TESTING OF	MEC/S/05/21/11	REVISION: 0
	PIPING SYSTEMS		EDITION: 1

Pressure tests shall be considered complete only after approved by the Engineer-in-charge. Defects, if any, noticed during testing shall be rectified immediately and retesting of the system/line shall be done by the contractor at his cost.

# 5.6 <u>Test Records</u>

Records in triplicate shall be prepared and submitted by the contractor for each piping system, for the pressure test done in the proforma provided / approved by the Engineer-in-charge.

# ADDITIONAL NOTES FOR VALVES



#### OIL & GAS SBU, DELHI

- **1.0** <u>Compliance with Specification:</u> The valve manufacturer shall be completely responsible for the design, materials, manufacture & fabrication, testing, inspection, preparation for shipment and transport of the valves. Minimum all pressure containing and pressure controlling parts of valves and actuators shall be provided with EN 10204-3.2 certificates.
- **2.0** <u>Valve manufacturer's Scope:</u> Valve manufacturer scope of work includes the equipment with all internals and accessories shown on the datasheets, specifications and all unmentioned parts necessary for a satisfactory operation and testing, except those which are indicated to be out of the valve manufacturer's supply.

#### 3.0 Inspection:

Inspection shall be in accordance with EN 10204 3.2 certification shall be issued for each dispatched valve. Valve manufacturer shall appoint anyone of the TPIA for inspection purpose. Valve manufacturer has to intimate the TPIA name to MECON prior to perform any inspection activity.

**4.0** Apart from inspection by TPIA, Inspection shall also be performed by MECON and or its authorized representative / MECON and or its authorized inspection agency (AIA), as set out and specified in the codes and particular documents forming this tender document.

Valve manufacturer shall provide office along with furniture for MECON/ TPIA personnel for entire duration of the manufacturing. AC Office shall have table/ chair/internet connectivity/stationary/courier and printing facilities for inspection officers for round the clock inspection during manufacturing.

- **5.0** For all values to be used in Gaseous Hydrocarbons service, impact & hardness tests / values as per clause 3.4, 3.5 & 3.6 of specification no. MEC/TS/05/21/002 shall be applicable.
- **6.0** Valve manufacturer shall inform separately spares for two years normal operation for valves & actuators, if applicable .
- **7.0** Valve manufacturer to include the start up and commissioning spares for valves & actuators (if applicable) in the quoted price for valves.
- **8.0** Valve manufacturer to indicate in his offer the gross weight (in kg or Metric Tonne) per unit, volume (in m3) per unit and dimensions (L x B x H) of package (wooden box, etc.) to accommodate unit quantity or number of quantities (as applicable).
- **9.0** Valve manufacturer must submit duly filled up & signed data sheets, check list and forms along with his offer.

# ADDITIONAL NOTES FOR VALVES



#### OIL & GAS SBU, DELHI

- **10.0** Valve manufacturer shall establish the equivalence/superiority of any material proposed (With justification of material properties and availability) other than that specified in Datasheet. Vendor shall also indicate the ASTM equivalent of his proposed material as well as of all the AISI designated materials specified in datasheets.
- **11.0** Valve manufacturer to note that for minimum inspection and testing requirement of the valves shall be governed by attached QAP. However, valve manufacturer shall submit their QAP for approval covering the requirement specified in attached QAP.
- **12.0** In the event of Conflict/inconsistency among the documents attached/ referred, the following order of precedence generally shall govern in interpretation of various requirements / data.
  - Additional notes for valves
  - Datasheets
  - Technical Specification
  - Codes and Standards
  - Vendor's Standards

However, Owner/Consultant reserves the right to consider most stringent requirement among the document attached / referred.

- **13.0** Successful bidder shall submit hard copies of all documents/ drawings to MECON for review and approval. The date of receipt of these documents/ drawings at MECON shall be deemed as the date of submission. If any documents/ drawings require re-submission due to any error/ deficiency noticed during review/ approval stage, in that event the additional time required by the successful bidder to get the revised document/ drawing reviewed/approved by MECON shall be solely to successful bidder's account and in no case the vendor/ supplier shall be entitled for any time or cost benefit.
- **14.0** Successful bidder to note that the valves supplied by them shall be capable to withstand the field hydro test pressure (i.e., 1.5 times of design pressure) for 6 to 24 hours test holding duration under field / site conditions. The valve's obturator shall be kept in either partial or full open condition for entire test duration and test medium will be non-corrosive water. Successful bidder shall be liable for repair/ replacement of valve if found faulty during site hydro test at his risk & cost. All cost for associated activities like packaging, transportation etc. in connection to repair / replacement of valve shall be borne by the successful bidder. No claim shall be entertained by the Owner / Purchaser in this regard.
- **15.0** Valve manufacturer to note that packing & transportation of the valves shall be done strictly as per attached technical specification for handling and transportation.
- **16.0** Valve manufacturer to note that the entire ordered quantity shall be offered for MECON's inspection as per following table. In case no. of visits of MECON engineer become more

# ADDITIONAL NOTES FOR VALVES

Page 3 of 3

#### OIL & GAS SBU, DELHI

0 - -

than as specified in table below for complete order quantity, vendor shall bear the touring expenditure of MECON engineers as per company rules. MECON reserves the right to waive off this requirement in case of project exigencies.

S.No.	Size	Minimum Quantity for one lot
1	30" and higher	Upto 5 valves
2	16" to 28"	Upto 8 valves
3	8" to 14"	Upto 20 valves
4	<sup>3</sup> ⁄ <sub>4</sub> " to 6"	Upto 200 valves

In case of any multiple of the ordered quantity the no. of valves shall be divided by quantity specified for one lot in above mentioned table to arrive at the no. of lots. No. of lots shall be determined by rounding off to next integer.

- **17.0** Valve manufacturer to note that TPI inspection is either to be conducted before MECON inspection or in parallel. In no case TPI inspection shall be permitted after MECON inspection. For the valves where MECON inspection extent is 100% witness, TPI inspection maybe allowed in parallel with MECON. However, for valves requiring 10% MECON witness inspection, vendor has to finish TPI inspection before raising call and upload TPI inspection report in Inspection Management System of MECON.
- **18.0** Extent of MECON witness during final inspection shall be as follows:

SI. No.	Size range	Class	Mecon Inspection extent
1.	2" to 8"	150	10% random witness and document review for 100% valves.
2.	10" and more	150	100% witness.
3.	2" to 6"	300 and higher	10% random witness and document review for 100% valves.
4.	8" and more	300 and higher	100% witness.
5.	Below 2"	All classes	10% random witness and document review for 100% valves.

**19.0** Strip Test: Valve manufacturer need to demonstrate strip test of bolted body ball valves. For this test one valve of each ordered type, size and rating shall be selected at random after successful hydro and pneumatic tests by TPI & MECON inspector. The valve shall be dismantled completely. Alloy steel parts shall be checked for compliance to relevant material code using Positive material identification technique. Selected valve(s) shall then be re-assembled after replacing sacrificial parts like gasket & O-rings and complete final inspection as per approved QAP shall be carried out once again to ensure the repeatability of body seals and seats.

MECON LIMITED	STANDARD TECHNICAL	SPECIFICATION						
REGD. OFF: RANCHI 834002	OFF: CHI 834002 OIL & GAS SBU, DELHI							
		DOCUMENT NO.	Page 1 of 20					
TITLE	BALL VALVE	MEC/TS/05/21/002	REVISION : 1					
			EDITION : 1					
<section-header><section-header><section-header><section-header><section-header><section-header><section-header><section-header></section-header></section-header></section-header></section-header></section-header></section-header></section-header></section-header>								
PREPARED BY:	CHECKED BY:	APPROVED B	Y: ISSUE DATE :					
	Page 426 of 644							

MECON LIMITED REGD. OFF:	STANDARD TECHNICAL		
RANCHI 834002	OIL & GAS SBU,	मेवर्डन	
		DOCUMENT NO.	Page 2 of 20
TITLE	TITLE BALL VALVE MEC/TS/05/21/002		REVISION : 1
			EDITION : 1

# **AMENDMENT STATUS**

SI.	Clause / Paragraph /	Page	Boy	Dato	Ву		Verifie	d
No.	Drawing Amended	No.	Nev.	Date	Name	Sig.	Name	Sig.
1.	Cl. No. 4.6	4	1	April 09	Gurdeep Singh		K.K. De	
2.	Overall Revision	All	1	July 20	K.P. Singh		A.K. Tyagi	

MECON LIMITED	STANDARD TECHNICAL		
RANCHI 834002	OIL & GAS SBU	मेक्ट्रेन के केवरा Committ	
		DOCUMENT NO.	Page 3 of 20
TITLE	BALL VALVE	REVISION : 1	
			EDITION : 1

# Abbreviations :

ASME	:	American Society of Mechanical Engineers
ASTM	:	American Society for Testing and Materials
API	:	American Petroleum Institute
BHN	:	Brinell hardness number
DN	:	Nominal Size
HAZ	:	Heat Affected Zone
LC	:	Lock Close (valve locked in full close position)
LO	:	Lock Open (valve locked in full open position)
MSS-SP	:	Manufacturers Standardization Society - Standard Practice
NDT	:	Non Destructive Testing
NPS	:	Nominal Pipe Size
RTJ	:	Ring Type Joint
SSPC	:	Steel Structures Painting Council

MECON LIMITED		STANDARD TECHNICAL		
RANCHI	334002	OIL & GAS SBU, DELHI		itatir
			DOCUMENT NO.	Page 4 of 20
ТІ	TLE	BALL VALVE	MEC/TS/05/21/002	REVISION : 1
				EDITION : 1
SI.	No.	Description		
1.0	)	SCOPE		
2.0	)	REFERENCE DOCUMENTS		
3.0	)			
4.0	)	DESIGN AND CONSTRUCTION		
5.0	)	INSPECTION AND TESTS		
6.0	)	EXTENT OF INSPECTION & TES	TING	
7.0	)	TEST CERTIFICATES		
8.0	)	PAINTING, MARKING AND SHIP		
9.0	)	SPARES AND ACCESSORIES		
10.	.0	DOCUMENTATION		
11.	.0	GUARANTEE		
FIG	SURE-1	VENT, DRAIN & SEALANT INJECT		

MECON LIMITED REGD. OFF:	STANDARD TECHNICAL		
RANCHI 834002	OIL & GAS SBU,	Hater to see comment	
		DOCUMENT NO.	Page 5 of 20
TITLE	BALL VALVE MEC/TS/05/21/002		REVISION : 1
			EDITION : 1

## 1.0 <u>SCOPE</u>

This specification covers the minimum requirements for design, manufacture, testing and supply of carbon steel ball valves of size DN 50 mm (2") and above and ANSI pressure rating class 150 to 900 to be used in on-shore pipeline systems handling non-sour hydrocarbons in liquid or gaseous phase, including Liquefied Petroleum Gas (LPG).

This specification does not cover ball valves for sour hydrocarbon (liquid / gas) service as defined in NACE standard MR-01-75.

# 2.0 **REFERENCE DOCUMENTS**

- 2.1 All valves shall be manufactured and supplied in accordance with the latest edition of American Petroleum Institute (API) Specification 6D / ISO 14313, with additions and modifications as indicated in the following sections of this specification.
- 2.2 Reference has also been made in this specification to the latest edition of the following Codes, Standards and Specifications:

ASME B 16.5	:	Pipe flanges and flanged fittings
ASMEB 16.10	:	Face-to-face and end-to-end dimensions of valves
ASME B 16.25	:	Butt welding ends
ASME B 16.34	:	Valves – flanged, threaded and welding ends
ASME B16.47	:	Large diameter steel flanges
ASME B 31.3	:	Process piping
ASME B 31.4	:	Pipeline transportation systems for liquid hydrocarbons and other liquids
ASME B 31.8	:	Gas transmission and distribution piping systems
ASME Sec VIII	:	Boiler and pressure vessel code - Rules for construction of pressure vessels
ASME Sec IX	:	Boiler and pressure vessel code - Welding and brazing qualifications
ASTM A 370	:	Standard test methods and definitions for mechanical testing of steel products
ASTM B 733	:	Autocatalytic nickel phosphorous coating on metals
API 6FA	:	Fire test for valves
		Page 430 of 644

MECON LIMITED REGD. OFF:	STANDARD TECHNICAL SPECIFICATION				
RANCHI 834002	OIL & GAS	S SBU, DELHI	मेवरून		
		DOCUMENT NO.	Page 6 of 20		
TITLE	BALL VALVE	MEC/TS/05/21/002	REVISION : 1		
			EDITION : 1		
	API 607 :	Fire test for soft-seated quarte	r-turn valves		
	API 1104 :	Welding of pipelines and relate	ed facilities		
	BS EN ISO 10497 :	Testing of valves – Fire type-te	esting requirements		
	MSS-SP-6 :	Standard finishes for contact to connecting-end flanges of values	faces of pipe flanges and /es and fittings		
	MSS-SP-44 :	Steel pipeline flanges			
	SSPC-VIS-1 :	Steel structures painting counc	cil-visual standard		
2.3	In case of conflict betweer Codes, Standards and Spec of this specification shall gov	n the requirements of this spec difications referred in clause 2.2 vern. Order of precedence shall	ification, API 6D and the above, the requirements be as follows :		
	<ul> <li>Valve Data Sheets</li> <li>Material Requisition</li> <li>This Specification</li> <li>API 6D Specification</li> <li>Other Referred Code</li> <li>Manufacturer's Stance</li> </ul>	<ul> <li>Valve Data Sheets</li> <li>Material Requisition</li> <li>This Specification</li> <li>API 6D Specification</li> <li>Other Referred Codes &amp; Standards</li> <li>Manufacturer's Standard</li> </ul>			
3.0	MATERIALS				
3.1	Material for major components of the valves shall be as indicated in Valve Data Sheet. Other components shall be as per Manufacturer's standard (suitable for the service conditions indicated in Data Sheet) and shall be subject to approval by Purchaser. In addition, the material shall also meet the requirements specified hereinafter.				
3.2	Carbon steel used for the ma	anufacture of valves shall be ful	ly killed.		
3.3	The Carbon Equivalent (CE) of valve end connections which are subject to further field welding by Purchaser, shall not exceed 0.43% (as calculated by the following formula) on check analysis for each heat of steel used:				
	%Mn %	%Ni + %V %Ni + %	6Cu		
	CE = %C + + 6	+ 5 15			
3.4	For Valves specified to be used for Gas service or LPG service, Charpy V-notch test, on each heat of base material shall be conducted as per API 6D Clause 8.5, for all pressure containing parts such as body, end flanges and welding ends as well as bolting material for pressure containing parts. Unless stated otherwise, the Charpy V-notch test shall be conducted at 0 °C. Test procedure shall conform to ASTM A370. The average absorbed energy value of three full sized specimens shall be 27 J. The				

Page 431 of 644

MECON LIMITED	STANDARD TECHNICAL SPECIFICATION		Trater B and common	
RANCHI 834002	OIL & GAS SBU, DELHI			
		DOCUMENT NO.	Page 7 of 20	
TITLE	BALL VALVE	MEC/TS/05/21/002	REVISION : 1	
			EDITION : 1	
	minimum impact energy value for any one specimen of the three specimens analysed as above, shall not be less than 22 J.			
	When Low Temperature Carbon Steel (LTCS) materials are specified in Valve Data Sheet or offered by Manufacturer, the Charpy V-notch test requirements of applicable material standard shall be complied with.			
3.5	For all such valves where carbon steel is used as ball material, the ball shall have 75 micrometer (0.003 inch) thick Electroless Nickel Plating (ENP) as per ASTM B733 with following classification : SC2, Type II, Class 2. The hardness of plating shall be minimum 50 RC.			
3.6	For valves specified to be used for Gas service or LPG service, hardness test shall be carried out as per ASTM A370 for each method of manufacture and each heat of steel used in the manufacture of valves. A full thickness cross-section shall be taken for this purpose and the maximum hardness of the materials of valve components shall not exceed 248 $HV_{10}$ .			
3.7	All process-wetted parts, metallic and non-metallic, shall be suitable for the fluids and service specified by the Purchaser. The service gas composition shall be as given elsewhere in the Material Requisition. In addition, Manufacturer shall confirm that all wetted parts are suitable for treated water / seawater environment, which may be used during field testing.			
3.8	Non-metallic parts of the valves (including O-rings, soft seal etc.) intended for hydrocarbon gas service at pressures of PN 100 (600 #) and above shall be resistant to explosive decompression.			
4.0	DESIGN AND CONSTRUCTION			
4.1	Valve design shall meet the requirements of API 6D and other referred codes and shall be suitable for the service conditions indicated in Valve Data Sheet. The ASME Boiler & Pressure Vessel Code, Section VIII, Division 1, may be used to design the valve body. Allowable stress requirements shall comply with the provisions of ASME B31.3. In addition, corrosion allowance indicated in Valve Data Sheet shall be considered in valve design. However, the minimum wall thickness shall not be less than the minimum requirement of ASME B16.34. The Manufacturer shall have a valid license to use API 6D monogram for manufacture of ball valves.			
4.2	For above ground valves, valve body design shall be either fully welded or bolted type, as indicated in Valve Data Sheet. Valve body joints with threads are not permitted.			
	For buried values, value body design shall be fully welded type only. Value body joints with bolts or threads are not permitted.			

4.3 Ball shall be of single piece, solid type construction.
MECON LIMITED	STANDARD TECHNICAL SPECIFICATION		
RANCHI 834002	OIL & GAS SBU, DELHI		मेवरेन 10 august comment
		DOCUMENT NO.	Page 8 of 20
TITLE	BALL VALVE	MEC/TS/05/21/002	REVISION : 1
			EDITION : 1

4.4 Valves shall be Full Opening (FO) or Reduced Opening (RO) as indicated in Valve Data Sheet. FO valves shall be suitable for the passage of all types of pipeline scraper and inspection pigs on regular basis without causing damage to either the valve component or the pig. The FO valve shall provide an unobstructed profile for pigging operations in either direction. FO valves shall be designed to minimize accumulation of debris in the seat ring region to ensure that valve movement is not impeded.

The opening size of RO valves shall be corresponding to that of a FO valve of smaller nominal diameter as indicated in table below. For sizes of a particular rating not covered in API 6D, the opening sizes of the RO valve shall be as per Manufacturer's standard.

Nominal Valve Size	Nominal Valve Size for Reduced Opening	Nominal Valve Size	Nominal Valve Size for Reduced Opening
DN mm (NPS inches)	DN <sub>mm</sub> (NPS <sub>inches</sub> )	DN <sub>mm</sub> (NPS <sub>inches</sub> )	DN <sub>mm</sub> (NPS <sub>inches</sub> )
50 (2)	50 (2)	600 (24)	500 (20)
80 (3)	50 (2)	650 (26)	550 (22)
100 (4)	80 (3)	700 (28)	600 (24)
150 (6)	100 (4)	750 (30)	600 (24)
200 (8)	150 (6)	800 (32)	650 (26)
250 (10)	200 (8)	850 (34)	700 (28)
300 (12)	250 (10)	900 (36)	750 (30)
350 (14)	250 (10)	950 (38)	800 (32)
400 (16)	300 (12)	1000 (40)	850 (34)
450 (18)	350 (14)	1050 (42)	900 (36)
500 (20)	400 (16)	1200 (48)	1050 (42)
550 (22)	450 (18)		

4.5

Ball mounting shall be trunnion / pivot type or as indicated in Valve Data Sheet. Ball mounting, either trunnion or floating, unless otherwise specified, shall be as follows.

SI.	ANSI Prossure Pating	Nominal Valve Size (NPS inches)		
No.	ANOI FIESSULE Mailing	Floating Ball	Trunnion Mounted	
1.	150#	<u>&lt;</u> 8"	> 8"	
2.	300#	<u>&lt;</u> 4"	> 4"	
3.	600#	Nil	<u>&gt;</u> 2"	

Valve design shall minimize the possibility of debris ingress into the trunnion as far as practicable.

MECON LIMITED REGD. OFF:	ED STANDARD TECHNICAL SPECIFICATION			
RANCHI 834002	OIL & GAS SBU	OIL & GAS SBU, DELHI		
		DOCUMENT NO.	Page 9 of 20	
TITLE	BALL VALVE	MEC/TS/05/21/002	REVISION : 1	
			EDITION : 1	
4.6	Valve seats shall have metal to metal contact. O-rings or other seals, if used for drip tight sealing, shall be encased in a suitable groove in such a manner that it can not be removed from seat ring and there is no extrusion during opening or closing operation of valve at maximum differential pressure corresponding to valve class rating. The seat rings shall be so designed as to ensure sealing at low as well as high differential pressures.			
4.7	Valves shall have double block an draining and venting of the valve b	Valves shall have double block and bleed feature to facilitate complete flushing, draining and venting of the valve body cavity.		
4.8	For valves to be used in liquid prevented by self relieving seat ri is not permitted. Self relieving se pressure not exceeding 50% of the	For valves to be used in liquid service, the body cavity over-pressure shall be prevented by self relieving seat rings / assemblies. A pressure relief hole in the ball is not permitted. Self relieving seat rings shall relieve at a body cavity differential pressure not exceeding 50% of the valve class rating pressure.		
4.9	Valves shall be designed to withstand a sustained internal vacuum of at least 1 (one) milli-bar in both open and closed positions.			
4.10	FO valves of nominal size DN 200 mm (8") & above and RO valves of nominal size DN 250 mm (10") & above shall have provision for secondary sealant injection under full line pressure for seat and stem seals. All sealant injection connections shall be provided with a needle valve, a grease fitting and non-return valve. Valve design shall have a provision to replace the sealant injection fitting under full line pressure. Location and arrangement of sealant points shall be as per Figure-1.			
4.11	Valves shall be provided with vent and drain connections. Location and arrangement of vents and drains shall be as per Figure-1. Body vent and drain shall be provided with valves (ball or plug type). Number and size shall be as per Figure-1.			
4.12	Valve design shall ensure repair of stem seals / packing under full line pressure.			
4.13 a)	Valve ends shall be either flanged or butt welded or one end flanged and one end butt welded as indicated in Valve Data Sheet. Flanges of the flanged end cast/ forged body valves shall be integrally cast / forged with the body of valve. Face-to- face/ end-to-end dimensions shall conform to API 6D. Face-to-face and end-to-end dimensions for valve sizes not specified in API 6D shall be in accordance with ASME B 16.10. Face-to-face and end-to-end dimensions not shown in API 6D or in ASME B 16.10 shall be as per Manufacturer Standard and shall be subject to approval by Purchaser.			
b)	Flanged ends shall have flanges mm (24 inches) excluding DN 550 16.47 series A for valve sizes DN and above. Flange face shall b indicated in Valve Data Sheet. F indicated in Valve Data Sheet. S microinches AARH. In case of RT 140 BHN.	as per ASME B16.5 for v mm (22 inches) and as p 550 mm (22 inches) & for e either raised face or lange face finish shall be smooth finish when speci TJ flanges, the groove ha	valve sizes up to DN 600 ber MSS-SP-44 / ASME B r DN 650 mm (26 inches) ring joint type (RTJ) as e serrated or smooth as fied shall be 125 to 200 rdness shall be minimum	

MECON LIMITED	STANDARD TECHNICA	STANDARD TECHNICAL SPECIFICATION		
RANCHI 834002	OIL & GAS SBU, DELHI		Hatir	
		DOCUMENT NO.	Page 10 of 20	
TITLE	BALL VALVE	MEC/TS/05/21/002	REVISION : 1	
			EDITION : 1	
c)	Butt weld end preparation shall be as per ASME B16.25. The thickness of the pip to which the valve has to be welded shall be as indicated in Valve Data Shee Valves shall be without transition pups, unless otherwise specified in Valve Data sheet. In case significant difference exists between thickness of welding ends valve and connecting pipe, the welding ends of valve shall have bevel preparation a per ASME B31.4 or ASME B31.8, as applicable.			
4.14	Design of weld end valves shall seals or plastic components of damaged. The Manufacturer sh post-weld test procedure to den field welding operations.	Design of weld end valves shall be such that during field welding operations, the soft seals or plastic components of the valve (where ever used) are not liable to be damaged. The Manufacturer shall furnish necessary field welding instructions and post-weld test procedure to demonstrate integrity and leak-tightness of valves after field welding operations.		
4.15	Valves shall be provided with ba at the fully open and fully closed	Valves shall be provided with ball position indicator and stops of rugged construction at the fully open and fully closed positions.		
4.16	FO valves of nominal size $\ge$ DN 200 mm (8") and RO valves of nominal size $\ge$ DN 250 mm (10") shall be equipped with support foot and lifting lugs. Tapped holes and eye bolts shall not be used for lifting lugs. Height of support foot shall be kept a minimum. The location and size of support foot / lifting lugs shall ensure unrestrictive operation of vent / drain valves.			
4.17	Valve design shall be such as to avoid bimetallic corrosion between carbon steel and high alloy steel components. Suitable insulation shall be provided as required.			
4.18	Valves shall be of fire resistant design as per API 607/BS EN ISO 10497/API 6FA, as indicated in Valve Data Sheet.			
4.19	Valves shall be provided with anti-static devices to ensure electrical continuity between stem / ball and valve body.			
4.20	Valves shall be suitable for either buried or above ground installation as indicated ir Valve Data Sheet.			
4.21	When stem extension requirements have the following provisions :	ent is indicated in Valve Da	ta Sheet, the valves sha	
	a) Valves provided with ste Length of stem extensio length indicated correspo opening and the top of operator / power actuator	em extension shall have v n shall be as indicated in onds to the distance betwee mounting flange for valve as applicable).	vater proof outer casing Valve Data Sheet. The en centerline of the valve e operating device (gea	
	b) Vent and drain connecti adjacent to the valve op- valve body. Pipe used si 80. Fittings shall be AS <sup>-</sup> ANSI class 6000.	ons and sealant injection erator by means of suitabl hall be API 5L Gr. B / AST TM A 105 / ASTM 234 G	lines shall be terminated e piping anchored to the M A 106 Gr. B, with Sch r. WPB, Socket Welded	

MECON LIMITED	STANDARD TECHNICAL SPECIFICATION		
RANCHI 834002	OIL & GAS SBU, DELHI		मेवरन
		DOCUMENT NO.	Page 11 of 20
TITLE	BALL VALVE	MEC/TS/05/21/002	REVISION : 1
			EDITION : 1

- c) Stem extension and stem housing design shall be such that the complete assembly will form a rigid unit giving a positive drive under all conditions with no possibility of free movement between valve body, stem extension or its operator.
- d) Outer casing of stem extension shall have 3/8" or ½" NPT plugs at the top and bottom, for draining and filling with oil to prevent internal corrosion.

### 4.22 **Operating Devices**

- a) Valves shall have a power actuator or manual operator as indicated in Valve Data Sheet. In case of manual operator, valve sizes ≤ DN 100 mm (4 inches) shall be wrench operated and valve sizes ≥ DN 150 mm (6 inches) shall be gear operated. Each wrench – operated valve shall be supplied with wrench. Valve design shall be such that damage due to malfunctioning of the operator or its controls will only occur in the operator gear train or power cylinder and that damaged parts can be replaced without the valve cover being removed.
- b) The power actuator shall be in accordance with the Purchaser specification issued for the purpose and as indicated in Valve and Actuator Data Sheet. Operating time shall be as indicated in Valve Data Sheet. Valve operating time shall correspond to full close to full open/full open to full close under maximum differential pressure corresponding to the valve rating. For actuated valves, the actuator torque output shall be 1.25 times the break torque required to operate the ball valve under the maximum differential pressure corresponding to the valve class rating.
- c) For manual operator of all valves, the diameter of the hand wheel or the length of operating wrench shall conform to API 6D requirements and be such that under maximum differential pressure, the total force required to operate the valve does not exceed 350 N. Manufacturer shall also indicate the number of turns of hand wheel (in case of gear operators) required for operating the valve from full open to full close position.
- d) Direction of operation of hand wheel or wrench shall be in clock-wise direction while closing the valve. Hand wheels shall not have protruding spokes.
- e) Gear operators, when provided, shall have a self locking provision and shall be fully encased, in water proof/ splash proof/ dust proof/ weather proof enclosure and shall be filled with suitable grease.
- f) Operating devices shall be designed for easy operation of the valve under maximum differential pressure corresponding to the valve rating.
- 4.23 All welds shall be made by welders and welding procedures qualified in accordance with the provisions of ASME Section IX. The procedure qualification shall include impact test and hardness test and shall meet the requirements of clauses 3.4 and 3.6 of this specification, respectively.
- 4.24 All welds shall be stress relieved in accordance with ASME Section VIII.

Page 436 of 644

MECON LIMITED REGD. OFF:	STANDARD TECHNICAL	STANDARD TECHNICAL SPECIFICATION		
RANCHI 834002	OIL & GAS SBU, DELHI		Hatter to allow comment	
		DOCUMENT NO.	Page 12 of 20	
TITLE	BALL VALVE	MEC/TS/05/21/002	REVISION : 1	
			EDITION : 1	
4.25	Repair by welding is not permitted for fabricated and forged body valves. However repair by welding as per ASME B16.34 is permitted for cast body valves. Sur repairs shall be carried out at casting supplier's care only. Repair shall be carried of before any heat treatment of casting is done. Repair welding procedure qualification shall also include impact test and hardness test and shall meet the requirements clauses 3.4 & 3.6 of this specification, respectively.			
4.26	The tolerance on internal diamete valves shall be as per applicable Data Sheet.	The tolerance on internal diameter and out of roundness at the ends for welded en valves shall be as per applicable connected pipe specification as indicated in Valv Data Sheet.		
4.27	When indicated in Material Requisition, valves shall have locking device to lock the valve either in full open (LO) or full close (LC) positions. Locking devices shall be permanently attached to the valve operator and shall not interfere with operation the valve.			
4.28	Valve stem shall be capable of withstanding the maximum operating torque require to operate the valve against the maximum differential pressure corresponding applicable class rating. The combined stress shall not exceed the maximu allowable stresses specified in ASME Section VIII, Division I. In case of pow actuated valves, the valve stem shall be designed for maximum output torque of the selected power actuator (including gear box, if any) at valve stem.			
5.0	INSPECTION AND TESTS			
5.1	The Manufacturer shall perform all inspection and tests as per the requirements this specification and the relevant codes, prior to shipment, at his works. Suc inspection and tests shall be, but not limited to, the following:			
5.1.1	All valves shall be visually inspected. The internal and external surfaces of the value shall be free from any strikes, gouges and other detrimental defects. The surfaces shall be thoroughly cleaned and free from dirt, rust and scales.			
5.1.2	Dimensional check on all valves shall be carried out as per the Purchaser approv drawings.			
5.1.3	Chemical composition and mechanical properties shall be checked as per releva material standards and this specification, for each heat of steel used.			
5.1.4	Non-destructive examination of in of, but not limited to castings, forg by the Manufacturer.	dividual valve material an ings, plate and assembly	d components consisti welds shall be carried o	
a)	Body castings of all valves shall be surface of critical areas as per AS shall be as per ASME B16.34. Th	e radiographically examine ME B16.34. Procedure ar le extent of radiography sh	ed on 100% of the nd acceptance criteria nall be as follows:	
			Page 437 of 644	

MECON LIMITED	STANDARD TECHNICAL SPECIFICATION		
RANCHI 834002	OIL & GAS SBU, DELHI		मेवर्डन
		DOCUMENT NO.	Page 13 of 20
TITLE	BALL VALVE	MEC/TS/05/21/002	<b>REVISION</b> : 1
			EDITION : 1

ANSI Pressure Rating	Valve Size	Extent of Radiography
150 #	All sizes	Nil
300 #	<u>&lt;</u> DN 400mm (16") <u>&gt;</u> DN 450mm (18")	Nil 100%
<u>&gt;</u> 600 #	All sizes	100%

All castings shall be wet magnetic particle inspected 100% of the internal surfaces. Method and acceptance shall comply with ASME B.16.34.

b) All valves, with body fabricated from plates or made by forgings, shall be ultrasonically examined in accordance with the procedure and acceptance standard of Annexure E of ASME B16.34.

All forgings shall be wet magnetic particle inspected 100% of the internal surfaces. Method and acceptance shall comply with ASME B 16.34

- c) Bodies and bonnets made by welded assembly of segments of castings, forgings, plates or combinations thereof shall be examined, as applicable, by methods of clause 5.1.4 a) for cast components or clause 5.1.4 b) for forged components and plates.
- 5.1.5 Full inspection by radiography shall be carried out on all welds of pressure containing parts. Acceptance criteria shall be as per ASME B 31.4 or ASME B31.8, as applicable, and API 1104.
- 5.1.6 Welds, which in Purchaser's opinion cannot be inspected by radiographic methods, shall be checked by ultrasonic or magnetic particle methods and acceptance criteria shall be as per ASME Section VIII, Division 1, Appendix 12 and Appendix 6, respectively.
- 5.1.7 a) All finished wrought weld ends subject to welding in field shall be 100% ultrasonically tested for lamination type defects for a distance of 50mm from the end. Laminations shall not be acceptable.
  - b) Weld ends of all cast valves subject to welding in field shall be 100% radiographically examined and acceptance criteria shall be as per ASME B16.34.
  - c) After final machining, all bevel surfaces shall be inspected by dye penetrant or wet magnetic particle methods. All defects longer than 6.35 mm are rejected, as are defects between 6.35 mm and 1.59mm that are separated by a distance less than 50 times their greatest length. Rejectable defects must be removed. Weld repair of bevel surface is not permitted.
- 5.1.8 All valves shall be tested in compliance with the requirements of API 6D. During pressure testing, valves shall not have sealant lines and other cavities filled with sealant, grease or other foreign material. The drain, vent and sealant lines shall be

	STANDARD TECHNICA	STANDARD TECHNICAL SPECIFICATION		
REGD. OFF. RANCHI 834002	OIL & GAS SBU, DELHI		Tratier Base comment	
		DOCUMENT NO.	Page 14 of 20	
TITLE	BALL VALVE	MEC/TS/05/21/002	REVISION : 1	
			EDITION : 1	
	either included in the hydrostation shall be held for at least 30 min testing. The body cavity self-reliev of this specification shall also be	c shell test or tested indep nutes. No leakage is perm eving feature meeting the re checked.	pendently. Test pressure issible during hydrostatic equirements of clause 4.8	
5.1.9	A supplementary air seat test as be carried out for all valves. A l sealant. No leakage is allowed.	s per API 6D (Annex B, Cl oubble tight seal is require Fest pressure shall be held	ause B.3.3, Type II) shall d without the use of any for at least 15 minutes.	
5.1.10	Manufacturer who intends biddin successful fire type-tests for valv API 6FA, as applicable in Valve [	g, must submit at bid stage res in accordance with API Data Sheet.	e, certificate and report for -607/ BS EN ISO 10497 /	
	Failure to comply with this re Bidder's offer.	equirement shall be a ca	use of rejection of the	
5.1.11	Valves shall be subjected to Operational Torque Test as per API 6D (Annex I Clause B.6) under hydraulic pressure equal to maximum differential pressur corresponding to the valve pressure class rating.			
	For manual operator of all valve operate the valve does not exce specification.	es, it shall be established t ed the requirements stated	that the force required to d in clause 4.22(c) of this	
5.1.12	Power actuated valves shall be tested after assembly of the valve and actuator at the valve Manufacturer's works. At least five Open-Close-Open cycles without internat pressure and five Open-Close-Open cycles with maximum differential pressure sha be performed on the valve actuator assembly. The time for Full Open to Full close shall be recorded during testing. If required, the actuator shall be adjusted to ensure that the opening and closing times are within the limits stated in Actuator Data Shee issued for the purpose.			
	Hand operator provided on the actuator shall also be checked after above test satisfactory manual over-ride performance.			
	These tests shall be conducted of the same size, rating and the the requirements, retesting / re Inspector.	on minimum one valve out actuator model / type. In c jection of the lot shall be	of a lot of five (5) valves ase the tests do not meet decided by Purchaser's	
5.1.13	Subsequent to successful testin one (1) valve out of the total of Purchaser's Representative for o	ng as specified in clause & ordered quantity shall be r cyclic testing as mentioned	5.1.11 and 5.1.12 above, andomly selected by the below :	
	a) The valve shall be subje maximum differential pres	ected to at least 100 Oper ssure corresponding to the	n-Close-Open cycles with valve rating.	
	b) Subsequent to the above supplementary air seat te	, the valve shall be subject st in accordance with claus	ed to hydrostatic test and e 5.1.8 and 5.1.9.	
	Page 439 of 644			

REGD. OFF:	STANDARD TECHNICAL SPECIFICATION OIL & GAS SBU, DELHI		
RANCHI 834002			मेक्सेन
		DOCUMENT NO.	Page 15 of 20
TITLE	BALL VALVE	MEC/TS/05/21/002	<b>REVISION</b> : 1
			EDITION : 1
	In case this valve fails to pass the valves shall be selected random both valves pass these tests, all va that failed) shall be deemed accep tests, all valves shall be rejected Manufacturer.	ese tests, the valve shall by y and subjected to testin alves manufactured for th otable. If either of the two d or each valve shall be	be rejected and two mo og as indicated above e order (except the va valves fails to pass the tested at the option
	Previously carried out test of sim same has been carried out by Ma below and two sizes above the siz one rating lower of valve tested pre	nilar nature shall be cons anufacturer in last two ye ze of valve previously tes eviously, shall be qualified	sidered acceptable if sears. Valves of two sized, and rating similar
5.1.14	Checks shall be carried out to demonstrate that the dissimilar metal used in the valves are successfully insulated as per the requirement of clause 4.17 of thi specification.		
5.1.15	When indicated in Valve Data Sheet, valves shall be subjected to anti-static testing as per supplementary test requirement of API 6D (Annex B, Clause B.5).		
5.2	Purchaser reserves the right to perform stage-wise inspection and witness tests as indicated in clause 5.1 above at Manufacturer's works prior to shipment Manufacturer shall give reasonable access and facilities required for inspection to the Purchaser's Inspector.		
	Purchaser reserves the right to require additional testing at any time to confirm of further investigate a suspected fault. The cost incurred shall be to Manufacturer' account.		
	In no case shall any action of Purchaser or his Inspector relieve the Manufacturer o his responsibility for material, design, quality or operation of valves.		
	Inspection and tests performed/ w way relieve the Manufacturer's o tests.	vitnessed by the Purchas obligation to perform the	er's Inspector shall in required inspection a
6.0	EXTENT OF INSPECTION & TES	TING	
6.1	Purchaser's Inspector shall perform inspection and witness tests on all valves or a indicated in the Quality Assurance Plan (QAP) attached with this specification.		
	•		•

	STANDARD TECHNICAL SPECIFICATION			
REGD. OFF: RANCHI 834002	OIL & GAS SBU, DELHI		मेवर्डन 10 april comman	
		DOCUMENT NO.	Page 16 of 20	
TITLE	BALL VALVE	MEC/TS/05/21/002	REVISION : 1	
			EDITION : 1	
7.0	TEST CERTIFICATES			
7.1	Manufacturer shall submit the follo	owing certificates:		
	<ul> <li>Mill test certificates relevant to the chemical analysis and mechanical properties of the materials used for valve construction as per the relevant standards.</li> </ul>			
	b) Test certificates of hydrost timing and pressure of eac	tatic and pneumatic tests h test.	complete with records of	
	c) Test reports on radiograph	and ultrasonic inspection		
	<ul> <li>Test report on operation of valves conforming to clause 5.1.11, 5.1.12 a</li> <li>5.1.13 of this specification.</li> </ul>			
	e) All other test reports and certificates as required by API 6D a specification.			
	The certificates shall be valid only when signed by Purchaser's Inspector. Only those valves which have been certified by Purchaser's Inspector shall be despatched from Manufacturer's works.			
8.0	PAINTING, MARKING & SHIPMENT			
8.1	Valve surface shall be thoroughly cleaned, freed from rust and grease and applied with sufficient coats of corrosion resistant paint. Surface preparation shall be carried out by shot blasting to SP-6 in accordance with "Steel Structures Painting Council – Visual Standard SSPC-VIS-1". For valves to be installed underground, when indicated in Valve Data Sheet, the external surfaces of the buried portion of valves shall be painted with three coats of suitable coal tar epoxy resin with a minimum dry film thickness of 300 microns.			
8.2	Manufacturer shall indicate the typ submitted for approval.	be of corrosion resistant pa	aint used, in the drawings	
8.3	All valves shall be marked as per API 6D. The units of marking shall be metric except Nominal Diameter which shall be in inches. Marking shall be done by diestamping on the bonnet or on the housing. However, for buried valves, the marking shall be done on the above ground portion of the stem housing only.			
8.4	Valve ends shall be suitably protected to avoid any damage during transit. All threaded and machined surfaces subject to corrosion shall be well protected by a coat of grease or other suitable material. All valves shall be provided with suitable protectors, for flange faces, securely attached to the valves. Bevel ends shall be protected with metallic or high impact plastic bevel protectors.			
8.5	All sealant lines and other cavities shipment.	es of the valve shall be	filled with sealant before	

MECON LIMITED	STANDA	STANDARD TECHNICAL SPECIFICATION				
RANCHI 834002		OIL & GAS SBU	, DELHI	मेवर्गन 10 aug comm		
			DOCUMENT NO.	Page 17 of 20		
TITLE	BALL V	ALVE	MEC/TS/05/21/002	REVISION : 1		
				EDITION : 1		
8.6	Packaging and sh	ipping instruction	ns shall be as per API 6D.			
8.7	On packages, follo	owing shall be m	arked legibly with suitable	e marking ink :		
	a) Order Nun b) Manufactu c) Valve Size d) Tag Numb e) Serial Nun	nber rer's Name and Rating er nber				
9.0	SPARES & ACCE	<u>ESSORIES</u>				
9.1	Manufacturer shall furnish list of recommended spares and accessories for valves required during start-up and commissioning and supply of such spares shall be included in the price quoted by Manufacturer.					
9.2	Manufacturer shall furnish list of recommended spares and accessories required for two years of normal operation and maintenance of valves and price for such spares shall be quoted separately.					
9.3	Manufacturer shall quote for spares & accessories as per Material Requisition.					
10.0	DOCUMENTATION					
10.1	At the time of bido	ling, Manufactur	er shall submit the followir	ng documents:		
	a) General a positions a and other of valve &	rrangement / as nd sizes of vents external parts to actuator.	sembly drawings showing s, drains, gear operator / a ogether with overall dimer	all features and relative actuator, painting, coating asions as well as weights		
	b) Sectional drawing showing major parts with reference numbers and material specification. In particular, a blow-up drawing of ball-seat assembly shall be furnished complying the requirement of clause 4.6 of this specification.					
	c) Reference years indi size, rating	list of similar b cating all releva g, service, etc.	oall valves manufactured int details including proje	and supplied in last five ct, year, client, location,		
	d) Torque cu maximum calculation	rves for the pow allowable stem s shall also be s	er actuated valves along v torque. In addition, si ubmitted for power actuat	with the break torque and zing criteria and torque ed valves.		
	e) Descriptive	e technical catalo	ogues of the Manufacturer			

f) Copy of valid API 6D certificate.

	STANDARD TECHNICA	STANDARD TECHNICAL SPECIFICATION OIL & GAS SBU, DELHI					
REGD. OFF: RANCHI 834002	OIL & GAS SB						
		DOCUMENT NO.	Page 18 of 20				
TITLE	BALL VALVE	MEC/TS/05/21/002	REVISION : 1				
			EDITION : 1				
	g) Details of support foot, in line to bottom of support	Details of support foot, including dimensions and distance from valve centre line to bottom of support foot.					
	h) Quality Assurance Plan accepted.	enclosed with this tender c	luly signed, stamped and				
	i) List of recommended spa	res required during start-up	and commissioning.				
	j) List of recommended sp maintenance.	ares required for 2 years	of normal operation and				
	k) Other documents / drawin	ngs / data as per Material R	equisition.				
	<ul> <li>Purchaser's final approval :</li> <li>a) Detailed sectional arrang numbers and material spots</li> <li>b) Assembly drawings with indicate the number of required for operating th painting scheme Comm</li> </ul>	<ul> <li>chaser's final approval :</li> <li>Detailed sectional arrangement drawings showing all parts with reference numbers and material specifications as referred to in clause 10.1 above.</li> <li>Assembly drawings with overall dimensions and features. Drawing shall also indicate the number of turns of hand wheel (in case of gear operators) required for operating the valve from full open to full close position and the</li> </ul>					
	applicable) shall be indic above.	applicable) shall be indicated in these drawings as referred to in clause 10.1 above.					
	c) Welding, heat treatment a	and testing procedures.					
	d) Procedure for cyclic testir	ng.					
	e) Details of corrosion resist	ant paint to be applied on t	he valves.				
	f) Design calculation for pre	ssure containing parts.					
	g) Other documents / drawin	ngs / data as per Material R	equisition.				
	Manufacture of valves sh indicated in clause 10.2a Purchaser, any changes be notified to Purchaser obtained before the valve	all commence only after a ) to 10.2c) above. Once ap in design, material and me whose approval in writing is manufactured.	pproval of the documents proval has been given by thod of manufacture shall g of all changes shall be				
10.3	Within 2 weeks from the approvide copies of the approved drawing 10.2 above.	al date, Manufacturer shal s, documents and specific	I submit to Purchaser six ations as listed in clause				

MECON LIMITED	STANDARD TECHNICAL					
RANCHI 834002	OIL & GAS SBU,	OIL & GAS SBU, DELHI				
		DOCUMENT NO.	Page 19 of 20			
TITLE	BALL VALVE	MEC/TS/05/21/002	REVISION : 1			
			EDITION : 1			
10.4	Prior to shipment, Manufacturer shall submit six hard copies and six soft copies (on CD-ROMs) of the following:					
	a) Test certificates as per clau	use 7.0 of this specification	ז.			
	<ul> <li>Manual for installation, erection, maintenance and operation instructions, including a list of recommended spares for the valves.</li> </ul>					
	c) Other documents / drawings / data as per Material Requisition.					
10.5	All documents shall be in English language.					
10.6	The above documents & data requirements shall also be supplemented by all requirements of clause 2.0 of the Material Requisition.					
11.0	GUARANTEE					
11.1	Manufacturer shall guarantee that the materials and machining of valves and fittings comply with the requirements in this specification and in the Purchase Order.					
11.2	Manufacturer is bound to replace or repair all valve parts which should result defective due to inadequate engineering or to the quality of materials and machining.					
11.3	If valve defect or malfunctioning cannot be eliminated, Manufacturer shall replace the valve without delay,					
11.4	Any defect occurring during the period of Guarantee shall be attended to by making all necessary modifications and repair of defective parts free of charge to the Purchaser as per the relevant clause of the bid document.					
11.5	All expenses shall be to Manufactu	urer's account.				



## PROCESS & PIPING DESIGN SECTION MECON LIMITED



# TECHNICAL SPECIFICATION FOR PLUG VALVES (NB $\geq 2^{"}$ )

## SPECIFICATION NO .: MEC/TS/05/62/003, Rev-2

<b>MECON LIMITED</b> Delhi	PROCESS & PIPING DESIGN SECTION	TECH	INICAL SPECIFICATION FOR PLUG VALVES	Atalia Bool Carp
TECHNICAL SPECIFICATION NO. : MEC/TS/05/62/003			REV-2	Page 1 of 13

<u>C O N T E N T S</u>				
<u>SI.No.</u>	Description	Page No.		
1.0	SCOPE	2		
2.0	REFERENCE DOCUMENTS	2		
3.0	MATERIALS & TEST PROCEDURES	3		
4.0	DESIGN & CONSTRUCTION	4		
5.0	INSPECTION & TESTS	7		
6.0	EXTENT OF INSPECTION & TESTING	9		
7.0	TEST CERTIFICATES	9		
8.0	PAINTING, MARKING & SHIPMENT	10		
9.0	SPARES & ACCESSORIES	10		
10.0	DOCUMENTATION	11		
11.0	GUARANTEE	12		

PREPARED BY :	CHECKED BY :	APPROVED BY :
Gurdeep Singh	A.K. Sarkar	A.K. Johri
Date	Date	Date Page 447 of 644
Duled data/Viiwant/Space mactor/Final Spac/SUDDLV ITEMS/02 DULC VALVE	V002 Dlug Voluce TS D2 dog	

D:\old data\Vijyant\Spec master\Final Spec\SUPPLY ITEMS\03 PLUG VALVE\003 - Plug Valves - TS-R2.doc

<b>MECON LIMITED</b> Delhi	PROCESS & PIPING DESIGN SECTION	TECH	NICAL SPECIFICATION FOR PLUG VALVES	
TECHNICAL SPECIFICATION NO. : MEC/TS/05/62/003			REV-2	Page 2 of 13

TECHNICAL SPECIFICATION NO. : MEC/TS/05/62/003

#### 1.0 **SCOPE**

This specification covers the minimum requirements for design, manufacture and supply of carbon steel plug valves of size DN 50mm (2") and above and ANSI Class 150# thru 900# for use in onshore pipeline systems handling non sour hydrocarbons in liquid phase or gaseous phase including Liquefied Petroleum Gas (LPG).

#### 2.0 **REFERENCE DOCUMENTS**

- 2.1 All valves shall be manufactured and supplied in accordance with the Twenty Second Edition, January, 2002, or the latest edition of American Petroleum Institute (API) Specification 6D, twenty first edition, 1994 including supplement 1 & 2 thereof with additions and modifications as indicated in the following sections of this specification.
- 2.2 Reference has also been made in this specification to the latest edition of the following Codes, Standards and Specifications :

ASME B 16.5	:	Pipe flanges and flanged fittings
ASME B 16.25	:	Buttwelding ends
ASME B 16.34	:	Valves – Flanged, threaded and welding end
ASME B16.47	:	Large diameter steel flanges
ASME B 31.3	:	Chemical & process plant piping system
ASME B 31.4	:	Liquid transportation systems for hydrocarbons and other liquids
ASME B 31.8	:	Gas transmission and distribution piping systems
ASME Sec.VIII	:	Boiler and pressure vessel code
ASTM A 370	:	Standard test methods and definitions for mechanical testing of steel products
ASTM B 733	:	Autocatalytic nickel phosphorous coating on metals
API 6FA	:	Fire test for valves
API 1104	:	Welding of pipelines and related facilities
BS:6755 (Part-II)	:	Testing of valves – Specification for fire type - testing requirements
MSS-SP-6	:	Standard finishes for contact faces of pipe flanges and connecting-end flanges of valves and fittings

D:\old data\Vijyant\Spec master\Final Spec\SUPPLY ITEMS\03 PLUG VALVE\003 - Plug Valves - TS-R2.doc

MECON LIMIT Delhi	ED	PROCESS & PIPING DESIGN SECTION	TECHN	ICAL SPECIFICATION FOR PLUG VALVES	Japir Jagir Jagir	
TECHNICAL SPE	CIFICATION NO. :	MEC/TS/05/62/003		REV-2	Page 3 of 13	
	MSS-SP-44 SSPC-VIS-1	: Steel p : Steel s	ipeline fla tructures	nges painting council-visu	al standard	
2.3	<ul> <li>In case of conflict between the requirements of this specification, API 6D and the Codes, Standards and Specifications referred in clause 2.2 above, the requirements of this specification shall govern. Order of precedence shall be as follows :</li> <li>Data Sheets</li> <li>This Specification</li> <li>API 6D Specification</li> <li>Other Referred Codes &amp; Standards</li> </ul>					
2.0						
3.0		A IESI PROCEDURES	<u>2</u>			
3.1	Material for major components of the valves shall be as indicated in Valve Data Sheet. Other components shall be as per Manufacturer's standard which will be subject to approval by Purchaser.					
3.2	Carbon steel u	used for the manufacture	e of valves	shall be fully killed.		
3.3	Chemical com further weldin used: a) Carbon b) Manga c) Silicon d) Phosp e) Sulphu Total percents elements shal a) Nitrog b) Nickel	nposition (check analysi g by Purchaser shall me n : anese : torus : ur : 0.030 % age of Vanadium, Niobi I not exceed the followin en :	s) of valv et the follo 0.22% (m 1.70 % (r 0.55 % (r 0.030 % ( 6 (max.) um and 1 g limits : 0.019 % 0.30 %	re end connection v owing requirements f nax.) nax.) (max.) (max.)	which are subject to for each heat of steel	
	c) Coppe d) Alumir	r : num :	0.20 % 0.070 %			
	e) Chrom f) Molybe	nium : denum :	0.15 % 0.05 %			
	Carbon equiva	ilent (CE) as calculated b	y the follo	wing shall not excee	d 0.45%.	
		Mn Cr + M	o + V	Ni + Cu		
	CE = C + -	6 5	+ ;	15		
				Pa	ae 449 of 644	

D:\old data\Vijyant\Spec master\Final Spec\SUPPLY ITEMS\03 PLUG VALVE\003 - Plug Valves - TS-R2.doc

MECON LIMITED Delhi	PROCESS & PIPING DESIGN SECTION	TECH	INICAL SPECIFICATION FOR PLUG VALVES	
TECHNICAL SPECIFICATION NO. : MEC/TS/05/62/003			REV-2	Page 4 of 13

3.4 For valves specified for Gas Service or high vapour pressure liquid service, charpy V-Notch test on each heat of base material shall be conducted as per API 6D, for all pressure containing parts such as body, end flanges and welding ends as well as the bolting material for pressure containing parts. Unless specified otherwise in Valve Data Sheets, the Charpy impact test shall be conducted at 0°C. The Charpy impact test specimen shall be taken in the direction of principal grain flow and notched perpendicular to the original surface of plate or forging.

Unless specified otherwise in Valve Data Sheets, the minimum average absorbed energy per set of three specimens shall be 27 J with an individual minimum per specimen of 22 J.

- 3.5 For valves specified for Gas Service or high vapour pressure liquid service, the hardness of base material of body and principal parts of the valve such as plug, stem, etc., shall not exceed 22 RC.
- 3.6 Plug for valve size DN 200mm (8") and above or as specified in Valve Data Sheets shall have Electroless Nickel Plating (ENP) or equivalent. The hardness of plating shall be minimum 50 RC. Manufacturer shall ensure that the adhesive strength of plating is sufficient so as to prevent peeling of plating during operation of the valve.
- 3.7 All process-wetted parts, metallic and non-metallic, shall be suitable for the fluids and service specified by the Purchaser. The service gas composition when applicable shall be as given in Annexure-I.

## 4.0 DESIGN & CONSTRUCTION

- 4.1 The Manufacturer shall have a valid license to use API 6D monogram for manufacture of Plug Valves.
- 4.2 Valve pattern shall be short, regular or venturi as specified in the following table:

Class	Size Range, NB mm (inch	) Pattern
	50-100 (2-4)	Short
150	150-300 (6-12)	Regular
	350 (14) & above	Venturi
	50-100 (2-4)	Short
300	150-250 (6-10)	Regular
	300 (12) & above	Venturi
	50-250 (2-10) F	Regular
600	300 (12) & above	Venturi
	50-250 (2-10) F	Regular
900	300 (12) & above	Venturi

<b>MECON LIMITED</b> Delhi	PROCESS & PIPING DESIGN SECTION	TECH	NICAL SPECIFICATION FOR PLUG VALVES	
TECHNICAL SPECIFICATION NO. : MEC/TS/05/62/003			REV-2	Page 5 of 13

4.3 Valve shall have an inherent feature using line pressure to ensure that the line pressure cannot cause taper locking of the plug/ plug movement into taper i.e. valves shall be of pressure balanced design. Cover shall be bolted to the body and screwed connections are not acceptable. 4.4 4.5 Soft seats to achieve a seal between plug and body are not permitted. 4.6 All valves shall have provisions for secondary sealant injection under full line pressure for seat and stem seals. Sealant injection points shall be provided with a ball type check valve or needle valve to replace the sealant injection fitting under full line pressure. 4.7 Valves shall have vent and drain connections as per API 6D. 4.8 When specified in the Valve Data Sheet, valves shall be designed to withstand a sustained internal vacuum of at least one milli-bar in both open and closed position. 4.9 Valve design shall ensure repair of gland packing under full line pressure. 4.10 Valve ends shall be either flanged or butt welded or one end flanged and one end butt a) welded as indicated in Valve Data Sheet. Flanges of the flanged end cast/ forged body valves shall be integrally cast/forged with the body of valve. Face-to-face/ end-to-end dimensions shall conform to API 6D. b) Flanged end shall have dimensions as per ASME B16.5 for valve sizes upto DN 600mm (24 inches) excluding DN 550mm (22 inches) and as per MSS-SP-44 for valve sizes DN 550mm (22 inches) & for DN 650mm (26 inches) and above. Flange face shall be either raised face or ring joint type as indicated in Valve Data Sheet. Flange face finish shall be serrated or smooth as indicated in Valve Data Sheet. Smooth finish when specified shall be 125 to 200 AARH. In case of RTJ flanges, the groove hardness shall be minimum 140 BHN. Butt weld end preparation shall be as per ASME B16.25. The thickness of the pipe to c) which the valve has to be welded shall be as indicated in Valve Data Sheet. Valves shall be without transition pups. In case significant difference exists between thickness of welding ends of valve and connecting pipe, the welding ends of valve shall have bevel preparation as per ASME B31.4 or ASME B31.8, as applicable. 4.11 Valves shall be provided with position indicator and stops at the fully open and fully closed positions. 4.12 Valves of size DN 200mm (8") and above shall be equipped with lifting lugs. Tapped holes and eye bolts shall not be used for lifting lugs. 4.13 Valves shall have locking devices to be locked either in full open or full close position when indicated in the Valve Data Sheets. Locking devices shall be permanently attached to the valve operator and shall not interfere with operation of the valve.

<b>MECON LIMITED</b> Delhi	PROCESS & PIPING DESIGN SECTION	TECH	NICAL SPECIFICATION FOR PLUG VALVES	2001 Conf
TECHNICAL SPECIFICATION NO. : MEC/TS/05/62/003			REV-2	Page 6 of 13

TECHNICAL SPECIFICATION NO. : MEC/TS/05/62/003

- 4.14 Valves shall be of fire safe design as per BS:6755 (Part-II)/ API 6FA, if indicated in Valve Data Sheet.
- 4.15 Valves shall be suitable for either buried or above ground installation as indicated in the Valve Data Sheet.
- 4.16 Valves with stem extension, when indicated in Valve Data Sheet shall have following provisions :
  - a) Valves provided with stem extension shall have water proof outer casing. Length of stem extension shall be as indicated in the Valve Data Sheet. The length indicated corresponds to the distance between the centreline of the valve opening and the top of the mounting flange for valve operating device (gear operator/ power actuator as applicable).
  - b) Vent and drain connections shall be terminated adjacent to the valve operator by means of suitable piping anchored to the valve body. Pipe used shall be API 5L Gr. B/ ASTM A106 Gr. B, with Sch. 160. Fittings shall be ASTM A105/ ASTM A 234 Gr. WPB, Socket Welded, ANSI class 6000.
  - Sealant injection lines shall be extended and terminated adjacent to the valve c) operator in manner as indicated in (b) above.
  - Stem extension and stem housing design shall be such that the complete d) assembly will form a rigid unit giving a positive drive under all conditions with no possibility of free movements between valve body stem extension or its operator.
  - Outer casing of stem extension shall have 3/8" or 1/2" NPT plugs at the top and e) bottom, for draining and filling with oil to prevent internal corrosion.

#### 4.17 **Operating Devices**

- a) Valves shall have a power actuator or manual operator as indicated in the Valve Data Sheet. Manual operated valves of size < DN 100mm (4") shall be wrench operated and valves of sizes > DN 150mm (6") shall be gear operated. Each wrench operated valve shall be supplied with wrench. Valve design shall be such that damage due to malfunctioning of the operator or its controls will only occur in the operator gear train or power cylinder and damaged parts can be replaced without the bonnet being removed.
- b) The power actuator shall be in accordance with the specification issued for the purpose and as indicated in the valve and actuator data sheet. Operating time shall be as indicated in valve data sheet. Valve operating time shall correspond to full close to full open / full open to full close under maximum differential pressure corresponding to the valve rating. For actuated valves, the actuator torque shall be atleast 1.25 times the maximum torgue required to operate the valve under maximum differential pressure corresponding to the valve class rating.
- Operating device shall be designed for easy operation of valve under maximum c) differential pressure corresponding to the valve rating.

MECON LIMITED Delhi		PROCESS & PIPING DESIGN SECTION	TECH	INICAL SPECIFICATION FOR PLUG VALVES		
TECHNICAL SP	ECIFICATION NO. :	MEC/TS/05/62/003		REV-2	Page 7 of 13	
<ul> <li>d) For manual operation of all valves, the diameter of the hand wheel or the length of operating lever shall be such that under the maximum differential pressure, the total force required to operate the valve does not exceed 350 N. Manufacture shall also indicate the number of turns of hand wheel (in case of gear operator) required to operate the valve from full open to full close position.</li> <li>e) Direction of operation of hand wheel or wrench shall be in clock-wise direction while closing the valve. Hand wheels shall not have protruding spokes.</li> <li>f) Gear operators, if specified, shall have a self locking provision and shall be full encased in waterproof/ dustproof/ weatherproof/ splashproof enclosure and shall be filled with suitable grease.</li> </ul>					I wheel or the length erential pressure, the 350 N. Manufacturer se of gear operator), ion. clock-wise direction g spokes. on and shall be fully f enclosure and shall	
4.18	Repair by welding is not permitted for fabricated and forged body valves. However repair by welding as per ASME B16.34 is permitted for cast body valves. Repair shall be carried out before any heat treatment of casting is done. Repair welding procedure qualification shall also include impact test and hardness test when required as per Clause 3.4 and 3.6 of this specification and shall meet the requirements as specified therein.					
4.19	The tolerance valves shall be	The tolerance on internal diameter and out of roundness at the ends for welded ends valves shall be as per connected pipe specification as indicated in the Valve Data Sheet.				
4.20	Valve stem sh operate the v class rating. specified in AS For Power Act of the selected	Valve stem shall be capable of withstanding the maximum operating torque required to operate the valve against the maximum differential pressure corresponding to applicable class rating. The combined stress shall not exceed the maximum allowable stresses specified in ASME section VIII, Division-1. For Power Actuated Valves, the valve stem shall be designed for maximum output torque of the selected power actuator (including gear box, if any) at the valves stem.				
5.0	INSPECTION	<u>N &amp; TESTS</u>				
5.1	The Manufacturer shall perform all inspection and tests as per the requirements of this specification and the relevant codes, prior to shipment at his works. Such inspection and tests shall be, but not limited to, the following :					
5.1.1	All valves shal	l be visually inspected.				
5.1.2	Dimensional c	heck shall be carried out a	as per t	he Purchaser approved	d drawings.	
5.1.3	Chemical con material stand	nposition and mechanica ards and this specification	al prop i, for ea	erties shall be check ch heat of steel used.	ed as per relevant	
5.1.4	a) Non-d consis be car	estructive examination ting of but not limited to o ried out by the Manufactu	of ind castings irer.	lividual valve materi s, forgings, plates and	al and component assembly welds shall	
				Pa	ge 453 of 644	

MECON LIMITED Delhi		PROCESS & P DESIGN SEC	PIPING TION	TECH	NICAL SPECIFIC FOR PLUG VALVES	ATION	्रिकॉन 19 2001 Carto
TECHNICAL SPE	ECIFICATION NO	. : MEC/TS/05/62/003			REV-2		Page 8 of 13
	b) Valv sea Proo radi	ves castings shall b t location, flanged cedure and accept ography shall be a	be radiogra body end tance crite s follows :	aphically Is and eria sha	y examined a circumference Il be as per a	t the cov e of ends ASME B1	ver and body portion, s to be field welded. 16.34. The extent of
	ANS	51 Class 150- 51 Class 300-	All Sizes ≤ DN 400 ≥	)mm (1 : DN 45	- 6") - 0mm (18")	Nil Nil -	100%
	ANS	I Class 600- above	All Sizes		-	100%	
	All surf	castings shall be aces. Method and	wet mag acceptanc	inetic p e shall	article inspection inspection in the comply with A	cted 100 ASME B10	) % of the internal 6.34.
	c) Valve forgings shall be examined by ultrasonic method. Inspection procedure and acceptance criteria shall be as per Annexure E of ASME B16.34.						
5.1.5	Areas which, in Purchaser's Inspector's opinion, cannot be inspected by radiographic methods shall be checked by ultrasonic or magnetic particle methods and acceptance criteria shall be as per ASME Sec-VIII, Division I, Appendix 12 and Appendix 6 respectively.						
5.1.6	a) Wel acco b) Afte wet and less perr c) All ultra end	Weld ends of all cast valves shall be 100% radiographically examined and acceptance criteria shall be as per ASME B16.34. After final machining all bevel surfaces shall be inspected by dye penetrant, or wet magnetic particle methods. Any defects longer than 6.35mm shall be rejected and also defects between 6.35mm and 1.59mm that are separated by a distance less than 50 times their greatest length. Weld repair of bevel surface is not permitted. Rejectable defects must be removed. All finished wrought weld ends subject to welding in the field shall be 100% ultrasonically tested for lamination type defects for a distance of 50mm from the end. Laminations shall not be acceptable.					
5.1.7	All valves shall be tested in compliance with the requirements of API 6D. Hydrostatic shell testing shall ensure that the whole of the shell is subjected to the test pressure. If necessary, the empty shell shall be pressure tested prior to assembly of the plug. The drain, vent and sealant lines shall be either included in the hydrostatic shell test or tested independently. No leakage is permissible during hydrostatic testing.						
5.1.8	A suppleme is allowed.	entary air seat test Test pressure shall	as per AP be held f	I 6D sh or at lea	all be carried ast 15 minutes	out for a s.	all valves. No leakage

<b>MECON LIMITED</b> Delhi	PROCESS & PIPING DESIGN SECTION	TECH	INICAL SPECIFICATION FOR PLUG VALVES	
TECHNICAL SPECIFICATION NO. : MEC/TS/05/62/003		REV-2	Page 9 of 13	

5.1.9 Manufacturer who intends bidding must submit at bid stage, certificate and report for successful fire safe tests for all types of valves in accordance with BS:6755 (Part-II)/ API 6FA, as applicable in Valve Data Sheet.

Failure to comply with the requirement shall be a cause of rejection of the offer.

- 5.1.10 Valve shall be subjected to Operational Torque Test as per supplementary test requirement of API 6D under hydraulic pressure equal to the maximum differential pressure corresponding to the valve rating. The maximum handwheel force shall not exceed 350 N.
- 5.1.11 Power actuated valves shall be tested after assembly at the valve Manufacturer's works. Actuator shall be capable to allow minimum five consecutive "opening" and "closing" cycles. To achieve this, the Manufacturer shall provide "closing" and "opening" operations. This test shall be conducted on one valve out of a lot of five valves of the same size, rating and actuator type. In case the test result dose not meet the requirements, retesting/ rejection of the lot shall be as decided by Purchaser's Inspector.

The actuator shall be adjusted to ensure that opening and closing time is within the limits stated in Actuator Data Sheet issued for the purpose.

The hand operator installed on the actuator shall also be checked after the cyclic testing, for satisfactory manual over-ride performance.

5.2 Purchaser reserves the right to perform stagewise inspection and witness tests as indicated in para 5.1 at Manufacturer's works prior to shipment. Manufacturer shall give reasonable access and facilities required for inspection to Purchaser's Inspector.

Purchaser reserves the right to request additional testing at any time to confirm or further investigate a suspected fault. If the suspected fault is confirmed, the cost incurred shall be to Manufacturer's account.

In no case shall any action of Purchaser or his representative relieve the Manufacturer of his responsibility for material, design, quality or operation of valves.

Inspection and tests performed/ witnessed by the Purchaser's Inspector shall in no way relieve the Manufacturer's obligation to perform the required inspection and tests.

## 6.0 EXTENT OF INSPECTION & TESTING

- 6.1 Purchaser's Inspector shall perform inspection and witness test on all valves as indicated in the Quality Assurance Plan (QAP) attached with this specification.
- 6.2 The hydrostatic testing and cyclic opening and closing of the valves with the operator shall be witnessed by Purchaser's Inspector.

## 7.0 **TEST CERTIFICATES**

- 7.1 Manufacturer shall submit the following certificates :
  - a) Mill test certificates relevant to the chemical analysis and mechanical properties of the materials used for valve construction as per the relevant standards.
  - b) Test certificates on hydrostatic and pneumatic test complete with records of timing and pressure of each test.
  - c) Test reports conforming to clause 5.1.9 of this specification, if applicable.
  - d) Test reports on radiographic and ultrasonic inspection.
  - e) Test reports on operation of valves conforming to clause 5.1.10 and 5.1.11 of this specification.
  - f) All other test reports and certificates as required by API 6D and this specification.

The certificates shall be valid only when signed by Purchaser's Inspector. Only those valves which have been certified by Purchaser's Inspector shall be dispatched from Manufacturer's works.

## 8.0 PAINTING, MARKING & SHIPMENT

- 8.1 Valve surface shall be thoroughly cleaned, freed from rust and grease and applied with sufficient coats of corrosion resistant paint. Surface preparation shall be carried out by shot blasting to SP 6 in accordance with "Steel Structures Painting Council Visual Standard SSPC-VIS-1". For the valves to be installed underground, when indicated in Valve Data Sheet, external surfaces of the buried portion of valves shall be painted with three coats of suitable coal tar epoxy resin with a minimum dry film thickness of 300 microns.
- 8.2 Manufacturer shall indicate the type of corrosion resistant paint used, in the drawings submitted for approval.
- 8.3 All valves shall be marked as per API 6D. The units of marking shall be metric except Nominal Diameter which shall be in inches. Marking shall be done by die-stamping on the bonnet or on the housing. However for buried valves the marking shall be done on the above ground portion of the stem housing only.
- 8.4 Valve ends shall be suitably protected to avoid any damage during transit. All threaded and machined surfaces subject to corrosion shall be well protected by a coat of grease or other suitable material. All valves shall be provided with suitable protectors, for flange faces, securely attached to the valves. Bevel ends shall be protected with metallic bevel protectors.

<b>MECON LIMITED</b> Delhi	PROCESS & PIPING DESIGN SECTION	TECH	INICAL SPECIFICATION FOR PLUG VALVES	Asir Bent Core
TECHNICAL SPECIFICATION NO. : MEC/TS/05/62/003			REV-2	Page 11 of 13

- 8.5 All sealant lines and other cavities of the valves shall be filled with sealant before shipment.
- 8.6 Packaging and shipping instructions shall be as per API 6D.
- 8.7 Packages shall be marked legibly, with suitable marking ink, the following.
  - a) Order Number
  - b) Manufacturer's Name
  - c) Valve Size and Rating
  - d) Tag Number
  - e) Serial Number

### 9.0 SPARES & ACCESSORIES

- 9.1 Manufacturer shall recommend and quote separately the spares for valves required for commissioning and two years of normal operation. List of such spares without price shall be indicated alongwith technical bid and separately with price.
- 9.2 Manufacturer shall recommend and quote unit price separately for the accessories (like wrench, sealant injector, etc.), sealant and special tools required for maintenance of valves.

## 10.0 **DOCUMENTATION**

- 10.1 At the time of bidding, the bidder shall submit the following documents :
  - General arrangement/ assembly drawings showing all features and relative positions & sizes of vents, drains, gear box & other external parts together with overall dimensions.
  - b) Sectional drawing showing major parts with reference numbers and material specification.
  - c) Reference list of similar plug valves manufactured and supplied in last five years, indicating all relevant details including project, year, client, location, size rating, service, etc.
  - d) Torque curves for the power actuated valves alongwith break torque and maximum allowable stem torque. In addition, sizing criteria and torque calculations shall also be submitted for power actuated valves.
  - e) Descriptive technical catalogues of the Manufacturer.
  - f) Copy of valid API 6D certificate, wherever applicable.

MECON LIMITED Delhi	PROCESS & PIPING DESIGN SECTION	TECH	NICAL SPECIFICATION FOR PLUG VALVES	
TECHNICAL SPECIFICATION NO. : MEC/TS/05/62/003		REV-2	Page 12 of 13	

- g) Details of support foot, including dimensions and distance from valve centre line to bottom of support foot.
- h) Quality Assurance Plan enclosed with this tender duly signed, stamped and accepted.

### **IMPORTANT**

The drawings to be submitted alongwith the bid shall be in total compliance with the requirement of technical specification and data sheets of the valves with no exception & deviation.

- 10.2 Within two weeks of placement of order, the manufacturer shall submit six copies of, but not limited to, the following drawings, documents and specifications for approval :
  - a) Design drawings and relevant calculations for pressure containing parts and other principle parts.
  - b) Detailed sectional arrangement drawing showing all parts with reference numbers and materials specification.
  - c) Assembly drawings with overall dimensions & clearances required and showing all features. Drawing shall also indicate the numbers of turns of handwheel (in case of gear operator) required for operating the valve from full open to full close position and the painting scheme.
  - d) Welding, heat treatment, testing and quality control procedures.
  - e) Details of corrosion resistant paint to be applied on the valves.
  - f) Design calculation for pressure containing parts.

Manufacture of valves shall commence only after approval of the above documents. Once approval has been given by Purchaser, any change in design, material and method of manufacture shall be notified to the Purchaser, whose approval in writing for all changes shall be obtained before the valves are manufactured.

- 10.3 Within 30 days from the approval date, Manufacturer shall submit one reproducible and six copies of the approved drawings, documents and specification as listed in clause 10.2 of this specification.
- 10.4 Prior to shipment, Manufacturer shall submit one reproducible and six copies of following
  - a) Test certificates as listed in clause 7.0 of this specification.
  - b) Manual for installation, erection instructions, maintenance and operation instructions, including a list of recommended spares for the valves.
- 10.5 All documents shall be in English Language.

۰.\_

MECON LIMITED Delhi	PROCESS & PIPING DESIGN SECTION	TECH	INICAL SPECIFICATION FOR PLUG VALVES	ABIT CONF
TECHNICAL SPECIFICATION NO. : MEC/TS/05/62/003			REV-2	Page 13 of 13

## 11.0 **GUARANTEE**

- 11.1 Manufacturer shall guarantee that the materials and machining of valves and fittings comply with the requirements in this specification and in the Purchase Order.
- 11.2 Manufacturer is bound to replace or repair all valve parts which should result defective due to inadequate engineering or to the quality of materials and machining.
- 11.3 If valve defect or malfunctioning cannot be eliminated, Manufacturer shall replace the valve without delay.
- 11.4 Any defect occurring during the period of Guarantee shall be attended to by making all necessary modifications and repair of defective parts free of charge to the Purchaser as per the relevant clause of the bid document.
- 11.5 All expenses shall be to Manufacturer's account.

## PROCESS & PIPING DESIGN SECTION MECON LIMITED DELHI – 110 092



# TECHNICAL SPECIFICATION FOR CHECK VALVES

# SPECIFICATION NO. : MEC/TS/05/62/004, Rev-2

MECON LIMITED Delhi	PROCESS & PIPING DESIGN SECTION	TECH	INICAL SPECIFICATION FOR CHECK VALVES	Agir Bon Conver
TECHNICAL SPECIFICATION NO. : MEC/TS/05/62/004		REV-2	Page 1 of 8	

<u>CONTENTS</u>					
SI.No.	Descriptior	1	Page No.		
1.0	SCOPE		2		
2.0	REFERENCE	DOCUMENTS	2		
3.0	MATERIALS		2		
4.0	DESIGN AND	O CONSTRUCTION	3		
5.0	INSPECTION	I AND TESTS	4		
6.0	TEST CERTI	FICATES	5		
7.0	PAINTING, M	MARKING AND SHIPMENT	6		
8.0	SPARES AND	D ACCESSORIES	6		
9.0	DOCUMENT	ATION	6		
10.0	GUARANTEE		7		
			Γ		
PREPARED BY		CHECKED BY	APPROVED BY		

<b>MECON LIMITED</b> Delhi	PROCESS & PIPING DESIGN SECTION	TECH	NICAL SPECIFICATION FOR CHECK VALVES	
TECHNICAL SPECIFICATION NO. : MEC/TS/05/62/004		REV-2	Page 2 of 8	

## 1.0 <u>SCOPE</u>

This specification covers the minimum requirements for design, manufacture and supply of carbon steel check valves of size DN 50mm (2") and above and ANSI class 150, 300 and 600, for use in onshore pipeline systems handling non-sour hydrocarbons in liquid phase or gaseous phase including Liquefied Petroleum Gas (LPG).

## 2.0 **REFERENCE DOCUMENTS**

2.1 All valves shall be manufactured and supplied in accordance with the latest edition of American Petroleum Institute (API) Specification 6D or 594 or British Standard BS:1868, with additions and modifications as indicated in the following sections of this specification.

For Contractual purpose, the edition in force at the time of floating of the enquiry shall be termed as "latest edition".

## 3.0 **MATERIALS**

- 3.1 Material for major components of the valves shall be as indicated in Valve Data Sheet. Other components shall be as per Manufacturer's standards which will be subject to approval by Purchaser.
- 3.2 Carbon steel used for the manufacture of valves shall be fully killed.
- 3.3 The Carbon Equivalent (CE) of valve end connections which are subject to further field welding by Purchaser, shall not exceed 0.45% (as calculated by the following formula) on check analysis for each heat of steel used :

 $CE = C + \frac{Mn}{6} + \frac{Cr + Mo + V}{5} + \frac{Ni + Cu}{15}$ 

3.4 Charpy V-Notch test on each heat of base material shall be conducted as per API 6D, clause 7.5, for all pressure containing parts such as body, end flanges and welding ends as well as bolting material for pressure containing parts. Unless specified otherwise, the Charpy impact test shall be conducted at 0°C. The Charpy impact test specimen shall be taken in the direction of principal grain flow and notched perpendicular to the original surface of plate or forging.

The minimum average absorbed energy per set of three specimens shall be 27 J with an individual minimum per specimen of 22 J. No specimen shall exhibit less than 80 percent shear area.

<b>MECON LIMITED</b> Delhi	PROCESS & PIPING DESIGN SECTION	TECH	NICAL SPECIFICATION FOR CHECK VALVES	
TECHNICAL SPECIFICATION NO. : MEC/TS/05/62/004		REV-2	Page 3 of 8	

3.5 All process – wetted parts, metallic and non-metallic, shall be suitable for the fluids and service specified by the Purchaser.

## 4.0 **DESIGN AND CONSTRUCTION**

- 4.1 Following types of check valves, meeting the requirements of applicable standards (refer clause 2.1 of this specification) are acceptable :
  - a) Swing check valve
  - b) Dual plate check valve
  - c) Axial flow (Nozzle) check valve

Valve design shall be suitable for the service conditions indicated in Valve Data Sheet. Corrosion allowance indicated in Valve Data Sheet shall be considered in valve design.

- 4.2 In case of swing check valves, the disc hinge shall be mounted on the valve body and shall not be attached to the valve body cover. Valve body cover joint shall be of bolted design. Screwed covers shall not be used.
- 4.3 Valves shall be provided with non-renewable integral type seats as indicated in Valve Data Sheet. Non-renewable seats shall be of a design which does not required renewal over the design life of the valve.
- 4.4 Valves shall be provided with drain connection as per the Manufacturer's standard. Drain tapping shall be provided in a position suitable to completely drain the valve with valve in horizontal position.
- 4.5 Valve ends shall be either flanged or butt welded or one end flanged and one end butt welded as indicated in Valve Data Sheet. Flanged end shall have dimensions as per ASME B16.5 for sizes upto DN 400mm (16"). Flanges of the flanged end cast body valves shall be integrally cast with the body of the valve.
- 4.6 Butt weld end preparation shall be as per ANSI B16.25. The thickness of the pipe to which the valve has to be welded shall be as indicated in Valve Data Sheet. Valves shall be without transition pups. In case difference exists between thickness of valve neck end and connecting pipe, the bevel end of valve shall be prepared as per ANSI B31.8 or ANSI B31.4, as applicable.
- 4.7 Valves of size DN 200mm (8") and above shall be equipped with lifting lugs. Tapped holes and eye bolts shall not be used for lifting lugs.

MECON LIMITED Delhi	PROCESS & PIPING DESIGN SECTION	TECH	INICAL SPECIFICATION FOR CHECK VALVES	2201 CUEND
TECHNICAL SPECIFICATION NO. : MEC/TS/05/62/004		REV-2	Page 4 of 8	

- 4.8 An arrow indicating the direction of flow shall be embossed or cast on the body of all valves.
- 4.9 All welds shall be made by welders and welding procedures qualified in accordance with the provisions of ASME Section IX. The welding and repair welding procedure qualification shall include impact test and shall meet the requirements of clause 3.4 of this specification.
- 4.10 Repair by welding is permitted for cast body valves subject to written approval by Purchaser and shall be carried out as per ANSI B16.34. Repair shall be carried out before any heat treatment of casting is done.

## 5.0 **INSPECTION AND TESTS**

- 5.1 The Manufacturer shall perform all inspection and tests as per the requirements of this specification and the relevant codes, prior to shipment at his works. Such inspection and tests shall be, but not limited to, the following :
- 5.1.1 All valves shall be visually inspected.
- 5.1.2 Dimensional check on all valves shall be carried out as per the Purchaser approved drawings.
- 5.1.3 Chemical compositions and mechanical properties shall be checked as per relevant material standards and this specification, for each heat of steel used.
- 5.1.4 a) Where applicable, the body castings of valves shall be radiographically examined on 100% of the surface of critical areas as per ANSI B16.34. Procedure and acceptance criteria shall be as per ANSI B16.34.
  - b) Where applicable, valve body made by forging and plate components shall be ultrasonically examined in accordance with procedure and acceptance standard of Annexure E of ANSI B16.34.
  - c) The extent of radiography/ ultrasonic examination shall be as follows :

ANSI	class 150	-	All sizes	-	Nil
ANSI	class 300	-	≤ DN 400mm (16") ≥ DN 450mm (18")	-	Nil 100%
ANSI	class 600	-	All sizes	-	100%

<b>MECON LIMITED</b> Delhi	PROCESS & PIPING DESIGN SECTION	TECHNICAL SPECIFICATION FOR CHECK VALVES		
TECHNICAL SPECIFICATION NO. : MEC/TS/05/62/004			REV-2	Page 5 of 8

5.1.5 All valves shall be tested in compliance with the requirements of applicable standard (refer clause 2.0).

5.2 Purchaser reserves the right to perform stage-wise inspection and witness tests as indicated in clause 5.1 above at Manufacturer's works prior to shipment. Manufacturer shall give reasonable access and facilities required for inspection to the Purchaser's Inspector.

> Purchaser reserves the right to require additional testing at any time to confirm or further investigate a suspected fault. The cost incurred shall be to Manufacturer's account.

> In no case shall any action of Purchaser or its Inspector relieve the Manufacturer of his responsibility for material, design, quality or operation of valves.

Inspection and tests performed/ witnessed by the Purchaser's Inspector shall in no way relieve the Manufacturer's obligation to perform the required inspection and tests.

## 6.0 **TEST CERTIFICATES**

Manufacturer shall submit the following certificates :

- a) Mill test certificates relevant to the chemical analysis and mechanical properties of the materials used for the valve construction as per the relevant standards.
- b) Hydrostatic test certificates complete with records of timing and pressure of each test.
- c) Test reports of radiograph and ultrasonic inspection, as applicable.
- d) All other test reports and certificates as required by applicable standard and this specification.

The certificates shall be valid only when signed by Purchaser's Inspector. Only those valves which have been certified by Purchaser's Inspector shall be despatched from Manufacturer's works.

TECHNICAL SPECIFICATION NO. : MEC/TS/05/62/004

REV-2

Page 6 of 8

## 7.0 PAINTING, MARKING AND SHIPMENT

- 7.1 Valve surface shall be thoroughly cleaned, freed from rust and grease and applied with sufficient coats of corrosion resistant paint. Surface preparation shall be carried out by shot blasting to SP-6 in accordance with "Steel Structures Painting Council Visual Standard SSPC-VIS-1".
- 7.2 All valves shall be marked as per applicable standard. The units of marking shall be metric except nominal diameter which shall be in inches.
- 7.3 Valve ends shall be suitably protected to avoid any damage during transit. All threaded and machined surfaces subject to corrosion shall be well protected by a coat of grease or other suitable material. All valves shall be provided with suitable protectors for flange faces, securely attached to the valves.
- 7.4 Packaging and shipping instructions shall be as per applicable standard.
- 7.5 On packages, the following shall be marked legibly with suitable marking ink:
  - a) Order Number
  - b) Manufacturer's Name
  - c) Valve Size and Rating
  - d) Tag Number

## 8.0 SPARES AND ACCESSORIES

8.1 Manufacturer shall recommend and quote separately the spares for valves required for commissioning and two years of normal operation.

## 9.0 **DOCUMENTATION**

- 9.1 At the time of bidding, Manufacturer shall submit the following documents :
  - a) General arrangement drawings showing all features together with overall dimensions and actual valve bore size.
  - b) Sectional drawing showing major parts with reference numbers and material specification.

<b>MECON LIMITED</b> Delhi	PROCESS & PIPING DESIGN SECTION	TECHNICAL SPECIFICATION FOR CHECK VALVES		asin bi
TECHNICAL SPECIFICATION NO. : MEC/TS/05/62/004			REV-2	Page 7 of 8

- c) Details of corrosion resistant paint proposed to be applied.
- d) Reference list of similar supplies of check valves, including project, year, client, location, size, rating, services, etc. shall be furnished by the Manufacturer for the last three years. (The valves shall be proven for service indicated in Valve Data Sheet).
- 9.2 Within three weeks of placement of order, the Manufacturer shall submit four copies of, but not limited to, the following drawings, documents and specifications for Purchaser's approval.
  - a) Detailed sectional drawings showing all parts with reference numbers and material specification.
  - b) Assembly drawings indicating overall dimensions, features and painting scheme.

Once the approval has been given by Purchaser, any changes in design, material and method of manufacture shall be notified to Purchaser whose approval in writing of all changes shall be obtained before the valve is manufactured.

- 9.3 Within 30 days from the approval date, Manufacturer shall submit to Purchaser one reproducible and six copies of all approved drawings, documents and specifications as listed in clause 9.2 above.
- 9.4 Prior to shipment, Manufacturer shall submit to Purchaser one reproducible and six copies of the following :
  - a) Test certificates as listed in clause 6.0 of this specification.
  - b) Manual for installation, erection, maintenance and operation instructions, including a list of recommended spares for the valves.
- 9.5 All documents shall be in English language.

## 10.0 **GUARANTEE**

10.1 Manufacturer shall guarantee that the materials and machining of valves and fittings comply with the requirements in this specification and in the Purchase Order.

<b>MECON LIMITED</b> Delhi	PROCESS & PIPING DESIGN SECTION	TECHNICAL SPECIFICATION FOR CHECK VALVES		
TECHNICAL SPECIFICATION NO. : MEC/TS/05/62/004			REV-2	Page 8 of 8

- 10.2 Manufacturer is bound to replace or repair all valve parts which should result defective due to inadequate engineering or to the quality of materials and machining.
- 10.3 If valve defect or malfunctioning cannot be eliminated, Manufacturer shall replace the valve without delay,
- 10.4 Any defect occurring during the period of Guarantee shall be attended to by making all necessary modifications and repair of defective parts free of charge to the Purchaser as per the relevant clause of the bid document.
- 10.5 All expenses shall be to Manufacturer's account.
## TECHNICAL NOTES FOR GATE & GLOBE VALVES

# 1.0 General

- 1.1 Valves shall be designed, manufactured, tested, inspected, marked and supplied as per the specifications, applicable design standards & codes and manufacturing standards (latest editions) as specified.
- 1.2 Material test certificates (Physical property, Chemical composition & Heat treatment report) of the pressure containing parts shall be furnished for the valves supplied. Material test certificates of other parts shall also be furnished for verification during inspection.
- 1.3 For heavy valves, provision for lifting shall be made by way of lugs, eyebolts, or similar standard devices.
- 1.4 Unless otherwise stated, all flanged valves shall have end flanges integral with valve body. Weld on flanges are not acceptable. Flange finish shall be serrated finish 250 AARH (250)AARH to 500 AARH) or 125 AARH (125)AARH to 250 AARH) or 63 MRH (32 AARH to 63 AARH) as per valve specification sheet.
- 1.5 For all weld end valves, with bevel end as per ANSI B 16.25, the bevel contour shall be as follows:

	Wall Thickness	Weld Contour
Material		
Carbon Steel (Except Low	Upto 22 mm	Figure 2 Type A
Temp. Carbon Steel)	> 22 mm	Figure 3 Type A
Alloy Steel	Upto 10 mm	Figure 4
Stainless Steel &	> 10 mm & upto 25 mm	Figure 5 Type A
Low Temp Carbon Steel	> 25mm	Figure 6 Type A

- 1.6 If an overlay weld-deposit is used for the body seat ring, seating surface, the seat ring base material shall be at least equal to the corrosion resistance of the materials of the shell.
- 1.7 For valve body/ bonnet, forging is acceptable where castings are specified but not vice versa.
- 1.8 Material of construction of yoke shall be as a minimum equivalent to body/ bonnet material.
- 1.9 Stem shall be forged or machined from a forged bar. Castings are not permitted except integral stem.
- 1.10 Stelliting/ hard facing by deposition shall have minimum 1.6mm thickness. Renewable seat rings shall be seal welded.
- 1.11 Face to face dimension of flanged valves shall conform to ANSI B 16.10 to the extent covered. For valves not covered in the ANSI specification, Contractor shall furnish certified dimensional drawings.

- 1.12 Flange dimensions of steel, alloy steel and stainless steel flanged valves shall conform to ANSI B 16.5 for sizes up to 24" and API 605 for size 26" and above.
- 1.13 Flange dimensions for cast iron flanged valves shall conform to ANSI B 16.1 for size up to 24" class 125 and API 605 with flat face for sizes greater than 24".
- 1.14 Unless otherwise mentioned, various valves should conform to following standards / codes.

SW gate valves (1 $1/2$ '' and below)	:	API 602
SW Globe valves (1 1/2" and below)	: BS 5	5352
Flanged gate valves	API 600 &	API-6D
Flanged Globe valves	:	BS 1873

- 1.15 Wherever stellite is specified, it means facing of seat and disc are welded by Cr-Co-W alloy. Stellite facing shall maintain minimum hardness of 375 BHN at high temperature.
- 1.16 All weld end valves shall have bevel ends as per ANSI B 16.25.
- 1.17 If an overlay weld deposit is used for the body seat ring or seating surface, the seat ring base material shall be at least equal to corrosion resistance of the material of the shell.
- 1.18 By Pass
  - Unless otherwise noted, by-pass requirement for gate valves shall be under -

150 Class	:	on sizes 26" and above
300 Class	:	on sizes 16" and above
600 Class	:	on sizes 6" and above
900 Class	:	on sizes 4" and above
1500 Class	:	on sizes 4' and above
2500 Class	:	on sizes 3" and above

- By-pass valve shall be a globe valve.
- Contractor shall supply the by-pass valve duly tested and fitted to the main valve. By-pass attachment to the main valve body shall not be screwed. All fillet welds for by-pass installation shall be 100% examined by DP / MP test.
- 1.19 Spiral wound bonnet gasket is to be provided with inner / outer ring except when encapsulated gaskets type body bonnet joints are employed. Outer ring may be avoided in case of non-circular spiral wound gasket used in 150# valve provided the outermost layer of spiral touches the bolts ascertaining the centering.

### 1.20 <u>Pressure Test</u>

- Valves covered under API codes shall be tested as per API 598 unless otherwise specified in the applicable valve code.
- Valves covered under BS code shall be tested as per BS 6755 unless otherwise specified in the applicable valve codes.
- 1.21 For all austenitic stainless valves, inter-granular corrosion test shall have to be

conducted as per following: -

- ASTM A 262 Practice 'E' with acceptance criteria of "60 mils / year (max.)". OR
- ASTM A 262 practice 'E' with acceptance criteria of "No cracks as observed from 20X magnification U & Microscopic structure to be observed from 250X magnification".
- 1.22 When specifically asked for high temperature application of some grades of austenitic stainless steel (like SS 309, 310, 316, 316H etc.) ASTM A 262 practice 'C' with acceptance criteria "15 MILS/YEAR" shall have to be conducted. When testing is conducted as per practice 'E' photograph of microscopic structure shall be submitted for record.
- 1.23 For the IGC test as described in 1.16.1 & 1.16.2 two sets of samples shall be drawn from each solution treatment lot, one set corresponding to highest carbon content and other set corresponding to the highest rating/ thickness.

## 2.0 **OPERATION**

2.1 Valves shall be supplied with gear operations based on the following requirements:

	Class	Size Requiring Gear
Valve Types		Operation
Gate & Diaphragm Valves	150	14" and larger
	300	14" and larger
	600	12" and larger
	900	6" and larger
	1500	3" and larger
	2500	3" and larger
Globe Valves	900	6" and larger
	1500	3" and larger
	2500	3" and larger

2.2 Gear operator shall be as under with position indicators for open / close positions, with limit stops.

For Gate / Globe / Diaphragm Valves	Totally enclosed bevel gear in grease
	case with grease nipples/ CHECK

- 2.3 Gear operators shall be so designed to operate effectively with the differential pressure across the closed valve equal to the cold non-shock pressure rating.
- 2.4 Hand wheel diameter shall not exceed 750 mm and effort to operate shall not exceed 35 kg at hand wheel periphery. In case these limits cannot be satisfied for any valve, a gear operation shall be provided.

## 3.0 **INSPECTION AND TESTING**

- 3.1 All the mandatory shop tests and inspection required by the respective data sheet and applicable standards & codes etc. shall be carried out.
- 3.2 The extent of inspection by shall be as under. However the exact extent with hold

points shall be decided during review of the inspection plan to be submitted to Company as part of the post-order documentation.

3.3 Valves under NACE should follow the requirements of MR-01-75

# FORGED VALVES

- Visual and dimensional inspection
- Review of material test certificates
- Any mandatory or supplementary test
- Hydrostatic test of all valves
- Strip check on 1% of total ordered quantity of valves at random to verify compliance with specification requirements.

#### CAST STEEL VALVES

- Visual and dimensional inspection
- Review of material test certificates
- Review of radiographs / radiographic reports and reports of any other NDT tests, wherever applicable as per data sheets
- Any mandatory or supplementary tests
- Hydrostatic test 100% for body
- Strip check on 1% of total ordered quantity of valves at random to verify compliance with specification requirements.
- 3.4 For motor /actuator operated valves, functional / operational checks as per the requirements of the specifications shall be made on each valve.

### 4.0 **RADIOGRAPHY OF CAST VALVES**

4.1 When specifically not mentioned in individual data sheets, valves castings shall undergo radiographic examination as specified hereunder:

	RATING	SIZE RANGE	RADIOGRAPHY
MATERIAL			
All	150#	24" and below	Nil
	150#	26" and above	100%
	300#	16" and below	Nil
	300#	18" and above	100%
	600# and above	All sizes	100%

4.2 Radiography procedure areas of casting to be radiographed shall be as per ANSI B 16.34 and acceptance criteria shall be as per ANSI B 16.34 Annexure — B. However for areas of casting to be radiographed for types of valve not covered in ANSI B 16.34, Contractor shall enclose details of areas to be radiographed in line with ANSI B 16.34.

# 5.0 **IBR VALVES**

- 5.1 All valves described as "IBR Valves" shall be in accordance with the latest IBR (Indian Boiler Regulations) as well as the other requirements specified in the specification.
- 5.2 For BW / SW end carbon steel valves under "IBR", the chemical composition shall conform to the following:

Carbon (Max.) : 0.25%

Others (S, B, Mn) : As per IBR

Above composition is not applicable for non-IBR valves.

- 5.3 For all "IBR Valves", test certificate in form III-C shall be furnished duly signed by IBR inspection authority or an IBR approved representative.
- 5.4 All valves shall be painted red.

### 6.0 MARKING

- 6.1 Valves markings, symbols, abbreviations, etc. shall be in accordance with MSS-SP-25 or the standard referred to in the specifications as applicable Manufacturer's name, valve size and rating, material designation, nominal size, direction of flow (if any) etc. shall be integral on the body.
- 6.2 Each valve shall have a corrosion resistant tag giving size and valve tag/code no. securely attached on the valve body.
- 6.3 Paint or ink used for marking shall not contain any harmful metal or metal salts such as zinc, lead or copper which may result in corrosive attack on heating.
- 6.4 Carbon steel valves shall be painted with two coats of red oxide zinc chromate primer.
- 6.5 All alloy steel high temp valves shall be painted with heat resistant silicone paint suitable for intended temperature.

### 7.0 **DESPATCH**

- 7.1 Valves shall be dry, clean and free from moisture, dirt and loose foreign material of any kind.
- 7.2 Valves shall be protected from rust, corrosion and any mechanical damage during transportation, shipment, and storage.
- 7.3 Rust preventative applied on machined surfaces to be welded shall be easily removable with a petroleum solvent or shall not be harmful to welding.
- 7.4 Each end of valves shall be protected as follows:

Flange Face	:	Wood, plastic or metal cover
Beveled End	:	Wood, plastic or metal cover
SW / Screwed End	:	Plastics cap

- 7.5 End protectors to be used on flange faces shall be attached by at least three bolts or wires through bolt holes and shall not be smaller than the outside diameter of the flange. Plastic caps for SW / Screwed and valves shall be press fit type.
- 7.6 End protectors to be used on beveled ends shall be securely attached.

MSS-SP-25 or the standard referred to in the specifications as applicable Manufacturer's name, valve size and rating, material designation, nominal size, direction of flow (if any) etc. shall be integral on the body.

- 6.2 Each valve shall have a corrosion resistant tag giving size and valve tag/code no. securely attached on the valve body.
- 6.3 Paint or ink used for marking shall not contain any harmful metal or metal salts such as zinc, lead or copper which may result in corrosive attack on heating.
- 6.4 Carbon steel valves shall be painted with two coats of red oxide zinc chromate primer.
- 6.5 All alloy steel high temp valves shall be painted with heat resistant silicone paint suitable for intended temperature.

# 7.0 **DESPATCH**

- 7.1 Valves shall be dry, clean and free from moisture, dirt and loose foreign material of any kind.
- 7.2 Valves shall be protected from rust, corrosion and any mechanical damage during transportation, shipment, and storage.
- 7.3 Rust preventative applied on machined surfaces to be welded shall be easily removable with a petroleum solvent or shall not be harmful to welding.
- 7.4 Each end of valves shall be protected as follows:

Flange Face	:	Wood, plastic or metal cover
Beveled End	:	Wood, plastic or metal cover
SW / Screwed End	:	Plastics cap

- 7.5 End protectors to be used on flange faces shall be attached by at least three bolts or wires through bolt holes and shall not be smaller than the outside diameter of the flange. Plastic caps for SW / Screwed and valves shall be press fit type.
- 7.6 End protectors to be used on beveled ends shall be securely attached.

MECON LIMITED	STANDARD TECHNICAL		
REGD. OFF: RANCHI 834002	OIL & GAS SB	OIL & GAS SBU, DELHI	
		DOCUMENT NO.	Page 1 of 7
TITLE	PACKING, TRANSPORTATION	MEC/TS/05/21/061	REVISION : 0
	& HANDLING OF VALVES		EDITION : 1

# STANDARD TECHNICAL SPECIFICATION

# FOR

# PACKING, TRANSPORTATION AND HANDLING OF VALVES

# SPECIFICATION NO .: MEC/TS/05/21/061



# (OIL & GAS SBU) MECON LIMITED DELHI 110 092

PREPARED BY:	CHECKED BY:	APPROVED BY:	ISSUE DATE :
(ASHISH MATHUR) SDE	Hawd (HARSH KUMAR) MGR	(A. K. GUPTA) DGM	11.09.2018

MECON LIMITED	STANDARD TECHNICAL			
REGD. OFF: RANCHI 834002	OIL & GAS SB	& GAS SBU, DELHI		
		DOCUMENT NO.	Page 2 of 7	
TITLE	PACKING, TRANSPORTATION	MEC/TS/05/21/061	REVISION : 0	
	& HANDLING OF VALVES		EDITION : 1	

# AMENDMENT STATUS

SI.	SI. Clause / Paragraph /	Page		Dette	BY		Verifie	ed	
No.	Drawing Amended	No.	Rev.	Rev.	Date	Name	Sig.	Name	Sig.
					1				
				A MARCE	a station				

MECON LIMITED	STANDARD TECHNICA			
REGD. OFF: RANCHI 834002	OIL & GAS SBU, DELHI		18 201-7 100 Compton	
<u>×</u>		DOCUMENT NO.	Page 3 of 7	
TITLE	PACKING, TRANSPORTATION	MEC/TS/05/21/061	REVISION : 0	
	& HANDLING OF VALVES		EDITION : 1	

# TABLE OF CONTENTS

1.0	SCOPE	4
2.0	PACKING	4
3.0	HANDLING	4
4.0	TRANSPORTATION	5

MECON LIMITED	STANDARD TECHNICA		
REGD. OFF: RANCHI 834002	OIL & GAS SB	OIL & GAS SBU, DELHI	
		DOCUMENT NO.	Page 4 of 7
TITLE	PACKING, TRANSPORTATION	MEC/TS/05/21/061	REVISION : 0
	& HANDLING OF VALVES		EDITION : 1

## 1.0 SCOPE

This specification covers the minimum requirements for Packing, Handling & transportation of valves and actuators. Though this specification covers the minimum requirement for packing, handling and transportation of valves, it is to be noted that any defect/ damage arising out of improper packing, handling & transportation shall be the responsibility of vendor. The delay due to rectification of such faults shall be to vendor's account. The date of delivery of material at site shall be considered as the day on which last such rectified material is delivered/ rectified at designated store.

# 2.0 PACKING

2.1 All valves shall be completely drained of test fluid and thoroughly dried after hydrotesting. The machined surfaces shall be coated with a light film of high viscosity rust inhibiting oil which will not become fluid and run off at temperatures below 80°C.

2.2 Flanged valves NPS 6" and smaller in Class 150 and Class 300 shall be fitted with UV resistant plastic covers. For other sizes, valve end flanges shall be fitted with plywood covers. The cover diameter shall be the same as the outside diameter of the flange and shall be at least 10 mm thick for valves up to NPS 24" and 12 mm thick for valves NPS 26" and larger. The cover shall be attached by machine bolts with a nut and washer fitted on the inside of the flange. There shall be minimum four (4) bolts on valves up to NPS 24" nominal size and eight (8) bolts on valves NPS 26 inch and larger. The bolts diameter shall not be less than 1/4 the size of the flange bolt hole.

2.3 In addition to the above, all flange facings (ring joint, raised and flat) shall be covered with NBR (based) rubber Self-Adhesive protection (see fig below) that meets the following:

Oil, ozone and weather resistant

□ Minimum thickness of 1.5 mm

□ Withstand temperatures up to 75°C

□ Non deforming, loosening or detaching

□ Proof against sand blasting

□ □ No glue residue

Chloride free

MECON LIMITED	STANDARD TECHNICA		
REGD. OFF: RANCHI 834002	OIL & GAS SB	1) 1001 7 1000 Control	
		DOCUMENT NO.	Page 5 of 7
TITLE	PACKING, TRANSPORTATION	MEC/TS/05/21/061	REVISION : 0
	& HANDLING OF VALVES		EDITION : 1

- 2.4 Buttweld end valves shall be blanked on each end by high impact plastic bevel protectors, so that bevels are protected from possible mechanical damage during transportation.
- 2.5 The ends of threaded and socket weld end valves shall be protected with tight fitting plastic caps.
- 2.6 Packing shall be strong and sturdy such that it can withstand loading/unloading, pushing and crane lifting etc. All packaging shall be done in such a manner as to reduce volume and weight as much as possible without jeopardizing the safety of the material. All packing materials shall be new.
- 2.7 Stacking of multiple valves in single box is permitted upto 4" NB. However, in such case suitable partitions are to be made inside packing box.
- 2.8 Where height limitations restrict transportation of valve with actuator in assembled condition, actuator should be dismantled after successful testing at shop. However, the same need to be proposed by valve manufacturer during inspection of said valves and take the approval for Client/ PMC.
- 2.9 When valve, extended stem and actuators are transported in dismantled condition, the same shall be reassembled after fitment of valve at site. Valve vendor to deploy their representative within 3 days once the intimation is sent from site. Any delay beyond 3 days shall be to supplier's account.
- 2.10 Valve manufacturers to note that the safe transportation of assembled valve with actuator is in their scope of work. It is therefore required that the valve manufacturer should order actuator meeting the packing guidelines given in this specification. No claim shall be entertained on account of actuator manufacturer's non compliance of requirements specified in this specification, and the valve with actuator shall leave manufacturer's workshop after meeting the terms given in this specification.
- 2.11 Valves shall not be packed in poly wrap irrespective of the increase in shipping/ transport volume. Box of wood/ ply board etc. shall only be used to pack the valves with/ without actuator irrespective of the size/ rating of the valve.
- 2.12 The packing shall have suitable lifting arrangement to enable the lifting of valve with the packing. Suitable provisions/ supports shall be provided from support foot/ lifting lugs to enable to lift the valve with packing.

MECON LIMITED	STANDARD TECHNICA	SPECIFICATION		
REGD. OFF: RANCHI 834002	OIL & GAS SBU, DELHI		100 1000 Control	
		DOCUMENT NO.	Page 6 of 7	
TITLE	PACKING, TRANSPORTATION	MEC/TS/05/21/061	REVISION : 0	
	& HANDLING OF VALVES		EDITION : 1	

- 2.13 Where it is required to transport valve and actuator separately, above clauses shall be individually applicable for valve and actuator.
- 2.14 Assembled Valves shall be properly secured inside packing in order to avoid any contact with packing material during transport.
- 2.15 For extended stem valves, it is permissible to dismantle stem extension and actuator and as such the valve may be transported in three parts, each part complying individually the requirements of this specification.
- 2.16 Actuators shall be packed in wooden box with proper cushioning of damage prone parts like sockets, tubing, panel boxes etc.
- 2.17 Actuator cylinders shall be mounted on base with the help of metallic U-clamps/ welding on reinforcement plate. Metallic U-clamps to be used with double bolts on either side of U clamp.
- 2.18 Actuator components layout shall be such that to minimize packing volume. Back-up tank shall be put in horizontal position only, wherever feasible.
- 2.19 The manufacturer shall exhibit the packing meeting to the requirement of this specification during inspection and take clearance.

## 3.0 HANDLING

- 3.1 Manufacturer to ensure that during lifting hooks for assembly are attached to body/ end piece casting/ forging only and not on the pup piece. Any pup piece having hook attachment mark may be rejected.
- 3.2 Assmebled valves, at all times, shall be lifted through lifting lugs only and not from the pup pieces.
- 3.3 Support foot shall be provided on body only in bolted design. In no case, the support foot shall be fastened in body bolting.
- 3.4 Lifting Lugs shall be provided on body/ tail piece in bolted design. In no case, the lifting lugs shall be fastened in body bolting.
- 3.5 Valve vendor to work in close coordination with actuator vendor to ensure that the sling put in lifting lug of valve do not interfere with the actuator/ tubing during lifting at site. Any breakage during site lifting due to fouling of tubing/

MECON LIMITED	STANDARD TECHNICA		
REGD. OFF: RANCHI 834002	OIL & GAS SB	J, DELHI	10 12000 Control
		DOCUMENT NO.	Page 7 of 7
TITLE	PACKING, TRANSPORTATION	MEC/TS/05/21/061	REVISION : 0
	& HANDLING OF VALVES		EDITION : 1

actuator components during site lifting shall be in supplier' account.

3.6 Extended Stem valves shall have placement of lifting lugs to ensure the lifting of valve in stem vertical condition only. Under no condition the valve is to be lifted in Stem horizontal/ inclined position.

# 4.0 TRANSPORTATION

- 4.1 If the valve and actuator in assembled condition can be accommodated on low bed trailer, low bed trailer shall only be used for inland transportation. Dismantling of valve and actuator shall not be permitted under such case.
- 4.2 Valve shall be secured on trailer/ truck bed with ropes suitably attached with valve boxes. Type of rope selection shall depend upon weight of valve.
- 4.3 Tack welds on trailer/ truck bed shall not be used as a fastening method.
- 4.4 Bolting may be used to securely fasten the valve base on trailer if the provision is available. No. and diameter of bolts shall be suitably chosen as per weight of valve to ensure that bolts do not shear off during transportation.
- 4.5 For large size valves, Loading shall be done preferably by hanging the valve in position and moving the vehicle to valve sitting position.
- 4.6 Since unloading of valves is under valve manufacturer's scope, it is to be ensured that valve manufacturer's representative shall be available at designated store to facilitate the same. Valve manufacturer has to keep the track of vehicle movement accordingly. If due to project exigency/ time constraint the unloading has to be done during manufacturer's representative's absence, any damage during such unloading shall be attributable to manufacturer only.

# SPECIFICATION FOR SHOP & FIELD PAINTING

SPECIFICATION NO.: MEC/S/05/21/07



(OIL & GAS SBU) MECON LIMITED DELHI 110 092

Page 482 of 644

SIL NO.DESCSIL NO.DESC1.0GENE2.0SCOP3.0CODE4.0EQUIF5.0SURF6.0PAINT7.0PAINT7.1PRE-ETEMPAND E7.2REPAWELDSTEEL	OIL & GAS SBU HOP & FIELD PAINTING CON RIPTION RAL E S & STANDARDS MENT ACE PREPARATION MATERIALS ING SYSTEMS RECTION/ PRE-FABRICATIO ERATURE CARBON STEEL & QUIPMENT ETC. R OF PRE-ERECTION/ FABRI ING FOR CARBON STEEL, LO	AND SHOLD AND SHOLD AND SHOLD AND SHOLD AND SHOLD AND SHOLD AND SHOLD AND SHOLD AND AND SHOLD AND AND AND AND AND AND AND AND AND AN	UMENT NO. S/05/21/07 S OP PRIMING F DY STEEL, STE	OR CARBON	ge 1 of 1 /ISION : 0 ITION : 1 N STEEL, LOW FURES, PIPINC
TITLE S	HOP & FIELD PAINTING CON RIPTION RAL E S & STANDARDS MENT ACE PREPARATION MATERIALS ING SYSTEMS RECTION/ PRE-FABRICATIOI ERATURE CARBON STEEL & QUIPMENT ETC. R OF PRE-ERECTION/ FABRI ING FOR CARBON STEEL, LO	DOC MEC/ TENTS TENTS	UMENT NO. S/05/21/07 S DP PRIMING F DY STEEL, STE	OR CARBON	ge 1 of 1 /ISION : 0 ITION : 1 N STEEL, LOW FURES, PIPINC R ERECTION/
TITLE S	HOP & FIELD PAINTING CON RIPTION RAL E S & STANDARDS MENT ACE PREPARATION MATERIALS ING SYSTEMS RECTION/ PRE-FABRICATIOI ERATURE CARBON STEEL & QUIPMENT ETC. R OF PRE-ERECTION/ FABRI ING FOR CARBON STEEL, LO	MEC/	S/05/21/07 S DP PRIMING F DY STEEL, STE ND SHOP PRIM	OR CARBON	ISION : 0 ITION : 1 N STEEL, LOW FURES, PIPINC R ERECTION/
SL. NO.         DESC           1.0         GENE           2.0         SCOP           3.0         CODE           4.0         EQUIF           5.0         SURF.           6.0         PAINT           7.1         PRE-E           TEMP         AND E           7.2         REPA           VELD         STEEL	<b>CON</b> <b>RIPTION</b> RAL E S & STANDARDS MENT ACE PREPARATION MATERIALS ING SYSTEMS RECTION/ PRE-FABRICATIO ERATURE CARBON STEEL & QUIPMENT ETC. R OF PRE-ERECTION/ FABRI ING FOR CARBON STEEL, LO	IN AND SH	<b>S</b> OP PRIMING F DY STEEL, STE ND SHOP PRIM	OR CARBON EEL STRUCT MING AFTEF	ITION : 1 N STEEL, LOW FURES, PIPINC R ERECTION/
SL. NO.DESC1.0GENE2.0SCOP3.0CODE4.0EQUIF5.0SURF.6.0PAINT7.0PAINT7.1PRE-ETEMPAND E7.2REPAWELDSTEEL	<b>CON</b> <b>RIPTION</b> RAL E S & STANDARDS MENT ACE PREPARATION MATERIALS ING SYSTEMS RECTION/ PRE-FABRICATION ERATURE CARBON STEEL & QUIPMENT ETC. R OF PRE-ERECTION/ FABRI ING FOR CARBON STEEL, LO	TENTS	<b>S</b> OP PRIMING F DY STEEL, STE ND SHOP PRIM	OR CARBON EEL STRUCT MING AFTEF	N STEEL, LOW FURES, PIPINC R ERECTION/
SL. NO.DESC1.0GENE2.0SCOP3.0CODE4.0EQUIF5.0SURF6.0PAINT7.0PAINT7.1PRE-ETEMPAND E7.2REPAWELDSTEEL	RIPTION RAL E S & STANDARDS MENT ACE PREPARATION MATERIALS ING SYSTEMS RECTION/ PRE-FABRICATIO ERATURE CARBON STEEL & QUIPMENT ETC. R OF PRE-ERECTION/ FABRI ING FOR CARBON STEEL, LO	N AND SH	DP PRIMING F DY STEEL, STE ND SHOP PRIM	OR CARBON EEL STRUCT MING AFTEF	N STEEL, LOW FURES, PIPINC R ERECTION/
1.0         GENE           2.0         SCOP           3.0         CODE           4.0         EQUIF           5.0         SURF,           6.0         PAINT           7.0         PAINT           7.1         PRE-E           TEMP         AND E           7.2         REPA           VELD         STEEL	RAL E S & STANDARDS MENT ACE PREPARATION MATERIALS ING SYSTEMS RECTION/ PRE-FABRICATIO ERATURE CARBON STEEL & QUIPMENT ETC. R OF PRE-ERECTION/ FABRI ING FOR CARBON STEEL, LO	N AND SH LOW ALLO	OP PRIMING F DY STEEL, STE ND SHOP PRII	OR CARBON EEL STRUCT MING AFTEF	N STEEL, LOW FURES, PIPINC R ERECTION/
2.0 SCOP 3.0 CODE 4.0 EQUIF 5.0 SURF 6.0 PAINT 7.0 PAINT 7.1 PRE-E TEMP AND E 7.2 REPA WELD STEEL	E S & STANDARDS MENT ACE PREPARATION MATERIALS ING SYSTEMS RECTION/ PRE-FABRICATIO ERATURE CARBON STEEL & QUIPMENT ETC. R OF PRE-ERECTION/ FABRI ING FOR CARBON STEEL, LO	N AND SH LOW ALL ICATION A DW TEMPE	OP PRIMING F DY STEEL, STE ND SHOP PRII	OR CARBON EEL STRUCT MING AFTEF	N STEEL, LOW FURES, PIPINC R ERECTION/
3.0 CODE 4.0 EQUIF 5.0 SURF 6.0 PAINT 7.0 PAINT 7.1 PRE-E TEMP AND E 7.2 REPA WELD STEEL	S & STANDARDS MENT ACE PREPARATION MATERIALS ING SYSTEMS RECTION/ PRE-FABRICATIO ERATURE CARBON STEEL & QUIPMENT ETC. R OF PRE-ERECTION/ FABRI ING FOR CARBON STEEL, LO	N AND SH LOW ALL ICATION A DW TEMPE	OP PRIMING F DY STEEL, STE ND SHOP PRII	OR CARBON EEL STRUCT MING AFTEF	N STEEL, LOW FURES, PIPINC R ERECTION/
4.0 EQUIF 5.0 SURF. 6.0 PAINT 7.0 PAINT 7.1 PRE-E TEMP AND E 7.2 REPA WELD STEEL	ACE PREPARATION MATERIALS ING SYSTEMS RECTION/ PRE-FABRICATIO ERATURE CARBON STEEL & QUIPMENT ETC. R OF PRE-ERECTION/ FABRI ING FOR CARBON STEEL, LO	N AND SH LOW ALL ICATION A DW TEMPE	OP PRIMING F DY STEEL, STE ND SHOP PRII	OR CARBON EEL STRUCT MING AFTEF	N STEEL, LOW FURES, PIPINC R ERECTION/
4.0 EQUIT 5.0 SURF. 6.0 PAINT 7.0 PAINT 7.1 PRE-E TEMP AND E 7.2 REPA WELD STEEL	ACE PREPARATION MATERIALS ING SYSTEMS RECTION/ PRE-FABRICATIO ERATURE CARBON STEEL & QUIPMENT ETC. R OF PRE-ERECTION/ FABRI ING FOR CARBON STEEL, LO	N AND SH LOW ALL ICATION A OW TEMPE	OP PRIMING F DY STEEL, STE ND SHOP PRII	OR CARBON EEL STRUCT MING AFTEF	N STEEL, LOW FURES, PIPING R ERECTION/
5.0 SURF. 6.0 PAINT 7.0 PAINT 7.1 PRE-E TEMP AND E 7.2 REPA WELD STEEI	ACE PREPARATION MATERIALS ING SYSTEMS RECTION/ PRE-FABRICATIO ERATURE CARBON STEEL & QUIPMENT ETC. R OF PRE-ERECTION/ FABRI ING FOR CARBON STEEL, LO	N AND SH LOW ALL ICATION A DW TEMPE	OP PRIMING F OY STEEL, STE ND SHOP PRII	OR CARBON EEL STRUCT MING AFTEF	N STEEL, LOW FURES, PIPING R ERECTION/
6.0 PAINT 7.0 PAINT 7.1 PRE-E TEMP AND E 7.2 REPA WELD STEEI	MATERIALS ING SYSTEMS RECTION/ PRE-FABRICATIO ERATURE CARBON STEEL & QUIPMENT ETC. R OF PRE-ERECTION/ FABRI ING FOR CARBON STEEL, LO	N AND SH LOW ALL ICATION A DW TEMPE	OP PRIMING F OY STEEL, STE ND SHOP PRII	OR CARBON EEL STRUCT MING AFTEF	N STEEL, LOW FURES, PIPING R ERECTION/
7.0 PAINT 7.1 PRE-E TEMP AND E 7.2 REPA WELD STEEI	ING SYSTEMS RECTION/ PRE-FABRICATIO ERATURE CARBON STEEL & QUIPMENT ETC. IR OF PRE-ERECTION/ FABRI ING FOR CARBON STEEL, LO	N AND SH LOW ALLO	OP PRIMING F OY STEEL, STE ND SHOP PRII	OR CARBON EEL STRUCT MING AFTEF	N STEEL, LOW FURES, PIPING R ERECTION/
7.1 PRE-E TEMP AND E 7.2 REPA WELD STEEL	RECTION/ PRE-FABRICATIO ERATURE CARBON STEEL & QUIPMENT ETC. R OF PRE-ERECTION/ FABRI ING FOR CARBON STEEL, LO	N AND SHO LOW ALLO	OP PRIMING F DY STEEL, STE ND SHOP PRII	OR CARBON EEL STRUCT MING AFTEF	N STEEL, LOW FURES, PIPING R ERECTION/
7.2 REPA STEEL	ERATURE CARBON STEEL & QUIPMENT ETC. IR OF PRE-ERECTION/ FABRI ING FOR CARBON STEEL, LO	ICATION A	ND SHOP PRI	EEL STRUCT	TURES, PIPING
7.2 REPA WELD STEEL	ERATURE CARBON STEEL & QUIPMENT ETC. IR OF PRE-ERECTION/ FABRI ING FOR CARBON STEEL, LC	ICATION A	ND SHOP PRI	VING AFTEF	RERECTION
AND E 7.2 REPA WELD STEEI	QUIPMENT ETC. IR OF PRE-ERECTION/ FABRI ING FOR CARBON STEEL, LC	ICATION A	ND SHOP PRI	MING AFTEF	R ERECTION/
7.2 REPA WELD STEEI	IR OF PRE-ERECTION/ FABRI	ICATION A	ND SHOP PRI	MING AFTEF	RERECTION/
WELD STEEI	ING FOR CARBON STEEL, LC	OW TEMPE			//
STEEL			RATURE CAR	BON STEEL	& I OW ALL O
SIEE					
	STEEL, ITEMS IN ALL ENVIRONMENTS.				
8.0 FIELD	FIELD PAINT SYSTEM FOR NORMAL CORROSIVE ENVIRONMENT (FOR CARBON				
STEEI	STEEL, LOW TEMPERATURE CARBON STEEL & LOW ALLOY STEEL)				
	ELD DAINT SYSTEM FOR CORROSIVE ENVIRONMENT (FOR CARBON STEEL LOW				
9.0 TIELD	ELD PAINT STSTEM FOR CORROSIDE ENVIRONMENT (FOR CARDON STEEL, LOW				
IEMP	TEMPERATURE CARBON STEEL AND LOW ALLOY STEEL)				
10.0 FIELD	PAINT SYSTEM FOR HIGHLY	Y CORROS	IVE AREA (FO	R CARBON S	STEEL, LOW
ALLO	ALLOY STEEL) EXTERNAL SURFACE OF UNINSULATED COLUMNS, VESSELS, HEAT				
EXCH	EXCHANGERS, BLOERS, PIPING. PUMPS. TOWERS. COMPRESSORS. FLARE LINES.				
CTDU		01111 0, 10		112000110,1	
SIRU	STRUCTURAL STEEL ETC.				
11.0 FIELD	HIELD PAINT SYSTEM FOR CARBON STEEL STORAGE TANKS (EXTERNAL) FOR ALL				
ENVIF	ENVIRONMENTS				
12.0 FIELD	FIELD PAINT SYSTEM FOR CARBON STEEL AND LOW ALLOY STEEL STORAGE TANK				
13.0 COAT	ING SYSTEM FOR EXTRNAL	SIDE OF U	NDERGROUN	D CARBON S	STEEL, PLAN
PIPIN	G AND TANKS.				
14.0 PAINT	ING UNDER INSULATION FO	R (HOT, C	OLD & SAFETY	() CARBON S	STEELLOW
	EDATI DE CADRON STEEL &				
		STAINLES	5 51 LLL FIFII		
ENVIE	ONMENT				
15.0 INTER	NAL PROTECTION OF CARB	ON STEEL	WATER BOXE	ES AND TUB	E SHEETS OF
COOL	ERS/ CONDENSERS.				
16.0 EIELD	PAINTING SYSTEM FOR GUT				ET
17.0 11220					- L- I
17.0 STOR	AGE				
18.0 COLO	URS CODE FOR PIPING				
19.0 IDENT	IFICATION OF VESSELS. PIP	ING ETC.			
20.0 DAINT	ING FOR CIVIL DEFENCE PE		NTS		
	TOTION AND TEATING		10		
21.0 INSPE					
22.0 GUAR	ANTEE				
23.0 QUAL	FICATION CRITERIA O PAIN	TING CON	FRACTOR.		
24.0 PROC	EDURE FOR APPROVAL OF !	NEW PAIN	T MANUFACTI	JRERS	
		ATURES			
ANNEXURE-II- LIST C	F RECOMMENDED MANUFA	UTURE'S F	KUDUCIS.		
				RV.	
FILFANLU DI.			AFFAUVED		1550L DATE
(Shalini Singh)	(Supil Kumar)		(A K Joh	ri)	Dec 2009
		<u> </u>	(היה)	··/	Dec. 2000

MECON LIMITED REGD. OFF: RANCHI	STANDARD TECHNICAL		
834002	OIL & GAS SBU	मिठ मिठान गर्भ 9001:2000 Contract	
		DOCUMENT NO.	Page 1 of 1
TITLE	SHOP & FIELD PAINTING	MEC/S/05/21/07	REVISION : 0
			EDITION: 1

# **AMENDMENT STATUS**

SI. No.	Clause / Paragraph / Annexure / Exhibit / Drawing Amended	Page No.	Revision	Date	By (Name)	Verified (Name)

MECON LIMITED REGD. OFF: RANCHI	STANDARD TECHNICAL		
834002	OIL & GAS SBU,	मित्रान मित्रान 1001:2000 Contrin	
		DOCUMENT NO.	Page 1 of 54
TITLE	SHOP & FIELD PAINTING	MEC/S/05/21/07	REVISION : 0
			EDITION: 1

# 1.0 **GENERAL**

- 1.1 These technical specifications shall be applicable for the work covered by the contract, and without prejudice to the various codes of practice, standard specifications etc. it is understood that contractor shall complete the work in all respects with the best quality of materials and workmanship and in accordance with the best engineering practice and instructions of Engineer-in-charge.
- 1.2 Wherever it is stated in the specification that a specific material is to be supplied or a specific work is to be done it shall be deemed that the same shall be supplied or carried out by the contractor.

Any deviation from this standard without within deviation permit from appropriate authority will result in rejection to job.

# 2.0 **SCOPE**

Scope of work covered in the specification shall include, but not limited to the following.

2.1 This specification defines the requirements for surface preparation, selection and application of paint on external surfaces of equipment, vessels, machinery, piping, ducts, steels structures, external & internal protection of storage tanks for all services RCC Chimney & MS Chimney with or without refractory lining and flare lines etc.

# 2.2 Extent of Works

- 2.2.1 The following surface and materials shall require shop, pre-erection and field painting.
  - a. All uninsulated C. S. & A.S. equipment like columns, vessels, drums, storage tanks, heat exchangers, pumps, compressors, electrical panels and motors etc.
  - b. All uninsulated carbon and low alloy piping fitting and valves (including painting of identification marks), furnace, ducts and stacks.
  - c. All items contained in a package unit as necessary.
  - d. All structural steel work, pipe, structural steel supports, walkways, handrails, ladders, platforms etc.

MECON LIMITED		STANDARD TECHNICAL	SPECIFICATION				
834002		OIL & GAS SBU,	DELHI	वित्र मेकान मेकान 9001:2000 Contract			
			DOCUMENT NO.	Page 2 of 54			
TITLE	9	SHOP & FIELD PAINTING	MEC/S/05/21/07	REVISION: 0			
				EDITION: 1			
	e.	RCC/ MS chimneys with or	without refractory lining 8	Flare lines.			
	f.	Identification colour bands aluminium clad, galvanised	Identification colour bands on all piping as required including insulated aluminium clad, galvanised, SS and non-ferrous piping.				
	g.	Identification lettering/ nur piping insulated aluminium	Identification lettering/ numbering on all painted surface of equipment/ piping insulated aluminium clad, galvanised, SS and non-ferrous piping.				
	h.	Marking/ identification signs on painted surfaces of equipment/ piping for hazardous service.					
	i.	Supply of all primers, paints and all other materials required for paintin other than owner's supply.					
	j.	Over insulation surface of equipments and pipes wherever required.					
	k.	Painting under insulation for carbon steel and stainless steel as specified.					
	I.	Repair work of damaged/ preerection/ fabrication shop primer and we joints at field.					
2.2.2	The speci	following surface and materials shall not be painted unless otherwise cified:					
	a.	Uninsulated austentic stainless steel.					
	b.	Plastic and/ or plastic coated materials.					
	C.	Non ferrous materials like aluminium, galvanised "piping", "gratings" and "handrails" etc. except G. I. Towers.					
2.3	Docι	cuments					
2.3.1	The docu	contractor shall perform the work in accordance with the following uments issued to him for executions of work.					
	a.	Bill of quantities for piping,	equipment, machinery an	d structure etc.			
	b.	Piping line list.					
	C.	Painting specifications inclu	uding special civil defence	requirement.			

MECON LIMITED	STANDARD TECHNICAL		
834002	OIL & GAS SBU,	मिकान 9001:2000 Comman	
		DOCUMENT NO.	Page 3 of 54
TITLE	SHOP & FIELD PAINTING	MEC/S/05/21/07	<b>REVISION: 0</b>
			EDITION: 1

- 2.4 Unless otherwise instructed final painting on pre-erection/ shop primed pipes and equipments shall be painted in the field, only after mechanical completion and testing on system are completed as well as, after completion of steam purging wherever required.
- 2.5 Changes and deviations required for any specific job due to clients requirement or otherwise shall be referred to MECON for deviation permit.

# 3.0 CODES & STANDARDS

3.1 Without prejudice to the provision of clause 1.1 above and the detailed specifications of the contract, the following codes and standards shall be followed for the work covered by this contract.

IS:5	:	Colour coding
IS-101	:	Methods of test for ready mixed paint and enamels.
IS-2379:1990	:	Indian standard for pipe line Identification –Colour code.
ASTM Vol. 6.01 and 6.03	:	American standard test methods for Paints and coatings.
ANSI A 13.1-1981	:	Scheme for Identification of piping systems : American National Standard Institution.

# 3.2 Surface Preparation Standards:

Following standards shall be followed for surface preparations:

3.2.1 Swedish Standard : SIS-05 5900-1967/ ISO-8501-1-1998 (Surface preparation standards for painting steel surfaces).

This standard contains photographs of the various standards on four different degrees of rusted steel and as such is preferable for inspection purpose by the Engineer-in-Charge.

	STANDARD TECHNICAL SPECIFICATION						
834002	OIL & GAS SBU	, DELHI	10 मेकान 9001:2000 CONN				
		DOCUMENT NO.	Page 4 of 54				
TITLE	SHOP & FIELD PAINTING	MEC/S/05/21/07	REVISION : 0				
			EDITION: 1				
3.2.2	Steel structure painting Council (SSPC-SP).	, U.S.A (surface prepa	rations specifications				
3.2.3	British standard (surface finish or	Blast-cleaned for painting	) BS:4232				
3.2.4	National Associations of Corrosion	n Engineers, U.S.A. (NAC	E)				
3.2.5	Various International Standards preparation are given in Table-I.	arious International Standards equivalent to swedish Standard for surface reparation are given in Table-I.					
3.3	The contractor shall arrange, at I any one of the above standards and	ne contractor shall arrange, at his own cost, to keep a set of latest edition of ny one of the above standards and codes at site.					
3.4	The paint manufacturer's instruction times. Particular attention shall be	e paint manufacturer's instructions shall be followed as far as practicable at all es. Particular attention shall be paid to the following:					
	a. Instructions for storage temperature.	to avoid exposure as v	well as extremes of				
	b. Surface preparations prior	to painting.					
	c. Mixing and thinning.						
	d. Application of paints and between coats.	d the recommended lin	nit on time intervals				
4.0	EQUIPMENT						
4.1	All tools, brushes, rollers, spray of leaning and all equipments, scaff water blasting equipments & air of suitable for the work and all in contractor at site and in sufficient	guns, abrasive materials folding materials, shot/ w compressors etc. required n good order and shall quantity.	hand/ power tools for et abrassive blasting, d to be used shall be be arranged by the				
	Mechanical mixing shall be used to systems except that the Engineer quantities at his discretion.	for paint mixing operation -in-Charge may allow the	s in case of two pack hand mixing of small				

# 5.0 SURFACE PREPARATION, SHOP COAT, COATING APPLICATION & REPAIR AND DOCUMENTATION

5.1 General

MECON LIMITED	STANDARD TECHNICAL			
834002	OIL & GAS SBU	03 मेकान 8001:2000 Control		
		DOCUMENT NO.	Page 5 of 54	
TITLE	SHOP & FIELD PAINTING	MEC/S/05/21/07	REVISION: 0	
			EDITION: 1	

- 5.1.1 In order to achieve the maximum durability, one or more of following methods of surface preparation shall be followed, depending on condition of steel surface and as instructed by Engineer-in-Charge. Adhesion of the paint film to surface depends largely on the degree of cleanliness of the metal surface. Proper surface preparation contributes more to the success of the paint protective system:
  - a. Manual or hand tools cleaning.
  - b. Mechanical or power tool cleaning.
  - c. Blast cleaning.
- 5.1.2 Mill scale, rust, rust scale and foreign matter shall be removed fully to ensure that a clean and dry surface is obtained. The minimum acceptable standard in case of manual or hand tool cleaning shall be St. 2 or equivalent, in case of mechanical or power tool cleaning it shall be St. 3 or equivalent, in case of blast cleaning it shall be Sa 21<sup>1</sup>/<sub>2</sub> or equivalent as per Swedish Standard SIS-055900-1967/ ISO-8501-1-1988. Where highly corrosive condition exits, then blast cleaning shall be Sa3 as per Swedish Standard.

Remove all other contaminants, oil, grease etc. by use of an aromatic solvent prior to surface cleaning.

- 5.1.3 Blast cleaning shall not be performed where dust can contaminate surfaces undergoing such cleaning or during humid weather conditions having humidity exceeding 85%.
- 5.1.4 Irrespective of the method of surface preparation, the first coat of primer must be applied on dry surface. This should be done immediately and in any case within 4 hours of cleaning of surface. However, at times of unfavourable weather conditions, the Engineer-in-Charge shall have the liberty to control the time period, at his sole discretion and / or to insist on recleaning, as may be required, before primer application is taken up. In general, during unfavourable weather conditions, blasting and painting shall be avoided as far as practicable.
- 5.1.5 The external surface of R.C.C. chimney to be painted be dry and clean. Any loose particle of stand, cement, aggregate etc. shall be removed by rubbing with soft wire brush if necessary, acid etching with 10-15% HCL solution about 15 minutes shall be carried out and surface must be thorought washed with water to remove acid & loose particles then dry completely before application of paint.

# 5.2 Procedure of Surface Preparation.

MECON LIMITED REGD. OFE: RANCHI	STANDARD TECHNICAL	कि मेकान का 1:2000 Contrin		
834002	OIL & GAS SBU,			
		DOCUMENT NO.	Page 6 of 54	
TITLE	SHOP & FIELD PAINTING	MEC/S/05/21/07	<b>REVISION: 0</b>	
			EDITION: 1	

# 5.2.1 Blast Cleaning

# 5.2.1.1 Air Blast Cleaning

The surface shall be blast cleaned using one of the abrasives: AL<sub>2</sub>O<sub>2</sub> particles chilled casts iron or malleable iron and steel at pressure of 7kg. Cm<sup>2</sup> at appropriate distance and angle depending on nozzle size maintaining constant velocity and pressure. Chilled cast iron, malleable iron and steel shall be in the form of shot or grit of size not greater than 0.055" maximum in case of steel and malleable iron and 0.04" maximum in case of chilled iron. Compressed air shall be free form moisture and oil. The blasting nozzles should be venture style with tungsten carbide or boron carbide as the material for liners. Nozzles orifice may vary from 3/16" to 3/4". On completion of blasting operation, the blasted surface shall be clean and free from any scale or rust and must show a grey white metallic lusture. Primer or first coat of paint shall be applied within 4 hours of surface preparation. Blast cleaning shall bot be done outdoors in bad weather without adequate protection or when there is dew on the metal which is to be cleaned, surface profile shall be uniform to provide good key to the paint adhesion (i.e.35to 50u). If possible vacuum collector shall be installed for collecting the abrasive and recycling.

# 5.2.1.2 Water Blast cleaning

Environmental, health and safety problems associated with abrassive blast cleaning limit the application of air blast cleaning in many installations. In such case water blast cleaning is resorted to.

Water blast cleaning can be applied with or without abrassive and high-pressure water blasting. The water used shall be inhibited with sodium chromate/ phosphate. The blast cleaned surface shall be washed thoroughly with detergents and wiped solvent and dried with compressed Air. For effective cleaning abrassives are used. The most commonly used pressure for high pressure water blast cleaning for maintenance surface preparation is 3000 to 6000 psi at 35-45 liters/ minute water volume and pressure upto 10000 psi and water volume of 45 liters/ minute provide maximum cleaning.

The wate blast cleaned surface shall be comparable to SSPC-SP-12/ NACE No. 5. The operation shall be carried out as per SSPC guidelines for water blast cleaning. The indicative values for sand injection is

Air : 300 to 400 Cu.ft/ min.

	STANDARD TECHNICAL	SPECIFICATION	
834002	OIL & GAS SBU,	DELHI	40 मकान 8001:2000 Confu
		DOCUMENT NO.	Page 7 of 54
TITLE	SHOP & FIELD PAINTING	MEC/S/05/21/07	REVISION: 0
			EDITION : 1
	Water : 5-10 Sand : 200- Nozzle : 0.5 t Special equipments for water bla be used.	liter/ min. with corrosion in 400 lbs/ hr. o 1" dia st cleaning with abrasive:	nhibitor s now available shall
5.2.2	Mechanical of Power tool cleaning		
	Power tool cleaning shall be done griding wheels or rotating steels w be avoided as it can reduce pa detached rust mill scale etc. shall water or stream and thoroughly d of paint.	e mechanical striking tools rire-brushes. Excessive bu int adhesion. On comple be removed by clean rag ried with compressed air	s, chipping hammers, irnish of surface shall tion of cleaning, the is and/ or washed by jet before application
5.2.3	Manual or hand tool cleaning		
	Manual or hand tool cleaning is application of other surface prepa in the specifications of paint system	s used only where safet ration procedure and hen ms.	y problems limit the ce dones not appear
	Hand tool cleaning normally consis	sts of the following:	
	<ul><li>a. Hand descaling and/ or has</li><li>b. Hand scraping</li><li>c. Hand wire brushing</li></ul>	mmering	
	Rust, mill scale spatters, old coats by hammering, scrapping tools combination of the above method shall be removed from the surfa- brushed, swept, deducted and blo all loose matter. Finally the surfa- effective cleaning.	ing and other foreign mat , emery paper cleaning s. On completion of clea ace by clean rags and own off with compressed ace may be washed with	ter, shall be removed g, wire brushing or ning, loose materials the surface shall be air/ steam to remove water and dried for
5.3	Non compatible shop coat prime	er	
	The compatibility of finishing manufacturer. In the event of use zinc silicate etc. as shop coat the coat, if shop coat is in satisfactory	coat should be confirm e of primer such as zinc pant system shall depend / condition showing no ma	ned from the paint rich epoxy, inorganic on condition of shop ajor defects, the shop

Page 491 of 644

MECON LIMITED REGD. OFF: RANCHI	STANDARD TECHNICAL			
834002	OIL & GAS SBU	मिकान मेकान 8001:2000 Contrib		
		DOCUMENT NO.	Page 8 of 54	
TITLE	SHOP & FIELD PAINTING	MEC/S/05/21/07	REVISION : 0	
			EDITION: 1	

coat shall not be removed. The touch up primer and finishing coat(s) shall be identified for application by Engineer-in-Charge.

- 5.4 Shop coated (coated with primer & finishing coat) equipment should not be repainted unless paint is damaged.
- 5.5 Shop primed equipment and surface will only be 'spot cleaned' in damaged areas by means of power tool brush cleaning and then spot primed before applying one coat of filed primer unless otherwise specified. If shop primer is not compatible with field primer then shop coated primer should be completely removed before applications of selected paints system for particular environment.
- 5.6 For packaged units/ equipment, shop primer should be as per the paint system given in this specification. However, manufacturer's standard can be followed after review.

# 5.7 **Coating Procedure and Application:**

- 5.7.1 Surface shall not be coated in rain, wind or in environment where injurious airbone elements exists, when the steel surface temperature is less than 5° F above dew point when the relative humidity is greater then 85% or when the temperature is below 40° F.
- 5.7.2 Blast cleaned surface shall be coated with one complete application of primer as soon as practicable but in no case later than 4 hrs. the same day.
- 5.7.3 To the maximum extent practicable, each coat of material shall be applied as a continuous film uniform thickness free of probes. Any spots or areas missed in application shall be recoated and permitted to dry before the next coat is applied. Applied paint should have the desired wet film thickness.
- 5.7.4 Each coat shall be proper state of cure or dryness before the application of succeeding coat. Material shall be considered dry for recoating when an additional coat can applied without the development of any detrimental film irregularities such as lifting or loose of adhesion of the under coat. Manufacturer instruction shall be followed for intercoat interval.
- 5.7.5 When the successive coat of the same colour have been specified, alternate coat shall be tinted, when practical, sufficiently to produce enough contrast to indicate complete coverage of the surface. The tinting material shall be compatible with the material and not detrimental to its service life.

MECON LIMITED	STANDARD TECHNICAL			
834002	OIL & GAS SBU,			
		DOCUMENT NO.	Page 9 of 54	
TITLE	SHOP & FIELD PAINTING	MEC/S/05/21/07	REVISION: 0	
			EDITION: 1	

# 5.7.6 Air spray application shall be in accordance with the following:

- a. The equipment used shall be suitable for the intended purpose, shall be capable of properly atomizing the paint to be applied, and shall be equipped with suitable pressure regulators and gauges. The air caps, nozzles, and needles shall be those recommended by the manufacturer of the equipment for the material beign sprayed. The equipment shall be kept in satisfactory condition to permit proper paint application.
- b. Traps or separators shall be provided to remove oil and condensed water from the air. These traps or separators must be of adequate size and must be drained periodically during operations. The air from the spray gun impinging against the surface shall show condensed water or oil.

# c. Ingredients shall be kept properly mixed in the spray pots or containers during application by continuous mechanical agitation.

- d. The pressure on the material in the pot and of the air at the gun shall be adjusted for optimum spraying effectiveness. The pressure on the material in the pot shall be adjusted when necessary for change in elevation of the gun above the pot. The atomizing air pressure at the gun shall be high enough to properly atomize the paint but not so high as to cause excessive fogging of paint, excessive evaporation of solvent, or less by overspray.
- e. Spray equipment shall be kept sufficiently clean so that dirt, dried paint, and other foreign materials are not deposited in the paint film.

Any solvents left in the equipment shall be completely removed before applying paint to the surface begin painted.

- f. Paint shall be applied in a uniform layer, with overlapping at the edge of the spray pattern. The spray patterns shall be adjusted so that the paint is deposited uniformly. During application the gun shall be held perpendicular to the surface and at a distance which will ensure that a wet layer of paint is deposited on the surface. The trigger of the gun should be released at the end of each stroke.
- g. All runs and sags shall be brushed out immediately or the paint shall be removed and the surface repainted.

MECON LIMITED	STANDARD TECHNICAL	SPECIFICATION				
834002	OIL & GAS SBU	, DELHI	मिलान मेलान बठा:2000 Control			
		DOCUMENT NO.	Page 10 of 54			
TITLE	SHOP & FIELD PAINTING	MEC/S/05/21/07	REVISION: 0			
			EDITION: 1			
	h. Areas inaccessible to the accessible by brush, daube	spray gun shall be pair ers or sheepking shall be u	nted by brush: if not used.			
	<ul> <li>All nameplates, manufac instrument glass, finished items shall be masked to p coated, the component s condition.</li> </ul>	turer's identification tags flange faces, control val prohibit coating disposition shall be cleaned and res	s, machined surface ve items and similar I. If these surface are stored to its original			
	j. Edges of structural shape first and an extra pass mad	s and irregular coated su de later.	face shall be coated			
	k. If spray gun shown chokin followed.	ng, immediately dechokin	g procedure shall be			
5.7.7	Airless spray application shall be per steel structure paint manual v relies on hydraulic pressure rather spray. An air compressor or electr pressures of 1,000 to 6.000 psi pressure through a single hose v into separate streams, which ar atomization of paint without the u with less overspray. Airless spray easier to use than conventional air	in accordance with the fol rol. 1 & vol. 2. By SSPC, I er than air atomization to ric motor is used to operat . Paint is delivered to the within the gun, a single par re forced through a sma use of air. This result in m usually is faster, cleaner, the rspray.	llowing procedure: as U.S.A., Air less spray produce the desired te a pump to produce he spray gun at this aint stream is divided Il orifice resulting in hore repaid coverage more economical and			
	Airless spray equipment is mounted on wheels, and paint is aspirated in a hot that sucks paint from any container, including drums. The unit shall have in bagitator that keep the paint uniformly mixed during the spraying. The unit should be fore spray. Incase of High Build epoxy coating (two pack), 30:1 pump rate and 0.020-0.023" tip size will provide a good spray pattern. Ideally fluid hos should no be less than 3/8" ID and not longer than 50ft to obtain optimized.					
	In case of gun choking, decoking	steps shall be followed im	mediately.			
5.7.8	Brush application of paint shall be	in accordance with the fol	lowing:			
	a. Brushes shall be of a style of paint	e and quality that will enab	ble proper application			

	STANDARD TECHNICAL	SPECIFICATION				
834002	OIL & GAS SBU	, DELHI	08 मेकान 18 मेकान 18 001:2000 Contract			
		DOCUMENT NO.	Page 11 of 54			
TITLE	SHOP & FIELD PAINTING	MEC/S/05/21/07	REVISION: 0			
			EDITION: 1			
	b. Round or oval brushes surfaces and rough or pi large flat areas, but they sl	are most suitable for ri tted steel. Wide flat brus hall not have width over fiv	vets, bolts, irregular shes are suitable for ve inches.			
	c. Paints shall be applied into	all corners.				
	d. Any runs or sags shall be l	orushed out.				
	e. There shall be minimum of	<sup>f</sup> brush marks left in the ap	oplied paint			
	f. Surface not accessible to sheepkin.	brushes shall be painted	by spray, duubers, or			
5.7.9	Manual application by sling (w approachable)	/here 6 O' clock posi	tion of pipe is not			
	A canvas strip (alternatively a tinplate strip) about 450mm wide and 1.5m longs is hold under the pipe by two men. Liquid coating poured on the sling at each side of the pipe. The men holding this sling move it up and down and walk slowly forward while fresh coating is poured on the pipe and they manipulate the sling so that an even coating is ontained all round the bottom. This work shall be done vey carefully and by experienced personnel. There shall bot be any formation of "Whiskers" and holes in the coating. The coating film shall be inspected by mirror.					
5.7.10	For each coat the painter should DFT and standardise the paint WFT. This is to be ensured in the	know the WFT correspor application technique to qualification trial.	nding to the specified achieve the desired			
5.8	Drying of Coated Surface					
5.8.1	No coat shall be applied unit the considered dry for re-coating wh development of any film irregula undercoats. Drying time of the ap for it as a first coat; if it exceeds mixing is faulty.	preceding coat has dried. Then another coat can be arities such as lifting or plied coat should not exce the paint material has po	The material shall be applied without the loss of adhesion of d maximum specified ssible deteriorated or			
5.8.2	No paint shall be force dried unde blistering formation of pores, or de	er condition which will caus etrimentally after the condi	se checking, wrinkling tion of the paint.			

MECON LIMITED	STANDARD TECHNICAL		
834002	OIL & GAS SBU,	DELHI	मित्र मेकान मेकान 9001:2000 Contract
		DOCUMENT NO.	Page 12 of 54
TITLE	SHOP & FIELD PAINTING	MEC/S/05/21/07	<b>REVISION: 0</b>
			EDITION: 1

No drier shall be added to a paint on the job unless specifically called for in the manufacturer's specification for the paint.

Paint shall be protected from rain, condensation, contamination snow and freezing until dry to the fullest extent practicable.

# 5.9 **Repair of damaged paint surface.**

- 5.9.1. Where paint has been damaged in handling and in transportation, the repair of damaged coating of pre-creation/ fabrication shall be as given below.
- 5.9.2. Repair of damaged inorganic zinc silicate primer after erection/ weldding:

Quickly remove the primer from damaged area by mechanical scraping and emery paper to expose the white metal. Blasts clean the surfaces possible. Feather the primer over the intact adjacent surface surrounding the damaged area by emery paper.

- 5.9.3 Repair of damaged pre-erection and shop priming in the design temperature of 90° C to 500° C.
  - Surface preparation shall be done as per procedure 5.9.2
  - One coat of F-9 shall be applied wherever damaged was observed on pre-erection/ pre-fabrication/ shop primer of inorganic zinc silicate coating (F-9) shall not be applied if damaged area is not more than 5 x 5 cm.

# 5.10 **PAINT APPLICATION**

- 5.10.1 Shop priming/ pre-erection priming with F9 of F12 shall be done only on blasted surface.
- 5.10.2 Shop priming/ pre-erection priming with F-9 or F-12 shall be done only with airless spray.
- 5.10.3 For large flat surface field painting shall be done by airless spray otherwise brush can be used.

# 5.11 Assessment of Painting Requirement

The paint system to be applied for a specific job shall be arrived as sequentially as given below :

MECON LIMITED REGD. OFF: RANCHI	STANDARD TECHNICAL			
834002	OIL & GAS SBU	00 मेकान 12000 Contract		
		DOCUMENT NO.	Page 13 of 54	
TITLE	SHOP & FIELD PAINTING	MEC/S/05/21/07	REVISION: 0	
			EDITION : 1	

- Identify the environment from area classification details and chose the appropriate table.
- Identify the design temperature from the technical documents.
- Identify the specific field paint system and surface preparation requirement from the above identified table and temperature range.
- Identify the shop priming requirement from Table 7.1 based on compatibility of the above paint system.
- Identify the need of repair of shop primer and execute as per Table 7.2.

# 5.12 **Documentation.**

A written quality plan with procedure for qualification trials and for the actual work.

Daily progress report with dedtails of weather condition, particular of application no of coats and type of materials applied, anomolies, progress of work versus programme.

Result of measurement of temperature relative humidity, surface profile, film thickness, holiday detection, adhesion tests with signature of appropriate authority.

Particular of surface preparation and paint application during trials and during the work.

Details of non-compliance, rejects and repairs.

Type of testing equipments and calibration.

Code and batch numbers of paint material used.

								PLIED	JE IS 1AL 1DITION 4ODS	ED AND EANING					
						REMAR		THIS METHOD IS AF	WHEN THE SURFAC EXPOSED TO NORN ATMOSPHERIC CON WHEN OTHER METI	CANNOT BE ADOPT ALSO FOR SPOT CL	PAINTING.		-00-		
	- D.	4	6		Ţ	ARDS	BRITISH STANDARD BS-4232: 1967		ı				ı		
	10 Hant	Page 14 of 5.	REVISION : (	EDITION: 1	0) NNDARDS	IONAL STAND ALENT)	NACÉ USA		1				ı		
		NO.	1/07		r clause 5. ATION ST/	IS INTERNAT (EQUIV/	SSPC-SP USA		SSPC-SP-	V			SSPC-SP-	ςΩ	
ECIFICATION	ELHI	DOCUMENT	MEC/S/05/2		TABLE-I (fo ACE PREPAR	VARIOU	SWEDISH STANDARD SIS-05-5900 1967		ST.2				ST.3		
STANDARD TECHNICAL SF	OIL & GAS SBU, D		SHOP & FIELD PAINTING		SURF	DESCRIPTION		OR HAND TOOL CLEANING	AL OF LOOSE RUST LOOSE MILL ND LOOSE PAINT, CHIPPING, IG, SANDING AND WIRE	IG, SURFACE SHOULD HAVE A ETALLIC SHEEN.		VICAL OR POWER TOOL CLEANING AL OF LOOSE RUST, LOOSE MILL ND LOOSE PAINT TO DEGREE	ED BY POWER TOOL CHIPPING,	ING, SANDING, WIRE BRUSHING NDIN, AFTER REMOVAL OF DUST,	E SHOULD HAVE A PRONOUNCED
LIMITED FF: RANCHI			ШЕ					MANUAL	REMOV/ SCALE A SCRAPIN	BRUSHI FAINT M		MECHAN REMOVA	SPECIFIL	DESCAL AND GRI	SURFAC
MECON I	834002		F			S N		-				5			

						REMARKS			WHERE EXTREMELY CLEAN SLIRFACE CAN RE EXPECTED	FOR PROLONG LIFE OF PAINT SYSTEMS.	THE MINIMUM REQUIREMENT	PAINT SYSTEM SUCH AS EPOXY, VINYL, POLYURETHANE BASED AND INORGANIC ZINC SILICATE PAINTS, ALSO FOR CONVENTIONAL PAINT SYSTEM USED UNDER FAIRLY CORROSIVE CONDITIONS TO OBTAIN DESIRED LIFE OF PAINT SYSTEM.
	A A A A A A A A A A A A A A A A A A A	54	0	1		DARDS	BRITISH STANDARD BS-4232: 1967		FIRST OLIALITY		SECOND	
	4001-2000 Control	Page 15 of 5	<b>REVISION:</b>	EDITION :	) NDARDS	IONAL STAN ALENT)	NACE USA		NACE#1		NACE #2	
		0.	/07		∵clause 5.0 \TION STA	S INTERNAT (EQUIV)	SSPC-SP USA		SSPC-SP- 5	)	SSPC-SP- 10	2
ECIFICATION	ELHI	DOCUMENT N	MEC/S/05/21		TABLE-I (for CE PREPAR/	VARIOU	SWEDISH STANDARD SIS-05-5900 1967		SA-3		SA 2 ½	
STANDARD TECHNICAL SP	OIL & GAS SBU, D		SHOP & FIELD PAINTING		SURFA	DESCRIPTION		:LEANING (AIR & WATER) ARE FOUR COMMON GRADES OF :LEANING	AETAL	LEANING TO WHITE METAL INESS REMOVAL OF ALL VISIBLE IILL SCALE PAINT & FOREIGN 100% CLEANLINESS WITH 2 SURFACE PROFILE.	HITE METAL	LEANING TO NEAR WHITE METAL INESS, UNIT AT LEAST 95% OF LEMENTS OF SURFACE AREA IS = ALL VISIBLE RESIDUES WITH D SURFACE PROFILE.
MITED			Е					BLAST C THERE ≠ BLAST C	WHITE N	BLAST C CLEANLI CLEANLI RUST, M MATTER DESIREC	NEAR W	BLAST C CLEANLI EACH EL FREE OF DESIREI
MECON LI REGD. OFF:	834002					S. ON.		ю́.	3.1		3.2	

MECON LIMITED REGD. OFF: RANCHI	STANDARD TECHNICAL	SPECIFICATION	
834002	OIL & GAS SBU	, DELHI	मिकान मेकान 8001:2000 Comp
		DOCUMENT NO.	Page 17 of 54
TITLE	SHOP & FIELD PAINTING	MEC/S/05/21/07	REVISION: 0
			EDITION: 1

# 6.0 **PAINT MATERIALS**

Paint manufacturers shall furnish all the characteristics of paint material on printed literature, alongwith the test certificate for all the specified characteristics given in this specifications. All the paint materials shall be of first quality and conform to the following general characteristics as per the table 6.1, 6.2 and 6.3.

MECON LIMITED	STANDARD TECHNICAL S	SPECIFICATION	
834002	OIL & GAS SBU,	DELHI	140FT
		DOCUMENT NO.	Page 18 of 54
TITLE	SHOP & FIELD PAINTING	MEC/S/05/21/07	REVISION: 0
			EDITION: 1

# PAINT MATERIALS TABLE NO.: 6.1 PRIMERS

	al Habit	Page 19 of 54	REVISION: 0	EDITION : 1	AINT	F-3 F-6 F-7	ed rubber Epoxy-High build High build coaltar sh paint finish paint epoxy coating.	ck plasticised Tow- pack Tow pack polyamide d rubber polyamide/ cured epoxy resin	dium with ployamine cured blended with coal/ tar	and weather epoxy resin medium, suitably	pigments. medium suitable pigmented. pigmented.	62% 65%	100-125μ 100-125μ	5-6 5-2-6.5	1.4 1.5	is 3 hrs. 4 hrs.	Overnight 48 hrs.	rnight Min.: Overnight Min.: 24 hrs. imited Max · 5 dav Max · 5 dav		cable 4-6 hrs. 4-6 hrs.	80°C 125°C
		NO.	1/07		ATERIALS 2FINISH P		Chlorinate based fini	Single pac	based me	chemical :	resistant	40%	40-50μ	8-10	1.2	30 minute	8 hrs.	Min.: Ove Max · Unli	)	Not applic	60°C
CAL SPECIFICATION	SBU, DELHI	DOCUMENT	MEC/S/05/2		PAINT M/ TABLE NO.: 6.	F-2	Acrylic polyurethane finish paint	Two-pack aliphatic isocynate cured	acrylic finish paint			40%	30-40µ	10-13	1.3	1 hrs.	Overnight	Min.: Overnight (12) hrs	Max.: Unlimited	6-8 hrs.	80°C
STANDARD TECHNIC	OIL & GAS (		SHOP & FIELD PAINTING			DESCRIPTION	cal Name	nd composition				solids (approx)	ry film thickness) per coat	tical covering capacity in at/ litre (approx)	per litre in kgs/ litre	dry at 30° C (approx)	y at 30° C (approx)	ating interval (approx)		at 30° C for two component approx).	rature Resistance
MITED			щ				Technic	Type an				Volume	DFT (Dr (approx)	Theoret M2/ coa	Welght (approx)	Touch d	Hard dry	Overcoa		Pot life ; paints (a	Temper
MECON LI REGD. OFF:	834002		TTTL			S. S	<del>.</del> .	2.				ю.	4.	5.	.9	7.	œ.	0		10.	11.

Page 20 of 54 **REVISION: 0** EDITION: 1 MEC/S/05/21/07 DOCUMENT NO. STANDARD TECHNICAL SPECIFICATION **OIL & GAS SBU, DELHI** SHOP & FIELD PAINTING MECON LIMITED REGD. OFF: RANCHI 834002 TITLE

# PAINT MATERIALS TABLE NO.: 6.3 FINISH PAINTS

s, ∾	DESCRIPTION	F-8	6-L	F-11	F-12				
<del>.</del>	Technical Name	Self priming type surface tolerant high build epoxy coating (Complete rust control coating)	Inorganic Zinc Slicate coating	Heat resistant synthetic medium based two pack aluminum paint suitable upto 250°C dry temperature	Heat resistant silicone aluminum paint suitable upto 500° C temperature				
N	Type and composition	Two-pack epoxy resin based suitable pigmented and capable pigmented and capable of adhering to manually prepared surface and old coating	A two-pack air drying self-curing solvent based inorganic zinc silicate coating.	Heat resistant synthetic medium based two pack aluminum paint suitable upto 250°C	Single pack silicone resin based medium with aluminum flakes.				
ю.	Volume solids (approx)	72%	60%	25%	20%				
4.	DFT (Dry film thickness) per coat (approx)	100-125μ	65-75μ	20-25μ	20-25μ				
5.	Theoretical covering capacity in M2/ coat/ litre	6.0-7.2	8-9	10-12	8-10				
6.	Welght per litre in kgs/ litre (approx)	1.4	2.3	1.2					
7.	Touch dry at 30° C (approx)	3 hrs.	30 min.	3 hrs.	30 min.				
œ.	Hard dry at 30° C (approx)	24 hrs.	12 hrs.	12 hrs.	24 hrs.				
9.	Overcoating interval (approx)	Min.: 10 hrs. Max.: 6 months	Min.: 8 hrs. at 20°C and 50% RH. Max.: Unlimited	Min.: 16 hrs. Max.: Unlimited	Min.: 16 hrs. Max.: Unlimited				
					-				
---------	----------------------	---------------------	----------------	---------------	-------------	------------	-----------------	----------------	--
MECON I	LIMITED F: RANCHI	STANDAR	D TECHNICAL S	SPECIFICATION			0		
834002		0	JIL & GAS SBU,	DELHI		CO BADI-	1		
				DOCUMENT N	ĮO.	Page 21 of	<sup>5</sup> 54		
F	ITLE	SHOP & FIELD F	PAINTING	MEC/S/05/21	1/07	REVISION	0 :		
						EDITION :	: 1		
S.	DE	ESCRIPTION		8°-	6- <b>1</b>		F-11	F-12	
Ň									
10.	. Pot life (al	pprox) at 30° C for	90 min.		4-6 hrs.		Not applicable	Not applicable	
	two comp	onent paints							
	(approx).								

F-14: Specially for mulated polyamine cured coal tal epoxy suitable for-45°C to 125°C for application under insulation F-15: Two pack cold curved epoxy phenolic coating suitable for 45°C to 125°C for application under insulation F-16: Eoxy siloxane anser coat 738

500° C

250° C

400°C

80°C

Temperature resistance

1.

MECON LIMITED REGD. OFF: RANCHI	STANDARD TECHNICAL	SPECIFICATION		
834002	OIL & GAS SBU	, DELHI	00 मेकान 1001:2000 Contrain	
		Page 22 of 54		
TITLE	SHOP & FIELD PAINTING	MEC/S/05/21/07	REVISION: 0	
			EDITION: 1	

### PAINT MATERIALS TABLE NO. 6.4 FINISH PAINTS

SI.	Description	F-14	F-15	F-16	F-17
<b>No.</b> 1.	Technical name	Polyamine cured coal tar epoxy	Two-component Epoxy phenolic coating cured with polyamine adduct hardner system (primer + intermediate coat + finish paint)	Ambient temperature curing Poly Siloxane coating / High build cold applied inorganic copolymer based aluminum coating suitable for under insulation coating of CS and SS piping for high temperature service.	Two component solvent free type high build epoxy phenolic / novalac epoxy phenolic coating cured with Polyamine adduct hardner system.
2.	Type & composition	Specially formulated polyamine cured coal tar epoxy suitable for application under insulation	Two pack ambient temperature curing epoxy phenolic coating system suitable for application under insulation of CS / SS piping.	Amercoat 738 from Ameron Products, USA / Berger 938 from Berger Paints Ltd., or Intertherm 751 CSA from Akzo Nobel coating. Note : 6	Two component solvent free type high build epoxy phenolic / novalac epoxy phenolic coating cured with Polyamine adduct hardner system.
3.	Volume Solids (minimum)	70%	65%	60%	98-100%
4.	DFT (Dry Film thickness) per coat (minimum)	125 μm	75 - 100 μm	75 - 100 μm	125- 150 μm
5.	Theoretical covering capacity in M <sup>2</sup> / coat / litre (minimum)	5.5	6.5-8.5	6.0-8.0	6.5-8.0
6.	Weight per liter in kgs/litre (max paint) (minimum)	1.5	1.7	1.3	1.7
7.	Touch dry at 30°C (maximum)	4 hrs.	2 hrs.	1 hr.	2 hrs.

MECON LIMI	TED NCHI	ST	ANDARD TECHNI	CALS	SPECIFICATI	ON		N N	
834002			OIL & GAS	SBU,	DELHI		मुकान मुकान 1001:2000 Confi	¢	
					DOCUM	ENT NO.	Page 23 of 54	4	
TITLE		SHOP 8	FIELD PAINTING	ì	MEC/S/C	5/21/07	<b>REVISION: 0</b>		
					EDITIO				
SI.	Desc	cription	F-14		F-15	F-16	F-17		

SI.	Description	F-14	F-15	F-16	F-17
No.					
8.	Hard dry at 30°C (maximum) Full cure	24 hrs.	24 hrs.	16 hrs.	24 hrs.
	30°C (for immersion / high temp. service)	168 hrs.(7 days)	168 hrs.(7 days)	-	168 hrs.(7 days)
9.	Over-coating interval	Min. 6 hrs.	Min. 36 hrs.	Min. 16 hrs.	Min. 16 hrs.
		Max. 5 days	Max. 21 days	Max. Not	Max. 21 days.
				applicable	
10.	Pot life at 30°C for	4 hrs.	1.5 hrs.	1 hr.	1 hr.
	two component				
	paints (minimum)				
11.	Temperature	-45°C to 125°C	-45°C to 125°C	Up to 400°C for	-45°C to 150°C
	Resistance (min.)	under insulation	under insulation	CS & SS under	for immersion
			(Note : 5)	insulation	service

### Notes:

- 1. Covering capacity and DFT depends on method of application. Covering capacity specified above are theoretical. Allowing the loose during the application, minimum specified DFT should be maintained.
- 2. All primers and finish coats should be cold cured and air drying unless otherwise specified.
- 3. All paints shall be applied in accordance with manufacturer's instruction for surface preparation, intervals, curing and application. The surface preparation, quality and workmanship should be ensured.
- 4. Technical data sheets for all paints shall be supplied at the time of submission of quotations.

### 6.4 List of recommended manufacturers

The paint shall conform to the specifications given above and the best quality in their products range of the manufacturers listed in Annexure-I.

### 7.0 PAINT SYSTEM

The paint system should vary with type of environment envisaged in and around the plants. Three types of environment as given below are considered for selection of paint system. The paint system is also given for specific requirements.

MECON LIMITED	STANDARD	D TECHNICAL	SPECIFICATION				
834002	0]	IL & GAS SBU,	DELHI	सु मेकान 9907:2000 Control			
			DOCUMENT NO.	Page 24 of 54			
TITLE	SHOP & FIELD P	AINTING	AL SPECIFICATION           SBU, DELHI         Image 24 of 54           DOCUMENT NO.         Page 24 of 54           MEC/S/05/21/07         REVISION : 0           EDITION : 1         EDITION : 1	REVISION: 0			
		OIL & GAS SBU, DELHI         DOCUMENT NO.         Page 24 of REVISION           LD PAINTING         MEC/S/05/21/07         REVISION (EDITION :           LD PAINTING         SUBLE AND AND AND AND AND AND AND AND AND AND					
Primer	s & finish coats co	overed in tal	ble nos. 7.0 to 15.0				
PRIME	RS						
P-2	:	Chlorinated	Rubber Zinc Phosphat	e Primer			
P-4	:	Etch Prime	r/ Wash Primer				
P-6	:	Epoxy Zic F	Phosphate Primer				
FINISH	COATS/ PAINTS						
F-2	:	Acrylic- Pol	yurethane finish paint				
F-3	:	Chlorinated	l Rubber Finish Paint				
F-6	:	High Build	Epoxy finish coating				
F-7	:	High Build	Coal Tar epoxy coating				
F-8	:	ouild epoxy					
F-9	:	Inorganic Z	inc Silicate Coating.				
F-11	:	Heat resist Aluminum p	ant Synthetic medium b paint.	based			
F-12	:	Heat resista	ant Silicone Aluminum p	paint.			
F-14	:	Specially fo Epoxy coat	ormulated polyamine-cu ing	red coal for			
F-15 F-16 F-17	:	Epoxy pher Epoxy Silox Two comp phenolic / r polyamine.	nolic coating kane Coating : Amercoa onent solvent free typ novalac epoxy phenolic	t 738 e high built epoxy coating cured with			

MECON LIMITED	STANDARD TECHNICAL	SPECIFICATION	(
834002	OIL & GAS SBU	, DELHI	to Habri
		DOCUMENT NO.	Page 25 of 54
TITLE	SHOP & FIELD PAINTING	MEC/S/05/21/07	REVISION: 0
			EDITION: 1

## TABLE 7.1: PRE-ERECTION/ PRE-FABRICATION AND SHOP PRIMING FOR CARBON STEEL, LOW TEMPERATURE CARBON STEEL & LOW ALLOY STEEL, STEEL STRUCTURE, PIPING AND EQUIPMENT ETC.

			DAINT CVCTTM		
о. No.	DESIGN IEMPERATURE IN °C	SURFACE PREPARATION	PAINI SYSLEM	NICRONS (MIN.)	KEMAKKS
7.1.1	-90 TO 400	SSPC-SP-10	1 COAT OF F-9	65-75	No overcoating is to
					be done
7.1.2	401 To 500	SSPC-SP-10	1 COAT OF F-12	40-50	Finish Coat at Site
7.1.3	-40 to 150 for Structures, hand rails and	SSPC-SP-3	1 COAT OF F-9 OR	65-75 OF F-9	For Damaged Area
	Grating only		2 COATS OF P-7 @	OR	of more than 5 x 5
			$40\mu$ DFT / COAT	80 (P-7)	Cm2.

## REPAIR OF PRE-ERECTION/ PRE- FABRICATION AND SHOP PRIMING AFTER ERECTION/ WELDING FOR CARBON STEEL LOW TEMPERATURE CARBON STEEL & LOW ALLOY STEEL, ITEMS IN ALL ENVIRONMENT. TABLE 7.2:

S. No.	DESIGN TEMPERATURE IN	SURFACE PREPARATION	PAINT SYSTEM	TOTAL DFT IN	REMARKS
	°			<b>MICRONS (MIN.)</b>	
7.2.1	-90 TO 400	SSPC-SP-3 (FOR REPAIR	1 COAT OF F-9	65-75	FOR DAMAGED
		ONLY)			AREA OF MORE
		SSPC-SP-10			THAN 5X5 CM.
7.2.2	401 TO 550	SSPC-SP-3	1 COAT OF F-12	20	FOR DAMAGED
					AREA OF MORE
					THAN 5X5 CM.

					LOW TEMPERATURE	TABLE 8.0: FIELD PAINT SYSTEM FOR NORMAL CORROSIVE ENVIRONMENT (FOR CARBON STEEL, LOW TEMPERATURE CARBON STEEL & LOW ALLOY STEEL)         CARBON STEEL & LOW ALLOY STEEL)         ALL NORMAL CORROSIVE AREAS SUCH AS OFF SITES EXTERNAL SURFACE OF UNINSULATED COLUMNS, VESSELS, HEAT EXCHANGERS, BLOWERS, PIPING, PUMPS, TOWERS, COMPRESSORS, STRUCTURAL STEEL WORKS, RCC CHIMNEY WITH OR WITHOUT REFRACTORY LINE INSIDE CHIMNEY (ALL ENVIRONMENTS), EXCLUDING TANK TOPS, FLARE LINES, D.M. PLANTS, INTERIOR OF TANKS ETC. FLARE LINES FOR NORMAL CORROSIVE ENVIRONMENT ALSO TO NE PAINTED AS PER TABLE 9.0         SUFFACE         SIGN         SIGN		No over coating to be done follow repair procedure only on damaged areas of pre- erection/ pre-fabrication primer/ coating F-9			
					ON STEEL,		DFT IN MICRONS (MIN.)	65-75	225	245	
	and the second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second s	Page 26 of 54	REVISION: 0	EDITION: 1	NMENT (FOR CARB	RFACE OF UNINSUL <sup>,</sup> , STRUCTURAL STE :NTS), EXCLUDING DSIVE ENVIRONMEN		FINISH PAINT	NONE	2 COATS OF F-3 @ 40 μ DFT/ COAT 2 X 40 = 80	1 COATS OF F-6 @ 100 μ DFT/ COAT
PECIFICATION	DELHI	DOCUMENT NO.	MEC/S/05/21/07		MAL CORROSIVE ENVIRC STEEL)	JCH AS OFF SITES EXTERNAL SURFA PUMPS, TOWERS, COMPRESSORS, ST NSIDE CHIMNEY (ALL ENVIRONMENT: FLARE LINES FOR NORMAL CORROSIV	PAINT SYSTEM	FIELD PRIMER	OF PRE-FABRICATION 1 COAT OF F-9 @65-75μ .Τ	OF PRE-FABRICATION 1 COAT OF F-9 @ 65-75μ .T + 2 COATS OF P-2 @ 40μ .T 0	OF PRE-FABRICATION 1 COAT OF F-9 @ 65-75μ 1 + 2 COATS OF P-6 @ 40μ 4T 0
HNICAL S	GAS SBU, I		<b>DNI</b>		STEM FOR NORMAL COR & LOW ALLOY STEEL) EAS SUCH AS OFF SITES PING, PUMPS, TOWERS, C LINE INSIDE CHIMNEY (A ETC. FLARE LINES FOR I			REPAIR PRIMER DFT/ COA	REPAIR PRIMER DFT/ COA DFT/ COA DFT/ COA	REPAIR PRIMER DFT/ COA DFT/ CO/ 2 X 40 = 8	
STANDARD TEC	OIL &		P & FIELD PAIN		AINT SYSTEM I I STEEL & LOM	SIVE AREAS SL ERS, PIPING, F CTORY LINE II F TANKS ETC. I	SURFACE	PREPARATIO N	SSPC-SP-10	SSPC-SP-10	SSPC-SP-10
<b>D</b> IHC			SHOI		.0: FIELD P, CARBON	(MAL CORRO GERS, BLOW HOUT REFRA INTERIOR OF	DESIGN	EMPERATUR E IN C	0 TO -15	4 TO 60	TO 80
SCON LIMITE	4002		TITLE		TABLE 8	CARBON ALL NORMAL CORROS EXCHANGERS, BLOWE OR WITHOUT REFRAC PLANTS, INTERIOR OF TABLE 9.0	Ś	L NO.	8.1 -9	8.2	8.3 61

					REMARKS					ED. BBTAIN 2 X 100=200µ IN 1.5 BE DELETED.
					TOTAL	DFT IN MICRONS (MIN.)	125	105	80	BE FOLOW COAT TO OI SUIDELINES O. 8.2 WILL
	abit 2000 County	Page 27 of 54	REVISION: 0	EDITION: 1		FINISH PAINT	3 COATS OF F-11 @ 20 μ DFT/ COAT 3 X 20 = 60	2 COATS OF F-12 @ 20 μ DFT/ COAT 2 X 20 = 40	2 COATS OF F-12 @ 20 μ DFT/ COAT 2 X 20 = 40	3, 8.4 AND 8.5 SHALL F F-6 @ 100 µ DFT/ ( PARATION AS PER G °C TO 80°C AND S.N
SPECIFICATION	DELHI	DOCUMENT NO.	MEC/S/05/21/07		PAINT SYSTEM	FIELD PRIMER	OF PRE-FABRICATION 1 COAT OF F-9 @ 65-75µ AT	OF PRE-FABRICATION 1 COAT OF F-9 @ 65-75µ AT	AS PER 7.2.2	T REFRACTORY LINING 8.3 CC CHMNEY: 2 COATS OF ER MAKING SURFACE PREI .3 SHALL BE USED FOR 14 .3 SHALL BE USED FOR 14
CHNICAL S	GAS SBU,		TING				REPAIR PRIMER DFT/ CO	REPAIR PRIMER DFT/ CO	REPAIR /	WITHOUT CE OF R( IED AFTE D S.NO. 8
STANDARD TE(	OIL &		> & FIELD PAIN		SURFACE	PREPARATIO N	SSPC-SP-10	SSPC-SP-10	SSPC-SP-10	CHIMNEY OR TENAL SURFA HALL BE APPL /ER REQUIREI
			SHO		SIGN	ERATUR IN C	250	0 400	0 500	FOR MS FOR EX <sup>-</sup> SI WHEREV
AITED RANCHI			ш		DE	TEMP	81 TO	251 TC	401 TC	
MECON LIM REGD. OFF: F	834002		TTTLE		Ś	ÖN	8.4	8.5	8.6	NOTE NOTE NOTE

					OW TEMPERATURE	ACID/ ALKALI/ SALT ULATED COLUMNS, ≷S, FLARE LINES,	REMARKS		Repair of pre-erection/ pre fabrication primer shall be done wherever damage is observed.	Surface preparation is required only for repairing of damaged pre-erection/	fabrication primer		
					NN STEEL, L(	LAGE'S OF , E OF UNINS OMPRESSOF	TOTAL DFT	IN MICRONS (MIN.)	65-75	225		105	80
	and the second	Page 28 of 54	REVISION: 0	EDITION: 1	INT (FOR CARBC	FUMES OR SPII TERNAL SURFAC 3, TOWERS, C(		FINISH PAINT	NONE	1 COATS OF F-6 @ 100μ DFT/ COAT	+ 1 COAT OF F-2 @ 40μ DFT/ COAT	2 COATS OF F-12 @ 20 μ DFT/ COAT 2 X 20 = 40	2 COATS OF F-12 @ 20 μ DFT/ COAT
SPECIFICATION	DELHI	DOCUMENT NO.	MEC/S/05/21/07		SORROSIVE ENVIRONME DY STEEL)	KOUND WHERE H <sub>z</sub> S, SO <sub>2</sub> I SURFACE SUCH AS EX NERS, PIPING, PUMPS	PAINT SYSTEM	FIELD PRIMER	R OF PRE-FABRICATION R 1 COAT OF F-9 @ 65-75μ :OAT	R OF PRE-FABRICATION R 1 COAT OF F-9 @ 65-75μ COAT + 1 COATS OF P-6	DFT/ COAT	R OF PRE-FABRICATION R 1 COAT OF F-9 @ 65-75μ :OAT	R 2S PER 7.2.2
STANDARD TECHNICAL	OIL & GAS SBU,		8. FIELD PAINTING		AINT SYSTEM FOR ( V STEEL & LOW ALL	e areas above gr ie in contact with :xchangers, blov - etc.	SURFACE	PREPARATION	SSPC-SP-10 REPAI PRIME DFT/ C	SSPC-SP-10 REPAI PRIME DFT/ (	@40 µ	SSPC-SP-10 REPAI PRIME DFT/ C	SSPC-SP-10 REPAI
TED			SHOP		E 9.0: FIELD P. CARBON	ALL CORROSIV IKELY TO COM ELS, HEAT E CTURAL STEEL	DESIGN	TEMPERATUR E IN °C	-90 TO –15	-14 TO 80		81 TO 400	401 TO 500
AECON LIMI	834002		TITLE		TABLE	FOR	S.	NO.	9.1	9.2		9.3	9.4

					LOW TEMPERATURE SULATED COLUMNS, MPRESSORS, FLARE	ELC.	REMARKS		Repair of pre-erection/	followed. No over coating is	Surface according in	required only for repairing	of damaged pre-erection/	fabrication primer.						
					STEEL, I OF UNINS /ERS, CON	CHLOKIDE	TOTAL	DFT IN MICRONS (MIN.)	65-75		21E	040				105		80	2	
	TO HADY	Page 29 of 54	REVISION: 0	EDITION: 1	(FOR CARBON AL SURFACES G PUMPS, TOW	IMPINGEMENI,		FINISH PAINT	NONE				$COAT = 2 \times 100=$	200 + 1 COAT OF	F-2 @ 40μ DF1/ COAT	2 COATS OF F-12	COAT ULLI	2 A 20 - 40 3 COATS OF E-19	@ 20 μ DFT/ COAT	2 X 20 = 40
L SPECIFICATION	U, DELHI	DOCUMENT NO.	MEC/S/05/21/07		DR HIGHLY CORROSIVE ALLOY STEEL) EXTERN GERS, BLOWERS, PIPIN ETC.	HCL H <sub>2</sub> SU <sub>4,</sub> SALIY WAIEK	PAINT SYSTEM	FIELD PRIMER	AIR OF PRE-FABRICATION	AER 1 COAT OF F-9 65-75µ / COAT		AIN OF FRE-FADRICATION	COAT + 1 COATS OF P-6	μ DFT/ COAT		AIR OF PRE-FABRICATION	/IEK 1 CUAT UF F-9 @ 63-73µ / COAT	AIR AS PER 7 2 2		
STANDARD TECHNICA	OIL & GAS SB		& FIELD PAINTING		PAINT SYSTEM FON STEEL & LOW S. HEAT EXCHAN STRUCTURE STEEL	AGE OR FUMES OF	SURFACE	PREPARATION	SSPC-SP-10 REP	DFT			DFT	@40		SSPC-SP-10 REP		SCDC_CD_10 RED		
ED			SHOF		10.0: FIELD F CARBOI VESSEL LINES, S		DESIGN	FEMPERATUR E IN °C	30 TO –15			14 1 0 00				1 TO 400		01 TO 500		
MECON LIMITI REGD. OFF: RAN	834002		TTTLE		TABLE	EXPOSI	Ś	NO.	10.1 -6		C C T					10.3 8		10 1	r 2	

					ERNAL) FOR ALL	N REMARKS	V.)	LUDING TOP SIDE OF STRUCTURAL WORK	ENTS FOR GRUDE OIL, BLE WATER, ACIDS,	F-6 should be	suitable for	occasional water	Immersion						F7 should be	suitable for	immersion service	of the products given.
					ranks (exti	TOTAL DFT II	MICRONS (MIP	JND TANK INCL	ALL ENVIRONM WATER, POTA	285					105				365			
	to Harring	Page 30 of 54	REVISION: 0	EDITION: 1	EL STORAGE 1	SYSTEM	FINISH PAINT	OPS OF ALL GROU	S, HAND TAILS FOR IREWATER, RAW	1 COATS OF F-6	@ 100μ DFT/	COAT + 2 COATS	DFT/ COAT	2 X 40 = 80	2 COATS OF F-12	@ 20 μ DFT/	CUAI 2 X 20 = 40	STORAGE TANKS.	3 COATS OF F - 7	@ 100 μ DFT/	COAT	3 X 100 = 300
CATION		CUMENT NO.	/S/05/21/07		CARBON STE	PAINT S	FILED PRIMER	RATUSES, ROOF T	SPIRAL STAIRWAY 1 WATER, F	1 COAT OF F-9 @	65-75μ DFT/	COAT + 1 COATS		65 X 40 = 105	1 COAT OF F-9 @	65-75μ DFT/	COAL	SOIL SIDE) FOR ALL	1 COAT OF F-9 @	65-75μ DFT	COAT	
<b>RD TECHNICAL SPECIFI</b>	OIL & GAS SBU, DELHI		DAINTING MEC		AINT SYSTEM FOR IVIRONMENTS.	SURFACE	PREPARATION	WIND GIRDERS APPA OPEN TANK AS WEL	DERS, DLINE, MOTOR SPIRIT, DN CALS ETC.	SSPC-SP-10					SSPC-SP-10			CE OF BOTTOM PLATE (	SSPC-SP-10			
	4		SHOP & FIELD		.0 : FIELD P	DESIGN	remperature IN °C	EXTERNAL SHELL ROOF OF	ND STATIONARY LADD ATF KEROSENE, GASC JLVENTS AND CHEMIC	TO 80					TO 500			EXTERNAL SURFA	TO 80			
MECON LIMITED REGD. OFF: RANCH	834002		TITLE		TABLE 11	S. NO.	•	1.1 FLOATING F	ROLLING AI LDO, HSD, / ALKALIS SC	11.1.1 -14					11.1.2 81			11.2	11.2 -14			

B34002     OIL & GAS SBU, DELHI       B34002     DIL & GAS SBU, DELHI       TITLE     BOCUMENT NO.       TITLE     SHOP & FIELD PAINTING       MEC/S/05/21/07     REVISION :       EDITION :     EDITION :	MECON LIMITED	STANDARD TECHNICAL	SPECIFICATION	
TITLE     BOCUMENT NO.     Page 31 of .       TITLE     SHOP & FIELD PAINTING     MEC/S/05/21/07     REVISION :       EDITION :     EDITION :	834002	OIL & GAS SBU,	, DELHI	and start
TITLE     SHOP & FIELD PAINTING     MEC/S/05/21/07     REVISION :       EDITION :     EDITION :			DOCUMENT NO.	Page 31 of 54
EDITION :	TITLE	SHOP & FIELD PAINTING	MEC/S/05/21/07	REVISION: 0
				EDITION: 1

# TABLE 12.0 : FIELD PAINT SYSTEM FOR CARBON STEEL AND ALLOY STORAGE TANK: (INTERNAL)

• C         • C         • C         • C         • C         • C         • C         • C         • C         • C         • C         • C         • C         • C         • C         • C         • C         • C         • C         • C         • C         • C         • C         • C         • C         • C         • C         • C         • C         • C         • C         • C         • C         • C         • C         • C         • C         · C         · C         · C         · C         · C         · C         · C         · C         · C         · C         · C         · C         · C         · C         · C         · C         · C         · C         · C         · C         · C         · C         · C         · C         · C         · C         · C         · C         · C         · C         · C         · C         · C         · C         · C         · C         · C         · C         · C         · C         · C         · C         · C         · C         · C         · C         · C         · C         · C         · C         · C         · C         · C         · C         · C         · C         · C         · C         · C         · C <th>S. NO.</th> <th>DESIGN TEMPERATURE IN</th> <th>SURFACE PREPARATION</th> <th>PAINT</th> <th>SYSTEM</th> <th>TOTAL DFT IN</th> <th>REMARKS</th>	S. NO.	DESIGN TEMPERATURE IN	SURFACE PREPARATION	PAINT	SYSTEM	TOTAL DFT IN	REMARKS
INTERNAL SURFACE OF FLOATING ROOF. INTERNAL STRUCTURAL OF CONE ROOF. BOTTOM PLA STELL. LADDERS SUPPORTS FOR STORING GRUIDE OIL. LDO AND HSD (EXCLUDING WHITE OIL PRODUCTS)       12.1     -14 TO 80     SSPC-SP-10     TGN/T OF 7     8.57C, 57     100,1     365       12.2     BARE SHEEL OF INSIDE FLOATING ROOF TANK AND CONE ROOF TANK FOR PRODUCTS MENTIONED IN 12.1     20     20       12.2.1     -14 TO 80     SSPC-SP-10     TREATMENT     WITH     2 X 10 = 20       12.2.1     -14 TO 80     SSPC-SP-10     TREATMENT     WITH     2 X 10 = 20       12.2.1     -14 TO 80     SSPC-SP-10     TREATMENT     WITH     2 X 10 = 20       12.2.1     -14 TO 80     SSPC-SP-10     TREATMENT     WITH     2 X 10 = 20       0.12.3.1     -14 TO 80     TREATMENT     WITH     2 X 10 = 20     200       12.3.1     -14 TO 80     SSPC-SP-10     TREATMENT     WITH     2 X 10 = 20       12.3.1     -14 TO 80     SSPC-SP-10     TREATMENT     WITH     2 X 10 = 20       12.3.1     -14 TO 80     SSPC-SP-10     TSPC-SP-10     TSPC-SP-10     2 COATS)       12.3.1     -14 TO 80     SSPC-SP-10     TSPC-SP-10     2 COATS     2 COATS       12.3.1     -14 TO 80     SSPC-SP-10     TSPC-SP-10     2 COATS     2 COATS       12.3.1 </th <th></th> <th>°</th> <th></th> <th>FILED PRIMER</th> <th>FINISH PAINT</th> <th>MICRONS (MIN.)</th> <th></th>		°		FILED PRIMER	FINISH PAINT	MICRONS (MIN.)	
12.1         -14 T0 80         SSPC-SP-10         1 COAT         0.5         3 COATS OF F.7 @ 100, la 365           12.2         BARE SHEEL OF INSIDE FLOATING ROOF TANK AND CONE ROOF TANK FOR PRODUCTS MENTIONED IN 12.1         20.010, a         3 X 100 = 300         20.010, a         200           12.2.1         -14 T0 80         SSPC-SP-10         PHOSPHATING         2 COATS OF @ 10, u         20           12.2.1         -14 T0 80         SSPC-SP-10         TREATMENT         WTH         2 X 10 = 20         2 COATS OF @ 10, u         20           12.2.1         -14 T0 80         SSPC-SP-10         TREATMENT         WTH         2 X 10 = 20         2 COATS OF @ 10, u         20           12.3.1         -14 T0 80         SSPC-SP-10         TANKS FOR PERCILEUM PRODUCTS SUCH AS ATF. GASOLINE. MAPHTHA, KEROSENE, MC         2 COATS OF \$60 100, u         365           12.3.1         -14 T0 80         SSPC-SP-10         1 COAT OF \$-9 @ 65.         3 C A00 = 300         3 C A00 = 300           12.3.1         -14 T0 80         SSPC-SP-10         1 COAT OF \$-9 @ 65.         3 C A00 = 300         557         3 C A00 = 300         1 COAT OF \$-9 @ 65.         1 COAT OF \$-1 @ 010, L         3 C A00 = 300         1 COAT OF \$-1 @ 010, L         3 C A00 = 300         1 COAT OF \$-1 @ 010, L         1 COAT OF \$-1 @ 010, L         1 COAT OF \$-1 @ 010, L		INTERNAL SURFACE ( STEEL, LADDERS SUPPORTS F	OF UNDERSIDE OF FLOATING R	OOF, INTERNAL STRUCTI AND HSD (EXCLUDING WI	URAL OF CONE ROOF, BOTT HITE OIL PRODUCTS)	FOM PLATE, ROOF STRUC	CTURE,
$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$	12.1	-14 TO 80	SSPC-SP-10	1 COAT OF F-9 @ 65-	3 COATS OF F-7 @ 100μ	365	F7 should be suitable for
122         BARE SHEEL OF INSIDE FLOATING ROOF TANK AND CONE ROOF TANK FOR PRODUCTS MENTIONED IN 12:1           122.1         -141 TO 80         3SPC-SP-10         PHOSPHATING         20           122.1         -141 TO 80         SSPC-SP-10         PHOSPHATING         20           12.2.1         -141 TO 80         SSPC-SP-10         PHOSPHATING         20           12.3.1         -141 TO 80         SSPC-SP-10         PHOSPHATING         210 = 20           0.F BOTTOM PLATE, UNDERSIDE OF FLOATING ROOF TANK FOR PEROLEUM PRODUCTS SUCH AS ATF, GASOLINE, MAPHTHA, KEROSENE, MC         20           12.3.1         -141 TO 80         SSPC-SP-10         10         100 = 300           12.3.1         -141 TO 80         SSPC-SP-10         10         100 = 300           12.3.1         -141 TO 80         SSPC-SP-10         10         100 = 300           12.4.1         -141 TO 80         SSPC-SP-10         10         100 = 300           12.4.1         -141 TO 80         SSPC-SP-10         10         100 = 300           12.4.1         -141 TO 80         SSPC-SP-10         10         100 = 300           12.4.1         -141 TO 80         SSPC-SP-10         10         100 = 300           12.4.1         -141 TO 80         SSPC-SP-10         10 <td></td> <td></td> <td></td> <td>75µ DFT/ COAT</td> <td>DFT/ COAT</td> <td></td> <td>immersion service of the</td>				75µ DFT/ COAT	DFT/ COAT		immersion service of the
12.2     BARE SHEEL OF INSIDE FLOATING ROOF TANK AND CONE ROOLUCTS MENTIONED IN 12.1       12.2.1     -14 TO 80     SSPC-SP-10     FREATMENT     WITH     2 X10 = 20     20       12.2.1     -14 TO 80     SSPC-SP-10     FREATMENT     WITH     2 X10 = 20     20       12.3.1     -14 TO 80     SSPC-SP-10     FREATMENT     WITH     2 X10 = 20     20       0.7.3     FLOATING CONE ROOF TANKS FOR PETROLEUM PRODUCTS SUCH AS ATF, GASOLINE, MAPHTHA, KEROSENE, MC     2 X10 = 20     20     20       12.3.1     -14 TO 80     SSPC-SP-10     T COATO F F9     6 S-10     36     3 X100 = 300       12.3.1     -14 TO 80     SSPC-SP-10     T COAT OF F9     6 S-10     36     3 X100 = 300       12.4.1     -14 TO 80     T 0.00     T 0.00 = 300     3 X100 = 300     36       12.4.1     -14 TO 80     SSPC-SP-10     T 0.00 T 0 F-9     6 S-10     37       12.4.1     -14 TO 80     SSPC-SP-10     T 0.00 T 0 F-9     6 S-10     1 0.01       12.4.1     -14 TO 80     SSPC-SP-10     T 0.00 T 0 F-9     6 S-10     1 0.01       12.4.1     -14 TO 80     SSPC-SP-10     T 0.00 T 0 F-9     6 S-10     1 0.01       12.5.1     -14 TO 80     SSPC-SP-10     T 0.00 T 0 T 0 F-9     6 S-10     1 0.01					3 X 100 = 300		products given.
12.2.1         -14 TO 80         SPC-SP-10         PHOSPHATING         2 COATS OF @10 µ         20           12.3.1         -14 TO 80         SPC-SP-10         PHOSPHATING         2 (2 COATS OF @10 µ         20           12.3         FLOATING CONE ROOF TANKS FOR PETROLEUM PROJUCTS SUCH AS ATF. CASOLINE, MAPHTHA, KEROSENE, MC         COATS)         2           12.3         FLOATING CONE ROOF AND SHELL ABOVE MAXIMULLUUDU EVEL. AND STRUCTURAL STEEL.         2           12.3.1         -14 TO 80         SSPC-SP-10         1 COAT OF F-9 @ 65-7         3 K 100 = 300           12.3.1         -14 TO 80         SSPC-SP-10         1 COAT OF F-9 @ 65-7         3 K 100 = 300           12.3.1         -14 TO 80         SSPC-SP-10         1 COAT OF F-9 @ 65-7         3 K 100 = 300           12.3.1         -14 TO 80         SSPC-SP-10         1 COAT OF F-9 @ 65-7         3 K 100 = 300           12.3.1         -14 TO 80         SSPC-SP-10         75µ DFT/COAT         3 K 100 = 300           12.4.1         -14 TO 80         SSPC-SP-10         7 5µ DFT/COAT         3 K 100 = 300           12.4.1         -14 TO 80         SSPC-SP-10         2 COAT OF F-9 @ 65-1         0 70µ           12.4.1         -14 TO 80         SSPC-SP-10         2 COAT OF F-9 @ 65-1         0 70µ         2 COAT OF F-0 <td>12.2</td> <td>BARE SHEEL OF INSIC</td> <td>DE FLOATING ROOF TANK AND (</td> <td>CONE ROOF TANK FOR PI</td> <td><b>RODUCTS MENTIONED IN 12</b></td> <td>.1</td> <td></td>	12.2	BARE SHEEL OF INSIC	DE FLOATING ROOF TANK AND (	CONE ROOF TANK FOR PI	<b>RODUCTS MENTIONED IN 12</b>	.1	
12.3     FLOATING CONE ROOF TANKS FOR PETROLEUM PROJUCTS SUCH AS ATF. GASOLINE. NAPHTHA, KEROSENE. MC       12.3     FLOATING CONE ROOF TANKS FOR PETROLEUM PRODUCTS SUCH AS ATF. GASOLINE. NAPHTHA, KEROSENE. MC       12.3.1     -14 TO 80       12.4.1     DATE       12.5.1     -14 TO 80       12.5.1     -14 TO 80       12.4.1     DATE       DATE     DATE	12.2.1	-14 TO 80	SSPC-SP-10	PHOSPHATING	2 COATS OF @10 μ	20	
12.3.1     FLOATING CONE ROOF TANKS FOR PETROLEUM PRODUCTS SUCH AS ATF, GASOLINE, NAPHTHA, KEROSENE, MC       12.3.1     FLOATING CONE ROOF AND SHELL ABOVE MAXIMUM LIQUUD LEVEL AND STRUCTURAL STEEL, LAD       12.3.1     -14 TO 80     SSPC-SP-10     75µ DFT/ COAT     3 COATS OF F-6 @ 100µ     365       12.3.1     -14 TO 80     SSPC-SP-10     75µ DFT/ COAT     3 X 100 = 300     365       12.4     BARE SHELL OF INSIDE     OF FLOATING CONE ROOF TANKS FOR PRODUCTS MENTIONED IN 12.3     3 X 100 = 300       12.4.1     -14 TO 80     SSPC-SP-10     75µ DFT/ COAT     3 X 100 = 300       12.4     BARE SHELL OF INSIDE     OF FLOATING CONE ROOF TANKS FOR PRODUCTS MENTIONED IN 12.3     56.75       12.4.1     -14 TO 80     SSPC-SP-10     75µ DFT/ COAT     55.75       12.5.1     -14 TO 80     SSPC-SP-10     75µ DFT/ COAT     56.75       12.6.1     -14 TO 80     SSPC-SP-10     75µ DFT/ COAT     56.75       12.6.1     -14 TO 80     SSPC-SP-10     75µ DFT/ COAT     27.00 S 200       12.6.1     -14 TO 80     SSPC-SP-10     75µ DFT/ COAT     27.00 S 200       12.6.1     -14 TO 80     SSPC-SP-10     75µ DFT/ COAT     27.00 S 200       12.6.1     -14 TO 80     SSPC-SP-10     2 COAT OF F-6 @ 10µ LATE     200       12.6.1     -14 TO 80     SSPC-SP-10				PHOSPHATING	7 Y 10 = 70		
12.3       FLOATING CONE ROOF TANKS FOR PETROLEUM PRODUCTS SUCH AS ATF, GASOLINE, NAPHTHA, KEROSENE, MC         0.F BOTTOM PLATE, UNDERSIDE OF FLOATING ROOF AND SHELL ABOVE MAXIMUM LIQUID LEVEL AND STRUCTURAL STEEL.       LAD         12.3.1       -14 TO 80       5SPC-SP-10       1 COAT OF F-9       65-       3 COATS OF F-6       100µ       365         12.4.1       -14 TO 80       SSPC-SP-10       1 COAT OF F-9       65-       3 COATS OF F-6       100µ       365         12.4.1       -14 TO 80       SSPC-SP-10       1 COAT OF F-9       65-       3 7 100 = 300       365         12.4.1       -14 TO 80       SSPC-SP-10       1 COAT OF F-9       65-       NONE       65-75         12.4.1       -14 TO 80       SSPC-SP-10       1 COAT OF F-9       65-       NONE       65-75         12.4.1       -14 TO 80       SSPC-SP-10       1 COAT OF F-9       65-       NONE       65-75         12.4.1       -14 TO 80       SSPC-SP-10       1 COAT OF F-9       65-       NONE       65-75         12.5.1       -14 TO 80       SSPC-SP-10       1 COAT OF F-9       66-75       75µ       75µ       75µ       75µ         12.5.1       -14 TO 80       SSPC-SP-10       2 COAT OF F       2 A 100 = 200       2 A 100 = 200       200 </td <td></td> <td></td> <td></td> <td>CHEMICALS (2 COATS)</td> <td></td> <td></td> <td></td>				CHEMICALS (2 COATS)			
12.3.1         -14 TO 80         SSPC-SP-10         1 COAT OF F-9         65-         3 COATS OF F-6         010μ         365           12.4         BARE SHELL OF INSIDE         SSPC-SP-10         75μ DFT/ COAT         3 X 100 = 300         365         5 X 100 = 300           12.4         BARE SHELL OF INSIDE         OF FLOATING CONE ROOF TANKS FOR PRODUCTS MENTIONED IN 12.3         3 X 100 = 300         3 X 100 = 300           12.4.1         -14 TO 80         SSPC-SP-10         1 COAT OF F-9         65-         NONE           12.4.1         -14 TO 80         SSPC-SP-10         1 COAT OF F-9         65-         NONE           12.5.1         -14 TO 80         SSPC-SP-10         1 COAT OF F-9         65-         NONE           12.5.1         -14 TO 80         SSPC-SP-10         2 COAT OF F-9         60         100µ         280           12.5.1         -14 TO 80         SSPC-SP-10         2 COAT OF F-9         60         100µ         280           12.5.1         -14 TO 80         SSPC-SP-10         2 COAT OF F-8         0 10µ         280           12.5.1         -14 TO 80         SSPC-SP-10         2 COAT OF F-8         0 10µ         280           12.5.1         -14 TO 80         SSPC-SP-10         2 COAT OF F-8         0 1	12.3 OF BOT	FLOATING CONE ROO	NE TANKS FOR PETROLEUM PRC	DDUCTS SUCH AS ATF, G/ F MAXIMUM LIQUID LEVEI	ASOLINE, NAPHTHA, KEROSI I AND STRUCTURAL STFFI	ENE, MOTOR SPIRIT, I ADDFRS FTC	INSIDE
12.4         BARE SHELL OF INSIDE         OF FLOATING CONE ROOF         TAUKS FOR PRODUCTS         DET/ COAT         TOW	12 3 1	-14 TO 80		1 COAT OF F-0 @ 65-	3 COATS OF E-6 @ 100.	365	E-6 should be suitable for
12.4     BARE SHELL OF INSIDE     OF FLOATING CONE ROOF     TANKS FOR PRODUCTS     MANTIONED IN 12.3       12.4.1     -14 TO 80     SSPC-SP-10     1 COAT OF F-9     65-7     65-75       12.5.1     -14 TO 80     SSPC-SP-10     1 COAT OF F-9     65-7     65-75       12.5.1     -14 TO 80     SSPC-SP-10     1 COAT OF F-9     65-75     65-75       12.5.1     -14 TO 80     SSPC-SP-10     2 COAT OF F-6     010µ     280       12.5.1     -14 TO 80     SSPC-SP-10     2 COAT OF F-6     010µ     280       12.5.1     -14 TO 80     SSPC-SP-10     2 COAT OF F-6     010µ     280       12.5.1     -14 TO 80     SSPC-SP-10     2 COAT OF F-6     010µ     280       12.5.1     -14 TO 80     SSPC-SP-10     2 COAT OF F-6     010µ     280       12.5.1     -14 TO 80     SSPC-SP-10     2 COAT OF F-6     010µ     280       12.6.1     -14 TO 80     SSPC-SP-10     2 COAT OF F-6     010µ     280       12.6.1     -14 TO 80     SSPC-SP-10     2 COAT OF F-6     010µ     280       12.6.1     -14 TO 80     SSPC-SP-10     2 COAT OF F-6     010µ     280       12.6.1     -14 TO 80     SSPC-SP-10     2 COAT OF F-6     010µ     280			5	75u DFT/ COAT	DFT/ COAT		immersion service of
12.4     BARE SHELL OF INSIDE     OF FLOATING CONE ROOF TANKS FOR PRODUCTS MENTIONED IN 12.3       12.4.1     -14 T0 80     SSPC-SP-10     1 COAT OF F-9 @ 65-     NONE       12.5.1     INTERNAL PROTECTION IF FIXED ROOF TYPE STORAGE TANKS FOR POTABLE WATER: INSIDE OF SHELL, UNDER     65-75       12.5.1     -14 T0 80     SSPC-SP-10     75µ DFT/ COAT     65-75       12.5.1     -14 T0 80     SSPC-SP-10     2 COAT OF F-6 @ 40µ     2 COATS OF F-6 @ 100µ       12.5.1     -14 T0 80     SSPC-SP-10     DFT/ COAT     2 COAT OF F-6 @ 40µ     2 COATS OF F-6 @ 100µ       12.5.1     -14 T0 80     SSPC-SP-10     DFT/ COAT     2 X 40 = 80     2 X 100 = 200       12.5.1     -14 T0 80     SSPC-SP-10     DFT/ COAT     2 X 40 = 80     2 X 100 = 200       12.6.1     -14 T0 80     SSPC-SP-10     T X 40 = 80     2 X 100 = 200       12.6.1     -14 T0 80     SSPC-SP-10     2 COAT OF F-6 @ 40µ     2 COATS OF F-6 @ 100µ       12.6.1     -14 T0 80     SSPC-SP-10     2 COAT OF F-6 @ 40µ     2 COATS OF F-6 @ 100µ       12.6.1     -14 T0 80     SSPC-SP-10     2 COAT OF F-6 @ 40µ     2 COATS OF F-6 @ 100µ       12.6.1     -14 T0 80     SSPC-SP-10     2 COAT OF F-6 @ 100µ     200       12.6.1     ALL     SSPC-SP-10     2 COAT OF F-6 @ 100µ     200					$3 \times 100 = 300$		petroleum produce like
12.4     BARE SHELL OF INSIDE     OF FLOATING CONE ROOF TANKS FOR PRODUCTS MENTIONED IN 12.3       12.4.1     -14 TO 80     SSPC-SP-10     1 COAT OF F-9 @ 65-     NONE     65-75       12.4.1     -14 TO 80     SSPC-SP-10     75µ DFT/ COAT     1 COAT OF F-9 @ 65-     NONE     65-75       12.5     INTERNAL PROTECTION IF FIXED ROOF TYPE STORAGE TANKS FOR POTABLE WATER: INSIDE OF SHELL, UNDER     NONE     65-75       12.5.1     -14 TO 80     SSPC-SP-10     2 COAT OF F-6 @ 40µ     2 COATS OF F-6 @ 100µ     280       12.5.1     -14 TO 80     SSPC-SP-10     2 COAT OF F-6 @ 40µ     2 COATS OF F-6 @ 100µ     280       12.5.1     -14 TO 80     SSPC-SP-10     2 COAT OF F-6 @ 40µ     2 COATS OF F-6 @ 100µ     280       12.5.1     -14 TO 80     SSPC-SP-10     2 X 40 = 80     2 X 100 = 200     2 X 100 = 200       12.6.1     -14 TO 80     SSPC-SP-10     2 X 40 = 80     2 X 100 = 200     2 X 100 = 200       12.6.1     -14 TO 80     SSPC-SP-10     2 X 40 = 80     2 X 100 = 200     2 X 100 = 200       12.6.1     -14 TO 80     SSPC-SP-10     A0 = 80     2 X 100 = 200     2 X 100 = 200       12.6.1     -14 TO 80     SSPC-SP-10     EBONITE RUBBER LINING AS PER SIMMS SPECIFICATION 6-       12.7     ALL     SSPC-SP-10     3 COATS VINYL     CHORIDE		_					ATF, Kerosene, petrol etc.
12.4.1     -14 TO 80     SSPC-SP-10     1 COAT OF F9 @ 65     NONE     65-75       12.5     INTERNAL PROTECTION IF FIXED ROOF TYPE STORAGE TANKS FOR POTABLE WATER: INSIDE OF SHELL, UNDER     055     056       12.5     INTERNAL PROTECTION IF FIXED ROOF TYPE STORAGE TANKS FOR POTABLE WATER: INSIDE OF SHELL, UNDER     057     056       12.5.1     -14 TO 80     SSPC-SP-10     2 COAT OF F6 @ 40µ     2 COATS OF F6 @ 100µ     280       12.5.1     -14 TO 80     SSPC-SP-10     2 TO 05     2 COAT OF F6 @ 40µ     2 COATS OF F6 @ 100µ     280       12.5.1     -14 TO 80     SSPC-SP-10     2 TO 05     2 COAT OF F6 @ 40µ     2 COATS OF F6 @ 100µ     280       12.5.1     -14 TO 80     D. M. (DEMINERALISED WATER) AND HYDROCHLORIC ACID (HCL): INTERNAL SHELL, BOTTOM PLATE AND ALL ACC     2 X 100 = 200     2 X 100 = 200       12.6.1     -14 TO 80     SSPC-SP-10     EBONITE RUBBER LINING AS PER SIMIS SPECIFICATION 6-       12.6.1     -14 TO 80     SSPC-SP-10     BONITE RUBBER LINING AS PER SIMIS SPECIFICATION 6-       12.7.1     ALL     ALL     -14 TO 80     SSPC-SP-10       12.7.1     ALL     ALL     SSPC-SP-10     3 COATS VINYL       12.7.1     ALL     ALL     SSPC-SP-10     3 COATS VINYL	12.4	BARE SHELL OF INSIC	DE OF FLOATING CONE ROOF T.	ANKS FOR PRODUCTS MI	ENTIONED IN 12.3		
12:5     INTERNAL PROTECTION IF FIXED ROOF TYPE STORAGE TANKS FOR POTABLE WATER: INSIDE OF SHELL, UNDER       12:5     INTERNAL PROTECTION IF FIXED ROOF TYPE STORAGE TANKS FOR POTABLE WATER: INSIDE OF SHELL, UNDER       12:5.1     -14 TO 80       12:6     D. M. (DEMINERALISED WATER) AND HYDROCHLORIC ACID (HCL): INTERNAL SHELL, BOTTOM PLATE AND ALL ACC       12:6     -14 TO 80       12:7     EG TANKS (INTERNAL SHELL, BOTTOM PLATE AND ALL ACCESSORIES)       12:7.1     ALL       12:7.1     ALL       12:7.1     ALL	12.4.1	-14 TO 80	SSPC-SP-10	1 COAT OF F-9 @ 65-	NONE	65-75	No over coating is allowed
12.5     INTERNAL PROTECTION IF FIXED ROOF TYPE STORAGE TANKS FOR POTABLE WATER: INSIDE OF SHELL, UNDER       12.5.1     -14 TO 80     STRURAL STEEL WORKS, LADDERS, WALKWAYS, PLATFORMS ETC.       12.5.1     -14 TO 80     SSPC-SP-10     2 COAT OF F-6 @ 40µ     2 COATS OF F-6 @ 100µ     280       12.5.1     -14 TO 80     SSPC-SP-10     2 X 40 = 80     2 X 100 = 200     240     280       12.5.1     -14 TO 80     SSPC-SP-10     2 X 40 = 80     2 X 100 = 200     21 A 40     280       12.6     D. M. (DEMINERALISED WATER) AND HYDROCHLORIC ACID (HCL): INTERNAL SHELL, BOTTOM PLATE AND ALL ACC     2 X 100 = 200     21 A 40     200       12.6     D. M. (DEMINERALISED WATER) AND HYDROCHLORIC ACID (HCL): INTERNAL SHELL, BOTTOM PLATE AND ALL ACC     2 X 100 = 200     21 A 40     200       12.6     TO M. (DEMINERALISED WATER) AND HYDROCHLORIC ACID (HCL): INTERNAL SHELL, BOTTOM PLATE AND ALL ACC     2 X 100 = 200     2 X 100 = 200       12.7.1     ALT TO 80     SSPC-SP-10     EBONITE RUBBER LINING AS PER SIMIS SPECIFICATION 6-       12.7.1     ALL     SSPC-SP-10     3 COATS VINYL     CHORIDE     225       12.7.1     ALL     SSPC-SP-10     3 COATS VINYL     CHORIDE     225				75μ DFT/ COAT			same as per pre-erection
12.5     14 TO 80     SSPC-SP-10     2 COAT OF F6 @ 40µ     2 COATS OF F-6 @ 100µ     280       12.5.1     -14 TO 80     SSPC-SP-10     2 COAT OF F6 @ 40µ     2 COATS OF F-6 @ 100µ     280       12.5.1     -14 TO 80     SSPC-SP-10     2 COAT OF F6 @ 40µ     2 COATS OF F-6 @ 100µ     280       12.5.1     -14 TO 80     SSPC-SP-10     2 X 40 = 80     2 X 100 = 200     210       12.6     D. M. (DEMINERALISED WATER) AND HYDROCHLORIC ACID (HCL): INTERNAL SHELL, BOTTOM PLATE AND ALL ACC     2 X 100 = 200     2 X 100 = 200       12.6     D. M. (DEMINERALISED WATER) AND HYDROCHLORIC ACID (HCL): INTERNAL SHELL, BOTTOM PLATE AND ALL ACC     2 X 100 = 200     2 X 100 = 200       12.7.1     ALL     -14 TO 80     SSPC-SP-10     EBONITE RUBBER LINING AS PER SIMIS SPECIFICATION 6-       12.7.1     ALL     ALL     SSPC-SP-10     3 COATS VINYL     CHORIDE       12.7.1     ALL     SSPC-SP-10     3 COATS 20 (FbL)     225	10 5		ON LE EIVED BOOF TVBE STOBA				
12.5.1       -14 TO 80       SSPC-SP-10       2 COAT OF F-6 @ 40µ       2 COATS OF F-6 @ 100µ       280         12.5.1       -14 TO 80       D. M. (DEMINERALISED WATER) AND HYDROCHLORIC ACID (HCL): INTERNAL SHELL, BOTTOM PLATE AND ALL ACC       2 X 40 = 80       2 X 100 = 200       200         12.6       D. M. (DEMINERALISED WATER) AND HYDROCHLORIC ACID (HCL): INTERNAL SHELL, BOTTOM PLATE AND ALL ACC       2 X 100 = 200       200         12.6.1       -14 TO 80       SSPC-SP-10       EBONITE RUBBER LINING AS PER SIMIS SPECIFICATION 6-         12.7.1       ALL       SSPC-SP-10       3 COATS VINYL       CHORIDE       225         12.7.1       ALL       SSPC-SP-10       3 COATS 23 @ 75µ / COAT       225	INSIDE 5	URFACE BOTTOM PLATE AND S	STRURAL STEEL WORKS, LADDE	ERS, WALKWAYS, PLATFC	DRMS ETC.		
DFT/COAT     DFT/COAT       12.6     D. M. (DEMINERALISED WATER) AND HYDROCHLORIC ACID (HCL): INTERNAL SHELL, BOTTOM PLATE AND ALL ACC       12.6.1     -14 TO 80       12.6.2     D. M. (DEMINERALISED WATER) AND HYDROCHLORIC ACID (HCL): INTERNAL SHELL, BOTTOM PLATE AND ALL ACC       12.6.1     -14 TO 80       12.7.1     EG TANKS (INTERNAL SHELL, BOTTOM PLATE ROOF AND ALL ACCESSORIES)       12.7.1     ALL       12.7.1     ALL       ASPC-SP-10     3       COATS     VINYL       CHORIDE     CO-POLYMER       225	12.5.1	-14 TO 80	SSPC-SP-10	2 COAT OF F-6 @ 40µ	2 COATS OF F-6 @ $100\mu$	280	F-6 shall be suitable for
2 X 40 = 80         2 X 100 = 200           12.6         D. M. (DEMINERALISED WATER) AND HYDROCHLORIC ACID (HCL): INTERNAL SHELL, BOTTOM PLATE AND ALL ACC           12.6.1         -14 TO 80         SSPC-SP-10         EBONITE RUBBER LINING AS PER SIMMS SPECIFICATION 6-           12.6.1         -14 TO 80         SSPC-SP-10         EBONITE RUBBER LINING AS PER SIMMS SPECIFICATION 6-           12.7.1         ALL         SSPC-SP-10         3         COATS         VINYL         CHORIDE         CO-POLYMER         225           12.7.1         ALL         SSPC-SP-10         3         COATS         VINYL         CHORIDE         CO-POLYMER         225				DFT/ COAT	DFT/ COAT		immersion service.
12.6     D. M. (DEMINERALISED WATER) AND HYDROCHLORIC ACID (HCL): INTERNAL SHELL, BOTTOM PLATE AND ALL ACC       12.6.1     -14 TO 80     SSPC-SP-10     EBONITE RUBBER LINING AS PER SMMS SPECIFICATION 6-       12.7.1     EG TANKS (INTERNAL SHELL, BOTTOM PLATE ROOF AND ALL ACCESSORIES)     3     COATS     VINYL     CHLORIDE     C0-POLYMER     225       12.7.1     ALL     SSPC-SP-10     3     COATS     VINYL     CHLORIDE     C0-POLYMER     225				2 X 40 = 80	2 X 100 = 200		
12.6.1         -14 TO 80         SSPC-SP-10         EBONITE RUBBER LINING AS PER SIMIS SPECIFICATION 6- 12.7           12.7         EG TANKS (INTERNAL SHELL, BOTTOM PLATE ROOF AND ALL ACCESSORIES)         3         COATS         VINYL         CHLORIDE         CO-POLYMER         225           12.7.1         ALL         SSPC-SP-10         3         COATS         VINYL         CHLORIDE         CO-POLYMER         225           12.7.1         ALL         AMERCOAT 23 @ 75µ / COAT         AMERCOAT 23 @ 75µ / COAT         225	12.6	D. M. (DEMINERALISEI	D WATER) AND HYDROCHLORIC	S ACID (HCL): INTERNAL S	HELL, BOTTOM PLATE AND ,	ALL ACCESSORIES	
12.7     EG TANKS (INTERNAL SHELL, BOTTOM PLATE ROOF AND ALL ACCESSORIES)       12.7.1     ALL     3     COATS     VINYL     CHLORIDE     CO-POLYMER     225       12.7.1     ALL     SSPC-SP-10     3     COATS     VINYL     CHLORIDE     CO-POLYMER     225	12.6.1	-14 TO 80	SSPC-SP-10	EBONITE RUBBER LININ	VG AS PER SMMS SPECIFICF	ATION 6-06-204	
12.7.1 ALL SSPC-SP-10 3 COATS VINYL CHLORIDE CO-POLYMER 225 AMERCOAT 23 @ 75µ / COAT	12.7	EG TANKS (INTERNAL	SHELL, BOTTOM PLATE ROOF	AND ALL ACCESSORIES)			
AMERCOAT 23 @ 75µ / COAT	12.7.1	ALL	SSPC-SP-10	3 COATS VINYL (	CHLORIDE CO-POLYMER	225	
				AMERCOAT 23 @ 75µ / (	COAT		

				(			
RANC		AKD LECHNICAL SI	PECIFICALION				
	4	OIL & GAS SBU, I	DELHI	and the second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second s			
			DOCUMENT NO.	Page 32 of 54			
Ë	SHOP & FIEL	D PAINTING	MEC/S/05/21/07	REVISION: 0			
				EDITION: 1			
D	ESIGN TEMPERATURE IN	SURFACE PREPAR	ATION	PAINT SYSTEM	TOTAL DFT IN	REMARKS	
	°		FILED PRIMER	FINISH PAINT	MICRONS (MIN.)		
	INSIDE PONTOON AND	D INSIDE OF DOUBLE D	DECK OF ALL FLOATING ROC	JFS.			
1 -14	TO 80	SSPC-SP-3	1 COAT OF F-	3 @ 1 COATS OF F-6 @ 100µ	200		
			100µ DFT/ COAT	DFT/ COAT			
				1 X 100 = 100			
	INTERNAL SURFACE (	<b>DF AMINE &amp; SOUR WA</b>	TER STORAGE TANKS				
1 -14	TO 80	SSPC-SP-10	1 COAT OF F-9 (	0 65- 2 COATS OF F-15 @ 75μ	215-225		
			75μ DFT/ COAT	DFT/ COAT			
				2 V 76 - 160			

					) CARBON STEEL PLANT PIPII	TOTAL DFT IN REMARK	() )		TAR 4mm CTE coating ING confirm to 12 203 Per BS: 4164 ACH		OF 4 mm	AS			65-75		<u>@</u> 365	65-75	250
	Population Contraction	Page 33 of 54	REVISION: 0	EDITION: 1	JNDERGROUND	SYSTEM FINISH PAINT			4mm THICK COAL COATING WRAPP AS PER AWWA-C- IN 2 LAYER OF E/ 2mm THICKNESS		2 LAYERS	TAPE COATING PFR AWWA-C-203			NONE		3 COATS OF F-7 100μ DFT/ COAT 3 X 100 = 300	NONE	NONE
CATION		CUMENT NO.	/S/05/21/07		ERNAL SIDE OF 1	PAINT	0)		1 COAT OF SYNTHETIC FAST DRYING PRIMER TYPE-B AS PER AWWA-C-203 (1991)		1 COAT OF SVNTHETIC EAST	DRYING PRIMER TYPE-R AS PER	AWWA-C-203 (1991)		1 COAT OF F-9 @ 65-75u DFT/ COAT	JND STORAGE TANKS:	1 COAT OF F-9 @ 65-75μ DFT/ COAT	1 COAT OF F-9 @ 65-75µ DFT/ COAT	1 CUAL UF AMERCOAT 738 @ 250μ DFT/ COAT
RD TECHNICAL SPECIFIC	OIL & GAS SBU, DELHI	OQ	PAINTING MEC		SYSTEM FOR EXTE KS	SURFACE PREPARATION	NT PIPING (UNDERGROUNI		SSPC-SP-10	ATING	SSPC-SP-10			<b>NT PIPING (UNDERGROUND</b>	SSPC-SP-10	JNINSULATED UNDERGROU	SSPC-SP-10	SSPC-SP-10	
STANDAF	<u>.</u>		SHOP & FIELD		3.0 : COATING AND TANI	ESIGN TEMPERATURE IN °C	CARBON STEEL PLA	YARD COATING	25 TO 60	OVER THE DITCH CO	25 Tto 60			CARBON STEEL PLAN	61 TO 400	EXTERNAL SIDE OF L	40 TO 80	-90 TO41 81 TO 400° c	
FGD. OFF: RANCH	34002		TITLE		TABLE 13	S. NO. DE	13.1	13.1.1	13.1.1.1	13.1.2	13.1.2.1			13.2	13.2.1	13.3	13.3.1	13.3.2	

					I STEEL, LOW TEEL PIPING,	REMARKS			For other temprature ranges no painting is	required under insulation.			If the piping & equipments are already	erected then surface shall be prepared by cleaning with emery	paper and wash/ flush with chloride free DM water followed by wiping with organic solvent
					AFETY CARBON STAINLESS S	TOTAL DFT IN	MICRONS (MIN.)		315		105-115		MINERAL SEALANT	250	
	and the state	Page 34 of 54	REVISION: 0	EDITION: 1	ED (HOT COLD S/ CARBON STEEL 8 ENVIRONMENT	SYSTEM	FINISH PINTS	D EQUIPMENT & TANKS	2 COATS OF F-14 @ 125μ DFT/ COAT	2 X 125 = 250 OR 3 COATS OF F-15= 3 X 80=240	3 COATS OF F-12 @ 20μ DFT/ COAT 3 X 20 = 60		AND CHLORIDE FREE LIED	2 COATS OF F-14 @ 125μ DFT/ COAT	2 X 125 = 250 OR 3 COATS OF F-15= 3 X 80 = 240
CATION		DCUMENT NO.	C/S/05/21/07		ON FOR INSULAT TEMPERATURE ( IPMENTS IN ALL E	PAINT	PRIMER	L AND LTCS PIPING ANI	REPAIR OF PRE- FABRICATION	75μ DFT9 @ 65-	REPAIR OF PRE- FABRICATION PRIMER F-9 @ 65- 75µ DFT	OY-20- PIPING	VITH ALUMINUM FOIL	NONE	
RD TECHNICAL SPECIFI	OIL & GAS SBU, DELHI	DO	DAINTING MEC		© UNDER INSULATIC LOY STEEL, LOW ORAGE TANKS EQUI	SURFACE PREPARATION		I STEEL, LOW ALLOY STEE	SSPC-SP-10		SSPC-SP-10	SS STEEL INCLUDING ALLO	ALUMINUM SHEETING W CONTAINING BARIUM CHR	SSPC-SP-10 ( 15-25μ SURFACE PROFILE)	
D STANDA	4		SHOP & FIELD		4.0 : PAINTING ALI STO	ESIGN TEMPERATURE	N °C	INSULATED CARBON	. TO 125		PERATING EMPERATURE –45 TO 25° C BUT DESIGN EMPERATURE 126-400°	INSULATED STAINLE	ELOW 0° C TO ALL INUS TEMPRATURE	TO 120	
MECON LIMITE	834002		TITLE		TABLE 1	S. NO. D		14.1	14.1.1 -4		14.1.2 0 11 12 12 12 12 12 12 12 12 12 12 12 12	14.2	14.2.1 BI	14.2.2 0	Te 518 of 6

MECON LIMI	TED STAND	ARD TECHNICAL SPECIF	ICATION			
834002		OIL & GAS SBU, DELH		Altoria Contractor		
			DOCUMENT NO.	Page 35 of 54		
TITLE	SHOP & FIEL	D PAINTING MI	EC/S/05/21/07	REVISION: 0		
				EDITION: 1		
S. NO.	DESIGN TEMPERATURE	SURFACE PREPARATIO	PAIN	T SYSTEM	TOTAL DFT IN	REMARKS
	S		PRIMER	FINISH PINTS	MICRONS (MIN.)	
14.2.3	121 TO 500	SSPC-SP-10	NONE	3 COATS OF F-12 @ 20μ DFT/COAT	60	No pre erection primer to be applied
				3 X 20 = 60		
14.2.4	501 TO 1000	SSPC-SP-10	NONE	1 СОАТ ОF АМЕRCОАТ 738 @ 150µ DFT/ COAT	150	Only Amorcoat 738 from Amoron is available for this temperature range.
14.2.5	CYCLIC SERVICE-196 TO 480 EXCEPTING -45 TO 120	SSPC-SP-10	NON	1 COAT OF AMERCOAT 738 @ 150μ DFT/ COAT	150	
14.3	NO PAINTING REQL	L JIRED FOR INSULATED MC	NEL, IN COLOY AND N	L ICKEL LINES		

					SHEETS OF COVERS AND	REMARKS		For C. S.	FOR NON FERROUS SURFACE		REMARKS	SHADE AS PER DEFENCE REQUIREMENTS	
					ES AND TUBE N PLATES, END	TOTAL DFT IN	MICRONS (MIN.)	290	300	F	TOTAL DFT IN	MICKONS (MIN.) 130	300
	To Habiri	Page 36 of 54	REVISION: 0	EDITION: 1	EL WATER BOXE INNELS, PARTITIO	SYSTEM	FINISH PAINT	2 COATS OF F-7 @ 125μ DFT/ COAT 2 x 125 = 250	2 COATS OF F-7 @ 125μ DFT/ COAT 2 x 125 = 250	ROUS TUBE SHEE	SYSTEM	<b>ΓΙΝΙΣΗ ΡΑΙΝΙ</b> 2 COATS OF F-2 @ 40μ DFT/ COAT 2 x 40 = 250	2 COATS OF F-7 @ 125μ DFT/ COAT 2 x 125 = 250
ATION		CUMENT NO.	/S/05/21/07		CARBON STEE TER BOXES, CHA	PAINT	PRIMER	1 COATS OF F-6 @ 40μ DFT/ COAT	1 COATS OF P-4 @ 8μ DFT/ COAT 1 COATS OF P-6 @ 40μ DFT/ COAT	WERS/ NON-FER	PAINT	FILEU PAINI 1 COATS OF P-4 @ 8-10µ DFT/ COAT + 1 COAT OF P-6 @ 4u DFT/ COAT	1 COATS OF P-4 @ 8µ DFT/ COAT 1 COATS OF P-6 @ 40µ DFT/ COAT
RD TECHNICAL SPECIFIC	OIL & GAS SBU, DELHI	DO	D PAINTING MEC		L PROTECTION OF S/ CONDENSERS WA EETS ETC.	SURFACE PREPARATION		SSPC-SP-10	SSPC-SP-10	G SYSTEM FOR GI TO	SURFACE PREPARATION	SSPC-SP-10	SSPC-SP-10
T STANDA	4		SHOP & FIELD		.0 : INTERNA COOLER TUBE SHI	DESIGN	TEMPERATURE IN °C	Upto 65	Upto 65 NON FERROUS AND BRASS TUBE SHEETS	.0 FIELD PAINTING	DESIGN	Upto 65	Upto 65 NON FERROUS AND BRASS TUBE SHEETS
MECON LIMITED	834002		TITLE		TABLE 15	S. NO.		15.1	15.2	TABLE 16	S. NO.	16.1	16.2

of 644

MECON LIMITED REGD. OFF: RANCHI	STANDARD TECHNICAL	SPECIFICATION	
834002	OIL & GAS SBU	, DELHI	93 9001:2000 Contrib
		DOCUMENT NO.	Page 37 of 54
TITLE	SHOP & FIELD PAINTING	MEC/S/05/21/07	REVISION : 0
			EDITION: 1

### 17.0 STORAGE

17.1 All paints and painting materials shall be stored only in rooms to be arranged by contractor and approved by Engineer-in-Charge for the purpose. All necessary precautions shall be taken to prevent fire. The storage building shall preferably be separate from adjacent building. A signboard bearing the words " PAINT STORAGE NO NAKED LIGHT-HIGHLY INFLAMMABLE" shall be clearly displayed outside.

### 18.0 COLOUR CODE FOR PIPING

For identification of pipeline, the colour code as per Table 18.1 shall be used. Paint material for color-coding shall be as specified in this standard in clause- 6.0.

18.1 Colour coding scheme for pipe, equipment, machinery & structure:

SR. NO.	DESCRIPTION	GROUND COLOUR	FIRST COLOUR BAND	SECOND COLOUR BAND
18.1.1	ALL KINDS OF WATER DRINKING WATER DE-MINERALISED WATER COOLING WATER BOILER FEED WATER CONDENSATE QUENCH WATER WASH WATER PROCESS WATER PROCESS WATER FIRE WATER SEA WATER	Sea Gree -do- -do- -do- -do- -do- -do- Fire red Sea Green	French Blue Gulf Red French Blue Gulf Red Light Brown Dark Grey Ganary Yellow Oxide Red Crimson Red White	Signal Red - - Signal Red - - - - -
18.1.2	STEAM VERY HIGH PRESSURE STEAM (VHP) HIGH PRESSURE STEAM (SH) MEDIUM PRESSURE STEAM (SH) LOW PRESSURE STEAM (SL) DILUTION STEAM/ PURGE STEAM	Aluminiumto IS2339 -do- -do- -do- -do-	Signal Red French Blue Gulf Red Canary Yellow Grey	- - Canary Yellow

	<b>D</b>	STANDARD TEC	HNICAL	SPECIFICAT	ION		
834002	CHI	OIL & C	GAS SBU,	DELHI			मिकान मेकान 9007:2000 Confu
				DOCUI	MENT NO.		Page 38 of 54
TITLE		SHOP & FIELD PAINT	ING	MEC/S/	05/21/07		REVISION: 0
							EDITION: 1
SR. NO.		DESCRIPTION		GROUND COLOUR	FIRST COLOUR B	AND	SECOND COLOUR BAND
18.1.3	COM PLAI INST NITF OXY CO <sub>2</sub>	IPRESSED AIR NT AIR TRUMENT AIR ROGEN GEN	Sky B -do- -do- Canar -do- -do-	lue y Yello	Signal Red Silver Grey French Blue Black White Light Grey	;	- - - - -
18.1.4	GAS FUE CHA RES ACE SWE	ES L GAS AND SOUR GAS RGE GAS IDUE GAS, LPG TYLENE EET GAS	Canar -do- -do- -do- -do-	y Yellow	Grey Signal Red Oxide Red Service Broy Grey	wn	Dark Violet French Blue White - -
18.1.5	ACIE SULI NITF HYD ACE CAU CHL	DS AND CHEMICALS FURIC ACID RIC ACID ROCHLORIC ACID TIC ACID STIC ORINE	DARK -do- -do- -do- smoke Canar	Violet e Grey y Yellow	Briliant Gree French Blue Signal Red Silver Grey Light Orang Dark Violet	en e	Light Orange -do- -do- -do- - - -do-
18.1.6	HYD NAP PRO ETH ETH BEN BUT ETH PRO ETH TAR ARO MET PYR MIXE LPG KER DIES DIES	RO CARBONS THAS PPYLENE PYLENE C.G. (LIQ) YLENE GLYCOL YLENE DICHLORIDE ZENE ADIENE ADIENE ANE(LIQ) PYLENE(LIQ) YLENE(LIQ) YLENE(LIQ) OLYSIS GASOLINE ED C4(LIQ) (LIQ) OSENE SEL OIL (WHITE) SEL OIL (BLACK)	Dark / Grey -do- -do- -do- -do- -do- Dark / Grey -do- -do- -do- -do- -do- -do- -do- -do	Admiralty Admiralty Brown	Brilliant Gre -do- -do- Gulf Red Canary Yello Black Light Grey Signal Red Light Grey Signal Grey Brilliant Gre White Brilliant Gre Signal Gree Brilliant Gre -do- -do-	en ow en n n	Black Smoke Grey Gulf Red - - - French Blue Black Black Brilliant Green Canary Yellow Gulf Red Black Light Brown Dark Violet -

MECON LIMITED REGD. OFF: RANCHI	STANDARD TECHNICAL		
834002	OIL & GAS SBU	10 401:2000 COMP	
		DOCUMENT NO.	
TITLE	SHOP & FIELD PAINTING	MEC/S/05/21/07	REVISION : 0
			EDITION : 1

- 18.2 The colour code scheme is intended for identification of the individual group of the pipeline. The system of colour coding of a ground colour and colour bands superimposed on it.
- 18.3 Ground colours as given in Table 18.1 shall be applied throughout the entire length for uninsulated pipes, on the metal cladding & on surfaces covered by Clause 2.2.2, ground colour coating of minimum 2m length or of adequate length not to be mistaken as colour band shall be applied at places requiring colour bands. Colour band(s) shall be applied at the following location.
  - a. At battery limit points
  - b. Intersection points & change of direction points in piping ways.
  - c. Other points, such as midway of each piping way, near valves, junction joints of services appliances, walls, on either side of pipe culverts.
  - d. For zong stretch/ xard piping at 50M interval.
  - e. At start and terminating points.

### 18.4Identification Sign

- 18.4.1 Flow direction shall be indicated by an arrow in the location stated in Para a,b,c & d and as directed by Engineer-in-charge.
- 18.4.2 Colours of arrows shall be black or white and in contrast to the colour on which they are superimposed.
- 18.4.3 Product names shall be marked at pump inlet, outlet and battery limit in a suitable size as approved by Engineer-in-charge.
- 18.4.4 Size of arrow shall be either of those given in 18.5.

### 18.5Colour Bands

MECON LIMITED REGD. OFE: RANCHI	STANDARD TECHNICAL	SPECIFICATION		
834002	OIL & GAS SBU	08 मेकान 9001:2000 Contract		
		DOCUMENT NO.	Page 40 of 54	
TITLE	SHOP & FIELD PAINTING	MEC/S/05/21/07	<b>REVISION: 0</b>	
			EDITION: 1	

18.5.1 As a rule minimum width of colour band shall conform to the following table:

Nominal Pipe Size	Width : L(mm)
3" NB and below	25mm
Above 3" NB upto 6" NB	50mm
Above 8" NB upto 12" OD	75mm
Above 12" OD	100mm

- Note: For insulated pipes, nominal pipe size means the outside diameter of insulation. Nominal pipe size figures are to be inches.
- 18.5.2 Colour band(s) shall be arranged in the sequence shown in Table 18.1 and the sequence follows the direction of flow. The relative proportional width of the first colour band to the subsequent bands shall be 4:1, minimum width of any band shall be as per Clause 18.5.1.
- 18.5.3 Whenever it is required by the Engineer-in-charge to indicate that a pipeline carries a hazardous material, a hazard marking of diagonal strips of black and golden yellow as epr IS:2379 shall be painted on the ground colour.
- 18.6 Wherever it is required by the Engineer-in-charge to indicate that a pipeline carries a hazardous material, a hazard marking of diagonal strips of black and golden yellow as per IS:2379 shall be painted on the ground colour.

### 19.0 **IDENTIFICATION OF VESSELS, PIPING ETC.**

19.1 Equipment number shall be stencilled in black or white on each vessel, column, equipment & machinery (insulated or uninsulated) after painting. Line number n black or white shall be stencilled on all the pipelines of more than one location as directed by Engineer-in-charge, size of letters printed shall be as below :

Column & Vessels	-	150mm(high)
Pump, Compressor and other machinery	-	50mm (high)
Piping	-	40-150mm

MECON LIMITED	STANDARD TECHNICAL								
834002	OIL & GAS SBU,	10 मेकान 10 1:2000 Californi							
		DOCUMENT NO.	Page 41 of 54						
TITLE	SHOP & FIELD PAINTING	MEC/S/05/21/07	<b>REVISION</b> : 0						
			EDITION: 1						
19.2	Identification of storage tanks The storage tanks shall be mar	<b>s</b> ked as detailed in the dr	awing.						
20.0	PAINTING FOR CIVIL DEFEN	CE REQUIREMENTS							
20.1	Following items shall be painted	l for camouflaging if req	uired by the client.						
	<ul> <li>a. All columns</li> <li>b. All tanks in offsites</li> <li>c. Large vessels</li> <li>d. Spheres</li> </ul>								
20.2	Two coats of selected finishing applied in a particular pattern Engineer-in-charge.	Two coats of selected finishing paint as per defence requirement shall be applied in a particular pattern as per 20.3 and as per the instructions of Engineer-in-charge.							
20.3	Method of Camouflaging								
20.3.1	Disruptive painting for camouflar ratio of 5:3:2 (all matt finish).	aging shall be done in t	hree colours in the						
	Dark GreenLight Green5:3:	n Dark Medium 2	Brown						
20.3.2	The patches should be asymmet	etrical and irregular.							
20.3.3	The patches should be inclin horizontal.	ned at 30 degree to	60 degree to the						
20.3.4	The patches should be continue	ous where two surfaces	meet at an angle.						
20.3.5	The patches should not coincid	e with corners.							
20.3.6	Slits and holes shall be painted	and dark shades.							
20.3.7	Width of patches should be 1 to	o 2 meters.							

MECON LIMITED REGD. OFF: RANCHI	STANDARD TECHNICAL	SPECIFICATION	
834002	OIL & GAS SBU,	य8 मेकान मेकान 9001:2000 Confu	
		DOCUMENT NO.	Page 42 of 54
TITLE	SHOP & FIELD PAINTING	MEC/S/05/21/07	REVISION: 0
			EDITION : 1

### 21.0 **INSPECTION AND TESTING**

- 21.1 All painting materials including primers and thinners brought to site by contractor for application shall be procured directly from manufacturers as per specifications and shall be accompanied by manufacturer's test certificates. Paint formulations without certificates are not acceptable.
- 21.2 Engineer-in-Charge at his discretion, may call for tests for paint formulations. Contractor shall arrange to have such tests performed including batchwise test of wet paints for physical & chemical analysis. All costs there shall be borne by the contractor.
- 21.3 The painting work shall be subject to inspection by Engineer-in-Charge at all times. In particular, following stagewise inspection will be performed and contractor shall offer the work for inspection and approval of every stage before proceeding with the next stage. The record of inspection shall ne maintained in the registers. Stages of inspection are as follows:
  - a. Surface preparation
  - b. Primer application
  - c. Each coat of paint

In addition to above, record should inculde type of shop primer already applied on equipment e. g. Redd oxide zinc chromate or zinc chromate or Red lead primer etc.

Any defect noticed during the various stages of inspection shall be rectified by the contractor to the entire satisfaction of Engineer-in-Charge before proceeding further. Irrespective of the inspection, repair and approval at intermidiate stages of work. Contractor shall be responsible for making good any defects found during final inspection/ guarantee period/ defect liability period as defined in general condition of contract. Dry film thickness (DFT) shall be checked and recorded after application of each coat and extra coat of paint should be applied to make-up the DFT specified without any extra cost to owner, the extra cost should have prior approval of Engineer-in-Charge.

MECON LIMITED	STANDARD TECHNICAL	SPECIFICATION	
834002	OIL & GAS SBU,	00 मेकान 8007:2000 Control	
		DOCUMENT NO.	Page 43 of 54
TITLE	SHOP & FIELD PAINTING	MEC/S/05/21/07	REVISION: 0
			EDITION: 1

### 21.4 **Primer Application**

After surface preparation the primer should be applied to cover the crevices, corners, sharp edges etc. in the presence of inspector nominated by Engineer-in-Charge.

- 21.5 The shades of successive coats should be slightly different in colour in order to ensure application of individual coats, the thickness of each coat and complete coverage should be checked as per provision of this specification. This should be approved by Engineer-in-Charge before application of successive coats.
- 21.6 The contractor shall provide standard thickness measurement instrument with appropriate ranges(s) for measuring.

Dry film thickness of each coat, surface profile guage for checking of surface profile in case of blast cleaning. Holiday directors and pinhole detector and positector whenever required for checking in case of immersion conditions.

- 21.7 Prior to application of paints on surface of chimneys the thickness of the individual coat shall be checked by application of each coat of same paint on M. S test panel. The thickness of paint on test panel shall be determined by using guage such as 'Elkomere'. This thickness of each coat shall be checked as per provision of this specification. This shall be approved by Engineer-in-Charge before application of paints on surface of chimney.
- 21.8 At the discretion of Engineer-in-Charge, the paint manufacturer must provide the expert technical service at site as and when required. This service should be free of cost and without any obligation to the owner, as it would be in the interest of the manufacturer to ensure that both surface preparation and application are carried out as per their recommendations.
- 21.9 Final inspection shall include measurement of paint dry film thickness. Adhesion Holiday detection check of finish and workmanship. The thickness should be measured at as many points/ locations as decided by Engineer-in-Charge and shall be within + 10% of the dry thickness, specified in the specifications.

MECON LIMITED REGD. OFE: RANCHI	STANDARD TECHNICAL	SPECIFICATION	
834002	OIL & GAS SBU	10 मेकान 8007:2000 Control	
		DOCUMENT NO.	Page 44 of 54
TITLE	SHOP & FIELD PAINTING	MEC/S/05/21/07	<b>REVISION: 0</b>
			EDITION: 1

21.10 The contractor shall arrange for spot checking of paint materials for Sp. Gr., flow time (ford cup) and spreading rate.

### 22.0 **GUARANTEE**

- 22.1 The contractor shall guarantee that the chemical and physical properties of paint materials used are in accordance with the specifications contained herein/ to be provided during execution of work.
- 22.2 The contractor shall produce test report from manufacturer regarding the quality of the particular batch of paint supplied. The Engineer-in-Charge shall have the right the test wet samples of paint at random, for quality of same as stipulated in clause 11 above. Batch test report of manufacturer's for each batch paint supplied shall be made available by the contractor.

### 23.0 QUALIFICATION CRITERIA OF PAINTING CONTRACTOR

Painting contractor who is awarded any job for MECON, projects under this standard must have necessary equipments, machinery, tool and tackles for surface preparation, paint application and inspection. The contractor must have qualified trained and experienced surface preparation, paint applicator, inspector, and supervisors. The contractor supervisor, inspector surface perpetrator and paint applicator must be conversant with the standards referred in this specification the contractors capacity, capability and competency requirements for the job shall be quantified in the tender document and shall be assessed by an MECON team before awarding any job.

### 24.0 PROCEDURE FOR APPROVAL OF NEW COATING MATERIALS AND MANUFACTURER'S

Following procedure is recommended to be followed for approval of new manufacturers.

24.1 The manufacturer should arrange testing of the inorganic zinc silicate coating materials as per the list of tests given in para 24.5 below from one of the reputed Government laboratories.

MECON LIMITED REGD. OFE: RANCHI	STANDARD TECHNICAL	STANDARD TECHNICAL SPECIFICATION					
834002	OIL & GAS SBU	100 月前11日 100 月前11日 100 月前11日					
		DOCUMENT NO.	Page 45 of 54				
TITLE	SHOP & FIELD PAINTING	MEC/S/05/21/07	REVISION: 0				
			EDITION: 1				

- 24.2 Samples of coating should be submitted to the Govt. laboratory in sealed containers with batch no. and test certificate on regular format of manufacturer's testing laboratory. The sampling shall be certificate and sealed by a citifying agency.
- 24.3 All test panels should be prepared by govt. testing agency coloured photographs of test panels should be taken before and after the test should be enclosed alongwith test report.

Sample batch. No. and manufacturer's test certificate should ne enclosed alongwith the report. Test reports contain details of observation and rusting if any, as per the testing code. Suggested government laboratories are:

RRL, Hayderabad HBTI, Kanpur DMSRDE, Kanpur IIT, Mumbai BIS Laboratory UDCT, Mumbai RITES, Calcutta PDIL

24.4 Manufacturers should intimate the company, details of sample submitted for testing name of Govt. testing agency, date, contact personnel of the Govt. testing agency. At the end of the test the manufacturer should submit the test report to the company for approval. The manufacturer(s) shall be qualified based on the result of these tests and other assessment and the Company's decision in this regard shall be final and binding on the manufacturer.

MECON LIMITED	STANDARD TECHNICAL S	मिकान 10 3001:2000 Control		
834002	OIL & GAS SBU,			
		DOCUMENT NO.	Page 46 of 54	
TITLE	SHOP & FIELD PAINTING	MEC/S/05/21/07	REVISION: 0	
			EDITION: 1	
24.5	Tests required for evaluation of offshore application.	acceptance of coating I	materials for	
	Test	ASTM Test Method		
	Density	D 1475		
	Dipping properties	D 823		
	Film Characteristics Drying time Flexibility Hardness	D 1640 D 1737/ D 522 D 3363		
	Adhesion Abrasion resistance DFT/ Coat Storage Stability	D 2197 D 968/ D 1044 AS PER SSPC GUID D 1849	ELINES	
	Resistance to Humidity for 2000 hrs. Salt Spray for 2000 hrs. Accelerated Weathering % Zn in DFT	D 2247 B 117 D 822 G 53		
24.6	Coating system for panel tes	t shall be decided aft	er discussion with	

6 Coating system for panel test shall be decided after discussion with MECON.

MECON LIMITED REGD. OFF: RANCHI 834002		STANDARD TECHNICAL	SPECIFICATION	
		OIL & GAS SBU,	, DELHI	मिकान मेकान 12000 Contra
			DOCUMENT NO.	Page 47 of 54
TITLE		SHOP & FIELD PAINTING	MEC/S/05/21/07	<b>REVISION: 0</b>
				EDITION: 1
				ANNEXURE-I
Indiar	vendo	rs		
$\begin{array}{c} 1.0\\ 2.0\\ 3.0\\ 4.0\\ 5.0\\ 6.0\\ 7.0\\ 8.0\\ 9.0\\ 10.0\\ 11.0\\ 12.0\\ 13.0\\ 14.0\\ 15.0\\ 16.0\\ 17.0\end{array}$	Asian F Berger Goodla Jensor Shalim Sigma CDC C Premie Corom Anupar Grand Bomba Vanapi Sunil F Courta Mark-c VCM F	Paints(I) Ltd. Paints Ltd. ass Nerlolac Paints Ltd. And Nicholson Paint Ltd & cho ar Paints Ltd. Coating, Mumabai Carboline Ltd. er Products Ltd. andel Paints & Chemicals Ltd. m Enterprises Polycoats ay Paints Ltd. rabha Esters & Glycer, Mumbai Paints and Varnishes Pvt. Ltd. ulds Coating & Sealants India (I chem Incorporated, Mumbai (for Polyurethane Paint (for polyureth	okuGu Jenson & Nichols Pvt.) Ltd. phosphating chemicals nane Paint only)	on Ltd. only)

- 1.0 Sigma Coating, Singapore
- 2.0 Ameron, USA
- 3.0 Kansai Paint, Japan
- 4.0 Hempel Paint, USA
- 5.0 Valspar Corporation, USA
- 6.0 Courtaulds Coating, UK.
- Note: This list subjected to revision based fresh approval which will be intimated to PDD/ Vendor Cell.

					ANNEXURE-II	F9	PH. INORGANIC ZINC SILICATE PRIMER/ COATING	APCOCIL 605	ZINC ANODE 304	DIMET COTE-9		INC TUFFKOTE ZILICATE TL	SIGMASIL MC (7568)	3 CARBOZINC 11	U17/ 92 ETHYL SILICATE INORGANIC ZINC	3XY -
	1-2000 Contra	48 of 54	0 : NOIS	I 0N : 1	PRODUCTS	P6	EPOXY ZINC PRIMER	APCODUR HB. RO.ZP-PC433	EPILUX 610	AMERCOAT 71	EPILAC ZINC PHOSPHATE PRIMER	EPIGUARD 4 Z PHOSPHATE PRIMER GREY	COLTURE CM PRIMER 7412	CARBOLINE 89	P-15/3A U-16/92	COROPEX EPC ZH. PH. HIGH E PRIMER
NOL		MENT NO. Page	(05/21/07 REVIS	EDIT	ED MANUFACTURER'S F	P4	ETCH PRIMER/ WASH PRIMER	APCONYL WP 636 (PC 335)	BISON WASH PRIMER	AMERCOAT 187	J & N ETCH PRIMER	TUFFKOTE ETC PRIMER	SIGMA COVER PRIMER (7413)			CPC WASH PRIMER
ARD TECHNICAL SPECIFICAL	OIL & GAS SBU, DELHI	DOCU	D PAINTING MEC/S		LIST OF RECOMMENDE	P2	CHLORINATED RUBBER Zp PRIMER	ASIOCHL OR HB. ZN.PH PRIMER RO PC 168	LINSOL HIGH BUILD ZP PRIMER		JENSOLAC CHLORINATED RUBBER HB ZN.PH. PRIMER	CHIOROKOTE ZINC PHOSPHATE PRIMER GREY	SIGMA NUCOL UNICOAT 7321		1	COROCLORE CR HB. ZN. PH. PRIMER
STAND			SHOP & FIEL			ANUFACTURER	NAME	N PAINTS (I) LTD.	GER PAINT LTD.	RON/ GODDLASS DLAC PAINTS	SON & NICHOSON TS LTD. AND KUGU JENSON	LIMAR PAINTS	1A COATING	CARBOLINE LTD.	11 ER PRODUCTS	AMANDEL PAINTS MICALS LTD.
CON LIMITED	002		TITLE			S. M	No.	1. ASIA	2. BAR(	3. AME NER( LTD.	4. JENE PAIN CHOI NICH	5. SHAI LTD.	6. SIGN	7. CDC	8. PRIN LTD.	9. COR. CHEI
ME	834															

Page 532 of 644

					ć	F9 INORGANIC ZINC SILICATE PRIMER/ COATING		1	GALVASOL 1570			INTERZINC			JOTACOTE – 2
	Han-I	19 of 54	10N : 0	ON : 1	ŝ	P6 EPOXY ZINC PH. PRIMER	ANUPAM ANILICOR A-EZP-500		HEMPEL'S SHOP PRIMER E-1530	VEGPOX 1241 Z/ P	SUNPOXY ZINC PHOSPHATE PRIMER	INTERGARD 251			ΕΡΟΧΥ CQ
NOI	1 Con	AENT NO. Page 4	05/21/07 REVIS	EDITI	ž	P4 ETCH PRIMER/ WASH PRIMER	ANUPRIME 291	GP PPRIME 401	PENTOLITE WASH PRIMER 8520	VEG WASH PRIMER 1181	SUN WASH	I			
<b>RD TECHNICAL SPECIFICAT</b>	OIL & GAS SBU, DELHI	DOCUN	D PAINTING MEC/S/		C	P2 CHLORINATED RUBBER Zp PRIMER	ANUCHLOR ZP PRIMER	GP CHILOROPRIME 601	HEMPA TEX HIGHBUILD 4633	VEGCHLOR HB PRIMER 1143	SUNCHLOR HB ZINC PHOSPHATE PRIMER		RUST PREVENTIVE LIQUID DRSAIO		
TED STANDA			SHOP & FIELC			MANUFAG IUKEK NAME	ANUPAM //	GRAND POLYCOATS (	BOMBAY PAINTS	VANAPRABHA ESTERS & GLYCERIDES.	SUNIL PAINTS AND ( VARNISHED PVT. F	COURTAULDS COATING LTD.	MARK-CHEM INCOPORATED, (FOR PHOSPHATING CHEMICAL ONLY)	VCM POLYURETHANE PAINTS (FOR POLY EURETHANE PAINTS ONLY)	JOTUŃ PAINTS
MECON LIMIT RFGD. OFF: RAI	834002		TITLE		(	ю. <mark>о</mark>	10.	11.	12.	13.	.41	15.		Page 533 c	

						F9	INORGANIC ZINC SILICATE PRIMER/	COATING			EZ 180(N)	
		and a state	je 50 of 54	VISION: 0	ITION: 1	P6	H EPOXY ZINC PH. PRIMER		SPECIAL ZINC PHOSPHATE	PRIMER		<b>JDUCTS (Contd)</b>
-	NOIL		MENT NO.	/ <b>05/21/07</b> RE	;/05/21/07 REVIS		ETCH PRIMER/ WASH PRIMER					ANUFACTURER'S PRO
	RD TECHNICAL SPECIFICA	OIL & GAS SBU, DELHI	DOCU	PAINTING MEC/S		P2	CHLORINATED RUBBER Zp PRIMER					T OF RECOMMENDED M/
-				SHOP & FIELD		MANUFACTURER	NAME				C PRODUCTS DREA)	<b>LIS</b> 1
	MECON LIMITED REGD. OFF: RANCH	834002		TITLE		S.	No.				19. KC( (KO	

DAL	0		Ð	COAT
F7 HIGH BUILD C TAR EPOX	APCODUR CF 3(	EPILUX 555	AMERCOAT 78 F	EPILAC SOI VENTI ESS (
F6 HIGH BUILD FINISH PAINT	APCODUR HB COATING 9466	EPILUX 04 AND 78 HB EPOXY COATING	AMER COAT 383 HS	EPILAC 981 ENAMEI
F3 CHLORINATED RUBBER FINISH	ASIOCHLOR CF 621 (PC 161)	LINOSOL CHLORINATED RUBBER HB COATING	AMERCOAT 515	JENSON HB
F2 ACRYLIC-POLY YURETHANE FINISH	APCOTHANE CF76 (PC 1109)	BARGER THANE ENAMEL (81)	AMERCOAT 450GL	J & N 993 HB POI VI IRETHANE
MANUFACTURER NAME	ASIAN PAINTS (I) LTD.	BARGER PAINT LTD.	AMERON/ GODDLASS NEROLAC PAINTS LTD.	JENSON & NICHOSON PAINTS
S. No.	<del>.</del> .	5	с.	4.

MECON I REGD. OF	LIMITED F: RANCHI	STAND	ARD TECHNICAL SP	ECIFICATION	~		
834002			OIL & GAS SBU, D	ELHI		Habit	
				DOCUMEN	T NO. Page	51 of 54	
F	ITLE	SHOP & FIEL	D PAINTING	MEC/S/05/	/21/07 REVIS	0 : NOI	
					EDIT	ION: 1	
C			C L	_	c L	¢ L	11
″ž	o. MAN	NAME	ACRYLIC-PO	۲	r3 CHLORINATED		F/ HIGH BUILD COAL
	;		YURETHANE FI	HSIN	RUBBER FINISH PAINT	FINISH PAINT	TAR EPOXY COATING
	LTD. A JENSC NICHO	ND CHOKUGU JN ILSON	FINISH PAINT.	RU PA	IBBER FINISH INT		TAR EPOXY COATING
£.	SHALI	MAR PAINTS	SHALITHANE FINI	ISH CH	ILORKOTE FINISH	EPIGARD KL	BIPIGARD'S BLACK
	- - -						COATING
.9	SIGM₽	A COATING	SIGMADOUR HS SEMIGLOSS 7530	SIC	3MA NUCOL VISH 7308	SIGMA COVER CM 7456	COLTURIET TCN 300
7.	CDC C LTD.	ARBOLINE	CARBOLINE 132	I		CARBOLINE 191	CARBOMASTIC-14
œ.	PRIME LTD.	ER PRODUCTS	U3/ 92 POLYURE1	THANE CR	R-71 FINISH PAINT	42B/ 4A HIGH BUILD EPOXY	350B/ 3A, COAL TAR EPOXY COATING
<u>ю</u>	CORA	MANDEL S CHEMICALS	I	CO	DROCLORE CR	COROPEX EPOXY HB COATING	COROPEX EPOXY COAL TAR COATING
10	. ANUP/ ENTEF	AM RPRISES	ANUTHANE ENAM	AEL AN EN	IUCHLOR HB AMEL	DURACOAT-6000	COROGUARD
11	. GRAN	D POLYCOATS	GP COAT 131, 132 GP BOND 141	2 63.	CHILOROGAURD	GP GUARD HP 234	POLYGUARD GE

MECON LIMITED	STANDARD TECHNICAL	SPECIFICATION	(
834002	OIL & GAS SBU	, DELHI	to Hant
		DOCUMENT NO.	Page 52 of 54
TITLE	SHOP & FIELD PAINTING	MEC/S/05/21/07	REVISION: 0
			EDITION: 1

## LIST OF RECOMMENDED MANUFACTURER'S PRODUCTS

NO.     NAME       12.     BOMBAY PAINTS     PENTAT       13.     LTD./ PAINTS     PENTAT       13.     VANAPRABHA     VEGTHA       14.     SUNIL PAINTS AND     SUNTHA       14.     SUNIL PAINTS AND     SUNTHA       15.     COURTAULDS     INTERTHA       15.     COURTAULDS     INTERTHA       16.     MARK-CHEM     INTERTHANE       16.     MARK-CHEM     INTERTHANE       17.     VCM     POLYURETHANE       17.     VCM     POLYURETHANE       17.     VCM     POLYURETHANE       18.     JOTUN PAINTS     POLYUF       18.     JOTUN PAINTS     POLYUF	ANUFACTURER'S	F2	F3	FG	F7
12.       BOMBAY PAINTS       PENTAT         LTD./ PAINTS       LTD./ PAINTS       PENTAT         13.       VANAPRABHA       VEGTHA         13.       VANAPRABHA       VEGTHA         13.       VANAPRABHA       VEGTHA         13.       VANAPRABHA       VEGTHA         13.       VANNSHED PVT.       NUTHA         14.       SUNIL PAINTS AND       SUNTHA         14.       SUNIL PAINTS AND       SUNTHA         15.       COURTAULDS       INTERTI         15.       COURTAULDS       INTERTI         16.       MARK-CHEM       INTERTI         17.       VCM       POLYURETHANE         POLYURETHANE       POLYUE       POLYUE         17.       VCM       POLYUE         18.	NAME	l	6		
LTD./ PAINTS     LTD./ PAINTS       13. VANAPRABHA     VEGTHA       13. VANAPRABHA     VEGTHA       ESTERS &     ESTERS &       GLYCERIDES,     VANISHED PVT.       14. SUNIL PAINTS AND     SUNTHA       VARNISHED PVT.     ITD.       15. COURTAULDS     INTERTI       16. MARK-CHEM     INTERTI       16. MARK-CHEM     INTERTI       17. VCM     POLYURETHANE       PAINTS (FOR POLY     POLYUF       POLYURETHANE     PAINTS (FOR POLY       18. JOTUN PAINTS     HARDTG	DMBAY PAINTS	PENTATHANE FP 4510	HEMPATEX HIBUILD	HEMPADUR HIGH	HEMPADUR 1510
13.       VANAPRABHA       VEGTHA         ESTERS &       ESTERS &         GLYCERIDES,       SUNTHA         14.       SUNIL PAINTS AND       SUNTHA         14.       SUNIL PAINTS AND       SUNTHA         15.       COURTAULDS       INTERTH         16.       MARK-CHEM       INTERTH         16.       MARK-CHEM       INTERTH         17.       VCM       PAINTS ONLY)         17.       VCM       ALIPHA         PAINTS (FOR PHOSPHATE       ALIPHA         PAINTS ONLY)       POLYUF         13.       VCM       POLYUF         14.       JOTUN PAINTS       HARDT(	D./ PAINTS		4633	BUILD 5520	
ESTERS & GLYCERIDES, GLYCERIDES, SUNTHA VARNISHED PVT. LTD. 15. COURTAULDS INTERTI 15. COURTAULDS INTERTI 15. COURTAULDS INTERTI 15. COURTAULDS INTERTI 15. COURTAULDS INTERTI 16. MARK-CHEM INCOPORATED, (FOR PHOSPHATE PAINTS ONLY) 17. VCM PAINTS ONLY) PIPCOT POLYURETHANE PAINTS (FOR POLY POLYUF EURETHANE PAINTS II. VCM 18. JOTUN PAINTS HARDTC	<b>NAPRABHA</b>	VEGTHANE FP 3641	VEGCHLOR FP 3140	VEGPOX- 3265	VEGPOX 4265
GLYCERIDES,       14.     SUNIL PAINTS AND       VARNISHED PVT.       VARNISHED PVT.       UTD.       15.       COURTAULDS       16.       MARK-CHEM       INCOPORATED,       (FOR PHOSPHATE       PAINTS ONLY)       17.       VCM       PAINTS ONLY)       POLYURETHANE       PAINTS (FOR POLY       POLYURETHANE       PAINTS (FOR POLY       PONLY)       18.       JOTUN PAINTS       18.       JOTUN PAINTS	STERS &			VEGPOX 3562	
14.       SUNIL PAINTS AND       SUNTHA         VARNISHED PVT.       VARNISHED PVT.       ITD.         15.       COURTAULDS       INTERTI         16.       MARK-CHEM       INTERTI         16.       MARK-CHEM       INTERTI         16.       MARK-CHEM       INTERTI         17.       VCM       PIOSPHATE         17.       VCM       PIPCOT         PAINTS ONLY)       PINCOT       PIPCOT         17.       VCM       PINCOT         18.       JOTUN PAINTS       FINISH F         18.       JOTUN PAINTS       HARDTC	-YCERIDES,				
VARNISHED PVT. LTD. 15. COURTAULDS INTERTI COATING LTD. 16. MARK-CHEM INCOPORATED, (FOR PHOSPHATE PAINTS ONLY) 17. VCM PAINTS ONLY) 17. VCM POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURET	JNIL PAINTS AND	SUNTHANE (ALIPHATIC)	SUNCHLOR HB CR	Sd, BH YXOQL	LPOXY BLACK P. S.
LTD. 15. COURTAULDS INTERTH COATING LTD. 16. MARK-CHEM INCOPORATED, (FOR PHOSPHATE PAINTS ONLY) 17. VCM PAINTS ONLY) 17. VCM PAINTS (FOR POLY POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POL	ARNISHED PVT.		COATING	901'	551
15.       COURTAULDS       INTERTH         16.       MARK-CHEM       INTERTH         16.       MARK-CHEM       INCOPORATED,         16.       MARK-CHEM       INCOPORATED,         16.       MARK-CHEM       INCOPORATED,         16.       MARK-CHEM       INCOPORATED,         17.       VCM       PIPCOT         17.       VCM       PIPCOT         POLYURETHANE       ALIPHA         PAINTS (FOR POLY       POLYUF         PONLY)       POLYUF         18.       JOTUN PAINTS         18.       JOTUN PAINTS	D.				
COATING LTD. 16. MARK-CHEM INCOPORATED, (FOR PHOSPHATE PAINTS ONLY) 17. VCM PAINTS ONLY) PIPCOT PAINTS (FOR POLY POLYUE POLYUETHANE POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF POLYUF	DURTAULDS	INTERTHANE		INTEGARD EM	INTERTUF JXA 006/
16. MARK-CHEM INCOPORATED, (FOR PHOSPHATE PAINTS ONLY) 17. VCM POLYURETHANE POLYURETHANE POLYURETHANE POLYURETHANE POLYUF FINISH FINISH	DATING LTD.			SERIES	007/010
INCOPORATED, (FOR PHOSPHATE PAINTS ONLY) 17. VCM POLYURETHANE PAINTS (FOR POLY PAINTS (FOR POLY POLYUF EURETHANE PAINTS (FOR POLY FINISH FINISH	ARK-CHEM				
(FOR PHOSPHATE PAINTS ONLY) 17. VCM POLYURETHANE POLYURETHANE ALIPHAT POLYUF FOR POLY POLYUF EURETHANE PAINTS 18. JOTUN PAINTS HARDTC	COPORATED,				
PAINTS ONLY)       17. VCM       POLYURETHANE	OR PHOSPHATE				
17. VCM POLYURETHANE ALIPHAT POLYURETHANE ALIPHAT PAINTS (FOR POLY POLYUF EURETHANE PAINTS FINISH F ONLY) 18. JOTUN PAINTS HARDTC	AINTS ONLY)				
POLYURETHANE ALIPHAT PAINTS (FOR POLY EURETHANE PAINTS FINISH F ONLY) 18. JOTUN PAINTS HARDTC	SM	PIPCOTHANE			
PAINTS (FOR POLY POLYUF EURETHANE PAINTS FINISH F ONLY) 18. JOTUN PAINTS HARDTC	DLYURETHANE	ALIPHATIC			
EURETHANE PAINTS FINISH F ONLY) 18. JOTUN PAINTS HARDTC	AINTS (FOR POLY	POLYURETHANE			
18. JOTUN PAINTS HARDTC	JRETHANE PAINTS	FINISH PAINT			
18. JOTUN PAINTS HARDTC	NLY)				
	TUN PAINTS	HARDTOP AS		PENGUARD	JOTAGUARD 85
	CC PRODUCTS			KOPOX TOPCOAT HB FT 5740	EH 173

	and the second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second s	Page 53 of 54	REVISION: 0	EDITION: 1
SPECIFICATION	, DELHI	DOCUMENT NO.	MEC/S/05/21/07	
STANDARD TECHNICAL	OIL & GAS SBU		SHOP & FIELD PAINTING	
MECON LIMITED	834002		TITLE	

## 6 Ľ C ŀ 2

		<u> </u>			Т	1		1	<u> </u>	
F-12 HEAT RESISTANCE SILICON AL. PAINT	HR SILICON ALUMINUM PAINT (PC 189)	BARGER HEAT RISISTANT SILICON ALUMINUM PAINT	AMERCOAT 878	FERRLOTECT SILICON HEAT RESISTANCE 1000	LUSTOTHERM HIGH TEMP ALUMINUM PAINT	AROSTA FINISH HR	CARBOLINE 4674		CPC SILICONE HR ALUMINUM PAINT	ANUPAM HEAT GUARD
F-11 HEAT RESISTANCE SYNTHETIC MEDIUM ALUMINUM PAINT	ASIAN HR ALUMINUM PAINT (PC 300)	FERROLOT HR ALUMINUM PANT		FERROTECT SYNTHETIC RUBBER H/R ALUMINUM PAINT 4000	HEAT RESISTING LUSTROL ALUMINUM	HIGH TEMPERATURE RESISTANT EPOXY SUSTEM UPTO 200° C 4062	CARBOLINE 1248		SILVOTOL HR ALUMINUM PAINT	
F-8 EPOXY MASTIC COATING SURFACE TOLERANT	APCODOR CF 640	PROTECTOMASTIC	AMERLOCK 400		EPIPLUS 56	SIGMA ETPC ALUMINUM	CARBOMASTIC-15	HB EPOXY MATIC 150B/ 150A	1	ANUMASTIC-102
ME				ĺ	1	1	1			
MANUFACTURER'S NA	ASIAN PAINTS (I) LTD.	BARGER PAINT LTD.	AMERON/ GODDLASS NEROLAC PAINTS LTD.	JENSON & NICHOSON PAINTS LTD. AND CHOKUGU JENSON NICHOLSON	SHALIMAR PAINTS LTD.	SIGMA COATING	CDC CARBOLINE LTD.	PRIMER PRODUCTS LTD.	CORAMANDEL PAINTS CHEMICALS	ANUPAM ENTERPRISES

MECON LIMITED	STANDARD TECHNICAL	SPECIFICATION	
834002	OIL & GAS SBU	, DELHI	and the second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second s
		DOCUMENT NO.	Page 54 of 54
TITLE	SHOP & FIELD PAINTING	MEC/S/05/21/07	REVISION: 0
			EDITION: 1

## LIST OF RECOMMENDED MANUFACTURER'S PRODUCTS

s. NO.	MANUFACTURER'S NAME	F8	F11	F12
11.	GRAND POLYCOATS	GP PRIME GUARD 235		
12.	BOMBAY PAINTS LTD./ HEMPEL MARINE PAINTS	HEMPADUR 1708	KANGAROO HHR ALUMINUM 4950	HEMPADUR HIGH BUILD 5520
13.	VANAPRABHA ESTERS & GLYCERIDES,	VEGEPOX MASTIC 2255	VEG HR AL PAINT TO IS211339	VEG HHR AL PAINT TO 600°C
14.	SUNIL PAINTS AND VARNISHED PVT. LTD.	P.S.901	-	
15.	COURTAULDS COATING LTD.	INTERPLUX	-	INTERTHERM 50
16.	MARK-CHEM INCOPORATED, (FOR PHOSPHATE PAINTS ONLY)			
17.	VCM POLYURETHANE PAINTS (FOR POLY EURETHANE PAINTS ONLY)	1		
18.	JOTUN PAINTS	JOTUMATIC 87		SOLVELITT HEAT RESISTANT SILICON PAINT
19.	KCC PRODUCTS (KOREA)	EH 4158H		QT 606

Page 538 of 644

### **SPECIFICATION**

### FOR

### **INSTALLATION OF INSTRUMENTS**

SPECIFICATION NO.: MEC/S/05/26/01



ELECTRICAL & INSTRUMENTATION (OIL & GAS SBU) MECON LIMITED DELHI 110 092

MECON LIMITED	STANDARD SPECI	FICATION				
REGD. OFF: RANCHI 834002	ELECTRICAL & INSTRU	UMENTATION	मेकान 🔉			
031002	OIL & GAS SBU,	DELHI	9001:2000 Comps			
	INSTALLATION OF	DOCUMENT NO.	Page 1 of 21			
TITLE	INSTRUMENTS	MEC/S/05/26/01	REVISION: 0			
			EDITION: 1			

### **AMENDMENT STATUS**

SI. No.	Clause / Paragraph / Annexure / Exhibit / Drawing Amended	Page No.	Revision	Date	By (Name)	Verified (Name)
MECON LIMITED	STANDARD SPECI	FICATION				
-----------------------------	------------------------------------------------------	----------------	----------------------------------			
REGD. OFF: RANCHI 834002	ELECTRICAL & INSTRUMENTATION OIL & GAS SBU, DELHI		मुन्न मेकान 9007:2000 Company			
	INSTALLATION OF	DOCUMENT NO.	Page 2 of 21			
TITLE	INSTRUMENTS	MEC/S/05/26/01	REVISION : 0			
			EDITION: 1			
	CONTENTS					

- 1) SCOPE
- 2) STANDARDS OF MATERIALS
- 3) INSTALLATION OF INSTRUMENTS
- 4) TESTING
- 5) CALIBRATION OF INSTRUMENTS

PREPARED BY:	CHECKED BY:	APPROVED BY:	ISSUE DATE :
(R. SANJAY BABU )	(RAKESH SHUKLA)	(PANKAJ SHIVASTAVA)	08 <sup>Th</sup> DEC 08
		Pag	e 541 of 644

MECON LIMITED	STANDARD SPECIFICATION		
REGD. OFF: RANCHI 834002	ELECTRICAL & INSTR OIL & GAS SBU	RUMENTATION J, DELHI	क मेकान 9007:2000 Contrint
	INSTALLATION OF	DOCUMENT NO.	Page 3 of 21
TITLE	INSTRUMENTS	MEC/S/05/26/01	REVISION : 0
			EDITION: 1

### INSTALLATION, TESTING AND CALIBRATION OF INSTRUMENTATION AND CONTROL SYSTEM

#### 1.0 SCOPE

- 1.1 The purpose of this specification is to define the general requirements for the installation, installation materials, testing and calibration of instruments and control system.
- 1.2 The work shall be carried out in accordance with the codes, standards and recommended practice listed in this specification and in accordance with local `Statutory regulations'.
- 1.3 For installation of instruments and control system, of the new material where quality is of the prescribed standards and which is in every way fit for its intended purpose shall be used.
- 1.4 Unless otherwise specified all the materials shall be indicated in this specification except where it is not compatible with fluids being handled. In such cases the selection of the material shall be approved by MECON.
- 1.5 Only the best trade practices shall be used. All the work shall be carried out in a neat, workman like manner and to the satisfaction of MECON.

### 2.0 STANDARDS OF MATERIALS

- 2.1 Instrument process piping / tubing upto and including the first block valve and `in-line' instrument equipment shall conform to the line class or vessel rating concerned instrument piping or tubing after the first lock valve may use alternate materials consistent with service conditions. In general they shall conform to the following specification as a minimum.
- 2.1.1 Stainless tubes shall be fully annealed and cold drawn seam less as per ASTM A 269 TP316 with size 1/2"OD x 0.65" WT (wall thickness).
- 2.1.2 Monel tubing shall be fully annealed seamless as per ASTM B165 with size 1/2" OD x 0.35"WT.
- 2.1.3 Carbon steel pipe shall be 1/2" seamless and shall be as per ASTM A106 Gr B min of sch 80 & dimensions as per ANSI B36.10.
- 2.1.4 Seamless stainless steel pipes shall be as per ASTM A 312 Gr TP 316L Sch 80S, dimensions as per ANSI B 36.19.

Page 542 of 644

MECON LIMITED	STANDARD SPEC	STANDARD SPECIFICATION ELECTRICAL & INSTRUMENTATION OIL & GAS SBU, DELHI	
REGD. OFF: RANCH 834002	I ELECTRICAL & INST		
	INSTALLATION OF	DOCUMENT NO.	Page 4 of 21
TITLE	INSTRUMENTS	MEC/S/05/26/01	REVISION : 0
			EDITION : 1
2.1.5 r	Instrument air supply piping from th neavy class pipes to IS 1239.	ne main instrument air hea	der shall be galvanised
2.2 li s	ndividual pneumatic signal and a pecifications:	air supply tubing shall co	nform to the following
2.2.1	Stainless tubes shall be used in ge eamless as per ASTM A269 TP 31	eneral and shall be fully an 6 with 6mmOD x 1mmWT	nealed and cold drawn
2.2.2 C	Copper tubing where specified s innealed as per ASTM 868.74a cd coloured Black.	shall be seamless 6mm No. 122 (DHP) sheathed	OD x 1.0mmWT soft with PVC 1.0mm thick
2.3 A s 1	All fittings shall be as a minimum of 100 rating except for tube fittings. The fittings shall have threading as per B2.1 and socket weld connections as per B 16.11. These shall conform to the following specifications in general.		
2.3.1 s	Tube fittings shall be flare type compression fittings double ferrule and pressure at type.		
P	All tube fittings in impulse lines shall be rated to 5000 PSIG at 38°C.		
2.3.2 C	Carbon steel pipe fittings shall be ittings shall be as per ASTM –182 (	forged as per ASTM A1 Grf 316L	05 stainless steel pipe
2.4 V r 8	Valve shall have normally Globe be ated to min. of 1500. These shall plug shall be integral with the ste 30mm. End connections shall be so	ody and shall be fabricate be screwed bonnet type v m. Face to face dimens ocket weld to ANSI 16.11 a	d out of Bar-stock and vith 13% GSS trim and ions shall be approx. nd threaded to B2.1
2.5 M 1 c	Aultibore tubing shall have a max mm numbered for easy identificat puter fire resistance PVC sheath. liameter flexible.	imum 19 single polyethyle ion. The bundle shall be They shall carry a pair of	ene tubes, 6mmOD x marked with inner and telephone wire 0.6mm
2.6 S c F	Single pair and multi pair extension alibrated in accordance with ISA M pair and 20 A for Multipair.	cables for Thermocouples IC 96.1. Conductor size s	shall be matched and hall be AWG for single
T a b	The cable shall be armoured, eac aluminium Mylar tape and a tinned be colour coded as per ISA recomm	h twisted pair shall be ind copper drain wire. The wi ended practices.	dividually shielded with res and the cable shall

MECON LIMITED	STANDARD SPECIFICATION		
REGD. OFF: RANCHI 834002	ELECTRICAL & INSTR OIL & GAS SBU	RUMENTATION J, DELHI	मेकान मेकान 9007:2000 Comman
	INSTALLATION OF	DOCUMENT NO.	Page 5 of 21
TITLE	INSTRUMENTS	MEC/S/05/26/01	REVISION: 0
			EDITION: 1

- 2.7 Instrument Electrical cables shall conform to the following specifications:
- 2.7.1 Instrument electronic signal cables single pair/ Multipair shall have copper conductor, twisted in pair and individually shielded with Aluminium Mylar tape with drain wire. In multipair cables, each pair shall be armoured with inner and cut PVC sheath. Minimum conductor size shall be 1.5 mm<sup>2</sup>.
- 2.7.2 Control Cables for control signal, alarms actuating devices and solenoid valves of the interlock and shutdown valves shall generally be 1.5 mm<sup>2</sup> copper conductors armoured with inner and cut PVC sheath.
- 2.7.3 All power supply cables shall have copper/Aluminium conductor depending upon the conductor size. The cables shall be armoured with inner and cut PVC sheath. The cables shall be sized adequately. Minimum conductor size shall be 2.5 mm<sup>2</sup>.
- 2.7.4 2-core armoured cable shall be used for illuminator on level gauges.
- 2.7.5 The material and construction of all electrical cables shall conform to IS- 1554 Part I or appropriate equivalent code and standard.

#### 3.0 INSTALLATION OF INSTRUMENTS

- 3.1 Instrument Mounting
- 3.1.1 No instrument shall be installed in such a way that it bends for support on the impulse piping or electrical connection on it.
- 3.1.2 Pressure gauges and temperature indicator shall normally be mounted directly on line. However direct on line mounting shall be avoided where vibrations are likely to be present.
- 3.1.3 Local mounted instruments shall be mounted on brackets, panels or placed on a suitable pedestal. Transmitters shall be mounted on 2" pipe supports where practical. Instruments to be mounted on steel columns, masonary structure etc. These shall not be mounted on heating equipments, pipelines and structures.
- 3.1.4 Blind transmitters shall be mounted at 130mm above graded platform. Local controllers, indicating transmitters and indicating instruments shall be mounted at approximately 1500 mm.
- 3.1.5 All the instruments shall be accessible from grade, ladder or platform etc. Pressures gauges and other local indicating instruments shall be readable from grade or operating level and if used for manual control shall be visible from the related valve.

MECON LIMIT	ED	STANDARD SPECIFICATION		
REGD. OFF: RAN 834002	ICHI	ELECTRICAL & INSTR	ELECTRICAL & INSTRUMENTATION	
		OIL & GAS SBU	J, DELHI	1:2000 Col
		INSTALLATION OF INSTRUMENTS	DOCUMENT NO.	Page 6 of 21
IIILE			MEC/S/05/26/01	REVISION : 0
	All ope	the instruments shall be locat ration.	ied such that they don't	impede the proces
3.1.6	Loc be r	al mounted instruments which a mounted inside a weather proof o	are not available in weath case.	er proof housing sha
3.1.7	lten app	ns such as pilot valves, solenoic lication or near to the device bei	l valves etc. shall be loca ng actuated by them.	ted local to its point o
3.1.8	For	blind transmitters output meters	shall be mounted on instru	ument supports.
3.1.9	Filte tran	er regulators shall be mounted smitter or on the control valve yo	on the instrument supp bke.	orts below pneumati
3.1.10	Inst	ruments or instrument lines shall	l not be supported on hand	l rails, in general.
3.1.11	The pos	e use of process piping to supp sible.	oort instrument lines shall	be avoided as far a
3.1.12	The	instrument impulse piping shall	be kept as short as possib	le.
3.1.13	Inst	ruments and impulse lines shall	be protected against mech	anical damage.
3.1.14	In c aga	ase of capillary tube instruments inst mechanical damage.	s, capillary tube is to be su	pported and protecte
3.1.15	Orif orifi	ice meters shall not be installed ce pressure taps shall be located	on the top of orifice fitting d as follows:	s. On horizontal line
	a) b)	On top for air and gas service Horizontal for liquid and conden	sible vapour service.	
3.2	Inst	rument Piping & Tubing.		
3.2.1	<u>Imp</u>	ulse Piping/tubing		
3.2.1.1	The spe	e primary instrument block val cifications.	ves for all instruments s	nall be as per pipin
3.2.1.23-	Valve mar fittir	e manifold in general shall be nifolds shall also be acceptable ngs.	e integral type. For pres instead of isolation valve	sure gauges, 2-valv , drain valve and pip
3.2.1.3Di	ifferer for o	ntial or static pressure sensing li direct connected or locally moun	nes shall not exceed 6 mt ted instruments.	rs. (20 feet) in gener

Page 545 of 644

MECON LIMITED	STANDARD SPECIFICATION		
REGD. OFF: RANCHI 834002	ELECTRICAL & INST OIL & GAS SB	RUMENTATION J, DELHI	क मेकान 9007:2000 Contrast
	INSTALLATION OF	DOCUMENT NO.	Page 7 of 21
TITLE	INSTRUMENTS	MEC/S/05/26/01	REVISION: 0
			EDITION: 1

- 3.2.1.4 All impulse lines shall be run with a slope not less than 1 in 12 except where otherwise specified. Direction of slope is to be downward from the process for liquid service and upward from the process for gas service.
- 3.2.1.5 Tubing shall be joined by compression fittings.
- 3.2.1.6 Piping shall be joined by pipe fittings/flanges as per the piping specifications.
- 3.2.1.7 All instruments pipes and tubes shall run in horizontal and vertical planes only and shall run with minimum number of changes in direction, consistent with good engineering practices and neat appearance.
- 3.2.1.8 Tubing shall be bent with correct size tube bender as far as possible to avoid use of fittings. Hot bending shall be totally avoided.

Tube cutter shall always be used to cut tubing. The use of short lengths of tubing in long runs shall be avoided in order to avoid the fittings.

- 3.2.1.9 All tubing shall run in such a manner as to give the maximum protection against mechanical damage. Tubing runs shall be grouped together and clamped.
- 3.2.1.10 Tubing shall be arranged so that the unions can be tightened without distorting lines.
- 3.2.1.11 Instrument tubing or piping shall not run on trays intended for cables and shall not share the same transit.
- 3.2.1.12 No pipe or tube shall be left with mechanical strain on them.
- 3.2.1.13 A mechanical ferrule seater shall be used on tubing for 140 kg/cm<sup>2</sup> (2000 psi) or more.
- 3.2.1.14 Pipe bushings shall not be used.
- 3.2.1.15 Pipe plugs shall be fabricated out of bar stock and shall have hex-head.
- 3.2.2 <u>Air/Signal Tubing</u>
- 3.2.2.1 Signal Transmission tubes shall be laid on perforated trays prefabricated out of min 2.5 mm. thick steel plates. The width of the tray shall be selected as per the number of tubes to be laid.
- 3.2.2.2 Where tubing is run in permanent enclosures, it should be ensured that entry and exit of such enclosures is clean and smooth.
- 3.2.2.3 Tubing run in permanent enclosures shall not have joints, except et special in stigans

MECON LIMITED STANDARD SPECIFICATION			
REGD. OFF: RANCHI 834002	ELECTRICAL & INSTR OIL & GAS SBI		मि मेकान 9007:2000 Company
	INSTALLATION OF	DOCUMENT NO.	Page 8 of 21
TITLE	INSTRUMENTS	MEC/S/05/26/01	REVISION: 0
			EDITION: 1

boxes provided for this purpose.

- 3.2.2.4 Where permanent enclosures are left with space for instrument tubing to be laid at some later date, a galvanised pull wire of adequate size shall be left in the tray.
- 3.2.2.5 Where the length of transmission tubing exceeds 60 mtrs (200ft) necessity of installing signal booster relays shall be considered.
- 3.2.2.6 In case of `Skidded' equipment or vessels with instrumentation, where off- skid alarms shutdown or control functions are provided the signal tubes shall be terminated on the control bulk head near the skid boundary.
- 3.2.3 All threaded pipe joints shall be joined after applying Teflon tape. It should be applied in a manner to ensure that the tape does not spill over the end of the male fitting. No other pipe joining compound shall be used except on high temperature service where graphite sealing compounds shall be used.
- 3.2.4 All reasonable precautions shall be taken to prevent foreign materials entering pipe lines or tubing before and during erection.
- 3.2.5 Pipes and tubes installed but not connected, shall have the ends clad in approved fashion to prevent the entry of foreign material. For a period upto one week adhesive tape may be used, for longer periods, caps or plugs shall be used.
- 3.2.6 <u>Piping/Tubing supports</u>
- 3.2.6.1 Piping and tubing shall be adequately supported and fixed at a distance not exceeding that in the following table:

Single tubing/Piping	Max. distance between supports
3/8" OD or less	Continuous
1/2" to 3/4" Nom. size	2 meters (6ft.)
3/4" to 1" Nom. size	3 meters (9ft.)
Multitube bundle	3 meters (9ft)

MECON LIMITED	STANDARD SPECIFICATION		
REGD. OFF: RANCHI 834002	ELECTRICAL & INSTR OIL & GAS SBU	RUMENTATION J, DELHI	मेकान 9007:2000 Comman
	INSTALLATION OF	DOCUMENT NO.	Page 9 of 21
TITLE	INSTRUMENTS	MEC/S/05/26/01	REVISION: 0
			EDITION: 1

3.2.6.2All field mounted instrument air tubing shall be supported with galvanised steel angles or channels of minimum 1/8" thickness fabricated to present neat appearance.

- 3.2.6.3 All instruments tubing supports shall be galvanised prior to installation
- 3.2.6.4 Trays shall be properly supported either from any rigid steel structure or concrete In case of non-availability of above, a suitable support shall be member. fabricated.
- 3.3 Instrument Air Supply Distribution
- 3.3.1 Piping material for instrument main and branched air headers up to the isolation valve at each take-off from main or branch header shall conform to piping specification.
- 3.3.2 The air header size shall be established in accordance with the table below, unless otherwise specified, for a header pressure of 4 to 8.5 kg/cm<sup>2</sup>

Max number of users	Nominal pipe size	
upto 5 upto 10 upto 25 upto 80 upto 150 upto 500	1/2" 3/4" 1" 1-1/2" 2" 3"	

Table \_ 2

- 3.2.3 All take off for branch lines are to be from the top of the main header with block valves equal in size to the branch line. All low point shall have a 1/2" valve installed as a drain and blow down point.
- 3.3.4 A minimum size of  $\frac{1}{2}$  pipe shall be run to the instrument with a  $\frac{1}{2}$  value for each user. Tubing from the isolation valve to the instrument shall be 6.0 mm.
- 3.2.5 Union shall be provided at convenient location in the air header.
- 3.3.6 Filter regulator shall be provided for individual field mounted consumer and shall be complete with an output gauge.
- 3.3.7 In case of skid mounted equipments or vessels which incorporate instrumentation requiring pneumatic supply, on skid supply piping shall terminate at the skid boundary location and size of the supply connections shall be noted on the vendor approval drawings.

MECON LIMITED	STANDARD SPECIFICATION		
REGD. OFF: RANCHI 834002	ELECTRICAL & INSTR OIL & GAS SBU	RUMENTATION J, DELHI	वि विवान 12000 Company
	INSTALLATION OF	DOCUMENT NO.	Page 10 of 21
TITLE	INSTRUMENTS	MEC/S/05/26/01	REVISION: 0
			EDITION: 1

- 3.4 Installation of multitude and Multicore cables.
- 3.4.1 Multicore/ Multitube cables shall generally be installed on trays or ducts and properly clamped. At bends minimum radius shall be maintained as per cable manufacturer's standards.
- 3.4.2 All cables shall be rigidly supported on structural steel and masonary. Drilling of steel member should normally be avoided. However, if the drilling of steel must be resorted to, it must be drilled where minimum of weakening of structure will result cables shall be support at every 500 mm. At every vertical drop these shall be clamped at more frequent intervals max of 300 mm.
- 3.4.3 Directly buried cables shall be laid underground in excavated cable trenches. Trenches shall have sufficient depth and width to accommodate all cables correctly spaced and arranged with a view of heat dissipation and economy of design construction of trenches laying of cables and filling up of trenches shall be as per relevant standard.
- 3.4.4 Each underground cable shall be provided with identifying tag of load securely fastened every 30 M of its underground length with at least one tag at each end before the cable enters the ground.

Before cables are placed, the trench bottom shall be filled with a layer of sand. The cables shall be covered with 150 mm of sand on the top of the largest dia. cable tube and sand shall be lightly pressed. A protective covering of 75 mm thick second class red bricks shall be laid flat and the balance portion of the trench shall be filled with soil, compacted and levelled.

- 3.4.5 At each road crossing and other places where cables enter pipe sleeves, adequate bed of sand shall be given so that the cables don't slack and get damaged by pipe ends after back filling.
- 3.4.6 At the entry into concrete blocks loops shall be provided at either end to prevent any damage to cable.
- 3.4.7 The cable entry to control room shall be suitably filled and sealed after laying of cables so as to achieve a positive sealing against the entry of gas/water.
- 3.4.8 All wiring, tubing, cables, Junctions boxes and auxiliary equivalent shall be suitably identified as per applicable codes and practices. All piping and tubing shall be tagged with slip-on or clip on wire marker at both ends.
- 3.4.9 Jointing of cables is generally not permitted. Cables shall be cut after the exact site measurements at the cable drums shall be so selected before cutting the lengths as to avoid any unnecessary wastage.

Page 549 of 644

MECON LIMITED	STANDARD SPEC		
REGD. OFF: RANCHI 834002	ELECTRICAL & INSTRUMENTATION OIL & GAS SBU, DELHI		10 मकान 9001:2000 Control
	INSTALLATION OF	DOCUMENT NO.	Page 11 of 21
TITLE	INSTRUMENTS	MEC/S/05/26/01	REVISION : 0
			EDITION: 1

- 3.4.10 Low signal cables like alarms, analysers cables, special cables for turbine meter, thermocouple compensating cables etc. shall be layed separated from power supply cables in ducts/trenches/trays.
- 3.4.11 Electric signal lines for electronic transmitters to receive and to final control element shall be continuously shielded with the shield grounded at the same point as the signal circuit generally at the control instrument.
- 3.4.12 Separate junction boxes shall be used for intrinsically sage cables.
- 3.4.13 Different intrinsically safe system e.g., systems having different rounds shall not be run in the same multicore cable, in general.

Recommended minimum separation distance between twisted pair signal leads and AC Power Lines.

AC Power Cable			Minimum Distance to Signal Lead	
\	Voltage (Volts)	Current (Am) in (cm)	)	
C	0 to 125	0 to 10	12" (30)	
1	125 to 250	0 to 50	15" (38)	
2	250 to 440	0 to 200	18" (46)	
5	5KV & Up	200 Amp. & Up	24" (61)	

Different intrinsically safe circuits e.g., circuits having different voltage levels, of the same intrinsically safe system shall not be run in the same cable unless each conductor insulation is at least 0.25mm or no hazard can result from interconnection.

- 3.4.14 The physical separation of power and signal cables shall be as per API 550 Part I Section VII. Cable in intrinsically safe circuits shall preferably be not run in the same tray where-- on intrinsically safe circuits cables are being run. If these are being run in the same tray, a metallic earthed separately shall be provided.
- 3.4.15 For temperature controllers, single pair thermocouple extension cable or cable for resistance thermometer, shall be layed directly from the element to the transducer in the control room without intermediate terminal blocks.

MECON LIMITED	MECON LIMITED REGD. OFF: RANCHI 834002STANDARD SPECIFICATIONREGD. OFF: RANCHI 001 & GAS SBU, DELHIELECTRICAL & INSTRUMENTATION OIL & GAS SBU, DELHI		
REGD. OFF: RANCHI 834002			मुकान मकान 8007:2000 Compart
	INSTALLATION OF	DOCUMENT NO.	Page 12 of 21
TITLE	INSTRUMENTS	MEC/S/05/26/01	REVISION: 0
			EDITION: 1

- 3.4.16 In case of skid mounted equipment or equipment which incorporate skid instrumentation like alarms, shutdown or control function shall terminate signals or control junction box near skid boundary for connection of off skid equipment.
- 3.4.17 No wire shall be terminated or left with mechanical strain within any conductor.
- 3.4.18 Splices shall be made only at terminals, in instruments or approval equipment/ junction boxes using lugs and screwed connections. No intermediate splices shall be made in cable trays or in conduct. Number of junction boxes in any cable path shall be limited to only one.
- 3.5 Installation of Zener barriers
- 3.5.1 Zener barriers shall be installed in the circuit to make the system intrinsically safe provided:
  - a) There is no energy storage system in excess to the minimum permitted by the barrier design on the hazardous side of the barrier. The same shall be met by taking intrinsically safe transmitters and selecting the cable electrical parameters like inductance L/R ratio & capacitance in accordance with the maximum parameters given in barrier specifications.
  - b) No power source exceeding the voltage rating of Zener barrier shall be connected on safe side of the Zener barrier.
  - c) No outside power source including other intrinsically safe circuits shall be connected to the hazardous side of the barrier.
- 3.5.2 Zener barriers shall be located as close as possible to the field wiring entry point in the control room.
- 3.5.3 Single barrier are bolted directly to copper bus bar and multiple barriers on the barrier mounting plates. Copper bus or barriers mounting plates shall be isolated from the panel frame.
- 3.5.4 The signal ground system for intrinsically safe system shall be separate from power ground system and shall be connected to the signal ground reference point. The maximum resistance allocable between the farthest point on intrinsically safe barrier ground bus and signal ground reference point shall be less than 1 ohm.
- 3.5.5 Field wires shall directly terminate at the barriers and not through intermediate terminals.

MECON LIMITED	STANDARD SPEC	STANDARD SPECIFICATION ELECTRICAL & INSTRUMENTATION OIL & GAS SBU, DELHI	
REGD. OFF: RANCH 834002	ELECTRICAL & INSTR OIL & GAS SBI		
	INSTALLATION OF	DOCUMENT NO.	Page 13 of 21
TITLE	INSTRUMENTS	MEC/S/05/26/01	REVISION: 0
			EDITION: 1
3.6.1	Installation of all analyser shall be	in general, as per APIP 55	50 Part II.
3.6.2	The analyser housing at its instal classifications.	llation shall meet all safet	y requirements as pe
3.6.3	Sampled process fluid, if not retu location. Piping shall be provided and liquids shall be drained in a c vapours shall not be vented to atm	rned to the process shall so that vapours can be ve clean and orderly fashion nosphere.	be disposed to a safe ented to a safe location to a safe place. Toxi
3.6.4	Analyser shall be located as near	to the sampling point as po	ossible.
3.6.5	Analyser equipment must be prote	ected from the following:	
	<ul> <li>a) Hot equipment</li> <li>b) Severe ambient temperat</li> <li>c) Shock</li> <li>d) Mechanical damage</li> <li>e) Vibration</li> </ul>	ure changes	
3.6.6	If a separate vent for the analyser of minimum air Turbulence. If th common vent, a back pressure rea	is used, the location of the vents of different analysing gulator shall be used.	at vent shall be in are sers are vented into
3.6.7	Vent piping shall be designed to point and obstruct a free vent flow	prevent condensate from	n accumulation in lo
3.7	Ducts, Trays and Supports		
3.7.1	Main cable duct shall be of botto side sheet and top cover of 3.2 mr	m open type with flat/ang n thickness.	le construction wit
3.7.2	The ducts and trays shall be pro insert plates are not available, so fixed with a minimum of 10 mm e fabricated from minimum of 40 mm	perly supported at regula upport on concrete struct xpansion bolts Angle supp n angle.	r intervals. Whereve ure or ceiling shall b ports for ducts shall b
3.7.3	All supports shall be neatly cut wi ends of angle supports shall not off.	ith hacksaw only and not have sharp ends and sha	with gas cutting. Fre Il be properly rounde
3.7.4	Ducts and supports shall be pair primer conforming to IS-2074 afte 2 coats of final enamel paint as give	nted with one coat of Rec r cleaning to remove scale ven below:	d oxide Zinc chromat and then painted wit

	MECON LIMITED	STANDARD SPECIFICATION		
	REGD. OFF: RANCHI 834002	ELECTRICAL & INSTR	RUMENTATION	गुरु मेकान गुरु
		OIL & GAS SBU	J, DELHI	001:2000 Col
		INSTALLATION OF	DOCUMENT NO.	Page 14 of 21
	TITLE	INSTRUMENTS	MEC/S/05/26/01	<b>REVISION: 0</b>
				EDITION: 1
		<ul><li>a) Duct - Dark admirately Grey a</li><li>b) Supports - Black.</li></ul>	s per IS0632.	
	3.8	Instrument Steam Tracing		
	3.8.1	Steam for Tracking of instruments valve through carbon steel pipes s	s shall be taken from main supported at regular interva	steam header take of als.
	3.8.2	Steam tracing around individual diameter.	instrument shall be by	copper tube of 1/8"
	3.8.3	Piping or tubing for steam tracin condensate pockets.	g shall be installed in su	ch a way as to avoid
	3.8.4	After steam tracing, the line is connected to drain funnel through steam trap.		
	3.9	Identification of Lines and Instruments		
	3.9.1	All site mounted instruments, ju terminations shall be labelled or ta	inction boxes, air heade gged.	rs, tubing and wiring
	3.9.2	Instruments shall be furnished wit manufacturer's name, and mode approximately 3"x1" size and sha stainless steel wire.	th stainless steel name tag I no. serial number. This all be attached to the inst	gs containing Tag no., s tag number shall be ruments with gauge
	3.9.3	Unused cable entries in junction b	oxes and field instruments	are to be plugged.
	4.0	TESTING		
	4.1	Instrument Impulse piping/Tubing		
	4.1.1	All process impulse lines shall t vessel/piping end and flushed with	be disconnected both from water.	m the instrument and
	4.1.2	After thorough flushing the imput and pressurised hydraulically to corrected for ambient temperat pressure source and the pressu fall at a rate exceeding one psig/	lse lines shall be isolated o 1.5 times the maximu ture. They shall then f re reading on a test pres hour.	from the instruments um working pressure be isolated from the ssure gauge shall not

In case no isolation valve is provided near the instrument, impulse piping/tubing shall be pressurised along with the instrument to the maximum pressure of scale in case of pressure transmitter and max. Operating pressure in case of differential pressure instrument with equalising valve open

MECON LIMITED	STANDARD SPEC	STANDARD SPECIFICATION	
REGD. OFF: RANCHI 834002	ELECTRICAL & INSTR		मकान कार्याः २००० Confu
	OIL & GAS SBU	J, DELHI	
	INSTALLATION OF	DOCUMENT NO.	Page 15 of 21
TITLE	INSTRUMENTS	MEC/S/05/26/01	REVISION: 0
			EDITION : 1

- 4.1.3 In special conditions where hydro- testing is not permissible due to service requirements, testing shall be carried out by using compressed air/nitrogen.
- 4.1.4 The external displacer type instruments and cage type level switches shall be tested to 1.5 times the operating pressure using air/nitrogen after thorough flushing.
- 4.2 <u>Instrument Air lines/signal tubing</u>.
- 4.2.1 Instrument air lines/signal tubing shall not be hydrostatically tested.
- 4.2.2 Instrument air tubing shall be disconnected upstream of all filter regulators and blown down to remove water, slag and mill scale, from lines at 7.0 kg/cm<sup>2</sup> G for fifteen minutes.

Air filter shall be taken in line and tubing shall be disconnected at instrument end, and blown for 3 minutes to remove traces of dirt.

- 4.2.3 Testing of instrument air shall be carried out with instrument air at 7 kg/cm<sup>2</sup>G upto the upstream of the filter regulator after thorough flushing. All lines shall be checked with soap solution and bubbler unit for possible leak at joints.
- 4.2.4 All signal tubing shall be checked with 1.5 kg/cm<sup>2</sup> after proper flushing. After pressuring, source shall be cut off and rate of fall in pressure shall be less than IPSL for each 100 feet of tubing for a test period of 2 minutes as per instrument society of American RP 7.1 `Pneumatic Control Circuit Pressure Test'

#### 4.3 <u>Cables</u>

- 4.3.1 All wiring shall be checked to ensure that it is correctly connected and properly grounded.
- 4.3.2 All cables shall be checked for continuity proper connection and insulation testing.

Insulation test shall be carried out on all wiring with a certified magger after disconnecting the cables at both ends.

- 4.4 All the results of the above mentioned testing shall be recorded and submitted for check.
- 4.5 All the in line instruments like orifice plates, turbine meters, Rotameters, Target meters, vortex meters, control valves, safety valves etc. shall be removed and spool pieces shall be provided prior to the flushing of the lines.

MECON LIMITED	STANDARD SPECIFICATION		
REGD. OFF: RANCHI 834002	ELECTRICAL & INSTR OIL & GAS SBU	ELECTRICAL & INSTRUMENTATION OIL & GAS SBU, DELHI	
	INSTALLATION OF	DOCUMENT NO.	Page 16 of 21
TITLE	INSTRUMENTS	MEC/S/05/26/01	REVISION: 0
			EDITION: 1

### 5.0 CALIBRATION OF INSTRUMENTS

- 5.1 All instruments shall be calibrated strictly as per manufacturer's instructions prior to the installation. In addition to calibration of instruments, setting of safety devices like process switches, safety valves etc. and simulation testing of all interlock and shutdown system shall be carried out.
- 5.2 In general, all tests shall simulate, as closely as possible, design process condition by the use of manometers, potentiometers, deadweight testers, test pressure gauges etc. Pour point calibration shall refer to the input signal to an instrument equivalent to 0, 25, 50, 75,100% of instrument range upscale (rising) and 75, 50, 25, 0% of instrument (downscale) (falling).

All instruments unless otherwise noted shall be calibrated in upscale and downscale direction and if necessary, adjusted until their accuracies conform to those limits state by the manufacturer.

Upon completion of these tests, the instruments shall be drained, completely.

- 5.3 <u>Temperature Instruments</u>
- 5.3.1 Temperature Gauges Filled type and Bi metallic dial type Thermometers shall be four point bench checked for proper operation and calibration using a temperature bath prior to installation.
- 5.3.2 Temperature Elements and Temperature Transmitters.

Temperature Elements and Transmitter shall be four point bench calibrated using a temperature bath precision meter or precision gauge prior to installation.

- 5.4 <u>Pressure Instruments</u>
- 5.4.1 <u>Pressure Gauges</u>
- 5.4.1.1Direct connected bourdon type pressure gauges shall be dead weight tested or tested against a test gauges prior to installation.
- 5.4.1.2 Receiver type pressure gauges shall be four points calibrated using a precision gauge and precision air regulator.
- 5.4.1.3 Pressure and Differential Pressure Transmitters. Pressure and differential pressure transmitters shall be four points calibrated using a hydraulic or dead weight tester or a precision pneumatic calibrator prior to the Page 555 of 644

MECON LIMITED	STANDARD SPECIFICATION         SD. OFF: RANCHI         834002		
REGD. OFF: RANCHI 834002			मुठे मेकान मेकान
	OIL & GAS SBU	, DELHI	Dage 17 of 21
		DOCUMENT NO.	
IIILE	INGINOPENIO	MEC/S/05/26/01	REVISION : 0
			EDITION : 1
in	stallation. A precision output me	ter or gauge shall be used	to monitor the output.
5.5.1 O sl	Prifice plates shall be checked v harp edge. Bore dia. shall be che	isually for the name plate cked for compliance with t	and for an upstream he specification.
5.5.2 D u: pi tr	ifferential pressure type of flow sing precision pneumatic calibra recision output meter or gauge ansmitter.	w instruments shall be t ator or a manometer and e shall be used to moni	four points calibrated precision regulator. A tor the output of the
5.5.3 a	) Rotameters shall be install confirm that shipping stops ha	ed as received. A cheo ave been removed and flo	ck shall be made to at has been installed.
b	) Where rotameters have tra and lowered mechanically a data/ curve shall be checked.	nsmitting mechanism, the nd output shall be checke	e float shall be raised ed. Vendor calibration
С	:) A check shall be conducted v	with plumb for a vertical ins	stallation.
5.5.4 T flo	Turbine meters, Annubar, positive displacement meters, vortex meter, ultrasonic flow meter, etc. shall be installed as received.		
5.5.5 T sl	Target meters shall be checked for calibration using calibration weights. Output shall be monitored using precision output meter.		
5.6 <u>L</u>	evel Instruments		
5.6.1 <u>L</u>	evel Gauge Glasses		
G	auge glasses shall be installed rotectors and other accessories s	d as received installation hall be checked.	of illuminators, frost
5.6.2 D	isplacer Type, Level Transmitter		
-	<ul> <li>Displacer type level transmitter shall be checked by raising and lowering mechanically the displacement and checking the pilot or transmitter action Check transmitter with out put gauge or meter for smooth and full output change.</li> </ul>		
-	- A check shall be conducted with plumb for a vertical installation.		
5.6.3 D tr in tr	ifferential pressure type level ansmitter shall be calibrated w istallation. A precision meter or g ansmitter.	transmitter Differential ith pneumatic calibrator a gauge shall be used to mo	pressure type level at four points prior to pnitor the output of the

MECON LIMITE	STANDARD SPEC	STANDARD SPECIFICATION ELECTRICAL & INSTRUMENTATION OIL & GAS SBU, DELHI			
REGD. OFF: RANCI 834002	ELECTRICAL & INSTR OIL & GAS SBU				
	INSTALLATION OF	DOCUMENT NO.	Page 18 of 21		
TITLE	INSTRUMENTS	MEC/S/05/26/01	REVISION : 0		
			EDITION: 1		
5.6.4	<ul> <li>5.6.4 Tank level gauges</li> <li>a) Tank level gauges shall be checked by raising and lowering mechanically the displacer and checking the indicator on the gauge board.</li> <li>b) Check for proper liquid seal prior to installation in case of liquid seal tank gauges.</li> <li>c) In case of servo type gauges, the displacer is hoisted from the tank into the calibration chamber</li> </ul>				
5.7	Control Valves, shutdown valves a	and self actuated valves			
5.7.1	All diaphragm and piston operate using a pressure regulator and pre on the name plate of the valve.	All diaphragm and piston operated control valve shall be stroked pneumatically using a pressure regulator and pressure gauge against the spring range specified on the name plate of the valve.			
5.7.2	Mechanical seating and travel of t indicator and the name plate	Vechanical seating and travel of the valve stem shall be checked against the side ndicator and the name plate			
5.7.3	Valve positioner shall be calibrated with the control valve in accordance with the name plate data and specifications with the help of pneumatic calibrator or gauge with precision regulator. Zero position or fully close position of the valve shall be a live zero i.e., the plug shall be just off the seat at the minimum setting.				
5.7.4	olume bottles, where used shall be checked for proper filling. The signal line hall be bled to zero pressure and failure action shall be confirmed.				
5.7.5	Control valve accessories such as handwheels, boosters, relays etc. shall be checked operationally. Declutch able handwheel shall be operable both with and without an air signal to the diaphragm.				
5.7.6	Self actuated control valves shall be installed as received, checking inlet and outlet points and name plate data. Regulators with external pressure connections shall be inspected for proper installation.				
5.7.7	Butterfly shall be checked carefu upstream and down stream pipin confirmed.	utterfly shall be checked carefully to see that the vane moves freely into the ostream and down stream piping. Proper vane movement to stroke shall be onfirmed.			
5.7.8	All control valves and regulators and during hydro testing.	shall be removed from th	e line prior to flushing		

MECON LIMITED	STANDARD SPECIFICATION		
REGD. OFF: RANCHI 834002	ELECTRICAL & INSTR OIL & GAS SBU	UMENTATION , DELHI	मिकान मेकान 8007:2000 Company
	INSTALLATION OF	DOCUMENT NO.	Page 19 of 21
TITLE	INSTRUMENTS	MEC/S/05/26/01	REVISION : 0
			EDITION: 1
5.8	Safety Relief Valves		
5.8.1 S	Safety relief valves shall be installe blate data. Pilots, if used, shall be safety valve.	ed as received after carefu e checked carefully for ins	Illy checking the name tallation on the proper
5.8.2	Valves, which are installed in such pressure tested after installation Compressed air or nitrogen shall b	n a manner as to permit o to determine proper o be used for testing of safety	on line testing, shall be peration and setting. y relief valves.
5.9	Switches		
5.9.1 l	_evel Switches shall be actuated n calibrated for level setting.	nechanically for switch ope	eration but shall not be
5.9.2 I	Pressure switches shall be calib precision air regulator and gauge. continuity tester.	rated using hydorlic or o The setting/trip point sha	dead weight tester or all be checked using a
5.9.3 <sup>-</sup> i	Temperature switches shall be nstallation and set to the required	calibrated using a temp alarm/ trip point using a co	erature both prior to ontinuity tester.
5.10 <u>I</u>	Receiver Instruments		
5.10.1	Receiver Indicator/Recorders		
5.10.1.1 I	Pneumatic indicators/ Recorders precision pressure regulator and g	shall be calibrated using auge.	pneumatic calibrator/
5.10.1.2 I	Electronics indicators/ Recorders s a precision meter.	shall be calibrated using a	current generator and
5.10.1.3	Chart drive assembly shall be che	cked for proper operation.	
5.10.2	<u>Controllers</u>		
5.10.2.1 I	Proper balancing of the controlle catalogues.	er shall be checked as p	er the manufacturers
5.10.2.2 (	Controllers shall be checked for r ransfer from manual to Auto and v	manual and Auto operation vice versa shall be bumple	on and Transfer. The ss and smooth.
5.10.3.1 I	Manual loader station Output of precision meter.	the manual loader sha	ll be checked with a

MECON LIMITED	STANDARD SPECIFICATION		
REGD. OFF: RANCHI 834002	D. OFF: RANCHI 834002 ELECTRICAL & INSTRUMENTATION OIL & GAS SBU, DELHI		मुकान अक्षता:2000 Compart
	INSTALLATION OF	DOCUMENT NO.	Page 20 of 21
TITLE	INSTRUMENTS	MEC/S/05/26/01	REVISION: 0
			EDITION: 1

5.10.4.1 Multipoint Temperature Recorders

Each point shall be calibrated using a temperature simulator/ decade box for RTD / voltage generator and precision meter for the thermocouples.

- 5.10.4.2 Point synchronisation shall be checked.
- 5.10.5 Pneumatic receiver switches shall be calibrated using precision air regulator and gauge. The setting/alarm/trip point shall be checked using continuity tester.
- 5.10.6 Trip Amplifiers Trip amplifiers shall be calibrated using a temperature simulators or voltage generator and precision meter for thermocouple or Resistance box for RTD's. The required setting/ alarm point/ trip point shall be checked using a continuity tester.
- 5.10.7 Receiver Switch module Receiver switch modules shall be calibrated using a current source and a precision meter. The required setting/alarm/trip point shall be checked using a continuity tester.
- 5.10.8 <u>Alarm and Annunciator system</u>
- 5.10.8.1 Alarm and annunciator system shall be checked for visual and audio alarm operation using dummy signals. Full alarm sequence of each alarm point shall be checked.
- 5.10.8.2 Each point shall be checked for proper engraving.
- 5.10.9 <u>Shutdown System</u>
- 5.10.9.1 Operation of final actuating elements shall be checked for proper operation using dummy signals.
- 5.10.9.2 All timers, push buttons and switches shall also be checked for their proper operation.
- 5.11 <u>Analytical Instruments</u>
- 5.11.1 Check the full analyser system including sample handling system for leakage.
- 5.11.2 Check the full sample handling system for its proper operation. Calibrate and check completely all analysers using zero and span samples as per vendor catalogues.

MECON LIMITED	STANDARD SPEC		
REGD. OFF: RANCHI 834002	ELECTRICAL & INSTRUMENTATION OIL & GAS SBU, DELHI		मेकान मेकान 9007:2000 Comman
	INSTALLATION OF	DOCUMENT NO.	Page 21 of 21
TITLE	INSTRUMENTS	MEC/S/05/26/01	REVISION: 0
			EDITION: 1

#### 5.12 Flow computer / Volume corrector

- 5.12.1 Corrected flow values shall be checked for various D.C. inputs and pressure and temperature variations for upscale and downscale ranges.
- 5.13 The list of test and calibration instruments with traceability certificates shall be submitted to MECON for approval before carrying out the tests / calibration of instruments at site.
- 5.14 The formats / description of tests / calibration of all instruments shall be submitted to MECON for approval.
- 5.15 Daily / weekly reports shall be submitted during execution of work at site.

\*\*\*\*\*

# **SPECIFICATION**

# FOR

# **INSTRUMENT TUBING**

## SPECIFICATION NO.: MEC/S/05/26/02



### ELECTRICAL & INSTRUMENTATION (OIL & GAS SBU) MECON LIMITED DELHI 110 092

Page 561 of 644

MECON LIMITED	STANDARD SPECIFICATION		
REGD. OFF: RANCHI 834002	ELECTRICAL & INSTR OIL & GAS SBU	मुत्र मेकान 9001:2000 Control	
	SPECIFICATION FOR	DOCUMENT NO.	Page 3 of 6
TITLE	INSTRUMENT TUBING	MEC/S/05/26/02	<b>REVISION: 0</b>
			EDITION: 1

### **AMENDMENT STATUS**

SI. No.	Clause / Paragraph / Annexure / Exhibit / Drawing Amended	Page No.	Revision	Date	By (Name)	Verified (Name)

MECON LIMITED	MECON LIMITED STANDARD SPECIFICATION		
REGD. OFF: RANCHI 834002	ELECTRICAL & INSTF OIL & GAS SBU	यत मेकान मकान 9007:2000 Comman	
	SPECIFICATION FOR	DOCUMENT NO.	Page 4 of 6
TITLE	INSTRUMENT TUBING	MEC/S/05/26/02	REVISION: 0
			EDITION: 1

### CONTENTS

- 1.0 GENERAL
- 2.0 CONSTRUCTION
- 3.0 TESTING
- 4.0 SHIPPING
- 5.0 REJECTION

PREPARED BY:	CHECKED BY:	APPROVED BY:	ISSUE DATE :
(R. SANJAY BABU )	(RAKESH SHUKLA)	(PANKAJ SHIVASTAVA)	08 <sup>Th</sup> DEC 08

MECON LIMITED	STANDARD SPECIFICATION				
REGD. OFF: RANCHI 834002	ELECTRICAL & INSTRUMENTATION		के मेकान		
	OIL & GAS SBL	0001:2000 Control			
	SPECIFICATION FOR	DOCUMENT NO.	Page 5 of 6		
TITLE	INSTRUMENT TUBING	MEC/S/05/26/02	REVISION: 0		
			EDITION: 1		
1.0	GENERAL				
1.1	Scope				
1.1	This standard specifications, to covers the requirements for the Instrument Tubing which include	gether with the data shee e design, materials, tes s the following types:-	ets attached herewith, ting and shipping of		
	<ul><li>a) SS tubes</li><li>b) Copper tubes</li></ul>				
1.1.2	The related standards referred latest edition prior to the date of l	to herein and mentioned Purchaser's enquiry:	below shall be of the		
	ASTM A 269 - Specification for seamless and welded ferriti				
	ASTM B 251 - Spec	<ul> <li>Specification for general requirements for wrought seamless copper and copper alloy tube.</li> <li>Specification for general requirements for wrought seamless copper and copper alloy tube (Motric)</li> </ul>			
	ASTM B 251M - wrou (Met				
	ASTM B 68 - Spec	cification for seamless	copper tube, bright		
	ASTM B 68M - Specarity anne	cification for seamless ealed. (Metric)	copper tube, bright		
1.1.3	In the event of any conflict bet standards, codes, etc., the ven clarifications and only after of manufacture of the items in ques	ween these specifications dor shall refer the matter obtaining the same sha stion.	, data sheets, related to the purchaser for all proceed with the		
1.2	Bids				
1.2.1	Vendor's quotation shall include tube which shall provide the follo	e a detailed specification s wing information:	sheet for each type of		
	<ul><li>a) All the details regarding th</li><li>b) Overall the dimensions in</li></ul>	ne type, construction, mate mm.	rials etc. of the items.		
1.2.2	All the units of measurement an vendor's specification sheets sheets	All the units of measurement and material specifications for various parts in the vendor's specification sheets shall be to same standards as in purchaser's data			
1.2.3	Vendor shall attach a list of item this specification and purchaser's provide reasons for these deviation	s, type wise, summing up s data sheets if there are a ons.	all the deviations from any. Also vendor shall		

Page 564 of 644

MECON LIMITED	STANDARD SPECIFICATION				
REGD. OFF: RANCHI 834002	ELECTRICAL & INSTR	के मेकान क			
	OIL & GAS SBU	"07:2000 Cov			
	SPECIFICATION FOR	DOCUMENT NO.	Page 6 of 6		
TITLE	INSTRUMENT TUBING	MEC/S/05/26/02	REVISION: 0		
			EDITION: 1		
1.2.4	Vendor shall enclose catalogue other information for each type o	es giving detailed technion f tube in the bid.	cal specifications and		
1.2.5	Vendor's quotation, catalogues, o	drawings etc. shall be in E	nglish language.		
1.3	Drawings, Data and Certification				
	Detailed drawings, data, catalogues and manuals etc. required from the vendor are indicated by the purchaser in vendor data requirement sheets. The required number of reproducible and prints shall be despatched to the address mentioned, adhering to the time limits indicated.				
2.0	CONSTRUCTION				
2.1	Stainless Steel Tubes				
2.1.1	SS tubes of the tubes shall be Rockwell RB 70-70. Tubes shall be free from scratches and to be suitable for bending.				
2.1.3	Tube wall thickness shall be 0.049" for 1/2" OD and 1mm for 6 mm unless otherwise specified.				
2.1.4	Maximum working pressure shall be 153.0 kg/cm <sup>2</sup> at 38°C for 1/2" OD Tube, unless otherwise specified and 80.0 kg/cm <sup>2</sup> at 38° for 6mm OD tube.				
2.1.5	Tubes shall be supplied in minimum length of 6 metres without brazing in between.				
2.1.6	Dimensional tolerances shall be as per ASTM A 269.				
2.1.7	The following shall be marked on the tube:				
	<ul> <li>a) Name of manufacturer</li> <li>b) Type and material grade of</li> <li>c) Tube O.D. and wall thickn</li> </ul>	of tube less			
2.2	Copper Tubes				
2.2.1	Copper Tubes (PVC Jacket)				

- 2.2.1.1 The tube shall be soft annealed copper with 6mm OD and a wall thickness of 1.0 mm as per ASTM B 68M Copper No.C12200.
- 2.2.1.2 The tube shall be jacketed with black PVC. The jacket thickness shall be 1.6mm. The PVC jacket shall confirm to ASTM D-1047.

Page 565 of 644

MECON LIMITED	STANDARD SPECIFICATION		
REGD. OFF: RANCHI 834002	ELECTRICAL & INST OIL & GAS SBU	रात मेकान अवग:2000 ComPart	
	SPECIFICATION FOR	DOCUMENT NO.	Page 7 of 6
TITLE	INSTRUMENT TUBING	MEC/S/05/26/02	REVISION: 0
			EDITION: 1

- 2.2.1.3 The tube ends shall be plugged prior to transportation.
- 2.2.1.4The tube shall be of continuous length without any brazing in between for 100 metres length.
- 2.2.1.5 Minimum length of single tube shall be 100 metres.
- 2.2.1.6 The dimensional tolerances shall be as per ASTM B 251M.
- 2.2.2 Bare Copper Tubes (For Steam Tracing)
- 2.2.2.1 The tube shall be soft annealed copper with 3/8" OD or 6mm OD with a wall thickness of 1.0 mm as per ASTM B68 copper No.C12200.
- 2.2.2.2The tube ends shall be plugged prior to transportation.
- 2.2.2.3The tube shall be of continuous length without any brazing in between for 100 metres length.
- 2.2.2.4 Minimum length of tube shall be 100 metres.
- 2.2.2.5The dimensional tolerances shall be as per ASTM B 251.

### 3.0 **TESTING**

- 3.1 The following tests shall be done for SS tubes.
  - a) Hardness test
  - b) Hydrostatic test at 153.0 kg/cm<sup>2</sup> at 38° C for 1/2" tube and at 80.0 kg/cm<sup>2</sup> at 38°C for 6mm tube, unless otherwise specified.
- 3.2 PVC jacketed copper tubes shall be tested at 7.0kg/cm<sup>2</sup>g with dry air for leak check.
- 3.3 Bare copper tubes shall be hydrostatically tested at 80.0 kg/cm<sup>2</sup>g at 38°C.
- 3.4 Final test before delivery shall include ball test to ensure clear opening of the tube for copper tubes. The O.D of the ball shall be minimum 1mm for 6mm O.D tube and 2mm for 3/8" tube.

### 4.0 SHIPPING

4.1 The tubes shall be plugged at both ends to avoid entry of foreign matter. The tubes shall be packed carefully so as to avoid damage during transport.

### 5.0 **REJECTION**

MECON LIMITED	STANDARD SPEC		
REGD. OFF: RANCHI 834002	ELECTRICAL & INST OIL & GAS SBU	मुत्र मेकान 9001:2000 Compart	
	SPECIFICATION FOR	DOCUMENT NO.	Page 8 of 6
TITLE	INSTRUMENT TUBING	MEC/S/05/26/02	REVISION: 0
			EDITION: 1

Vendor shall make his offer in detail, with respect to every item of the purchaser's specifications. Any offer not conforming to this shall be summarily rejected.

# **SPECIFICATION**

# FOR

# INLET / OUTLET SECTIONS AND FLOW STRAIGHTENER

## SPECIFICATION NO.: MEC/S/05/26/03



### ELECTRICAL & INSTRUMENTATION (OIL & GAS SBU) MECON LIMITED DELHI 110 092

Page 568 of 644

MECON LIMITED	STANDARD SPECI		
REGD. OFF: RANCHI 834002	ELECTRICAL & INSTR	गुरु मेकान	
	OIL & GAS SBU,	DELHI	001:2000 Com
	SPECIFICATIO FOR	DOCUMENT NO.	Page 1 of 3
TITLE	INLET/ OUTLET SECTIONS AND FLOW STRAIGHTENER	MEC/S/05/26/03	REVISION: 0
			EDITION: 1

### **AMENDMENT STATUS**

SI. No.	Clause / Paragraph / Annexure / Exhibit / Drawing Amended	Page No.	Revision	Date	By (Name)	Verified (Name)

MECON LIMITED	STANDARD SPECI		
REGD. OFF: RANCHI 834002	ELECTRICAL & INSTR OIL & GAS SBU	रत मेकान 9007:2000 Comput	
	SPECIFICATIO FOR	DOCUMENT NO.	Page 2 of 3
TITLE	INLET/ OUTLET SECTIONS AND FLOW STRAIGHTENER	MEC/S/05/26/03	REVISION: 0
			EDITION: 1

### <u>CONTENTS</u>

- 1.0 FLOW STRAIGHTNER
- 2.0 INLET SECTION/OUTLET SECTION

PREPARED BY:	CHECKED BY:	APPROVED BY:	ISSUE DATE :
(R. SANJAY BABU )	(RAKESH SHUKLA)	(PANKAJ SRIVASTAVA)	08 <sup>Th</sup> DEC 08

MECON LIMITED	STANDARD SPEC	IFICATION	
REGD. OFF: RANCHI 834002	ELECTRICAL & INSTR OIL & GAS SBU	18 मेकान 9007:2000 Contract	
	SPECIFICATIO FOR	DOCUMENT NO.	Page 3 of 3
TITLE	INLET/ OUTLET SECTIONS AND FLOW STRAIGHTENER	MEC/S/05/26/03	REVISION: 0
			EDITION: 1

### **1.0 FLOW STRAIGHTNER**

- 1.1 Flow straightener shall be installed in the inlet section upstream of flow meter.
- 1.2 Flow straightener shall consist of a bundle of tubes designed and assembled as per ISO 5167-1980 (E) or AGA Report no.3 or AGA Report no.7.
- 1.3 Tube diameter shall be 0.2 D or less where D is the diameter of Meter inlet section pipe. Tube shall be seamless quality of SS 316 material thickness 1.5 or 1 mm Tube shall be hydrostatically tested for 75kg/cm<sup>2</sup> pressure and then cut to the required length for assembly.
- 1.4 On both the sides of tube bundle a very thin retaining rings of SS316 shall be provided for avoiding the dislocation of pipes.
- 1.5 The tube bundle shall be kept in position by 6mm threaded bolt. After tightening the bolt shall seal welded.
- 1.6 Tube ends shall be tempered at  $45^{\circ}$  angle.

#### 2.0 INLET SECTION/OUTLET SECTION

- 2.1 Inlet/Outlet sections of pipe for flow meter shall be of same inside diameter (ID) as turbine meter.
- 2.2 The pipe used should be circular and with no ovality or any other deformation, the inside surface of pipe shall be smooth, welded portion shall be grinded to smooth finish.

# **SPECIFICATION**

# FOR

# **INSTRUMENT TUBE FITTINGS**

SPECIFICATION NO.: MEC/S/05/26/04



ELECTRICAL & INSTRUMENTATION (OIL & GAS SBU) MECON LIMITED DELHI 110 092

MECON LIMITED	STANDARD SPECI		
REGD. OFF: RANCHI 834002	ELECTRICAL & INSTR	यठ मेकान 9001:2000 Contract	
	SPECIFICATION FOR	DOCUMENT NO.	Page 1 of 5
TITLE	INSTRUMENT TUBE FITTINGS	MEC/S/05/26/04	REVISION: 0
			EDITION: 1

### **AMENDMENT STATUS**

SI. No.	Clause / Paragraph / Annexure / Exhibit / Drawing Amended	Page No.	Revision	Date	By (Name)	Verified (Name)

MECON LIMITED	STANDARD SPECI		
REGD. OFF: RANCHI 834002	ELECTRICAL & INSTR OIL & GAS SBU,	मुठ मेकान 9007:2000 Contribut	
	SPECIFICATION FOR	DOCUMENT NO.	Page 2 of 5
TITLE	INSTRUMENT TUBE FITTINGS	MEC/S/05/26/04	<b>REVISION: 0</b>
			EDITION: 1

### CONTENTS

- 1.0 GENERAL
- 2.0 CONSTRUCTION
- 3.0 TESTING
- 4.0 SHIPPING
- 5.0 REJECTION

PREPARED BY:	CHECKED BY:	APPROVED BY:	ISSUE DATE :
(R. SANJAY BABU )	(RAKESH SHUKLA)	(PANKAJ SHIVASTAVA)	08 <sup>™</sup> DEC 08
			•

MECON LIMITED	STANDARD SPEC	IFICATION	
REGD. OFF: RANCHI 834002	ELECTRICAL & INSTR	10 मेकान 10 मेकान 10 मिकान	
	OIL & GAS SBU	, DELHI	712000 000
	SPECIFICATION FOR	DOCUMENT NO.	Page 3 of 5
TITLE	INSTRUMENT TUBE FITTINGS	MEC/S/05/26/04	REVISION: 0
			EDITION: 1

### 1.0 GENERAL

- 1.1 <u>Scope</u>
- 1.1.1 This standard specifications, together with the data sheets attached herewith, covers the requirements for the design, materials, testing and shipping of instrument tube fittings which includes the following types:
  - a) SS compression fittings (SS tube)
  - b) Brass compression fittings (copper tube)
- 1.1.2 The related standards referred to herein and mentioned below shall be of the latest edition prior to the date of Purchase's enquiry:

ANSI B 2.1	-	Pipe Threads
B16.11		- Forged steel fittings-socket welding and threaded.
IS:319	-	Specification for free cutting brass bars, rods and sections.
ISA RP 42.1	-	Nomenclature for instrument tubing - fittings.

- 1.1.3 In the event of any conflict between these specifications, data sheets, related standards, codes etc., the vendor shall refer the matter to the purchaser for clarifications and only after obtaining the same shall proceed with the manufacture of the items in question.
- 1.2 <u>Bids</u>
- 1.2.1 Vendor's quotation shall include a detailed specification sheet for each type of tube fittings which shall provide the following information:
  - a) All the details regarding the type, construction, materials, etc. of the items.
  - b) Overall dimensions in mm.
- 1.2.2 All the units of measurement and material specifications for various parts in the vendor's specification sheets shall be to same standards as in purchaser's data sheets.
- 1.2.3 Vendor shall attach a list of items, typewise, summing up all the deviations from this specification and purchaser's data sheets if there are any. Also vendor shall provide reasons for these deviations.
- 1.2.4 Vendor shall enclose catalogues giving detailed technical specifications and other information for each type of fitting in the bid.
- 1.2.5 Vendor's quotation, catalogues, drawings, etc. shall be in Englishdengruage644

MECON LIMITED	STANDARD SPEC		
REGD. OFF: RANCHI 834002	ELECTRICAL & INSTR OIL & GAS SBU	मेकान 93 9001:2000 Contrint	
	SPECIFICATION FOR	DOCUMENT NO.	Page 4 of 5
TITLE	INSTRUMENT TUBE FITTINGS	MEC/S/05/26/04	<b>REVISION: 0</b>
			EDITION: 1

#### 1.3 Drawings, Data and Certification

Detailed drawings, data, catalogues and manuals etc., required from the vendor are indicated by the purchaser in vendor data requirement sheets. The required number of reproducibles and points shall be despatched to the address mentioned, adhering to the time limits indicated.

#### 2.0 CONSTRUCTION

- 2.1 <u>SS Tube fittings</u>:
- 2.1.1 Nomenclature of all tube fittings shall be as per ISA RP 42.1.
- 2.1.2 Fittings shall be flareless compression type and of three piece construction with ferrule, nut and body suitable for use on SS tubes conforming to ASTM A 269 TP316, hardness not exceeding RB80.
- 2.1.3 All parts shall be of SS 316.
- 2.1.4 Hardness of the ferrules shall be in the range of RB 85-90 so as to ensure a minimum hardness difference of 5 to 10 between tube and fittings, for better sealing.
- 2.1.5 Nuts and ferrules of particular size shall be interchangeable for each type.
- 2.1.6 Spanner hold shall be metric.
- 2.1.7 Threaded ends of fittings shall be NPT as per ANSI B 2.1.
- 2.1.8 <u>Copper Tube Fittings</u>
- 2.2.1 Nomenclature of all tube fittings shall be as per ISA 42.1.
- 2.2.2 Fittings shall be of flareless compression type and of three-piece construction consisting of ferrule, nut and body suitable for use on copper tubes conforming to ASTM B 68/B 68M hardnesss not exceeding RB 50.
- 2.2.3 All parts shall be manufactured from Brass as per IS 319 barstock and Nickel plated.
- 2.2.4 For better grip, vendor shall maintain hardness difference between tube & ferrule and indicate the same along with the offer.
- 2.2.5 Nuts & ferrules of particular size shall be interchangeable for each type.

Page 576 of 644
MECON LIMITED	STANDARD SPEC	IFICATION	
REGD. OFF: RANCHI 834002	मिकान मेकान 8007:2000 Company		
	SPECIFICATION FOR	DOCUMENT NO.	Page 5 of 5
TITLE	INSTRUMENT TUBE FITTINGS	MEC/S/05/26/04	REVISION: 0
			EDITION: 1

- 2.2.6 Threaded ends of fittings shall be NPT as per ANSI B 2.1.
- 2.2.7 Spanner hold shall be metric.
- 2.2.8 Vendor shall ensure that the ferrules and nuts supplied for fittings shall be suitable for sample tube which shall be supplied during manufacture.

#### 3.0 TESTING

3.1 Random samples of SS tube fittings shall be hydrostatically tested as follows:-

For 6 mm fittings at 80.0 kg/cm<sup>2</sup>, 1/2" fittings at 153.0 kg/cm<sup>2</sup> at 38°c unless otherwise specified.

3.2 Random samples of brass compression fittings shall be hydrostatically tested as follows:-

For 1/4" fittings, at 10 kg./cm<sup>2</sup>, 3/8" at 80.0 Kg/cm<sup>2</sup> and all at  $38^{\circ}$ C.

#### 4.0 SHIPPING

4.1 All thread/ends shall be protected with plastic caps to prevent damage/entry of foreign matter.

#### 5.0 REJECTION

Vendor shall make his offer in detail, with respect to every item of the purchaser's specifications. Any offer not conforming to this shall be summarily rejected.

# **SPECIFICATION**

# FOR

# **INSTRUMENT VALVES AND MANIFOLDS**

**SPECIFICATION NO.: MEC/S/05/26/05** 



ELECTRICAL & INSTRUMENTATION (OIL & GAS SBU) MECON LIMITED DELHI 110 092

Page 578 of 644

MECON LIMITED	STANDARD SPECI	FICATION	
REGD. OFF: RANCHI 834002	ELECTRICAL & INSTR OIL & GAS SBU,	मकान मकान 9001:2000 Compton	
	SPECIFICATION FOR	DOCUMENT NO.	Page 1 of 6
TITLE	INSTRUMENT VALVES AND MANIFOLDS	MEC/S/05/26/05	REVISION: 0
			EDITION: 1

# **AMENDMENT STATUS**

SI. No.	Clause / Paragraph / Annexure / Exhibit / Drawing Amended	Page No.	Revision	Date	By (Name)	Verified (Name)

MECON LIMITED	STANDARD SPECI		
REGD. OFF: RANCHI 834002	ELECTRICAL & INSTR OIL & GAS SBU,	18 मेकान 9007:2000 Contract	
	SPECIFICATION FOR	DOCUMENT NO.	Page 2 of 6
TITLE	INSTRUMENT VALVES AND MANIFOLDS	IENT VALVES AND ANIFOLDS MEC/S/05/26/05	
			EDITION: 1

# CONTENTS

- 1.0 GENERAL
- 2.0 CONSTRUCTION
- 3.0 TESTING
- 4.0 SHIPPING
- 5.0 REJECTION

	CHECKED BT:	APPROVED BY:	ISSUE DATE :
(R. SANJAY BABU ) (	RAKESH SHUKLA)	(PANKAJ SRIVASTAVA)	08 <sup>Th</sup> DEC 08

MECON LIMITED	MECON LIMITED STANDARD SPECIFICATION				
REGD. OFF: RANCHI 834002	RUMENTATION	क मेकान क			
	OIL & GAS SBU	J, DELHI	0001:2000 Com		
	SPECIFICATION FOR	DOCUMENT NO.	Page 3 of 6		
TITLE	INSTRUMENT VALVES AND MANIFOLDS	MEC/S/05/26/05	REVISION: 0		
			EDITION : 1		

# 1.0 GENERAL

- 1.1 <u>Scope</u>
- 1.1.1 This standard specifications, together with the data sheets attached herewith, covers the requirements for the design, materials, testing and shipping of Instrument Valves & Manifolds which includes the following types:
  - a) Miniature instrument valves
  - b) Instrument valve manifolds
  - c) Instrument air valves
- 1.1.2 The related standards referred to herein and mentioned below shall be of the latest edition prior to the date of Purchaser's enquiry:

ANSI B 2.1	-	Pipe threads
ANSI B 16.11	-	Forged steel fittings-socket welding and threaded.

- 1.1.3 In the event of any conflict between these specifications, data sheets, related standards, codes etc, the vendor shall refer the matter to the purchaser for clarifications and only after obtaining the same shall proceed with the manufacture of the items in question.
- 1.2 <u>Bids</u>
- 1.2.1 Vendor's quotation shall include a detailed specification sheet for each type of Valves & Manifolds which shall provide the following information:
  - a) All the details regarding the type, construction, materials etc. of the items.
  - b) Overall dimensions in mm.
- 1.2.2 All the units of measurement and material specifications for various parts in the vendor's specification sheets shall be to same standards as in purchaser's data sheets.
- 1.2.3 Vendor shall attach a list of items, typewise, summing up all the deviations from this specification and purchaser's data sheets if there are any. Also vendor shall provide reasons for these in the bid.
- 1.2.5 Vendor's quotation, catalogues, drawings etc. shall be in English language.

MECON LIMITED	STANDARD SPEC		
REGD. OFF: RANCHI 834002	18 101:2000 Company		
	SPECIFICATION FOR	DOCUMENT NO.	Page 4 of 6
TITLE	INSTRUMENT VALVES AND MANIFOLDS	MEC/S/05/26/05	REVISION: 0
			EDITION: 1

Detailed drawings, data, catalogues and manual etc. required from the vendor are indicated by the purchaser in vendor data requirement sheets. The required number of reproducible and prints shall be despatched to the address mentioned, adhering to the time limits indicated.

## 2.0 CONSTRUCTION

- 2.1 <u>Instrument Valves (Miniature)</u>
- 2.1.1 The instrument valves shall be globe pattern-needle valves forged/ barstock with inside screwed bonnet.
- 2.1.2 Body and trim material shall be 316 SS unless otherwise specified.
- 2.1.3 The valve body rating shall be 3000 lbs unless specified in piping material specification which shall govern in case it is specified.
- 2.1.4 The end connection shall be 1/2" NPTF to ANSI B2.1.
- 2.1.5 The packing material shall be teflon unless otherwise specified.
- 2.1.6 The hand wheel material shall be carbon steel zinc plated.
- 2.1.7 Flow direction shall be marked on the body.
- 2.1.8 The valve dimension shall be as follows:
  - a) End to end dimensions 76 mm (approximately).
  - b) Height in fully open condition 135mm maximum.

## 2.2 VALVE MANIFOLDS

- 2.2.1 <u>3-Valve & 5-Valve manifolds</u>:
- 2.2.1.13 Valve manifold shall be designed for direct coupling to differential pressure transmitters having 2 bolt flanges with 54 mm (2-1/8") centre to centre connections and 41.3 mm (1-5/8") bolt to bolt distance. The manifold shall contain two main block valves and an equalizing by-pass valve. The valves shall be needle valves. They shall use self aligning 316SS ball seats.
- 2.2.1.25 Valve manifold shall contain two main line block valves and a combination double block and bleed for the bypass line.

2.2.1.3The manifold shall be suitably for mounting directly on the stanchion (2" pipe) Page 582 of 644

MECON LIMITED	STANDARD SPEC		
REGD. OFF: RANCHI 834002	मिकान मिकान सिंहा 2000 Company		
	SPECIFICATION FOR	DOCUMENT NO.	Page 5 of 6
TITLE	INSTRUMENT VALVES AND MANIFOLDS	MEC/S/05/26/05	REVISION: 0
			EDITION: 1

- 2.2.1.4All bonnets shall have teflon packing unless otherwise specified.
- 2.2.1.5The material of construction shall be 316 SS unless otherwise specified.
- 2.2.1.5The material of construction shall be 316SS unless otherwise specified.
- 2.2.1.6The flanges shall be integral part of the block.
- 2.2.1.7The process connection shall be 1/2" NPTF to ANSI B2.1.
- 2.2.1.8The manifolds shall be supplied alongwith mounting accessories. The bolts and nuts shall be alloy steel as per ASTM A 193 Gr B ASTM A 194 GR 2H respectively. Rings shall be teflon and other accessories shall be cadmium plated.
- 2.2.1.9Vendor shall furnish the material certificate for body.
- 2.2.2 <u>3 Way 2 Valve Manifold for pressure gauges.</u>
- 2.2.2.1The manifold shall be designed for use with pressure gauges.
- 2.2.2.2The valve shall be a ball valve.
- 2.2.2.3 The body shall be either straight or angle as specified in data sheets.
- 2.2.2.4The body and trim material shall be 316SS, packing material shall be teflon unless otherwise specified.
- 2.2.2.5 The inlet connection shall be 3/4" plain end (female) for socket weld as per ANSI B 16.11.
- 2.2.2.6 The gauge connections shall be with union nut & tail piece threaded 1/2" NPT (F).
- 2.2.2.7The drain connection shall 1/2"NPTF.
- 2.3 Instrument Air Isolation Valves
- 2.3.1 The valves shall be full bore ball valves.
- 2.3.2 Body material shall be Nickel or Cadmium plated carbon steel.
- 2.3.3 Trim material shall be 316SS.

## 2.3.4 The end connection shall be 1/2" NPTF to ANSI B2.1 unless otherwise specified. Page 583 of 644

MECON LIMITED	STANDARD SPECIFICATION			
REGD. OFF: RANCHI 834002	ELECTRICAL & INSTR	RUMENTATION	क मेकान	
	OIL & GAS SBU	J, DELHI	"001:2000 Con	
	SPECIFICATION FOR	DOCUMENT NO.	Page 6 of 6	
TITLE	INSTRUMENT VALVES AND MANIFOLDS	MEC/S/05/26/05	REVISION: 0	
			EDITION: 1	
2.3.5 The packing material shall be teflon.				
2.3.6	The handle/wrench material shall be cadmium or nickel plated carbon steel.			
2.3.7	The valve body rating shall be ANSI 800 lb.			
2.3.8	End to end dimensions shall be 70mm (approximately).			
3.0	TESTING			

- 3.1 The instrument valves (miniature) shall be hydrostatically tested at 200kg/cm<sup>2</sup> g at 38°C.
- 3.2 All manifolds (3 valves, 5 valves and 3 ways, 2 valves) shall be hydrostatically tested at 200 kg/cm2 at 38C.
- 3.3 The instrument air valves shall be hydrostatically tested at 15.0 kg/cm<sup>2</sup>g at 38°C and at 10.5 kg/cm<sup>2</sup>g with dry air.

#### 4.0 SHIPPING

4.1 All threads/ends shall be protected with plastic caps to prevent damage/entry of foreign matter.

#### 5.0 REJECTION

Vendor shall make his offer in detail, with respect to every item of the purchaser's specifications. Any offer not conforming to this shall be summarily rejected.

# **SPECIFICATION**

# FOR

# JUNCTION BOXES AND CABLE GLANDS

# SPECIFICATION NO.: MEC/S/05/26/06



# ELECTRICAL & INSTRUMENTATION (OIL & GAS SBU) MECON LIMITED DELHI 110 092

MECON LIMITED	STANDARD SPECI	STANDARD SPECIFICATION		
REGD. OFF: RANCHI 834002	EGD. OFF: RANCHI 834002 OIL & GAS SBU, DELHI			
	SPECIFICATION FOR	DOCUMENT NO.	Page 1 of 7	
TITLE	JUNCTION BOXES AND CABLE GLANDS	MEC/S/05/26/06	REVISION: 0	
			EDITION: 1	

# **AMENDMENT STATUS**

SI. No.	Clause / Paragraph / Annexure / Exhibit / Drawing Amended	Page No.	Revision	Date	By (Name)	Verified (Name)

MECON LIMITED	STANDARD SPECIFICATION		
REGD. OFF: RANCHI 834002	ELECTRICAL & INSTR		मिकान बुरु बेरेग:2000 Control
	UIL & GAS SBU		
	SPECIFICATION FOR	DOCUMENT NO.	Page 2 of 7
TITLE	JUNCTION BOXES AND CABLE GLANDS	MEC/S/05/26/06	REVISION: 0
			EDITION: 1

# CONTENTS

- 1.0 GENERAL
- 2.0 JUNCTION BOXES
- 3.0 CABLE GLANDS & PLUGS, REDUCERS/ ADAPTORS
- 4.0 NAME PLATE
- 5.0 SHIPPING
- 6.0 REJECTION

PREPARED BY:	CHECKED BY:	APPROVED BY:	ISSUE DATE :
(R. SANJAY BABU )	(RAKESH SHUKLA)	(PANKAJ SRIVASTAVA)	08 <sup>Th</sup> DEC 08

MECON LIMITED STANDARD SPECIFICATION		IFICATION	
REGD. OFF: RANCHI 834002	ELECTRICAL & INSTR OIL & GAS SBU	RUMENTATION J, DELHI	18 9007:2000 Contrin
	SPECIFICATION FOR	DOCUMENT NO.	Page 3 of 7
TITLE	JUNCTION BOXES AND CABLE GLANDS	MEC/S/05/26/06	REVISION: 0
			EDITION: 1

### 1.0 General

## 1.1 <u>Scope</u>

- 1.1.1 This standard specifications, together with the data sheets attached herewith, covers the requirements for the design, materials, nameplate marking, testing and shifting of junction boxes & cable glands which include the following types:
  - a) Electrical junction boxes.
  - b) Pneumatic junction boxes
  - c) Cable glands (whenever specified)
- 1.1.2 The related standards referred to herein and mentioned below shall be of the latest editions prior to the date of the purchaser's enquiry:

ANSI B 2.1	:	Pipe threads
IS-5	:	Colours for ready mixed paints and enamels
IS-2147	:	Degrees of protection provided by enclosures for Low voltage switchgear and control gear
IS-2148	:	Flame proof enclosure of electrical apparatus.

- 1.1.3 In the event of any conflict between specifications, data sheets, related standards, codes etc., the vendor shall refer the matter to the purchaser for clarifications and only after obtaining the same should proceed with the manufacture of the items in questions.
- 1.2 <u>Bids</u>
- 1.2.1 Vendor's quotation shall include a detailed specification sheet for each type of junction box and cable gland which shall provide the following information:
  - a) All the details regarding the type, construction, materials, housing, entries, etc.
  - b) All dimensions in millimetre.
  - c) Sketch for each type of JB with dimensional details showing the terminal and entries arrangement.
  - d) Mounting details.
  - e) Vendor shall furnish certificate from statutory body for explosion proof enclosure, indicating the gas group and temperature class.
- 1.2.2 All the material specifications for various parts in the vendor's specification sheets shall be to the same standards as those in purchaser's data sheets (e.g. BS IS, etc.)

MECON LIMITED	STANDARD SPECIFICATION ELECTRICAL & INSTRUMENTATION OIL & GAS SBU, DELHI		
REGD. OFF: RANCHI 834002			मकान मकान कार्या:2000 Company
	SPECIFICATION FOR	DOCUMENT NO.	Page 4 of 7
TITLE	JUNCTION BOXES AND CABLE GLANDS	MEC/S/05/26/06	<b>REVISION: 0</b>
			EDITION: 1
1.2.3	Vendor shall attach a list of items, tag number wise, summing up all the deviations from the purchaser's data sheets, if there are any. Also vendor shall furnish reasons for these deviations.		
1.2.4	Vendor shall enclose catalogue other information for each type o the bid.	es giving detailed technic f JB/cable glands and its a	cal specifications and accessories covered in
1.2.5	Vendor's quotation, catalogues, o	drawings, etc. shall be in E	nglish language.
1.3	Drawings, Data and Certification		
1.3.1	Detailed drawings, data, catalogues required from the vendor are indicated in vendor data requirements sheets. The required number of reproducible and prints shall be despatched to the address mentioned, adhering to the time limits indicated.		
1.3.2	After placement of purchaser order, vendor shall submit certified drawings and specifications sheets for each type of JB/cable gland which shall include the following:		
	<ul> <li>a) Detailed dimensional draw</li> <li>b) Weight of each in grams/l</li> <li>c) Certificate from statutor</li> <li>hazardous area.</li> </ul>	vings Kg. y body suitable for ins	stallation in specified
2.1	Junction Boxes		
2.0	Junction boxes shall be either of	the following type as spec	ified in data sheets.
	I. Weather proof junction bo II. Weather proof & Explosio	oxes. n proof junction boxes.	
2.2	The enclosure shall be as per IS-2147 for weather proof junction boxes and for Explosion proof it shall be as per IS-2148 suitable for the area classification specified.		
2.3	Number of entries and locations	shall be as per data sheet	S.
2.4	Junction boxes shall be prov connection of hand powered tele	vided with telephone so phone set.	ckets and plugs for

MECON LIMITED	ANCHI ELECTRICAL & INSTRUMENTATION		
REGD. OFF: RANCHI 834002			वे मेकान
	OIL & GAS SBU	I, DELHI	001:2000 Cont
	SPECIFICATION FOR	DOCUMENT NO.	Page 5 of 7
TITLE	JUNCTION BOXES AND CABLE GLANDS	MEC/S/05/26/06	REVISION: 0
			EDITION: 1
2.5	Electrical Junction boxes		
2.5.1	Material shall be die-cast alumini	ium of minimum 5 mm thic	k (LM-6 alloy)
2.5.2	Explosion proof junction boxes s box by means of cadmium plated	hall have detachable cove d triangular head/hexagon	er which is fixed to the al head screws.
2.5.3	Weather proof junction boxes sl	hall have doors which sha	all be hinged type and
2.5.4	Explosion proof junction boxes on the cover as given below:	shall have a warning engl	aved/integrally cast
	"Isolate power supply elsewhere	before opening"	
2.5.5	Terminals shall be spring loaded plated steel rails complete with e	l, vibration proof, clip-on ty nd cover and clamps for e	pe, mounted on nickel ach row.
2.5.6	All terminals shall be suitable conductor, in general. Howeve detail shall be as per job specific	for accepting minimum or for power supply distrit ation/Data sheets.	2.5 sq. mm copper pution boxes, terminal
2.5.7	Sizing shall be done with due consideration for accessibility and maintenance in accordance with the following guidelines.		
	i) 50 to 60 mm between ter for upto 50 terminals a terminals.	minals and sides of box p and additional 25mm fo	arallel to terminal strip r each additional 25
	ii) 100 to 120mm between 25mm for each additional	terminals for upto 50 ter 25 terminals	minals and additional
	iii) Bottom/top of terminal sh the junction box.	all not be less than 100 r	nm from bottom/top of
2.5.8	Terminals shall be marked as pe	r the various types indicat	ed in data sheets.
2.5.9	Shall be provided with external e	arthing lugs.	
2.6	Pneumatic junction boxes		
2.6.1	Pneumatic junction boxes shall shall have necessary neoprene flush with the box and shall be hi	be made of 3mm thick h gasket between door and nged type and provided w	not rolled steel. They I body. Door shall be ith wing nuts.
2.6.2	Single tube entries shall be suit fittings. Multi tube bundle entry sheets.	able for 6mm O.D. coppe shall be suitable for the	er tube with bulk head data furnished in data

MECON LIMITED	STANDARD SPECIFICATION		
REGD. OFF: RANCHI 834002	ELECTRICAL & INSTRUMENTATION		मेकान क
	OIL & GAS SBL	J, DELHI	0007:2000 Contre
	SPECIFICATION FOR	DOCUMENT NO.	Page 6 of 7
TITLE	JUNCTION BOXES AND CABLE GLANDS	MEC/S/05/26/06	REVISION: 0
			EDITION: 1
2.7	Painting		
2.7.1	Surface shall be prepared for pa scale.	inting. It shall be smooth	and devoid of rust and
2.7.2	Two coats of lead-free base prin paint shall be applied both for int	ner and two final coats of erior and exterior surfaces	lead free epoxy based
2.7.3	The colour shall be as specified	in data sheets.	
3.0	Cable glands & plugs, Reduce	rs/Adaptors	
3.1	Cable glands shall be supplied b	y vendor whenever specifi	ed.
3.2	Cable glands shall be double compression type for use with armoured cables.		
3.3	The cable glands shall be of Nickel plated brass.		
3.4	The cable gland shall be weather proof. Whenever specified they shall be explosion proof and certificate from statutory body shall be furnished.		
3.5	Cable glands shall be supplied to suit the cable dimensions indicated along with tolerance indicated in data sheets. Various components like rubber ring, metallic ring, metallic cone and the outer/inner nuts etc. shall be capable of adjusting to the above tolerances of cable dimensions.		
3.6	Reducers/Adaptors shall be su They shall be nickel plated bra These shall also be explosion statutory body for explosion shal	pplied as per details indi ass. These shall be wea proof wherever specified I be furnished.	cated in data sheets. ther proof in general. d and certificate from
3.7	Plugs shall be provided whereve	r specified. They shall be	of Nickel plated brass.
3.8	Plugs shall be certified explosio boxes.	n proof when used with e	explosion and junction
4.0	Name Plate		
4.1	Each junction box shall have a fixed to it at a visible place bea shall also bear the stamp of certi	n anodised aluminium na aring the tag no. & enclos fying agency with certificat	me plate permanently sure. The name plate te number.
5.0	Shipping		Page 501 of 614

Page 591 of 644

MECON LIMITED	STANDARD SPEC	IFICATION	
REGD. OFF: RANCHI 834002	ELECTRICAL & INSTR OIL & GAS SBU	RUMENTATION J, DELHI	18 मेकान 9007:2000 Contract
	SPECIFICATION FOR	DOCUMENT NO.	Page 7 of 7
TITLE	JUNCTION BOXES AND CABLE GLANDS	MEC/S/05/26/06	REVISION: 0
			EDITION: 1

- 5.1 All threaded openings shall be suitably protected to prevent entry of foreign material.
- 5.2 All threaded components shall be protected with plastic caps to prevent damage of threads.

# 6.0 Rejection

Vendor shall furnish his offer in detail, with respect to every item of the purchaser's specifications. Any offer not conforming to this shall be summararily rejected.

# ELECTRICAL & INSTRUMENTATION SECTION MECON LIMITED DELHI - 110 092



# SPECIFICATION

FOR

# SIGNAL CABLES

SPECIFICATION NO.: MEC/S/05/26/07

D:\projects\std & spec\Standard spec on Ritish (Ritish)\Standard spec\Specification for Signal Cabling07.docx

Page 593 of 644



REV-0

Page 1 of 7

# <u>C O N T E N T S</u>

- 1.0 GENERAL
- 2.0 CONSTRUCTION
- 3.0 ELECTRICAL CHARACTERISTICS
- 4.0 TESTING
- 5.0 SHIPPING
- 6.0 REJECTION



REV-0

Page 2 of 7

- 1.0 <u>GENERAL</u>
- 1.1 <u>Scope</u>:

This specification together with the job Specifications attached herewith forms the requirements for design, materials, manufacturing, testing and shipping of PVC insulated signal cables.

1.2 <u>Standards</u>:

The cables shall conform to the latest editions of the various standards mentioned in the specification.

In case of any conflict between any standard and this - specification the matter shall be referred to the purchaser before proceeding with the manufacture of the cables.

- 1.3 <u>Bids</u>:
- 1.3.1 Vendor's quotation shall include the following as a minimum.

Completed job specs., Pair identification method, type test certificates, technical literatures, various testing methods and cross sectional dimensional drawings. All information/data shall be in English language.

- 1.3.2 Vendor's quotation shall include a list of deviations if any from purchaser's specifications and shall also indicate the reasons for such deviations for consideration to arrive at mutually agreed deviations. However vendor shall note that no deviation shall be accepted in respect of the permissible limits of resistance capacitance and L/R ratio of cables.
- 1.3.3 Vendor shall quote unit price per meter for each type of cable.
- 1.4 <u>Instructions to Bidder</u>:
- 1.4.1 The quantity indicated against each type of cable in the job specification may vary by<u>+</u> 25% at the time of placement of order. Vendor shall confirm that there shall be no price implication on this account in unit prices type wise.
- 1.4.2 Drum length for each type of cable shall be 500 to 1000 meters. Vendor shall indicate the maximum drum length possible for each type of cable in his bid. Exact requirements of drum length will be specified after purchase order during detailed engineering and vendor shall confirm that the same shall not affect the price or delivery schedule. The actual produced drum length shall not vary by more than  $\pm$  5% from the value indicated in the purchase order.



REV-0

Page 3 of 7

1.4.3	Tolerance over the total ordered length shall be as follows: ± 5% for total length less than 5 km. ± 2% for total length 5 km or more.
1.4.4	Cable dimensions furnished by vendor in his bid shall be firm. Vendor shall comply with all the values during execution.
1.5	All cables shall be suitable for laying in open air, corrosive hydrocarbon plant atmosphere, direct sun and in trenches. The cable shall also be designed for prolonged use in tropical atmosphere.
1.6	On demand vendor shall furnish documents such as invoice and test certificates to prove the quality and composition of the materials used for manufacturing the cable to the satisfaction of client/ consultant or authorized representative during various stages of expediting and inspection.
2.0	Construction:
2.1	Type 1 (single pair/triad shielded cable)
2.1.1	Each core shall be 1.5 sq. mm made of 7 stranded annealed electrolytic copper conductor. Each strand shall be 0.53 mm dia.
2.1.2	Primary insulation shall be $85^{\circ}$ C polyvinyl chloride (PVC) as per IS-5831 Type C. Thickness shall be 0.5 mm minimum.
2.1.3	Each wire shall have twisted cores and No. Of twists shall be not less than 10 per meter. Color of cores insulation shall be black blue in a pair and black, blue and brown in a triad.
2.1.4	Individual pair and triad shall be shielded. Shield shall be Aluminium backed by mylar/polyester tape bonded together with the metallic side down helically applied with either side 25% overlap and 100% coverage. Minimum shielded thickness shall be 0.05mm. Drain wire shall be 0.5 sq.mm multistrand bare tinned annealed copy conductor. The drain wire shall be in continuous contact with Aluminium side of the shield.
2.1.5	Inner and outer jacket shall be made of extruded flame retardant low smoke 90°C PVC to IS 5831-Type ST2. Oxygen index of PVC shall be 30. Temp. Index shall be over 250° C. Inner jacket color shall be black. Outer jacket color shall be black except for cable to be
	used in intrinsically safe systems it shall be light blue. Rip cord shall be provided.
2.1.6	Armour over inner jacket shall be of galvanised steel wire/flat as per IS-1554 part-I.

MECON LIMITED Delhi		ELECTRICAL & INSTRUMENTATION SECTION DELHI	SPECIFICATION FOR SIGNAL CABLES	मेकॉन 3001 Carty 51
SPECIFICATIO	ON NO. : M	EC/S/05/26/07	REV-0	Page 4 of 7
<u>г</u>				
2.1.7	A pair 250 m	or triad identification shall k m as per vendor's standard.	be with numbers at interva	l of not more than
2.1.8	Toleran cables	ice in overall diameter of cable with OD less than 30mm and $\pm$	e shall be within ± 2mm ov 3mm for cables with OD more	er offered value for e than 30mm.
2.2	Type-II (Multipa	: air/multitraid cable with individu	al pair shield and overall shie	ld)
	The cal sq.mm be of 0.	ble shall be same as single pair made of 16 strands of anneale .2mm dia.	shielded cable except conduct d electrolytic copper conducto	ctor size shall be 0.5 or. Each strand shall
	Additio	nal feature shall be as follows:		
2.2.1	Overall with th shield t and sha	shield shall be of Aluminum ba e metallic side down either s hickness shall be 0.075mm Dra all be in continuous contact with	acked up by mylar/polyester to side 25% overlap and 100% in wire shall be similar to indiv the Aluminium side of the ov	ape helically applied coverage. Minimum vidual pair drain wire erall shield.
2.2.2	Overall twist of all pair/triads shall be as per vendor's standard.			
2.2.3	A pair wire sh 0.4 mm	of communication wire shall be all be 0.5 sq. mm of plain ann n thick 85°C PVC insulation. Insu	e provided for multipair/mul ealed single or multistrand co Ilation shall be green and red	titriad cables. Each opper conductor with colour coded.
2.3	Type-II (Multipa	I air/Multitriad cable with only ove	erall shield)	
	These cables shall be same as type-II cables except that the individual pair/triad shan not have shielding.			vidual pair/triad shall
2.4	Type-I\ (Multipa	/ air/ multitriad cable with individe	ual pair shield and overall shie	ld)
	The cal strande	ole shall be same as Type II ex ed annealed electrolytic copper o	cept conductor size shall be 1 conductor. Each strand shall b	.5 sq.mm made of 7 be 0.53 mm dia.
2.5	Type-V (Multipa	air/ Multitriad cable with overall	shield only)	
	The cal	ble shall be same as type IV e elding.	xcept that the individual pair	/triad shall not have



REV-0

Page 5 of 7

3.0	Electrical Characteristics
3.1	Maximum d.c. resistance of the conductor of the completed cable shall not exceed 12.3 ohms/km at $20^{\circ}$ C for cables with 1.5 sq.mm conductor and 39.7 ohms/km at $20^{\circ}$ C for cables with 0.5 sq.mm conductor.
3.2	Capacitance
3.2.1	Mutual Capacitance
	The mutual capacitance of the pairs or adjacent cores shall not exceed a maximum of 250 pF/Meter at a frequency of 1KHz.
3.2.2	Capacitance between any core or screen.
	The capacitance between any core or screen shall not exceed a maximum of 400 pF/Meter at a frequency of 1KHz.
3.3	L/R ratio of adjacent core shall not exceed 40 micro henry/ohm for cables with 1.5 sq. mm conductor & shall not exceed 25 micro henry/ohm for cables with 0.5 sq. mm conductor.
3.4	The drain wire resistance including shield shall not exceed 30 ohms/km. Insulation resistance on drum length @ 20 Deg. C shall be 25 M-Ohm/Km (as per BS 5308 Part-2).
3.5	Electrostatic noise rejection ratio shall be over 76 dB.
3.6	High Voltage Test shall be carried out between conductors & between core & screen @ 1 KV-AC for one minute (as per BS 5308 Part-2).
3.7	Flammability test shall be as per IEC part 3.
4.0	Testing
4.1	<u>Type test</u> : Cable shall be flame retardant to IEC 332 Part III Cat. A. For qualification, certificates from third party or client /consultants authorised representative for this test shall be furnished by vendor for cables similar to those being offered.
4.2	<u>Routine tests</u> : (To be carried out by vendor during various stages of manufacture. Purchaser / purchaser's representative shall review the related documentation).
4.2.1	Insulation and jackets: All tests as per IS-5831 except insulation resistance, voltage and spark test shall be as per BS-5308. Part-II(1986)



REV-0

Page 6 of 7

4.2.2	Armour test as per IS-3975.
4.2.3	Conductor resistance.
4.2.4	Cable capacitance and L/R ratio.
4.3	<u>Acceptance test</u> (The test shall be carried out in the presence of purchaser or his authorised representative.)
4.3.1	Continuity test
4.3.2	Voltage test as per BS-5308 Part II.
4.3.3	Conductor resistance, Insulation resistance and drain wire resistance.
4.3.4	Cable capacitance and L/R ratio test.
4.3.5	Tests for uniformity of galvanisation of armour as per IS 2633.
4.3.6	Oxygen and temp. index test as per ASTM D 2863 & Flammability test.
4.3.7	Dimensional check for overall diameter and under armour/over armour diameter.
4.3.7	Visual Check of outer sheath marking (Manufacturer's name, voltage grade, type & size of cable, length marking, year of manufacturing, etc.), colour coding, construction as per datasheet / tech. spec. etc.
4.3.8	Overall finish check. In case of any lump purchaser will have the right to cut outer sheath for lump portion and reject the cable.
4.3.9	Check of drum length and overall length tolerances.
	Immediately after completion of electrical tests, the ends of the cable shall be sealed to prevent ingress of moisture with suitable PVC/Rubber caps.
5.0	Shipping
	Cables shall be despatched in wooden drums, securely battened with take off end fully protected against damage. Each drum shall be marked with following:
	<ul> <li>a. Manufacturer's name.</li> <li>b. Details of the cable.</li> <li>c. Length of the cable contained in the drum in metres.</li> <li>d. Gross wt.</li> <li>e. Direction of rotation of drum for unwinding by means of an arrow.</li> <li>f. Purchase order no.</li> </ul>

MECON LIM Delhi	ITED	ELECTRICAL & INSTRUMENTATION SECTION DELHI	SPECIFICATION FOR SIGNAL CABLES	र्मित स्वॉन 30 3001 CONVS	
SPECIFICATION NO. : MEC/S/05/26/07		EC/S/05/26/07	REV-0	Page 7 of 7	
6.0	Rejection				
	Vendor shall make his offer in detail with respect to every item of the purchaser's specifications. Any offer not conforming to this shall be summarily rejected.				

# GENERAL SPECIFICATIONS FOR INSTRUMENTATION

SPECIFICATION NO.: MEC/S/05/26/08



ELECTRICAL & INSTRUMENTATION (OIL & GAS SBU) MECON LIMITED DELHI 110 092

MECON LIMITED	STANDARD SPECIFICATION		
REGD. OFF: RANCHI 834002	ELECTRICAL & INST	के मेकान	
	OIL & GAS SBU	J, DELHI	001:2000 Contre
	SPECIFICATION FOR	DOCUMENT NO.	Page 1 of 17
TITLE	INSTRUMENTATION	MEC/S/05/26/08	REVISION: 0
			EDITION: 1

# **AMENDMENT STATUS**

SI. No.	Clause / Paragraph / Annexure / Exhibit / Drawing Amended	Page No.	Revision	Date	By (Name)	Verified (Name)

MECON LIMITED		STANDARD SPEC	IFICATION		
REGD. OFF: RANCHI 834002		<b>ELECTRICAL &amp; INSTRUMENTATION</b>			मेकान
	SDEC	OIL & GAS SBU	J, DELHI	Pag	2000 Compo
TITLE	INS		MEC/S/05/26/08	Pay RE\	
			FILC/ 3/ 03/ 20/ 00	ED	ITION : 1
		<u>C O N</u>	<u>TENTS</u>		
1.0	GENERA	L			
2.0	DEFINITIO	ONS			
3.0	SCOPE C	OF WORK			
4.0	DESCRIP	TION OF WORK			
5.0	DRAWING	GS AND DOCUMENT	S TO BE PROVIDED B	Y OWNEF	R
6.0	DRAWING	GS AND DOCUMENT	S TO BE PROVIDED B	Y CONTR	ACTOR
7.0	SCRAP A	ND EXCESS MATER	TIAL		
8.0	SPECIAL	INSTRUCTIONS TO	CONTRACTOR		
PREPARED BY	:	CHECKED BY:	APPROVED	BY:	ISSUE DATE

MECON LIMITED	STANDARD SPEC	IFICATION	
REGD. OFF: RANCHI 834002	ELECTRICAL & INSTRUMENTATION OIL & GAS SBU, DELHI		10 मेकान 9001:2000 Company
	SPECIFICATION FOR	DOCUMENT NO.	Page 3 of 17
TITLE	INSTRUMENTATION	MEC/S/05/26/08	REVISION : 0
			EDITION: 1

# 1.0 GENERAL

# 1.1 INTRODUCTION

- 1.1.1 This part of the technical specifications covers in general, definitions, standards, scope of works, specifications of work, documentation, scope of supply of materials and scrap and excess materials and different requirements to be adhered to during the course of execution of instrumentation works.
- 1.1.2 Instrumentation works shall be performed in accordance with this technical specification and various other drawings and schedules supplied during the execution and time to time instructions from Engineer-in-Charge or his authorised representative(s) during the progress of the work.

## 2.0 DEFINITIONS

# 2.1 <u>MANIFOLDS</u>

- 2.1.1 For close coupled instruments "Manifold" shall mean complete piping of instruments from first block valve upto the instruments, if the distance of the Instrument is within 2 feet (0.6m), from the Instrument tapping. If the distance of instrument is more than 2 feet (0.6m) from primary tapping such as orifice, then the installation is to be considered under remote installation.
- 2.1.2 For remote mounted instrument, "Manifold" shall mean the assembly of nipples, valves and fittings around the instrument to form a block and bleed or by pass manifold or drain manifold as the case may be. These shall be generally according to the hook up drawings enclosed with tender. Wherever the instruments are with 3--way-valve manifold, this definition shall not be applicable as 3-way manifold forms part of instrument.

## 2.2 FIRST BLOCK VALVE

First block valve shall mean the valve/valves that are mounted directly on equipment, columns, pipe, standpipe etc. and shall be operated to isolate the instrument and connected instrument piping from the above items.

#### 2.3 <u>SUPPORTS</u>

Supports shall mean the MS angles, flats, channels that are generally provided to support the main cable ways, cable ducts, junction boxes, angle trays, perforated trays, instrument piping, signal tubing, instrument air supply lines etc., at specified intervals from the structures, concrete columns etc. to keep all items firmly secured against vibration, warping, bending etc.

MECON LIMITED	STANDARD SPEC	CIFICATION	
REGD. OFF: RANCHI 834002	ELECTRICAL & INSTR	मेकान क	
	OIL & GAS SBU	J, DELHI	001:2000 Cont
	SPECIFICATION FOR	DOCUMENT NO.	Page 4 of 17
TITLE	INSTRUMENTATION	MEC/S/05/26/08	REVISION: 0
			EDITION: 1

# 2.4 <u>SCRAP</u>

## 2.4.1 <u>Salvageable scraps</u>

Salvage scrap shall mean lengths of tubes, pipes, multicables, other cables etc. that can be used one time or other at later date and normally they are recovered from the cut-pieces of tubes, pipes, multicables, cables, etc.

#### 2.4.2 Non Salvageable Scrap

Non salvageable scrap shall mean the lengths of tubes, pipes, multicables, cables, etc. that cannot be used at all one time or other.

## 2.5 <u>Standards</u>

The instrumentation erection and calibration works shall be carried out generally in accordance with various international and Indian standards in instrumentation listed below but not limited to the following:

- 2.5.1 API -RP-550 Manual on Practices for instrumentation.
- 2.5.2 ISA standards and Practices for instrumentation
- 2.5.3 Instrumentation hook-up standards enclosed.
- 2.5.4 Instrumentation supports standard enclosed.
- 2.5.5 Manufacturer's standards and Practices.

## 3.0 SCOPE OF WORK

- 3.1.0 The Scope of work shall consist of supply of instrument items (as per schedule of quantities/rates and SCC), instruments, their erection, testing, calibration and commissioning and making it ready for commercial operation. The scope covers various jobs listed under the schedule of quantities/rates. However to ensure proper execution and completeness of instrument--work any or all of the following shall also form the part of the scope and shall be covered in the quoted rates.
- 3.1.1 Fabrication of pipe nipples, including threading whenever required.
- 3.1.2 Fabrication of seal pot/syphon/drain pot as per standards. Filling of seal pots with filling liquids as per instructions from Engineer-in-charge.
- 3.1.3 Back/seal welding of screwed fittings as required by standards.

Page 605 of 644

MECON LIMITED	STANDARD SPECIFICATION			
REGD. OFF: RANCHI 834002	ELECTRICAL & INSTR	के मेकान		
	OIL & GAS SBL	J, DELHI	9001:2000 Comps	
	SPECIFICATION FOR	DOCUMENT NO.	Page 5 of 17	
TITLE	INSTRUMENTATION	MEC/S/05/26/08	REVISION: 0	
			EDITION: 1	
3.1.4	Laying of cable underground in back filling.	cluding excavation, sand t	filling, brick laying and	
3.1.5	Connection of purging devices drawings.	for instruments to the sy	stems provide as per	
3.1.6	Civil works including the cas instruments supports where pave	ating of foundation as ed surface do not exist.	per requirements for	
3.1.7	Minor civil works like chipping of pavement and grouting on the pavements the instrument panels/supports/stanchions, and chipping and refilling of the pavement for conduits.			
3.1.8	Sealing of cables/ tube entries into the control room after laying and testing of all tubes, cables etc.			
3.1.9	Degreasing of handwheels of control valves, stud bolts, nuts of side and bottom flange of control valves, orifice plates, other primary elements flanges, oxygen service impulse lines, instruments as per manufacturers instructions and other items as required by Engineer- in-charge.			
3.1.10	Rotation of control valve bonnet	wherever required.		
3.1.11	Reversing the action of contraccessories or in positioner when	ol valves either the rep rever required.	placement of springs,	
3.1.12	Minor modification/repairs required to be done on the instruments namely, changing the dial, glasses for pressure gauges, temperature gauges and other instruments, replacement of rotameter tubes, level gauge glasses, replacement of damaged signal tubes, threads, couplings etc.			
3.1.13	Painting of all structural suppor etc, as per painting specification.	ts for trays, pipes, junctic	on boxes, instruments,	
3.1.14	Identification with approved c manifold connected with alarm/ items shall be carried out as per	olour of paint the instri trap circuit. Also, punchin instructions of Engineer-in	uments/impulse, lines ng of tag numbers on a-charge.	
3.1.15	Coordination with mechanical ar line/ vessels/ equipment mou assemblies, turbine meters, PD switches etc. which involves tubes/cables, reconnection for al	nd other sub-contractors fo inted instruments like o meters, level transmitter removal of instrumen ignment proper installatior	or proper installation of control valves, orifice rs, level gauges, level its, disconnection of n etc.	
3.1.16	Drilling holes on all panels, shut panels pneumatic enclosures etc	t down cabinets, power si c., for cables/ multitubes/ g	upply cabinets, control lands/ groomats.	

Page 606 of 644

MECON LIMITED	STANDARD SPEC	IFICATION		
REGD. OFF: RANCHI 834002	ELECTRICAL & INSTRUMENTATION		मेकान	
031002	OIL & GAS SBL	J, DELHI	9001:2000 ComP	
	SPECIFICATION FOR	DOCUMENT NO.	Page 6 of 17	
TITLE	INSTRUMENTATION	MEC/S/05/26/08	REVISION : 0	
			EDITION: 1	
3.1.17	Grounding of shield of all shield provided in the control room/loca	ded cables to respective Il panel/thermocouple head	instruments earth bus d.	
3.1.18	Laying and termination at both e control room/ local panel to instru	nds between instrument e ument earth pit provided b	arth buses provided in y others.	
3.1.19	Supply of all types of consumable	es required for the execution	on of the job.	
3.1.20	Submission of monthly material appropriation statements for cables, piping materials fittings, including the quantity issued and expended in standard proforma.			
3.1.21	Completion of owners drawings/documents, as per the execution of work at site.			
3.1.22	Preparation and submission of as built drawings as required.			
3.1.23	Start-up and commissioning.			
3.1.24	Submission of final material app by the owner.	Submission of final material appropriation statements for all the materials issued by the owner.		
3.1.25	Any other work not mentioned al works.	pove, but required for the p	proper execution of the	
3.1.26	Where requested by ow representatives, all or any of the shall also be performed on particular to the shall by owner or by others.	ner/Engineer-in-charge e works detailed above ar ckage units, local panels/	or his authorised nd schedule quantities cabinets/gauge board	
3.1.27	Sealing of safety valves/switche the presence of Engineer- in-cha	es with standard lead sea arge.	ls after final setting in	
4.0	DESCRIPTION OF WORK			
4.1.0	INSTRUMENT PIPING			
4.1.1	All primary piping shall be insta follow installation standards in standard, the instruction of the E	lled in the best workman each case. Where th ngineer-in-charge shall be	like manner and shall ere is no installation followed.	
4.1.2.1 Horiz	zontal and vertical lines shall be ins	stalled using levels and plu	imo bobs.	
4.1.3	Unless otherwise specified in th on the horizontal runs.	ne drawings pipelines sha	ll have a slope of 8%	

MECON LIMITED	STANDARD SPEC	IFICATION		
REGD. OFF: RANCHI 834002	ELECTRICAL & INSTRUMENTATION OIL & GAS SBU, DELHI		मिकान मेकान से01:2000 Contrain	
	SPECIFICATION FOR	DOCUMENT NO.	Page 7 of 17	
TITLE	INSTRUMENTATION	MEC/S/05/26/08	REVISION : 0	
			EDITION: 1	

- 4.1.4 All welding shall be carried out as per welding procedures and codes with electrodes approved by Engineer-in-charge. Only qualified welders approved by Engineer-in-charge shall carry out welding. Charges for non-destructive testing like radiography, Dye penetration tests, post heat treatment tests and stress relieving shall be carried out on the basis of actual man hours spent towards these works and man-hour charges with cost of all materials, test equipments, etc. shall be used. However, any materials like electrode, equipments, testing charges for various tests, etc., required for the initial qualification of the welder/welders shall be or the scope of the contractor.
- 4.1.3.1Pipe shall be bent using pipe benders only and any bending will be totally rejected. Pipes shall be cut using pipe cutting device. Hot cutting will not be allowed.
- 4.1.6 Piping for steam tracing shall be installed according to the standards and avoiding condensate pockets.
- 4.1.7 All threaded joints shall be jointed with Teflon tape and no other pipe jointing compound shall be used except on high temperature service where graphited sealing compounds shall be used.
- 4.1.8 All primary piping shall be properly supported at regular intervals of 1.0 meters. Angle supports shall be fabricated from 40mmx40mmx5mm MS angles as minimum.
- 4.2.0 <u>PVC COVERED/BARE TUBE (COPPER/SS/ALUMINIUM)</u>
- 4.2.1 Single copper/SS/Aluminium tubes shall be laid as per standards on trays. Fabricated out of 2.5 mm thick perforated steel plate. The width of the trays shall be selected as per the number of tubes laid. Tubes shall be clamped to the trays at every 300 mm using clamps made of galvanized steel/Aluminium strips. The practice of flattening tubes for clamping purposes shall be avoided. In case of PVC covered tubes, any exposed portion at ends and connection shall be neatly taped to appropriate thickness.
- 4.2.2 Trays shall be properly supported either from any rigid steel structure or concrete member as detailed under trays and supports below.
- 4.2.3 All male/female tube connectors shall be installed with Teflon tape only. Identification tag plates/ferrules shall be provided on either side of copper tubing as per tubing/junction box schedules. Ferrules shall be single sleeve type with letters and numbers neatly printed.

MECON LIMITED	STANDARD SPEC	IFICATION	
REGD. OFF: RANCHI 834002	ELECTRICAL & INST OIL & GAS SBU	मिकान मेकान 9007:2000 Contract	
	SPECIFICATION FOR	DOCUMENT NO.	Page 8 of 17
TITLE	INSTRUMENTATION	MEC/S/05/26/08	REVISION: 0
			EDITION: 1

- 4.3.1 Multiple cables/ multitubes shall always be installed on ducts/trays and properly clamped. At every vertical drop to junction boxes, they shall be clamped at more frequency intervals (Maximum of 300mm). They shall be connected inside junction boxes strictly according to the number system as mentioned in cable schedule. At bends minimum radius shall be maintained as per manufacturer's standard. The angle tray supports shall be fabricated from 40mmx40mmx5mm angles minimum size.
- 4.3.2 Identification tags shall be provided on either end of multitubes, multicore cables as per cable/tubing/cable schedules. Engraved tag plates or PVC ferrules shall be used for identification of tubes/cables.
- 4.3.3 All Multitubes and Multicables shall be cut after the exact site measurements are taken between ends and the cable/tube drums shall be selected before cutting the lengths so as to avoid any wastage.
- 4.3.4 In the field, the cables shall be laid in perforated trays as per layout drawings. Cables shall also be buried or laid in concrete trenches. Inside control room, these shall be laid in concrete trenches or under false floorings.
- 4.3.5 In the field, the cables shall be laid in perforated trays as per layout drawings. Cables shall also be buried or laid in concrete trenches. Inside control room, these shall be laid in concrete trenches or under false floorings.
- 4.4.0 INSTALLATION OF INSTRUMENTS
- 4.4.1 All instruments shall be generally installed on supports as per installation standards in each case, and shall be accessible.
- 4.4.2 Receiver gauges shall be mounted on instrument support itself as per tubing hook up standards.
- 4.4.3 Filter regulators shall be mounted on the instruments support itself below the instruments or on the control yoke.
- 4.5.0 INSTRUMENT AIR SUPPLY
- 4.5.1 The main instrument air header in each area is laid by other contractor. Air supply from the main air header take off valve to individual instrument shall be through either galvanized steel pipe or 1/4" OD PVC covered copper tube or SS tubes.
- 4.5.2 Individual takes off valves shall always be located on top of the main air header. Unions shall be provided at convenient locations. There shall be one isolation valve at each instrument end. The galvanised pipe shall be supported at a minimum interval of 1000 mm with 40mmx40mmx5mm Magaretiges of 6#4nal

MECON LIMITED	STANDARD SPEC	IFICATION	
REGD. OFF: RANCHI 834002	ELECTRICAL & INST OIL & GAS SB	मि मेकान मेकान 8007:2000 Contrin	
	SPECIFICATION FOR	DOCUMENT NO.	Page 9 of 17
TITLE	INSTRUMENTATION	MEC/S/05/26/08	REVISION: 0
			EDITION: 1

connection to be instrument shall be copper/SS tubing as per tubing hookup standards.

4.5.3 Teflon tapes shall be used on all threaded joints.

## 4.6.0 INSTRUMENT STEAM TRACING

- 4.6.1 The mainsteam header in each area is laid by the other contractor. From the main steam header take off valve, steam to individual instrument shall be taken through carbon steel pipes supported at regular intervals. Steam tracing around individual instruments shall be to copper tubes. After steam tracing, the line is connected to the drain funnel through individual steam trap/condensate return header/tapper point as the case may be.
- 4.6.2 Electrical tracing shall be done by others.

## 4.7.0 PERFORATED TRAYS AND SUPPORTS

- 4.7.1 The perforated trays / angle trays shall be properly supported at a regular interval of max. 1000mm from insert plates or steel structures. Wherever insert plates are not available supports on concrete structures on ceiling shall be fixed with a minimum 10mm diameter expansion bolts. Angle supports for perforated trays/angle trays shall be fabricated from 40mmx40mmx5mm M.S. angles minimum size.
- 4.7.2 All supports shall be cut with hacksaw and any work executed by gas cutting for cutting and drilling holes will be totally rejected. Free ends of angle support shall not have sharp edges and shall be properly rounded off.
- 4.7.3 Perforated trays/angle trays shall be used for branching cables and tubes from main trays. Perforated trays shall be used for branching cables and tubes from main trays. Perforated trays shall be fabricated out of 2.5 mm perforated steel sheet. Width of trays shall be selected according to number of tubes and cables. Trays shall be laid generally as per site conditions with the approval of Engineer-in-charge.
- 4.8.0 LAYING OF CABLES
- 4.8.1 All cables shall be laid in accordance with installation drawings and cable schedules. Before laying, cable/multicable on drums shall be meggered and tested to ascertain the transit damages.
- 4.8.2 All cables routes shall be carefully measured and cables cut to the required lengths, leaving sufficient amount for the final connection of the cable to the terminals on either end. The various cable lengths cut from the cable reels shall be carefully selected to prevent undue wastage of cables. Sufficient extra length Page 610 of 644

MECON LIMITED	STANDARD SPEC	IFICATION		
REGD. OFF: RANCHI 834002	ELECTRICAL & INSTR	गुरु मेकान		
	OIL & GAS SBL	J, DELHI	2001:2000 Con	
	SPECIFICATION FOR	DOCUMENT NO.	Page 10 of 17	
TITLE	INSTRUMENTATION	MEC/S/05/26/08	REVISION: 0	
			EDITION: 1	
	of cable shall be kept at the term	inal on points.		
4.8.3	Cables shall have complete uncu	ut lengths from one termina	al to the other.	
4.8.4	All cables shall be identified close to their termination point by cables number as per cable schedules/junction boxes schedules. PVC ferrule/tag plate shall be used and these identification tags shall be securely fastened to the cables.			
4.8.5	4.8.5 All cores of electrical cables shall be identified by their wire numbers by means of the PVC ferrules. Wire numbers shall be as per schedules. All temporary ends of cables shall be protected against dirt and moisture. For this purpose, ends of all PVC insulation cables shall be taped with an approved PVC or rubber insulating tape. Use of function type or other fabric type is not permitted.			
4.8.6	The cable shall be bent in a large radius. Cables installed above ground shall be run exposed on walls, ceilings, structures and shall run parallel or at right angles with beams, walls or columns.			
4.8.7	Cables shall be rigidly supported on structural steel and masonary individually or in groups as required using galvanised clips, multiple cable supports or cable trays. If drilling of steel must be resorted to, approval must be obtained and steel must be drilled where the minimum of weakening of the structure will result.			
4.8.8	All special cables and power instrument without any junction b	supply cables will be lai boxes, unless otherwise sp	d directly to the field ecified.	
4.8.9	While laying cable in trenches or low signal cables like alarm, and turbine meters, compensating ca cables.	r burying them care shall b lyser cables, special cable able etc. are separated fro	be taken to ensure that es, special cables from om other power supply	
4.8.10	Each underground cable (either in concrete trenches or burried) shall be provided with identifying tag of lead securely fastened every 30m of its underground length with atleast one tag at each end before the cable			
4.8.11	Directly buried cables shall be wherever specified in layout draw width to accommodate all cables trench bottom shall be filled with of cables shall be covered with pressed. A protective covering placed flat on the final layer of trench shall be then back filled we every group of cable laying and insulation test in the presence of defective should be replaced be route markers indicating number	e laid underground in ex- wings. Trenches shall have s correctly spaced. Before 100 mm layer of sand an 150 mm of sand on top ar of 75 mm thick second-cla sand and cable. The re- vith soil compacted and leve d before sand filling, every of Engineer-in-charge. An efore the next groups of de- er of cables, depth and d	cavated cable trench ve sufficient depth and cables are placed the d leveled. Each layer and sand shall be lightly ass red bricks shall be maining portion of the veled. On complete of y cable shall be given by cable proved to be cables are laid. Cable irequipe will be plaged	

			$\bigcirc$
MECON LIMITED	STANDARD SPECIFICATION		
834002	ELECTRICAL & INSTRUMENTATION		वि मेकान
	OIL & GAS SBU	DOCUMENT NO	Page 11 of 17
	INSTRUMENTATION	MEC/S/05/26/08	
	INSTROPLETATION	MLC/ 3/ 03/ 20/ 00	FDITION : 1
enroute, on crossovers/turnings, etc. to mark the cable route.			
4.8.12	At each road crossings and other places, where cables enter pipe sleeves, adequate bed of sand shall be given so that the cables do not slack and get damaged by pipe ends after pack filling.		
4.8.13	At the entry into concrete blocks at road crossings cable loops shall be provided at either end to prevent any damage to cables. Each cable shall have one tag at each end before the cable enters/leaves conduct pipes		
4.8.14	After laying of all the cables and multitubings, cables, the cable entry to control room shall be suitably filled and sealed so as to achieve a positive seal against the entry of gas/water.		
4.8.15	All cables and tubes shall be laid in accordance with the layout drawings with sand and precast concrete slabs shall be placed on the trench.		
4.8.16	On completion of cable laying in concrete trenches, the trenches shall be filled with sand and precast.		
4.9.0	EARTHING		
4.9.1	Earthing of junction boxes, local cabinets as per the documents and instruction from Engineer-in-charge.		
4.10.0	PAINTING		
4.10.1	This part of the specification is applicable to cable ducts, MS cable ways, angle trays, instrument supports, perforated trays, all structural supports for the above items, etc.		
4.10.2	The surface to be painted shall be thoroughly cleaned with wire brush, sand paper to remove all scales. After cleaning, the surface is painted with one coat of red oxide zinc chromate primer conforming to IS- 207 and allowed to dry completely.		
4.10.3	Primer coated surface is painted with one coat of paint to the colour nearest to the final paint and allowed to dry. The colour number shall be specified from IS-5.		
4.10.4	Final second coating shall be with the paint of desired colours and shall be selected from IS-5.		
4.10.5	It shall be noted that final second coating of external surfaces not covered by cables, copper tubes etc. shall be applied just before handling over the plant or commissioning of the plant whichever is earlier.		
4.10.6	The name of manufacturer, colour and quality of all types of pp்தகு நகற்கொகுகியும்		
MECON LIMITED	STANDARD SPEC		
-----------------------------	-------------------------------------	------------------------------	---------------
REGD. OFF: RANCHI 834002	ELECTRICAL & INSTF OIL & GAS SBU	00 मेकान 9007:2000 Comman	
	SPECIFICATION FOR	DOCUMENT NO.	Page 12 of 17
TITLE	INSTRUMENTATION	MEC/S/05/26/08	REVISION : 0
			EDITION: 1

subject to approval of Engineer-in-charge.

#### 4.11.0 <u>TESTING</u>

- 4.11.1 Electrical cables for signal power supply alarms, and compensating cables for thermocouples; resistance thermometer cables shall be checked for megger values and continuity before proper termination and ferruling.
- 4.11.2 Testing shall be carried out after the installation of instrument with primary piping complete in all respects and approved by Engineer-in-charge.
- 4.11.3 Primary piping shall be tested hydraulically pneumatically to 1.5 times the operating pressure after isolating the instruments. Flushing of piping shall be carried out as per instructions of Engineer-in-charge. Lines shall be blown after hydro-testing. All external displacement /float type level instrument level gauges shall also be tested as per instructions of Engineer- in-charge.
- 4.11.4 Tubes and air line shall be tested with compressed air to 7 kg/cm<sup>2</sup> upto the filter regulator. The down steam side of the filter regulator shall be tested for 1.5kg/cm<sup>2</sup>. The lines shall be blown with the instrument air upto the regulator for 15 minutes to remove any traces of oil, dust & moisture. All lines shall be checked with soap solution and bubbler unit for possible leak at joints. After pressurizing, source shall be cut off and rate of fall in pressure shall be less than 1 p.s.i. for each 100 ft. of copper tubing for a test period of 2 minutes as per I.S.A.R.P.7-1 "Pneumatic control circuit pressure test".
- 4.11.5 All test results shall be recorded in the approved format.

#### 4.12.0 <u>CALIBRATION</u>

- 4.12.1 All instruments shall be calibrated strictly as per manufacturer's instructions prior to installation. The scope of calibration includes all field and control rooms of all types namely, pneumatic, electronic, electrical etc.
- 4.12.2 Contractor shall use his own oil free instruments, air compressor for calibration purposes.
- 4.12.3 The level switches (external cage type) shall be set by filling the cage with water to the desired alarm/trip level, while setting the switches, it shall be ensured that the micro switches do not reset for full rated travel of the float.

Page 613 of 644

MECON LIMITED	STANDARD SPEC			
REGD. OFF: RANCHI 834002	REGD. OFF: RANCHI 834002 ELECTRICAL & INSTRUMENTATION OIL & GAS SBU, DELHI			
	SPECIFICATION FOR	DOCUMENT NO.	Page 13 of 17	
TITLE	INSTRUMENTATION	MEC/S/05/26/08	REVISION: 0	
			EDITION: 1	

4.12.4 Control valves and positioners shall be checked for hysterisis and linearity and calibration for rated strokes. Prior to calibration, valves shall be cleaned externally. The stem is then lubricated if required, and stroked few times to extreme positions of plug to ensure that movement is free from friction. The valve shall then be calibrated for rated stroke and linearity also. Subsequently the valves shall be checked for hysterisis to the accuracy of 1% FS with positioners and 5% FS without positioners.

Stroke speed has to be evaluated for all trip/shutdown valves.

- 4.12.5 All calibrations reading shall be recorded in the enclosed format and submitted to Engineer-in-Charge for approval. Where significant deviations from specifications are obtained, the matter shall be brought to the immediate notice of the Engineer-in- Charge for corrective actions.
- 4.12.6 Furnished hereunder is a list of recommended calibration and test equipments required as a minimum for calibration work. The contractor shall clearly state in his offers the complete list of calibration and test equipments along with the range, accuracy and quantity, which he proposes to use for this job. Contractor should also ensure that any equipment not listed below but required at the time of calibration shall be made available at his own cost.
- 4.12.7 All test equipments/kits shall be approved by NPL authorities.

4.12.7.1	Controller test stands	Mft. Standard
4.12.7.2	Indicator/recorder test stands	-do-
4.12.7.3	Squeeze bulb (Flow calibrator Range: 0-770, 10,000 mm wg.	-
4.12.7.4	Dead weight testers (Budenberg or equivalent) For ranges upto 350 kg/cm <sup>2</sup>	- +/- 0.1%
4.12.7.5	Gauge comparator for pressure gauges Rating : upto 350 kg/cm <sup>2</sup>	-
4.12.7.6	Oil bath for temperature calibrations max. Temp 350°C.	Mfr's Std
4.12.8.7	Standard Mercury in glass thermometers Range : -50 to + 50°C. 0 to 100°C (NPL certified) 0-250°C. 0-350°C	<u>+</u> 0.25%
4.12.7.8	Standard gauges for Ranges upto 350kg/cm <sup>2</sup>	<u>+</u> 0.25%
4.12.7.9	U-tube differential manometers/inclined	Page 614 of 644

			Lange -	
MECON LIMITED	IMITED STANDARD SPECIFICATION			
REGD. OFF: RANCHI 834002	ELECTRICAL & INSTR	के मेकान		
	OIL & GAS SBU	J, DELHI	"01:2000 Cov."	
	SPECIFICATION FOR	DOCUMENT NO.	Page 14 of 17	
TITLE	INSTRUMENTATION	MEC/S/05/26/08	REVISION: 0	
			EDITION: 1	
	tube manometer Static pr. rating	: 7 kg/cm <sup>2</sup>	+_ 1mm	
4.12.7.10	Single leg manometers Scale: -1500 mm water and 1500 Static pr. rating : 7 kg/cm <sup>2</sup> .	Single leg manometers Scale: -1500 mm water and 1500 mm hg. Static pr. rating : 7 kg/cm <sup>2</sup> .		
4.12.7.11	Decade resistance box	Decade resistance box		
4.12.7.12	Millimeters	<u>+</u> 0.05Mv		
4.12.7.13	Potentiometer (Cable of generating and measu	-		
4.12.7.14	Meggers 500V/1000V		-	
4.12.7.15	Air hydro pump/hydraulic pump		-	
4.12.7.16	Vacuum pump	Vacuum pump		
4.12.7.17	Instrument air compressor with find Regulators and deoilers.	-		
4.12.7.18	Current generator (instrument ch 4-20mA dc(YEW make or equiva	-		

#### 4.13.0 <u>LOOP TEST</u>

- 4.13.1 Loop test shall be performed after calibration of all instruments and leak test of signal lines. Loop tests are conducted to check the functional performance of all elements comprising the loop, thereby ensuring proper connections and operations.
- 4.13.2 Before proceeding for loop tests the calibration results of individual elements shall be recorded on the enclosed proforma and shall get it approved by Engineer-in-Charge for correctness of installation, measurements and calibration results.
- 4.13.3 Loop testing for all control loops shall be generally by sinRagio 615f of 644 ss

MECON LIMITED	STANDARD SPEC					
REGD. OFF: RANCHI 834002	ELECTRICAL & INSTR	मेकान				
054002	OIL & GAS SBU, DELHI					
	SPECIFICATION FOR	DOCUMENT NO.	Page 15 of 17			
TITLE	INSTRUMENTATION	MEC/S/05/26/08	REVISION: 0			
			EDITION: 1			
	conditions and shall fix points r scale inputs. Detailed procedur approval before proceeding with	namely 0%, 25%, 50%, 7 re shall be submitted to I the loop testing.	5% and 100% of full- Engineer-in-charge for			
	In case of shutdown system fie abnormality by disconnecting the systems are checked.	ld/receiver pressure switc e wires at terminal and fur	hes are simulated for action of all associated			
4.13.5	Performance of individual loops $\pm$ 1.5% where deviations exist, c form part of loop testing wherever	may be accepted for an ov ontractor shall recalibrate er required, at no extra cos	rerall accuracy of the instruments, which t.			
4.13.5.1	After the loop test is completerminations and connections rer	ete, the contractor shal moved for loop test.	ll connect back any			
4.13.7	A loop shall be considered as particular loop are complete and loop sheets being duly filled in Engineer-in-Charge and client.	handed over only after indicentified by Engineer-in- n all respects and approv	measurements in that Charge, in addition to ved and accepted by			
4.13.8	In case of loops in which certain agency, loop testing shall be per Any defect in the calibration of t it shall be rectified to the satis defect in calibration of the instru be rectified by the agency involv the other agency/agencies the satisfaction of Engineer-in-Charg contract.	instruments of the loops a formed in coordination wit the instrument in contractor faction of the Engineer-ir ments in the scope of oth red. After the calibration h e loop checking would ge, and this part covers u	are calibrated by other h the agency involved. or's scope is observed, n- Charge. However, er agency, same shall as been rechecked by be performed to the inder the scope of the			
4.13.9	Final certified loop sheets shall b	e submitted in 4 copies an	d one transparency.			
5.0.0	DRAWINGS AND DOCUMENT OWNER/ENGINEER IN CHARG	rs to be supplied f Ge	OR EXECUTION BY			
5.1.1	Piping ad Instrumentation diagra	ms.				
5.1.2	General layout plan for all unit instruments junction boxes indicated	s, showing all information ative routes of cables, main	n like position of field n ducts/cable trays.			
5.1.3	Cable schedules for alarm, sig cables, earthing guide lines.	gnal, shutdown, power s	upply and pneumatic			

5.1.4 Termination details/drawings for connecting at control room end.

Page 616 of 644

	STANDARD SPECIFICATION ELECTRICAL & INSTRUMENTATION OIL & GAS SBIL DELHI				
REGD. OFF: RANCHI 834002			मिकान मेकान 9007:2000 Company		
	SPECIFICATION FOR	DOCUMENT NO.	Page 16 of 17		
TITLE	INSTRUMENTATION	MEC/S/05/26/08	REVISION: 0		
			EDITION: 1		
5.1.5	Individual Instrument specificatio	ns			
5.1.6	Bill of materials				
5.1.7	Installation standards/ Hook-up				
5.1.8	Manufacturers hand book wit wherever necessary for reference	h instructions for install e.	ation and calibratio		
6.0	DRAWINGS AND DOCUMENTS	S TO BE PROVIDED BY	CONTRACTOR		
6.1.1	The drawings for materials that namely local control panel, juncti	are included on their supp on boxes and local cabine	oly and erection scop ts.		
6.1.2	The detailed engineering drawing	g wherever such drawing is	s assigned.		
6.1.3	Two sets of layout drawings, standards bill of materials cable schedules etc., duly incorporating the changes/modification carried out during the course of execution of works.				
6.1.4	Final material appropriation sta shortages of any in the proforma	atement for all free issue duly approved by Enginee	e materials indicatir er-in-Charge.		
7.1.0	SCRAP AND EXCESS MATER	AL			
7.1.1	Every month, the contractor sha by the Owner in the standard Engineer- in-Charge.	ll submit an account for a proforma prescribed for	Il the materials issue this purpose by th		
7.1.2	On completion of the work, the statements for all materials issu the Engineer-in-Charge.	contractor shall submit `l ed by the Owner in the p	Material appropriatic roforma prescribed l		
7.1.3	The following scrap allowances a	are permissible.			
	Length below Non-sa 0.5 mm	alvageable	Unaccountable		
	Steel pipes, SS       2%       0.5%         Tubes single pair/       Twocore / Three       Core cables.				

MECON LIMITED	STANDARD SPEC			
REGD. OFF: RANCHI 834002	ELECTRICAL & INSTR	ELECTRICAL & INSTRUMENTATION		
	OIL & GAS SBU	J, DELHI	"001:2000 Col	
	SPECIFICATION FOR	DOCUMENT NO.	Page 17 of 17	
TITLE	INSTRUMENTATION	MEC/S/05/26/08	REVISION: 0	
			EDITION: 1	
	length below Non-sa 20 m	gth below Non-salvageable		
	Multitube, 2º Multicables	2%		

#### 8.0 SPECIAL INSTRUCTIONS TO CONTRACTOR:

- 8.1.4 All excess materials and scrap shall be returned after duly accounting for, to the storage points designated by the Owner. Where materials are to be weighed before return, the contractor shall be responsible for making the necessary section obtained during the course of construction for fabricating temporary supports or other items, without prior permission of the Engineer in -Charge.
- 8.1.5 If the contractor fails to return the surplus material as aforesaid, the owner will charge the contractor for such unreturned materials at panel rates, which will deducted from whatever amount is due to the contractor. In case any material issued by the Owner deteriorates during storage by the contractor, new materials will be issued to him at penal rates, but the delay in procuring such materials will be at the contractor's account only.

\*\*\*\*\*

# STANDARD SPECIFICATION FOR CABLING

### SPECIFICATION NO.: MEC/S/05/E5/021



ELECTRICAL & INSTRUMENTATION (OIL & GAS SBU) MECON LIMITED DELHI 110 092

Page 619 of 644

MECON LIMITED	STANDARD SPECI			
REGD. OFF: RANCHI 834002	ELECTRICAL & INSTR OIL & GAS SBU,	ELECTRICAL & INSTRUMENTATION OIL & GAS SBU, DELHI		
		DOCUMENT NO.	Page 1 of 13	
TITLE	CABLING	MEC/S/05/E5/021	REVISION: 0	
			EDITION: 1	

### **AMENDMENT STATUS**

SI. No.	Clause / Paragraph / Annexure / Exhibit / Drawing Amended	Page No.	Revision	Date	By (Name)	Verified (Name)

MECON LIMITED	STANDARD SPECI	FICATION	
REGD. OFF: RANCHI 834002	ELECTRICAL & INSTR	ELECTRICAL & INSTRUMENTATION	
	OIL & GAS SBU,	DELHI DOCUMENT NO.	Page 2 of 13
TITLE	CABLING	MEC/S/05/E5/021	REVISION : 0
			EDITION: 1
	<u>C O N <sup>-</sup></u>	<u> </u>	
1.0	SCOPE		
2.0	STANDARD		
3.0	CABLE SPECIFICATIONS		
4.0	MISCELLANEOUS MATERIALS	SPECIFICATIONS	
5.0	CABLE LAYING		
6.0	TERMINATION		
7.0	TESTING		
PREPARED BY	: CHECKED BY:	APPROVED BY	: ISSUE DATE :
(R. SANJAY BAB	U ) (RAKESH SHUKLA)	(PANKAJ SRIVAST	AVA) 08 <sup>Th</sup> DEC 08
			Page 621 of 644

MECON LIMITED	STANDARD SPEC				
REGD. OFF: RANCHI 834002	ELECTRICAL & INSTR OIL & GAS SBU	ELECTRICAL & INSTRUMENTATION OIL & GAS SBU, DELHI			
		DOCUMENT NO.	Page 3 of 13		
TITLE	CABLING	MEC/S/05/E5/021	REVISION: 0		
			EDITION: 1		

#### 1.0 SCOPE

This is to define the requirements for supply, wherever applicable, the installation, testing and commissioning of the cabling system.

#### 2.0 STANDARDS

The work shall be carried out in the best workman like manner in conformity with this specification, the relevant specifications, codes of practice of Indian Standards Institution, approved drawings and instructions of Engineer-in-Charge or his authorized representative issued from time to time. In case of any conflict between the standards, the instruction of Engineer-in-Charge shall be binding.

#### 3.0 CABLE SPECIFICATIONS

#### 3.1 <u>Power Cables</u>

Power cables for use on 415 V systems shall be of 1100 Volts grade, aluminium stranded conductor, PVC insulated, PVC sheathed, armoured and overall PVC sheathed. Power cables for 3.3 KV 6.6 KV and 11 KV system shall be aluminium conductor, XLPL insulated, screened, PVC bedded galvanized steel flat armoured and PVC sheathed cable. All L.T. Cables conform to standard specification and relevant sections of IS: 1554 Part-I and H.T. Cables shall conform to IS: 7098 (Part II). Unarmoured cables will be used wherever specified on the cable schedule.

#### 3.2 <u>Control Cables:</u>

Control cables shall be 1100 Volt Grade, 2.5 mm<sup>2</sup> copper conductor PVC insulated PVC sheathed, single wire armoured with an overall PVC sheath, as per IS: 1554 Pt. Unarmoured cables shall be used wherever specified on the cable schedule.

#### 3.3 <u>Communication cables:</u>

Communication cables shall comprise 1 pair unarmoured, 2-pair, 5-pair and multipair armoured cables of sizes as specified in the cable schedule. Minimum conductor size shall be 0.5 mm telephone system and 0.71 for plant communication system.

MECON LIMITED	STANDARD SPECIFICATION				
REGD. OFF: RANCHI 834002	ELECTRICAL & INSTR OIL & GAS SBU	ELECTRICAL & INSTRUMENTATION OIL & GAS SBU, DELHI			
		DOCUMENT NO.	Page 4 of 13		
TITLE	CABLING	MEC/S/05/E5/021	REVISION: 0		
			EDITION: 1		

#### 4.0 MISCELLANEOUS MATERIALS SPECIFICATIONS

#### 4.1 <u>Connectors:</u>

Cable terminations shall be made with aluminium / tinned copper crimped type solder less lugs of M/s. Dowell's make or approved equivalent for all aluminium conductors and stud type terminals.

#### 4.2 <u>Cable Identification</u>

Cable tags shall be of 2 mm thick, 20 mm wide aluminium strap of suitable length to contain cable number, equipment no., etc.

#### 4.3 <u>Ferrules</u>

Ferrules shall be of approved type size to suit core size mentioned and shall be employed to designate the various cores of control cable by the terminal numbers to which the cores are connected for case in identification and maintenance.

#### 4.4 <u>Cable Glands</u>:

Cable glands to be supplied shall be nickel-plated Brass double compression type of approved/ reputed make. Glands for classified hazardous areas shall be certified by CMRS.

#### 4.5 <u>Cable Trays</u>:

This shall be either prefabricated hot dip galvanized sheet steel trays or site fabricated angle iron trays as specified elsewhere. Prefabricated hot dip galvanized sheet steel cable trays shall be used for maximum support span of 2000 mm unless design is approved for larger span. For requirements of larger than 750 mm width two trays shall be run side by side. Cable trays shall be suitable for a cable weight of 50 kg/meter running length of tray. Minimum thickness of sheet steel/galvanizing shall be 2mm/86 microns respectively.

Cable trays fabricated from standard rolled sections shall use 50x50x6 /ISMC 100 Sections for runners for supporting spans limited to 2000 mm/more than 2000 mm respectively. Cross support shall be 32 x 6 mm flat/ 25x25x6 angle for width upto 500 mm/ more than 500 mm respectively.

Vertical supports for both the above type of trays shall be fabricated out of ISMC 100 and horizontal supports with 75 x 50 x 6 angle iron/ ISMC 75 as approved by Engineer-in-Charge.

If unit rate is not included in schedule of rates, then cable trays if required, shall be fabricated and installed at site as per tone rate for electrical structural supports etc. Page 623 of 644

MECON LIMITED	STANDARD SPECIFICATION ELECTRICAL & INSTRUMENTATION OIL & GAS SBU, DELHI		
REGD. OFF: RANCHI 834002			80 मेकान 9001:2000 Control
		DOCUMENT NO.	Page 5 of 13
TITLE	CABLING	MEC/S/05/E5/021	<b>REVISION: 0</b>
			EDITION: 1

#### 5.0 CABLE LAYING

- 5.1 Cable network shall include power, control, lighting and communication cables, which shall be laid in trenches, cable trays or conduits as detailed in the relevant drawings and cable schedules. Erection of cable trays as required shall be checked after erection and marked in as built drawings. Cable routing given on the layout drawings shall be checked in the field to avoid interference with structures, heat sources, drains, piping, air-conditioning duct etc. and minor adjustments shall be done to suit the field conditions wherever deemed necessary without any extra cost.
- 5.2 High voltage, medium voltage and other control cables shall be separated from each other by adequate spacing or running through independent pipes, trenches or cables trays, as applicable.

All communication cables (telephones, P.A.S.) RTD Cables shall run on instrument trays/ducts/trenches. Wherever these are not available, cables shall be taken in a separate trench with a minimum clearance of 300 mm away from electrical trench as per the direction of Engineer-in-Charge and Communication cables shall cross power cables at right angles.

All cable routes shall be carefully measured and cables cut to the required lengths, leaving sufficient lengths for the final connection of the cable to the terminal of the equipment. The various cable lengths cut from the cable reels shall be carefully selected to prevent undue wastage of cables. The quantity indicated in the cable schedule is only approximate. The contractor shall ascertain the exact requirement of cable for a particular feeder by measuring at site and avoiding interference with structure, foundation, pipelines or any other works. Before the start of cable lying, cable drum schedule; shall be prepared be electrician contractor and get that approved by Engineer-in-Charge to minimize/avoid straight through joints required. Contractor shall work out the actual number of straight through joints required.

- 5.4 Cables as far as possible shall be laid in complete, uncut lengths from one termination to the other.
- 5.5 Cables shall be neatly arranged in the trenches/trays in such a manner so that criss-crossing is avoided and final take off to the motor/switchgear is facilitated. Arrangement of cables within the trenches/trays shall be the responsibility of the Contractor. Cable routing between lined cable trench and equipment/motors shall be taken through GI pipe sleeves of adequate size. Pipe sleeves shall be laid at an angle of maximum 45° to the trench wall. In case of larger dia cables, i.e., 50 mm and above, adequately sized pipe with larger bend radius shall be provided for ease of drawing of cable or for replacement. In places where it is not possible, a smaller trench may be provided if approved by Engineer-in-Charge. Page 624 of 644

MECON LIMITED	STANDARD SPECIFICATION		
REGD. OFF: RANCHI 834002	ELECTRICAL & INST	ELECTRICAL & INSTRUMENTATION	
	OIL & GAS SBU	J, DELHI	- 1.2000 000
		DOCUMENT NO.	Page 6 of 13
TITLE	CABLING	MEC/S/05/E5/021	<b>REVISION: 0</b>
			EDITION: 1

5.6 All cables will be identified close to their termination point by cable numbers as per cable schedule. Cable numbers will be punched on aluminium straps (2 mm thick) securely fastened to the cable and wrapped around it. Alternatively cable tags shall be circular in construction to which cable numbers can be conveniently punched.

Each underground cable shall be provided with identity tags of lead securely fastened every 30 m of its underground length with at least one tag at each end before the cable enters the ground. In unpaved areas, cable trenches shall be identified by means of markers as per standard drawing. These posts shall be placed at location of changes in the direction of cables and at intervals of not more than 30 M and at cable joint locations.

- 5.7 All temporary ends of cables must be protected against dirt and moisture to prevent damage to the insulation. For this purpose, ends of all PVC insulated cables shall be taped with an approved PVC or rubber insulating tape. Use of friction type or other fabric type tape is not permitted. Lead sheathed cables shall be plumbed with lead alloy.
- 5.8 RCC cable trenches shall be with removable covers. Cables shall be laid in 3 or 4 tiers in these trenches as indicated on the sectional drawings. Concrete cable trenches shall be filled with sand where specified to avoid accumulation of hazardous gases, RCC covers of trenches in process area shall be effectively sealed to avoid ingress of chemicals etc. The electrical Contractor at no extra cost shall do removal of concrete covers for purpose of cable laying and reinstating them in their proper positions after the cables are laid.

Cables shall be handled carefully during installation to prevent mechanical injury to the cables. Ends of cables leaving trenches shall be coiled and provided with a protective pipe or cover, until such times the final termination to the equipment is connected.

5.9 Directly buried cables shall be laid underground in excavated cable trenches where specified in layout drawings. Trenches shall be of sufficient depth and width for accommodation of all cables correctly spaced and arranged with a view of heat dissipation and economy of design.

Minimum depth of buried cable trench shall be 750 mm for low voltage and 900 mm for H.V. Cables, the depth and the width of the trench shall vary depending upon the number of layers of cables.

Cables shall be laid in trenches at depth as shown in the drawing. Before cables are placed, the trenches bottom shall be filled with a layer of sand. This sand shall be levelled and cables laid over it. These cables shall be shall be shall be approximately with  $\frac{1}{2}$ 

MECON LIMITED	STANDARD SPECIFICATION				
REGD. OFF: RANCHI 834002	ELECTRICAL & INSTR	ELECTRICAL & INSTRUMENTATION			
	OIL & GAS SBU	DOCUMENT NO.	Page 7 of 13		
TITLE	CABLING	MEC/S/05/E5/021	REVISION : 0		
		-, -, -, -, -, -	EDITION : 1		
	mm of sand on top of the larg pressed. A protective covering of be laid flat. The remainder of rammed and levelled.	gest diameter cable and f 75 mm thick second clas the trench shall then be	sand shall be lightly s red bricks shall then e back-filled with soil,		
5.10	As each row of cables is laid in p shall be given an insulation test Any cable, which proves defect cables is laid.	blace and before covering in the presence of Engine ive, shall be replaced bef	with sand every cable er-in-Charge / Owner. Fore the next group of		
	All wall openings / pipe sleeves cables to avoid seepage of water	s shall be effectively seal inside building/-lined trend	ed after installation of ch.		
t 3	Where cables rise from trenches to motor, control station, lighting panels etc., they shall be taken in G.I. Pipes for mechanical protection upto a minimum of 300 mm above finished ground level.				
	Cable ends shall be carefully pulled through the conduits, to prevent damage to the cable. Where required, approved cable lubricant shall be used for this purpose. Where cable enters conduit the cable should be bent in large radius. Radius shall not be less than the recommended bending radius of the cables specified by the manufacturer.				
	Following grade of the pipe fill sh	all be used for sizing the p	ipe size:		
	<ul> <li>a) 1 cable in pipe - 53% full</li> <li>b) 2 cables in pipe - 31% full</li> <li>c) 3 or more cables - 43% full</li> <li>d) Multiple cables - 40% full</li> </ul>				
	After the cables are installed a grade shall be plugged with a st for sealing purpose. Alternativel sealing purposes. The cost for the installation of G.I. Pipe and	nd all testing is complete uitable weatherproof plast y G.I. Lidsor PVC bushes ne same shall be deemed I no separate payment sha	e, conduit ends above ic compound/ `PUTTI' shall be employed for to have been included all be allowed.		
5.11	Where cables pass through four the necessary ducts or opening However, should it become new structures, the electrical contra approval of the Engineer-in-Char	ndation walls or other ur gs will be provided in ac cessary to cut holes in e ctor shall determine thei ge before cutting is done.	derground structures, dvance for the same. xisting foundations or r location and obtain		

5.12 At road crossing and other places where cables enter pipe sleeves adequate bed of sand shall be given so that the cables do not slack and get damaged by pipe ends.

MECON LIMITED	STANDARD SPECIFICATION		
REGD. OFF: RANCHI 834002	ELECTRICAL & INSTRUMENTATION		88 मेकान 9001:2000 Control
		DOCUMENT NO.	Page 8 of 13
TITLE	CABLING	MEC/S/05/E5/021	REVISION : 0
			EDITION: 1
5.13	Drum number of each cable fror cable number in the cable sched	n which it is taken shall be ule.	e recorded against the
5.14	Cables installed above grade sh structures and shall be run paral	all be run in trays, expose lel or at right angles to bea	ed on walls, ceilings or ims, walls or columns.
h	Cables shall be so routed that t ot piping or vessels.	hey will not be subjected	to heat from adjacent
5.15	Individual cables or small group clamped by means of 10 SWG 0 saddle and saddle bars shall be of cables and no separate paym small group of cables can be tak They shall be rightly supported of groups as required, if drilling of secured and steel must be drilled will result.	os which run along struct GI saddles on 25x6 mm sa deemed to have been incl ent shall be made on this en through 100 mm slotted on structural steel and ma f steel must be resorted d where the minimum wea	ures/walls etc. will be ddle bars. The cost of uded in the installation account. Alternatively d channel/ISMC 100. asonry, individual or in to, approval must be kening of the structure
	Cables shall be supported so distance between supports shall mm diameter and maximum 450	as to prevent unsightly be approximately 300 m mm for cables larger than	sagging. In general m for cables up to 25 25 mm dia.
5.16	All G.I. Pipes shall be laid as pe fabrication of various profiles of (which is to be arranged by the removed. GI Pipes with bends the bends shall be totally cond coating shall be applied on the be undertaken well before paving paving agency shall be the response of pipes shall be suitably plugg position. The Contractor at no en	r layout drawings and site pipe by hydraulically oper contractor), all the burrs fi shall be buried in soil/con cealed. For G.I. Pipes b buried lengths. Installation is completed and necess onsibility of Electrical Contr ged with G.I. Plugs after xtra cost shall supply G.I. I	requirements. Before ated bending machine rom the pipes shall be crete in such way that buried in soil, bitumen of G.I. Pipes shall be eary co-ordination with ractor. The open ends they are laid in final Plugs.
5.17	Cable laid on supporting angle vertical run of cable trays sl Saddles/Clamps, whereas cable means of nylon cords.	e in cable trenches, stru hall be suitably clampe in horizontal run of cable	uctures, columns and d by means of G.I. e trays shall be tied by
5.18	Supporting steel shall be painted done with one coat of red lead aluminium paint unless otherwise	d before laying of cables. d paint and two coats of e specified.	The painting shall be approved bituminous

### 6.0 TERMINATION

Page 627 of 644

MECON LIMITED	STANDARD SPECIFICATION		
REGD. OFF: RANCHI 834002	ELECTRICAL & INSTR OIL & GAS SBL	RUMENTATION J, DELHI	मिकान मेकान 3007:2000 Contribut
		DOCUMENT NO.	Page 9 of 13
TITLE	CABLING	MEC/S/05/E5/021	REVISION: 0
			EDITION: 1
6.1	All PVC cables up to 1.1 KV g means of double compression in nipple with conduit electrical three	rade shall be terminated type cable glands. They ads and check nut.	at the equipments by shall have a screwed
r	All Cable entries shall be throu made only after getting approval	igh bottom only and top of Engineer-in-Charge.	entry terminations are
6.2	Power cables wherever colour c yellow and blue PVC tapes. W necessary bimetallic washers additional red ferrules shall be u at the termination points in the S	oding is not available shall here copper to aluminium shall be used. For tri sed only in the particular o witchgear/Control panels a	be identified with red, connections is made, circuit identification cores of control cables and Control Switches.
6.3	In case of control cables all cables shall be identified at both ends by their terminal numbers by means of PVC ferrules or Self-sticking cable markers. Wire numbers shall be as per schematic/ wiring /inter- connection diagram. Bidders shall have the samples of PVC ferrules/cable markers approved before starting the work. All unused spare cores of control cables shall be neatly bunched and ferruled with cable tag at both ends.		
6.4	Where threaded cable gland is screwed into threaded opening of different size, suitable galvanized threaded reducing bushing shall be used of approved type, at no extra cost. All switchgear and control panels shall have undrilled gland plate.		
	Contractor shall drill holes for fix Gland plate shall be of non-mag core cables.	ing glands wherever nece gnetic material/aluminium s	ssary at no extra cost. sheet in case of single
6.5	The cable shall be taken through equipment such as motors. T taken along the cable ways (if polyethylene straps. Only contro be directly terminated on to the t	h glands inside the panels he individual cores shall provided) or shall be fix I cables of single strand a erminals.	or any other electrical then be dressed and ed to the panels with nd lighting cables may
	In case of termination of cables having no access from the botto plate for all the cables in one lin along the centre line of holes. should be sealed with cold settin open armouring to connect it to e	s at the bottom of a pane om close fit hole should b le, then bottom plate shou After installation of botto ng compound. Cables sha earth bus.	el over a cable trench e drilled in the bottom ld be split in two parts m plate and cables it ll be clamped over the
6.6	Cable leads shall be terminate crimped type solder less conne	ed at the equipment ter ector as manufactured by	minals, by means of y M/s. Dowell Electro Page 628 of 644

MECON LIMITED	STANDARD SPECIFICATION		
REGD. OFF: RANCHI 834002	ELECTRICAL & INSTR OIL & GAS SBU	मुठ मकान मुठ मकान मेठा:2000 Contract	
		DOCUMENT NO.	Page 10 of 13
TITLE	CABLING	MEC/S/05/E5/021	REVISION: 0
			EDITION: 1

works or approved equivalent.

Crimping shall be done by hand crimping hydraulically operated tool and conducting jelly shall be applied on the conductor. Insulation of the leads should be removed immediately before the crimping. Conductor surface shall be cleaned and shall not be left open.

- 6.7 <u>Cable accessories for H.V. Systems</u>
- 6.7.1 The 11, skilled and experienced jointers duly approved by the Engineer-in-Charge shall do 6.6 and 3.3 KV cables terminations joints. Termination including supplying of jointing kit shall be threaded in Contractor scope unless specified otherwise.
- 6.7.2 The termination and straight thro' joint kit. For use on high voltage system shall be suitable for the type of cables red by the contractor or the type of cables issued by owner for installation. The materials required for termination and straight through joints shall be supplied in kit form. The kit shall include all insulating and sealing materials apart from conductor fitting and consumables items. An installation instruction shall be included in each sheet.
- 6.7.3 The termination kits shall be suitable for termination of the cables to indoor switchgear or to a weatherproof cable box of an outdoor mounted transformer motor. The terminating kits shall preferably be of the following types:
  - a) TAPLEX' of M-seal make using non-linear resistance material fortress grading.
  - b) `PUSH-ON' type of CCI make using factory moulded silicone rubber insulators.
  - c) `TROPOLINK' type of CCI makes.
  - d) Heat-shrinkable sleeve type of M/s. Raychem.

For outdoor installations, weather shields/sealing ends and any other accessories required shall also form part of the kit.

MECON LIMITED	STANDARD SPECIFICATION				
REGD. OFF: RANCHI 834002	ELECTRICAL & INSTR	<b>अ</b> नेकान गुरु			
	OIL & GAS SBU	J, DELHI	507:2000 COV		
		DOCUMENT NO.	Page 11 of 13		
TITLE	CABLING	MEC/S/05/E5/021	REVISION : 0		
6.7.5	<ul> <li>EDITION : 1</li> <li>with uncontrolled backfill and possibility of flooding by water. The jointing kit shall be one of the following types.</li> <li>a) `TAPLEX' of M-seal make</li> <li>b) `TROPOLINK' type of CCI make</li> <li>c) Heat-shrinkable sleeve type of M/s. Raychem.</li> <li>6.7.5 Makes of kits other than those specified in 6.7.3 and 6.7.4 may be considered provided the Contractor furnishes type test certificates, along with the offer.</li> </ul>				
6.7.6	<ul> <li>lype tests are to be carried out qualities and design of a given tests shall include the following to VDE 0278 specifications. The system considered shall submit to a) A.C. Voltage withstand dr</li> <li>a) A.C. Voltage withstand dr</li> <li>b) Partial discharge test - Disconstructed out and the system considered shall submit to a) A.C. high voltage test - Disconstructed out and the system constructed out and the system considered shall submit to a) A.C. high voltage test - Disconstructed out and the system constructed out and the system considered shall submit to a) A.C. high voltage test - Disconstructed out and the system constructed li></ul>	It at manufacturer's works type of termination/jointinests conforming to the late Contractor along with the Contractor along with the che type test certificates. y test for 1 minute scharge magnitude small to scharge magnitude small to test with 10 impulses of e following load cycling of 250°C for 1 second. t for 30 minutes.	to prove the general ng system. The type est IEC 502.2, 466 and e offer for the jointing be less than 20 p.c. each polarity. test with conductor		

MECON LIMITED	STANDARD SPECIFICATION ELECTRICAL & INSTRUMENTATION			
REGD. OFF: RANCHI 834002			मेकान	
	OIL & GAS SBU	3001:2000 Corne		
		DOCUMENT NO.	Page 12 of 13	
TITLE	CABLING	MEC/S/05/E5/021	REVISION: 0	
			EDITION: 1	
7.1	Before energizing, the insulation from phase to phase and from pl	n resistance of every circ hase to ground.	uit shall be measured	
7.2	Where splices or termination a measure insulation resistance of terminating. Repeat measure completed.	are required in circuits ra of each length of cable b ement after splices and	ted above 600 volts, efore splicing and or/ d/or terminations are	
7.3	Measure the insulation resistant trenches are back-filled. Repeat	ce of directly buried cable measurement after back-	e circuits before cable filling.	
	For cables up to 1.1 KV grade, 1KV Megger and for H.V. Cables 2.5 KV/5 KV, 2.4 KV/4.9 KV Megger shall be used			
7.4	D.C. High Voltage Test shall be conducted after installation on the following and test results are recorded.			
	a) All 1000volts grade cables in which straight through joints have been made.			
	b) All cables above 1100 V grade.			
	For record purposes test data shall include the measure values of leakage current versus time.			
	The D.C. High Voltage test shall be performed as detailed below in the presence of the Engineer-in- Charge or his authorized representative only.			
	Cables shall be installed in final position with the entire straight through joints complete. Terminations shall be kept unfinished so that motors, switchgears, transformers etc. are not subjected to test voltage.			
	The test voltage shall be as under: -			
	i) For cables 3.3 KV grade	5.4 KV DC		
	ii) For cables 6.6 KV grade	10.8 KV DC		
	iii) For cables 11 KV grade	18 KV DC		

MECON LIMITED	STANDARD SPECIFICATION ELECTRICAL & INSTRUMENTATION OIL & GAS SBU, DELHI		
REGD. OFF: RANCHI 834002			10 मेकान 9001:2000 Confund
		DOCUMENT NO.	Page 13 of 13
TITLE	CABLING	MEC/S/05/E5/021	REVISION: 0
			EDITION: 1

engineer.

7.6 Cable schedule and layout drawings must be marked for AS BUILT conditions during the installation work and shall be approved by Site Engineer.

### SPECIFICATION FOR EARTHING AND LIGHTNING PROTECTION

## SPECIFICATION NO. MEC/S/05/26/23A



## (ELECTRICAL SECTION) MECON LIMITED DELHI 110 092

C:\Documents and Settings\Administrator\Desktop\shared\MEETERING SKID PART-C\Standard. specification\Standard Inst. & electrical\spec for earthing and lightning protection Final 23 A.doc

				<b>]</b>
MECON LIMITED REGD. OFF: RAN (JHARKHAND)	) NCHI	PROCESS & PIPING DESIGN SECTION NEW DELHI	STANDARD SPECIFICATION	Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin Patin
TITLE	EARTHIN PROTECT	G AND LIGHTING	SPECIFICATION NO.	PAGE 1 OF 10
			MEC/S/05/26/23A	REVISION 1
		CONTEN	TS	
1.0	SCOPE			
2.0	STANDARDS			
3.0	EARTHING CONDUCTOR/ELECTRODE			
4.0	EARTHING NET WORK			
5.0	INSTALLATION OF EARTH ELECTRODE			
6.0	CONNECTION			
7.0	TESTING			
8.0	TEST PROFORMA			

MECON LIMITED REGD. OFF: RANCHI (JHARKHAND)		PROCESS & PIPING DESIGN SECTION NEW DELHI	STANDARD SPECIFICATION	Anti- Segur Castle
TITLE	EARTHIN PROTECT	G AND LIGHTING TON	SPECIFICATION NO.	PAGE 2 OF 10
			MEC/S/05/26/23A	REVISION 1

#### 1.0 <u>SCOPE</u>:

The intent of this specification is to define the requirements for the supply, installation, testing, and commissioning of the Earthing System.

#### 2.0 <u>STANDARDS</u>:

The work shall be carried out in the best workmanlike manner in conformity with this specification, the relevant specifications/codes of practice of Indian Standard Institution, approved drawings and instructions of the Engineer-in-Charge or his authorized representative issued from time to time. In case of any conflict between the standards, the instructions of Engineer-in-Charge shall be binding.

#### 3.0 <u>CONDUCTOR ELECTRODE</u>:

The main grid conductor shall be hot dip galvanized G.I. Flat or PVC insulated aluminum conductor/copper conductor. Sizes for main conductors shall be marked on the drawings. Thickness of hot dip galvanizing shall not be less than 75 microns.

#### 4.0 <u>EARTHING NETWORK</u>:

4.1 The earthing installation shall be done in accordance with the earthing drawings, specifications and the standard drawings of reference attached with this document. The entire earthing system shall fully comply with the Indian Electricity Act and Rules framed thereunder. The contractor shall carry out any changes desired by the Electrical Inspector or the owner, in order to make the installation conform to the Indian Electricity Rules at no extra cost. The exact location on the determined equipment shall be in field, in consultation with the Engineer-in-Charge or his authorized representative. Any changes in the methods, routing, size of conductors etc. shall be subject to approval of the Owner/Engineer-in-Charge before execution.

Excavation and refilling of earth, necessary for laying underground earth bus loops shall be the responsibility of the contractor.

4.3 The earth loop impedance to any point in the electrical system shall have a value which will ensure satisfactory operation of protective devices.

MECON LIMITED REGD. OFF: RAN (JHARKHAND)	<b>)</b> NCHI	PROCESS & PIPING DESIGN SECTION NEW DELHI	STANDARD SPECIFICATION	AND CONTRACT
TITLE	EARTH	ING AND LIGHTING PROTECTION	SPECIFICATION NO.	PAGE 3 OF 10
			MEC/S/05/26/23A	REVISION 1
4.4	The main Wherever trenches earthing s	earth loop shall be laid at cable trenches are availa and shall be firmly cleared to trip shall be protected agains	a depth of 500 mm ble, the earth lead sh the walls of concrete li st mechanical damage.	below grade level. nall be laid in the ned trenches. The
4.5	In process specified in electrically	unit areas, the earthing cab the layout drawings. The e bonded to the cable tray at r	le shall be run along ca earthing cable shall be s egular intervals.	able trays wherever suitably cleated and
4.6	Joints and tapping in the main earth loop shall be made in such a way that reliable and good electrical connections are permanently ensured. All joints below grade shall be welded and suitably protected by giving two coats of bitumen and covering with Hessian tape. All joints above ground shall be by means of connectors/lugs as far as practicable. Tee connectors shall be used for tapping, earth leads from the main earth loop wherever it is installed above ground. Earthing plates shall be provided for earthing of two or more equipmen at a place from earth grid. Where aluminum cable risers are to be connected to the underground GI earth bus, the aluminum cable riser shall be taken to the nearest earth pit and terminated through a bolted joint. If this is not practicable then a G.I. risers shall be brought above grade and a bolted joint shall be made between this GI riser and the aluminum cable termination. This G.I. Riser shall be protected applying two coats of bituminous paint/bitumen on the exposed particable			n such a way that nsured. All joints iving two coats of ground shall be by rs shall be used for is installed above or more equipment to be connected to all be taken to the s is not practicable, joint shall be made this G.I. Riser shall n on the exposed
4.7	Conduits i earthed.	n which cables have been Cable arm ours shall be earth	installed, shall be effec ed at both ends.	ctively bonded and
5.0	EARTH EI	<u>ECTRODES</u> :		
5.1	Earth pipe and in acc location sh	electrodes shall be installed ordance with the standard of all be marked to enable accu	as shown in the earthi drawings of reference a urate location by permar	ng layout drawings and IS:3043. Their nent markers.
5.2	All earth electrodes shall preferably be driven to sufficient depth to reach permanently moist soil. Electrodes shall preferably be situated in a soil which has a fine texture and which is packed by watering and ramming as tightly as possible. Wherever practicable, the soil shall be dug up, all lumps broken and stones removed from the immediate vicinity of the electrodes.			

MECON LIMITED REGD. OFF: RANCHI (JHARKHAND)			PROCESS & PIPING DESIGN SECTION NEW DELHI	STANDARD SPECIFICATION	Andra Begy Control		
TITLE		EARTHIN	G AND LIGHTING PROTECTION	SPECIFICATION NO.	PAGE 4 OF 10		
				MEC/S/05/26/23A	REVISION 1		
5.3 All earth ele earth test n protracted d parallel to r shall not be			ectrodes shall be tested for leter. The tests shall take ry spell. If necessary, a nur educe the earth resistance. ess than twice the length of	earth resistance by r place in dry months, nber of electrodes sha The distance betwe electrode.	neans of standard preferably after a all be connected in en two electrodes		
5.4	The or c	e electrod other mate	es shall have a clean surface erials of poor conductivity.	ce, not covered by pai	nt, enamel, grease		
5.5	The exact location and number of earth electrodes required at each location shall be determined in the field in consultation with the owner/Engineer-in- Charge, depending on the soil strata and resistively, to meet the ohmic values prescribed in clause 5.3. Earth Electrodes shall be located avoiding interference with road, building foundation, column etc. Individual earth electrode shall be provided for each lightning arrestor and lightning mast. The electrodes shall be so placed that all lightning protective earths may be brought to earth electrode by a short and straight a path as possible to minimize surge impedance.						
5.6	The ear and	The disconnect facility shall be provided for the individual earth pits to check their earth resistance periodically. All the earth electrodes shall be suitably numbered and this should be indicated in as built drawings.					
6.0	<u>CO</u>	NNECTIO	<u>NC</u> :				
6.1	All equ sev in t con	All electrical equipment is to be doubly earthed by connecting two points on equipment to a main earthing ring. The earthing ring will be connected via links to several earth electrodes. The earth grid formed shall be a closed loop as shown in the drawing with earth electrodes connected to the grid with double strip connection. The cable armour will be earthed through the cable glands.					
6.2	In hazardous areas all major process equipments shall be connected to the earthing ring by means of anti- loosening connections and all pipelines will be bonded and earthed on entering the battery limit of the process area.						
6.3	The following shall be earthed.						
	1.	Trans	former neutrals, CT/PT neut	rals.			
	2.	2. Neutral Grounding Resistors.					

MECON LIMITED REGD. OFF: RANCHI (JHARKHAND)		PROCESS & PIPING DESIGN SECTION NEW DELHI	STANDARD SPECIFICATION	Antir BEDI Estim		
TITLE	EARTHING AND LIGHTING PROTECTION		SPECIFICATION NO.	PAGE 5 OF 10		
			MEC/S/05/26/23A	REVISION 1		
3.	Trans	former Housing.				
4.	4. Lightning Arrestors.					
5.	5. All switchgear and their earth buses, bus duct.					
6.	Motor	Frames.				
7.	7. Non-current carrying metallic parts of electrical equipment such as switchgear, switch racks, panel boards, motor control centers, lighting, power and instrument panels, push button stations, cable trays, pipes, conduits, terminal boxes, etc.					
8.	8. All fences, gates/enclosures, housing electrical equipment					
9.	9. All steel structures, rails etc. including bonding between sections.					
10.	10. Shield Wire					
11.	11. Structural steel and Columns.					
12.	Loadi	ng racks.				
13.	Lighti	ng Mast, poles.				
14.	Lighti	ng rods (Mast).				
15.	Tanks	s and vessels containing	g flammable materials.			
16.	16. Rotating parts of the agitators, pumps etc. through spring loaded brushes of suitable grade.					
17.	17. Earth continuity conductor shall be provided for flanges.					
Con Eart	Conductor size for connection to various equipments shall be as indicated on Earthing Layout Drawings.					
6.4 Two whice	Two distinct conductors directly connected to independent earth electrodes, which in turn, shall be connected to the earth too, shall earth system.					

MECON LIMITED REGD. OFF: RANCHI (JHARKHAND)		PROCESS & PIPING DESIGN SECTION NEW DELHI	STANDARD SPECIFICATION	A SECT CONT			
TITLE	EARTHING A	ND LIGHTING PROTECTION	SPECIFICATION NO.	PAGE 6 OF 10			
			MEC/S/05/26/23A	REVISION 1			
	The earth connection shall be properly made. A small flexible aluminum cable loops to bridge the top cover of the transformer and the tank shall be provided to avoid earth fault current passing through fastening bolts when there is a lightning surge, high voltage surge or failure of the bushings.						
6.5	Each Lightning Arrestor shall be connected to a separate electrode located as close as possible to it and within the fenced area for each set of arrestors. The three nos. electrodes for each set of arrestors shall be spaced about 5 meters apart so that they are all within the enclosing fence. Each of these electrodes shall be connected to the main earth grid.						
6.6	The shield wire shall be connected with the main grid solidly and not through supporting steel structures.						
6.7	All paint, scale and enamel shall be removed from the contact surface before the earthing connections are made.						
6.8	All earthing connections for equipment earthing shall be preferably from the earth plate mounted above ground. In case of G.I. Earth Loop all underground "T" connections shall be of the same size as main loop however in case of PVC insulated aluminum conductor loops underground joints shall be completely avoided. Connections to motors from earth plate or main loop conductor brought above ground shall not be less than following:						
	i) No.8 SWG G.I. Wire upto 3.7 KW motors.						
	ii) 3/8" DIA G.I. FINE WIRE ROPE for all motors above 3.7 KW upto 30 KW with tinned copper lug at both ends or 35 mm <sup>2</sup> PVC insulated stranded aluminum conductor with crimped lug.						
	iii) 5/8" DIA G.I. FINE WIRE ROPE OR 70 mm <sup>2</sup> PVC insulated aluminum stranded conductor for motors above 30 KW upto 75 KW terminated as described above.						
	iv) For all motors above 75 KW conductor size shall be same as that of loc conductor with equivalent size flexible, if required.						
	Anchor bolts or fixing bolts shall not be used for earthing connection.			ection.			

MECON LIMITED REGD. OFF: RANCHI (JHARKHAND)			PROCESS & PIPING DESIGN SECTION NEW DELHI	STANDARD SPECIFICATION	Registree Conversion	
TITLE EARTHIN			G AND LIGHTING PROTECTION	SPECIFICATION NO.	PAGE 7 OF 10	
				MEC/S/05/26/23A	REVISION 1	
6.9	6.9 All hardware used for earthing installations shall be hot dip galvanized or passivated. Spring washers shall be used for all earthing connection equipment.					
6.10	Ligl cab	nting fixtu le for this	rres shall be earthed throus purpose.	ugh the extra core prov	ided in the lighting	
7.0	<u>TE</u> Ear	<u>STING</u> : thing sys	tems/connections shall be	tested as follows:		
7.1	Resistance of individual electrodes shall be measured after disconnecting it from the grid.					
7.2	Earthing resistance of the grid shall be measured after connecting all the electrodes to the grid. The resistance between any point on the metallic earth grid and the general mass of earth shall not exceed 1 ohm.					
7.3	The	The resistance to earth shall be measured at the following:				
	a)	a) At each electrical system earth or system neutral earth.				
	b)	At ea	ch earth provided for struct	ture lightning protections	S.	
	<li>c) At one point on each earthing system used to earth electrical equi enclosures.</li>					
	d) At one point on each earthing system used to earth wiring sys enclosures such as metal conduits and cable sheaths or armor.					
	e) At one point on each fence enclosing electrical equipment.					
	Measurement shall be made before connection is made between the ground a the object to be grounded.				een the ground and	

MECON REGD. (JHARK	N LIMITED OFF: RANCHI (HAND)		PROCESS & PIPING DESIGN SECTION NEW DELHI	STANDARD SPECIFICATION		
TITLE		EARTHIN	G AND LIGHTING ION	SPECIFICATION NO.	PAGE 8 OF 10	
				MEC/S/05/26/23A	REVISION 1	
8.0	LIGHTNIN	IG PROT	ECTION			
	8.1	Lightning protection system shall generally comprise lightning finials (air terminals), roof conductors, down conductors, test links, and earth electrodes. the number, types, materials and sizes shall be in accordance with the drawings				
	8.2	Air terminals shall be mounted on top of buildings or structure as required. All air terminals shall be inter-connected with roof conductors,pipes,hands rails or any other metallic projection above the roofs shall also be bonded to the roof conductors.				
	8.3	Down conductors from air terminals or from roof conductors shall be routed as directly as possible to the test links on earth buses, with minimum bends				
	8.4	All provisions regarding connections of conductors for equipment earthing system shall also apply to lightning protection system.				
	8.5	In corrosive atmospheres, plumbing metal for corrosion protection shall cover lightning finials or air terminals.				

MECON LIMITED REGD. OFF: RANCHI (JHARKHAND)		PROCESS & PIPING DESIGN SECTION NEW DELHI	STANDARD SPECIFICATION	A SUCT CANTER	
TITLE EARTHIN PROTEC		EARTHIN	G AND LIGHTING TON	SPECIFICATION NO.	PAGE 9 OF 10
				MEC/S/05/26/23A	REVISION 1
9.0	<u>1</u> (INS	<u>EST PR</u>	<u>OFORMA</u> ION TESTING REPOR	RT EARTHING INSTALLAT	<u>10NS)</u>
	1.	<u>Earth</u>	system data		
		Туре	of electrode	:	
		Total	number of electrodes	:	
		Main	grid size	:	
		Mater	ial	:	
	2.	Gene	<u>ral checks</u> Put Tick $$ if	O.K.; otherwise give details	S.
		Const electr Stand	truction of earth odes as per lard.		
		Size o for va O.K. a	of earth conductor rious equipment as per Standard.		
		Minim betwe	num distance kept een two electrodes.		
		Clean of cor Inspe conne	liness and tightness nnectors. ct bolted & clamped ectors.		

3.	<u>TESTS</u>		
	3.1	Mea: of ea	sured earth resistance ach electrode in ohms
	No.	1	
		2	
		3	
		4	
		5	
	3.2	<u>Mea</u> (with	surement of earth grid resistance all electrodes connected to grid)
		a)	At each electrical system earth or system neutral earth.
		b)	At each point provided for structure lightning protection
		c)	At one point on each earthing systems used to earth electrical equipment enclosure
		d)	At one point on each earthing systems used to earth wiring systems

# C:\Documents and Settings\Administrator\Desktop\shared\MEETERING SKID PART-C\Standard. specification\Standard Inst. & electrical\spec for earthing and lightning protection Final 23 doc

such as metal conduits

At one point on each fence enclosing

electrical equipment.

etc.

e)

2

:

:

:

2

MECON LIMITED REGD. OFF: RANCHI (JHARKHAND)		PROCESS & PIPIN DESIGN SECTION NEW DELHI	G I	STANDARD SPI	ECIFICATION	A SUBT CONT
TITLE	EARTHING AMD LIGHTING PROTECTION			SPECIFICATION NO.		PAGE 10 OF 10
				MEC/S/05/26/23	A	REVISION 1
PROJEC	4. T:	Remarks:	UNI			
I TESTED	BY	 אדועע	NESS	SED BY	DATE	ĺ
    CONTRA 	CTOR	     MECO 	   N	OWNER	   	
1		I				I